#### **RESPONSE TO COMMENTS**

## MARTIN OPERATING PARTNERSHIP L.P. PERMIT #1227-AOP-R0 AFIN: 70-00039

On November 15, 2013 the Director of the Arkansas Department of Environmental Quality gave notice of a draft permitting decision for the above referenced facility. During the comment period, written comments on the draft permitting decision were submitted by HSG Environmental on behalf of the facility and Allen Chang on behalf of the USEPA Region 6. The Department's response to these issues follows.

Note: The following page numbers and condition numbers refer to the draft permit. These references have changed in the final permit based on changes made during the comment period and are referenced in parenthesis.

#### Comment #1:

<u>Draft Specific Condition No. 5:</u> For internal accounting purposes, Martin requests that the requirement to update records by the 15th day of the month be changed to the 30th day of the month.

This change is also requested for the following conditions: Draft Specific Condition Nos. 8, 13, 18, 23, 28, 35, 41, 47, 58, 60, 85, 87, 91, 94, 98, 105, 109, 114, 118, 124, 129, 133, 137, 141, 145, 158, 166, 174, 182, 191, 202, 220, and 224, and Draft Plantwide Condition Nos. 10 and 15.

## **Response to Comment #1:**

The changes have been made.

#### Comment #2:

<u>Draft Specific Condition Nos. 7 and 8:</u> Draft Specific Condition No. 7 limits the amount of crude oil the facility can process, and Draft Specific Condition No. 8 requires Martin to maintain records of the amount of crude oil processed. Martin requests that these conditions be moved to the Plantwide Conditions section of the permit, which apply to the facility as a whole, rather than listing them as specific conditions for one emission source; i.e., the Crude Charge Heater (SN-01).

#### Response to Comment #2:

The conditions have been moved to the Plantwide Conditions section and are listed as Plantwide Condition #7 & #8.

#### Comment #3:

<u>Vacuum Tower Charge Heater (SN-02)</u>: Draft Specific Condition No. 12 limits the amount of natural gas burned to 63 MMscf/yr. This is a limit that Martin is taking because Martin does not operate the heater at maximum capacity, and this limit is not based on maximum design capacity. Therefore, a specific condition needs to be added to this emission source that requires the facility to install a dedicated meter on the natural gas piping that feeds SN-02. Martin proposes the following condition be added to the permit.

The permittee shall install and maintain a dedicated meter on the natural gas piping which feed SN-02. [§19.703, A.C.A. §8-4-203 as reference by §8-4-304 and §8-4-311]

## **Response to Comment #3:**

The condition has been added to the permit as Specific Condition #12.

#### Comment #4:

<u>Draft Specific Condition No. 29 (SC#28):</u> As specified in §60.105(e)(ii), the H<sub>2</sub>S limit of 230 mg/dscm (0.10 gr/dscf) is based on a 3-hour rolling average. Please add the following wording to Specific Condition 29 for clarification purposes.

The permittee shall not burn any fuel gas that contains  $H_2S$  in excess of 230 mg/dscm (0.10 gr/dscf) on a 3-hour rolling average basis at SN-12. The permittee shall comply with the  $H_2S$  emission limit on or after the day on which the initial performance test, required by 60.8, is completed, but not later than 60 days after achieving maximum production rate at the facility, or 180 days after initial startup, whichever comes first. [§19.304 and 40 CFR §60.104(a)(1)]

#### **Response to Comment #4:**

The language has been added.

## Comment #5:

<u>Draft Specific Condition No. 30 (SC #29):</u> This condition requires Martin to "maintain records of the specific exemption chosen for each fuel gas stream". Martin will comply with NSPS Subpart J by continuously monitoring the  $H_2S$  concentration of the fuel gas as allowed by  $\S60.105(a)(3)$ , rather than using an exemption (i.e., burning only pipeline quality natural gas). Therefore, Specific Condition 30 should be removed from the permit.

## **Response to Comment #5:**

The condition has been removed and replaced with the following language:

"In compliance with §60.104, the permittee shall install, calibrate, maintain and operate a continuous monitoring system for  $H_2S$  in fuel gases before being burned at SN-12. [§19.304 and 40 CFR §60.105(a)(3)]"

#### Comment #6:

<u>Draft Specific Condition Nos. 33, 39, and 45:</u> These conditions require that Martin conduct daily opacity observations of loading rack sources. Martin requests that these conditions be removed from the permit. At these sources, a tanker truck or railcar is loaded with a liquid material through a hose connected directly to the truck or railcar. Absent a catastrophic event, such as a fire, there is no possibility for visible emissions. Requiring daily observations of these sources places an unnecessary compliance/administrative burden on the facility and results in no environmental benefit.

## **Response to Comment #6:**

The conditions have been removed.

#### Comment #7:

<u>Draft Specific Condition Nos. 52 and 53 (SC#48 and #49)</u>: These conditions contain a typo. The last specific condition listed for the compliance demonstration is Draft Specific Condition No. 0. The compliance demonstration reference should be Specific Condition No. 144 (as listed in the draft permit).

## **Response to Comment #7:**

The corrections have been made.

#### Comment #8:

<u>Draft Specific Condition No. 59:</u> This condition requires Martin to calculate NOx emissions for SN-25 by monitoring fuel inlet flow and applying a factor of 184.6 lb NOx/MMcf of natural gas. This condition should be removed for the following reasons.

- Draft Specific Condition No. 59 is an old condition that was originally listed in Permit 1227-AR-3, because Martin had taken a bubbled NOx limit of 53.0 ton/yr at three emission sources (SN-01, SN-25, and SN-26) to remain a minor source.
- With the issuance of this permit (1227AOP-R0), Martin will no longer be a minor source facility, and this source no longer has a bubbled NOx emission limit with other sources.
- The hourly emission limits for this source (SN-25) are based on maximum design fuel usage, and the yearly emission limits are based on maximum design fuel usage and operating the maximum number of hours a year.

## **Response to Comment #8:**

The condition has been removed from the permit.

#### Comment #9:

<u>Draft Specific Condition No. 60:</u> Draft Specific Condition No. 60 requires Martin to maintain records required by Draft Specific Condition No. 59. This condition should be removed along with Draft Specific Condition No. 59.

## **Response to Comment #9:**

The condition has been removed from the permit.

#### Comment #10:

<u>Draft Specific Condition No. 61:</u> This condition requires Martin to maintain a dedicated meter on the natural gas piping that feeds SN-25. The natural gas emission limit of 432 MMscf/yr listed in Specific Condition 57 is the maximum amount of fuel that can be burned at SN-25 based on maximum burner design and maximum number of operating hours in a year. Considering guidance on practical enforceability, a dedicated meter is not necessary, and this condition should be removed from the permit. This change is consistent with SN-03, SN-08 and SN-12, which also have natural gas limits (Draft Specific Condition Nos. 17, 22, and 27) listed in the permit that are based on maximum burner design and maximum number of operating hours in a year. Those sources are not required to have a dedicated meter to demonstrate compliance.

## **Response to Comment #10:**

The condition has been removed from the permit.

#### Comment #11:

<u>Draft Specific Condition No. 65 (SC#59)</u>: Add the regulatory reference §60.334(h)(3) to this condition, which is the NSPS Subpart GG reference that requires the cogeneration unit to only fire pipeline quality natural gas.

## **Response to Comment #11:**

The requested reference does not specifically require the cogeneration unit to only fire pipeline natural gas; therefore the citation has not been added to the condition.

#### Comment #12:

<u>Draft Specific Condition No. 69 (SC#63)</u>: The regulatory reference should be §60.335(b)(10) not §60.335(d). MOP also requests the following wording changes to be consistent with what is allowed by §60.335(b)(10).

Analysis for fuel sulfur content of the natural gas shall be conducted using one of the approved ASTM reference methods for the measurement of sulfur in gaseous fuels, or an approved alternative method. The approved reference methods are: ASTM D1072-80, 90 (reapproved 1994); ASTM D3031-81; ASTM D 3246-81, 92, 96; and ASTM D4084-82D4468-85 9(reapproved 2000); or D6667-01 as referenced in 40 CFR 60.335 (b) (210). [40 CFR Part 60, Subpart GG, §60.335 (d)]

## **Response to Comment #12:**

The revision has been made.

#### Comment #13:

<u>Draft Specific Condition No. 69 and 70 (SC#63 and #64):</u> These conditions require MOP to sample the natural gas for a period of time using an approved method to prove the fuel supplied is low in sulfur content. As allowed by §60.334(h)(3)(i), MOP would like the option to demonstrate the low sulfur fuel through a valid purchase contract, tariff sheet, or transportation contract for the gaseous fuel. MOP requests the following condition be added to the permit.

In lieu of complying with Specific Condition Nos. 69 and 70, the permittee may elect to not monitor the total sulfur content of the gaseous fuel combusted in the turbine if the gaseous fuel is demonstrated to be low sulfur fuel by obtaining a current, valid purchase contract, tariff sheet or transportation contract for the gaseous fuel, specifying that the maximum total sulfur content of the fuel is 20.0 grains/100 scf or less: [40 CFR Part 60, Subpart GG, §60.334(h)(3)]

## **Response to Comment #13:**

The condition has been added as Specific Condition #65.

#### Comment #14:

<u>Draft Specific Condition No. 71 (SC#66):</u> The regulatory reference should be §60.334(h)(4)(i)(3), not §60.335(d).

## **Response to Comment #14:**

The correction has been made.

#### Comment #15:

<u>Draft Specific Condition No. 76 (SC#71):</u> Martin requests the regulatory reference to be changed from §60.334 to §60.334(j)(1)(i) for clarification purposes.

## **Response to Comment #15:**

The change has been made.

#### Comment #16:

<u>Draft Specific Condition No. 78 (SC #58):</u> This condition requires Martin to maintain records for each period for which Draft Specific Condition No. 64 (SC #57) applies. Please move this condition to follow after Draft Specific Condition No. 64 (SC #57), so that those conditions are grouped together.

#### **Response to Comment #16:**

The condition has been repositioned to follow Specific Condition #57.

#### Comment #17:

<u>Draft Specific Condition No. 79 (SC#73):</u> This specific condition states that the NOx emissions shall be determined at each of the load conditions specified in §60.335(c)(2). The regulatory reference is incorrect. The reference should be §60.335(c).

#### **Response to Comment #17:**

The condition has been corrected to list §60.335(b) as the applicable regulatory citation.

#### Comment #18:

<u>Draft Specific Condition No. 80 (SC#74):</u> This condition contains a typo. The reference to the initial test required by Specific Condition No. 62(o) should be required by Draft Specific Condition No. 79.

## **Response to Comment #18:**

The correction has been made.

#### Comment #19:

<u>Draft Specific Condition No. 86:</u> This condition requires Martin to calculate NOx emissions for SN-26 by monitoring fuel inlet flow and applying 40.0 lb NOx/MMcf of natural gas. This condition should be removed for the following reasons.

- Draft Specific Condition No. 86 is an old condition that was originally listed in Permit 1227-AR-3 because Martin had taken a bubbled NOx limit of 53.0 ton/yr at three emission sources (SN-01, SN-25, and SN-26) in order to remain a minor source.
- With the issuance of this permit (1227-AOP-R0), Martin will no longer be a minor source facility and this source no longer has a bubbled NOx emission limit with other sources.
- The hourly emission limits for this source (SN-26) are based on maximum design fuel usage.
- Yearly emission limits are based on burning a maximum of 438 MMscf/yr of natural gas. Draft Specific Condition No. 85 requires Martin to maintain records to demonstrate that the natural gas limit is not exceeded on a 12-month rolling total basis.

#### **Response to Comment #19:**

The condition has been removed.

#### Comment #20:

<u>Draft Specific Condition No. 87:</u> Draft Specific Condition No. 87 requires Martin to maintain records required by Draft Specific Condition No. 86. This condition should be removed along with Draft Specific Condition No. 86.

## **Response to Comment #20:**

The condition has been removed.

#### Comment #21:

<u>Draft Specific Condition No. 101 (SC#93)</u>: Martin requests that the regulatory reference be changed from §60.474(c) to §60.474(c)(5) for clarification purposes.

## **Response to Comment #21:**

The change has been made.

#### Comment #22:

<u>Draft Specific Condition No. 125 (SC#117):</u> Additional NSPS Subpart Kb requirements for visual inspections [§60.113b(a)(4)] and notification to administrator [§60.113b(a)(5)] need to be added to this condition.

## **Response to Comment #22:**

The following conditions have been added to the permit:

- f. Visually inspect the internal floating roof, the primary seal, the secondary seal, gaskets, slotted membranes and sleeve seals each time the storage vessel is emptied and degassed. If the internal floating roof has defects, the primary seal has holes, tears, or other openings in the seal or the seal fabric, or the seal has holes, tears, or other openings in the seal or the seal fabric, or the gaskets no longer close off the liquid surfaces from the atmosphere, or the slotted membrane has more than 10 percent open area, the permittee shall repair the items as necessary so that none of the conditions specified in §60.113b(a)(4) exist before refilling the storage vessel with VOL. In no event shall inspections conducted in accordance with this provision occur at intervals greater than 10 years in the case of vessels conducting the annual visual inspection as specified in §60.113b(a)(2) and (a)(3)(ii) and at intervals no greater than 5 years in the case of vessels specified in paragraph (a)(3)(i) of this section.
- g. Notify ADEQ in writing at least 30 days prior to the filling or refilling of each storage vessel for which an inspection is required by §60.113b(a)(1) and (a)(4) to afford ADEQ the opportunity to have an observer present. If the inspection required by §60.113b(a)(4) is not planned and the permittee could not have known about the inspection 30 days in advance or refilling the tank, the permittee shall notify ADEQ at least 7 days prior to the refilling of the storage vessel. Notification shall be made by telephone immediately followed by written documentation demonstrating why the inspection was unplanned. Alternatively, this notification including the written documentation may be made in writing and sent by express mail so that it is received by ADEQ at least 7 days prior to the refilling.

#### Comment #23:

<u>Draft Specific Condition Nos. 142, 143, and 145 (SC#134, #135, and #137):</u> These conditions contain a typo. The condition states that compliance is demonstrated by compliance with Specific Condition No. 0. The compliance demonstration reference should be Specific Condition No. 144 (as listed in the draft permit).

#### **Response to Comment #23:**

The corrections have been made.

#### Comment #24:

<u>Specific Condition 146:</u> Martin requests that this condition be removed based on the following reasons.

- All crude oil that is off loaded/delivered to SN-28 and SN-29 feed Tank 109 (SN-27e).
- Specific Condition 110 requires that Martin sample and analyze the vapor concentration of benzene at Tank 109.
- The vapor concentration of benzene should not vary significantly from one crude oil storage tank to the next. Therefore, sampling one crude oil storage tank (Tank 109) will provide representative benzene data for crude oil storage that can be applied to all crude oil storage tanks.

## **Response to Comment #24:**

The condition has been removed.

#### Comment #25:

<u>Lube Charge Heater No. 1 (SN-36)</u>: Draft Specific Condition No. 157 limits the amount of natural gas burned to 95.67 MMscf/yr. This is a limit that Martin is taking because Martin does not to operate the heater at maximum capacity, and this limit is not based on maximum design capacity. Therefore, a specific condition needs to be added to this emission source that requires the facility to install a dedicated meter on the natural gas piping that feeds SN-36. Martin proposes that the following condition be added to the permit.

The permittee shall install and maintain a dedicated meter on the natural gas piping which feed SN-36. [ $\S$ 19.703, A.C.A.  $\S$ 8-4-203 as reference by  $\S$ 8-4-304 and  $\S$ 8-4-311]

#### **Response to Comment #25:**

The condition has been added has been added as Specific Condition #152.

#### Comment #26:

Draft Specific Condition Nos. 159, 167, 175, and 183 (SC#150, 159, 167, and 176): Martin requests that these conditions be amended to allow calculation of CO<sub>2</sub> emissions, with the option of installing a CEMS if the facility so chooses. This compliance demonstration method is consistent with greenhouse gas PSD permits recently issued by Region 6 USEPA in Texas, where the federal agency administers the GHG program. There is no NAAQS for CO<sub>2</sub>, nor is there any underlying federal requirements, including the GHG regulations, that require continuous monitoring using a CEMS. EPA Region 6 has consistently required emission calculation methods, rather than continuous monitoring, to demonstrate compliance with GHG permit limits. We request that the ADEQ follow calculation methodologies consistent with the EPA Region 6 approach.

The following Texas permits are attached as examples.

- PSD Permit Number PSD-TX-104949-GHG issued to Copano Processing, L.P.;
- PSD Permit Number PSD-TX-861-GHG, issued to Diamond Shamrock Refining Company, L.P.;
- PSD Permit Number PSD-TX-955-GHG, issued to Calpine Corporation; and
- PSD Permit Number PSD-TX-903-GHG, issued to BASF FINA Petrochemicals LP.

Martin proposes the following condition as a replacement to Draft Specific Condition Nos. 159, 167, 175, and 183.

The permittee shall calculate, on a monthly basis, the amount of CO emitted from combustion in tons/yr using equation C-2a in 40 CFR Part 98 Subpart C, converted to short tons. Compliance shall be based on a 12-month rolling basis to be updated by the last day of the following month. As an alternative, the permittee may install a CO<sub>2</sub> CEMS and volumetric stack gas flow monitoring system with an automated data acquisition and handling system for measuring and recording CO<sub>2</sub> emissions discharged to the atmosphere, and use these values to show compliance with the annual emission limit. The CO<sub>2</sub> monitor shall be operated in accordance with the ADEQ CEMS conditions and shall be operated at all times during startup and shutdown. Compliance will be demonstrated on a 12-month rolling basis. [Regulation 19, §19.703; 40 CFR Part 52, Subpart E; and A.C.A. §8-4-203 as referenced by §8-4-304 and §18-4-311]

## **Response to Comment #26:**

Since these are natural gas combustion only sources, GHG rates are based on worst case emissions, and their rates will not vary. The proposal is acceptable.

#### Comment #27:

<u>Draft Specific Condition Nos. 160, 176, and 184 (SC#151, #160, #168, and #177)</u>: To avoid significant process disruption, Martin requests that this condition be amended to allow preventative maintenance to be performed during planned shutdowns, which typically occur every eighteen to twenty-four months, and according to vendor specifications. Otherwise, compliance with this condition would likely require shutting down the entire process for flare maintenance only. Martin proposes the following wording change to Draft Specific Condition Nos. 160, 176, and 184.

The permittee shall during each scheduled facility shutdown – but no more than 24 months apart - annually-calibrate and perform preventative maintenance check of the fuel gas flow meters, perform heater burner tune ups, and perform a preventative maintenance check. The permittee shall maintain documentation on-site and records shall be made available to Department personnel upon request. [§19.703, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

#### **Response to Comment #27:**

The revision has been made.

#### Comment #28:

<u>Draft Specific Condition No. 168 (SC#160)</u>: To avoid significant process disruption, Martin requests that this condition be amended to allow preventative maintenance to be performed during planned shutdowns, which typically occur every eighteen to twenty-four months, and according to vendor specifications. Otherwise, compliance with this condition would likely require shutting down the entire process for flare maintenance only. Also, this condition states that preventative maintenance checks shall be performed on the fuel gas flow meters; however, a fuel gas flow meter is not required for this emission source. Martin proposes the following wording change to Draft Specific Condition No. 168.

The permittee shall annually calibrate and perform preventative maintenance check of the fuel gas flow meters, during each scheduled facility shutdown – but no more than 24 months apart – perform heater burner tune ups and perform a preventative maintenance check. The permittee shall maintain documentation on-site and records shall be made available to Department personnel upon request. [§19.703, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

## **Response to Comment #28:**

The revision has been made.

#### Comment #29:

<u>Lube Charge Heater No. 2 (SN-38)</u>: Draft Specific Condition No. 173 (#165) limits the amount of natural gas burned to 167 MMscf/yr. This is a limit that Martin is taking because Martin does not operate the heater at maximum capacity, and this limit is not based on maximum design capacity. Therefore, a specific condition needs to be added to this emission source that requires that the facility install a dedicated meter on the natural gas piping that feeds SN-38. Martin proposes that the following condition be added to the permit.

The permittee shall install and maintain a dedicated meter on the natural gas piping which feed SN-38. [§19.703, A.C.A. §8-4-203 as reference by §8-4-304 and §8-4-311]

#### **Response to Comment #29:**

The condition has been added as Specific Condition #169.

#### Comment #30:

Stripper Charge Heater No. 2 (SN-39): Draft Specific Condition No. 181 (#174) limits the amount of natural gas burned to 41.3 MMscf/yr. This is a limit that Martin is taking because Martin does not to operate the heater at maximum capacity, and this limit is not based on maximum design capacity. Therefore, a specific condition needs to be added to this emission source that requires that the facility install a dedicated meter on the natural gas piping that feeds SN-39. Martin proposes the following condition be added to the permit.

The permittee shall install and maintain a dedicated meter on the natural gas piping which feed SN-38. [§19.703, A.C.A. §8-4-203 as reference by §8-4-304 and §8-4-311]

#### Response to Comment #30:

The condition has been added as Specific Condition #178.

## Comment #31:

<u>Draft Specific Condition No. 193 (SC#187)</u>: This condition requires that a flame be present at the flare at all times, and that the piping system to the flare be automatically shut down in the event of a flame failure. To comply with this condition would be an OSHA violation and would potentially cause an explosion at the refinery. Martin has an Emergency Action Plan in place, which includes an evacuation of the plant as well as the community within the impact area, in the event of a flame failure and venting of uncontrolled gasses. Martin proposes the following changes to Draft Specific Condition No. 193.

The Flare (SN-40) must have a flame present at all times of operation or if no flame is present, the orifice of the unlit flare must be closed and the piping to the unlit flare shutdown to prevent passive venting of uncontrolled gases. The presence of a flare pilot light shall be monitored continuously using a thermocouple, an ultraviolet sensor, or any other equivalent device to detect the presence of a flame. In the event of a flame failure, the piping system directed to the flare must automatically shut down to prevent passive venting of ammonia gas the permittee shall maintain and follow emergency procedures until the flame is present again. [Regulation 19, §19.303, §19.304, §60.18(b) through (f), and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]

## **Response to Comment #31:**

The condition has been revised as follows:

The Flare (SN-40) must have a flame present at all times of operation. The presence of a flare pilot light shall be monitored continuously using a thermocouple, an ultraviolet sensor or any other equivalent device to detect the presence of a flame. In the event of a flame failure, the permittee shall maintain and follow emergency procedures until the flame is present again. The permittee shall report all upset conditions to the Department by the end of the next business day after the discovery of the occurrence. [Regulation 19, §19.303, §19.304, §19.601, §60.18(b) through (f), and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]

#### Comment #32:

<u>Draft Specific Condition No. 196 (SC#190)</u>: To avoid significant process disruption, Martin requests that this condition be amended to allow calibration and preventative maintenance to be performed during planned shutdowns, which typically occur every eighteen to twenty-four months, and according to vendor specifications. Otherwise, compliance with this condition would likely require shutting down the entire process for flare maintenance only. Martin proposes the following wording change to Draft Specific Condition No. 196.

The permittee shall annually during each scheduled plant shutdown—but no more than 24 months apart—calibrate and perform preventative maintenance. In accordance with manufacturer's specifications, the permittee shall ensure the flare operates with efficient combustion control and shall implement procedures to minimize the amount of carbon in the unburned gas stream through steps including proper burner tuning and the use of natural gas as the combustion fuel. The permittee shall maintain documentation on-site, and records shall be made available to Department personnel upon request. [§19.703, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

#### **Response to Comment #32:**

The condition has been revised as requested.

#### Comment #33:

<u>Draft Specific Condition No. 203 (SC#197):</u> Martin requests that the regulatory reference to be changed from §63.6603 to §60.6603(a) for clarification purposes.

## **Response to Comment #33:**

The change has been made.

#### Comment #34:

<u>Draft Specific Condition No. 227 (SC#222)</u>: Martin currently operates under NPDES Permit #AR00000591. Martin proposes the following wording changes to Draft Specific Condition No. 227.

Dust suppression activities should be conducted in a manner and at a rate of application that will not cause runoff from the area being applied. Best Management Practices (40 CFR §122.44(k)) should be used around streams and waterbodies to prevent the dust suppression agent from entering Waters of the State. Except for potable water, no agent shall be applied within 100 feet of wetlands, lakes, ponds, springs, streams, or sinkholes. Failure to meet this condition may require the permittee to obtain a National Pollutant Discharge Elimination System (NPDES) permit in accordance with 40 CFR §122.1(b). [A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311] The permittee operates under NPDES Permit # AR00000591. Compliance with the pertinent portions of this NPDES permit and/or its subsequent iterations shall be deemed compliance with this condition.

## **Response to Comment #34:**

The reference condition is standard language. The condition will remain as written.

#### Comment #35:

<u>Page 82, Source Description for SN-28 and SN-29</u>: Please make the following changes to source descriptions for SN-28 and SN-29.

MOP receives crude oil by barge, **pipeline**, and by tanker trucks. Trucks are loaded from the pipeline with daily delivery to the tanks at Sandyland (SN-28). Crude oil delivered by barge is off loaded to the tanks at Miller's (SN-29).

At Sandyland, trucks are off loaded to Tank 104, 117, Tank 110, or Tank 111. Tanks 117, 104, 110, and 111 feed Tank 104, while the refinery (Tank 109). is continuously pulling crude from Tank 104. Tanks 110 and Tank 111 can feed crude oil to the refinery, if needed.

At Miller's, barges off load to Tank 114 and Tank 115. These tanks feed crude oil to the refinery (Tank 109), Tank 104 and/or Sandyland.

#### Response to Comment #35:

The language has been revised as requested.

#### Comment #36:

<u>Page 104, Source Description, SN-41, SN-42, and SN-43:</u> Please remove the statement, "The engines will never operate more than 100 hours per year." Draft Specific Condition No. 201 limits operation of the engines to 500 hours per year.

## Response to Comment #36:

The statement has been removed.

#### Comment #37:

<u>Page 108, SN-41 and SN-43, NSPS Subpart IIII:</u> A specific condition needs to be added to address the requirement to install a non-resettable meter, as required by §60.4209(a).

## Response to Comment #37:

The condition has been added as Specific Condition #208.

#### Comment #38:

<u>Page 109 (and globally as appropriate)</u>, <u>Source name:</u>. Please change the source name to SN-44 Packaging Plant Railcar Loading and Unloading Rack.

## **Response to Comment #38:**

The change has been made.

#### Comment #39:

<u>Page 110 (and globally as appropriate)</u>, <u>Source name:</u> Please change the source name to SN-45 Packaging Plant Truck Loading and Unloading Rack.

## Response to Comment #39:

The change has been made.

#### Comment #40:

<u>Draft Plantwide Condition No. 9:</u> Please remove this condition. The only conditions requiring daily opacity observations are those referenced in Comment No. 6.

#### Response to Comment #40:

The condition has been removed

#### Comment #41:

<u>Draft Plantwide Condition No. 10:</u> Martin requests that this condition be removed. This condition requires Martin to maintain monthly VOC and HAP emissions from the loading rack sources (SN-14, SN-15, SN-16, SN-17, SN-18, and SN-21). Each emission source listed in this condition has an associated specific condition (Draft Specific Condition Nos. 34, 40, and 46) that limits the material throughput at these sources. Requiring the facility to keep both throughput records (required by the listed specific conditions) and monthly VOC and HAP emissions (required by this plantwide condition) is redundant, places an unnecessary compliance/administrative burden on the facility, and results in no environmental benefit.

#### Response to Comment #41:

The condition has been removed.

#### Comment #42:

<u>Draft Plantwide Condition No. 11:</u> This condition requires Martin to sample and analyze the vapor concentration of benzene in tanks SN-27a – SN-27l, SN-28, and SN-29. Martin requests this condition be removed for the following reasons.

- Tanks that are permitted as SN-27a and SN-27b store refinery additives and packaging plant additives. Based on MSDS information, these additives do not contain benzene. Therefore, testing for benzene in the vapor concentration of these storage tanks is unnecessary and results in no environmental benefit.
- Tanks that permitted as SN-27c store asphalt and is required to be tested for benzene in the vapor concentration by Draft Specific Condition No. 99; therefore, this plantwide condition is redundant.
- Tanks that are permitted as SN-27d store black oil. Black oil is a mixture of asphalt and lube oil. The emission calculations for black oil are based on site specific testing that was conducted at the asphalt tanks and the lube oil tanks. Therefore, as long as Martin demonstrates compliance with Specific Condition No. 99 (for asphalt) and Plantwide Condition No. 12 (for lube oils), then compliance is demonstrated for black oil. Additional testing of black oil would be redundant and results in no environmental benefit.
- Tanks that are permitted as SN-27e store crude oil and are required to be tested for benzene in the vapor concentration by Specific Condition No. 110. Therefore this plantwide condition is redundant.
- Tanks that are permitted as SN-27f store untreated distillate and liquid sampling is required by Plantwide Condition No. 12. Therefore this plantwide condition is redundant.
- Tanks that are permitted as SN-27g store gasoline and are required to be tested for benzene in the vapor concentration by Specific Condition No. 119. Therefore this plantwide condition is redundant.
- Tanks that are permitted as SN-27h store untreated lube oil and sampling is required by Plantwide Condition No. 12. Therefore this plantwide condition is redundant.
- Tanks that are permitted as SN-27i and SN-27j store treated lube oil and sampling is required by Plantwide Condition No. 12. Therefore this plantwide condition is redundant.
- Tanks that are permitted as SN-27k store reclaimed oil. The HAP emission limits for these storage tanks are based on site specific sampling data for lube oil, which ultimately ended up as reclaimed oil. Plantwide Condition No. 12 requires liquid sampling of lube oil. Therefore, demonstrating compliance with Plantwide Condition No. 12 is also a demonstration of compliance with the HAP emission limits for the reclaimed oil storage tanks. Additional testing of reclaimed oil would be redundant and results in no environmental benefit.
- Tanks that are permitted as SN-27j store treated distillate and liquid sampling is required by Plantwide Condition No. 12. Therefore this plantwide condition is redundant.
- Tanks that are permitted as SN-28 and SN-29 store crude oil that is sent to Tank 109 (SN-27e). Tank 109 is required to be tested for benzene in the vapor concentration by Specific Condition No. 110. Therefore, this plantwide condition is redundant.

## Response to Comment #42:

The condition has been removed.

#### Comment #43:

<u>Draft Plantwide Condition No. 12 (PWC#11):</u> This condition requires Martin to sample and analyze the materials stored in the distillate, heavy condensate, and lube oil storage tanks. Martin requests that the portion pertaining to heavy condensate be removed. The heavy condensate tanks are no longer vented to the atmosphere.

## Response to Comment #43:

The change has been made.

#### Comment #44:

<u>Draft Plantwide Condition No. 13 (PWC#12)</u>: Martin produces seven different types of lube oils with various vapor pressures. Martin requests to demonstrate compliance with the vapor pressure limit by averaging the vapor pressure for all lube oils produced. Martin proposes the following revised wording.

The permittee shall not exceed the following vapor pressure limits for the materials stored at this facility. The permittee shall maintain documentation to demonstrate compliance with this specific condition. [Regulation 19 §19.705 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Material	Vapor Pressure (psia)	
Additive	0.0003	
Asphalt	0.002	
Crude Oil	0.3198	
Distillate (not heavy condensate)	0.0115	
Gasoline	6.5377	
Lube Oil	0.001*	
Reclaimed Oil	0.0218	

<sup>\*</sup>Limit for the average of all the lube oils produced.

#### **Response to Comment #44:**

The condition has been revised as requested.

#### Comment #45:

<u>Draft Plantwide Condition No. 15:</u> Martin requests that this condition be removed. This condition requires Martin to maintain monthly VOC and HAP emissions from the loading rack sources (SN-32, 33, SN-34, SN-44, and SN-45). Each emission source listed in this condition has an associated specific condition (Draft Specific Condition Nos. 97, 104, 113, 123, 132, 219, and 223) that limits the material throughput. Requiring that the facility keep both throughput records (required by the listed specific conditions) and monthly VOC and HAP emissions (required by this plantwide condition) is redundant, places an unnecessary compliance/administrative burden on the facility, and results in no environmental benefit.

## **Response to Comment #45:**

The condition has been removed.

## Comment #46:

<u>Draft Plantwide Condition No. 18:</u> This condition is addressed by Draft Plantwide Condition No. 19; therefore, it is redundant and should be removed. If ADEQ does not agree with removing this condition, the regulatory reference should be §60.108a(c)(6)(ix), not §60.108a(c)(1).

## **Response to Comment #46:**

The condition has been revised to cite the correct regulatory reference as requested.

#### Comment #47:

Draft Plantwide Condition No. 19: The wording of this condition should be revised as follows.

The permittee shall record and maintain records of discharges greater than 500 lb/day  $SO_2$  from and discharges to an affected flare in excess of 500,000 scfd. The records shall include all items listed in  $\S60.108a(c)(6)(i)$  through (vixi), as applicable.  $[\S60.108a(c)(6)]$ 

## Response to Comment #47:

The condition has been revised.

#### Comment #48:

<u>Draft Plantwide Condition No. 21a:</u> The span value for the instrument should be 300 ppmv H<sub>2</sub>S, not 320 ppmv H<sub>2</sub>S pursuant to §60.107a(a)(2)(i).

## **Response to Comment #48:**

The condition has been corrected.

#### Comment #49:

<u>Plantwide Conditions for NSPS Subpart Ja</u>: For the Flare (SN-40), a condition needs to be added to address §60.107a(e). Martin will monitor total reduced sulfur as allowed by §60.107a(e)(1). Also, the flare does not meet the definition of an emergency flare (used 4 times or less); therefore, a condition needs to be added to require flow monitoring as required by §60.107a(f).

#### **Response to Comment #49:**

The following conditions have been added to the permit:

14. The permittee shall determine the total reduce sulfur concentration for each gas line directed to the affected flare. The permittee shall install, operate, calibrate, and maintain an instrument for continuously monitoring and recording the concentration of total reduced sulfur in gas discharged to the flare.

- a. The permittee shall install operate and maintain each total reduced sulfur monitor according to Performance Specification 5 of Appendix B to Part 60.
- b. The permittee shall conduct performance evaluations of each total reduced sulfur monitor according to the requirements of §60.13(c) and Performance Specification 5 of Appendix b to Part 60. For flares that routinely have flow, the permittee of each total reduced sulfur monitor shall use EPA Method 15A of Appendix A-5 to Part 60 for conducting the relative accuracy evaluations.
- c. The permittee shall comply with the applicable quality assurance procedures in Appendix F to Part 60 for each total reduced sulfur monitor. [§60.107a(e)(1)]
- 15. The permittee shall install, calibrate, operate and maintain a CPMS to measure and record the flow rate of the gas discharged to the flare. If a flow monitor is not already in place, the permittee shall comply with §60.107a(f) by no later than November 15, 2015, or upon startup of the flare, whichever is later. The permittee shall install, calibrate, operate and maintain each flow monitor according to the manufacturer's procedures and specifications and the following requirements:
  - a. Locate the monitor in a position that provides a representative measurement of the total gas flow rate.
  - b. Use a flow sensor with a measurement sensitivity of no more than 5 percent of the flow rate or 10 cubic feet per minute, whichever is greater.
  - c. Use a flow monitor that is maintainable online, is able to continuously correct for temperature and pressure and is able to record flow in standard conditions (as defined in §60.2) over one-minute averages.
  - d. At least quarterly, perform a visual inspection of all components of the monitor for physical and operational integrity and all electrical connections for oxidation and galvanic corrosion of the flow monitor is not equipped with a redundant flow sensor.
  - e. Recalibrate the flow monitor in accordance with the manufacturers' procedures and specification biennially (every two years) or at the frequency specified by the manufacturer.

 $[\S60.107a(f)(1)]$ 

## Comment #50:

<u>Draft Plantwide Condition No. 22:</u> This condition can be removed from the permit. Martin is not using the exemption allowed by  $\S60.102a(h)$ . Martin will monitor the  $H_2S$  concentration.

#### **Response to Comment #50:**

The condition has been removed.

#### Comment #51:

<u>Draft Plantwide Condition No. 23 (PWC#22)</u>: Martin requests that the excess emissions report required by this condition be reported in accordance with General Provision 7 and submitted with the Semi-Annual Monitoring Report. The following revised wording is proposed.

The permittee shall submit an excess emissions report for all periods of excess emissions according to the requirements of  $\S60.7(c)$  except that the report shall contain the information specified in  $\S60.108a(d)(1)$  through (7). The excess emissions report shall be submitted in accordance with General Provision No. 7.  $[\S60.108a(d)]$ 

## **Response to Comment #51:**

The language has been added as requested.

#### Comment #52:

<u>Draft Plantwide Condition No. 24 (PWC#23)</u>: NSPS Subpart GGGa applies to equipment associated with the Lube Hydrotreater Unit No. 2 (LHT2), not just SN-38 and SN39. Therefore, the following revised wording is proposed.

All equipment associated with Lube Hydrotreater Unit No. 2, which includes but is not limited to SN-38 and SN-39, shall comply with the requirements of §60.482-1a to §60.482-10a as soon as practicable, but no later than 180 days after initial startup. [40 CFR Part §60.592a(a)]

## Response to Comment #52:

The language has been added for clarification.

#### Comment #53:

Draft Plantwide Condition No. 48: This condition references §60.697(f)(3), which does not apply to the wastewater treatment system at Martin; therefore, this condition should be removed from the permit. Pursuant to §60.692-3(b), a control device is not required for the wastewater treatment system if the oil water separator tank is less than 250 gpm and pursuant to §60.692-3(f) the oil water separator tank will be equipped with a pressure control valve. Therefore, the wastewater treatment system tanks will not be equipped with a fixed roof or with a control device that would be subject to the requirements of §60.697(f)(3).

#### **Response to Comment #53:**

The condition has been removed from the permit.

#### Comment #54:

<u>Draft Plantwide Condition No. 54 (PWC#52)</u>: Martin requests that the initial and semi-annual reports required by Subpart QQQ be submitted in accordance with General Provision 7. The following revised wording is proposed.

The permittee shall submit an initial and semiannual report to the Department that summarizes all inspections when a water seal was dry or otherwise breached, when a drain cap or plug was missing or improperly installed, or when cracks, gaps, or other problems were identified that could result in VOC emissions, including information about the repairs or corrective action taken. The initial and semiannual report shall be submitted in accordance with General Provision No. [40 CFR Part 60, Subpart QQQ §60.698(c) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

## Response to Comment #54:

The language has been added as requested.

#### Comment #55:

<u>Page 129, Applicable Regulations:</u> 40 CFR Part 60, Subpart GGGa applies to the Lube Hydrotreater Unit No. 2, not the individual heaters SN-38 and SN-39. Please remove the reference to SN-38 and SN-39, and replace it with Lube Hydrotreater Unit No. 2.

## **Response to Comment #55:**

The reference has been changed to Lube Hydrotreater Unit No. 2 which includes the individual heaters SN-38 and SN-39 as referenced in Comment #52.

The following comment was submitted by Allen Chang on behalf of EPA

#### Comment #56:

## Draft TV/PSD Permit #1227-AOP-R0

Page 107, Specific Condition #207 (SC#201), it states, "The permittee may operate the emergency stationary RICE engines for the purpose of maintenance checks and readiness testing, provided that the tests are recommended by Federal, State, or local government, the manufacturer, the vendor or the insurance company associated with the engine... [Regulation No.19 §19.304 and 40 CFR Part 63, Subpart ZZZZ, §63.6640(f)(l)(i, ii, iii)]"

There is no (i. ii. iii) in the text of 40 CFR Part 63, Subpart ZZZZ, §63.6640(f)(l), the above stated requirements are similar to that of in §63.6640(f)(4), please verify the citation and make appropriate correction.

#### **Response to Comment #56:**

The regulatory citation has been corrected to reference  $\S63.6640(f)(4)$ .



February 11, 2014

Charles Clark
Vice President of Human Resources & Government Affairs
Martin Operating Partnership L.P.
484 East 6th Street
Smackover, AR 71762

Dear Mr. Clark:

The enclosed Permit No. 1227-AOP-R0 is your authority to construct, operate, and maintain the equipment and/or control apparatus as set forth in your application initially received on 3/11/2013.

After considering the facts and requirements of A.C.A. §8-4-101 et seq., and implementing regulations, I have determined that Permit No. 1227-AOP-R0 for the construction, operation and maintenance of an air pollution control system for Martin Operating Partnership L.P. to be issued and effective on the date specified in the permit, unless a Commission review has been properly requested under Arkansas Department of Pollution Control & Ecology Commission's Administrative Procedures, Regulation 8, within thirty (30) days after service of this decision.

The applicant or permittee and any other person submitting public comments on the record may request an adjudicatory hearing and Commission review of the final permitting decisions as provided under Chapter Six of Regulation No. 8, Administrative Procedures, Arkansas Pollution Control and Ecology Commission. Such a request shall be in the form and manner required by Regulation 8.603, including filing a written Request for Hearing with the APC&E Commission Secretary at 101 E. Capitol Ave., Suite 205, Little Rock, Arkansas 72201. If you have any questions about filing the request, please call the Commission at 501-682-7890.

Sincerely,

Mike Bates

Chief, Air Division

# ADEQ OPERATING AIR PERMIT

Pursuant to the Regulations of the Arkansas Operating Air Permit Program, Regulation 26:

Permit No.: 1227-AOP-R0

IS ISSUED TO:

Martin Operating Partnership L.P.
484 East 6th Street
Smackover, AR 71762
Union County
AFIN: 70-00039

THIS PERMIT AUTHORIZES THE ABOVE REFERENCED PERMITTEE TO INSTALL, OPERATE, AND MAINTAIN THE EQUIPMENT AND EMISSION UNITS DESCRIBED IN THE PERMIT APPLICATION AND ON THE FOLLOWING PAGES. THIS PERMIT IS VALID BETWEEN:

February 11,2014

AND

February 10, 2019

THE PERMITTEE IS SUBJECT TO ALL LIMITS AND CONDITIONS CONTAINED HEREIN.

Signed:

Mike Bates

Chief, Air Division

February 11, 2014

Date

AFIN: 70-00039

# Table of Contents

SECTION I: FACILITY INFORMATION	
SECTION II: INTRODUCTION	
Summary of Permit Activity	
Process Description	9
Regulations	11
Emission Summary	
SECTION III: PERMIT HISTORY	26
SECTION IV: SPECIFIC CONDITIONS	34
SN-01 Crude Charge Heater	34
SN-02 Vacuum Tower Charge Heater	
SN-03 Asphalt Charge Heater	38
SN-08 Hydrogen Reformer No.1	40
SN-12 Stripper Charge Heater No. 1	42
SN-14 Distillate Loading Rack	44
SN-15 & SN-16 Asphalt/Black Oil Truck Loading Rack No.1 and Asphalt/Black Oil	il Truck
Loading Rack No. 2	
SN-17, SN-18, and SN-21 Lube Oil Truck Loading Rack, Lube Oil Truck Loading F	Rack, and
Lube Oil Rail Car Loading Rack	48
SN-23 Fugitive Emissions	50
SN-24 Wastewater Emissions	53
SN-25 Cogeneration Unit	54
SN-26 Boiler #4	59
SN-27a Refinery Additives Storage Tanks	61
SN-27b Packaging Plant Additive Storage Tanks	62
SN-27c Asphalt Storage Tanks	63
SN-27d Black Oil Storage Tanks	65
SN-27e Crude Oil Storage Tank	67
SN-27f Untreated Distillate Storage Tank	69
SN-27g Gasoline Storage tank	71
SN-27h Untreated Lube Oil Storage Tank	74
SN-27i Refinery Treated Lube Oil Storage Tanks	
SN-27j Packaging Plant Treated Lube Oil Storage Tanks	79
SN-27k Reclaim Oil Storage Tanks	80
SN-271 Treated Distillate Storage Tanks	
SN-28 & SN-29 Sandyland Storage Tanks and Miller's Storage Tanks	83
SN-32 Asphalt/Black Oil Tank Car Loading Rack	85
SN-33 Lube Oil & Distillate Rail Car Loading and Additive Railcar Unloading	86
SN-34 Packaging Plant Lube Oil Loading/Unloading Docks	87
SN-36 Lube Charge Heater No. 1	88
SN-37 Hydrogen Reformer No. 2	91
SN-38 Lube Charge Heater No. 2	94
SN-39 Stripper Charge Heater No. 2	97
SN-40 Flare	100

SN-41, SN-42, and SN-43 Fire Pump Engine No.1, Fire Pump Engine No.2	, and Fire Pump
Engine No. 3	
SN-44 Packaging Plant Railcar Loading Rack	109
SN-45 Packaging Plant Truck Loading Rack	110
SN-46 Haul Roads	111
SECTION V: COMPLIANCE PLAN AND SCHEDULE	112
SECTION VI: PLANTWIDE CONDITIONS	
Title VI Provisions	
Permit Shield	
SECTION VII: INSIGNIFICANT ACTIVITIES	
SECTION VIII: GENERAL PROVISIONS	
Appendix A - 40 CFR Part 60, Subpart J	
Appendix B - 40 CFR Part 60, Subpart Ja	
Appendix C - 40 CFR Part 60, Subpart Kb	
Appendix D - 40 CFR Part 60, Subpart UU	
Appendix E - 40 CFR Part 60, Subpart GG	
Appendix F - 40 CFR Part 60, Subpart GGGa	
Appendix G- 40 CFR Part 60, Subpart QQQ	
Appendix H - 40 CFR Part 60 Subpart IIII	
Appendix I - 40 CFR Part 61, Subpart FF	
Appendix J - 40 CFR Part 63, Subpart ZZZZ	
Appendix K- 40 CFR Part 63, Subpart CCCCCC	
Appendix L – Examples of Loading Rack Calculations	
Appendix M - Inventory of Tanks	
**	

Permit #: 1227-AOP-R0

AFIN: 70-00039

# List of Acronyms and Abbreviations

A.C.A. Arkansas Code Annotated

AFIN ADEQ Facility Identification Number

CFR Code of Federal Regulations

CO Carbon Monoxide

HAP Hazardous Air Pollutant

lb/hr Pound Per Hour

MVAC Motor Vehicle Air Conditioner

No. Number

NO<sub>x</sub> Nitrogen Oxide

PM Particulate Matter

PM<sub>10</sub> Particulate Matter Smaller Than Ten Microns

SNAP Significant New Alternatives Program (SNAP)

SO<sub>2</sub> Sulfur Dioxide

SSM Startup, Shutdown, and Malfunction Plan

Tpy Tons Per Year

UTM Universal Transverse Mercator

VOC Volatile Organic Compound

Permit #: 1227-AOP-R0

AFIN: 70-00039

## SECTION I: FACILITY INFORMATION

PERMITTEE: Martin Operating Partnership L.P.

AFIN: 70-00039

PERMIT NUMBER: 1227-AOP-R0

FACILITY ADDRESS: 484 East 6th Street

Smackover, AR 71762

MAILING ADDRESS: 484 East 6th Street

Smackover, AR 71762

COUNTY: Union County

CONTACT NAME: Charles Clark

CONTACT POSITION: Vice President of Human Resources & Government Affairs

TELEPHONE NUMBER: 870-881-8700

REVIEWING ENGINEER: Kimberly O'Guinn

UTM North South (Y): Zone 15: 3691692.87 m

UTM East West (X): Zone 15: 526334.70 m

Permit #: 1227-AOP-R0

AFIN: 70-00039

#### SECTION II: INTRODUCTION

#### **Summary of Permit Activity**

Martin Operating Partnership L.P. (MOP) operates an oil refinery at 484 East Sixth Street, in Smackover, Union County, Arkansas, 71762. This permitting action is to allow the facility to operate as a Title V source due to the facility becoming a major source of VOC and greenhouse gas (GHG) emissions. This modification also addresses modifications to existing sources and the installation of new sources that are subject to a Prevention of Significant Deterioration (PSD) review.

With this modification, the following changes occurred:

- Removal of the Lube Precharge Heater (SN-07) and the Lube Charge Heater (SN-30), both associated with the existing lube hydrotreater unit (LHT1) and install one new heater, Lube Charge Heater No. 1 (SN-36).
- Removal of the High Pressure Flare (SN-09) and Low Pressure Flare (SN-10) and install a new flare, (SN-40).
- Renaming of the existing Hydrogen Plant Heater/Reformer (SN-08) to be "Hydrogen Reformer No.1".
- Renaming of the existing Stripper Charge Heater (SN-12) to be "Stripper Charge Heater No. 1".
- Addition of the black oil loading to SN-15 and SN-16 and rename these sources:
  - o Asphalt/Black Oil Truck Loading Rack No. 1 (SN-15)
  - o Asphalt/Black Oil truck Loading Rack No. 2 (SN-16)
- Re-permitting all onsite storage tanks that are currently permitted as two source numbers, SN-27 and SN-35, as sources SN-27a through SN-27l, grouped according to what is stored in each tank, as follows:
  - o Refinery Additives Storage Tanks (SN-27a)
  - o Packaging Plant Additive Storage Tanks (SN-27b)
  - o Asphalt Storage Tanks (SN-27c)
  - o Black Oil Storage Tanks (SN-27d)
  - o Crude Oil Storage Tank (SN-27e)
  - o Untreated Distillate Storage Tanks (SN-27f)
  - o Gasoline Storage Tanks (SN-27g)
  - o Untreated Lube Oil Storage Tanks (SN-27h)
  - o Refinery Treated Lube Oil Storage Tanks (SN-27i)
  - o Packaging Plant Treated Lube Oil Storage Tanks (SN-27i)
  - o Reclaim Oil Storage Tanks (SN-27k)
  - o Treated Distillate Storage Tanks (SN-271)
- Installation of new emission sources:
  - o Hydrogen Reformer No. 2 (SN-37)
  - Lube Hydrotreater No. 2 Sources: Lube Charge Heater No. 2 (SN-38) and Stripper Charge Heater No. 2 (SN-39)
  - o Fire Pump Engine No. 1 (SN-41)

Permit #: 1227-AOP-R0

AFIN: 70-00039

- o Fire Pump Engine No. 2 (SN-42)
- o Fire Pump Engine No. 3 (SN-43)
- Addition of natural gas limits for the following sources.
  - o Crude Charge Heater (SN-01) 250.0 MMcf/yr
  - o Vacuum Charge Heater (SN-02), 63.0 MMcf/yr
  - o Boiler No. 4 (SN-26), 438.0 MMcf/yr
  - o Lube Charge heater No. 1 (new source, SN-36), 95.67 MMcf/yr
  - o Lube Charge heater No. 2 (new source, SN-38) 167.0 MMcf/yr
  - o Stripper Charge heater No. 2 (new source, SN-39), 41.30 MMcf/yr
- Changes to tanks Storage Inventory
  - O Distillate tanks: Add new tanks, 120, 121, and 122. Existing lube oil tanks 206, 266, and 292 will now be distillate tanks
  - o Treated Lube Oil tanks: Add new tanks 344, 345, and 346. Replace existing tanks 247, 248, 271, 272, and 274) with three new tanks,
  - o New lube oil tanks: tanks 348, 349, 350, 351, 352, and 353.
  - o New Packaging Plant tanks PO80, PO81, and PO82
- Relocation of the Distillate Loading Rack (SN-14). Naphtha and Heavy Condensate will continue to be loaded at the current SN-14 location; however, because of the vapor recovery system, there are no emissions associated with Naphtha and Heavy Condensate.
- Rebuilding of the loading rack SN-33 and install two new loading racks, SN-44 and SN-45 for the Packaging Plant.
- Relocation of the Lube Oil Truck Loading Rack.
- Annual NOx emissions for SN-01, SN-25, and SN-26 were previously permitted as a bubble limit. This modification assigns individual annual limits are for each sources.

## PSD Applicability Analysis

The physical modifications with this permit will allow MOP to make higher quality lubricant streams than the current Reformer and Hydrotreater can make. However, MOP is still limited by the crude unit on the amount of crude oil that can be processed, which is 7,700 bbl/day.

The PSD Applicability Analysis demonstrates that the increase in criteria pollutant emissions as a result of the proposed modifications are below the PSD significance increase level (SIL) for each pollutant, except for greenhouse gas (CO<sub>2e</sub>) pollutants.

To determine PSD applicability, regulated pollutant emission increases were compared to the PSD SILs. Under PSD regulations, emissions increases for new emission units and existing emissions units are calculated differently.

Permit #: 1227-AOP-R0

AFIN: 70-00039

#### **Baseline Actual Emissions**

Baseline actual emissions are defined by 40 CFR part 52.21(b)(48)(ii) as the average rate (in tons per year) at which the emissions unit actually emitted the pollutant during any consecutive 24-month period within the 10-year period immediately preceding the date construction begins or the date the permit application is received by the reviewing authority, whichever is earlier. In relation to the actual emissions of criteria pollutants, the 10-year period of October 2002 through September 2012 was used for the PSD analysis.

#### Potential-To-Emit Emissions

Pursuant to 40 CFR 52.21(b)(41)(ii)(d), the facility may elect to use the emissions unit's potential to emit (PTE), in tons per year, as the projected actual emissions. PTE is defined by 40 CFR Part 52.21(b)(4) as the maximum capacity of a stationary source under its physical and operational design. The definition further states that a federally enforceable limit on the amount of fuel combusted is considered an operational limitation for PTE purposes. With this modification, MOP proposed federally enforceable natural gas usage limits for some of the combustion sources.

The emission increases from all affected sources were calculated as the PTE minus the baseline actual emission.

Pollutant	Baseline Actual Emissions	PTE <sup>1</sup> Emissions	Emissions Increase	PSD Significance Level	PSD Review Required?
	ton/yr	ton/yr	ton/yr	ton/yr	ton/yr
PM	3.7	11.8	8.1	25	No
$PM_{10}$	3.7	9.1	5.4	15	No
PM <sub>2.5</sub>	3.7	9.1	5.4	10	No
$SO_2$	9.2	17.3	8.1	40	No
VOC	86.3	108.5	14.3	40	No
СО	35.5	77.4	41.9	100	No
NOx	53.6	90.7	37.1	40	No
CO <sub>2e</sub>	73,053	208,765	135,712	75,000	Yes

<sup>1</sup>PTE as limited by the federally enforceable fuel combustion limits taken at some of the combustion sources

Permit #: 1227-AOP-R0

AFIN: 70-00039

As demonstrated in the above table, the baseline actual-to-PTE analysis indicates the modifications will not result in emissions exceeding the PSD SILs for all criteria pollutants, with the exception of greenhouse gas pollutants (CO<sub>2e</sub>).

## **BACT Analysis**

This permit addresses the physical expansion of the MOP facility, which results in emissions from new process equipment that exceeds the significant emission rate for greenhouse gases (GHG). Specifically, a BACT analysis is required due to the installation of the Hydrogen Reformer No. 2 (SN-37).

SN	Pollutant	BACT Determination		
37	CO <sub>2e</sub>	Efficient Combustion Control	0.05 lb CO <sub>2e</sub> /scf H <sub>2</sub> production	
36 38 39	CO <sub>2e</sub>	Efficient Combustion Control	117 lb CO <sub>2e</sub> /MMBtu	
40	CO <sub>2e</sub>	Efficient Combustion Control	300.1 lb/hr	

With this modification permitted emissions will decrease by 1.0 tons/year (tpy)  $PM_{10}$  and 10.0 tpy CO. Permitted emissions will increase by 1.7 tpy of PM, 12.4 tpy  $SO_2$ , 44.4 tpy VOC, 19.4 tpy NOx, 73,035.8 tpy of  $CO_{2e}$ , 0.02 tpy Acetone, and 1.22 tpy Total HAPs.

## **Process Description**

MOP is a refinery which processes produced crude oil into naphtha, distillate, lube oils, and asphalt. MOP also operates an adjoining packaging facility. Crude oil is charged from storage and is preheated with heat exchangers. The crude passes through an electrostatic desalting unit which separates the saltwater from the crude. From the desalters, the crude is heated through a series of exchangers and then through the Crude Charge Heater (SN-01). Finally, the crude is sent to the atmospheric distillation tower where the oil is separated into naphtha, distillate, No. 2, No. 3 and No. 4 lube oils.

Naphtha and other non-condensable gases flow overhead from the atmospheric tower. The naphtha is cooled in a condenser and then flows to an accumulator from which a portion of the naphtha liquid is pumped back to the tower as reflux. Excess naphtha is drawn off of the accumulator and pumped to product storage for sales. Distillate and Lube Oils No. 2, 3, and 4 are drawn off of the side of the tower, routed through strippers to remove non-condensable gases, and then pumped through a series of exchangers and on to storage where Distillate is loaded at SN-14. The naphtha loading rack is equipped with a vapor recovery system and all emissions are captured and routed back to the naphtha storage tanks.

Permit #: 1227-AOP-R0

AFIN: 70-00039

The bottom stream (residue) off of the atmospheric tower is pumped through the Vacuum Tower Charge Heater (SN-02) before being charged to the vacuum distillation column. The residue is separated in the vacuum tower to produce the heavier grades of lube oil (No. 7, 9, 10, and 11) and pumped through a series of exchangers and then to storage for product sales. The vacuum tower bottoms are asphalt flux which is pumped through heat exchangers to storage and then circulated through the Asphalt Heater (SN-03) to maintain tank temperature. The asphalt product is loaded into trucks and tank cars at one of the asphalt loading facilities (SN-15, SN-16, and SN-32).

The lube oils produced by both atmospheric and vacuum distillation are further processed in a set of exchangers and then are passed by two hydrotreaters. The oils are heated through the heat exchangers and the Lube Charge Heater No. 1 (SN-36) and the Lube Charge Heater No. 2 (SN-38) before being pumped into the top of each reactor (LHT1 or LHT2). The hot lube oils combine with hydrogen at the top of each reactor before passing through a catalyst bed. Sulfur in the oil reacts with the hydrogen to form hydrogen sulfide gas. The hydrogen gas also saturates the aromatic compounds in the oil, removes heavy metals, and converts some nitrogen to ammonia.

Each reactors effluent flows to a high pressure separator where the excess hydrogen, hydrogen sulfide, and ammonia gases flash off. From the high pressure separator, the oil flows to a low pressure separator where additional light ends flash off. The oil then flows to a lube stripper where the remaining hydrogen sulfide is removed by stream stripping. The Stripper Charge Heater No. 1 (SN-12) and Stripper Charge Heater No. 2 (SN-39) supply heat to the lube oil stripper. From the lube stripper, the oil flows to a vacuum flash tower where any entrained water is vacuum stripped from the product. The bottoms from the vacuum stripper tower are routed to finished lube storage for blending and sales. The finished lube products are loaded at lube oil loading racks (SN-17, SN-18, SN-21, and SN-33) and also pumped to the Packaging Plant.

The waste gas from the high pressure separator is routed to a series of amine and caustic scrubber where the gas enters the bottom of the column. The gas flows countercurrent to the amine and caustic solution that removes the hydrogen sulfide in the gas. The waste gas from the low pressure separator combines with the gases from the lube stripper and the lube vacuum flash tower. The combined gas stream is then treated in a two stage caustic scrubber system. The clean hydrogen gas from the scrubbers is then sent back to the hydrotreater reactors (LHT1 and LHT2).

The primary hydrogen is supplied to the two hydrotreaters by two steam/methane reformers (SN-08 and SN-37). At each reformer, natural gas is compressed, heated in a preheat exchanger, and combined with steam. The mixture is charged to the Hydrogen Plant Heaters (SN-08 and SN-37) where it passes over a catalyst and reacts with steam to produce hydrogen and carbon oxides. The gases leaving the reactor are routed to a shift converter which contains an iron-chromium catalyst. Most of the carbon monoxide (CO) in the gas is converted to carbon dioxide (CO<sub>2</sub>) and hydrogen (H<sub>2</sub>). The CO<sub>2</sub> and H<sub>2</sub> gas then flow to a pressure swing absorption (PSA) system where the CO<sub>2</sub> and other impurities are removed.

Permit #: 1227-AOP-R0

AFIN: 70-00039

Steam is produced in a boiler (SN-26) and a cogeneration unit (SN-25) at the facility. Steam is also produced by the reformers to be used at the reformers with excess steam routed to processes at the facility. The cogeneration unit contains a gas fired turbine, which along with the boiler, uses natural gas as fuel. The facility has one emergency flare (SN-40) for emergency releases from the crude unit and the hydrogen reformer units. The facility also has three fire pump emergency engines (SN-41, SN-42, and SN-43).

## Regulations

The following table contains the regulations applicable to this permit.

Regulations
Arkansas Air Pollution Control Code, Regulation 18, effective June 18, 2010
Regulations of the Arkansas Plan of Implementation for Air Pollution Control, Regulation 19, effective July 27, 2013
Regulations of the Arkansas Operating Air Permit Program, Regulation 26, effective November 18, 2012
40 CFR Part 60, Subpart J – Standards of Performance for Petroleum Refineries
40 CFR Part 60, Subpart Ja, Standards of Performance for Petroleum Refineries for Which Construction, Reconstruction, or Modification Commenced After May 14, 2007
40 CFR Part 60, Subpart Kb – Standards of Performance for Volatile Organic Liquid Storage Vessels (Including Petroleum Liquid Storage Vessels) for Which Construction, Reconstruction, or Modification Commenced After July 23, 1984
40 CFR Part 60, Subpart GG – Standards of Performance for Stationary Gas Turbines
40 CFR Part 60, Subpart UU – Standards of Performance for Asphalt Processing and Asphalt Roofing Manufacture
40 CFR Part 60, Subpart GGGa – Standards of Performance for Equipment Leaks for VOC in Petroleum Refineries for Which Construction, Reconstruction, or Modification Commenced After November 7, 2006
40 CFR Part 60, Subpart QQQ - Standards of Performance for VOC Emissions From Petroleum Refinery Wastewater Systems
40 CFR Part 60 Subpart IIII – Standards of Performance for Stationary Compression Ignition Internal Combustion Engines
40 CFR Part 61, Subpart FF – National Emission Standards for Hazardous Air Pollutants, Benzene Waste Operations
40 CFR Part 63, Subpart ZZZZ – National Emission Standard for Stationary Reciprocating Internal Combustion Engines
40 CFR Part 63, Subpart CCCCC - National Emission Standards for Hazardous Air Pollutants for Source Category: Gasoline Dispensing Facilities

Permit #: 1227-AOP-R0

AFIN: 70-00039

Subpart FF applies because MOP is a petroleum refinery. The boiler SN-26 was manufactured in 1971 and is therefore not subject to the requirements of NSPS Subpart Dc. The reformer SN-08 burns pipeline quality gas so it is not subject to 40 CFR Part 60, Subpart J – Standards of Performance for Petroleum Refineries.

Although the lube oil tanks #026p, #027p, #028p, #029p (1,182 barrel), #031p, #032p, #229, #230, #231, #232, #233, #332, #333, #334, #335, #336, #337, #338, #339, #340, #P041, #P042, #P043, #P044, #P045, #P046, #P047, #P048, #P049, #P050, #P051, #P052, #P060, #P061, #P062 and #P063 are greater than 75 cubic meters (19,813 gallons), the vapor pressure of the lube oil (<0.007 kPa) is below the NSPS Subpart Kb threshold of 3.5 kPa, therefore, the lube oil tanks are not subject to any other requirements of Subpart Kb.

40 CFR Part 60, Subpart GGG – Standards Performance for Equipment Leaks of VOC in Petroleum Refineries. This regulation applies to certain compressors and other equipment in VOC service installed after January 4, 1983. The permittee consultant's letter dated April 25, 2007 states that all compressors installed after the effective date are in hydrogen service and are not in VOC service. Therefore, no compressors at this facility are subject to Subpart GGG.

This facility is classified as a major source of greenhouse gas emissions.

AFIN: 70-00039

# **Emission Summary**

The following table is a summary of emissions from the facility. This table, in itself, is not an enforceable condition of the permit.

EMISSION SUMMARY				
Source	Description	Pollutant	Emission Rates	
Number	Description	ronutant	lb/hr	tpy
		PM	4.5	12.5
		PM <sub>10</sub>	3.8	9.8
		SO <sub>2</sub>	9.0	21.0
Total A	Allowable Emissions	VOC	107.9	127.5
		СО	26.7	76.7
		$NO_X$	33.4	92.5
		$\mathrm{CO}_{2\mathrm{e}}$	57,176.7	210,745.8
		Benzene*	0.114	0.62
		Beryllium*	4.62E-06	0.11
		Cadmium*	5.73E-04	0.11
	Chromium*	4.94E-04	0.11	
		Ethylbenzene*	0.80	2.43
	Formaldehyde*	0.14	0.15	
	HAPs	Hexane*	4.89	6.68
		MTBE*	1.21	0.01
		POM*	9.60E-03	0.37
	Styrene*	0.03	0.01	
		Toluene*	1.40	2.80
		2,2,4-Trimethylpentane*	0.45	0.15
		Xylene*	1.12	2.37
Air	Contaminants **	Acetone	0.01	0.02

EMISSION SUMMARY				
Source	Description	Pollutant	Emission Rates	
Number	Description	1 Ondiant	lb/hr	tpy
		PM	0.3	1.0
		$PM_{10}$	0.3	1.0
		$\mathrm{SO}_2$	1.0	3.8
į		VOC	0.2	0.7
		CO	2.7	10.5
		$NO_x$	3.2	12.5
	C 1 Cl II - 4	$\mathrm{CO}_{2\mathrm{e}}$	3,765.7	15,003.5
01	Crude Charge Heater	Benzene	7.0E-05	0.01
	32 MMBtu/hr	Beryllium	4.0E-07	0.01
		Cadmium	0.0001	0.01
		Chromium	5.0E-05	0.01
		Formaldehyde	0.01	0.01
}		Hexane	0.06	0.23
		POM (including Naphthalene)	2.0E-05	0.01
		Toluene	0.01	0.01
		PM	0.2	0.5
		$\mathrm{PM}_{10}$	0.2	0.5
		$\mathrm{SO}_2$	3.2	8.1
		VOC	0.3	0.7
		CO	0.5	1.3
		$NO_x$	0.4	0.9
	Vacuum Tower	$\mathrm{CO}_{2\mathrm{e}}$	1,482.7	3,780.9
02	Charge Heater 12.6 MMBtu/hr	Benzene	3.0E-05	0.01
		Beryllium	2.0E-07	0.01
		Cadmium	0.0001	0.01
		Chromium	2.0E-05	0.01
		Formaldehyde	0.01	0.01
		Hexane	0.03	0.06
		POM (including Naphthalene)	8.0E-06	0.01
		Toluene	0.01	0.01

		EMISSION SUMMARY		
Source	Degarintian	Pollutant	Emission Rates	
Number	Description		lb/hr	tpy
		PM	0.1	0.3
		$PM_{10}$	0.1	0.3
		$SO_2$	0.3	1.1
		VOC	0.1	0.2
		СО	0.7	2.9
		$NO_x$	0.8	3.5
	A ambolt Change	$\mathrm{CO}_{2\mathrm{e}}$	941.4	4,123.3
03	Asphalt Charge	Benzene	2.0E-05	0.01
	Heater	Beryllium	1.0E-07	0.01
		Cadmium	9.0E-06	0.01
		Chromium	2.0E-05	0.01
		Formaldehyde	0.01	0.01
		Hexane	0.02	0.07
		POM (including Naphthalene)	6.0E-06	0.01
		Toluene	0.01	0.01
-		PM	0.3	1.0
		$PM_{10}$	0.3	1.0
		$SO_2$	0.1	0.1
		VOC	0.2	0.8
		СО	2.5	10.9
		$NO_x$	1.3	5.5
	Hydrogen Reformer	$\mathrm{CO}_{2\mathrm{e}}$	6,245.3	27,354.3
08	No.1	Benzene	7.0E-05	0.01
	30 MMBtu/hr	Beryllium	4.0E-07	0.01
		Cadmium	4.0E-05	0.01
		Chromium	5.0E-05	0.01
		Formaldehyde	0.01	0.01
		Hexane	0.06	0.24
		POM (including Naphthalene)	3.0E-05	0.01
		Toluene	0.01	0.01

EMISSION SUMMARY				
Source	Description	Pollutant	Emission Rates	
Number	Description		lb/hr	tpy
		PM	0.1	0.5
		$PM_{10}$	0.1	0.5
		$SO_2$	0.5	1.9
		VOC	0.1	0.4
		СО	1.2	5.1
		$NO_x$	1.4	6.1
	Stripper Charge	$CO_{2e}$	1,652.2	7,236.4
12	Heater No. 1	Benzene	3.0E-05	0.01
	13 MMBtu/hr	Beryllium	2.0E-07	0.01
		Cadmium	2.0E-05	0.01
		Chromium	2.0E-05	0.01
		Formaldehyde	0.01	0.01
		Hexane	0.03	0.11
I		POM (including Naphthalene)	9.0E-06	0.01
		Toluene	0.01	0.01
		VOC	0.5	0.2
i		Benzene	6.0E-04	0.01
		Ethylbenzene	0.01	0.01
14	Distillate/Naphtha	Hexane	0.01	0.01
	Loading Rack <sup>1</sup>	POM (including Naphthalene)	8.0E-05	0.01
'		Toluene	0.01	0.01
		Xylene	0.01	0.01
		VOC	0.1	0.2
	Asphalt/Black Oil Truck Loading Rack No.1	Benzene	3.0E-04	0.01
		Ethylbenzene	0.07	0.07
15		Hexane	0.01	0.01
		POM (including Naphthalene)	1.0E-04	0.01
		Toluene	0.10	0.09
		Xylene	0.10	0.09
		VOC	0.1	0.2
16	Asphalt/Black Oil Truck Loading Rack No.2	Benzene	3.0E-04	0.01
		Ethylbenzene	0.07	0.07
		Hexane	0.01	0.01
		POM (including Naphthalene)	1.0E-04	0.01
		Toluene	0.10	0.09
		Xylene	0.10	0.09

EMISSION SUMMARY				
Source	Description	Pollutant	Emission Rates	
Number	er Beschiphon Tondant	lb/hr	tpy	
		VOC	0.1	0.3
		Benzene	4.0E-04	0.01
	I 1 O'IT 1	Ethylbenzene	0.01	0.01
17	Lube Oil Truck	Hexane	0.01	0.01
	Loading Rack	POM (including Naphthalene)	6.0E-06	0.01
		Toluene	0.01	0.01
		Xylene	0.01	0.01
		VOC	0.1	
		Benzene	4.0E-04	
		Ethylbenzene	0.01	
18	Lube Oil Truck	Hexane	0.01	
	Loading Rack	POM (including Naphthalene)	6.0E-06	
		Toluene	0.01	
		Xylene	0.01	
		VOC	0.2	***
		Benzene	7.0E-04	
		Ethylbenzene	0.01	
21	Lube Oil Rail Car	Hexane	0.01	
-1	Loading Rack	POM (including Naphthalene)	2.0E-04	
,		Toluene	0.01	
		Xylene	0.01	
		VOC	19.9	87.2
		Benzene	0.06	0.26
		Ethylbenzene	0.26	1.14
		Hexane	0.68	2.94
23	Fugitive Emissions	POM (including Naphthalene)	0.005	0.02
		Toluene	0.42	1.83
		2,2,4-Trimethylpentane	0.02	0.08
		Xylene	0.40	1.73
		VOC	0.1	0.2
		Acetone	0.01	0.02
		Benzene	7.0E-04	0.01
24	Wastewater	Ethylbenzene	0.01	0.01
- '	Emissions	POM (including Naphthalene)	8.0E-04	0.01
		Toluene	0.04	0.16
		Xylene	0.01	0.01

	EMISSION SUMMARY			
Source	Description	Pollutant	Emissio	on Rates
Number	Description	Tonumit	lb/hr	tpy
		PM	0.4	1.5
		$PM_{10}$	0.4	1.5
		$SO_2$	0.1	0.2
		VOC	0.5	2.0
		CO	2.8	12.3
25	Coconomation Unit	NOx	9.1	39.9
23	Cogeneration Unit	$CO_{2e}$	5,919.9	25,925.3
		Benzene	7.0E-04	0.01
		Ethylbenzene	0.01	0.81
		POM (including Naphthalene)	7.0E-05	0.01
		Toluene	0.01	0.01
		Xylene	0.01	0.01
		PM	0.7	1.7
		$PM_{10}$	0.7	1.7
		$SO_2$	0.1	0.2
		VOC	0.6	1.2
		СО	7.8	18.4
		$NO_x$	3.7	8.8
	D '1 //4	$CO_{2e}$	11,096.7	26,286.1
26	Boiler #4	Benzene	2.0E-04	0.01
	94 MMBtu/hr	Beryllium	2.0E-06	0.01
		Cadmium	2.0E-04	0.01
		Chromium	2.0E-04	0.01
		Formaldehyde	0.01	0.01
		Hexane	0.17	0.40
		POM (including Naphthalene)	6.0E-05	0.01
		Toluene	0.01	0.01
27a	Refinery Additive Storage Tanks	VOC	2.6	0.3
	Packaging Plant		<del></del>	
27b	Additive Storage	VOC	3.1	0.6
	Tanks	Voc	0.1	
		VOC	0.1	0.1
27-	Asphalt Storage	Benzene	4.0E-05	0.01
27c	Tanks	Ethylbenzene	0.08	0.08
		Toluene	0.10	0.10
		Xylene	0.10	0.10

	EMISSION SUMMARY			
Source	Description	Pollutant	Emissic	n Rates
Number	Description	Tondant	lb/hr	tpy
		VOC	0.1	0.1
		Benzene	3.0E-04	0.01
	D11- O'1 G/	Ethylbenzene	0.02	0.02
27d	Black Oil Storage Tanks	Hexane	0.01	0.01
	ranks	POM (including Naphthalene)	5.0E-05	0.01
		Toluene	0.03	0.03
		Xylene	0.03	0.03
		VOC	3.1	6.4
		Benzene	2.0E-03	0.01
	0 1 03.0	Ethylbenzene	0.01	0.01
27e	Crude Oil Storage	Hexane	0.23	0.46
	Tanks	POM (including Naphthalene)	4.0E-05	0.01
		Toluene	0.02	0.04
		Xylene	0.01	0.02
		VOC	0.7	0.9
		Benzene	1.0E-03	0.01
	17. (17. (17. (11. (	Ethylbenzene	0.01	0.01
27f	Untreated Distillate	Hexane	0.01	0.01
	Storage Tanks	POM (including Naphthalene)	2.0E-04	0.01
		Toluene	0.01	0.01
		Xylene	0.01	0.01
		VOC	33.1	0.2
		Benzene	5.0E-04	0.01
		Ethylbenzene	0.03	0.01
07	Gasoline Storage	Hexane	1.48	0.01
27g	Tank	MBTE	1.21	0.01
		POM (including Naphthalene)	2.0E-04	0.01
		2,2,4-Trimethylpentane	0.32	0.01
		Styrene	0.03	0.01
		VOC	0.4	0.4
		Benzene	2.0E-03	0.01
	11.44.11.1.01	Ethylbenzene	0.01	0.01
27h	Untreated Lube Oil	Hexane	0.02	0.02
	Storage Tanks	POM (including Naphthalene)	3.0E-04	0.01
		Toluene	0.01	0.01
		Xylene	0.01	0.01

	EMISSION SUMMARY			
Source	Description	Pollutant	Emission Rates	
Number	Description	1 officialit	lb/hr	tpy
		VOC	0.2	0.2
		Benzene	7.0E-04	0.01
	Refinery Treated	Ethylbenzene	0.01	0.01
27i	Lube Oil Storage	Hexane	0.01	0.01
	Tanks	POM (including Naphthalene)	2.0E-04	0.01
		Toluene	0.01	0.01
		Xylene	0.01	0.01
		VOC	0.2	0.2
		Benzene	7.0E-04	0.01
	D1 -1 -1 -1 -1	Ethylbenzene	0.01	0.01
27j	Packaging Plant Lube	Hexane	0.01	0.01
	Oil Storage Tanks	POM (including Naphthalene)	2.0E-04	0.01
		Toluene	0.01	0.01
		Xylene	0.01	0.01
		VOC	0.8	0.4
		Benzene	4.0E-03	0.01
		Ethylbenzene	0.01	0.01
27k	Reclaim Oil Storage	Hexane	0.06	0.01
2/K	Tanks	POM (including Naphthalene)	7.0E-05	0.01
1		2,2,4-Trimethylpentane	0.01	0.01
		Toluene	0.03	0.01
		Xylene	0.01	0.01
		VOC	0.8	0.7
		Benzene	9.0E-04	0.01
	Treated Distillate	Ethylbenzene	0.01	0.01
271	Storage Tanks	Hexane	0.01	0.01
	Storage ranks	POM (including Naphthalene)	2.0E-04	0.01
·		Toluene	0.01	0.01
		Xylene	0.01	0.01
		VOC	11.9	19.7
,		Benzene	0.006	0.01
		Ethylbenzene	0.01	0.02
28	Sandyland Storage	Hexane	0.85	1.40
20	Tanks	POM (including Naphthalene)	2.0E-04	0.01
		2,2,4-Trimethylpentane	0.03	0.05
	i	Toluene	0.06	0.10
		Xylene	0.03	0.05

	EMISSION SUMMARY			
Source	Danasintian	Pollutant	Emissic	n Rates
Number	Description	Tonutant	lb/hr	tpy
	Miller's Storage Tanks	VOC Benzene Ethylbenzene	25.0 0.02 0.02	***
29	(2 Total Crude Oil Tanks) Tanks #114 and #115	Hexane POM (including Naphthalene) 2,2,4-Trimethylpentane Toluene Xylene	1.78 3.0E-04 0.07 0.12 0.06	****
32	Asphalt/Black Oil Tank Car Loading Rack	VOC Benzene Ethylbenzene Hexane POM (including Naphthalene) Toluene Xylene	0.1 3.0E-04 0.07 0.01 5.0E-05 0.10 0.10	0.2 0.01 0.07 0.01 0.01 0.09 0.09
33	Lube Oil and Distillate Railcar Loading and Additive Railcar Unloading	VOC Benzene Ethylbenzene Hexane POM (including Naphthalene) Toluene Xylene	0.7 2.0E-03 0.01 0.01 2.0E-04 0.01 0.01	0.6 0.01 0.01 0.01 0.01 0.01
34	Packaging Plant Lube Oil Loading/Unloading Docks	VOC Benzene Ethylbenzene Hexane POM (including Naphthalene) Toluene Xylene	0.2 7.0E-04 0.01 0.01 2.0E-04 0.01 0.01	0.3 0.01 0.01 0.01 0.01 0.01 0.01

	EMISSION SUMMARY				
Source	Dominica	Pollutant Emission R		on Rates	
Number	Description	Pollutant	lb/hr	tpy	
		PM	0.1	0.4	
		$PM_{10}$	0.1	0.4	
		$SO_2$	0.4	1.5	
		VOC	0.1	0.3	
		СО	1.0	4.1	
		NOx	0.6	2.4	
	I who Change Heaten	$CO_{2e}$	1,310.9	5,741.6	
36	Lube Charge Heater	Benzene	3.0E-05	0.01	
	No. 1	Beryllium	2.0E-07	0.01	
		Cadmium	2.0E-05	0.01	
		Chromium	2.0E-05	0.01	
		Formaldehyde	0.01	0.01	
		Hexane	0.02	0.09	
		POM (including Naphthalene)	8.0E-06	0.01	
		Toluene	0.01	0.01	
- "		PM	0.2	0.8	
		$PM_{10}$	0.2	0.8	
		$\mathrm{SO}_2$	0.1	0.1	
		VOC	0.1	0.3	
		CO	0.3	1.0	
		NOx	1.3	5.5	
	Hydrogen Reformer	$\mathrm{CO}_{2\mathrm{e}}$	18,591.7	81,431.5	
37	No. 2	Benzene	8.0E-05	0.01	
	No. 2	Beryllium	5.0E-07	0.01	
		Cadmium	4.0E-05	0.01	
		Chromium	5.0E-05	0.01	
		Formaldehyde	0.01	0.02	
		Hexane	0.07	0.28	
		POM (including Naphthalene)	3.0E-05	0.01	
		Toluene	0.01	0.01	

	EMISSION SUMMARY				
Source	Description	Pollutant		on Rates	
Number	Description	Pollutant	lb/hr	tpy	
· <del></del>		PM	0.3	0.7	
		$PM_{10}$	0.3	0.7	
		$SO_2$	1.1	2.6	
		VOC	0.2	0.5	
		CO	2.9	7.1	
		NOx	1.8	4.2	
	I who Chausa Hastan	$\mathrm{CO}_{2\mathrm{e}}$	4,118.6	10,022.4	
38	Lube Charge Heater	Benzene	8.0E-05	0.01	
	No. 2	Beryllium	5.0E-07	0.01	
		Cadmium	4.0E-05	0.01	
		Chromium	5.0E-05	0.01	
		Formaldehyde	0.01	0.01	
		Hexane	0.07	0.16	
		POM (including Naphthalene)	3.0E-05	0.01	
		Toluene	0.01	0.01	
		PM	0.1	0.2	
		$PM_{10}$	0.1	0.2	
		$SO_2$	0.2	0.7	
		VOC	0.1	0.2	
		CO	0.6	1.8	
		NOx	0.7	2.1	
	Ctuinn on Chanca	$\mathrm{CO}_{2\mathrm{e}}$	804.9	2,478.6	
39	Stripper Charge	Benzene	2.0E-05	0.01	
	Heater No. 2	Beryllium	9.0E-08	0.01	
		Cadmium	8.0E-07	0.01	
		Chromium	1.0E-05	0.01	
		Formaldehyde	0.01	0.01	
		Hexane	0.02	0.04	
		POM (including Naphthalene)	5.0E-06	0.01	
		Toluene	0.01	0.01	

EMISSION SUMMARY				
Source	Description	Pollutant	Emissic	on Rates
Number	Description	Tonutant	lb/hr	tpy
		PM	0.1	0.1
		$PM_{10}$	0.1	0.1
		$SO_2$	0.1	0.4
		VOC	0.1	0.1
		СО	0.3	1.0
		NOx	0.2	0.6
		$CO_{2e}$	300.1	1,314.3
40	Flare	Benzene	6.0E-06	0.01
		Beryllium	3.0E-08	0.01
		Cadmium	3.0E-06	0.01
		Chromium	4.0E-06	0.01
		Formaldehyde	0.01	0.01
		Hexane	0.01	0.02
		POM (including Naphthalene)	2.0E-06	0.01
		Toluene	0.01	0.01
		PM	0.1	0.1
		$PM_{10}$	0.1	0.1
		$SO_2$	0.7	0.1
		VOC	0.2	0.1
		CO	1.1	0.1
41	Fire Pump Engine	NOx	1.8	0.1
41	No.1	$CO_{2e}$	376.1	18.9
		Benzene	3.0E-03	0.01
		Formaldehyde	0.01	0.01
		POM (including Naphthalene)	2.0E-04	0.01
		Toluene	0.01	0.01
		Xylene	0.01	0.01
		PM	0.4	0.1
		$PM_{10}$	0.4	0.1
		$SO_2$	0.4	0.1
		VOC	0.5	0.1
		CO	1.2	0.1
42	Fire Pump Engine	NOx	5.3	0.3
12	No. 2	$\mathrm{CO}_{2\mathrm{e}}$	194.4	9.8
		Benzene	2.0E-03	0.01
		Formaldehyde	0.01	0.01
		POM (including Naphthalene)	1.0E-04	0.01
		Toluene	0.01	0.01
		Xylene	0.01	0.01

Permit #: 1227-AOP-R0

	EMISSION SUMMARY			
Source	Description	Pollutant	Emission Rates	
Number	Description	1 Onutant	lb/hr	tpy
		PM	0.1	0.1
		$PM_{10}$	0.1	0.1
		$SO_2$	0.7	0.1
		VOC	0.2	0.1
		CO	1.1	0.1
43	Fire Pump Engine	NOx	1.8	0.1
40	No. 3	$\mathrm{CO}_{2\mathrm{e}}$	376.1	18.9
		Benzene	3.0E-03	0.01
		Formaldehyde	0.01	0.01
İ		POM (including Naphthalene)	2.0E-04	0.01
		Toluene	0.01	0.01
		Xylene	0.01	0.01
		VOC	0.1	0.1
		Benzene	6.68E-05	0.01
	Packaging Plant	Ethylbenzene	0.01	0.01
44	Railcar Loading and	Hexane	0.01	0.01
	Unloading Rack	POM (including Naphthalene)	5.72E-05	0.01
		Toluene	0.01	0.01
		Xylene	0.01	0.01
		VOC	0.1	0.1
		Benzene	2.79E-04	0.01
	Packaging Plant	Ethylbenzene	0.01	0.01
45	Truck Loading and	Hexane	0.01	0.01
	Unloading Rack	POM (including Naphthalene)	5.89E-05	0.01
	_	Toluene	0.01	0.01
		Xylene	0.01	0.01
46	Haul Roads	PM	1.0	3.5
40	Haul Koaus	$PM_{10}$	0.3	0.8

<sup>\*</sup>HAPs included in the VOC or PM totals. Other HAPs are not included in any other totals unless specifically stated.

<sup>\*\*</sup>Air Contaminants such as ammonia, acetone, and certain halogenated solvents are not VOCs or HAPs.

<sup>&</sup>lt;sup>1</sup>No naphtha loading emissions. Naphtha loading emissions are routed to one of Tanks #7130, #7131, or #7132.

<sup>\*\*\*</sup>Combined total for SN-17, SN-18, and SN-21.

<sup>\*\*\*\*</sup>Combined total for SN-28 and SN-29.

Permit #: 1227-AOP-R0

AFIN: 70-00039

#### SECTION III: PERMIT HISTORY

- Issued on December 9, 1991, this was the first operating permit for CORC. This permit included the inclusion of a recently installed naphtha storage tank.
- This modification, issued on July 14, 1992, covered the installation of a lube-oil stripper/reboiler/heater at the facility.
- 1227-AR-2 Issued on November 20, 1992, this permit allowed the installation of a replacement boiler. The installation of the replacement boiler classified the facility as a major source subject to Title V permitting since NO<sub>x</sub> emissions exceeded 100 tons per year.
- This permit was issued on August 5, 1997 due to an emissions inventory that discovered that the facility did not have actual emissions greater than the major source threshold. Therefore, Cross Oil Refining and Marketing, Inc. was removed from major source status. Additionally, a cogeneration unit and the #4 boiler were added as sources at the facility.
- This modification was issued on June 29, 2000 and covers the relocation of a 94.3 MMBTU/hr natural gas fired boiler to the facility. Several boilers at the plant had reached the end of their useful life. This new boiler incorporates a low NO<sub>x</sub> burner and flue gas recirculation to minimize emissions. Additionally, it was planned that a duct burner would work in conjunction with a cogeneration unit, but the duct burner was never installed and is being removed from the permit and the cogeneration unit calculations adjusted accordingly. In order for CORC to install the duct burner, a new application must be submitted. Also, the existing #3 Boiler (SN-06) has been retired from operation and so the emissions from this source have been removed.
- 1227-AR-5 This permit was issued on April 29, 2002 and addressed a proposal to make the following changes to some storage tanks:
  - Two tanks which stored lube oil product were destroyed in a fire in 1999 and have not yet been replaced. The refinery plans to move two existing identical tanks to replace these tanks. These tanks will be designated as #328 and #329. Both tanks have a capacity of 1,000 barrels each (42,000 gallons) and will be subject to the record keeping provisions of 40 CFR Part 60, Subpart Kb, since they will store organic liquids;
  - Two tanks (#330 and #331), which will store lube oil product, are planned for installation. The tanks have a capacity of 10,000 barrels each (420,000 gallons) and will be subject to record keeping provisions of 40 CFR 60, Subpart Kb, since they will store organic liquids;
  - Two tanks (#332 and #333), which will store lube oil product, are planned for installation. The tanks have a capacity of 500 barrels each (21,000 gallon) and will be subject to the record keeping provisions of 40 CFR 60, Subpart Kb, since they will store organic liquids; and

Permit #: 1227-AOP-R0

AFIN: 70-00039

• One existing tank (#284), which is currently permitted to store lube oil, will be removed from service.

The total emissions VOC from the six tanks amounted to less than 0.02 tons annually.

- 1227-AR-6 This permit was issued on August 2, 2002 and addressed the following modifications to the facility:
  - Tank #329 was recently permitted as a lube oil storage tank. This tank will be equipped with an internal floating roof and will store naphtha. The tank will be subject to the provisions of NSPS Subpart Kb. A floating roof meeting the requirements of 40 CFR 60 60.112b (a) (1) will be installed;
  - The existing naphtha tank #206 will be converted to a lube oil storage tank. It was constructed in 1980 and will not be modified with this project. Therefore, the tank will not be subject to NSPS Subpart Kb after the change of service;
  - Tanks #291 and #292 will be changing service from diesel to lube oil storage. The tanks were constructed in 1980. Therefore, the tanks will not be subject to NSPS Subpart Kb after the change of service; and
  - Tank #113 is currently permitted as a crude oil storage tank subject to NSPS
     Subpart Ka. It will be changing service to store Cross Oil's B Series lube oil (a
     mixture of lube oil and diesel). The tank was constructed in 1980. Therefore, the
     tank will not be subject to NSPS Subpart Kb (or NSPS Subpart Ka due to the low
     vapor pressure of the lube oil) after the change of service.

The above changes in tank service resulted in a decrease in VOC emissions of 2.9 tons per year. Without considering the reduction in emissions due to the change in service of the tanks, the total increase associated with this project is 0.74 tons VOC per year.

- 1227-AR-7 This permit was issued on October 29, 2002 and addressed the following modifications to the permit:
  - Addition of six tanks (001 through 006), which will store lube oil product, are
    planned for installation. The tanks have a capacity of 15,250 gallons each and
    will be subject to the record keeping provisions of 40 CFR 60, Subpart Kb, since
    they will store organic liquids; and
  - Addition of three tanks (007 through 009), which will store lube oil product, are planned for installation. The tanks have a capacity of 2,000 gallons each and will not be subject to the record keeping provisions of 40 CFR 60, Subpart K.

The above changes resulted in an increase of VOC emissions of 0.2 tons per year.

Permit #: 1227-AOP-R0

- This permit was issued on May 30, 2003 and allowed the facility to modify its existing permitted emission rates based upon emission factors, physical property data, facility operating conditions, and revised emissions modeling. In addition, the facility proposed to include hazardous air pollutant (HAP) emissions, which were not included in permit 1227-AR-7, to permit emissions from offsite storage tanks, and to correct opacity limits. No production increases were proposed. The proposed changes resulted in an increase of 0.7 tons per year of SO<sub>2</sub> emissions, 31.0 tons per year of CO emissions, and 15.8 tons per year of HAP emissions.
- 1227-AR-9 This permit was issued on April 19, 2004, and it allowed the facility to install two new 3,500 gallon lube oil storage tanks (#010p and #011p). The proposed change resulted in no production and negligible annual emissions increases.
- This permit was issued September 28, 2004. CORC's proposal included the installation of one new 21,000 gallon reclaimed oil storage tank, one new 42,000 gallon reclaimed oil storage tank, and six new 16,800 gallon lube oil storage tanks. In addition to installing the new storage tanks, Cross Oil requested to remove #500, redesignate #332 as #500, and to re-designate #312 and #333 as #512 and #513, respectively.
- This permit was issued April 11, 2005. The permit revision contained the following changes: converted two tanks containing asphalt to lube oil (Tank #223 and #224); added a new lube oil tank (Tank #331); converted a tank currently containing lube oil to naphtha (Tank #312); corrected the current tank numbering by shifting Tank #012P though #017P each up one number, resulting in Tank #013P through #018P; added two new lube oil tanks (Tanks #012P and #019P of SN-27); added a seasonal 50 horsepower (0.125 MMBTU/hr) low pressure boiler as an insignificant activity; added a pre-heat lube charge heater, with a design rating of 6.0 MMBTU/hr (SN-30) with low NO<sub>x</sub> burners; added 7 heat exchangers (no source number) to the process in order to increase efficiency and reduce reliance on the crude heaters; and removed the crude oil throughput limit. Total annual emission increases were 0.2 ton/yr PM/PM<sub>10</sub>, 0.1 ton/yr SO<sub>2</sub>, 0.4 ton/yr VOC, 1.3 ton/yr NO<sub>x</sub>, and 2.2 ton/yr CO.
- This permit was issued on September 23, 2005. The facility modified their permit in order to increase the annual asphalt throughput at the Blow Still Incinerator Waste Heat Boiler (SN-04) and to remove the testing requirements for SO<sub>2</sub> at the Cogeneration Unit (SN-25), which is a natural gas source. Annual particulate emissions increased by 14.6 tons/year as a result of the asphalt throughput.

Permit #: 1227-AOP-R0

AFIN: 70-00039

- This permit was issued October 24, 2006. The permitting action was necessary to: install five 700 barrel (29,400 gallons) lube oil storage tanks at SN-27 (Tanks #020p, #022p, #023p, #024p, and #025p), install one 500 barrel (21,000 gallons) lube oil storage tank at SN-27 (Tank #021p), install one 5,000 barrel (210,000 gallons) lube oil storage tank at SN-27 (Tank #030p), and remove the 1000 barrel (42,000 gallon) reclaimed oil storage tank and replace it with a 400 barrel (16,800 gallon) reclaimed oil storage tank at SN-27 (Tank #503). In addition, the permitted HAP lb/hr emission rate limits were corrected for the Diesel/Naphtha/Kerosene Loading Rack (SN-14), and various typographical errors were corrected. The typographical errors included adding SN-23's VOC emissions back to Specific Condition #1; adding naphthalene to the Total Allowable Emissions Table; and correcting the VOC, NO<sub>x</sub>, and toluene total emission rates in the Total Allowable Emissions Table. The total permitted annual emission rate increases included: 0.1 tpy NO<sub>x</sub>, 0.01 tpy cumene, 0.01 tpy phenol, and 0.01 tpy toluene.
- 1227-AR-14 This permit was issued August 27, 2007. The permit revision contained the following changes:
  - Installation of Six (6) lube oil storage tanks at SN-27 with the following storage capacities: Tanks #026p (1,182 barrel), #027p (1,049 barrel), #028p (862 barrel), #029p (1,182 barrel), #031p (1,182 barrel), #032p (1,049 barrel);
  - Installation of one 1,000 barrel distillate oil storage tank (Tank #319) at SN-27;
  - Removing of the 1000 barrel reclaimed oil storage tank and replacing it with a 400 barrel reclaimed oil storage tank at SN-27 Tank #504;
  - Installation of four (4) 280 barrel reclaimed oil storage tanks at SN-27; Tanks #514, #515, #516, and #517;
  - Add an asphalt tank car loading rack;
  - Add a distillate lube oil loading rack; and
  - Revise insignificant activity list in the existing permit.

Cross Oil also submitted a DeMinimis application to perform the necessary piping modifications to allow for pipeline quality natural gas to be fired at Blow Still Incinerator Waste Heat Boiler (SN-04), Lube Stripper Reboiler (SN-12), and Lube Charge Heater (SN-30); to include use of mixed gas at the Crude Charge Heater (SN-01), Vacuum Tower Charge Heater (SN-02), the Asphalt Below Charge Heater (SN-03), and the Lube Precharge Heater (SN-07); to make necessary improvement to NASH plant to ensure the sulfur content in the waste gas produced onsite remain below 0.1 gr/dscf; and to change the source descriptions for SN-07 to the "Lube Precharge Heater" and SN-30 to the "Lube Charge Heater".

The permitted emission increases were due to these modifications: 3.8 tons per year (tpy) PM/PM10, 5.8 tpy SO<sub>2</sub>, 1.1 tpy VOC, 0.01 tpy 2,2,4-trimethylpentane, 0.18 tpy benzene, 0.01 tpy cumene, 0.01 tpy ethylbenzene, 0.34 tpy hexane, 0.06 tpy toluene, and 0.04 tpy xylene.

- 1227-AR-15 This permit was issued January 26, 2009. This permitting action was necessary to:
  - Add two (2) 400 barrel asphalt plasticizer tanks #227 and #228 at SN-27,

Permit #: 1227-AOP-R0

AFIN: 70-00039

- Replace three (3) 280 barrel reclaimed oil storage tanks #514, #515, and #516 (previously permitted) with three (3) reclaimed oil tanks of different size (400 barrel) and throughput at SN-27,
- Remove the 280 barrel reclaimed oil storage tank #517at SN-27,
- Add a lube oil and additive loading/unloading rack (SN-33), and
- Add a lube oil packaging operation into insignificant activities list.

The permitted emission increased due to this De Minimis modification is 0.2 tons per year (tpy) VOC. The new tanks #227, #228, #514, #515, and #516 are not subject to 40 CFR 60 Subpart Ka and Kb.

Also, Cross Oil submitted another DeMinimis application on September 3, 2008 to modify its existing minor source to address the throughput at the Distillate Lube Oil Truck Loading Rack and associated storage tanks:

- Increase the throughput at the distillate lube oil storage tank (Tank # 319) from 1,302,000 gal/yr to 2,730,000 gal/yr.
- Increase the throughput at the Distillate Lube Oil loading Rack SN-31.
- Add a condition to address the throughput and tracking for the gasoline tank at SN-27.
- Record keeping requirements for SN-31 and SN-32 have been added to the permit.

The permitted emissions for SN-31 increased by: 1.2 tons per year (tpy) VOC, Benzene 0.15 tpy, Ethylbenzene 0.02 tpy, Hexane 0.3 tpy, Toluene 0.05, and Xylene 0.03 tpy.

Additionally, Cross Oil submitted a summary of all of the tanks with the revised capacities, turnovers, and emissions for Onsite Storage Tanks SN-27, Sandyland Storage Tanks SN-28, and Miller's Storage Tanks SN-29 by email dated June 26, 2008. Due to this revision the emission limits for these sources were decreased. The overall permitted emissions decreased due to this DeMinimis modification by: 16.2 tpy VOC, 2,2,4-Trimethylpentane 0.06 tpy, Benzene 0.6 tpy, Cumene 0.02 tpy, Ethylbenzene 0.04 tpy, Hexane 1.02 tpy, Toluene 0.48, and Xylene 0.17 tpy.

1227-AR-16 This permit was issued March 27, 2009. This permitting action was necessary to:

- Install fourteen (14) onsite lube oil storage tanks #229, #230, #231, #232, #233, #332, #333, #334, #335, #336, #337, #338, #339, and #340 at SN-27.
- Allow the distillate to be loaded at the Asphalt Tank Car Loading Rack SN-32.

There was an increase in lube oil throughput as a result of the additional storage tanks because the tanks were being installed to allow for reduced throughput at other lube oil tanks. The introduction of distillate to SN-32 resulted in an increase of 1,500,000 gallons per year of distillate. The permitted emissions increased 1.3 ton per year (tpy) of VOC, 0.02 tpy of 2,2,4-Trimethylpentane, 0.14 tpy of Benzene, 0.01 tpy Cumene, 0.02 tpy of Hexane, 0.07 tpy of Toluene, and 0.03 tpy of Xylene.

Permit #: 1227-AOP-R0

AFIN: 70-00039

- 1227-AR-17 This permit was issued on December 21, 2009. This de minimis change was necessary to:
  - Install an additional onsite distillate storage Tank #328 and lube oil storage Tanks #P033 through #P059 at SN-27
  - Rename the Lube Stripper Reboiler (SN-12) to Lube Stripper Charge Heater (SN-12)
  - Convert the existing lube oil storage Tanks #113 to a distillate storage tank at SN-27
  - Install an offsite distillate Tank #116 at SN-29
  - Install a new lube oil loading/unloading dock (SN-34)
  - Change the permittee ownership and name from Cross Oil Refining and Marketing, Inc. to Martin Operating Partnership L.P.

The above listed changes may have resulted in potential emission rate increases of 0.5 ton per year (tpy) of VOC, 0.05 tpy of benzene, 0.1 tpy of hexane, 0.02 tpy of toluene, and 0.01 tpy of xylene.

1227-AR-18 This permit was issued on May 9, 2011. This permitting action was necessary to: install three new lube oil storage tanks (SN-35: Tanks #246, #341, and #342), change one existing tank from lube oil service to diesel service (SN-35: Tank #321), and correct typographical errors in the Total Allowable Emissions table. The total permitted annual emission rate limit changes associated with this de minimis modification included: 0.4 ton per year (tpy) of VOC, 0.0003 tpy of benzene, 0.092 tpy of hexane, 0.001 tpy of toluene, and 0.0012 tpy of xylene.

Permit #: 1227-AOP-R0

AFIN: 70-00039

This permit was issued on January 12, 2012. This permitting action was necessary to: 1227-AR-19 correct the hourly permitted emission rate limits for SN-27, SN-28, and SN-29 based on the maximum hourly throughput; update the annual storage tanks throughputs; modify SN-27: removed existing Tank #100 and install a new 80,000 bbl asphalt storage tank (Tank #100), removed Tank #501 and renamed Tank #514 to Tank #501, removed Tank #502 and renamed Tank #515 to Tank #502, removed Tank #504 and installed one new 28,224 gallon reclaimed oil tank (Tank #504), removed Tank #321 from SN-27 since it is permitted under SN-35, installed three 42,301 gallon naphtha tanks (Tanks #7130, #7131, and #7132), installed one 77,400 gallon heavy condensate tank (Tank #7133), installed four new 31,500 gallon reclaimed oil tanks (Tanks #514, #515, #516, #517), changed Tanks #319 and #320 from HHD storage to lube oil, changed Tanks #312 and #329 from naphtha storage to lube oil, changed Tank #227 from asphalt storage to additive; remove Tank #116 (SN-29); rename the following sources: SN-03 to Asphalt Charge Heater (MOP will not conduct blow still operations), SN-08 to Hydrogen Plant Heater/Reformer, SN-12 to Stripper Charge Heater, SN-14 to Distillate/Naphtha Loading Rack, and SN-32 to Asphalt/Black Oil Tank Car Loading Rack; remove the Blow Still Incinerator Waste Heat Boiler SN-04 (removed from service in May 2011); route emissions from the naphtha loading (SN-14) and heavy condensate loading (SN-31) to Tanks #7130 through #7132 for naphtha and Tank #7133 for heavy condensate instead of venting to the atmosphere; revise emission limits for SN-14, SN-17, SN-18, SN-21, SN-23, SN-27, SN-28, and SN-29 to reflect site specific lube oil and distillate analysis; revise benzene emission limits for asphalt, gasoline, and crude oil storage tanks to reflect site specific analysis; install a 42,301 gallon crude oil tank (bullet tank) at the Sandyland Storage Tanks (SN-28); bubble the annual emission rate limits for the lube oil loading rack sources (SN-17, SN-18, and SN-21); bubble the annual emission limits for the Sandyland Storage Tanks (SN-28) and the Miller's Storage Tanks (SN-29); revise the emission limits at SN-32 to reflect the removal of all distillate lines; replace SN-02; add 40 CFR Part 63, Subpart CCCCCC conditions; revise reportable HAPs throughout the permit; rename the Recycle Water Evaporators in the insignificant activities list to Cooling Tower No. 1 and Cooling Tower No. 2; replace Cooling Tower No. 2 (insignificant activity A-13) with a newer model; add a new asphalt tank heater under insignificant activity A-13; and add a 50% safety factor to the pollutant content limits for SN-35 and change the HAP emission rate limits to correspond. The total permitted annual emission rate limit changes associated with this modification included: decreases of 36.6 ton per year (tpy) PM/PM<sub>10</sub>, 6.3 tpy CO, 8.5 tpy NO<sub>X</sub>, 3.7003 tpy benzene, 0.09 tpy cumene, 2.352 tpy hexane, 0.01 tpy phenol and increases of 0.2 tpy SO<sub>2</sub>, 14.4 tpy VOC, 0.11 tpy beryllium, 0.11 tpy cadmium, 0.11 tpy chromium, 1.38 tpy ethylbenzene, 0.01 tpy methyl tert butyl ether, 0.19 tpy POM, 0.939 tpy toluene, 0.01 tpy 2,2,4-trimethylpentane, and 1.6688 tpy xylene.

Permit #: 1227-AOP-R0

AFIN: 70-00039

1227-AR-20 This permit was issued on May 1, 2012. This permitting action was necessary to: install four new 49,250 gallon lube oil storage tanks (Tank #P060, #P061, #P062, and #P063) at SN-27, and remove Scenario #1 from SN-14, SN-27, and SN-31. The total permitted annual emission rate limits for Scenario #2 did not change with this modification.

1227-AR-21 This permit was issued November 9, 2012. This permitting action was necessary to:

- Increase the black oil production limit (SN-27: Tanks #275 through #278) from 4,872,000 gallons per year to 9,240,000 gallons per year,
- Increase the annual distillate limit for Tank #321 (SN-35) from 15,120,000 gallons per year to 26,219,328 gallons per year,
- Add distillate loading to the railcar loading source (SN-33), and
- Remove Scenario #1.

The total permitted annual emission rate increases associated with this modification included: 0.5 tons per year (tpy) VOC and 0.01 tpy hexane.

1227-AR-22 This permit was issued on May 14, 2013. This permitting action was necessary to add six lube oil storage tanks and one reclaimed oil storage tank to SN-27, to replace six existing bolted storage tanks (#247, #248, #271, #272, #273 and #27s) with three new welded storage tanks (#350, #T-343, and #T-347) in SN-27, to rebuild the existing railcar loading rack (SN-33), to add a new railcar loading rack (SN-46), to add a new Packaging Plant lube oil truck loading rack (SN-47), to add two new wastewater surge tanks (#519, #508, #510) and to remove the existing wastewater surge tank (#507) in SN-27. The permitted emissions increased by 1.2 tpy VOC. The CO<sub>2</sub>e permitted emission limit established for sources affected by this modification was 137,710 tpy.

Permit #: 1227-AOP-R0

AFIN: 70-00039

#### SECTION IV: SPECIFIC CONDITIONS

## SN-01 Crude Charge Heater

#### Source Description

MOP operates a crude unit that processes 7,700 bb/day of crude. The crude unit is equipped with a crude charge heater that heats the crude prior to the vacuum distillation tower. Crude Charge Heater (SN-01) is the charge heater associated with the existing crude unit. The heater has a maximum heat input rate of 32 MMBtu/hr and burns natural gas.

The crude charge heater has the capability of burning refinery off gas that is mixed with pipeline quality natural gas. The crude charge heater was installed in 1972 and has not been modified or reconstructed; therefore NSPS Subpart J/Ja does not apply. However, MOP maintains the H<sub>2</sub>S concentration of the mixed gas below the NSPS Subpart J/Ja standard of 0.10 gr/dscf.

### Specific Conditions

1. The permittee shall not exceed the emission rates set forth in the following table. Hourly emissions are based on maximum design heat input rate. The permittee shall demonstrate compliance with annual limits by compliance with Specific Condition #4. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
PM <sub>10</sub>	0.3	1.0
$SO_2$	1.0	3.8
VOC	0.2	0.7
СО	2.7	10.5
$NO_x$	3.2	12.5
$\mathrm{CO}_{2\mathrm{c}}$	3,765.7	15,003.5

2. The permittee shall not exceed the emission rates set forth in the following table. Hourly emissions are based on maximum design heat input rate. The permittee shall demonstrate compliance with annual limits by compliance with Specific Condition #4. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
PM	0.3	1.0

Permit #: 1227-AOP-R0

Pollutant	lb/hr	tpy
Benzene	7.0E-05	0.01
Beryllium	4.0E-07	0.01
Cadmium	0.0001	0.01
Chromium	5.0E-05	0.01
Formaldehyde	0.01	0.01
Hexane	0.06	0.23
POM (including Naphthalene)	2.0E-05	0.01
Toluene	0.01	0.01

- 3. The permittee shall not exceed 5% opacity from Crude Charge Heater (SN-01) as measured by EPA Reference Method 9. Compliance with the opacity limit for SN-01 shall be demonstrated by burning natural gas mixed with refinery off gas. [§18.501 of Regulation 18, and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR Part 52, Subpart E]
- 4. The permittee shall not exceed a throughput of 250 MMscf/year of natural gas mixed with refinery off gas at SN-01 per rolling 12 month period. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]
- 5. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #4. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [Regulation No. 19 §19.705 and 40 CFR Part 52, Subpart E]
- 6. The permittee shall install and maintain a dedicated meter on the natural gas piping which feeds SN-01. [§19.703, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

## SN-02 Vacuum Tower Charge Heater

### Source Description

The Vacuum Tower Charge Heater (SN-02) is the charge heater associated with the existing vacuum distillate tower in line with the existing crude unit. The heater has a maximum heat input rate of 12.6 MMBtu/hr and burns mixed gas (refinery off gas that is mixed with pipeline quality natural gas).

### Specific Conditions

7. The permittee shall not exceed the emission rates set forth in the following table. Hourly emissions are based on maximum design heat input rate. The permittee shall demonstrate compliance with annual limits by compliance with Specific Condition #10 and Plantwide Conditions #14 through #22. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
PM <sub>10</sub>	0.2	0.5
SO <sub>2</sub>	3.2	8.1
VOC	0.3	0.7
СО	0.5	1.3
NO <sub>x</sub>	0.4	0.9
CO <sub>2e</sub>	1,482.7	3,780.9

8. The permittee shall not exceed the emission rates set forth in the following table. Hourly emissions are based on maximum design heat input rate. The permittee shall demonstrate compliance with annual limits by compliance with Specific Condition #10 and Plantwide Conditions #14 through #22. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
PM	0.2	0.5
Benzene	3.0E-0.5	0.01
Beryllium	2.0E-07	0.01
Cadmium	0.0001	0.01
Chromium	2.0E-05	0.01

Permit #: 1227-AOP-R0

Pollutant	lb/hr	tpy
Formaldehyde	0.01	0.01
Hexane	0.03	0.06
POM (including Naphthalene)	8.0E-06	0.01
Toluene	0.01	0.01

- 9. The permittee shall not exceed 5% opacity from Vacuum Tower Heater (SN-02) as measured by EPA Reference Method 9. Compliance with the opacity limit for SN-02 shall be demonstrated by burning natural gas mixed with refinery off gas. [§18.501 of Regulation 18, and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR Part 52, Subpart E]
- 10. The permittee shall not exceed a throughput of 63 MMscf/yr of natural gas mixed with refinery off gas at SN-02 per rolling 12 month period. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]
- 11. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #10. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 12. The permittee shall install and maintain a dedicated meter on the natural gas piping which feed SN-02. [§19.703, A.C.A. §8-4-203 as reference by §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-03 Asphalt Charge Heater

#### Source Description

The bottoms off of the Vacuum Tower are asphalt flux, which is pumped through a series of exchangers to storage and then circulated through the Asphalt Charge Heater (SN-03) to maintain tank temperature. The heater has a maximum heat input rate of 8 MMBtu/hr and burns mixed gas (refinery off gas that is mixed with pipeline quality natural gas). The asphalt charge heater was installed in 1978 and has not been modified or reconstructed; therefore, NSPS Subpart J/Ja does not apply.

### **Specific Conditions**

13. The permittee shall not exceed the emission rates set forth in the following table. Hourly emissions are based on the maximum design heat input rate. The permittee shall demonstrate compliance with the annual emission limits by complying with Specific Condition #16. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
PM <sub>10</sub>	0.1	0.3
SO <sub>2</sub>	0.3	1.1
VOC	0.1	0.2
СО	0.7	2.9
NO <sub>x</sub>	0.8	3.5
CO <sub>2e</sub>	941.4	4,123.3

14. The permittee shall not exceed the emission rates set forth in the following table. Hourly emissions are based on the maximum design heat input rate. The permittee shall demonstrate compliance with the annual emission limits by complying with Specific Condition #16. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
PM	0.1	0.3
Benzene	2.0E-05	0.01
Beryllium	1.0E-07	0.01
Cadmium	9.0E-06	0.01

Permit #: 1227-AOP-R0

Pollutant	lb/hr	tpy
Chromium	2.0E-05	0.01
Formaldehyde	0.01	0.01
Hexane	0.02	0.07
POM (including Naphthalene)	6.0E-06	0.01
Toluene	0.01	0.01

- 15. Visible emissions from SN-03 shall not exceed 5% opacity as measured by EPA Reference Method 9. Compliance with the opacity limit for SN-03 shall be demonstrated by burning natural gas mixed with refinery off gas. [§18.501 of Regulation 18, and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR Part 52, Subpart E]
- 16. The permittee shall not exceed a throughput of 68.71 MMscf/yr of natural gas mixed with refinery off gas at SN-03 per rolling 12 month period. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]
- 17. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #16. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-08 Hydrogen Reformer No.1

#### Source Description

Lube oils produced by the atmospheric and vacuum distillation towers are hydrotreated by passing through the Lube Hydrotreater Units, either LHT1 or LHT2. At the hydrotreaters, lube oils combine with hydrogen to make hydrotreated lube oils. The hydrogen is supplied to LHT1 and LHT2 by a steam/methane reformer. Natural gas is compressed and heated in a preheat exchanger and combined with steam. The natural gas mixture is then charged to the Hydrogen Reformer, where it passes over a nickel catalyst and reacts to produce hydrogen and carbon oxides.

The hydrogen reformer is equipped with a natural gas fired burner that has a maximum heat input rate of 30.0 MMBtu/hr. With the exception of GHG emissions, all emissions occur as a result of the natural gas combustion at the burner. GHG emissions occur from the CO<sub>2</sub> generated during the reforming of the natural gas to make hydrogen.

The reformer burner will burn pipeline quality natural gas that is combined with gas from the pressure swing absorption (PSA) system, where CO<sub>2</sub> and other impurities are removed from the hydrogen. The CO<sub>2</sub> emissions from the combustion of the PSA gas at the burner is accounted for in the GHG emissions. Mixed gas, which is refinery off gas that is mixed with pipeline quality natural gas, will not be burned at the reformer burner.

### **Specific Conditions**

18. The permittee shall not exceed the emission rates set forth in the following table. Hourly emissions are based on the maximum design heat input rate. The permittee shall demonstrate compliance with the annual emission limits by complying with Specific Condition #21. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
PM <sub>10</sub>	0.3	1.0
$SO_2$	0.1	0.1
VOC	0.2	0.8
СО	2.5	10.9
NO <sub>x</sub>	1.3	5.5
CO <sub>2e</sub>	6,245.3	27,354.3

Permit #: 1227-AOP-R0

AFIN: 70-00039

19. The permittee shall not exceed the emission rates set forth in the following table. Hourly and yearly emissions are based on the maximum design heat input rate. The permittee shall demonstrate compliance with the annual emission limits by complying with Specific Condition #21. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
PM	0.3	1.0
Benzene	7.0E-05	0.01
Beryllium	4.0E-07	0.01
Cadmium	4.0E-05	0.01
Chromium	5.0E-05	0.01
Formaldehyde	0.01	0.01
Hexane	0.06	0.24
POM (including Naphthalene)	3.0E-05	0.01
Toluene	0.01	0.01

- 20. Visible emissions from SN-08 shall not exceed 5% opacity as measured by EPA Reference Method 9. Compliance will be demonstrated by only burning natural gas. [§18.501 of Regulation 18, and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR Part 52, Subpart E]
- 21. The permittee shall not exceed a throughput of 257.65 MMscf/yr of natural gas at SN-08 per rolling 12 month period. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]
- 22. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #21. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

## SN-12 Stripper Charge Heater No. 1

### Source Description

The Stripper Charge Heater No. 1 (SN-12) is part of the existing Lube Hydrotreater Unit (LHT1). The lube oil from LHT1 flows to a lube oil stripper, where the remaining hydrogen sulfide is removed by steam stripping. The charge heater supplies heat to the lube oil stripper. The heater has a maximum heat input rate of 14.04 MMBtu/hr.

MOP will burn mixed gas (refinery off gas that is mixed with pipeline quality natural gas) at the stripper charge heater. The stripper charge heater was installed in 1998; therefore, NSPS Subpart J applies to this emission source.

## Specific Conditions

23. The permittee shall not exceed the emission rates set forth in the following table. Hourly and annual emissions are based on maximum design heat input capacity. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #26and Specific Condition #28. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
PM <sub>10</sub>	0.1	0.5
SO <sub>2</sub>	0.5	1.9
VOC	0.1	0.4
СО	1.2	5.1
NO <sub>x</sub>	1.4	6.1
CO <sub>2e</sub>	1,652.2	7,236.4

24. The permittee shall not exceed the emission rates set forth in the following table. Hourly and annual emissions are based on maximum design heat input capacity. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #26. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
PM	0.1	0.5
Benzene	3.0E-05	0.01
Beryllium	2.0E-07	0.01

Permit #: 1227-AOP-R0

AFIN: 70-00039

Pollutant	lb/hr	tpy
Cadmium	2.0E-05	0.01
Chromium	2.0E-05	0.01
Formaldehyde	0.01	0.01
Hexane	0.03	0.11
POM (including Naphthalene)	9.0E-06	0.01
Toluene	0.01	0.01

- 25. Visible emissions from SN-12 shall not exceed 5% opacity as measured by EPA Reference Method 9. Compliance will be demonstrated by only burning natural gas mixed with refinery off gas. [§18.501 of Regulation 18, and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR Part 52, Subpart E]
- 26. The permittee shall not exceed a throughput of 120.58 MMscf/yr of natural gas mixed with refinery off gas at SN-12 per rolling 12 month period. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]
- 27. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #26. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [Regulations 19 §19.705 and 40 CFR Part 52, Subpart E]

#### **NSPS Subpart J**

- 28. The permittee shall not burn any fuel gas that contains H<sub>2</sub>S in excess of 230 mg/dscm (0.10 gr/dscf) on a 3-hour rolling average basis at SN-12. The permittee shall comply with the H<sub>2</sub>S emission limit on or after the day on which the initial performance test, required by 60.8, is completed, but not later than 60 days after achieving maximum production rate at the facility, or 180 days after initial startup, whichever comes first. [§19.304 and 40 CFR §60.104(a)(1)]
- 29. In compliance with §60.104, the permittee shall install, calibrate, maintain and operate a continuous monitoring system for H2S in fuel gases before being burned at SN-12. [§19.304 and 40 CFR §60.105(a)(3)]. [§19.304 and 40 CFR §60.105(a)(3)]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-14 Distillate Loading Rack

Source Description

Distillate is loaded at SN-14.

### **Specific Conditions**

30. The permittee shall not exceed the emission rates set forth in the following table. Hourly emissions are based on pump equipment maximum. The permittee shall demonstrate compliance with annual emissions by compliance with Specific Condition #32. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	0.5	0.2

31. The permittee shall not exceed the emission rates set forth in the following table. Hourly emissions are based on pump equipment maximum. The permittee shall demonstrate compliance with annual emissions by compliance with Specific Condition #32, and Plantwide Condition #9. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant**	lb/hr	tpy
Benzene	6.0E-04	0.01
Ethylbenzene	0.01	0.01
Hexane	0.01	0.01
POM (including Naphthalene)	8.0E-05	0.01
Toluene	0.01	0.01
Xylene	0.01	0.01

<sup>\*\*</sup> No naphtha loading emissions. Naphtha loading emissions are routed to one of Tanks #7130, #7131, or #7132.

32. The permittee shall not exceed a throughput of 26,219,328 gallons of distillate at SN-14 per rolling 12 month period. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

33. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #32. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-15 & SN-16 Asphalt/Black Oil Truck Loading Rack No.1 and Asphalt/Black Oil Truck Loading Rack No. 2

#### Source Description

Asphalt and black oil are loaded into trucks at two loading racks (SN-15 and SN-16).

### **Specific Conditions**

35. The permittee shall not exceed the emission rates set forth in the following table. Hourly emissions are based on pump equipment maximum. The permittee shall demonstrate compliance with annual emissions by compliance with Specific Condition #. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

SN	Pollutant	lb/hr	tpy
15	VOC	0.1	0.2
16	VOC	0.1	0.2

36. The permittee shall not exceed the emission rates set forth in the following table. Hourly emissions are based on pump equipment maximum. The permittee shall demonstrate compliance with annual emissions by compliance with Specific Condition #37 and Plantwide Condition #9. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

SN	Pollutant	lb/hr	tpy
	Benzene	3.0E-04	0.01
	Ethylbenzene	0.07	0.07
	Hexane	0.01	0.01
15	POM (including Naphthalene)	1.0E-04	0.01
	Toluene	0.10	0.09
	Xylene	0.10	0.09
	Benzene	3.0E-04	0.01
	Ethylbenzene	0.07	0.07
	Hexane	0.01	0.01
16	POM (including Naphthalene)	1.0E-04	0.01
	Toluene	0.10	0.09
	Xylene	0.10	0.09

Permit #: 1227-AOP-R0

AFIN: 70-00039

37. The permittee shall not exceed the throughputs listed in the following table. [Regulation 19 §19.705 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Material Throughput (gallons per rolling 12 month pe	
Asphalt	19,987,800
Black Oil	9,240,000

38. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #37. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

SN-17, SN-18, and SN-21 Lube Oil Truck Loading Rack, Lube Oil Truck Loading Rack, and Lube Oil Rail Car Loading Rack

# Source Description

Lube oil is loaded into trucks (SN-17 and SN-18) or onto railcars (SN-21).

## **Specific Conditions**

40. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #42. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

SN	Description	Pollutant	lb/hr	tpy
17	Lube Oil Truck Loading Rack	VOC	0.1	
18	Lube Oil Truck Loading Rack	VOC	0.1	0.3
21	Lube Oil Truck Loading Rack	VOC	0.2	

41. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #42 and Plantwide Condition #9. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

SN	Description	Pollutant	lb/hr	tpy*		
	Lube Oil Truck Loading Rack	Benzene	4.0E-04	0.01		
		Ethylbenzene	0.01	0.01		
		Loading Rack	Lube Oil Truck	Hexane	0.01	0.01
17			POM (including Naphthalene)	6.0E-06	0.01	
			Toluene	Toluene	0.01	0.01
			Xylene	0.01	0.01	

Permit #: 1227-AOP-R0

SN	Description	Pollutant	lb/hr	tpy*		
***		Benzene	4.0E-04			
		Ethylbenzene	0.01			
	Lube Oil Truck	ha Oil Truck Hexane	0.01			
18	Loading Rack	POM (including Naphthalene)	6.0E-06			
	Toluene	Toluene	0.01			
					Xylene	0.01
	21 Lube Oil Rail Car Loading Rack		Benzene	7.0E-04		
		Ethylbenzene	0.01			
		Hexane	0.01			
21		POM (including Naphthalene)	2.0E-04			
		Toluene	0.01			
		Xylene	0.01			

<sup>\*</sup>Combined total for SN-17, SN-18, and SN-21

- 42. The permittee shall not exceed a throughput of 97,885,008 gallons of lube oil at SN-17, SN-18, and SN-21 combined per rolling 12 month period. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]
- 43. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #42. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

## SN-23 Fugitive Emissions

## Source Description

Fugitive Emissions occur through equipment leaks. SN-23 accounts for the entire facility fugitive emissions. Fugitive emissions are based on the number of valves, pump seals, compressor seals, pressure relief valves, connectors, open-ended Lines, and sampling connections.

### Specific Conditions

44. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #46. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	19.9	87.2

The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #46. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Benzene	0.06	0.26
Ethylbenzene	0.26	1.14
Hexane	0.68	2.94
POM (including Naphthalene)	0.005	0.02
Toluene	0.42	1.83
2,2,4-Trimethylpentane	0.02	0.08
Xylene	0.40	1.73

46. The permittee shall perform an annual facility count of valves, pump seals, compressor seals, pressure relief valves, connectors, open-ended lines, and sampling connections in order to modify the permit if necessary for any significant changes in emissions due to changes in piping at the facility. At issuance of this permit, the submitted count is as follows:

Permit #: 1227-AOP-R0

AFIN: 70-00039

Type	Crude Oil	Naphtha	Condensate	Distillate Diesel/Kerosene	Asphalt
Valves	309	94	96	61	72
Pump Seals	32	3	5	7	4
Compressor Seals	0	0	0	0	0
Pressure Relief Valves	1	0	0	0	0
Connectors	684	221	96	217	176
Open-ended Lines	23	0	0	0	0
Sampling Connections	1	0	0	0	0
	Lube Oil	Lube Oil	Lube Oil	Lube Oil	Lube Oil
	No. 2	No. 3	No. 4	No. 6	No. 7
Valves	136	333	268	79	252
Pump Seals	5	9	7	3	6
Compressor Seals	0	0	0	0	0
Pressure Relief Valves	0	0	0	3	0
Connectors	431	1,071	860	224	789
Open-ended Lines	0	0	0	0	0
Sampling Connections	0	0	0	0	0
	Lube Oil No. 9	Lube Oil No. 11	Lube Oil (various) Based on No. 3 Stream	Residue Based on Lube Oil No. 7	Slop Oil Based on Lube Oil No. 7
Valves	152	100	307	14	23
Pump Seals	5	5	18	3	4
Compressor Seals	$\frac{3}{0}$	0	0	0	0
Pressure Relief Valves	0	0	1	0	0
Connectors	457	294	887	45	78
Open-ended Lines	0	0	13	0	0
Sampling Connections	0	0	0	0	0
	Wash Oil Based on 11 Stream	Seal Oil Based on No.3 Stream	Fuel Gas	Mix Gas	Off Gas
Valves	49	0	8	74	149
Pump Seals	3	2	0	0	0
Compressor Seals	0	0	0	0	12
Pressure Relief Valves	0	0	0	3	5
Connectors	118	10	8	217	500
Open-ended Lines	0	0	0	4	2
- Open-ended Lines					

[Regulation No. 19 §19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

47. Records of the annual facility count of valves, pump seals, compressor seals, pressure relief valves, connectors, open-ended lines, and sampling connections shall be maintained on an annual basis. Such records shall be maintained on-site and submitted in accordance with General Provision #7. [Regulation 19 §19.705 and 40 CFR Part 52, Subpart E]

Permit #: 1227-AOP-R0

AFIN: 70-00039

## SN-24 Wastewater Emissions

#### Source Description

All storm water and process water flows to a sump, pumped to surge tanks (tanks 508, 510, and 519), and then routed to the API separator. The oil in the influent wastewater is removed by the API separator, which has an estimated removal efficiency of 99%. The oil drawn off at the API separator is routed to a gun barrel tank for additional oil removal (tanks 500, 509, and 513) and oil removed is routed to the slop oil tanks (emissions from the slop oil tanks have been estimated as part of SN-27). The wastewater that is removed at the API separator is routed to a series of other treatment components including a primary clarifier, DAF tank, a tank with surface aerators, and a series of 4 lagoons.

#### **Specific Conditions**

48. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Conditions #82, #85, #89, #96, #100, #105, #109, #115, #120, #124, #128, #132, and #136. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	0.1	0.2

49. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Conditions #82, #85, #89, #96, #100, #105, #109, #115, #120, #124, #128, #132, and #136. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Acetone	0.01	0.02
Benzene	7.0E-04	0.01
Ethylbenzene	0.01	0.01
POM (including Naphthalene)	8.0E-04	0.01
Toluene	0.04	0.16
Xylene	0.01	0.01

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-25 Cogeneration Unit

#### Source Description

Steam is provided to various parts of the facility by a Cogeneration Unit (SN-25) with a maximum heat input rate of 50.3 MMBtu/hr. The cogeneration unit burns only pipeline quality natural gas; therefore, NSPS Subpart J/Ja does not apply. The cogeneration unit is subject to NSPS Subpart GG for gas turbines.

# Specific Conditions

50. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Conditions #53, #55 and equipment limitations. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
PM <sub>10</sub>	0.4	1.5
SO <sub>2</sub>	0.1	0.2
VOC	0.5	2.0
СО	2.8	12.3
NO <sub>x</sub>	9.1	39.9
CO <sub>2e</sub>	5,919.9	25,925.3

51. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Conditions #53 and equipment limitations. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
PM	0.4	1.5
Benzene	7.0E-04	0.01
Ethylbenzene	0.01	0.81
POM (including Naphthalene)	7.0E-05	0.01
Toluene	0.01	0.01
Xylene	0.01	0.01

Permit #: 1227-AOP-R0

AFIN: 70-00039

Visible emissions from SN-25 shall not exceed 5% opacity as measured by EPA Reference Method 9. Compliance will be demonstrated by burning only natural gas. [§18.501 of Regulation 18, and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR Part 52, Subpart E]

- 53. The permittee shall not use more than 432 MMscf/yr of natural gas per consecutive 12-month period at SN-25. [§19.705 of Regulation #19, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR 70.6]
- 54. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #53. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [Regulation 19 §19.705 and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]

#### **NSPS GG Conditions**

- 55. The cogeneration unit (SN-25) is an affected source of 40 CFR Part 60, Subpart GG (Appendix C) Standards of Performance for Stationary Gas Turbines. [Regulation 19 §19.304 and 40 CFR Part 60, Subpart GG]
- 56. The turbine shall not discharge any gases which contain nitrogen oxides in excess of 209 ppm by volume at 15 percent oxygen on a dry basis. [40 CFR Part 60, Subpart GG, §60.332 (a) (2)]
- 57. The turbine is exempt from Specific Condition #56 when ice fog is deemed a traffic hazard by the owner or operator of the gas turbine. [40 CFR Part 60, Subpart GG, §60.332 (f)]
- 58. Records shall be kept for each period which Specific Condition #57 applies. For each period the ambient conditions existing during the period, the date and time the air pollution control system was deactivated, and the date and time the air pollution control system was reactivated shall be recorded. These records shall be retained for a period of five years, and be available for inspection. [40 CFR Part 60, Subpart GG, §60.334]
- 59. The cogeneration unit shall only be fired with pipeline quality natural gas. [Regulation No. 19 §19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR 70.6]
- 60. The facility shall not discharge into the atmosphere from SN-25 any gases which contain SO<sub>2</sub> in excess of 0.015 percent by volume at 15 percent oxygen and on a dry basis or not burn fuel which contains total sulfur in excess of 0.8 percent by weight. [40 CFR Part 60, Subpart GG, §60.333(a) & (b)]

Permit #: 1227-AOP-R0

AFIN: 70-00039

61. The facility shall install, calibrate, maintain and operate a continuous monitoring system to monitor and record the fuel consumption and the ratio of water or steam to fuel being fired in SN-25. [40 CFR Part 60, Subpart GG, §60.334 (a)]

- 62. The facility shall monitor the steam or water to fuel ration or other parameters during the performance test required under §60.8 to establish acceptable values and ranges. The facility shall develop and keep on-site a parameter monitoring plan which explains the procedures used to document proper operation of the NOx emission controls. [40 CFR Part 60, Subpart GG, §60.334(g)]
- Analysis for fuel sulfur content of the natural gas shall be conducted using one of the approved ASTM reference methods for the measurement of sulfur in gaseous fuels, or an approved alternative method. The approved reference methods are: ASTM D1072-80, 90 (reapproved 1994); ASTM D 3246-81, 92, 96; and ASTM D4468-85 (reapproved 2000); or D6667-01 as referenced in 40 CFR 60.335 (b) (10). [40 CFR Part 60, Subpart GG, 60.335(d)]
- 64. The fuel supply shall be initially sampled daily for a period of two weeks to establish that the pipeline quality natural gas fuel supply is low in sulfur content. [Regulation 19 §19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR 70.6]
- 65. In lieu of complying with Specific Condition #63 and #64, the permittee may elect to not monitor the total sulfur content of the gaseous fuel combusted in the turbine if the gaseous fuel is demonstrated to be low sulfur fuel by obtaining a current, valid purchase contract, tariff sheet or transportation contract for the gaseous fuel, specifying that the maximum total sulfur content of the fuel is 20.0 grains/100 scf or less. [40 CFR Part 60, Subpart GG, §60.334(h)(3)]
- 66. After the monitoring required in Specific Condition #64, sulfur monitoring shall be conducted twice monthly for six months. If this monitoring shows little variability in the fuel sulfur content, and indicates consistent compliance with 40 CFR 60.333, then sulfur monitoring shall be conducted once per quarter for six quarters. [40 CFR Part 60, Subpart GG, §60.334(h)(4)(i)(3)]
- 67. If after the monitoring required in Specific Condition #66, the sulfur content of the fuel shows little variability and, calculated as sulfur dioxide, represents consistent compliance with the sulfur dioxide emission limits specified under 40 CFR 60, Subpart GG, §60.333, sample analysis shall be conducted twice per annum. This monitoring shall be conducted during the first and third quarters of each calendar year. [40 CFR 60, Subpart GG, §60.333]

Permit #: 1227-AOP-R0

AFIN: 70-00039

68. Should any sulfur analysis as required in Specific Condition #64 or #66 indicate noncompliance with 40 CFR 60, Subpart GG, §60.333, the owner or operator shall notify ADEQ of such excess emissions and the custom schedule shall be re-examined. Sulfur monitoring shall be conducted weekly during the interim period when this custom schedule is being re-examined. [Regulation 19 §19.303 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

- 69. If there is a change in fuel supply (supplier), the fuel shall be sampled daily for a period of two weeks to re-establish for the record that the fuel supply is low in sulfur content. If the fuel supply's low sulfur content is re-established, then the custom fuel monitoring schedule can be resumed. [Regulation 19 §19.705, 40 CFR Part 52, Subpart E, and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 70. Records of sample analysis and fuel supply pertinent to this custom schedule shall be retained for a period of three years, and be available for inspection. [Regulation 19 §19.705, 40 CFR Part 52, Subpart E, and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 71. Any one hour period during which the average water-to-fuel ratio, as measured by the continuous monitoring system, falls below the water-to-fuel ratio determined to demonstrate compliance with §60.332 by the performance test required in §60.8 or any period during which the fuel-bound nitrogen of the fuel is greater than the maximum nitrogen content allowed by the fuel-bound nitrogen allowance used during the performance test required in §60.8 shall be recorded. Each record entry shall include the average water-to-fuel ratio, average fuel consumption, ambient conditions gas turbine load, and nitrogen content of the fuel during the period of excess emissions, and the graphs or figures developed under §60.335(a). These records shall be retained for a period of five years, and be available for inspection. [40 CFR Part 60, Subpart GG, §60.334(j)(1)(i)]
- 72. Records shall be kept of any daily period during which the sulfur content of the fuel being fired in the gas turbine exceeds 0.8 percent. These records shall be retained for a period of five years, and be available for inspection. [40 CFR Part 60, Subpart GG, §60.334]
- 73. The permittee shall perform an initial test of NO<sub>x</sub> and SO<sub>2</sub> to verify emissions. EPA Method 20 shall be used to determine the nitrogen oxides and oxygen concentrations. The span values shall be 300 ppm of nitrogen oxide and 21 percent oxygen. NO<sub>x</sub> emissions shall be determined at each of the load conditions specified in §60.335(b). The testing shall be coordinated in advance with the Compliance Inspector Supervisor. This initial test was performed in March 2005. [40 CFR Part 60, Subpart GG, §60.335(b)]

Permit #: 1227-AOP-R0

AFIN: 70-00039

74. The permittee shall test NO<sub>x</sub> emissions once every five years to verify that the unit is operating within permitted limits. The permittee shall utilize the procedure outlined in 40 CFR Part 60, Subpart GG §60.335 and as previously conducted in the initial test required by Specific Condition #73. The testing shall be coordinated in advance with the Compliance Inspector Supervisor. [Regulation 19 §19.702, 40 CFR Part 52, Subpart E and 40 CFR Part 60, Subpart GG, §60.335]

Permit #: 1227-AOP-R0

AFIN: 70-00039

## SN-26 Boiler #4

## Source Description

Steam is provided to various parts of the facility by Boiler #4 (SN-26) with a maximum heat input rate of 94.3 MMBtu/hr. The boiler burns only pipeline quality natural gas.

## **Specific Conditions**

75. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #78 and equipment limitations. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
PM <sub>10</sub>	0.7	1.7
$SO_2$	0.1	0.2
VOC	0.6	1.2
СО	7.8	18.4
NO <sub>x</sub>	3.7	8.8
CO <sub>2e</sub>	11,096.7	26,286.1

76. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #78 and equipment limitations. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
PM	0.7	1.7
Benzene	2.0E-04	0.01
Beryllium	2.0E-06	0.01
Cadmium	2.0E-04	0.01
Chromium	2.0E-04	0.01
Formaldehyde	0.01	0.01
Hexane	0.17	0.40
POM (including Naphthalene)	6.0E-05	0.01
Toluene	0.01	0.01

Permit #: 1227-AOP-R0

AFIN: 70-00039

- 77. Visible emissions may not exceed 5% at SN-26 as measured by EPA Reference Method 9. Compliance will be demonstrated by burning only natural gas. [A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 78. The permittee shall not use more than 438 MMscf/yr of natural gas per consecutive 12-month period at SN-26. [§19.705 of Regulation #19, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR 70.6]
- 79. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #78. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [Regulation 19 §19.705 and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]
- 80. The permittee shall install and maintain dedicated meters on the natural gas piping which feeds the Boiler #4, SN-26. [Regulation 19 §19.703 and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-27a Refinery Additives Storage Tanks

#### Source Description

MOP adds additives to the lube oil, as needed, for quality control purposes. The majority of these additives do not contain VOCs; however MOP has estimated VOC emissions from these additives based on a worst case additive. Additives at the Refinery are stored in 4 storage tanks (Tank 256, 259-261). They do not fill or pull from all 4 additive storage tanks at one time. The additives are added to the lube oil through a chemical injection system. The maximum hourly emissions occur during the filling of one additive storage tank (25 gpm), and pumping from one additive tank to the lube oil processing (25 gpm).

#### **Specific Conditions**

81. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #83. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	2.6	0.3

- 82. The permittee shall not exceed a throughput of 150,000 gallons of refinery additive at SN-27a per rolling 12 month period. [§19.705 of Regulation #19, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR 70.6]
- 83. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #82. For each tank, these records shall include: the gallons per month of material produced for all tanks combined storing that material and the 12 month rolling totals of material produced for all tanks combined storing that material. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705 of Regulation #19 and 40 CFR Part 52]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-27b Packaging Plant Additive Storage Tanks

# Source Description

MOP adds additives to the lube oil, as needed, for quality control purposes. The additives are added to the lube oil through a chemical injection system. The majority of these additives do not contain VOCs; however, MOP has estimated VOC emissions from these additives based on a worse case additive. Additives at the Packaging Plant are stored in 7 storage tanks (P053-P059). They do not fill or pull from all 7 additive storage tanks at one time.

#### Specific Conditions

84. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #85. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	3.1	0.6

- 85. The permittee shall not exceed a throughput of 500,000 gallons of additives at SN-27b per rolling 12 month period. [§19.705 of Regulation #19, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR 70.6]
- 86. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #85. For each tank, these records shall include: the gallons per month of material produced for all tanks combined storing that material and the 12 month rolling totals of material produced for all tanks combined storing that material. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705 of Regulation #19 and 40 CFR Part 52]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-27c Asphalt Storage Tanks

#### Source Description

MOP stores asphalt in 7 storage tanks (Tank 100, 222, 225, 228, and 279-281). The asphalt storage tanks are subject to NSPS Subpart UU. MOP does not fill all 7 asphalt storage tanks at one time. They fill one tank at a time from the asphalt production unit at a rate of 25 gpm. If needed, they may also pump from the storage tank to another at a pump rate of 225 gpm; however, this is not a daily routine of moving asphalt from one tank to another.

#### Specific Conditions

87. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #89. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	0.1	0.1

88. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #89 and Specific Condition #91. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Benzene	4.0E-05	0.01
Ethylbenzene	0.08	0.08
Toluene	0.10	0.10
Xylene	0.10	0.10

89. The permittee shall not exceed a throughput of 19,987,800 gallons of asphalt at SN-27c per rolling 12 month period. [§19.705 of Regulation #19, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

- 90. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #89. For each tank, these records shall include: the gallons per month of material produced for all tanks combined storing that material and the 12 month rolling totals of material produced for all tanks combined storing that material. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705 of Regulation #19 and 40 CFR Part 52]
- 91. The permittee shall sample and analyze the vapor concentration of benzene in the tanks at SN-27c. The permittee shall collect at least one sample from the material. The analyses shall be conducted annually and the results shall be less than the values in the following table. The test methods used shall be approved in advance by the Department. The request for test method approval shall be submitted to the Compliance Inspector Supervisor at least fifteen working days in advance of the test. [Regulation 18 §18.1004 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Source	Material	Vapor Benzene Concentration Limit
		(% by weight)
SN-27c	Asphalt	0.0345

- 92. Within 60 days after achieving the maximum production rate, but not later than 180 days after initial startup of such facility, the permittee shall not cause to be discharged into the atmosphere from any asphalt storage tank exhaust gases with opacity greater than 0 percent, except for one consecutive 15-minute period in any 24-hour period when the transfer lines are being blown for clearing. The control device shall not be bypassed during this 15-minute period. [40 CFR Part §60.472(c)]
- 93. The permittee shall use Method 9 and the procedures in §60.11 to determine opacity at SN-27c. [40 CFR Part 60§60.474(c)(5)]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-27d Black Oil Storage Tanks

#### Source Description

MOP stores black oil in 7 storage tanks (Tanks 231-233, and 275-278) that are all the same size (42,000 gallon capacity); however, MOP does not fill all 7 black oil storage tanks at one time. They fill one tank at a time from the black oil production unit at a rate of 9 gpm. If needed, they may also pump from one storage tank to another at a pump rate of 240 gpm or 175 gpm (different pumping rate for different grades of black oil); however this is not a daily routine of moving black oil from one tank to another.

## Specific Conditions

94. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #96. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	0.1	0.1

95. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #96. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Benzene	3.0E-04	0.01
Ethylbenzene	0.02	0.02
Hexane	0.01	0.01
POM (including Naphthalene)	5.0E-05	0.01
Toluene	0.03	0.03
Xylene	0.03	0.03

96. The permittee shall not exceed a throughput of 9,240,000 gallons of black oil at SN-27d per rolling 12 month period. [§19.705 of Regulation #19, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

97. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #96. For each tank, these records shall include: the gallons per month of material produced for all tanks combined storing that material and the 12 month rolling totals of material produced for all tanks combined storing that material. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705 of Regulation #19 and 40 CFR Part 52]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-27e Crude Oil Storage Tank

## Source Description

MOP processes crude oil into naphtha, distillate, lube oils, and asphalt. The crude oil is stored in one large storage tank, Tank 109, with a capacity of 1,260,000 gal. Crude oil from Sandyland storage tanks (SN-28) is being pumped to Tank 109, while crude is continuously being pumped to the refinery for processing.

## Specific Conditions

98. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #100. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	3.1	6.4

99. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #100 and Plantwide Condition #9. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Benzene	2.0E-03	0.01
Ethylbenzene	0.01	0.01
Hexane	0.23	0.46
POM (including Naphthalene)	4.0E-05	0.01
Toluene	0.02	0.04
Xylene	0.01	0.02

100. The permittee shall not exceed a throughput of 118,041,000 gallons of crude oil at SN-27e per rolling 12 month period. [§19.705 of Regulation #19, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

101. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #100. For each source and each material, these records shall include: the gallons per month of material produced for all tanks combined storing that material and the 12 month rolling totals of material produced for all tanks combined storing that material. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705 of Regulation #19 and 40 CFR Part 52]

102. The permittee shall sample and analyze the vapor concentration of benzene in the tanks at SN-27e. The permittee shall collect at least one sample from the material. The analyses shall be conducted annually and the results shall be less than the values in the following table. The test methods used shall be approved in advance by the Department. The request for test method approval shall be submitted to the Compliance Inspector Supervisor at least fifteen working days in advance of the test. [Regulation 18 §18.1004 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Source	Material	Vapor Benzene Concentration Limit
		(% by weight)
SN-27e	Crude Oil	0.0465

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-27f Untreated Distillate Storage Tank

#### Source Description

MOP stores untreated distillate in two storage tanks (Tanks 120 and Tank 321). MOP fills only one tank at a time from the distillate production unit at a rate of 50 gpm. From the storage tank they feed untreated distillate to the hydrotreater at a rate of 375 gpm.

# Specific Conditions

103. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #105. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	0.7	0.9

The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #105 and Plantwide Condition #9. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Benzene	1.0E-03	0.01
Ethylbenzene	0.01	0.01
Hexane	0.01	0.01
POM (including Naphthalene)	2.0E-04	0.01
Toluene	0.01	0.01
Xylene	0.01	0.01

105. The permittee shall not exceed a throughput of 26,219,328 gallons of untreated distillate at SN-27f per rolling 12 month period. [§19.705 of Regulation #19, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

106. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #105. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [Regulation 19 §19.705 and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-27g Gasoline Storage tank

## Source Description

MOP stores gasoline in 1 storage tank. Gasoline is delivered to the plant in 4,200 gallon tanker trucks. Only one track can off load in a given hour. Maximum hourly emissions occur during the filling of the gasoline storage tank.

# Specific Conditions

107. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #109. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	33.1	0.2

108. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #109 and Specific Condition #111. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Benzene	5.0E-04	0.01
Ethylbenzene	0.03	0.01
Hexane	1.48	0.01
MTBE	1.21	0.01
POM (including Naphthalene)	2.0E-04	0.01
2,2,4-Trimethylpentane	0.32	0.01
Styrene	0.03	0.01

109. The permittee shall not exceed a throughput of 24,066 gallons of gasoline at SN-27g per rolling 12 month period. [Regulation 19 §19.705 and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

110. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #109. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [Regulation 19 §19.705 and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]

111. The permittee shall sample and analyze the vapor concentration of benzene in tank SN-27g. The permittee shall collect at least one sample from material. The analyses shall be conducted annually and the results shall be less than the value in the following table. The test methods used shall be approved in advance by the Department. The request for test method approval shall be submitted to the Compliance Inspector Supervisor at least fifteen working days in advance of the test. [Regulation 18 §18.1004 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Source	Material	Vapor Benzene Concentration Limit
		(% by weight)
SN-27g	Gasoline	0.0015

#### **NESHAP CCCCCC Conditions**

- The gasoline tank at SN-27g is subject to 40 CFR Part 63, Subpart CCCCC. The permittee shall comply with all applicable provisions of 40 CFR Part 63, Subpart CCCCCC which includes, but is not limited to the following: [40 CFR Part 63, Subpart CCCCCC and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
  - a. The permittee shall keep records to demonstrate that the monthly gasoline throughput is less than 10,000 gallons. These records shall be kept for a period of 5 years. [§63.11111]
  - b. The permittee must, at all times, operate and maintain any affected source in a manner consistent with safety and good air pollution control practices for minimizing emissions. [§63.11115(a)]
  - c. The permittee must not allow gasoline to be handled in a manner that would result in vapor releases to the atmosphere for extended periods of time. Measures to be taken include, but are not limited to, the following: [§63.11116(a)]
    - i. Minimize gasoline spills;
    - ii. Clean up spills as expeditiously as practicable;
    - iii. Cover all open gasoline containers and all gasoline storage tank fill-pipes with a gasketed seal when not in use;
    - Minimize gasoline sent to open waste collection systems that collect and transport gasoline to reclamation and recycling devices, such as oil/water separators.

Permit #: 1227-AOP-R0

AFIN: 70-00039

d. The permittee is not required to submit notifications or reports as specified in 40 CFR §63.11125, §63.11126, or Subpart A, but the permittee must have records available within 24 hours of a request by the Administrator to document the gasoline throughput. [§63.11116(b)]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-27h Untreated Lube Oil Storage Tank

## Source Description

MOP stores untreated lube oil in 20 storage tanks (Tanks #210, #216, #218, #246, #314, #315, #318, #319, #320, #322, #323, #325, #326, #328, #330, #332, #333, #334, #335, #336). MOP does not fill all 20 storage tanks at the same time. They fill one tank at a time from the production line at a rate of 150 gpm. Some of lube oils are blended together, which means they may pump lube oil from one tank to another tank to blend the lube oils. MOP has 3 pumps, each with a pumping rate of 225 gpm, to pump lube oil from one tank to another.

# Specific Conditions

113. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #115. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	0.4	0.4

114. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #115 and Plantwide Condition #9. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Benzene	2.0E-03	0.01
Ethylbenzene	0.01	0.01
Hexane	0.02	0.02
POM (including Naphthalene)	3.0E-04	0.01
2,2,4-Trimethylpentane	0.01	0.01
Xylene	0.01	0.01

115. The permittee shall not exceed a throughput of 82,628,700 gallons of untreated lube oil at SN-27h per rolling 12 month period. [§19.705 of Regulation #19, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

116. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #115. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

## NSPS Kb Conditions: Tank #322 is Subject to 40 CFR Part 60, Subpart Kb:

- 117. SN-27h (Tank #322) shall meet all applicable requirements of 40 CFR Part 60, Subpart Kb (Appendix B) Standards of Performance for Volatile Organic Liquid Storage Vessels (Including Petroleum Liquid Storage Vessels) for Which Construction, Reconstruction, or Modification Commenced After July 23, 1984. These requirements include, but are not limited to the following: [Regulation 19 §19.304 and 40 CFR Part 60, Subpart Kb]
  - a. The owner or operator shall keep copies of all records required by § 60.116b, except for the record required by paragraph of §60.116b(b), for at least 2 years. The record required by paragraph (b) of §60.116b will be kept for the life of the source.
  - b. The owner or operator of each storage vessel as specified in §60.110b(a) shall keep readily accessible records showing the dimension of the storage vessel and an analysis showing the capacity of the storage vessel.
  - c. The owner or operator of each storage vessel either with a design capacity greater than or equal to 151 m³ storing a liquid with a maximum true vapor pressure greater than or equal to 3.5 kPa or with a design capacity greater than or equal to 75 m³ but less than 151 m³ storing a liquid with a maximum true vapor pressure greater than or equal to 15.0 kPa shall maintain a record of the VOL stored, the period of storage, and the maximum true vapor pressure of that VOL during the respective storage period.
  - d. The owner or operator of each storage vessel either with a design capacity greater than or equal to 151 m³ storing a liquid with a maximum true vapor pressure that is normally less than 5.2 kPa or with a design capacity greater than or equal to 75 m³ but less than 151 m³ storing a liquid with a maximum true vapor pressure that is normally less than 27.6 kPa shall notify the Administrator within 30 days when the maximum true vapor pressure of the liquid exceeds the respective maximum true vapor pressure values for each volume range.
  - e. Available data on the storage temperature may be used to determine the maximum true vapor pressure as determined below.

Permit #: 1227-AOP-R0

AFIN: 70-00039

- (1) For vessels operated above or below ambient temperatures, the maximum true vapor pressure is calculated based upon the highest expected calendarmonth average of the storage temperature. For vessels operated at ambient temperatures, the maximum true vapor pressure is calculated based upon the maximum local monthly average ambient temperature as reported by the National Weather Service.
- (2) For refined petroleum products the vapor pressure may be obtained by the following:
- i. Available data on the Reid vapor pressure and the maximum expected storage temperature based on the highest expected calendar-month average temperature of the stored product may be used to determine the maximum true vapor pressure from nomographs contained in API Bulletin 2517 (incorporated by reference—see §60.17), unless the Administrator specifically requests that the liquid be sampled, the actual storage temperature determined, and the Reid vapor pressure determined from the sample(s).
- ii. The true vapor pressure of each type of lube oil with a Reid vapor pressure less than 13.8 kPa or with physical properties that preclude determination by the recommended method is to be determined from available data and recorded if the estimated maximum true vapor pressure is greater than 3.5 kPa.
- f. Visually inspect the internal floating roof, the primary seal, the secondary seal, gaskets, slotted membranes and sleeve seals each time the storage vessel is emptied and degassed. If the internal floating roof has defects, the primary seal has holes, tears, or other openings in the seal or the seal fabric, or the secondary seal has holes, tears, or other openings in the seal or the seal fabric, or the gaskets no longer close off the liquid surfaces from the atmosphere, or the slotted membrane has more than 10 percent open area, the permittee shall repair the items as necessary so that none of the conditions specified in §60.113b(a)(4) exist before refilling the storage vessel with VOL. In no event shall inspections conducted in accordance with this provision occur at intervals greater than 10 years in the case of vessels conducting the annual visual inspection as specified in §60.113b(a)(2) and (a)(3)(ii) and at intervals no greater than 5 years in the case of vessels specified in paragraph (a)(3)(i) of this section.

Permit #: 1227-AOP-R0

AFIN: 70-00039

g. Notify ADEQ in writing at least 30 days prior to the filling or refilling of each storage vessel for which an inspection is required by §60.113b(a)(1) and (a)(4) to afford ADEQ the opportunity to have an observer present. If the inspection required by §60.113b(a)(4) is not planned and the permittee could not have known about the inspection 30 days in advance or refilling the tank, the permittee shall notify ADEQ at least 7 days prior to the refilling of the storage vessel. Notification shall be made by telephone immediately followed by written documentation demonstrating why the inspection was unplanned. Alternatively, this notification including the written documentation may be made in writing and sent by express mail so that it is received by ADEQ at least 7 days prior to the refilling.

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-27i Refinery Treated Lube Oil Storage Tanks

#### Source Description

MOP stores treated lube oil at the refinery in 52 storage tanks (Tanks #113, #197, #206, #223, #224, #229, #230, #247, #263, #266, #269, #270, #287-295, #299-310, #327, #331, #337-353). MOP does not fill all 48 storage tanks at the same time. They fill one tank at a time from the hydrotreated production line at a rate of 150 gpm. Some of the lube oils are blended together, which means they may pump lube oil from one tank to another tank to blend the lube oils. MOP has 3 pumps each with a pumping rate of 225 gpm, to pump lube oil from one tank to another.

#### Specific Conditions

118. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #120. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	0.2	0.2

119. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #120 and Plantwide Condition #9. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Benzene	7.0E-04	0.01
Ethylbenzene	0.01	0.01
Hexane	0.01	0.01
POM (including Naphthalene)	2.0E-04	0.01
2,2,4-Trimethylpentane	0.01	0.01
Xylene	0.01	0.01

- 120. The permittee shall not exceed a throughput of 82,628,700 gallons of treated lube oil at SN-27i per rolling 12 month period. [§19.705 of Regulation #19, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR 70.6]
- 121. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #120. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-27j Packaging Plant Treated Lube Oil Storage Tanks

## Source Description

MOP stores treated lube oil at the packaging plant in 62 tanks (Tanks #P001- #P052, #P060 - #P063, #P080 - #P082, #P296 - #P298) MOP does not fill all 62 storage tanks at the same time. They fill one tank at a time from the Refinery using 3 pumps at a rate of 225 gpm.

#### Specific Conditions

122. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #124. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	0.2	0.2

123. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #124 and Plantwide Condition #9. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Benzene	7.0E-04	0.01
Ethylbenzene	0.01	0.01
Hexane	0.01	0.01
POM (including Naphthalene)	2.0E-04	0.01
Toluene	0.01	0.01
Xylene	0.01	0.01

- 124. The permittee shall not exceed a throughput of 60,000,000 gallons of treated lube oil at SN-27j per rolling 12 month period. [§19.705 of Regulation #19, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR 70.6]
- 125. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #124. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-27k Reclaim Oil Storage Tanks

#### Source Description

MOP stores reclaimed oil from the waste water treatment system in 16 storage tanks (Tanks 226, 284, 500-505, 509, and 512-518). MOP does not fill all 16 reclaimed oil storage tanks at one time. They fill one tank at a time from the waste water treatment system at a pump rate of 500gal/hr. Only one truck can be loaded at a time.

# Specific Conditions

126. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #128. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	0.8	0.4

The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #128 and Plantwide Condition #9. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Benzene	4.0E-03	0.01
Ethylbenzene	0.01	0.01
Hexane	0.06	0.01
POM (including Naphthalene)	7.0E-05	0.01
2,2,4-Trimethylpentane	0.01	0.01
Toluene	0.03	0.01
Xylene	0.01	0.01

- 128. The permittee shall not exceed a throughput of 4,035,229 gallons of reclaimed oil at SN-27k per rolling 12 month period. [§19.705 of Regulation #19, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR 70.6]
- 129. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #128. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.753, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-271 Treated Distillate Storage Tanks

# Source Description

MOP stores treated distillate in 2 storage tanks (Tanks 121 and 122). They fill one tank at a time from the hydrotreater. Normal operations would be to fill a tank from the hydrotreater, and then from that storage tank, the distillate is sent to the loading rack.

#### Specific Conditions

130. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #132. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	0.8	0.7

131. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #132 and Plantwide Condition #9. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Benzene	9.0E-04	0.01
Ethylbenzene	0.01	0.01
Hexane	0.01	0.01
POM (including Naphthalene)	2.0E-04	0.01
Toluene	0.01	0.01
Xylene	0.01	0.01

The permittee shall not exceed a throughput of 26,219,328 gallons of treated distillate at SN-27l per rolling 12 month period. [Regulation 19 §19.705 and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

133. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #132. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-28 & SN-29 Sandyland Storage Tanks and Miller's Storage Tanks

#### Source Description

MOP receives crude oil by barge, pipeline and by tanker trucks. Trucks are loaded from the pipeline with daily deliver to the tanks at Sandyland (SN-28). Crude oil delivered by barge is off loaded to the tanks at Miller's (SN-29).

At Sandyland trucks are off loaded to Tank 104, Tank 110, or Tank 111. Tanks 104, 110, and 111 feed the refinery (Tank 109).

At Miller's, barges off load to Tank 114 and Tank 115. These tanks feed crude oil to the refinery (Tank 109), Tank 104 and/or Sandyland.

## Specific Conditions

134. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by complying with Specific Condition #136. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

SN	Description	Pollutant	lb/hr	tpy
28	Sandyland Storage Tanks	VOC	11.9	19.7
29	Miller's Storage Tanks	VOC	25.0	17.7

The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by complying with Specific Condition #136 and Plantwide Condition #9. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

SN	Description	Pollutant	lb/hr	tpy	
		Benzene		0.01*	
		Ethylbenzene	0.01	0.02*	
		Hexane	0.85	1.40*	
28	Sandyland Storage Tanks	P	POM (including Naphthalene)	2.0E-04	0.01*
		2,2,4-Trimethylpentane	0.03	0.05*	
			Toluene	0.06	0.10*
		Xylene	0.03	0.05*	

Permit #: 1227-AOP-R0

AFIN: 70-00039

SN	Description	Pollutant	lb/hr	tpy	
		Benzene	0.02	*	
		Ethylbenzene	0.02	*	
			Hexane	1.78	*
29	Miller's Storage Tanks	POM (including Naphthalene)	3.0E-04	*	
		2,2,4-Trimethylpentane	0.07	*	
			Toluene	0.12	*
		Xylene	0.06	*	

<sup>\*</sup>Combined total for SN-28 and SN-29

- 136. The permittee shall not exceed a throughput of 118,041,000 gallons of crude oil at SN-28 and 29,510,250 gallons of crude oil at SN-29 per rolling 12 month period. [Regulation 19 §19.705 and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]
- 137. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #136. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [Regulation 19 §19.705 and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-32 Asphalt/Black Oil Tank Car Loading Rack

## Source Description

Asphalt and Black Oil are loaded into railcars (SN-32).

#### **Specific Conditions**

138. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by complying with Specific Condition #89 and Specific Condition #96. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	0.1	0.2

139. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by complying with Specific Condition #89 and Specific Condition #96. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Benzene	3.0E-04	0.01
Ethylbenzene	0.07	0.07
Hexane	0.01	0.01
POM (including naphthalene)	5.0E-05	0.01
Toluene	0.10	0.09
Xylene	0.10	0.09

Permit #: 1227-AOP-R0

AFIN: 70-00039

SN-33 Lube Oil & Distillate Rail Car Loading and Additive Railcar Unloading

#### Source Description

Emissions from distillate loaded onto railcars and lube oil loaded onto railcars are permitted as SN-33.

#### **Specific Conditions**

140. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #105 and Specific Condition #115. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	0.7	0.6

141. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #105 and Specific Condition #115. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Benzene	2.0E-03	0.01
Ethylbenzene	0.01	0.01
n-Hexane	0.01	0.01
POM (including naphthalene)	2.0E-04	0.01
Toluene	0.01	0.01
Xylene	0.01	0.01

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-34 Packaging Plant Lube Oil Loading/Unloading Docks

#### Source Description

Purchased paraffinic oils and lube oils are unloaded at the Packaging Plant loading docks. As well, finished lube oil products are loaded at the Packaging Plant loading docks.

Paraffinic oils are "cleaner" oils than naphthenic lube oils (the type of lube oils produced at MOP). Emissions from the unloading/loading activities at this source will be conservatively estimated by assuming all material loaded/unloaded is naphthenic lube oils.

## **Specific Conditions**

142. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by with Specific Condition #124. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	0.2	0.3

143. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #124. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Benzene	7.0E-04	0.01
Ethylbenzene	0.01	0.01
n-Hexane	0.01	0.01
POM (including naphthalene)	2.0E-04	0.01
Toluene	0.01	0.01
Xylene	0.01	0.01

Permit #: 1227-AOP-R0

AFIN: 70-00039

## SN-36 Lube Charge Heater No. 1

#### Source Description

With this application, MOP proposes to install a new heater for the Lube Charge heater No. 1 (SN-36). The Lube Charge Heater No. 1 is part of the existing Lube Hydrotreater Unit (LHT1). This heater will replace both the existing Lube Charge Heater (SN-07) and the Lube Charge Heater (SN-30). Prior to lube oils being pumped to LHT1, they will be heated by the Lube Oil Charge Heater No. 1 (SN-36). The heater has a maximum heat input rate of 11.14 MMBtu/hr. MOP will burn mixed gas (refinery off gas that is mixed with pipeline quality natural gas) at this heater. The heater will be installed in 2013; therefore, NSPS Subpart Ja will apply to this emission source. MOP maintains the H<sub>2</sub>S concentration of the mixed gas is below the NSPS Subpart Ja standard of 0.10 gr/dsf.

## **Specific Conditions**

144. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by complying with Specific Condition #146, #148 and Plantwide Conditions #14 through #22. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
$PM_{10}$	0.1	0.4
SO <sub>2</sub>	0.4	1.5
VOC	0.1	0.3
СО	1.0	4.1
NO <sub>x</sub>	0.6	2.4
CO <sub>2e</sub>	1,310.9	5,741.6

145. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by complying with Specific Condition #148, Plantwide Conditions #14 through #22. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
PM	0.1	0.4
Benzene	3.0E-05	0.01

Permit #: 1227-AOP-R0

AFIN: 70-00039

Pollutant	lb/hr	tpy
Beryllium	2.0E-07	0.01
Cadmium	2.0E-05	0.01
Chromium	2.0E-05	0.01
Formaldehyde	0.01	0.01
Hexane	0.02	0.09
POM (including naphthalene)	8.0E-06	0.01
Toluene	0.01	0.01

146. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with the CO<sub>2</sub> emission limits by complying with Specific Condition #150. [Regulation 19, §19.901 et seq. and 40 CFR Part 52, Subpart E]

	BACT Analysis Summary				
Source	Description	Pollutant	Control Technology	BACT Limit	
36	Lube Charge Heater No. 1	CO <sub>2e</sub>	Efficient Hydrogen Purification  Good Combustion Practices  Maintenance and Fouling Control  Conduct Annual	117 lb CO <sub>2e</sub> /MMBtu	

- 147. The permittee shall not exceed 5% opacity from Lube Charge Heater No. 1 (SN-36) as measured by EPA Reference Method 9. Compliance with the opacity limit for SN-36 shall be demonstrated by burning natural gas mixed with refinery gas. [Regulation 18, §18.501 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- The permittee shall not exceed a throughput of 95.67 MMscf/yr of natural gas per rolling 12 month period at SN-36. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

- 149. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #148. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.703, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]
- 150. The permittee shall calculate, on a monthly basis, the amount of CO emitted from combustion in tons/yr using equation C-2a in 40 CFR Part 98 Subpart C, converted to short tons. Compliance shall be based on a 12-month rolling basis to be updated by the last day of the following month. As an alternative, the permittee may install a CO<sub>2</sub> CEMS and volumetric stack gas flow monitoring system with an automated data acquisition and handling system for measuring and recording CO<sub>2</sub> emissions discharged to the atmosphere, and use these values to show compliance with the annual emission limit. The CO<sub>2</sub> monitor shall be operated in accordance with the ADEQ CEMS conditions and shall be operated at all times during startup and shutdown. Compliance will be demonstrated on a 12-month rolling basis. [Regulation 19, §19.703; 40 CFR Part 52, Subpart E; and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 151. The permittee shall during each scheduled facility shutdown, but no more than 24 months apart, calibrate and perform preventative maintenance check of the fuel gas flow meters, perform heater burner tune ups and perform a preventative maintenance check. The permittee shall maintain documentation on-site and records shall be made available to Department personnel upon request. [§19.703, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]
- 152. The permittee shall install and maintain a dedicated meter on the natural gas piping which feed SN-36. [§19.703, A.C.A. §8-4-203 as reference by §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-37 Hydrogen Reformer No. 2

### Source Description

With this application, MOP proposes to install a second Hydrogen reformer that will supply hydrogen to one of two hydrotreater units (referred to as LHT1 or LHT2).

Lube oils produced by the atmospheric and vacuum distillation towers are hydrotreated by passing through the hydrotreater units, either LHT1 or LHT2. At the hydrotreater, lube oils combine with hydrogen to make hydrotreated lube oils. The hydrogen is supplied to LHT1 and LHT2 by a steam/methane reformer. Natural gas is compressed and heated in a preheat exchanger and combined with steam. The natural gas mixture is charged to the hydrogen reformer, where is passes over a nickel catalyst and reacts to produce hydrogen and carbon oxides.

The hydrogen reformer is equipped with a natural gas fired burner that has a maximum heat input rate of 35.25 MMBtu/hr. With the exception of GHG emissions, all emissions occur as a result of the natural gas combustion at the burner. GHG emissions occur from the CO<sub>2</sub> generated during the reforming of the natural gas to make hydrogen.

The reformer burner will burn pipeline quality natural gas that is combined with gas from the pressure swing absorption (PSA) system, where the CO<sub>2</sub> and other impurities are removed from the hydrogen. The CO<sub>2</sub> emissions from the combustion of the PSA gas at the burner are accounted for in the GHG emission calculations. Mixed gas, which is refinery off gas that is mixed with pipeline quality natural gas, will not be burned at the reformer burner.

#### Specific Conditions

153. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #155 and #157. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
PM <sub>10</sub>	0.2	0.8
$SO_2$	0.1	0.1
VOC	0.1	0.3
СО	0.3	1.0
NO <sub>x</sub>	1.3	5.5
CO <sub>2e</sub>	18,591.7	81,431.5

Permit #: 1227-AOP-R0

AFIN: 70-00039

154. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #157. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
PM	0.2	0.8
Benzene	8.0E-05	0.01
Beryllium	5.0E-07	0.01
Cadmium	4.0E-05	0.01
Chromium	5.0E-05	0.01
Formaldehyde	0.01	0.02
Hexane	0.07	0.28
POM (including naphthalene)	3.0E-05	0.01
Toluene	0.01	0.01

155. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with the CO<sub>2</sub> emission limits by complying with Specific Condition #159. [Regulation 19, §19.901 et seq. and 40 CFR Part 52, Subpart E]

	BACT Analysis Summary				
Source	Description	Pollutant	Control Technology	BACT Limit	
37	Hydrogen Reformer No. 2	CO <sub>2e</sub>	Efficient Hydrogen Purification, Good Combustion Practices, Maintenance and Fouling Control	0.05 lb/scf H <sub>2</sub> Production	

156. The permittee shall not exceed 5% opacity from Hydrogen Reformer No. 2 (SN-37) as measured by EPA Reference Method 9. Compliance with the opacity limit for SN-37 shall be demonstrated by burning natural gas. [Regulation 18, §18.501 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

157. The permittee shall not exceed a throughput of 302.74 MMcf/yr of natural gas per rolling 12 month period at SN-37. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

- 158. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #158. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]
- The permittee shall calculate, on a monthly basis, the amount of CO emitted from combustion in tons/yr using equation C-2a in 40 CFR Part 98 Subpart C, converted to short tons. Compliance shall be based on a 12-month rolling basis to be updated by the last day of the following month. As an alternative, the permittee may install a CO<sub>2</sub> CEMS and volumetric stack gas flow monitoring system with an automated data acquisition and handling system for measuring and recording CO<sub>2</sub> emissions discharged to the atmosphere, and use these values to show compliance with the annual emission limit. The CO<sub>2</sub> monitor shall be operated in accordance with the ADEQ CEMS conditions and shall be operated at all times during startup and shutdown. Compliance will be demonstrated on a 12-month rolling basis. [Regulation 19, §19.703; 40 CFR Part 52, Subpart E; and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 160. The permittee shall during each scheduled facility shutdown, but no more than 24 months apart, perform heater burner tune ups and perform a preventative maintenance check. The permittee shall maintain documentation on-site and records shall be made available to Department personnel upon request. [§19.703, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-38 Lube Charge Heater No. 2

# Source Description

With this application, MOP proposes to install a second Lube Hydrotreater Unit (LHT2), which will have two emission sources: the Lube Charge Heater No. 2 (SN-38) and the Stripper Charge Heater No. 2 (SN-39). Prior to lube oil being pumped to LHT2, it will be heated by the Lube Charge heater No. 2 (SN-38). The heater has a maximum heat input rate of 35.0 MMBtu/hr. MOP will burn mixed gas (refinery off gas that is mixed with pipeline quality natural gas) at this heater. The heater will be installed in 2013; therefore, NSPS Subpart Ja will apply to this emission source.

# **Specific Conditions**

161. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #163 and Plantwide Conditions #14 through #22. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
PM <sub>10</sub>	0.3	0.7
SO <sub>2</sub>	1.1	2.6
VOC	0.2	0.5
СО	2.9	7.1
NO <sub>x</sub>	1.8	4.2
CO <sub>2e</sub>	4,118.6	10,022.4

The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #163. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
PM	0.3	0.7
Benzene	8.0E-05	0.01
Beryllium	5.0E-07	0.01
Cadmium	4.0E-05	0.01

Permit #: 1227-AOP-R0

AFIN: 70-00039

Pollutant	lb/hr	tpy
Chromium	5.0E-05	0.01
Formaldehyde	0.01	0.01
Hexane	0.07	0.16
POM (including naphthalene)	3.0E-05	0.01
Toluene	0.01	0.01

163. The permittee shall not exceed the emission rates set forth in the following table permittee shall demonstrate compliance with the CO<sub>2</sub> emission limits by complying with Specific Condition #167. [Regulation 19, §19.901 et seq. and 40 CFR Part 52, Subpart E]

	BACT Analysis Summary				
Source	Description	Pollutant	Control Technology	BACT Limit	
38	Lube Charge Heater No. 2	CO <sub>2e</sub>	Efficient Hydrogen Purification  Good Combustion Practices  Maintenance and Fouling Control  Conduct Annual Tune-ups	117 lb CO <sub>2e</sub> /MMBtu	

- The permittee shall not exceed 5% opacity from Lube Charge Heater No. 2 (SN-38) as measured by EPA Reference Method 9. Compliance with the opacity limit for SN-38 shall be demonstrated by burning natural gas mixed with refinery gas. [Regulation 18, §18.501 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- The permittee shall not exceed a throughput of 167 MMscf/yr of natural gas at SN-38 per rolling 12 month period. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

166. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #165. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

- 167. The permittee shall calculate, on a monthly basis, the amount of CO emitted from combustion in tons/yr using equation C-2a in 40 CFR Part 98 Subpart C, converted to short tons. Compliance shall be based on a 12-month rolling basis to be updated by the last day of the following month. As an alternative, the permittee may install a CO<sub>2</sub> CEMS and volumetric stack gas flow monitoring system with an automated data acquisition and handling system for measuring and recording CO<sub>2</sub> emissions discharged to the atmosphere, and use these values to show compliance with the annual emission limit. The CO<sub>2</sub> monitor shall be operated in accordance with the ADEQ CEMS conditions and shall be operated at all times during startup and shutdown. Compliance will be demonstrated on a 12-month rolling basis. [Regulation 19, §19.703; 40 CFR Part 52, Subpart E; and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 168. The permittee shall during each scheduled facility shutdown, but no more than 24 months apart, calibrate and perform preventative maintenance check of the fuel gas flow meters, perform heater burner tune ups and perform a preventative maintenance check. The permittee shall maintain documentation on-site and records shall be made available to Department personnel upon request. [§19.703, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]
- 169. The permittee shall install and maintain a dedicated meter on the natural gas piping which feed SN-38. [§19.703, A.C.A. §8-4-203 as reference by §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-39 Stripper Charge Heater No. 2

# Source Description

The Stripper Charge Heater No. 2 (SN-39) is a part of the new Lube Hydrotreater Unit (LHT2). The lube oil from LHT2 flows to a lube oil stripper, where the remaining hydrogen sulfide is removed by steam stripping. The Stripper Charge Heater (SN-39) supplies heat to the lube oil stripper. The heater has a maximum heat input rate of 6.84 MMBtu/hr.

MOP will burn mixed gas (refinery off gas that is mixed with pipeline quality natural gas) at the stripper charge heater. The stripper charge heater will be installed in 2013 therefore; NSPS Subpart Ja applies to this emission source. MOP maintains the H<sub>2</sub>S concentration of the mixed gas is below the NSPS Subpart Ja standard of 0.10 gr/dscf.

## **Specific Conditions**

170. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #172, #174, and Plantwide Conditions #14 through #22. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
$PM_{10}$	0.1	0.2
$SO_2$	0.2	0.7
VOC	0.1	0.2
СО	0.6	1.8
NO <sub>x</sub>	0.7	2.1
CO <sub>2e</sub>	804.9	2,478.6

171. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #174. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
PM	0.1	0.2
Benzene	2.0E-05	0.01

Permit #: 1227-AOP-R0

AFIN: 70-00039

Pollutant	lb/hr	tpy
Beryllium	9.0E-08	0.01
Cadmium	8.0E-07	0.01
Chromium	1.0E-05	0.01
Formaldehyde	0.01	0.01
Hexane	0.02	0.04
POM (including naphthalene)	5.0E-06	0.01
Toluene	0.01	0.01

172. The permittee shall not exceed the emission rates set forth in the following table permittee shall demonstrate compliance with the CO<sub>2</sub> emission limits by complying with Specific Condition #176. [Regulation 19, §19.901 et seq. and 40 CFR Part 52, Subpart E]

	BACT Analysis Summary				
Source	Description	Pollutant	Control Technology	BACT Limit	
39	Stripper Charge Heater No. 2	CO <sub>2e</sub>	Efficient Hydrogen Purification  Good Combustion Practices  Maintenance and Fouling Control  Conduct Annual Tune-ups	117 lb CO <sub>2e</sub> /MMBtu	

- 173. The permittee shall not exceed 5% opacity from Stripper Charge Heater No. 2 (SN-39) as measured by EPA Reference Method 9. Compliance with the opacity limit for SN-39 shall be demonstrated by burning natural gas mixed with refinery gas. [Regulation 18, §18.501 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 174. The permittee shall not exceed a throughput of 41.3 MMscf/yr of natural gas at SN-39 per rolling 12 month period. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

- 175. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #174. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]
- 176. The permittee shall calculate, on a monthly basis, the amount of CO emitted from combustion in tons/yr using equation C-2a in 40 CFR Part 98 Subpart C, converted to short tons. Compliance shall be based on a 12-month rolling basis to be updated by the last day of the following month. As an alternative, the permittee may install a CO<sub>2</sub> CEMS and volumetric stack gas flow monitoring system with an automated data acquisition and handling system for measuring and recording CO<sub>2</sub> emissions discharged to the atmosphere, and use these values to show compliance with the annual emission limit. The CO<sub>2</sub> monitor shall be operated in accordance with the ADEQ CEMS conditions and shall be operated at all times during startup and shutdown. Compliance will be demonstrated on a 12-month rolling basis. [Regulation 19, §19.703; 40 CFR Part 52, Subpart E; and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 177. The permittee shall during each scheduled facility shutdown, but no more than 24 months apart, calibrate and perform preventative maintenance check of the fuel gas flow meters, perform heater burner tune ups and perform a preventative maintenance check. The permittee shall maintain documentation on-site and records shall be made available to Department personnel upon request. [§19.703, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]
- 178. The permittee shall install and maintain a dedicated meter on the natural gas piping which feed SN-38. [§19.703, A.C.A. §8-4-203 as reference by §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

SN-40 Flare

# Source Description

MOP operates a flare that is designed specifically to control emergency releases from the Crude Unit and the Hydrogen Reformers. The flare will have a pilot maximum heat input rate of 2.55 MMBtu/hr. MOP will burn mixed gas (refinery off gas that is mixed with pipeline quality natural gas) as the pilot gas for the flare. The flare for the crude unit will be installed in 2013; therefore, NSPS Subpart Ja applies to this emission source. MOP maintains the  $H_2S$  concentration of the mixed gas is below the NSPS Subpart Ja standard of  $0.10 \, \text{gr/dscf}$ .

# **Specific Conditions**

179. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #184 and Plantwide Conditions #14 through #22. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
PM <sub>10</sub>	0.1	0.1
SO <sub>2</sub>	0.1	0.4
VOC	0.1	0.1
СО	0.3	1.0
NO <sub>x</sub>	0.2	0.6

180. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #184. [Regulation 19, §19.901 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
CO <sub>2e</sub>	300.1	1,314.3

Permit #: 1227-AOP-R0

AFIN: 70-00039

181. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #184. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
PM	0.1	0.1
Benzene	6.0E-06	0.01
Beryllium	3.0E-08	0.01
Cadmium	3.0E-06	0.01
Chromium	4.0E-06	0.01
Formaldehyde	0.01	0.01
Hexane	0.01	0.02
POM (including naphthalene)	2.0E-06	0.01
Toluene	0.01	0.01

- 182. The permittee shall not exceed 5% opacity from SN-40 as measured by EPA Reference Method 22. Compliance with the opacity limit is demonstrated by compliance with Specific Conditions #183. [Regulation 19, §19.503 and 40 CFR 52, Subpart E]
- 183. The permittee shall conduct weekly observations of the opacity at SN-40 and keep a record of these observations. The flare (SN-40) shall be designed for and operated with no visible emissions, except for periods not to exceed a total of five (5) minutes during any two (2) consecutive hours. EPA Reference Methods 22 shall be used to determine compliance with the visible emission provisions of the flare. If the permittee detects visible emissions in excess of their permitted limit, the permittee must immediately take action to identify and correct the cause of the visible emissions. After implementing the corrective action, the permittee must document that the source complies with the visible emissions requirements. The permittee shall maintain records of the cause of the visible emissions and the corrective action taken. The permittee must keep these records onsite and make them available to Department personnel upon request. [Regulation 18, §18.501 and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]
- 184. The permittee shall not exceed a throughput of 21.9 MMcf/yr of natural gas at SN-40 per rolling 12 month period. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

- 185. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #184. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]
- 186. The permittee must operate the flare (SN-40) pilot flame within the design limitations and manufacturer's specifications. [Regulation 19, §19.303 and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]
- 187. The Flare (SN-40) must have a flame present at all times of operation. The presence of a flare pilot light shall be monitored continuously using a thermocouple, an ultraviolet sensor or any other equivalent device to detect the presence of a flame. In the event of a flame failure, the permittee shall maintain and follow emergency procedures until the flame is present again. The permittee shall report all upset conditions to the Department by the end of the next business day after the discovery of the occurrence. [Regulation 19, §19.303, §19.304, §19.601, §60.18(b) through (f), and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]
- 188. The permittee shall install and operate alarm system to notify the operator of the presence of a pilot flame or other possible flare malfunction. The permittee shall perform monthly visual confirmation of the pilot lights, semi-annually remove the strainer and check for debris, and annual test fire to ensure pilot light. The permittee shall maintain logs of all flare inspection and maintenance activities. These logs shall be kept on site, in accordance with General Provision 7, and made available to Department personnel upon request. [Regulation 19, §19.702; 40 CFR 52, Subpart E; and 40 CFR Part 64]

Permit #: 1227-AOP-R0

AFIN: 70-00039

189. The permittee shall report an upset condition any time that the flare is used. The permittee shall report the upset (by telephone, facsimile, or overnight delivery) to the Department by the end of the next business day after the occurrence or the discovery of the occurrence. The permittee shall submit to the Department, within five business days after the occurrence or the discovery of the occurrence, a full, written report of such occurrence, including a statement of all known causes and of the scheduling and nature of the actions to be taken to minimize or eliminate future occurrences, including, but not limited to, action to reduce the frequency of occurrence of such conditions, to minimize the amount by which said limits are exceeded, and to reduce the length of time for which said limits are exceeded. If the information is included in the initial report, it need not be submitted again. [Regulation 19 §19.601 and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]

190. The permittee shall during each scheduled plant shutdown, but no more than 24 months apart, calibrate and perform preventative maintenance. In accordance with manufacturer's specification, the permittee shall ensure the flare operates with efficient combustion control and shall implement procedures to minimize the amount of carbon in the unburned gas stream through steps including proper burner turning and the use of natural gas as the combustion fuel. The permittee shall maintain documentation on-site and records shall be made available to Department personnel upon request. [§19.703, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

SN-41, SN-42, and SN-43 Fire Pump Engine No.1, Fire Pump Engine No.2, and Fire Pump Engine No. 3

# Source Description

Fire Pump Engine No. 1 (SN-41) and Fire Pump Engine No.3 (SN-43) are classified as new stationary RICE located at an area source that is rate at <500 hp, which are subject to NESHAP Subpart ZZZZ. Pursuant to 63.6590(c), compliance with this subpart is demonstrated by comply with NSPS Subpart IIII. Fire Pump Engine No. 2 (SN-42) is classified as an existing stationary RICE located at an area source that is rated at <500 hp, which is subject to NESHAP Subpart ZZZZ.

All three fire pump engines are diesel fired and are used to supply water to the refinery in the event of a fire. The engines are tested on a weekly basis for 30 minutes.

#### **Specific Conditions**

191. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #193, #195, #205, and #206. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

SN	Description	Pollutant	lb/hr	tpy
		PM <sub>10</sub>	0.1	0.1
		$\mathrm{SO}_2$	0.7	0.1
41	Eiro Duma Engino No. 1	VOC	0.2	0.1
41	Fire Pump Engine No. 1	СО	1.1	0.1
		$NO_x$	1.8	0.1
		$CO_{2e}$	376.1	18.9
	Fire Pump Engine No. 2	PM <sub>10</sub>	0.4	0.1
		$\mathrm{SO}_2$	0.4	0.1
42		VOC	0.5	0.1
42		СО	1.2	0.1
		$NO_x$	5.3	0.3
		CO <sub>2e</sub>	194.4	9.8

Permit #: 1227-AOP-R0

AFIN: 70-00039

SN	Description	Pollutant	lb/hr	tpy
		PM <sub>10</sub>	0.1	0.1
		SO <sub>2</sub>	0.7	0.1
42		VOC	0.2	0.1
Fire Pump Engine No. 2	СО	1.1	0.1	
	NO <sub>x</sub>	1.8	0.1	
	CO <sub>2e</sub>	376.1	18.9	

The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #193, #195, #205, and #206. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

SN	Description	Pollutant	lb/hr	tpy
41		PM	0.1	0.1
		Benzene	3.0E-03	0.01
	Fire Pump Engine No. 1	Formaldehyde	0.01	0.01
41	The Fump Engine No. 1	POM (including naphthalene)	2.0E-04	0.01
		Toluene	0.01	0.01
		Xylene	0.01	0.01
		PM	0.4	0.1
	Fire Pump Engine No. 2	Benzene	2.0E-03	0.01
40		Formaldehyde	0.01	0.01
42		POM (including naphthalene)	1.0E-04	0.01
		Toluene	0.01	0.01
		Xylene	0.01	0.01
		PM	0.1	0.1
		Benzene	3.0E-03	0.01
42	D' D E ' M 4	Formaldehyde	0.01	0.01
43	Fire Pump Engine No. 3	POM (including naphthalene)	2.0E-04	0.01
		Toluene	0.01	0.01
		Xylene	0.01	0.01

Permit #: 1227-AOP-R0

AFIN: 70-00039

- 193. The permittee shall not exceed 20% opacity from SN-41, SN-42, and SN-43 as measured by EPA Reference Method 9. Compliance with this Specific Condition shall be demonstrated by compliance with Specific Condition 194. [Regulation 18, §18.501, and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]
- 194. Annual observations of the opacity from SN-41, SN-42, and SN-43 shall be conducted by a person trained but not necessarily certified in EPA Reference Method. If visible emissions in excess of the permitted levels are detected, the permittee shall immediately take action to identify the cause of the visible emissions in excess of the permit limit, implement corrective action, and document that visible emissions did not appear to be in excess of the permitted opacity following the corrective action. The permittee shall maintain records which contain the following items in order to demonstrate compliance with this specific condition. These records shall be updated annually kept on site, and made available to Department personnel upon request.
  - a. The date and time of the observation.
  - b. If visible emissions which appeared to be above the permitted limit were detected.
  - c. If visible emissions which appeared to be above the permitted limit were detected, the cause of the exceedance of the opacity limit, the corrective action taken, and if the visible emissions appeared to be below the permitted limit after the corrective action was taken.
  - d. The name of the person conducting the opacity observations.
- 195. The permittee shall not operate the emergency engines SN-41, SN-42, and SN-43 in excess of 500 hours each per calendar year. [Regulation 19, §19.705 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 196. The permittee shall maintain monthly and annual records to demonstrate compliance with Specific Condition #195. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. A twelve month calendar total and each individual month's data shall be in a spreadsheet, database, or other well-organized format, maintained on site, made available to Department personnel upon request and submitted in accordance with General Provision #21. [Regulation 19, §19.705 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

# SN-42: NEHSAP Subpart ZZZZ

- 197. The permittee shall meet the following requirements of Table 2d of 40 CFR Part 63 Subpart ZZZZ:
  - a. Change oil and filter every 500 hours of operation or annually, whichever comes first.
  - b. Inspect air cleaner every 1,000 hours of operation or annually, whichever comes first.
  - c. Inspect all hoses and belts every 500 hours of operation or annually, whichever comes first, and replace as necessary.

[Regulation No. 19 §19.304 and 40 CFR Part 63, Subpart ZZZZ, §63.6603(a)]

Permit #: 1227-AOP-R0

AFIN: 70-00039

- 198. The permittee shall operate and maintain SN-42 according to the manufacturer's emission-related written instructions or develop maintenance plan which must provide to the extent practicable for the maintenance and operation of the engine in a manner consistent with good air pollution control practice for minimizing emissions. [Regulation No. 19 §19.304 and 40 CFR Part 63, Subpart ZZZZ, §63.6625(e)]
- 199. The permittee shall install a non-resettable hour meter if one is not already installed at SN-42. [Regulation No. 19 §19.304 and 40 CFR Part 63, Subpart ZZZZ, §63.6625(f)]
- 200. The permittee shall minimize the engines time spent at idle during startup and minimize the engines startup time to a period needed for appropriate and safe loading of the engines, not to exceed 30 minutes, after which time the emission standards applicable to all times other than startup in Table 2c of 40 CFR Part 63 Subpart ZZZZ apply. [Regulation No. 19 §19.304 and 40 CFR Part 63, Subpart ZZZZ, §63.6625(h)]
- 201. The permittee may operate the emergency stationary RICE engines for the purpose of maintenance checks and readiness testing, provided that the tests are recommended by Federal, State, or local government, the manufacturer, the vendor or the insurance company associated with the engine. Maintenance checks and readiness testing of such units is limited to 100 hours per year. SN-42 may be operated up to 50 hours per year in non-emergency situations, but those hours per year are counted towards the 100 hours per year provided for maintenance and testing. The hours cannot be used for peak shaving or to generate income for the facility to supply power to the electric grid or otherwise supply power as part of a financial arrangement with another entity. There are no time limits for the use of an emergency stationary RICE in emergency situations. The engines may not be operated for more than 30 minutes prior to the time when the emergency conditions expected to occur, and the engines operation must be terminated immediately after the facility is notified that the emergency condition is no longer imminent. [Regulation No. 19 §19.304 and 40 CFR Part 63, Subpart ZZZZ, §63.6640(f)(4)]
- 202. The permittee shall submit deviations to 40 CFR Part 63 Subpart ZZZZ in the semiannual monitoring report and annual compliance report. [Regulation No. 19 §19.304 and 40 CFR Part 63, Subpart ZZZZ, §63.6650(f)]
- 203. The permittee shall keep records of the maintenance conducted on SN-42 in order to demonstrate that they were operated and maintained according to the maintenance plan. [Regulation No. 19 §19.304 and 40 CFR Part 63, Subpart ZZZZ, §63.6655(e)(2)]
- 204. The permittee shall keep records of the hours of operation of the engines that is recorded through the non-resettable hour meter. The permittee must document how many hours are spent for emergency operation; including what classified the operation as emergency and how many hours are spent for non-emergency operation. If the engines are used for demand response operation, records of notification of the emergency situation, and the time the engines were operated as part of demand response. [Regulation No. 19 §19.304 and 40 CFR Part 63, Subpart ZZZZ, §63.6655(f)(2)]

Permit #: 1227-AOP-R0

AFIN: 70-00039

205. The permittee shall maintain files of all information required by 40 CFR Part 63 Subpart ZZZZ recorded in a form suitable and readily available for expeditious inspection and review. The files shall be retained for at least 5 years following the date of each occurrence, measurement, maintenance, corrective action, report, or record. At a minimum, the most recent 2 years of data shall be retained on site. [Regulation No. 19 §19.304 and 40 CFR Part 63, Subpart ZZZZ, §63.6660]

#### SN-41 and SN-43: NSPS Subpart IIII

- 206. The permittee shall comply with and maintain the emission standards in Table 4 of 40 CFR Part 60 Subpart IIII for all pollutants over the entire life of the engine. [Regulation No. 19 §19.304 and 40 CFR Part 60 Subpart IIII, §60.4205(c),§60.4206]
- The permittee shall only use diesel fuel that meets the requirements of 40 CFR §80.510 (b). [Regulation No. 19 §19.304 and 40 CFR Part 60 Subpart IIII, §60.4207(b)]
- 208. The permittee shall install a non-resettable hour meter prior to startup of the engines. [Regulation No. 19 §19.304 and 40 CFR Part 60 Subpart IIII, §60.4209(a)]
- 209. The permittee shall demonstrate compliance with the emission standards in Subpart IIII by complying with the following:
  - a. Operate and maintain SN-41 and SN-43 according to the manufacturer's emission-related written instructions;
  - b. Change only those emission related settings that are permitted by the manufacturer;
  - c. Meet the requirements of 40 CFR parts 89, 94, and/or 1068, as applicable.

[Regulation No. 19 §19.304 and 40 CFR Part 60 Subpart IIII, §60.4211(a)(1)(2)(3)]

- 210. The permittee may operate the engine for the purpose of maintenance checks and readiness testing, provided that the tests are recommended, for a maximum of 100 hours per year. The engines may be operated up to 50 hours per year in non-emergency situation, but those hours are counted towards the 100 hours per year provided for maintenance and testing. [Regulation No. 19 §19.304 and 40 CFR Part 60 Subpart IIII, §60.4211(f)]
- 211. If the permittee does not install, configure, operate, and maintain SN-41 and SN-43 according to the manufacturer's emission related written instructions, or if the permittee changes emission related settings in a way that is not permitted by the manufacturer, the permittee shall demonstrate compliance through compliance with §40.4211(g). [Regulation No. 19 §19.304 and 40 CFR Part 60 Subpart IIII, §60.4211(g)]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-44 Packaging Plant Railcar Loading and Unloading Rack

# Source Description

## **Specific Conditions**

212. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #214. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	0.1	0.1

213. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #214. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Benzene	6.68E-05	0.01
Ethylbenzene	0.01	0.01
n-Hexane	0.01	0.01
POM (including Naphthalene)	5.72E-05	0.01
Toluene	0.01	0.01
Xylene	0.01	0.01

- 214. The permittee shall not exceed a throughput of 12,000,000 gallons of lube oil at SN-44 per rolling 12 month period. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]
- 215. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #214. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.703, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-45 Packaging Plant Truck Loading and Unloading Rack

# Source Description

# **Specific Conditions**

216. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #218. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
VOC	0.1	0.1

217. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #218. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
Benzene	2.79E-04	0.01
Ethylbenzene	0.01	0.01
n-Hexane	0.01	0.01
POM (including Naphthalene)	5.89E-05	0.01
Toluene	0.01	0.01
Xylene	0.01	0.01

- 218. The permittee shall not exceed a throughput of 24,300,000 gallons of lube oil at SN-45 per rolling 12 month period. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]
- 219. The permittee shall maintain monthly records to demonstrate compliance with Specific Condition #218. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [§19.703, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SN-46 Haul Roads

# Source Description

MOP has various types of trucks that travel throughout the plant to deliver or pickup production. Fugitive dust or PM emissions are generated as a result of the vehicle travel.

# **Specific Conditions**

220. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #222. [Regulation 19 §19.501 et seq. and 40 CFR Part 52, Subpart E]

Pollutant	lb/hr	tpy
PM <sub>10</sub>	1.0	3.5

221. The permittee shall not exceed the emission rates set forth in the following table. The permittee shall demonstrate compliance with this condition by compliance with Specific Condition #222. [Regulation 18 §18.801 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Pollutant	lb/hr	tpy
PM	0.3	0.8

Dust suppression activities should be conducted in a manner and at a rate of application that will not cause runoff from the area being applied. Best Management Practices (40 CFR §122.44(k)) should be used around streams and waterbodies to prevent the dust suppression agent from entering Waters of the State. Except for potable water, no agent shall be applied within 100 feet of wetlands, lakes, ponds, springs, streams, or sinkholes. Failure to meet this condition may require the permittee to obtain a National Pollutant Discharge Elimination System (NPDES) permit in accordance with 40 CFR §122.1(b). [A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# SECTION V: COMPLIANCE PLAN AND SCHEDULE

Martin Operating Partnership L.P. will continue to operate in compliance with those identified regulatory provisions. The facility will examine and analyze future regulations that may apply and determine their applicability with any necessary action taken on a timely basis.

Permit #: 1227-AOP-R0

AFIN: 70-00039

#### SECTION VI: PLANTWIDE CONDITIONS

- 1. The permittee shall notify the Director in writing within thirty (30) days after commencing construction, completing construction, first placing the equipment and/or facility in operation, and reaching the equipment and/or facility target production rate. [Regulation 19 §19.704, 40 CFR Part 52, Subpart E, and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 2. If the permittee fails to start construction within eighteen months or suspends construction for eighteen months or more, the Director may cancel all or part of this permit. [Regulation 19 §19.410(B) and 40 CFR Part 52, Subpart E]
- 3. The permittee must test any equipment scheduled for testing, unless otherwise stated in the Specific Conditions of this permit or by any federally regulated requirements, within the following time frames: (1) new equipment or newly modified equipment within sixty (60) days of achieving the maximum production rate, but no later than 180 days after initial startup of the permitted source or (2) operating equipment according to the time frames set forth by the Department or within 180 days of permit issuance if no date is specified. The permittee must notify the Department of the scheduled date of compliance testing at least fifteen (15) business days in advance of such test. The permittee shall submit the compliance test results to the Department within thirty (30) calendar days after completing the testing. [Regulation 19 §19.702 and/or Regulation 18 §18.1002 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 4. The permittee must provide:
  - a. Sampling ports adequate for applicable test methods;
  - b. Safe sampling platforms;
  - c. Safe access to sampling platforms; and
  - d. Utilities for sampling and testing equipment.

[Regulation 19 §19.702 and/or Regulation 18 §18.1002 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

- 5. The permittee must operate the equipment, control apparatus and emission monitoring equipment within the design limitations. The permittee shall maintain the equipment in good condition at all times. [Regulation 19 §19.303 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 6. This permit subsumes and incorporates all previously issued air permits for this facility. [Regulation 26 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

7. The permittee shall not exceed a throughput of 7,700 bbl/day of crude oil at the facility per rolling 12 month period. [§19.705, A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311 & 40 CFR Part 70.6]

- 8. The permittee shall maintain monthly records to demonstrate compliance with Plantwide Condition #7. The permittee shall update these records by the thirtieth day of the month following the month to which the records pertain. The twelve month rolling totals and each individual month's data shall be maintained on-site, made available to Department personnel upon request, and submitted in accordance with General Provision #7. [Regulation No. 19 §19.705 and 40 CFR Part 52, Subpart E]
- 9. The naphtha loading rack emissions shall be routed to Tanks #7130, #7131, and #7132. The naphtha storage tanks (Tanks #7130, #7131, and #7132) shall not vent to the atmosphere. [Regulation 19 §19.705 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 10. The heavy condensate loading rack emissions shall be routed to Tank #7133. The heavy condensate storage tank (Tanks #7133) shall not vent to the atmosphere. [Regulation 19 §19.705 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 11. The permittee shall sample and analyze the materials stored in the distillate and lube oil (treated and untreated) tanks at SN-27c SN-27l for the following pollutants: 2,2,4-trimethylpentane; benzene; 1,3-butadiene; tert-butylmethylether; cresol; cumene; cyclohexane; ethylbenzene; n-hexane; naphthalene; phenol; toluene; m,p-xylene; and o-xylene. These analyses shall be conducted annually and the results shall either be less than the values in the following table or the permittee shall show that the values result in emissions that are less than the permitted emission rates in Specific Condition #88, #95, #99, #104, #108, #114, #119, #123, #127, and #131. The test methods used shall be approved in advance by the Department. The request for test method approval shall be submitted to the Compliance Inspector Supervisor at least fifteen working days in advance of the test. [Regulation 18 §18.1004 and A.C.A. §8-4-203 as referenced by A.C.A. §8-4-304 and §8-4-311]

Martin Operating Partnership L.P. Permit #: 1227-AOP-R0

AFIN: 70-00039

Material	Pollutant	Limit (mg/kg)
Lube Oil	Benzene	1.67
¥	1,3-Butadiene	Below Detection
	tert-Butylmethylether	Below Detection
	Cresol	Below Detection
	Cumene	6.11
	Cyclohexane	28.8
	Ethylbenzene	7.44
	n-Hexane	8.61
	Naphthalene	95.1
	Phenol	Below Detection
	2,2,4-Trimethylpentane	Below Detection
	Toluene	10.7
	m,p-Xylene	33.3
	o-Xylene	12.4
Distillate	Benzene	3.68
	1,3-Butadiene	Below Detection
	tert-Butylmethylether	Below Detection
	Cresol	Below Detection
	Cumene	11.4
	Cyclohexane	37.1
	Ethylbenzene	17.9
	n-Hexane	9.2
	Naphthalene	162
	Phenol	Below Detection
	2,2,4-Trimethylpentane	Below Detection
	Toluene	23.3
	m,p-Xylene	84.8
	o-Xylene	35.4
Untreated Lube Oil	Benzene	1.67
	1,3-Butadiene	Below Detection
	tert-Butylmethylether	Below Detection
	Cresol	Below Detection
	Cumene	6.11
	Ethylbenzene	7.44
	n-Hexane	8.61
	Naphthalene	23.0
	Phenol	Below Detection
	2,2,4-Trimethylpentane	Below Detection
	Toluene	10.7
	m,p-Xylene	33.3
	o-Xylene	12.4

Permit #: 1227-AOP-R0

AFIN: 70-00039

Material	Pollutant	Limit (mg/kg)
Treated Lube Oil	Benzene	Below Detection
	1,3-Butadiene	Below Detection
	tert-Butylmethylether	Below Detection
	Cresol	Below Detection
	Cumene	1.16
	Ethylbenzene	2.01
	n-Hexane	Below Detection
	Naphthalene	95.1
	Phenol	Below Detection
	2,2,4-Trimethylpentane	Below Detection
	Toluene	2.12
	m,p-Xylene	11.8
	o-Xylene	6.24

12. The permittee shall not exceed the following vapor pressure limits for the materials stored at this facility. The permittee shall maintain documentation to demonstrate compliance with this specific condition. [Regulation 19 §19.705 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Material	Vapor Pressure (psia)
Additive	0.0003
Asphalt	0.002
Crude Oil	0.3198
Distillate (not heavy condensate)	0.0115
Gasoline	6.5377
Lube Oil	0.001*
Reclaimed Oil	0.0218

<sup>\*</sup>Limit for the average of all the lube oils produced.

13. The permittee shall maintain the tank number easily visible at each storage tank (each tank at SN-27a-1, SN-28, and SN-29). The permittee shall maintain a cross reference tank inventory of all storage tanks that identifies the tank number, size, installation date, and contents of the tank. The permittee shall not have any tanks that are not listed in the tank inventory in Appendix M. [Regulation 19 §19.705 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

#### **NSPS Subpart Ja**

14. The permittee shall determine the total reduce sulfur concentration for each gas line directed to the affected flare. The permittee shall install, operate, calibrate, and maintain an instrument for continuously monitoring and recording the concentration of total reduced sulfur in gas discharged to the flare.

Permit #: 1227-AOP-R0

AFIN: 70-00039

- a. The permittee shall install operate and maintain each total reduced sulfur monitor according to Performance Specification 5 of Appendix B to Part 60.
- b. The permittee shall conduct performance evaluations of each total reduced sulfur monitor according to the requirements of §60.13(c) and Performance Specification 5 of Appendix b to Part 60. For flares that routinely have flow, the permittee of each total reduced sulfur monitor shall use EPA Method 15A of Appendix A-5 to Part 60 for conducting the relative accuracy evaluations.
- c. The permittee shall comply with the applicable quality assurance procedures in Appendix F to Part 60 for each total reduced sulfur monitor. [§60.107a(e)(1)]
- 15. The permittee shall install, calibrate, operate and maintain a CPMS to measure and record the flow rate of the gas discharged to the flare. If a flow monitor is not already in place, the permittee shall comply with §60.107a(f) by no later than November 15, 2015, or upon startup of the flare, whichever is later. The permittee shall install, calibrate, operate and maintain each flow monitor according to the manufacturer's procedures and specifications and the following requirements:
  - a. Locate the monitor in a position that provides a representative measurement of the total gas flow rate.
  - b. Use a flow sensor with a measurement sensitivity of no more than 5 percent of the flow rate or 10 cubic feet per minute, whichever is greater.
  - c. Use a flow monitor that is maintainable online, is able to continuously correct for temperature and pressure and is able to record flow in standard conditions (as defined in §60.2) over one-minute averages.
  - d. At least quarterly, perform a visual inspection of all components of the monitor for physical and operational integrity and all electrical connections for oxidation and galvanic corrosion of the flow monitor is not equipped with a redundant flow sensor.
  - e. Recalibrate the flow monitor in accordance with the manufacturers' procedures and specification biennially (every two years) or at the frequency specified by the manufacturer.

 $[\S60.107a(f)(1)]$ 

16. The permittee shall not burn in any fuel that contains H<sub>2</sub>S in excess of 162 ppmv determined hourly on a 3-hour rolling average basis and H<sub>2</sub>S in excess of 60 ppmv determined daily on a 365 successive calendar day rolling average. The permittee shall comply with the notification, recordkeeping, and reporting requirements in §60.7. [§60.102a(g)(1)(ii) and §60.108a(a)]

Permit #: 1227-AOP-R0

AFIN: 70-00039

17. The permittee shall develop and implement a written flare management plan in accordance with §60.103a(a)(1) through (6) no later than the date the flare becomes an affected flare. The permittee shall maintain a copy of the flare management plan on site and made available to the Department upon request. [§60.103a(a) and §60.108a(c)(1)]

- 18. The permittee shall conduct a root cause analysis of any emission limit exceedance or process start-up, shutdown, upset, or malfunction that causes a discharge to the atmosphere in excess of 500 lb/day of SO<sub>2</sub>. The permittee shall record the identification of the affected facility, the date and duration of the discharge, the results and the action taken as a result of the root cause analysis. The permittee shall maintain a copy of each root cause analysis of a discharge onsite and made available to the Department upon request. [§60.103a(d) and §60.108a(c)(6)(ix)]
- 19. The permittee shall record and maintain records of discharges greater than 500 lb/day SO<sub>2</sub> and discharges to an affected flare in excess of 500,000 scfd. The records shall include all items listed in §60.108a(c)(6)(i) through (xi), as applicable. [§60.108a(c)(6)]
- 20. The permittee shall conduct a performance test to demonstrate initial compliance with the applicable emissions limits in §60.102a according to the requirements of §60.8. The notification requirements of §60.8d apply to the initial performance test. The permittee shall use the test methods in 40 CFR Part 60 Appendices A-1 through A-8. [§60.104a(a), (c)]
- 21. The permittee shall determine compliance with the H<sub>2</sub>S emissions limit in §60.102a(g), in accordance with §60.104a(j). [§60.104a(j)]
  - a. The permittee shall install, operate, and maintain each H<sub>2</sub>S monitor according to Performance Specification 7 of Appendix B to Part 60. The span value for this instrument is 300 ppmv H<sub>2</sub>S.
  - b. The permittee shall conduct performance evaluations for each  $H_2S$  monitor according to the requirements of  $\S60.13(c)$  and Performance Specification 7 of Appendix B to Part 60.
  - c. The permittee shall comply with the applicable quality assurance procedures in Appendix F to Part 60 for each H<sub>2</sub>S monitor.
  - d. Fuel gas combustion device having a common source of fuel gas may be monitored at only one location, if monitoring at this location accurately represents the concentration of H<sub>2</sub>S in the fuel gas being burned.
- 22. The permittee shall submit an excess emissions report for all periods of excess emissions according to the requirements of §60.7(c) except that the report shall contain the information specified in §60.108a(d)(1) through (7). The excess emissions report shall be submitted in accordance with General Provision No. 7. [§60.108a(d)]

Permit #: 1227-AOP-R0

AFIN: 70-00039

# **NSPS Subpart GGGa**

- All equipment associated with Lube Hydrotreater Unit No. 2, which includes but is not limited to SN-38 and SN-39 shall comply with the requirements of §60.482-1a to §60.482-10a as soon as practicable, but no later than 180 days after initial startup. [40 CFR Part §60.592a(a)]
- 24. The facility may elect to comply with §60.483-1a, §60.483-2a or Phase III provisions in §63.168, except the facility may elect to follow the provisions in §60.482-7a(f) instead of §63.168 for any valve that is designated as being leakless. [40 CFR Part §60.592a(b)]
- 25. The facility shall comply with the provisions on §60.485a except as provided in §60.593a
- The permittee shall demonstrate compliance with the requirements of §60.482-1a through 60.482-10a or §60.480a(e) for all equipment within 180 days of initial startup. Compliance will be determined by review of performance test results, and inspection using the methods and procedures specified in §60.485a. [40 CFR Part 60.§60.482-1a]
- 27. If the permittee elects to comply with an allowable percentage of valves leaking of equal or less than 2.0%, the permittee must notify ADEQ before implementing as specified in §60.487a(d). A performance test shall be conducted initially upon designation, annually, and at other times requested by ADEQ. If a valve leak is detected, it shall be repaired in accordance with §60.482-7a(d) and (e). The permittee shall not have leak percentage greater than 2%, determined as described in §60.485a(h). [40 CFR Part §60.483-1a]
- 28. The permittee shall comply with the recordkeeping requirements of 40 CFR Part 60 Subpart VVa.
  - a. The permittee shall record the operator identification, equipment identification, date of monitoring, and instrument reading for each monitoring event.[§60.486a(a)]
  - b. For each leak that is detected a weather-proof visible id must be attached. For a valve it can be removed after two successive months of monitoring without a leak. [§60.486a(b)]
  - c. Maintain a log of information required by §60.486a(c) for each leak detected for a period of two years.
  - d. Maintain a log of the information pertaining to all equipment subject to NSPS Subpart VVa as listed in §60.486a(e).
  - e. For valves, pump, and connectors that are designated as unsafe or difficult to monitor maintain a log of these, reason for the designation, and a plan/schedule for monitoring. [§60.486a(f)]

Permit #: 1227-AOP-R0

AFIN: 70-00039

- f. Maintain a schedule of monitoring, percent of valves found leaking during each monitoring period for facilities choosing the alternative monitoring requirements of §60.483-2a. [60.486a(g)]
- g. Maintain a log of design criterion for pumps in light liquid service or compressors with dual seals that are exempt. [§60.486a(h)]
- h. Maintain records demonstrating design capacity, raw materials, and/or analysis demonstrating affected facilities are exempt or not in VOC service. [40.486a(i)]
- i. Maintain readily accessible records to demonstrate a piece of equipment is not in VOC service. [40.486a(j)]
- 29. The permittee shall submit semiannual reports beginning 6 months after the initial startup date. The initial report shall include the information in §60.487a(b)(1) through (5). All semiannual reports shall include the items listed in §60.487a(c) [40 CFR Part §60.487a(a),(b),(c)]
- 30. The permittee shall notify ADEQ of the schedule for the initial performance tests at least 30 days before the initial performance test. [40 Part §60.487a(e)].

#### **NESHAP FF Conditions**

- 31. The facility is an affected source according to 40 CFR Part 61, Subpart FF (Appendix I) *National Emission Standard for Benzene Waste Operations*. [Regulation 19 §19.304 and 40 CFR Part 61, Subpart FF]
  - a. The owner and operator shall determine the total annual benzene quantity from facility waste by the procedures outlined in §61.355 (a). [40 CFR Part 61, Subpart FF, §61.355]
  - b. The facility shall comply with all record keeping requirements outlined in §61.356 (b). [40 CFR Part 61, Subpart FF, §61.355 (a)]
  - c. The facility shall submit reports to the Department by following the procedures of §61.357 (a) (1)-(4). In cases where the total annual benzene quantity is less than 1 Mg/yr [as determined in Plantwide Condition #31a.], reports will comply with §61.357 (b). In cases where the total annual benzene quantity is greater than 1 Mg/yr but less than 10 Mg/yr, reports will comply with §61.357 (c). And when the total annual benzene quantity is greater than 10 Mg/yr, reports will comply with §61.357 (d). [40 CFR Part 61, Subpart FF, §61.357]

4

Permit #: 1227-AOP-R0

AFIN: 70-00039

# **NSPS Subpart QQQ**

- 32. The wastewater system drain systems Reformer/LHT-2 expansion project, oil-water separator (SN-24), aggregated flow lines/junction boxes and associated equipment are subject to 40 CFR Part 60, Subpart QQQ —Standards of Performance for VOC Emissions From Petroleum Refinery Wastewater Systems. The permittee shall comply with all applicable provisions of 40 CFR Part 60, Subpart QQQ which includes, but is not limited to Plantwide Condition #33 through #53. [40 CFR Part 60, Subpart QQQ and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- The permittee shall comply with the requirements of Plantwide Conditions #34 through #53 except during periods of startup, shutdown, or malfunction. [40 CFR Part 60, Subpart QQQ §60.692-1(a) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 34. The permittee shall demonstrate compliance with the following exclusions through compliance with Plantwide Conditions #48, #49, and #50. [40 CFR Part 60, Subpart QQQ §60.690 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
  - a. Storm water sewer systems are not subject to the requirements of this subpart.
  - b. Ancillary equipment, which is physically separate from the wastewater system and does not come in contact with or store oily wastewater, is not subject to the requirements of this subpart.
  - c. Non-contact cooling water systems are not subject to the requirements of this subpart.
- 35. The permittee shall insure that the individual drain systems subject to this subpart are meet the following requirements: [40 CFR Part 60, Subpart QQQ §60.692-2 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
  - a. Each drain shall be equipped with water seal controls.
  - b. Each drain in active service shall be checked by visual or physical inspection initially and monthly thereafter for indications of low water levels or other conditions that would reduce the effectiveness of the water seal controls.
  - c. Except as provided in paragraph (d) of this condition, each drain out of active service shall be checked by visual or physical inspection initially and weekly thereafter for indications of low water levels or other problems that could result in VOC emissions.
  - d. As an alternative to the requirements in paragraph (c) of this condition, if an owner or operator elects to install a tightly sealed cap or plug over a drain that is out of service, inspections shall be conducted initially and semiannually to ensure caps or plugs are in place and properly installed.
  - e. Whenever low water levels or missing or improperly installed caps or plugs are identified, water shall be added or first efforts at repair shall be made as soon as practicable, but not later than 24 hours after detection, except as provided in § 60.692-6.

Permit #: 1227-AOP-R0

AFIN: 70-00039

- f. The Junction boxes shall be equipped with a cover and may have an open vent pipe. The vent pipe shall be at least 90 cm (3 ft) in length and shall not exceed 10.2 cm (4 in) in diameter.
- g. Junction box covers shall have a tight seal around the edge and shall be kept in place at all times, except during inspection and maintenance.
- h. Junction boxes shall be visually inspected initially and semiannually thereafter to ensure that the cover is in place and to ensure that the cover has a tight seal around the edge.
- i. If a broken seal or gap is identified, first effort at repair shall be made as soon as practicable, but not later than 15 calendar days after the broken seal or gap is identified, except as provided in § 60.692-6.
- j. Sewer lines shall not be open to the atmosphere and shall be covered or enclosed in a manner so as to have no visual gaps or cracks in joints, seals, or other emission interfaces.
- k. The portion of each unburied sewer line shall be visually inspected initially and semiannually thereafter for indication of cracks, gaps, or other problems that could result in VOC emissions.
- 1. Whenever cracks, gaps, or other problems are detected, repairs shall be made as soon as practicable, but not later than 15 calendar days after identification, except as provided in § 60.692-6.
- m. Except as provided in paragraph (n) of this condition, each modified or reconstructed individual drain system that has a catch basin in the existing configuration prior to May 4, 1987 shall be exempt from the provisions of this section.
- n. Refinery wastewater routed through new process drains and a new first common downstream junction box, either as part of a new individual drain system or an existing individual drain system, shall not be routed through a downstream catch basin.
- 36. The permittee shall insure that the Oil-water separators subject to this subpart are meet the following requirements: [40 CFR Part 60, Subpart QQQ §60.692-3 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
  - a. Each oil-water separator tank, slop oil tank, storage vessel, or other auxiliary equipment subject to the requirements of this subpart shall be equipped and operated with a fixed roof, which meets the following specifications, except as provided in paragraph (d) of this section or in § 60.693-2.
  - b. The fixed roof shall be installed to completely cover the separator tank, slop oil tank, storage vessel, or other auxiliary equipment with no separation between the roof and the wall.
  - c. The vapor space under a fixed roof shall not be purged unless the vapor is directed to a control device.
  - d. If the roof has access doors or openings, such doors or openings shall be gasketed, latched, and kept closed at all times during operation of the separator system, except during inspection and maintenance.

Permit #: 1227-AOP-R0

AFIN: 70-00039

e. Roof seals, access doors, and other openings shall be checked by visual inspection initially and semiannually thereafter to ensure that no cracks or gaps occur between the roof and wall and that access doors and other openings are closed and gasketed properly.

- f. When a broken seal or gasket or other problem is identified, first efforts at repair shall be made as soon as practicable, but not later than 15 calendar days after it is identified, except as provided in § 60.692-6.
- g. Slop oil from an oil-water separator tank and oily wastewater from slop oil handling equipment shall be collected, stored, transported, recycled, reused, or disposed of in an enclosed system. Once slop oil is returned to the process unit or is disposed of, it is no longer within the scope of this subpart. Equipment used in handling slop oil shall be equipped with a fixed roof meeting the requirements of paragraph (a) of this condition.
- h. Each oil-water separator tank, slop oil tank, storage vessel, or other auxiliary equipment that is required to comply with paragraph (a) of this condition, and not paragraph (b) of this condition, may be equipped with a pressure control valve as necessary for proper system operation. The pressure control valve shall be set at the maximum pressure necessary for proper system operation, but such that the value will not vent continuously.
- 37. The permittee may delay the repair required by this subpart if the repair is technically impossible without a complete or partial refinery or process unit shutdown. Repair of such equipment shall occur before the end of the next refinery or process unit shutdown. [40 CFR Part 60, Subpart QQQ §60.692-6 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 38. The permittee may delay compliance of modified individual drain systems with ancillary downstream treatment components if compliance with the provisions of this subpart cannot be achieved without a refinery or process unit shutdown. Installation of equipment necessary to comply with the provisions of this subpart shall occur no later than the next scheduled refinery or process unit shutdown. [40 CFR Part 60, Subpart QQQ § 60.692-7 and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 39. The permittee shall inspect equipment subject to Subpart QQQ for indications of potential emissions, defects, or other problems that may cause the requirements of this subpart not to be met before the applicable equipment is used. Points of inspection shall include, but are not limited to, seals, flanges, joints, gaskets, hatches, caps, and plugs. [40 CFR Part 60, Subpart QQQ §60.696(a) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 40. The permittee shall maintain records required for compliance of this subpart for a minimum of 2 years after being recorded unless otherwise noted. [40 CFR Part 60, Subpart QQQ § 60.697(a) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

- 41. The permittee shall maintain records of inspections for the drain systems specified in Plantwide Condition #35. These records shall include the inspection location, date, and any corrective action for each drain when the water seal is dry or otherwise breached, when a drain cap or plug is missing or improperly installed, or other problem is identified that could result in VOC emissions, as determined during the initial and periodic visual or physical inspection. [40 CFR Part 60, Subpart QQQ §60.697(b)(1) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 42. The permittee shall maintain records of inspections for the junction boxes specified in Plantwide Condition #35. These records shall include the inspection location, date, and any corrective action when a broken seal, gap, or other problem is identified that could result in VOC emissions. [40 CFR Part 60, Subpart QQQ §60.697(b)(2) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 43. The permittee shall maintain records of inspections for the sewer lines specified in Plantiwde Condition #35. These records shall include the inspection location, date, and any corrective action required when a problem is identified that could result in VOC emissions. [40 CFR Part 60, Subpart QQQ §60.697(b)(3) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 44. The permittee shall maintain records of inspections for the oil-water separators specified in Plantwide Condition #36. These records shall include the inspection location, date, and any corrective action required when a problem is identified that could result in VOC emissions. [40 CFR Part 60, Subpart QQQ §60.697(c) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 45. If an emission point cannot be repaired or corrected without a process unit shutdown as specified in Plantwide Condition #37, the permittee shall maintain a record containing the following information: [40 CFR Part 60, Subpart QQQ §60.697(e) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
  - a. The expected date of a successful repair
  - b. The reason for the delay
  - c. The signature of the owner or operator (or designee) whose decision it was that repair could not be effected without refinery or process shutdown
  - d. The date of successful repair or corrective action
- 46. The permittee shall maintain a copy of the design specifications for all equipment used to comply with the provisions of this subpart for the life of the source in a readily accessible location. These design specifications must contain the following information: [40 CFR Part 60, Subpart QQQ §60.697(f)(1-2) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
  - a. Detailed schematics, and piping and instrumentation diagrams
  - b. The dates and descriptions of any changes in the design specifications

Permit #: 1227-AOP-R0

AFIN: 70-00039

53. If compliance with the provisions of this subpart is delayed pursuant to Plantwide Condition #32, the notification required under 40 CFR 60.7(a)(4) shall include the estimated date of the next scheduled refinery or process unit shutdown after the date of notification and the reason why compliance with the standards is technically impossible without a refinery or process unit shutdown. [40 CFR Part 60, Subpart QQQ §60.698(e) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

#### Title VI Provisions

- 54. The permittee must comply with the standards for labeling of products using ozone-depleting substances. [40 CFR Part 82, Subpart E]
  - a. All containers containing a class I or class II substance stored or transported, all products containing a class I substance, and all products directly manufactured with a class I substance must bear the required warning statement if it is being introduced to interstate commerce pursuant to §82.106.
  - b. The placement of the required warning statement must comply with the requirements pursuant to §82.108.
  - c. The form of the label bearing the required warning must comply with the requirements pursuant to §82.110.
  - d. No person may modify, remove, or interfere with the required warning statement except as described in §82.112.
- 55. The permittee must comply with the standards for recycling and emissions reduction, except as provided for MVACs in Subpart B. [40 CFR Part 82, Subpart F]
  - a. Persons opening appliances for maintenance, service, repair, or disposal must comply with the required practices pursuant to §82.156.
  - b. Equipment used during the maintenance, service, repair, or disposal of appliances must comply with the standards for recycling and recovery equipment pursuant to §82.158.
  - c. Persons performing maintenance, service repair, or disposal of appliances must be certified by an approved technician certification program pursuant to §82.161.
  - d. Persons disposing of small appliances, MVACs, and MVAC like appliances must comply with record keeping requirements pursuant to §82.166. ("MVAC like appliance" as defined at §82.152)
  - e. Persons owning commercial or industrial process refrigeration equipment must comply with leak repair requirements pursuant to §82.156.
  - f. Owners/operators of appliances normally containing 50 or more pounds of refrigerant must keep records of refrigerant purchased and added to such appliances pursuant to §82.166.
- 56. If the permittee manufactures, transforms, destroys, imports, or exports a class I or class II substance, the permittee is subject to all requirements as specified in 40 CFR Part 82, Subpart A, Production and Consumption Controls.

Permit #: 1227-AOP-R0

AFIN: 70-00039

- 47. If the permittee elects to install a tightly sealed cap or plug over a drain that is out of active service, the permittee shall maintain for the life of a facility in a readily accessible location, plans or specifications which indicate the location of such drains. [40 CFR Part 60, Subpart QQQ §60.697(g) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 48. The permittee shall maintain for the life of the facility in a readily accessible location, plans or specifications which demonstrate that no wastewater from any process units or equipment is directly discharged to any applicable stormwater sewer system. [40 CFR Part 60, Subpart QQQ §60.697(h) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 49. The permittee shall maintain for the life of the facility in a readily accessible location, plans or specifications which demonstrate that any applicable ancillary equipment does not come in contact with or store oily wastewater. [40 CFR Part 60, Subpart QQQ §60.697(i) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 50. The permittee shall maintain for the life of the facility in a readily accessible location, plans or specifications which demonstrate that the applicable cooling water does not contact hydrocarbons or oily wastewater and is not recirculated through a cooling tower. [40 CFR Part 60, Subpart QQQ §60.697(j) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 51. The permittee shall submit to the Administrator within 60 days after initial startup a certification that the equipment necessary to comply with these standards has been installed and that the required initial inspections or tests of process drains, sewer lines, junction boxes, oil-water separators, and closed vent systems and control devices have been carried out in accordance with these standards. Thereafter, the owner or operator shall submit to the Administrator semiannually a certification that all of the required inspections have been carried out in accordance with these standards. [40 CFR Part 60, Subpart QQQ §60.698(b) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 52. The permittee shall submit an initial and semiannual report to the Department that summarizes all inspections when a water seal was dry or otherwise breached, when a drain cap or plug was missing or improperly installed, or when cracks, gaps, or other problems were identified that could result in VOC emissions, including information about the repairs or corrective action taken. The initial and semiannual report shall be submitted in accordance with General Provision No. 7. [40 CFR Part 60, Subpart QQQ §60.698(c) and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]

Permit #: 1227-AOP-R0

AFIN: 70-00039

57. If the permittee performs a service on motor (fleet) vehicles when this service involves ozone depleting substance refrigerant (or regulated substitute substance) in the motor vehicle air conditioner (MVAC), the permittee is subject to all the applicable requirements as specified in 40 CFR part 82, Subpart B, Servicing of Motor Vehicle Air Conditioners.

The term "motor vehicle" as used in Subpart B does not include a vehicle in which final assembly of the vehicle has not been completed. The term "MVAC" as used in Subpart B does not include the air tight sealed refrigeration system used as refrigerated cargo, or the system used on passenger buses using HCFC 22 refrigerant.

58. The permittee can switch from any ozone depleting substance to any alternative listed in the Significant New Alternatives Program (SNAP) promulgated pursuant to 40 CFR Part 82, Subpart G.

#### Permit Shield

59. Compliance with the conditions of this permit shall be deemed compliance with all applicable requirements, as of the date of permit issuance, included in and specifically identified in the following table of this condition. The permit specifically identifies the following as applicable requirements based upon the information submitted by the permittee in an application dated March 11, 2013.

**Applicable Regulations** 

Source No.	Regulation	Description	
SN-12	40 CFR Part 60, Subpart J	NSPS for Petroleum Refineries	
SN-36 SN-38 SN-39 SN-40	40 CFR Part 60, Subpart Ja	NSPS for Petroleum Refineries for Which Construction, Reconstruction, or Modification Commenced After May 14, 2007	
SN-27h	40 CFR Part 60, Subpart Kb	NSPS for Storage Vessels for Petroleum Liquids for Which Construction, Reconstruction, or Modification Commenced After July 23, 1984	
SN-25	40 CFR Part 60, Subpart GG	NSPS for Stationary Gas Turbines	
SN-27c	40 CFR Part 60, Subpart UU	NSPS for Asphalt Processing and Asphalt Roofing Manufacture	
Lube Hydrotreater Unit No. 2	40 CFR Part 60, Subpart GGGa	NSPS for Equipment Leaks of VOC in Petroleum Refineries for Which Construction, Reconstruction, or Modification Commenced After November 7, 2006	

Martin Operating Partnership L.P. Permit #: 1227-AOP-R0

AFIN: 70-00039

Source No.	Regulation	Description
SN-24	40 CFR Part 60, Subpart QQQ	NSPS for VOC Emissions From Petroleum refinery Wastewater Systems
SN-41 SN-43	40 CFR Part 60, Subpart IIII	NSPS for Stationary Compression Ignition Internal Combustion Engines
Facility	40 CFR Part 61, Subpart FF	National Emission Standards for Benzene Waste Operations
SN-42	40 CFR Part 63, Subpart ZZZZ	NESHAP for Stationary Reciprocating Internal Combustion Engines
SN-27g	40 CFR Part 63, Subpart CCCCCC	NESHAP for Source Category: gasoline Dispensing Facilities

The permit specifically identifies the following as inapplicable based upon information submitted by the permittee in an application dated March 11, 2013.

## Inapplicable Regulations

Source No.	Regulation	Description	
SN-26	40 CFR Part 60, Subpart Dc	NSPS for Small Industrial-Commercial- Institutional Steam Generating Units	
SN-27a-1, SN-28, SN-29	40 CFR Part 60, Subpart K	NSPS for Storage Vessels for Petroleum Liquids for Which Construction, Reconstruction, or Modification Commenced After June 11, 1973, and prior to May 19, 1978	
SN-27a-1 SN-28, SN-29	40 CFR Part 60, Subpart Ka	NSPS for Storage vessels for Petroleum Liquids for Which Construction, Reconstruction, or Modification Commenced After May 18, 1978 and Prior to July 23, 1984	
SN-38 SN-39	40 CFR Part 60, Subpart GGG	NSPS for Equipment Leaks for VOCs in Petroleum Refineries After January 4, 1983 and before November 7, 2006	
Facility	40 CFR Part 60, Subpart KKK	NSPS for Stationary Combustion Turbines	
Facility	40 CFR Part 61, Subpart J	National Emission Standard for Equipment Leaks (Fugitive Emission Sources) of Benzene	
Facility	40 CFR Part 63, Subpart CC	NESHAP for Petroleum Refineries	
Facility	40 CFR Part 63, Subpart UUU	NESHAP for Petroleum refineries: Catalytic Cracking Units, Catalytic reforming Units, and Sulfur Recovery Units	

Martin Operating Partnership L.P. Permit #: 1227-AOP-R0 AFIN: 70-00039

Source No.	Regulation	Description
SN-25	40 CFR Part 63, Subpart YYYY	NESHAP for Station Combustion Turbines
SN-26	40 CFR Part 63, Subpart DDDDD	NESHAP for Industrial, Commercial, and Institutional Boilers and Process Heaters
SN-26	40 CFR Part 63, Subpart JJJJJJ	NESHAP for Industrial, Commercial, and Institutional Boilers Area Sources

Permit #: 1227-AOP-R0

AFIN: 70-00039

#### SECTION VII: INSIGNIFICANT ACTIVITIES

The following sources are insignificant activities. Any activity that has a state or federal applicable requirement shall be considered a significant activity even if this activity meets the criteria of §26.304 of Regulation 26 or listed in the table below. Insignificant activity determinations rely upon the information submitted by the permittee in an application dated March 13, 2013.

Description	Category
50 HP (0.125 MMBtu/HR) Low Pressure Boiler	A-1
Lab Equipment	A-5
Cooling Tower No. 1 and Cooling Tower No. 2	A-13
Asphalt Tank Heater	A-13
Lube Oil Packaging Operation	A-13

Permit #: 1227-AOP-R0

AFIN: 70-00039

#### SECTION VIII: GENERAL PROVISIONS

- 1. Any terms or conditions included in this permit which specify and reference Arkansas Pollution Control & Ecology Commission Regulation 18 or the Arkansas Water and Air Pollution Control Act (A.C.A. §8-4-101 et seq.) as the sole origin of and authority for the terms or conditions are not required under the Clean Air Act or any of its applicable requirements, and are not federally enforceable under the Clean Air Act. Arkansas Pollution Control & Ecology Commission Regulation 18 was adopted pursuant to the Arkansas Water and Air Pollution Control Act (A.C.A. §8-4-101 et seq.). Any terms or conditions included in this permit which specify and reference Arkansas Pollution Control & Ecology Commission Regulation 18 or the Arkansas Water and Air Pollution Control Act (A.C.A. §8-4-101 et seq.) as the origin of and authority for the terms or conditions are enforceable under this Arkansas statute. [40 CFR 70.6(b)(2)]
- 2. This permit shall be valid for a period of five (5) years beginning on the date this permit becomes effective and ending five (5) years later. [40 CFR 70.6(a)(2) and Regulation 26 §26.701(B)]
- 3. The permittee must submit a complete application for permit renewal at least six (6) months before permit expiration. Permit expiration terminates the permittee's right to operate unless the permittee submitted a complete renewal application at least six (6) months before permit expiration. If the permittee submits a complete application, the existing permit will remain in effect until the Department takes final action on the renewal application. The Department will not necessarily notify the permittee when the permit renewal application is due. [Regulation 26 §26.406]
- 4. Where an applicable requirement of the Clean Air Act, as amended, 42 U.S.C. 7401, et seq. (Act) is more stringent than an applicable requirement of regulations promulgated under Title IV of the Act, the permit incorporates both provisions into the permit, and the Director or the Administrator can enforce both provisions. [40 CFR 70.6(a)(1)(ii) and Regulation 26 §26.701(A)(2)]
- 5. The permittee must maintain the following records of monitoring information as required by this permit.
  - a. The date, place as defined in this permit, and time of sampling or measurements;
  - b. The date(s) analyses performed;
  - c. The company or entity performing the analyses;
  - d. The analytical techniques or methods used;
  - e. The results of such analyses; and
  - f. The operating conditions existing at the time of sampling or measurement.

[40 CFR 70.6(a)(3)(ii)(A) and Regulation 26 §26.701(C)(2)]

6. The permittee must retain the records of all required monitoring data and support information for at least five (5) years from the date of the monitoring sample,

Permit #: 1227-AOP-R0

AFIN: 70-00039

measurement, report, or application. Support information includes all calibration and maintenance records and all original strip chart recordings for continuous monitoring instrumentation, and copies of all reports required by this permit. [40 CFR 70.6(a)(3)(ii)(B) and Regulation 26 §26.701(C)(2)(b)]

7. The permittee must submit reports of all required monitoring every six (6) months. If the permit establishes no other reporting period, the reporting period shall end on the last day of the month six months after the issuance of the initial Title V permit and every six months thereafter. The report is due on the first day of the second month after the end of the reporting period. The first report due after issuance of the initial Title V permit shall contain six months of data and each report thereafter shall contain 12 months of data. The report shall contain data for all monitoring requirements in effect during the reporting period. If a monitoring requirement is not in effect for the entire reporting period, only those months of data in which the monitoring requirement was in effect are required to be reported. The report must clearly identify all instances of deviations from permit requirements. A responsible official as defined in Regulation No. 26, §26.2 must certify all required reports. The permittee will send the reports to the address below:

Arkansas Department of Environmental Quality Air Division ATTN: Compliance Inspector Supervisor 5301 Northshore Drive North Little Rock, AR 72118-5317

[40 CFR 70.6(a)(3)(iii)(A) and Regulation 26 §26.701(C)(3)(a)]

- 8. The permittee shall report to the Department all deviations from permit requirements, including those attributable to upset conditions as defined in the permit.
  - a. For all upset conditions (as defined in Regulation 19, § 19.601), the permittee will make an initial report to the Department by the next business day after the discovery of the occurrence. The initial report may be made by telephone and shall include:
    - i. The facility name and location;
    - ii. The process unit or emission source deviating from the permit limit;
    - iii. The permit limit, including the identification of pollutants, from which deviation occurs;
    - iv. The date and time the deviation started:
    - v. The duration of the deviation:
    - vi. The average emissions during the deviation;
    - vii. The probable cause of such deviations;
    - viii. Any corrective actions or preventive measures taken or being taken to prevent such deviations in the future; and
    - ix. The name of the person submitting the report.

Permit #: 1227-AOP-R0

AFIN: 70-00039

The permittee shall make a full report in writing to the Department within five (5) business days of discovery of the occurrence. The report must include, in addition to the information required by the initial report, a schedule of actions taken or planned to eliminate future occurrences and/or to minimize the amount the permit's limits were exceeded and to reduce the length of time the limits were exceeded. The permittee may submit a full report in writing (by facsimile, overnight courier, or other means) by the next business day after discovery of the occurrence, and the report will serve as both the initial report and full report.

b. For all deviations, the permittee shall report such events in semi-annual reporting and annual certifications required in this permit. This includes all upset conditions reported in 8a above. The semi-annual report must include all the information as required by the initial and full reports required in 8a.

[Regulation 19 §19.601 and §19.602, Regulation 26 §26.701(C)(3)(b), and 40 CFR 70.6(a)(3)(iii)(B)]

- 9. If any provision of the permit or the application thereof to any person or circumstance is held invalid, such invalidity will not affect other provisions or applications hereof which can be given effect without the invalid provision or application, and to this end, provisions of this Regulation are declared to be separable and severable. [40 CFR 70.6(a)(5), Regulation 26 §26.701(E), and A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 10. The permittee must comply with all conditions of this Part 70 permit. Any permit noncompliance with applicable requirements as defined in Regulation 26 constitutes a violation of the Clean Air Act, as amended, 42 U.S.C. §7401, et seq. and is grounds for enforcement action; for permit termination, revocation and reissuance, for permit modification; or for denial of a permit renewal application. [40 CFR 70.6(a)(6)(i) and Regulation 26 §26.701(F)(1)]
- 11. It shall not be a defense for a permittee in an enforcement action that it would have been necessary to halt or reduce the permitted activity to maintain compliance with the conditions of this permit. [40 CFR 70.6(a)(6)(ii) and Regulation 26 §26.701(F)(2)]
- 12. The Department may modify, revoke, reopen and reissue the permit or terminate the permit for cause. The filing of a request by the permittee for a permit modification, revocation and reissuance, termination, or of a notification of planned changes or anticipated noncompliance does not stay any permit condition. [40 CFR 70.6(a)(6)(iii) and Regulation 26 §26.701(F)(3)]
- This permit does not convey any property rights of any sort, or any exclusive privilege. [40 CFR 70.6(a)(6)(iv) and Regulation 26 §26.701(F)(4)]

Permit #: 1227-AOP-R0

AFIN: 70-00039

- 14. The permittee must furnish to the Director, within the time specified by the Director, any information that the Director may request in writing to determine whether cause exists for modifying, revoking and reissuing, or terminating the permit or to determine compliance with the permit. Upon request, the permittee must also furnish to the Director copies of records required by the permit. For information the permittee claims confidentiality, the Department may require the permittee to furnish such records directly to the Director along with a claim of confidentiality. [40 CFR 70.6(a)(6)(v) and Regulation 26 §26.701(F)(5)]
- 15. The permittee must pay all permit fees in accordance with the procedures established in Regulation 9. [40 CFR 70.6(a)(7) and Regulation 26 §26.701(G)]
- 16. No permit revision shall be required, under any approved economic incentives, marketable permits, emissions trading and other similar programs or processes for changes provided for elsewhere in this permit. [40 CFR 70.6(a)(8) and Regulation 26 §26.701(H)]
- 17. If the permit allows different operating scenarios, the permittee shall, contemporaneously with making a change from one operating scenario to another, record in a log at the permitted facility a record of the operational scenario. [40 CFR 70.6(a)(9)(i) and Regulation 26 §26.701(I)(1)]
- 18. The Administrator and citizens may enforce under the Act all terms and conditions in this permit, including any provisions designed to limit a source's potential to emit, unless the Department specifically designates terms and conditions of the permit as being federally unenforceable under the Act or under any of its applicable requirements. [40 CFR 70.6(b) and Regulation 26 §26.702(A) and (B)]
- 19. Any document (including reports) required by this permit must contain a certification by a responsible official as defined in Regulation 26, §26.2. [40 CFR 70.6(c)(1) and Regulation 26 §26.703(A)]
- 20. The permittee must allow an authorized representative of the Department, upon presentation of credentials, to perform the following: [40 CFR 70.6(c)(2) and Regulation 26 §26.703(B)]
  - a. Enter upon the permittee's premises where the permitted source is located or emissions related activity is conducted, or where records must be kept under the conditions of this permit;
  - b. Have access to and copy, at reasonable times, any records required under the conditions of this permit;
  - c. Inspect at reasonable times any facilities, equipment (including monitoring and air pollution control equipment), practices, or operations regulated or required under this permit; and

Permit #: 1227-AOP-R0

AFIN: 70-00039

- d. As authorized by the Act, sample or monitor at reasonable times substances or parameters for assuring compliance with this permit or applicable requirements.
- 21. The permittee shall submit a compliance certification with the terms and conditions contained in the permit, including emission limitations, standards, or work practices. The permittee must submit the compliance certification annually. If the permit establishes no other reporting period, the reporting period shall end on the last day of the anniversary month of the initial Title V permit. The report is due on the first day of the second month after the end of the reporting period. The permittee must also submit the compliance certification to the Administrator as well as to the Department. All compliance certifications required by this permit must include the following: [40 CFR 70.6(c)(5) and Regulation 26 §26.703(E)(3)]
  - a. The identification of each term or condition of the permit that is the basis of the certification;
  - b. The compliance status;
  - c. Whether compliance was continuous or intermittent;
  - d. The method(s) used for determining the compliance status of the source, currently and over the reporting period established by the monitoring requirements of this permit; and
  - e. Such other facts as the Department may require elsewhere in this permit or by §114(a)(3) and §504(b) of the Act.
- 22. Nothing in this permit will alter or affect the following: [Regulation 26 §26.704(C)]
  - a. The provisions of Section 303 of the Act (emergency orders), including the authority of the Administrator under that section;
  - b. The liability of the permittee for any violation of applicable requirements prior to or at the time of permit issuance;
  - c. The applicable requirements of the acid rain program, consistent with §408(a) of the Act; or
  - d. The ability of EPA to obtain information from a source pursuant to §114 of the Act.
- 23. This permit authorizes only those pollutant emitting activities addressed in this permit. [A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311]
- 24. The permittee may request in writing and at least 15 days in advance of the deadline, an extension to any testing, compliance or other dates in this permit. No such extensions are authorized until the permittee receives written Department approval. The Department may grant such a request, at its discretion in the following circumstances:
  - a. Such an extension does not violate a federal requirement;
  - b. The permittee demonstrates the need for the extension; and

Permit #: 1227-AOP-R0

AFIN: 70-00039

c. The permittee documents that all reasonable measures have been taken to meet the current deadline and documents reasons it cannot be met.

[Regulation 18 §18.314(A), Regulation 19 §19.416(A), Regulation 26 §26.1013(A), A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR Part 52, Subpart E]

- 25. The permittee may request in writing and at least 30 days in advance, temporary emissions and/or testing that would otherwise exceed an emission rate, throughput requirement, or other limit in this permit. No such activities are authorized until the permittee receives written Department approval. Any such emissions shall be included in the facility's total emissions and reported as such. The Department may grant such a request, at its discretion under the following conditions:
  - a. Such a request does not violate a federal requirement;
  - b. Such a request is temporary in nature;
  - c. Such a request will not result in a condition of air pollution;
  - d. The request contains such information necessary for the Department to evaluate the request, including but not limited to, quantification of such emissions and the date/time such emission will occur;
  - e. Such a request will result in increased emissions less than five tons of any individual criteria pollutant, one ton of any single HAP and 2.5 tons of total HAPs; and
  - f. The permittee maintains records of the dates and results of such temporary emissions/testing.

[Regulation 18 §18.314(B), Regulation 19 §19.416(B), Regulation 26 §26.1013(B), A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR Part 52, Subpart E]

- 26. The permittee may request in writing and at least 30 days in advance, an alternative to the specified monitoring in this permit. No such alternatives are authorized until the permittee receives written Department approval. The Department may grant such a request, at its discretion under the following conditions:
  - a. The request does not violate a federal requirement:
  - b. The request provides an equivalent or greater degree of actual monitoring to the current requirements; and
  - c. Any such request, if approved, is incorporated in the next permit modification application by the permittee.

[Regulation 18 §18.314(C), Regulation 19 §19.416(C), Regulation 26 §26.1013(C), A.C.A. §8-4-203 as referenced by §8-4-304 and §8-4-311, and 40 CFR Part 52, Subpart E]

Appendix A
40 CFR Part 60, Subpart J
Standards of Performance for Petroleum Refineries

#### **Environmental Protection Agency**

§ 60.92 Standard for particulate matter.

§ 60.100

- (1) If a source processes elemental sulfur or an ore that contains elemental sulfur and uses air to supply oxygen, the following procedure may be used instead of determining the volumetric flow rate and production rate:
- (i) The integrated technique of Method 3 is used to determine the  $O_2$  concentration and, if required,  $CO_2$  concentration.
- (ii) The  $SO_2$  or acid mist emission rate is calculated as described in  $\S 60.84(d)$ , substituting the acid mist concentration for  $C_s$  as appropriate.

[54 FR 6666, Feb. 14, 1989]

Subpart I—Standards of Performance for Hot Mix Asphalt Facilities

## § 60.90 Applicability and designation of affected facility.

- (a) The affected facility to which the provisions of this subpart apply is each hot mix asphalt facility. For the purpose of this subpart, a hot mix asphalt facility is comprised only of any combination of the following: dryers; systems for screening, handling, storing, and weighing hot aggregate; systems for loading, transferring, and storing mineral filler, systems for mixing hot mix asphalt; and the loading, transfer, and storage systems associated with emission control systems.
- (b) Any facility under paragraph (a) of this section that commences construction or modification after June 11, 1973, is subject to the requirements of this subpart.

[42 FR 37936, July 25, 1977, as amended at 51 FR 12325, Apr. 10, 1986]

#### § 60.91 Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the Act and in subpart A of this part.

(a) Hot mix asphalt facility means any facility, as described in §60.90, used to manufacture hot mix asphalt by heating and drying aggregate and mixing with asphalt cements.

[51 FR 12325, Apr. 10, 1986]

- (a) On and after the date on which the performance test required to be conducted by §60.8 is completed, no owner or operator subject to the provisions of this subpart shall discharge or cause the discharge into the atmosphere from any affected facility any gases which:
- (1) Contain particulate matter in excess of 90 mg/dscm (0.04 gr/dscf).
- (2) Exhibit 20 percent opacity, or greater.

[39 FR 9314, Mar. 8, 1974, as amended at 40 FR 46259, Oct. 6, 1975]

#### $\S 60.93$ Test methods and procedures.

- (a) In conducting the performance tests required in §60.8, the owner or operator shall use as reference methods and procedures the test methods in appendix A of this part or other methods and procedures as specified in this section, except as provided in §60.8(b).
- (b) The owner or operator shall determine compliance with the particulate matter standards in §60.92 as follows:
- (1) Method 5 shall be used to determine the particulate matter concentration. The sampling time and sample volume for each run shall be at least 60 minutes and 0.90 dscm (31.8 dscf).
- (2) Method 9 and the procedures in §60.11 shall be used to determine opacity.

[54 FR 6667, Feb. 14, 1989]

#### Subpart J—Standards of Performance for Petroleum Refineries

# § 60.100 Applicability, designation of affected facility, and reconstruction.

(a) The provisions of this subpart are applicable to the following affected facilities in petroleum refineries: fluid catalytic cracking unit catalyst regenerators, fuel gas combustion devices, and all Claus sulfur recovery plants except Claus plants with a design capacity for sulfur feed of 20 long tons per day (LTD) or less. The Claus sulfur recovery plant need not be physically located within the boundaries of a petroleum refinery to be an affected facility, provided it processes gases produced within a petroleum refinery.

#### §60.101

- (b) Any fluid catalytic cracking unit catalyst regenerator or fuel gas combustion device under paragraph (a) of this section other than a flare which commences construction, reconstruction or modification after June 11, 1973, and on or before May 14, 2007, or any fuel gas combustion device under paragraph (a) of this section that is also a flare which commences construction, reconstruction or modification after June 11, 1973, and on or before June 24, 2008, or any Claus sulfur recovery plant under paragraph (a) of this section which commences construction, reconstruction or modification after October 4, 1976, and on or before May 14, 2007, is subject to the requirements of this subpart except as provided under paragraphs (c) through (e) of this section.
- (c) Any fluid catalytic cracking unit catalyst regenerator under paragraph (b) of this section which commences construction, reconstruction, or modification on or before January 17, 1984, is exempted from §60.104(b).
- (d) Any fluid catalytic cracking unit in which a contact material reacts with petroleum derivatives to improve feedstock quality and in which the contact material is regenerated by burning off coke and/or other deposits and that commences construction, reconstruction, or modification on or before January 17, 1984, is exempt from this subpart.
- (e) Owners or operators may choose to comply with the applicable provisions of subpart Ja of this part to satisfy the requirements of this subpart for an affected facility.
- (f) For purposes of this subpart, under §60.15, the "fixed capital cost of the new components" includes the fixed capital cost of all depreciable components which are or will be replaced pursuant to all continuous programs of component replacement which are commenced within any 2year period following January 17, 1984. For purposes of this paragraph, "commenced" means that an owner or operator has undertaken a continuous program of component replacement or that an owner or operator has entered into a contractual obligation to undertake and complete, within a reasonable

time, a continuous program of component replacement.

[43 FR 10868, Mar. 15, 1978, as amended at 44 FR 61543, Oct. 25, 1979; 54 FR 34026, Aug. 17, 1989; 73 FR 35865, June 24, 2008; 77 FR 56463, Sep. 12, 2012]

#### § 60.101 Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the Act and in subpart A

- (a) Petroleum refinery means any facility engaged in producing gasoline, kerosene, distillate fuel oils, residual fuel oils, lubricants, or other products through distillation of petroleum or through redistillation, cracking or reforming of unfinished petroleum derivatives.
- (b) *Petroleum* means the crude oil removed from the earth and the oils derived from tar sands, shale, and coal.
- (c) *Process gas* means any gas generated by a petroleum refinery process unit, except fuel gas and process upset gas as defined in this section.
- (d) Fuel gas means any gas which is generated at a petroleum refinery and which is combusted. Fuel gas includes natural gas when the natural gas is combined and combusted in any proportion with a gas generated at a refinery. Fuel gas does not include gases generated by catalytic cracking unit catalyst regenerators and fluid coking burners. Fuel gas does not include vapors that are collected and combusted in a thermal oxidizer or flare installed to control emissions from wastewater treatment units or marine tank vessel loading operations.
- (e) Process upset gas means any gas generated by a petroleum refinery process unit as a result of start-up, shut-down, upset or malfunction.
- (f) Refinery process unit means any segment of the petroleum refinery in which a specific processing operation is conducted.
- (g) Fuel gas combustion device means any equipment, such as process heaters, boilers and flares used to combust fuel gas, except facilities in which gases are combusted to produce sulfur or sulfuric acid.
- (h) Coke burn-off means the coke removed from the surface of the fluid catalytic cracking unit catalyst by

combustion in the catalyst regenerator. The rate of coke burn-off is calculated by the formula specified in §60.106.

- (i) Claus sulfur recovery plant means a process unit which recovers sulfur from hydrogen sulfide by a vapor-phase catalytic reaction of sulfur dioxide and hydrogen sulfide.
- (j) Oxidation control system means an emission control system which reduces emissions from sulfur recovery plants by converting these emissions to sulfur dioxide.
- (k) Reduction control system means an emission control system which reduces emissions from sulfur recovery plants by converting these emissions to hydrogen sulfide.
- (1) Reduced sulfur compounds means hydrogen sulfide  $(H_2S)$ , carbonyl sulfide (COS) and carbon disulfide  $(CS_2)$ .
- (m) Fluid catalytic cracking unit means a refinery process unit in which petroleum derivatives are continuously charged; hydrocarbon molecules in the presence of a catalyst suspended in a fluidized bed are fractured into smaller molecules, or react with a contact material suspended in a fluidized bed to improve feedstock quality for additional processing; and the catalyst or contact material is continuously regenerated by burning off coke and other deposits. The unit includes the riser, reactor, regenerator, air blowers, spent catalyst or contact material stripper, catalyst or contact material recovery equipment, and regenerator equipment for controlling air pollutant emissions and for heat recovery.
- (n) Fluid catalytic cracking unit catalyst regenerator means one or more regenerators (multiple regenerators) which comprise that portion of the fluid catalytic cracking unit in which coke burn-off and catalyst or contact material regeneration occurs, and includes the regenerator combustion air blower(s).
- (0) Fresh feed means any petroleum derivative feedstock stream charged directly into the riser or reactor of a fluid catalytic cracking unit except for petroleum derivatives recycled within the fluid catalytic cracking unit, fractionator, or gas recovery unit.
- (p) Contact material means any substance formulated to remove metals,

sulfur, nitrogen, or any other contaminant from petroleum derivatives.

(q) Valid day means a 24-hour period in which at least 18 valid hours of data are obtained. A "valid hour" is one in which at least 2 valid data points are obtained.

[39 FR 9315, Mar. 8, 1974, as amended at 43 FR 10868, Mar. 15, 1978; 44 FR 13481, Mar. 12, 1979; 45 FR 79453, Dec. 1, 1980; 54 FR 34027, Aug. 17, 1989; 73 FR 35865, June 24, 2008; 77 FR 56463, Sep. 12, 2012]

### § 60.102 Standard for particulate mat-

Each owner or operator of any fluid catalytic cracking unit catalyst regenerator that is subject to the requirements of this subpart shall comply with the emission limitations set forth in this section on and after the date on which the initial performance test, required by §60.8, is completed, but not later than 60 days after achieving the maximum production rate at which the fluid catalytic cracking unit catalyst regenerator will be operated, or 180 days after initial startup, whichever comes first.

- (a) No owner or operator subject to the provisions of this subpart shall discharge or cause the discharge into the atmosphere from any fluid catalytic cracking unit catalyst regenerator:
- (1) Particulate matter in excess of 1.0 kg/Mg (2.0 lb/ton) of coke burn-off in the catalyst regenerator.
- (2) Gases exhibiting greater than 30 percent opacity, except for one sixminute average opacity reading in any one hour period.
- (b) Where the gases discharged by the fluid catalytic cracking unit catalyst regenerator pass through an incinerator or waste heat boiler in which auxiliary or supplemental liquid or solid fossil fuel is burned, particulate matter in excess of that permitted by paragraph (a)(1) of this section may be emitted to the atmosphere, except that the incremental rate of particulate matter emissions shall not exceed 43 grams per Gigajoule (g/GJ) (0.10 lb/million British thermal units (Btu)) of

#### § 60,103

heat input attributable to such liquid or solid fossil fuel.

[39 FR 9315, Mar. 8, 1974, as amended at 42 FR 32427, June 24, 1977; 42 FR 39389, Aug. 4, 1977; 43 FR 10868, Feb. 15, 1978; 54 FR 34027, Aug. 17, 1989; 65 FR 61753, Oct. 17, 2000; 73 FR 35866, June 24, 2008]

## § 60.103 Standard for carbon monoxide.

Each owner or operator of any fluid catalytic cracking unit catalyst regenerator that is subject to the requirements of this subpart shall comply with the emission limitations set forth in this section on and after the date on which the initial performance test, required by §60.8, is completed, but not later than 60 days after achieving the maximum production rate at which the fluid catalytic cracking unit catalyst regenerator will be operated, or 180 days after initial startup, whichever comes first.

(a) No owner or operator subject to the provisions of this subpart shall discharge or cause the discharge into the atmosphere from any fluid catalytic cracking unit catalyst regenerator any gases that contain carbon monoxide (CO) in excess of 500 ppm by volume (dry basis).

[54 FR 34027, Aug. 17, 1989, as amended at 55 FR 40175, Oct. 2, 1990]

#### § 60.104 Standards for sulfur oxides.

Each owner or operator that is subject to the requirements of this subpart shall comply with the emission limitations set forth in this section on and after the date on which the initial performance test, required by §60.8, is completed, but not later than 60 days after achieving the maximum production rate at which the affected facility will be operated, or 180 days after initial startup, whichever comes first.

- (a) No owner or operator subject to the provisions of this subpart shall:
- (1) Burn in any fuel gas combustion device any fuel gas that contains hydrogen sulfide ( $\rm H_2S$ ) in excess of 230 mg/dscm (0.10 gr/dscf). The combustion in a flare of process upset gases or fuel gas that is released to the flare as a result of relief valve leakage or other emergency malfunctions is exempt from this paragraph.

- (2) Discharge or cause the discharge of any gases into the atmosphere from any Claus sulfur recovery plant containing in excess of:
- (i) For an oxidation control system or a reduction control system followed by incineration, 250 ppm by volume (dry basis) of sulfur dioxide (SO<sub>2</sub>) at zero percent excess air.
- (ii) For a reduction control system not followed by incineration, 300 ppm by volume of reduced sulfur compounds and 10 ppm by volume of hydrogen sulfide ( $H_2S$ ), each calculated as ppm  $SO_2$  by volume (dry basis) at zero percent excess air.
- (b) Each owner or operator that is subject to the provisions of this subpart shall comply with one of the following conditions for each affected fluid catalytic cracking unit catalyst regenerator:
- (1) With an add-on control device, reduce  $SO_2$  emissions to the atmosphere by 90 percent or maintain  $SO_2$  emissions to the atmosphere less than or equal to 50 ppm by volume (ppmv), whichever is less stringent; or
- (2) Without the use of an add-on control device to reduce  $SO_2$  emission, maintain sulfur oxides emissions calculated as  $SO_2$  to the atmosphere less than or equal to 9.8 kg/Mg (20 lb/ton) coke burn-off. or
- (3) Process in the fluid catalytic cracking unit fresh feed that has a total sulfur content no greater than 0.30 percent by weight.
- (c) Compliance with paragraph (b)(1), (b)(2), or (b)(3) of this section is determined daily on a 7-day rolling average basis using the appropriate procedures outlined in \$60.106.
- (d) A minimum of 22 valid days of data shall be obtained every 30 rolling successive calendar days when complying with paragraph (b)(1) of this section.

[43 FR 10869, Mar. 15, 1978, as amended at 54 FR 34027, Aug. 17, 1989; 55 FR 40175, Oct. 2, 1990; 65 FR 61754, Oct. 17, 2000; 73 FR 35866, June 24, 2008]

## § 60.105 Monitoring of emissions and operations.

(a) Continuous monitoring systems shall be installed, calibrated, maintained, and operated by the owner or

operator subject to the provisions of this subpart as follows:

- (1) For fluid catalytic cracking unit catalyst regenerators subject to §60.102(a)(2), an instrument for continuously monitoring and recording the opacity of emissions into the atmosphere. The instrument shall be spanned at 60, 70, or 80 percent opacity.
- (2) For fluid catalytic cracking unit catalyst regenerators subject to §60.103(a), an instrument for continuously monitoring and recording the concentration by volume (dry basis) of CO emissions into the atmosphere, except as provided in paragraph (a)(2) (ii) of this section.
- (i) The span value for this instrument is 1,000~ppm CO.
- (ii) A CO continuous monitoring system need not be installed if the owner or operator demonstrates that the average CO emissions are less than 50 ppm (dry basis) and also files a written request for exemption to the Administrator and receives such an exemption. The demonstration shall consist of continuously monitoring CO emissions for 30 days using an instrument that shall meet the requirements of Performance Specification 4 of appendix B of this part. The span value shall be 100 ppm CO instead of 1,000 ppm, and the relative accuracy limit shall be 10 percent of the average CO emissions or 5 ppm CO, whichever is greater. For instruments that are identical to Method 10 and employ the sample conditioning system of Method 10A, the alternative relative accuracy test procedure in §10.1 of Performance Specification 2 may be used in place of the relative accuracy test.
- (3) For fuel gas combustion devices subject to  $\S60.104(a)(1)$ , either an instrument for continuously monitoring and recording the concentration by volume (dry basis, zero percent excess air) of  $SO_2$  emissions into the atmosphere or monitoring as provided in paragraph (a)(4) of this section). The monitor shall include an oxygen monitor for correcting the data for excess.
- (i) The span values for this monitor are 50 ppm  $SO_2$  and 25 percent oxygen  $(O_2)$ .
- (ii) The  $SO_2$  monitoring level equivalent to the  $H_2S$  standard under

§60.104(a)(1) shall be 20 ppm (dry basis, zero percent excess air).

- (iii) The performance evaluations for this SO<sub>2</sub> monitor under §60.13(c) shall use Performance Specification 2. Methods 6 or 6C and 3 or 3A shall be used for conducting the relative accuracy evaluations. Method 6 samples shall be taken at a flow rate of approximately 2 liters/min for at least 30 minutes. The relative accuracy limit shall be 20 percent or 4 ppm, whichever is greater, and the calibration drift limit shall be 5 percent of the established span value.
- (iv) Fuel gas combustion devices having a common source of fuel gas may be monitored at only one location (i.e., after one of the combustion devices), if monitoring at this location accurately represents the  $SO_2$  emissions into the atmosphere from each of the combustion devices.
- (4) Instead of the  $SO_2$  monitor in paragraph (a)(3) of this section for fuel gas combustion devices subject to  $\S 60.104(a)(1)$ , an instrument for continuously monitoring and recording the concentration (dry basis) of  $H_2S$  in fuel gases before being burned in any fuel gas combustion device.
- (i) The span value for this instrument is 425 mg/dscm  $\rm H_2S.$
- (ii) Fuel gas combustion devices having a common source of fuel gas may be monitored at only one location, if monitoring at this location accurately represents the concentration of  $H_2S$  in the fuel gas being burned.
- (iii) The performance evaluations for this  $H_2S$  monitor under  $\S 60.13(c)$  shall use Performance Specification 7. Method 11, 15, 15A, or 16 shall be used for conducting the relative accuracy evaluations.
- (iv) The owner or operator of a fuel gas combustion device is not required to comply with paragraph (a)(3) or (4) of this section for fuel gas streams that are exempt under §60.104(a)(1) and fuel gas streams combusted in a fuel gas combustion device that are inherently low in sulfur content. Fuel gas streams meeting one of the requirements in paragraphs (a)(4)(iv)(A) through (D) of this section will be considered inherently low in sulfur content. If the composition of a fuel gas stream changes such that it is no longer exempt under §60.104(a)(1) or it no longer meets one

#### § 60.105

- of the requirements in paragraphs (a)(4)(iv)(A) through (D) of this section, the owner or operator must begin continuous monitoring under paragraph (a)(3) or (4) of this section within 15 days of the change.
  - (A) Pilot gas for heaters and flares.
- (B) Fuel gas streams that meet a commercial-grade product specification for sulfur content of 30 ppmv or less. In the case of a liquefied petroleum gas (LPG) product specification in the pressurized liquid state, the gas phase sulfur content should be evaluated assuming complete vaporization of the LPG and sulfur containing-compounds at the product specification concentration.
- (C) Fuel gas streams produced in process units that are intolerant to sulfur contamination, such as fuel gas streams produced in the hydrogen plant, the catalytic reforming unit, the isomerization unit, and HF alkylation process units.
- (D) Other fuel gas streams that an owner or operator demonstrates are low-sulfur according to the procedures in paragraph (b) of this section.
- (5) For Claus sulfur recovery plants with oxidation control systems or reduction control systems followed by incineration subject to \$60.104(a)(2)(i), an instrument for continuously monitoring and recording the concentration (dry basis, zero percent excess air) of SO<sub>2</sub> emissions into the atmosphere. The monitor shall include an oxygen monitor for correcting the data for excess air.
- (i) The span values for this monitor are 500 ppm  $SO_2$  and 25 percent  $O_2$ .
- (ii) The performance evaluations for this  $SO_2$  monitor under §60.13(c) shall use Performance Specification 2. Methods 6 or 6C and 3 or 3A shall be used for conducting the relative accuracy evaluations.
- (6) For Claus sulfur recovery plants with reduction control systems not followed by incineration subject to  $\S 60.104(a)(2)(ii)$ , an instrument for continuously monitoring and recording the concentration of reduced sulfur and  $O_2$  emissions into the atmosphere. The reduced sulfur emissions shall be calculated as  $SO_2$  (dry basis, zero percent excess air).

- (i) The span values for this monitor are 450 ppm reduced sulfur and 25 percent  $O_2$ .
- (ii) The performance evaluations for this reduced sulfur (and  $O_2$ ) monitor under §60.13(c) shall use Performance Specification 5 of appendix B of this part(and Performance Specification 3 of appendix B of this partfor the  $O_2$  analyzer). Methods 15 or 15A and Method 3 shall be used for conducting the relative accuracy evaluations. If Method 3 yields  $O_2$  concentrations below 0.25 percent during the performance specification test, the  $O_2$  concentration may be assumed to be zero and the reduced sulfur CEMS need not include an  $O_2$  monitor.
- (7) In place of the reduced sulfur monitor under paragraph (a)(6) of this section, an instrument using an air or O<sub>2</sub> dilution and oxidation system to convert the reduced sulfur to SO<sub>2</sub> for continuously monitoring and recording the concentration (dry basis, zero percent excess air) of the resultant SO<sub>2</sub>. The monitor shall include an oxygen monitor for correcting the data for excess oxygen.
- (i) The span values for this monitor are  $375 \text{ ppm SO}_2$  and  $25 \text{ percent O}_2$ .
- (ii) For reporting purposes, the  $SO_2$  exceedance level for this monitor is 250 ppm (dry basis, zero percent excess air).
- (iii) The performance evaluations for this  $SO_2$  (and  $O_2$ ) monitor under  $\S60.13(c)$  shall use Performance Specification 5. Methods 15 or 15A and Method 3 shall be used for conducting the relative accuracy evaluations.
- (8) An instrument for continuously monitoring and recording concentrations of SO<sub>2</sub> in the gases at both the inlet and outlet of the SO<sub>2</sub> control device from any fluid catalytic cracking unit catalyst regenerator for which the owner or operator seeks to comply specifically with the 90 percent reduction option under § 60.104(b)(1).
- (i) The span value of the inlet monitor shall be set at 125 percent of the maximum estimated hourly potential SO<sub>2</sub> emission concentration entering the control device, and the span value of the outlet monitor shall be set at 50 percent of the maximum estimated hourly potential SO<sub>2</sub> emission concentration entering the control device.

- (ii) The performance evaluations for these  $SO_2$  monitors under  $\S60.13(c)$  shall use Performance Specification 2. Methods 6 or 6C and 3 or 3A shall be used for conducting the relative accuracy evaluations.
- (9) An instrument for continuously monitoring and recording concentrations of  $SO_2$  in the gases discharged into the atmosphere from any fluid catalytic cracking unit catalyst regenerator for which the owner or operator seeks to comply specifically with the 50 ppmv emission limit under §60.104 (b)(1).
- (i) The span value of the monitor shall be set at 50 percent of the maximum hourly potential  $SO_2$  emission concentration of the control device.
- (ii) The performance evaluations for this  $SO_2$  monitor under §60.13 (c) shall use Performance Specification 2. Methods 6 or 6C and 3 or 3A shall be used for conducting the relative accuracy evaluations.
- (10) An instrument for continuously monitoring and recording concentrations of oxygen  $(O_2)$  in the gases at both the inlet and outlet of the sulfur dioxide control device (or the outlet only if specifically complying with the 50 ppmv standard) from any fluid catalytic cracking unit catalyst regenerator for which the owner or operator has elected to comply with  $\S 60.104(b)(1)$ . The span of this continuous monitoring system shall be set at 10 percent.
- (11) The continuous monitoring systems under paragraphs (a)(8), (a)(9), and (a)(10) of this section are operated and data recorded during all periods of operation of the affected facility including periods of startup, shudown, or malfunction, except for continuous monitoring system breakdowns, repairs, calibration checks, and zero and span adjustments.
- (12) The owner or operator shall use the following procedures to evaluate the continuous monitoring systems under paragraphs (a)(8), (a)(9), and (a)(10) of this section.
- (i) Method 3 or 3A and Method 6 or 6C for the relative accuracy evaluations under the §60.13(e) performance evaluation.

- (ii) appendix F, Procedure 1, including quarterly accuracy determinations and daily calibration drift tests.
- (13) When seeking to comply with §60.104(b)(1), when emission data are not obtained because of continuous monitoring system breakdowns, repairs, calibration checks and zero and span adjustments, emission data will be obtained by using one of the following methods to provide emission data for a minimum of 18 hours per day in at least 22 out of 30 rolling successive calendar days.
- (i) The test methods as described in §60.106(k);
- (ii) A spare continuous monitoring system; or
- (iii) Other monitoring systems as approved by the Administrator.
- (b) An owner or operator may demonstrate that a fuel gas stream combusted in a fuel gas combustion device subject to §60.104(a)(1) that is not specifically exempted in §60.105(a)(4)(iv) is inherently low in sulfur. A fuel gas stream that is determined to be low-sulfur is exempt from the monitoring requirements in paragraphs (a)(3) and (4) of this section until there are changes in operating conditions or stream composition.
- (1) The owner or operator shall submit to the Administrator a written application for an exemption from monitoring. The application must contain the following information:
- (i) A description of the fuel gas stream/system to be considered, including submission of a portion of the appropriate piping diagrams indicating the boundaries of the fuel gas stream/system, and the affected fuel gas combustion device(s) to be considered;
- (ii) A statement that there are no crossover or entry points for sour gas (high H<sub>2</sub>S content) to be introduced into the fuel gas stream/system (this should be shown in the piping diagrams);
- (iii) An explanation of the conditions that ensure low amounts of sulfur in the fuel gas stream (i.e., control equipment or product specifications) at all times;
- (iv) The supporting test results from sampling the requested fuel gas stream/ system demonstrating that the sulfur content is less than 5 ppmv. Sampling

data must include, at minimum, 2 weeks of daily monitoring (14 grab samples) for frequently operated fuel gas streams/systems; for infrequently operated fuel gas streams/systems, seven grab samples must be collected unless other additional information would support reduced sampling. The owner or operator shall use detector tubes ("length-of-stain tube" type measurement) following the "Gas Processors Association Standard 2377-86. Test for Hydrogen Sulfide and Carbon Dioxide in Natural Gas Using Length of Stain Tubes," 1986 Revision (incorporated by reference—see §60.17), with ranges 0-10/0-100 ppm (N = 10/1) to test the applicant fuel gas stream for H2S; and

- (v) A description of how the 2 weeks (or seven samples for infrequently operated fuel gas streams/systems) of monitoring results compares to the typical range of H2S concentration (fuel quality) expected for the fuel gas stream/system going to the affected fuel gas combustion device (e.g., the 2 weeks of daily detector tube results for a frequently operated loading rack included the entire range of products loaded out, and, therefore, should be representative of typical operating conditions affecting H2S content in the fuel gas stream going to the loading rack flare).
- (2) The effective date of the exemption is the date of submission of the information required in paragraph (b)(1) of this section).
- (3) No further action is required unless refinery operating conditions change in such a way that affects the exempt fuel gas stream/system (e.g., the stream composition changes). If such a change occurs, the owner or operator will follow the procedures in paragraph (b)(3)(i), (b)(3)(ii), or (b)(3)(iii) of this section.
- (i) If the operation change results in a sulfur content that is still within the range of concentrations included in the original application, the owner or operator shall conduct an H<sub>2</sub>S test on a grab sample and record the results as proof that the concentration is still within the range.
- (ii) If the operation change results in a sulfur content that is outside the range of concentrations included in the

original application, the owner or operator may submit new information following the procedures of paragraph (b)(1) of this section within 60 days (or within 30 days after the seventh grab sample is tested for infrequently operated process units).

(iii) If the operation change results in a sulfur content that is outside the range of concentrations included in the original application and the owner or operator chooses not to submit new information to support an exemption, the owner or operator must begin H2S monitoring using daily stain sampling to demonstrate compliance. The owner or operator must begin monitoring according to the requirements in paragraphs (a)(1) or (a)(2) of this section as soon as practicable but in no case later than 180 days after the operation change. During daily stain tube sampling, a daily sample exceeding 162 ppmv is an exceedance of the 3-hour H<sub>2</sub>S concentration limit. The owner or operator must determine a rolling 365day average using the stain sampling results; an average H2S concentration of 5 ppmv must be used for days prior to the operation change.

- (c) The average coke burn-off rate (Mg (tons) per hour) and hours of operation shall be recorded daily for any fluid catalytic cracking unit catalyst regenerator subject to §60.102, §60.103, or §60.104(b)(2).
- (d) For any fluid catalytic cracking unit catalyst regenerator under §60.102 that uses an incinerator-waste heat boiler to combust the exhaust gases from the catalyst regenerator, the owner or operator shall record daily the rate of combustion of liquid or solid fossil-fuels and the hours of operation during which liquid or solid fossil-fuels are combusted in the incinerator-waste heat boiler.
- (e) For the purpose of reports under §60.7(c), periods of excess emissions that shall be determined and reported are defined as follows:

NOTE: All averages, except for opacity, shall be determined as the arithmetic average of the applicable 1-hour averages, e.g., the rolling 3-hour average shall be determined as the arithmetic average of three contiguous 1-hour averages.

(1) Opacity. All 1-hour periods that contain two or more 6-minute periods during which the average opacity as

measured by the continuous monitoring system under §60.105(a)(1) exceeds 30 percent.

- (2) Carbon monoxide. All 1-hour periods during which the average CO concentration as measured by the CO continuous monitoring system under §60.105(a)(2) exceeds 500 ppm.
- (3) Sulfur dioxide from fuel gas combustion. (i) All rolling 3-hour periods during which the average concentration of SO<sub>2</sub> as measured by the SO<sub>2</sub> continuous monitoring system under §60.105(a)(3) exceeds 20 ppm (dry basis, zero percent excess air); or
- (ii) All rolling 3-hour periods during which the average concentration of  $H_2S$  as measured by the  $H_2S$  continuous monitoring system under 60.105(a)(4) exceeds 230 mg/dscm (0.10 gr/dscf).
- (4) Sulfur dioxide from Claus sulfur recovery plants. (i) All 12-hour periods during which the average concentration of  $SO_2$  as measured by the  $SO_2$  continuous monitoring system under  $\S 60.105(a)(5)$  exceeds 250 ppm (dry basis, zero percent excess air); or
- (ii) All 12-hour periods during which the average concentration of reduced sulfur (as SO<sub>2</sub>) as measured by the reduced sulfur continuous monitoring system under §60.105(a)(6) exceeds 300 ppm; or
- (iii) All 12-hour periods during which the average concentration of  $SO_2$  as measured by the  $SO_2$  continuous monitoring system under  $\S 60.105(a)(7)$  exceeds 250 ppm (dry basis, zero percent excess air).

[39 FR 9315, Mar. 8, 1974, as amended at 40 FR 46259, Oct. 6, 1975; 42 FR 32427, June 24, 1977; 42 FR 39389, Aug. 4, 1977; 43 FR 10869, Mar. 15, 1978; 48 FR 23611, May 25, 1983; 50 FR 31701, Aug. 5, 1985; 54 FR 34028, Aug. 17, 1989; 55 FR 40175, Oct. 2, 1990; 65 FR 61754, Oct. 17, 2000; 73 FR 35866, June 24, 2008]

#### § 60.106 Test methods and procedures.

- (a) In conducting the performance tests required in §60.8, the owner or operator shall use as reference methods and procedures the test methods in appendix A of this part or other methods and procedures as specified in this section, except as provided in §60.8(b).
- (b) The owner or operator shall determine compliance with the particulate matter (PM) standards in §60.102(a) as follows:

(1) The emission rate (E) of PM shall be computed for each run using the following equation:

$$E = \frac{c_s Q_{sd}}{KR_c}$$

Where:

- E = Emission rate of PM, kg/Mg (lb/ton) of coke burn-off.
- c<sub>s</sub> = Concentration of PM, g/dscm (gr/dscf).
- Q<sub>sd</sub> = Volumetric flow rate of effluent gas, dscm/hr (dscf/hr).
- $R_{\rm c}$  = Coke burn-off rate, Mg/hr (ton/hr) coke. K=Conversion factor, 1,000 g/kg (7,000 gr/lb).
- (2) Method 5B or 5F is to be used to determine particulate matter emissions and associated moisture content from affected facilities without wet FGD systems; only Method 5B is to be used after wet FGD systems. The sampling time for each run shall be at least 60 minutes and the sampling rate shall be at least 0.015 dscm/min (0.53 dscf/min), except that shorter sampling times may be approved by the Administrator when process variables or other factors preclude sampling for at least 60 minutes.
- (3) The coke burn-off rate  $(R_c)$  shall be computed for each run using the following equation:

$$\begin{array}{lll} R_c &= K_1 Q_r \ (\%CO_2 \ + \ \%CO) \ + \ K_2 Q_a - K_3 Q_r \\ (\%CO/2 \ + \ \%CO_2 \ + \ \%O_2) \ + \ K_3 Q_{oxy} \\ (\%O_{oxy}) \end{array}$$

Where:

- $R_{\rm c}$  = Coke burn-off rate, kilograms per hour (kg/hr) (lb/hr).
- Qr = Volumetric flow rate of exhaust gas from fluid catalytic cracking unit regenerator before entering the emission control system, dscm/min (dscf/min).
- Q<sub>a</sub> = Volumetric flow rate of air to fluid catalytic cracking unit regenerator, as determined from the fluid catalytic cracking unit control room instrumentation, dscm/min (dscf/min).
- Q<sub>oxy</sub> = Volumetric flow rate of O<sub>2</sub> enriched air to fluid catalytic cracking unit regenerator, as determined from the fluid catalytic cracking unit control room instrumentation, dscm/min (dscf/min).
- %CO<sub>2</sub> = Carbon dioxide concentration in fluid catalytic cracking unit regenerator exhaust, percent by volume (dry basis).
- %CO = CO concentration in FCCU regenerator exhaust, percent by volume (dry basis).
- $\%O_2 = O_2$  concentration in fluid catalytic cracking unit regenerator exhaust, percent by volume (dry basis).

#### § 60.106

- $%O_{oxy} = O_2$  concentration in  $O_2$  enriched air stream inlet to the fluid catalytic cracking unit regenerator, percent by volume (dry basis).
- K<sub>1</sub> = Material balance and conversion factor, 0.2982 (kg-min)/(hr-dscm-%) [0.0186 (lbmin)/(hr-dscf-%)].
- K<sub>2</sub> = Material balance and conversion factor, 2.088 (kg-min)/(hr-dscm) [0.1303 (lb-min)/ (hr-dscf)].
- $K_3$  = Material balance and conversion factor, 0.0994 (kg-min)/(hr-dscm-%) [0.00624 (lbmin)/(hr-dscf-%)].
- (i) Method 2 shall be used to determine the volumetric flow rate (Qr).
- (ii) The emission correction factor, integrated sampling and analysis procedure of Method 3B shall be used to determine CO2, CO, and O2 concentrations.
- (4) Method 9 and the procedures of §60.11 shall be used to determine opacity.
- (c) If auxiliary liquid or solid fossilfuels are burned in an incineratorwaste heat boiler, the owner or operator shall determine the emission rate of PM permitted in §60.102(b) as fol-
- (1) The allowable emission rate (E<sub>s</sub>) of PM shall be computed for each run using the following equation:

 $E_s = F + A (H/R_c)$ 

Whore.

- $E_s$  = Emission rate of PM allowed, kg/Mg (lb/ ton) of coke burn-off in catalyst regenerator.
- F = Emission standard, 1.0 kg/Mg (2.0 lb/ton) of coke burn-off in catalyst regenerator. A = Allowable incremental rate of PM emissions, 43 g/GJ (0.10 lb/million Btu).
- H = Heat input rate from solid or liquid fossil fuel, GJ/hr (million Btu/hr).

  R<sub>c</sub> = Coke burn-off rate, Mg coke/hr (ton
- coke/hr).
- (2) Procedures subject to the approval of the Administrator shall be used to determine the heat input rate.
- (3) The procedure in paragraph (b)(3) of this section shall be used to determine the coke burn-off rate (R<sub>c</sub>).
- (d) The owner or operator shall determine compliance with the CO standard in §60.103(a) by using the integrated sampling technique of Method 10 to determine the CO concentration (dry basis). The sampling time for each run shall be 60 minutes.
- (e)(1) The owner or operator shall determine compliance with the H2S

- standard in §60.104(a)(1) as follows: Method 11, 15, 15A, or 16 shall be used to determine the H2S concentration. The gases entering the sampling train should be at about atmospheric pressure. If the pressure in the refinery fuel gas lines is relatively high, a flow control valve may be used to reduce the pressure. If the line pressure is high enough to operate the sampling train without a vacuum pump, the pump may be eliminated from the sampling train. The sample shall be drawn from a point near the centroid of the fuel gas line.
- (i) For Method 11, the sampling time and sample volume shall be at least 10 minutes and 0.010 dscm (0.35 dscf). Two samples of equal sampling times shall be taken at about 1-hour intervals. The arithmetic average of these two samples shall constitute a run. For most fuel gases, sampling times exceeding 20 minutes may result in depletion of the collection solution, although fuel gases containing low concentrations of H2S may necessitate sampling for longer periods of time.
- (ii) For Method 15 or 16, at least three injects over a 1-hour period shall constitute a run.
- (iii) For Method 15A, a 1-hour sample shall constitute a run.
- (2) Where emissions are monitored by §60.105(a)(3), compliance §60.104(a)(1) shall be determined using Method 6 or 6C and Method 3 or 3A. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is acceptable alternative to EPA Method 6. A 1-hour sample shall constitute a run. Method 6 samples shall be taken at a rate of approximately 2 liters/min. The ppm correction factor (Method 6) and the sampling location in paragraph (f)(1) of this section apply. Method 4 shall be used to determine the moisture content of the gases. The sampling point for Method 4 shall be adjacent to the sampling point for Method 6 or 6C.
- (f) The owner or operator shall determine compliance with the SO<sub>2</sub> and the H<sub>2</sub>S and reduced sulfur standards in  $\S60.104(a)(2)$  as follows:
- (1) Method 6 shall be used to determine the SO<sub>2</sub> concentration. The concentration in mg/dscm obtained by

Method 6 or 6C is multiplied by 0.3754 to obtain the concentration in ppm. The sampling point in the duct shall be the centroid of the cross section if the cross-sectional area is less than 5.00 m<sup>2</sup> (53.8 ft2) or at a point no closer to the walls than 1.00 m (39.4 in.) if the crosssectional area is 5.00 m<sup>2</sup> or more and the centroid is more than 1 m from the wall. The sampling time and sample volume shall be at least 10 minutes and 0.010 dscm (0.35 dscf) for each sample. Eight samples of equal sampling times shall be taken at about 30-minute intervals. The arithmetic average of these eight samples shall constitute a run. For Method 6C, a run shall consist of the arithmetic average of four 1hour samples. Method 4 shall be used to determine the moisture content of the gases. The sampling point for Method 4 shall be adjacent to the sampling point for Method 6 or 6C. The sampling time for each sample shall be equal to the time it takes for two Method 6 samples. The moisture content from this sample shall be used to correct the corresponding Method 6 samples for moisture. For documenting the oxidation efficiency of the control device for reduced sulfur compounds, Method 15 shall be used following the procedures of paragraph (f)(2) of this section.

(2) Method 15 shall be used to determine the reduced sulfur and H2 S concentrations. Each run shall consist of 16 samples taken over a minimum of 3 hours. The sampling point shall be the same as that described for Method 6 in paragraph (f)(1) of this section. To ensure minimum residence time for the sample inside the sample lines, the sampling rate shall be at least 3.0 lpm (0.10 cfm). The SO<sub>2</sub> equivalent for each run shall be calculated after being corrected for moisture and oxygen as the arithmetic average of the SO2 equivalent for each sample during the run. Method 4 shall be used to determine the moisture content of the gases as the paragraph (f)(1) of this section. The sampling time for each sample shall be equal to the time it takes for four Method 15 samples.

(3) The oxygen concentration used to correct the emission rate for excess air shall be obtained by the integrated sampling and analysis procedure of Method 3 or 3A. The samples shall be

taken simultaneously with the SO2, reduced sulfur and H2S, or moisture samples. The SO<sub>2</sub>, reduced sulfur, and H<sub>2</sub>S samples shall be corrected to zero percent excess air using the equation in paragraph (h)(6) of this section.

(g) Each performance test conducted for the purpose of determining compliance under §60.104(b) shall consist of all testing performed over a 7-day period using Method 6 or 6C and Method 3 or To determine compliance, the arithmetic mean of the results of all the tests shall be compared with the applicable standard.

(h) For the purpose of determining compliance with §60.104(b)(1), the following calculation procedures shall be

used:

- (1) Calculate each 1-hour average concentration (dry, zero percent oxygen, ppmv) of sulfur dioxide at both the inlet and the outlet to the add-on control device as specified in §60.13(h). These calculations are made using the collected emission data §60.105(a).
- (2) Calculate a 7-day average (arithmetic mean) concentration of sulfur dioxide for the inlet and for the outlet to the add-on control device using all of the 1-hour average concentration values obtained during seven successive 24-hour periods.
- (3) Calculate the 7-day average percent reduction using the following equation:

 $Rso_2 = 100(Cso_2(i) - Cso_2(o))/Cso_2(i)$ 

where:

Rso<sub>2</sub> = 7-day average sulfur dioxide emission reduction, percent

Cso<sub>2</sub>(i) = sulfur dioxide emission concentration determined in §60.106(h)(2) at the inlet to the add-on control device, ppmv Cso<sub>2</sub>(o) = sulfur dioxide emission concentration determined in §60.106(h)(2) at the

outlet to the add-on control device, ppmv 100 = conversion factor, decimal to percent

(4) Outlet concentrations of sulfur dioxide from the add-on control device for compliance with the 50 ppmv standard, reported on a dry, O2-free basis, shall be calculated using the procedures outlined in §60.106(h)(1) and (2) above, but for the outlet monitor only.

(5) If supplemental sampling data are used for determining the 7-day averages under paragraph (h) of this section and such data are not hourly averages,

#### § 60.106

then the value obtained for each supplemental sample shall be assumed to represent the hourly average for each hour over which the sample was obtained.

(6) For the purpose of adjusting pollutant concentrations to zero percent oxygen, the following equation shall be used:

$$C_{adj} = C_{meas} [20.9 \text{c}/(20.9 - \% O_2)]$$

where:

 $C_{adj}$  = pollutant concentration adjusted to zero percent oxygen, ppm or g/dscm

C<sub>meas</sub> = pollutant concentration measured on a dry basis, ppm or g/dscm

20.9c = 20.9 percent oxygen - 0.0 percent oxygen (defined oxygen correction basis), percent

20.9 = oxygen concentration in air, percent  $^{9}\text{O}_2 = \text{oxygen}$  concentration measured on a dry basis, percent

- (i) For the purpose of determining compliance with §60.104(b)(2), the following reference methods and calculation procedures shall be used except as provided in paragraph (i)(12) of this section:
- (1) One 3-hour test shall be performed each day.
- (2) For gases released to the atmosphere from the fluid catalytic cracking unit catalyst regenerator:
- (i) Method 8 as modified in §60.106(i)(3) for moisture content and for the concentration of sulfur oxides calculated as sulfur dioxide,
- (ii) Method 1 for sample and velocity traverses,
- (iii) Method 2 calculation procedures (data obtained from Methods 3 and 8) for velocity and volumetric flow rate, and
  - (iv) Method 3 for gas analysis.
- (3) Method 8 shall be modified by the insertion of a heated glass fiber filter between the probe and first impinger. The probe liner and glass fiber filter temperature shall be maintained above 160 °C (320 °F). The isopropanol impinger shall be eliminated. Sample recovery procedures described in Method 8 for container No. 1 shall be eliminated. The heated glass fiber filter also shall be excluded; however, rinsing of all connecting glassware after the heated glass fiber filter shall be retained and included in container No. 2. Sampled volume shall be at least 1 dscm.

- (4) For Method 3, the integrated sampling technique shall be used.
- (5) Sampling time for each run shall be at least 3 hours.
- (6) All testing shall be performed at the same location. Where the gases discharged by the fluid catalytic cracking unit catalyst regenerator pass through an incinerator-waste heat boiler in which auxiliary or supplemental gaseous, liquid, or solid fossil fuel is burned, testing shall be conducted at a point between the regenerator outlet and the incinerator-waste heat boiler. An alternative sampling location after the waste heat boiler may be used if alternative coke burn-off rate equations, and, if requested, auxiliary/supplemental fuel SOx credits, have been submitted to and approved by the Administrator prior to sampling.
- (7) Coke burn-off rate shall be determined using the procedures specified under paragraph (b)(3) of this section, unless paragraph (i)(6) of this section applies.
- (8) Calculate the concentration of sulfur oxides as sulfur dioxide using equation 8-3 in Section 6.5 of Method 8 to calculate and report the total concentration of sulfur oxides as sulfur dioxide (Cso $_{\rm x}$ ).
- (9) Sulfur oxides emission rate calculated as sulfur dioxide shall be determined for each test run by the following equation:

$$E_{so_x} = C_{so_x} Q_{sd}/K$$

Where

E<sub>SOx</sub> = sulfur oxides emission rate calculated as sulfur dioxide, kg/hr (lb/hr)

C<sub>Sox</sub> = sulfur oxides emission concentration calculated as sulfur dioxide, g/dscm (gr/dscf)

Q<sub>sd</sub> = dry volumetric stack gas flow rate corrected to standard conditions, dscm/hr (dscf/hr)

K=1,000 g/kg (7,000 gr/lb)

(10) Sulfur oxides emissions calculated as sulfur dioxide shall be determined for each test run by the following equation:

$$R_{so_x} = (E_{so_x}/R_c)$$

Where:

R<sub>SOx</sub> = Sulfur oxides emissions calculated as kg sulfur dioxide per Mg (lb/ton) coke burn-off  $E_{SOx}$  = Sulfur oxides emission rate calculated as sulfur dioxide, kg/hr (lb/hr).

- $R_c$  = Coke burn-off rate, Mg/hr (ton/hr).
- (11) Calculate the 7-day average sulfur oxides emission rate as sulfur dioxide per Mg (ton) of coke burn-off by dividing the sum of the individual daily rates by the number of daily rates summed.
- (12) An owner or operator may, upon approval by the Administrator, use an alternative method for determining compliance with §60.104(b)(2), as provided in §60.8(b). Any requests for approval must include data to demonstrate to the Administrator that the alternative method would produce results adequate for the determination of compliance.
- (j) For the purpose of determining compliance with §60.104(b)(3), the following analytical methods and calculation procedures shall be used:
- (1) One fresh feed sample shall be collected once per 8-hour period.
- (2) Fresh feed samples shall be analyzed separately by using any one of the following applicable analytical test methods: ASTM D129-64, 78, or 95, ASTM D1552-83 or 95, ASTM D2622-87, 94, or 98, or ASTM D1266-87, 91, or 98. (These methods are incorporated by reference: see §60.17.) The applicable range of some of these ASTM methods is not adequate to measure the levels of sulfur in some fresh feed samples. Dilution of samples prior to analysis with verification of the dilution ratio is acceptable upon prior approval of the Administrator.
- (3) If a fresh feed sample cannot be collected at a single location, then the fresh feed sulfur content shall be determined as follows:
- (i) Individual samples shall be collected once per 8-hour period for each separate fresh feed stream charged directly into the riser or reactor of the fluid catalytic cracking unit. For each sample location the fresh feed volumetric flow rate at the time of collecting the fresh feed sample shall be measured and recorded. The same method for measuring volumetric flow rate shall be used at all locations.
- (ii) Each fresh feed sample shall be analyzed separately using the methods specified under paragraph (j)(2) of this section.

(iii) Fresh feed sulfur content shall be calculated for each 8-hour period using the following equation:

$$S_f = \sum_{i=1}^n \frac{S_i Q_i}{Q_f}$$

where:

- $\mathbf{S}_{\text{f}}$  = fresh feed sulfur content expressed in percent by weight of fresh feed.
- n = number of separate fresh feed streams charged directly to the riser or reactor of the fluid catalytic cracking unit.
- Q<sub>f</sub> = total volumetric flow rate of fresh feed charged to the fluid catalytic cracking unit.
- S<sub>i</sub> = fresh feed sulfur content expressed in percent by weight of fresh feed for the "ith" sampling location.
- Q<sub>i</sub> = volumetric flow rate of fresh feed stream for the "ith" sampling location.
- (4) Calculate a 7-day average (arithmetic mean) sulfur content of the fresh feed using all of the fresh feed sulfur content values obtained during seven successive 24-hour periods.
- (k) The test methods used to supplement continuous monitoring system data to meet the minimum data requirements in §60.104(d) will be used as described below or as otherwise approved by the Administrator.
- (1) Methods 6, 6B, or 8 are used. The sampling location(s) are the same as those specified for the monitor.
- (2) For Method 6, the minimum sampling time is 20 minutes and the minimum sampling volume is 0.02 dscm (0.71 dscf) for each sample. Samples are taken at approximately 60-minute intervals. Each sample represents a 1-hour average. A minimum of 18 valid samples is required to obtain one valid day of data.
- (3) For Method 6B, collection of a sample representing a minimum of 18 hours is required to obtain one valid day of data.
- (4) For Method 8, the procedures as outlined in this section are used. The equivalent of 16 hours of sampling is

#### § 60.107

required to obtain one valid day of data.

[39 FR 9315, Mar. 8, 1974, as amended at 43 FR 10869, Mar. 15, 1978; 51 FR 42842, Nov. 26, 1986; 52 FR 20392, June 1, 1987; 53 FR 41333, Oct. 21, 1988; 54 FR 34028, Aug. 17, 1989; 55 FR 40176, Oct. 2, 1990; 56 FR 4176, Feb. 4, 1991; 65 FR 61754, Oct. 17, 2000; 71 FR 55127, Sept. 21, 2006; 73 FR 35867, June 24, 2008; 77 FR 56463, Sep. 12, 20121

## § 60.107 Reporting and recordkeeping requirements.

- (a) Each owner or operator subject to \$60.104(b) shall notify the Administrator of the specific provisions of \$60.104(b) with which the owner or operator seeks to comply. Notification shall be submitted with the notification of initial startup required by \$60.7(a)(3). If an owner or operator elects at a later date to comply with an alternative provision of \$60.104(b), then the Administrator shall be notified by the owner or operator in the report described in paragraph (c) of this section.
- (b) Each owner or operator subject to §60.104(b) shall record and maintain the following information:
  - (1) If subject to §60.104(b)(1),
- (i) All data and calibrations from continuous monitoring systems located at the inlet and outlet to the control device, including the results of the daily drift tests and quarterly accuracy assessments required under appendix F, Procedure 1;
- (ii) Measurements obtained by supplemental sampling (refer to §60.105(a)(13) and §60.106(k)) for meeting minimum data requirements; and
- (iii) The written procedures for the quality control program required by appendix F, Procedure 1.
- (2) If subject to §60.104(b)(2), measurements obtained in the daily Method 8 testing, or those obtained by alternative measurement methods, if §60.106(i)(12) applies.
- (3) If subject to \$60.104(b)(3), data obtained from the daily feed sulfur tests.
- (4) Each 7-day rolling average compliance determination.
- (c) Each owner or operator subject to §60.104(b) shall submit a report except as provided by paragraph (d) of this section. The following information shall be contained in the report:
  - (1) Any 7-day period during which:

- (i) The average percent reduction and average concentration of sulfur dioxide on a dry, O2-free basis in the gases discharged to the atmosphere from any fluid cracking unit catalyst regenerator for which the owner or operator seeks to comply with §60.104(b)(1) is below 90 percent and above 50 ppmv, as measured by the continuous monisystem prescribed under §60.105(a)(8), or above 50 ppmv, as measured by the outlet continuous moniprescribed under toring system §60.105(a)(9). The average percent reduction and average sulfur dioxide concentration shall be determined using specified the procedures under §60.106(h);
- (ii) The average emission rate of sulfur dioxide in the gases discharged to the atmosphere from any fluid catalytic cracking unit catalyst regenerator for which the owner or operator seeks to comply with \$60.104(b)(2)\$ exceeds 9.8 kg  $SO_x$  per 1,000 kg coke burnoff, as measured by the daily testing prescribed under \$60.106(i)\$. The average emission rate shall be determined using the procedures specified under \$60.106(i)\$; and
- (iii) The average sulfur content of the fresh feed for which the owner or operator seeks to comply with §60.104(b)(3) exceeds 0.30 percent by weight. The fresh feed sulfur content, a 7-day rolling average, shall be determined using the procedures specified under §60.106(i).
- (2) Any 30-day period in which the minimum data requirements specified in §60.104(d) are not obtained.
- (3) For each 7-day period during which an exceedance has occurred as defined in paragraphs (c)(1)(i) through (c)(1)(iii) and (c)(2) of this section:
- (i) The date that the exceedance occurred;
- (ii) An explanation of the exceedance;
- (iii) Whether the exceedance was concurrent with a startup, shutdown, or malfunction of the fluid catalytic cracking unit or control system; and
- (iv) A description of the corrective action taken, if any.
  - (4) If subject to  $\S 60.104(b)(1)$ ,
- (i) The dates for which and brief explanations as to why fewer than 18 valid hours of data were obtained for

the inlet continuous monitoring system:

- (ii) The dates for which and brief explanations as to why fewer than 18 valid hours of data were obtained for the outlet continuous monitoring system:
- (iii) Identification of times when hourly averages have been obtained based on manual sampling methods;
- (iv) Identification of the times when the pollutant concentration exceeded full span of the continuous monitoring system; and
- (v) Description of any modifications to the continuous monitoring system that could affect the ability of the continuous monitoring system to comply with Performance Specifications 2 or 3.
- (vi) Results of daily drift tests and quarterly accuracy assessments as required under appendix F, Procedure 1.
- (5) If subject to §60.104(b)(2), for each day in which a Method 8 sample result required by §60.106(i) was not obtained, the date for which and brief explanation as to why a Method 8 sample result was not obtained, for approval by the Administrator.
- (6) If subject to §60.104(b)(3), for each 8-hour period in which a feed sulfur measurement required by §60.106(j) was not obtained, the date for which and brief explanation as to why a feed sulfur measurement was not obtained, for approval by the Administrator.
- (d) For any periods for which sulfur dioxide or oxides emissions data are not available, the owner or operator of the affected facility shall submit a signed statement indicating if any changes were made in operation of the emission control system during the period of data unavailability which could affect the ability of the system to meet the applicable emission limit. Operations of the control system and affected facility during periods of data unavailability are to be compared with operation of the control system and affected facility before and following the period of data unavailability.
- (e) For each fuel gas stream combusted in a fuel gas combustion device subject to §60.104(a)(1), if an owner or operator determines that one of the exemptions listed in §60.105(a)(4)(iv) applies to that fuel gas stream, the owner or operator shall maintain records of

the specific exemption chosen for each fuel gas stream. If the owner or operator applies for the exemption described in §60.105(a)(4)(iv)(D), the owner or operator must keep a copy of the application as well as the letter from the Administrator granting approval of the application.

- (f) The owner or operator of an affected facility shall submit the reports required under this subpart to the Administrator semiannually for each sixmonth period. All semiannual reports shall be postmarked by the 30th day following the end of each sixmonth period.
- (g) The owner or operator of the affected facility shall submit a signed statement certifying the accuracy and completeness of the information contained in the report.

[54 FR 34029, Aug. 17, 1989, as amended at 55 FR 40178, Oct. 2, 1990; 64 FR 7465, Feb. 12, 1999; 65 FR 61755, Oct. 17, 2000; 73 FR 35867, June 24, 2008]

## § 60.108 Performance test and compliance provisions.

- (a) Section 60.8(d) shall apply to the initial performance test specified under paragraph (c) of this section, but not to the daily performance tests required thereafter as specified in §60.108(d). Section 60.8(f) does not apply when determining compliance with the standards specified under §60.104(b). Performance tests conducted for the purpose of determining compliance under §60.104(b) shall be conducted according to the applicable procedures specified under §60.106.
- (b) Owners or operators who seek to comply with §60.104(b)(3) shall meet that standard at all times, including periods of startup, shutdown, and malfunctions.
- (c) The initial performance test shall consist of the initial 7-day average calculated for compliance with §60.104(b)(1), (b)(2), or (b)(3).
- (d) After conducting the initial performance test prescribed under §60.8, the owner or operator of a fluid catalytic cracking unit catalyst regenerator subject to §60.104(b) shall conduct a performance test for each successive 24-hour period thereafter. The daily performance tests shall be conducted according to the appropriate

#### § 60.109

procedures specified under §60.106. In the event that a sample collected under §60.106(i) or (j) is accidentally lost or conditions occur in which one of the samples must be discontinued because of forced shutdown, failure of an irreplaceable portion of the sample train, extreme meteorological conditions, or other circumstances, beyond the owner or operators' control, compliance may be determined using available data for the 7-day period.

(e) Each owner or operator subject to §60.104(b) who has demonstrated compliance with one of the provisions of §60.104(b) but a later date seeks to comply with another of the provisions of §60.104(b) shall begin conducting daily performance tests as specified under paragraph (d) of this section immediately upon electing to become subject to one of the other provisions of §60.104(b). The owner or operator shall furnish the Administrator with a written notification of the change in the semiannual report required §60.107(f).

[54 FR 34030, Aug. 17, 1989, as amended at 55 FR 40178, Oct. 2, 1990; 64 FR 7466, Feb. 12, 1999; 73 FR 35867, June 24, 2008]

#### § 60.109 Delegation of authority.

- (a) In delegating implementation and enforcement authority to a State under section 111(c) of the Act, the authorities contained in paragraph (b) of this section shall be retained by the Administrator and not transferred to a State
- (b) Authorities which shall not be delegated to States:
  - (1) Section 60.105(a)(13)(iii),
  - (2) Section 60.106(i)(12).

[54 FR 34031, Aug. 17, 1989, as amended at 55 FR 40178, Oct. 2, 1990]

# Subpart Ja—Standards of Performance for Petroleum Refineries for Which Construction, Reconstruction, or Modification Commenced After May 14, 2007

SOURCE: 73 FR 35867, June 24, 2008, unless otherwise noted.

## § 60.100a Applicability, designation of affected facility, and reconstruction

- (a) The provisions of this subpart apply to the following affected facilities in petroleum refineries: fluid catalytic cracking units (FCCU), fluid coking units (FCU), delayed coking units, fuel gas combustion devices (including process heaters), flares and sulfur recovery plants. The sulfur recovery plant need not be physically located within the boundaries of a petroleum refinery to be an affected facility, provided it processes gases produced within a petroleum refinery.
- (b) Except for flares and delayed coking units, the provisions of this subpart apply only to affected facilities under paragraph (a) of this section which commence construction, modification or reconstruction after May 14, 2007. For flares, the provisions of this subpart apply only to flares which commence construction, modification or reconstruction after June 24, 2008. For the purposes of this subpart, a modification to a flare commences when a project that includes any of the activities in paragraphs (c)(1) or (2) of this section is commenced. For delayed coking units, the provisions of this subpart apply to delayed coking units that commence construction, reconstruction or modification on the earliest of the following dates:
- (1) May 14, 2007, for such activities that involve a "delayed coking unit" defined as follows: one or more refinery process units in which high molecular weight petroleum derivatives are thermally cracked and petroleum coke is produced in a series of closed, batch system reactors;
- (2) December 22, 2008, for such activities that involve a "delayed coking unit" defined as follows: a refinery process unit in which high molecular weight petroleum derivatives are thermally cracked and petroleum coke is produced in a series of closed, batch system reactors. A delayed coking unit consists of the coke drums and associated fractionator;
- (3) September 12, 2012, for such activities that involve a "delayed coking unit" as defined in \$60.101a.

Appendix B
40 CFR Part 60, Subpart Ja
Standards of Performance for Petroleum Refineries for Which Construction, Reconstruction, or
Modification Commenced After May 14, 2007



#### § 60, 109

procedures specified under §60.106. In the event that a sample collected under §60.106(i) or (j) is accidentally lost or conditions occur in which one of the samples must be discontinued because of forced shutdown, failure of an irreplaceable portion of the sample train, extreme meteorological conditions, or other circumstances, beyond the owner or operators' control, compliance may be determined using available data for the 7-day period.

(e) Each owner or operator subject to §60.104(b) who has demonstrated compliance with one of the provisions of §60.104(b) but a later date seeks to comply with another of the provisions of §60.104(b) shall begin conducting daily performance tests as specified under paragraph (d) of this section immediately upon electing to become subject to one of the other provisions of §60.104(b). The owner or operator shall furnish the Administrator with a written notification of the change in the semiannual report required §60.107(f).

[54 FR 34030, Aug. 17, 1989, as amended at 55 FR 40178, Oct. 2, 1990; 64 FR 7466, Feb. 12, 1999; 73 FR 35867, June 24, 2008]

#### § 60.109 Delegation of authority.

- (a) In delegating implementation and enforcement authority to a State under section 111(c) of the Act, the authorities contained in paragraph (b) of this section shall be retained by the Administrator and not transferred to a State.
- (b) Authorities which shall not be delegated to States:
- (1) Section 60.105(a)(13)(iii),
- (2) Section 60.106(i)(12).

[54 FR 34031, Aug. 17, 1989, as amended at 55 FR 40178, Oct. 2, 1990]

#### Subpart Ja—Standards of Performance for Petroleum Refineries for Which Construction, Reconstruction, or Modification Commenced After May 14, 2007

SOURCE: 73 FR 35867, June 24, 2008, unless otherwise noted.

# § 60.100a Applicability, designation of affected facility, and reconstruction.

- (a) The provisions of this subpart apply to the following affected facilities in petroleum refineries: fluid catalytic cracking units (FCCU), fluid coking units (FCU), delayed coking units, fuel gas combustion devices (including process heaters), flares and sulfur recovery plants. The sulfur recovery plants are not be physically located within the boundaries of a petroleum refinery to be an affected facility, provided it processes gases produced within a petroleum refinery.
- (b) Except for flares and delayed coking units, the provisions of this subpart apply only to affected facilities under paragraph (a) of this section which commence construction, modification or reconstruction after May 14, 2007. For flares, the provisions of this subpart apply only to flares which commence construction, modification or reconstruction after June 24, 2008. For the purposes of this subpart, a modification to a flare commences when a project that includes any of the activities in paragraphs (c)(1) or (2) of this section is commenced. For delayed coking units, the provisions of this subpart apply to delayed coking units that commence construction, reconstruction or modification on the earliest of the following dates:
- (1) May 14, 2007, for such activities that involve a "delayed coking unit" defined as follows: one or more refinery process units in which high molecular weight petroleum derivatives are thermally cracked and petroleum coke is produced in a series of closed, batch system reactors:
- (2) December 22, 2008, for such activities that involve a "delayed coking unit" defined as follows: a refinery process unit in which high molecular weight petroleum derivatives are thermally cracked and petroleum coke is produced in a series of closed, batch system reactors. A delayed coking unit consists of the coke drums and associated fractionator;
- (3) September 12, 2012, for such activities that involve a "delayed coking unit" as defined in §60.101a.

#### **Environmental Protection Agency**

- (c) For all affected facilities other than flares, the provisions in §60.14 regarding modification apply. As provided in §60.14(f), the special provisions set forth under this subpart shall supersede the provisions in §60.14 with respect to flares. For the purposes of this subpart, a modification to a flare occurs as provided in paragraphs (c)(1) or (2) of this section.
- (1) Any new piping from a refinery process unit, including ancillary equipment, or a fuel gas system is physically connected to the flare (e.g., for direct emergency relief or some form of continuous or intermittent venting). However, the connections described in paragraphs (c)(1)(i) through (vii) of this section are not considered modifications of a flare.
- (i) Connections made to install monitoring systems to the flare.
- (ii) Connections made to install a flare gas recovery system or connections made to upgrade or enhance components of a flare gas recovery system (e.g., addition of compressors or recycle lines).
- (iii) Connections made to replace or upgrade existing pressure relief or safety valves, provided the new pressure relief or safety valve has a set point opening pressure no lower and an internal diameter no greater than the existing equipment being replaced or upgraded.
- (iv) Connections made for flare gas sulfur removal.
- (v) Connections made to install backup (redundant) equipment associated with the flare (such as a back-up compressor) that does not increase the capacity of the flare.
- (vi) Replacing piping or moving an existing connection from a refinery process unit to a new location in the same flare, provided the new pipe diameter is less than or equal to the diameter of the pipe/connection being replaced/moved.
- (vii) Connections that interconnect two or more flares.
- (2) A flare is physically altered to increase the flow capacity of the flare.
- (d) For purposes of this subpart, under §60.15, the "fixed capital cost of the new components" includes the fixed capital cost of all depreciable components which are or will be re-

placed pursuant to all continuous programs of component replacement which are commenced within any 2-year period following the relevant applicability date specified in paragraph (b) of this section.

[73 FR 35867, June 24, 2008, as amended at 77 FR 56464, Sep. 12, 2012]

#### § 60.101a Definitions.

Terms used in this subpart are defined in the Clean Air Act (CAA), in §60.2 and in this section.

Air preheat means a device used to heat the air supplied to a process heater generally by use of a heat exchanger to recover the sensible heat of exhaust gas from the process heater.

Ancillary equipment means equipment used in conjunction with or that serve a refinery process unit. Ancillary equipment includes, but is not limited to, storage tanks, product loading operations, wastewater treatment systems, steam- or electricity-producing units (including coke gasification units), pressure relief valves, pumps, sampling vents and continuous analyzer vents.

Cascaded flare system means a series of flares connected to one flare gas header system arranged with increasing pressure set points so that discharges will be initially directed to the first flare in the series (i.e., the primary flare). If the discharge pressure exceeds a set point at which the flow to the primary flare would exceed the primary flare's capacity, flow will be diverted to the second flare in the series. Similarly, flow would be diverted to a third (or fourth) flare if the pressure in the flare gas header system exceeds a threshold where the flow to the first two (or three) flares would exceed their capacities.

Co-fired process heater means a process heater that employs burners that are designed to be supplied by both gaseous and liquid fuels on a routine basis. Process heaters that have gas burners with emergency oil back-up burners are not considered co-fired process heaters.

Coke burn-off means the coke removed from the surface of the FCCU catalyst by combustion in the catalyst regenerator. The rate of coke burn-off is calculated by the formula specified in §60.104a.

§60.101a

Contact material means any substance formulated to remove metals, sulfur, nitrogen, or any other contaminant from petroleum derivatives.

Delayed coking unit means one or more refinery process units in which high molecular weight petroleum derivatives are thermally cracked and petroleum coke is produced in a series of closed, batch system reactors.

Corrective action means the design, operation and maintenance changes that one takes consistent with good engineering practice to reduce or eliminate the likelihood of the recurrence of the primary cause and any other contributing cause(s) of an event identified by a root cause analysis as having resulted in a discharge of gases to an affected flare in excess of specified thresholds.

Corrective action analysis means a description of all reasonable interim and long-term measures, if any, that are available, and an explanation of why the selected corrective action(s) is/are the best alternative(s), including, but not limited to, considerations of cost effectiveness, technical feasibility, safety and secondary impacts.

Delayed coking unit means a refinery process unit in which high molecular weight petroleum derivatives are thermally cracked and petroleum coke is produced in a series of closed, batch system reactors. A delayed coking unit includes, but is not limited to, all of the coke drums associated with a single fractionator; the fractionator, including the bottoms receiver and the overhead condenser; the coke drum cutting water and quench system, including the jet pump and coker quench water tank; process piping and associated equipment such as pumps, valves and connectors; and the coke drum blowdown recovery compressor system.

Emergency flare means a flare that combusts gas exclusively released as a result of malfunctions (and not startup, shutdown, routine operations or any other cause) on four or fewer occasions in a rolling 365-day period. For purposes of this rule, a flare cannot be categorized as an emergency flare unless it maintains a water seal.

Flare means a combustion device that uses an uncontrolled volume of air to burn gases. The flare includes the foun-

dation, flare tip, structural support, burner, igniter, flare controls, including air injection or steam injection systems, flame arrestors and the flare gas header system. In the case of an interconnected flare gas header system, the flare includes each individual flare serviced by the interconnected flare gas header system and the interconnected flare gas header system.

Flare gas header system means all piping and knockout pots, including those in a subheader system, used to collect and transport gas to a flare either from a process unit or a pressure relief valve from the fuel gas system, regardless of whether or not a flare gas recovery system draws gas from the flare gas header system. The flare gas header system includes piping inside the battery limit of a process unit if the purpose of the piping is to transport gas to a flare or knockout pot that is part of the flare.

Flare gas recovery system means a system of one or more compressors, piping and the associated water seal, rupture disk or similar device used to divert gas from the flare and direct the gas to the fuel gas system or to a fuel gas combustion device.

Flexicoking unit means a refinery process unit in which high molecular weight petroleum derivatives are thermally cracked and petroleum coke is continuously produced and then gasified to produce a synthetic fuel gas

Fluid catalytic cracking unit means a refinery process unit in which petroleum derivatives are continuously charged and hydrocarbon molecules in the presence of a catalyst suspended in a fluidized bed are fractured into smaller molecules, or react with a contact material suspended in a fluidized bed to improve feedstock quality for additional processing and the catalyst or contact material is continuously regenerated by burning off coke and other deposits. The unit includes the riser, reactor, regenerator, air blowers, spent catalyst or contact material stripper, catalyst or contact material recovery equipment, and regenerator equipment for controlling air pollutant emissions and for heat recovery. When fluid catalyst cracking unit regenerator exhaust from two separate fluid catalytic cracking units share a common exhaust treatment (e.g., CO boiler or wet scrubber), the *fluid catalytic cracking unit* is a single affected facility.

Fluid coking unit means a refinery process unit in which high molecular weight petroleum derivatives are thermally cracked and petroleum coke is continuously produced in a fluidized bed system. The fluid coking unit includes the coking reactor, the coking burner, and equipment for controlling air pollutant emissions and for heat recovery on the fluid coking burner exhaust vent.

Forced draft process heater means a process heater in which the combustion air is supplied under positive pressure produced by a fan at any location in the inlet air line prior to the point where the combustion air enters the process heater or air preheat. For the purposes of this subpart, a process heater that uses fans at both the inlet air side and the exhaust air side (i.e., balanced draft system) is considered to be a forced draft process heater.

Fuel gas means any gas which is generated at a petroleum refinery and which is combusted. Fuel gas includes natural gas when the natural gas is combined and combusted in any proportion with a gas generated at a refinery. Fuel gas does not include gases generated by catalytic cracking unit catalyst regenerators, coke calciners (used to make premium grade coke) and fluid coking burners, but does include gases from flexicoking unit gasifiers and other gasifiers. Fuel gas does not include vapors that are collected and combusted in a thermal oxidizer or flare installed to control emissions from wastewater treatment units other than those processing sour water, marine tank vessel loading operations or asphalt processing units (i.e., asphalt blowing stills).

Fuel gas combustion device means any equipment, such as process heaters and boilers, used to combust fuel gas. For the purposes of this subpart, fuel gas combustion device does not include flares or facilities in which gases are combusted to produce sulfur or sulfuric acid.

Fuel gas system means a system of compressors, piping, knock-out pots, mix drums, and units used to remove sulfur contaminants from the fuel gas

(e.g., amine scrubbers) that collects refinery fuel gas from one or more sources for treatment as necessary prior to combusting in process heaters or boilers. A fuel gas system may have an overpressure vent to a flare but the primary purpose for a fuel gas system is to provide fuel to the refinery.

Natural draft process heater means any process heater in which the combustion air is supplied under ambient or negative pressure without the use of an inlet air (forced draft) fan. For the purposes of this subpart, a natural draft process heater is any process heater that is not a forced draft process heater, including induced draft systems.

Non-emergency flare means any flare that is not an emergency flare as defined in this subpart.

Oxidation control system means an emission control system which reduces emissions from sulfur recovery plants by converting these emissions to sulfur dioxide ( $SO_2$ ) and recycling the  $SO_2$  to the reactor furnace or the first-stage catalytic reactor of the Claus sulfur recovery plant or converting the  $SO_2$  to a sulfur product.

Petroleum means the crude oil removed from the earth and the oils derived from tar sands, shale, and coal.

Petroleum refinery means any facility engaged in producing gasoline, kerosene, distillate fuel oils, residual fuel oils, lubricants, asphalt (bitumen) or other products through distillation of petroleum or through redistillation, cracking or reforming of unfinished petroleum derivatives. A facility that produces only oil shale or tar sands-derived crude oil for further processing at a petroleum refinery using only solvent extraction and/or distillation to recover diluent is not a petroleum refinery.

Primary flare means the first flare in a cascaded flare system.

Process heater means an enclosed combustion device used to transfer heat indirectly to process stream materials (liquids, gases, or solids) or to a heat transfer material for use in a process unit instead of steam.

Process upset gas means any gas generated by a petroleum refinery process unit or by ancillary equipment as a result of startup, shutdown, upset or malfunction.

#### §60.102a

Purge gas means gas introduced between a flare's water seal and a flare's tip to prevent oxygen infiltration (backflow) into the flare tip. For flares with no water seals, the function of purge gas is performed by sweep gas (i.e., flares without water seals do not use purge gas).

Reduced sulfur compounds means hydrogen sulfide ( $H_2S$ ), carbonyl sulfide, and carbon disulfide.

Reduction control system means an emission control system which reduces emissions from sulfur recovery plants by converting these emissions to  $\rm H_2S$  and either recycling the  $\rm H_2S$  to the reactor furnace or the first-stage catalytic reactor of the Claus sulfur recovery plant or converting the  $\rm H_2S$  to a sulfur product.

Refinery process unit means any segment of the petroleum refinery in which a specific processing operation is conducted.

Root cause analysis means an assessment conducted through a process of investigation to determine the primary cause, and any other contributing cause(s), of a discharge of gases in excess of specified thresholds.

Secondary flare means a flare in a cascaded flare system that provides additional flare capacity and pressure relief to a flare gas system when the flare gas flow exceeds the capacity of the primary flare. For purposes of this subpart, a secondary flare is characterized by infrequent use and must maintain a water seal.

Sulfur pit means the storage vessel in which sulfur that is condensed after each Claus catalytic reactor is initially accumulated and stored. A sulfur pit does not include secondary sulfur storage vessels downstream of the initial Claus reactor sulfur pits.

Sulfur recovery plant means all process units which recover sulfur from  $\rm H_2S$  and/or  $\rm SO_2$  from a common source of sour gas produced at a petroleum refinery. The sulfur recovery plant also includes sulfur pits used to store the recovered sulfur product, but it does not include secondary sulfur storage vessels or loading facilities downstream of the sulfur pits. For example, a Claus sulfur recovery plant includes: Reactor furnace and waste heat boiler, catalytic reactors, sulfur pits and, if

present, oxidation or reduction control systems or incinerator, thermal oxidizer or similar combustion device. Multiple sulfur recovery units are a single affected facility only when the units share the same source of sour gas. Sulfur recovery plants that receive source gas from completely segregated sour gas treatment systems are separate affected facilities.

Sweep gas means the gas introduced in a flare gas header system to maintain a constant flow of gas to prevent oxygen buildup in the flare header. For flares with no water seals, sweep gas also performs the function of preventing oxygen infiltration (backflow) into the flare tip.

[73 FR 35867, June 24, 2008, as amended at 77 FR 56464, Sep. 12, 2012]

#### §60.102a Emissions limitations.

- (a) Each owner or operator that is subject to the requirements of this subpart shall comply with the emissions limitations in paragraphs (b) through (i) of this section on and after the date on which the initial performance test, required by §60.8, is completed, but not later than 60 days after achieving the maximum production rate at which the affected facility will be operated or 180 days after initial startup, whichever comes first.
- (b) An owner or operator subject to the provisions of this subpart shall not discharge or cause the discharge into the atmosphere from any FCCU or FCU:
- (1) Particulate matter (PM) in excess of the limits in paragraphs (b)(1)(i), (ii), or (iii) of this section.
- (i) 1.0 kilogram per Megagram (kg/Mg)(1 pound (lb) per 1,000 lb) coke burnoff or, if a PM continuous emission monitoring system (CEMS) is used, 0.040 grain per dry standard cubic feet (gr/dscf) corrected to 0 percent excess air for each modified or reconstructed FCCII
- (ii) 0.5 gram per kilogram (g/kg) coke burn-off (0.5 lb PM/1,000 lb coke burnoff) or, if a PM CEMS is used, 0.020 gr/ dscf corrected to 0 percent excess air for each newly constructed FCCU.
- (iii) 1.0 kg/Mg (1 lb/1,000 lb) coke burn-off; or if a PM CEMS is used, 0.040 grain per dry standard cubic feet (gr/

dscf) corrected to 0 percent excess air for each affected FCU.

- (2) Nitrogen oxides  $(NO_X)$  in excess of 80 parts per million by volume (ppmv), dry basis corrected to 0 percent excess air, on a 7-day rolling average basis.
- (3) Sulfur dioxide (SO<sub>2</sub>) in excess of 50 ppmv dry basis corrected to 0 percent excess air, on a 7-day rolling average basis and 25 ppmv, dry basis corrected to 0 percent excess air, on a 365-day rolling average basis.
- (4) Carbon monoxide (CO) in excess of 500 ppmv, dry basis corrected to 0 percent excess air, on an hourly average basis.
- (c) The owner or operator of a FCCU or FCU that uses a continuous parameter monitoring system (CPMS) according to §60.105a(b)(1) shall comply with the applicable control device parameter operating limit in paragraph (c)(1) or (2) of this section.
- (1) If the FCCU or FCU is controlled using an electrostatic precipitator:
- (i) The 3-hour rolling average total power and secondary current to the entire system must not fall below the level established during the most recent performance test; and
- (ii) The daily average exhaust coke burn-off rate must not exceed the level established during the most recent performance test.
- (2) If the FCCU or FCU is controlled using a wet scrubber:
- (i) The 3-hour rolling average pressure drop must not fall below the level established during the most recent performance test; and
- (ii) The 3-hour rolling average liquidto-gas ratio must not fall below the level established during the most recent performance test.
- (d) If an FCCU or FCU uses a continuous opacity monitoring system (COMS) according to the alternative monitoring option in §60.105a(e), the 3-hour rolling average opacity of emissions from the FCCU or FCU as measured by the COMS must not exceed the site-specific opacity limit established during the most recent performance test.
- (e) The owner or operator of a FCCU or FCU that is exempted from the requirement for a CO continuous emissions monitoring system under §60.105a(h)(3) shall comply with the pa-

- rameter operating limits in paragraph (e)(1) or (2) of this section.
- (1) For a FCCU or FCU with no post-combustion control device:
- (i) The hourly average temperature of the exhaust gases exiting the FCCU or FCU must not fall below the level established during the most recent performance test.
- (ii) The hourly average oxygen (O<sub>2</sub>) concentration of the exhaust gases exiting the FCCU or FCU must not fall below the level established during the most recent performance test.
- (2) For a FCCU or FCU with a post-combustion control device:
- (i) The hourly average temperature of the exhaust gas vent stream exiting the control device must not fall below the level established during the most recent performance test.
- (ii) The hourly average  $O_2$  concentration of the exhaust gas vent stream exiting the control device must not fall below the level established during the most recent performance test.
- (f) Except as provided in paragraph (f)(3), each owner or operator of an affected sulfur recovery plant shall comply with the applicable emission limits in paragraphs (f)(1) or (2) of this section.
- (1) For a sulfur recovery plant with a capacity greater than 20 long tons per day (LTD):
- (i) For a sulfur recovery plant with an oxidation control system or a reduction control system followed by incineration, the owner or operator shall not discharge or cause the discharge of any gases into the atmosphere in excess of 250 ppm by volume (dry basis) of sulfur dioxide (SO2) at zero percent excess air. If the sulfur recovery plant consists of multiple process trains or release points the owner or operator shall comply with the 250 ppmv limit for each process train or release point or comply with a flow rate weighted average of 250 ppmv for all release points from the sulfur recovery plant;
- (ii) For a sulfur recovery plant with a reduction control system not followed by incineration, the owner or operator shall not discharge or cause the discharge of any gases into the atmosphere in excess of 300 ppmv of reduced sulfur compounds and 10 ppmv of  $\rm H_2S$ ,

each calculated as ppmv SO<sub>2</sub> (dry basis) at 0-percent excess air; or

(iii) For systems using oxygen enrichment, the owner or operator shall calculate the applicable emission limit using Equation 1 of this section:

$$E_{LS} = k_1 \times (-0.038 * (\%O_2)^2 + 11.53 * \%O_2 + 25.6)$$
 (Eq. 1)

Where:

E<sub>LS</sub> = Emission rate of SO<sub>2</sub> for large sulfur recovery plant, ppmv;

 $\begin{array}{lll} k_1 = Constant \ factor \ for \ emission \ limit \ conversion: \ k_1 = 1 \ for \ converting \ to \ SO_2 \ limit \ and \ k_1 = 1.2 \ for \ converting \ to \ the \ reduced \ sulfur \ compounds \ limit; \ and \end{array}$ 

 $\%O_2 = O_2$  concentration to the SRP, percent by volume (dry basis).

(2) For a sulfur recovery plant with a capacity of 20 LTD or less:

(i) For a sulfur recovery plant with an oxidation control system or a reduction control system followed by incineration, the owner or operator shall not discharge or cause the discharge of any gases into the atmosphere in excess of 2,500 ppm by volume (dry basis) of SO<sub>2</sub> at zero percent excess air. If the sulfur recovery plant consists of multiple process trains or release points

the owner or operator shall comply with the 2,500 ppmv limit for each process train or release point or comply with a flow rate weighted average of 2,500 ppmv for all release points from the sulfur recovery plant; or

(ii) For sulfur recovery plant with a reduction control system not followed by incineration, the owner or operator shall not discharge or cause the discharge of any gases into the atmosphere in excess of 3,000 ppm by volume of reduced sulfur compounds and 100 ppm by volume of hydrogen sulfide  $(H_2S)$ , each calculated as ppm  $SO_2$  by volume (dry basis) at zero percent excess air; or

(iii) For systems using oxygen enrichment, the owner or operator shall calculate the applicable emission limit using Equation 2 of this section:

$$E_{ss} = k_1 \times (-0.38 * (\%O_2)^2 + 115.3 * \%O_2 + 256$$
 (Eq. 2)

Where:

 $E_{SS}$  = Emission rate of  $SO_2$  for small sulfur recovery plant, ppmv.

(3) Periods of maintenance of the sulfur pit, during which the emission limits in paragraphs (f)(1) and (2) shall not apply, shall not exceed 240 hours per year. The owner or operator must document the time periods during which the sulfur pit vents were not controlled and measures taken to minimize emissions during these periods. Examples of these measures include not adding fresh sulfur or shutting off vent fans.

(g) Each owner or operator of an affected fuel gas combustion device shall comply with the emissions limits in paragraphs (g)(1) and (2) of this section.

(1) Except as provided in (g)(1)(iii) of this section, for each fuel gas combustion device, the owner or operator shall comply with either the emission limit in paragraph (g)(1)(i) of this section or the fuel gas concentration limit in paragraph (g)(1)(ii) of this section.

(i) The owner or operator shall not discharge or cause the discharge of any gases into the atmosphere that contain  $SO_2$  in excess of 20 ppmv (dry basis, corrected to 0-percent excess air) determined hourly on a 3-hour rolling average basis and  $SO_2$  in excess of 8 ppmv (dry basis, corrected to 0-percent excess air), determined daily on a 365 successive calendar day rolling average basis; or

(ii) The owner or operator shall not burn in any fuel gas combustion device

any fuel gas that contains H<sub>2</sub>S in excess of 162 ppmv determined hourly on a 3-hour rolling average basis and H<sub>2</sub>S in excess of 60 ppmv determined daily on a 365 successive calendar day rolling average basis.

(iii) The combustion in a portable generator of fuel gas released as a result of tank degassing and/or cleaning is exempt from the emissions limits in paragraphs (g)(1)(i) and (ii) of this section.

(2) For each process heater with a rated capacity of greater than 40 million British thermal units per hour (MMBtwhr) on a higher heating value basis, the owner or operator shall not discharge to the atmosphere any emissions of  $NO_X$  in excess of the applicable limits in paragraphs (g)(2)(i) through (iv) of this section.

(i) For each natural draft process heater, comply with the limit in either paragraph (g)(2)(i)(A) or (B) of this section. The owner or operator may comply with either limit at any time, provided that the appropriate parameters for each alternative are monitored as specified in \$60.107a; if fuel gas composition is not monitored as specified in \$60.107a(d), the owner or operator must comply with the concentration limits in paragraph (g)(2)(i)(A) of this section.

(A) 40 ppmv (dry basis, corrected to 0percent excess air) determined daily on a 30-day rolling average basis; or (B) 0.040 pounds per million British thermal units (lb/MMBtu) higher heating value basis determined daily on a 30-day rolling average basis.

(ii) For each forced draft process heater, comply with the limit in either paragraph (g)(2)(ii)(A) or (B) of this section. The owner or operator may comply with either limit at any time, provided that the appropriate parameters for each alternative are monitored as specified in  $\S 60.107a$ ; if fuel gas composition is not monitored as specified in  $\S 60.107a$ (d), the owner or operator must comply with the concentration limits in paragraph (g)(2)(ii)(A) of this section.

(A) 60 ppmv (dry basis, corrected to 0percent excess air) determined daily on a 30-day rolling average basis; or

(B) 0.060 lb/MMBtu higher heating value basis determined daily on a 30-day rolling average basis.

(iii) For each co-fired natural draft process heater, comply with the limit in either paragraph (g)(2)(iii)(A) or (B) of this section. The owner or operator must choose one of the emissions limits with which to comply at all times:

(A) 150 ppmv (dry basis, corrected to 0-percent excess air) determined daily on a 30 successive operating day rolling average basis; or

(B) The daily average emissions limit calculated using Equation 3 of this section:

$$ER_{NOx} = \frac{0.06 Q_{gas} HHV_{gas} + 0.35 Q_{oil} HHV_{oil}}{Q_{gas} HHV_{gas} + Q_{oil} HHV_{oil}}$$
(Eq. 3)

Where:

 $\mathrm{ER}_{\mathrm{NOx}}=\mathrm{Daily}$  allowable average emission rate of  $\mathrm{NO}_{\mathrm{X}},\ \mathrm{lb/MMBtu}$  (higher heating value basis);

Qgas = Daily average volumetric flow rate of fuel gas, standard cubic feet per day (scf/ day);

Q<sub>oil</sub> = Daily average volumetric flow rate of fuel oil, sof/day;

HHV<sub>gas</sub> = Daily average higher heating value of gas fired to the process heater, MMBtu/scf; and

HHV<sub>oil</sub> = Daily average higher heating value of fuel oil fired to the process heater, MMBtu/scf. (iv) For each co-fired forced draft process heater, comply with the limit in either paragraph (g)(2)(iv)(A) or (B) of this section. The owner or operator must choose one of the emissions limits with which to comply at all times:

(A) 150 ppmv (dry basis, corrected to 0-percent excess air) determined daily on a 30 successive operating day rolling average basis: or

(B) The daily average emissions limit calculated using Equation 4 of this section:

$$ER_{NOx} = \frac{0.11Q_{gas} HHV_{gas} + 0.40Q_{oil} HHV_{oil}}{Q_{gas} HHV_{gas} + Q_{oil} HHV_{oil}}$$
(Eq. 4)

Where:

 $\mathrm{ER}_{NOx}$  = Daily allowable average emission rate of  $\mathrm{NO}_{x}$ , lb/MMBtu (higher heating value basis);

Q<sub>gus</sub> = Daily average volumetric flow rate of fuel gas, scf/day;

Q<sub>oii</sub> = Daily average volumetric flow rate of fuel oil, scf/day;

HHV<sub>gas</sub> = Daily average higher heating value of gas fired to the process heater, MMBtu/scf; and

HHV<sub>oil</sub> = Daily average higher heating value of fuel oil fired to the process heater, MMBtu/scf.

- (h) [Reserved]
- (i) For a process heater that meets any of the criteria of paragraphs (i)(1)(i) through (iv) of this section, an owner or operator may request approval from the Administrator for a  $NO_X$  emissions limit which shall apply specifically to that affected facility. The request shall include information as described in paragraph (i)(2) of this section. The request shall be submitted and followed as described in paragraph (i)(3) of this section.
- (1) A process heater that meets one of the criteria in paragraphs (i)(1)(i) through (iv) of this section may apply for a site-specific  $NO_X$  emissions limit:
- (i) A modified or reconstructed process heater that lacks sufficient space to accommodate installation and proper operation of combustion modification-based technology (e.g., ultra-low NO<sub>x</sub> burners); or
- (ii) A modified or reconstructed process heater that has downwardly firing induced draft burners; or
- (iii) A co-fired process heater; or
- (iv) A process heater operating at reduced firing conditions for an extended period of time (i.e., operating in turndown mode). The site-specific  $NO_X$  emissions limit will only apply for those operating conditions.
- (2) The request shall include sufficient and appropriate data, as determined by the Administrator, to allow the Administrator to confirm that the process heater is unable to comply with the applicable NO<sub>X</sub> emissions

limit in paragraph (g)(2) of this section. At a minimum, the request shall contain the information described in paragraphs (i)(2)(i) through (iv) of this section.

- (i) The design and dimensions of the process heater, evaluation of available combustion modification-based technology, description of fuel gas and, if applicable, fuel oil characteristics, information regarding the combustion conditions (temperature, oxygen content, firing rates) and other information needed to demonstrate that the process heater meets one of the four classes of process heaters listed in paragraph (i)(1) of this section.
- (ii) An explanation of how the data in paragraph (i)(2)(i) demonstrate that ultra-low  $NO_X$  burners, flue gas recirculation, control of excess air or other combustion modification-based technology (including combinations of these combustion modification-based technologies) cannot be used to meet the applicable emissions limit in paragraph (g)(2) of this section.

(iii) Results of a performance test conducted under representative conditions using the applicable methods specified in §60.104a(i) to demonstrate the performance of the technology the owner or operator will use to minimize NO<sub>2</sub> emissions.

(iv) The means by which the owner or operator will document continuous compliance with the site-specific emissions limit.

(3) The request shall be submitted and followed as described in paragraphs (i)(3)(i) through (iii) of this section.

(i) The owner or operator of a process heater that meets one of the criteria in paragraphs (i)(1)(i) through (iv) of this section may request approval from the Administrator within 180 days after initial startup of the process heater for a  $NO_X$  emissions limit which shall apply specifically to that affected facility.

(ii) The request must be submitted to the Administrator for approval. The owner or operator must comply with the request as submitted until it is approved.

- (iii) The request shall also be submitted to the following address: U.S. Environmental Protection Agency, Ofice of Air Quality Planning and Standards, Sector Policies and Programs Division, U.S. EPA Mailroom (E143-01), Attention: Refinery Sector Lead, 109 T.W. Alexander Drive, Research Triangle Park, NC 27711. Electronic copies in lieu of hard copies may also be submitted to refinerynsps@epa.gov.
- (4) The approval process for a request for a facility-specific  $NO_X$  emissions limit is described in paragraphs (i)(4)(i) through (iii) of this section.
- (i) Approval by the Administrator of a facility-specific  $NO_X$  emissions limit request will be based on the completeness, accuracy and reasonableness of the request. Factors that the EPA will consider in reviewing the request for approval include, but are not limited to, the following:
- (A) A demonstration that the process heater meets one of the four classes of process heaters outlined in paragraphs (i)(1) of this section:
- (B) A description of the low- $NO_X$  burner designs and other combustion modifications considered for reducing  $NO_X$  emissions;
- (C) The combustion modification option selected; and
- (D) The operating conditions (firing rate, heater box temperature and excess oxygen concentration) at which the  $NO_X$  emission level was established.
- (ii) If the request is approved by the Administrator, a facility-specific  $NO_X$  emissions limit will be established at the  $NO_X$  emission level demonstrated in the approved request.
- (iii) If the Administrator finds any deficiencies in the request, the request must be revised to address the deficiencies and be re-submitted for approval.

[73 FR 35867, June 24, 2008, as amended at 77 FR 56466, Sep. 12, 2012]

## § 60.103a Design, equipment, work practice or operational standards.

(a) Except as provided in paragraph (g) of this section, each owner or operator that operates a flare that is subject to this subpart shall develop and implement a written flare management

- plan no later than the date specified in paragraph (b) of this section. The flare management plan must include the information described in paragraphs (a)(1) through (7) of this section.
- (1) A listing of all refinery process units, ancillary equipment, and fuel gas systems connected to the flare for each affected flare.
- (2) An assessment of whether discharges to affected flares from these process units, ancillary equipment and fuel gas systems can be minimized. The flare minimization assessment must (at a minimum) consider the items in paragraphs (a)(2)(i) through (iv) of this section. The assessment must provide clear rationale in terms of costs (capital and annual operating), natural gas offset credits (if applicable), technical feasibility, secondary environmental impacts and safety considerations for selected minimization alternative(s) or a statement, with justifications, that flow reduction could not be achieved. Based upon the assessment, each owner or operator of an affected flare shall identify the minimization alternatives that it has implemented by the due date of the flare management plan and shall include a schedule for the prompt implementation of any selected measures that cannot reasonably be completed as of that date.
- (i) Elimination of process gas discharge to the flare through process operating changes or gas recovery at the source.
- (ii) Reduction of the volume of process gas to the flare through process operating changes.
- (iii) Installation of a flare gas recovery system or, for facilities that are fuel gas rich, a flare gas recovery system and a co-generation unit or combined heat and power unit.
- (iv) Minimization of sweep gas flow rates and, for flares with water seals, purge gas flow rates.
- (3) A description of each affected flare containing the information in paragraphs (a)(3)(i) through (vii) of this section
- (i) A general description of the flare, including the information in paragraphs (a)(3)(i)(A) through (G) of this section.
- (A) Whether it is a ground flare or elevated (including height).

### §60.103a

- (B) The type of assist system (e.g., air, steam, pressure, non-assisted).
- (C) Whether it is simple or complex flare tip (e.g., staged, sequential).
- (D) Whether the flare is part of a cascaded flare system (and if so, whether the flare is primary or secondary).
- (E) Whether the flare serves as a backup to another flare.
- (F) Whether the flare is an emergency flare or a non-emergency flare.
- (G) Whether the flare is equipped with a flare gas recovery system.
- (ii) Description and simple process flow diagram showing the interconnection of the following components of the flare: flare tip (date installed, manufacturer, nominal and effective tip diameter, tip drawing); knockout or surge drum(s) or pot(s) (including dimensions and design capacities); flare header(s) and subheader(s); assist system; and ignition system.
- (iii) Flare design parameters, including the maximum vent gas flow rate; minimum sweep gas flow rate; minimum purge gas flow rate (if any); maximum supplemental gas flow rate; maximum pilot gas flow rate; and, if the flare is steam-assisted, minimum total steam rate.
- (iv) Description and simple process flow diagram showing all gas lines (including flare, purge (if applicable), sweep, supplemental and pilot gas) that are associated with the flare. For purge, sweep, supplemental and pilot gas, identify the type of gas used. Designate which lines are exempt from sulfur,  $\rm H_2S$  or flow monitoring and why (e.g., natural gas, inherently low sulfur, pilot gas). Designate which lines are monitored and identify on the process flow diagram the location and type of each monitor.
- (v) For each flow rate, H<sub>2</sub>S, sulfur content, pressure or water seal monitor identified in paragraph (a)(3)(iv) of this section, provide a detailed description of the manufacturer's specifications, including, but not limited to, make, model, type, range, precision, accuracy, calibration, maintenance and quality assurance procedures.
- (vi) For emergency flares, secondary flares and flares equipped with a flare gas recovery system designed, sized and operated to capture all flows ex-

- cept those resulting from startup, shutdown or malfunction:
- (A) Description of the water seal, including the operating range for the liquid level.
- (B) Designation of the monitoring option elected (flow and sulfur monitoring or pressure and water seal liquid level monitoring).
- (vii) For flares equipped with a flare gas recovery system:
- (A) A description of the flare gas recovery system, including number of compressors and capacity of each compressor.
- (B) A description of the monitoring parameters used to quantify the amount of flare gas recovered.
- (C) For systems with staged compressors, the maximum time period required to begin gas recovery with the secondary compressor(s), the monitoring parameters and procedures used to minimize the duration of releases during compressor staging and a justification for why the maximum time period cannot be further reduced.
- (4) An evaluation of the baseline flow to the flare. The baseline flow to the flare must be determined after implementing the minimization assessment in paragraph (a)(2) of this section. Baseline flows do not include pilot gas flow or purge gas flow (i.e., gas introduced after the flare's water seal) provided these gas flows remain reasonably constant (i.e., separate flow monitors for these streams are not required). Separate baseline flow rates may be established for different operating conditions provided that the management plan includes:
- (i) A primary baseline flow rate that will be used as the default baseline for all conditions except those specifically delineated in the plan;
- (ii) A description of each special condition for which an alternate baseline is established, including the rationale for each alternate baseline, the daily flow for each alternate baseline and the expected duration of the special conditions for each alternate baseline; and
- (iii) Procedures to minimize discharges to the affected flare during each special condition described in paragraph (a)(4)(ii) of this section, unless procedures are already developed for these cases under paragraph (a)(5)

through (7) of this section, as applicable.

- (5) Procedures to minimize or eliminate discharges to the flare during the planned startup and shutdown of the refinery process units and ancillary equipment that are connected to the affected flare, together with a schedule for the prompt implementation of any procedures that cannot reasonably be implemented as of the date of the submission of the flare management plan.
- (6) Procedures to reduce flaring in cases of fuel gas imbalance (i.e., excess fuel gas for the refinery's energy needs), together with a schedule for the prompt implementation of any procedures that cannot reasonably be implemented as of the date of the submission of the flare management plan.
- (7) For flares equipped with flare gas recovery systems, procedures to minimize the frequency and duration of outages of the flare gas recovery system and procedures to minimize the volume of gas flared during such outages, together with a schedule for the prompt implementation of any procedures that cannot reasonably be implemented as of the date of the submission of the flare management plan.
- (b) Except as provided in paragraph (g) of this section, each owner or operator required to develop and implement a written flare management plan as described in paragraph (a) of this section must submit the plan to the Administrator as described in paragraphs (b)(1) through (3) of this section.
- (1) The owner or operator of a newly constructed or reconstructed flare must develop and implement the flare management plan by no later than the date that the flare becomes an affected facility subject to this subpart, except for the selected minimization alternatives in paragraph (a)(2) and/or the procedures in paragraphs (a)(5) though (a)(7) of this section that cannot reasonably be implemented by that date. which the owner or operator must implement in accordance with the schedule in the flare management plan. The owner or operator of a modified flare must develop and implement the flare management plan by no later than November 11, 2015 or upon startup of the modified flare, whichever is later.

- (2) The owner or operator must comply with the plan as submitted by the date specified in paragraph (b)(1) of this section. The plan should be updated periodically to account for changes in the operation of the flare. such as new connections to the flare or the installation of a flare gas recovery system, but the plan need be re-submitted to the Administrator only if the owner or operator adds an alternative baseline flow rate, revises an existing baseline as described in paragraph (a)(4) of this section, installs a flare gas recovery system or is required to change flare designations and monitoring methods as described in §60.107a(g). The owner or operator must comply with the updated plan as submitted.
- (3) All versions of the plan submitted to the Administrator shall also be submitted to the following address: U.S. Environmental Protection Agency, Office of Air Quality Planning and Standards, Sector Policies and Programs Division, U.S. EPA Mailroom (E143-01), Attention: Refinery Sector Lead, 109 T.W. Alexander Drive, Research Triangle Park, NC 27711. Electronic copies in lieu of hard copies may also be submitted to refinerynsps@epa.gov.
- (c) Except as provided in paragraphs (f) and (g) of this section, each owner or operator that operates a fuel gas combustion device, flare or sulfur recovery plant subject to this subpart shall conduct a root cause analysis and a corrective action analysis for each of the conditions specified in paragraphs (c)(1) through (3) of this section.
  - (1) For a flare:
- (i) Any time the  $SO_2$  emissions exceed 227 kilograms (kg) (500 lb) in any 24-hour period; or
- (ii) Any discharge to the flare in excess of 14,160 standard cubic meters (m³) (500,000 standard cubic feet (scf)) above the baseline, determined in paragraph (a)(4) of this section, in any 24-hour period; or
- (iii) If the monitoring alternative in §60.107a(g) is elected, any period when the flare gas line pressure exceeds the water seal liquid depth, except for periods attributable to compressor staging that do not exceed the staging time specified in paragraph (a)(3)(vii)(C) of this section

### §60.103a

- (2) For a fuel gas combustion device, each exceedance of an applicable short-term emissions limit in §60.102a(g)(1) if the SO<sub>2</sub> discharge to the atmosphere is 227 kg (500 lb) greater than the amount that would have been emitted if the emissions limits had been met during one or more consecutive periods of excess emissions or any 24-hour period, whichever is shorter.
- (3) For a sulfur recovery plant, each time the  $SO_2$  emissions are more than 227 kg (500 lb) greater than the amount that would have been emitted if the  $SO_2$  or reduced sulfur concentration was equal to the applicable emissions limit in  $\S60.102a(f)(1)$  or (2) during one or more consecutive periods of excess emissions or any 24-hour period, whichever is shorter
- (d) Except as provided in paragraphs (f) and (g) of this section, a root cause analysis and corrective action analysis must be completed as soon as possible, but no later than 45 days after a discharge meeting one of the conditions specified in paragraphs (c)(1) through (3) of this section. Special circumstances affecting the number of root cause analyses and/or corrective action analyses are provided in paragraphs (d)(1) through (5) of this section.
- (1) If a single continuous discharge meets any of the conditions specified in paragraphs (c)(1) through (3) of this section for 2 or more consecutive 24-hour periods, a single root cause analysis and corrective action analysis may be conducted.
- (2) If a single discharge from a flare triggers a root cause analysis based on more than one of the conditions specified in paragraphs (c)(1)(i) through (iii) of this section, a single root cause analysis and corrective action analysis may be conducted.
- (3) If the discharge from a flare is the result of a planned startup or shutdown of a refinery process unit or ancillary equipment connected to the affected flare and the procedures in paragraph (a)(5) of this section were followed, a root cause analysis and corrective action analysis is not required; however, the discharge must be recorded as described in §60.108a(c)(6) and reported as described in §60.108a(d)(5).
- (4) If both the primary and secondary flare in a cascaded flare system meet

- any of the conditions specified in paragraphs (c)(1)(i) through (iii) of this section in the same 24-hour period, a single root cause analysis and corrective action analysis may be conducted.
- (5) Except as provided in paragraph (d)(4) of this section, if discharges occur that meet any of the conditions specified in paragraphs (c)(1) through (3) of this section for more than one affected facility in the same 24-hour period, initial root cause analyses shall be conducted for each affected facility. If the initial root cause analyses indicate that the discharges have the same root cause(s), the initial root cause analyses can be recorded as a single root cause analysis and a single corrective action analysis may be conducted.
- (e) Except as provided in paragraphs (f) and (g) of this section, each owner or operator of a fuel gas combustion device, flare or sulfur recovery plant subject to this subpart shall implement the corrective action(s) identified in the corrective action analysis conducted pursuant to paragraph (d) of this section in accordance with the applicable requirements in paragraphs (e)(1) through (3) of this section.
- (1) All corrective action(s) must be implemented within 45 days of the discharge for which the root cause and corrective action analyses were required or as soon thereafter as practicable. If an owner or operator concludes that corrective action should not be conducted, the owner or operator shall record and explain the basis for that conclusion no later than 45 days following the discharge as specified in §60.108a(c)(6)(ix).
- (2) For corrective actions that cannot be fully implemented within 45 days following the discharge for which the root cause and corrective action analyses were required, the owner or operator shall develop an implementation schedule to complete the corrective action(s) as soon as practicable.
- (3) No later than 45 days following the discharge for which a root cause and corrective action analyses were required, the owner or operator shall

record the corrective action(s) completed to date, and, for action(s) not already completed, a schedule for implementation, including proposed commencement and completion dates as specified in §60.108a(c)(6)(x).

(f) Modified flares shall comply with the requirements of paragraphs (c) through (e) of this section by November 11, 2015 or at startup of the modified flare, whichever is later. Modified flares that were not affected facilities subject to subpart J of this part prior to becoming affected facilities under §60.100a shall comply with the requirements of paragraph (h) of this section and the requirements of §60.107a(a)(2) by November 11, 2015 or at startup of the modified flare, whichever is later. Modified flares that were affected facilities subject to subpart J of this part prior to becoming affected facilities under §60.100a shall comply with the requirements of paragraph (h) of this section and the requirements of §60.107a(a)(2) by November 13, 2012 or at startup of the modified flare, whichever is later, except that modified flares that have accepted applicability of subpart J under a federal consent decree shall comply with the subpart J requirements as specified in the consent decree, but shall comply with the requirements of paragraph (h) of this section and the requirements of §60.107a(a)(2) by no later than November 11, 2015.

(g) An affected flare subject to this subpart located in the Bay Area Air Quality Management District (BAAQMD) may elect to comply with both BAAQMD Regulation 12, Rule 11 and BAAQMD Regulation 12, Rule 12 as an alternative to complying with the requirements of paragraphs (a) through (e) of this section. An affected flare subject to this subpart located in the South Coast Air Quality Management District (SCAQMD) may elect to comply with SCAQMD Rule 1118 as an alternative to complying with the requirements of paragraphs (a) through (e) of this section. The owner or operator of an affected flare must notify the Administrator that the flare is in compliance with BAAQMD Regulation 12, Rule 11 and BAAQMD Regulation 12, Rule 12 or SCAQMD Rule 1118. The owner or operator of an affected flare shall also submit the existing flare management plan to the following address: U.S. Environmental Protection Agency, Office of Air Quality Planning and Standards, Sector Policies and Programs Division, U.S. EPA Mailroom (E143-01), Attention: Refinery Sector Lead, 109 T.W. Alexander Drive, Research Triangle Park, NC 27711. Electronic copies in lieu of hard copies may also be submitted to refinerynsps@epa.gov.

- (h) Each owner or operator shall not burn in any affected flare any fuel gas that contains  $H_2S$  in excess of 162 ppmv determined hourly on a 3-hour rolling average basis. The combustion in a flare of process upset gases or fuel gas that is released to the flare as a result of relief valve leakage or other emergency malfunctions is exempt from this limit.
- (i) Each owner or operator of a delayed coking unit shall depressure each coke drum to 5 lb per square inch gauge (psig) or less prior to discharging the coke drum steam exhaust to the atmosphere. Until the coke drum pressure reaches 5 psig, the coke drum steam exhaust must be managed in an enclosed blowdown system and the uncondensed vapor must either be recovered (e.g., sent to the delayed coking unit fractionators) or vented to the fuel gas system, a fuel gas combustion device or a flare.
- (j) Alternative means of emission limitation. (1) Each owner or operator subject to the provisions of this section may apply to the Administrator for a determination of equivalence for any means of emission limitation that achieves a reduction in emissions of a specified pollutant at least equivalent to the reduction in emissions of that pollutant achieved by the controls required in this section.
- (2) Determination of equivalence to the design, equipment, work practice or operational requirements of this section will be evaluated by the following guidelines:
- (i) Each owner or operator applying for a determination of equivalence shall be responsible for collecting and verifying test data to demonstrate the equivalence of the alternative means of emission limitation.

### § 60.104a

- (ii) For each affected facility for which a determination of equivalence is requested, the emission reduction achieved by the design, equipment, work practice or operational requirements shall be demonstrated.
- (iii) For each affected facility for which a determination of equivalence is requested, the emission reduction achieved by the alternative means of emission limitation shall be demonstrated.
- (iv) Each owner or operator applying for a determination of equivalence to a work practice standard shall commit in writing to work practice(s) that provide for emission reductions equal to or greater than the emission reductions achieved by the required work practice.
- (v) The Administrator will compare the demonstrated emission reduction for the alternative means of emission limitation to the demonstrated emission reduction for the design, equipment, work practice or operational requirements and, if applicable, will consider the commitment in paragraph (j)(2)(iy) of this section.
- (vi) The Administrator may condition the approval of the alternative means of emission limitation on requirements that may be necessary to ensure operation and maintenance to achieve the same emissions reduction as the design, equipment, work practice or operational requirements.
- (3) An owner or operator may offer a unique approach to demonstrate the equivalence of any equivalent means of emission limitation.
- (4) Approval of the application for equivalence to the design, equipment, work practice or operational requirements of this section will be evaluated by the following guidelines:
- (i) After a request for determination of equivalence is received, the Administrator will publish a notice in the FEDERAL REGISTER and provide the opportunity for public hearing if the Administrator judges that the request may be approved.
- (ii) After notice and opportunity for public hearing, the Administrator will determine the equivalence of a means of emission limitation and will publish the determination in the FEDERAL REGISTER.

- (iii) Any equivalent means of emission limitations approved under this section shall constitute a required work practice, equipment, design or operational standard within the meaning of section 111(h)(1) of the CAA.
- (5) Manufacturers of equipment used to control emissions may apply to the Administrator for determination of equivalence for any alternative means of emission limitation that achieves a reduction in emissions achieved by the equipment, design and operational requirements of this section. The Administrator will make an equivalence determination according to the provisions of paragraphs (j)(2) through (4) of this section.

[77 FR 56467, Sep. 12, 2012]

### §60.104a Performance tests.

- (a) The owner or operator shall conduct a performance test for each FCCII FCU, sulfur recovery plant, flare and fuel gas combustion device to demonstrate initial compliance with each applicable emissions limit in §60.102a according to the requirements of §60.8. The notification requirements §60.8(d) apply to the initial performance test and to subsequent performance tests required by paragraph (b) of this section (or as required by the Administrator), but does not apply to performance tests conducted for the purpose of obtaining supplemental data because of continuous monitoring system breakdowns, repairs, calibration checks and zero and span adjustments.
- (b) The owner or operator of a FCCU or FCU that elects to monitor control device operating parameters according to the requirements in §60.105a(b), to use bag leak detectors according to the requirements in §60.105a(c), or to use COMS according to the requirements in §60.105a(e) shall conduct a PM performance test at least once every 12 months and furnish the Administrator a written report of the results of each test.
- (c) In conducting the performance tests required by this subpart (or as requested by the Administrator), the owner or operator shall use the test methods in 40 CFR part 60, Appendices A-1 through A-8 or other methods as specified in this section, except as provided in §60.8(b).

- (d) The owner or operator shall determine compliance with the PM,  $NO_X$ ,  $SO_2$ , and CO emissions limits in  $\S 60.102a(b)$  for FCCU and FCU using the following methods and procedures:
- (1) Method 1 of appendix A-1 to part 60 for sample and velocity traverses.
- (2) Method 2 of appendix A-1 to part 60 for velocity and volumetric flow rate.
- (3) Method 3, 3A, or 3B of appendix A-2 to part 60 for gas analysis. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 3B of appendix A-2 to part 60.
- (4) Method 5, 5B, or 5F of appendix A-3 to part 60 for determining PM emissions and associated moisture content from a FCCU or FCU without a wet scrubber subject to the emissions limit in §63.102a(b)(1). Use Method 5 or 5B of appendix A-3 to part 60 for determining PM emissions and associated moisture content from a FCCU or FCU with a wet scrubber subject to the emissions limit in §63.102a(b)(1).
- (i) The PM performance test consists of 3 valid test runs; the duration of each test run must be no less than 60 minutes.
- (ii) The emissions rate of PM  $(E_{PM})$  is computed for each run using Equation 5 of this section:

$$E = \frac{c_s Q_{sd}}{K R_c}$$
 (Eq. 5)

Where:

- E = Emission rate of PM, g/kg (lb/1,000 lb) of coke burn-off;
- c<sub>s</sub> = Concentration of total PM, grams per dry standard cubic meter (g/dscm) (gr/ dscf);
- Q<sub>sd</sub> = Volumetric flow rate of effluent gas, dry standard cubic meters per hour (dry standard cubic feet per hour);
- $R_c$  = Coke burn-off rate, kilograms per hour (kg/hr) [lb per hour (lb/hr)] coke; and
- K = Conversion factor, 1.0 grams per gram (7,000 grains per lb).
- (iii) The coke burn-off rate  $(R_c)$  is computed for each run using Equation 6 of this section:

$$R_{c} = K_{1}Q_{c}(\%CQ_{c} + \%CQ_{c}) + K_{2}Q_{a} - K_{3}Q_{c}(\%CQ_{c}) + \%CQ_{c} + \%CQ_{c} + \%Q_{c}) + K_{3}Q_{oxy}(\%Q_{oxy})$$
(Eq. 6)

Where:

 $R_c$  = Coke burn-off rate, kg/hr (lb/hr);

- Qr = Volumetric flow rate of exhaust gas from FCCU regenerator or fluid coking burner before any emissions control or energy recovery system that burns auxiliary fuel, dry standard cubic meters per minute (dscm/min) [dry standard cubic feet per minute (dscf/min)];
- Q<sub>a</sub> = Volumetric flow rate of air to FCCU regenerator or fluid coking burner, as determined from the unit's control room instrumentation, dscm/min (dscf/min);
- $Q_{\rm oxy} = {
  m Volumetric}$  flow rate of  $O_2$  enriched air to FCCU regenerator or fluid coking unit, as determined from the unit's con-

trol room instrumentation, dscm/min (dscf/min);

- %CO<sub>2</sub> = Carbon dioxide (CO<sub>2</sub>) concentration in FCCU regenerator or fluid coking burner exhaust, percent by volume (dry basis);
- %CO = CO concentration in FCCU regenerator or fluid coking burner exhaust, percent by volume (dry basis);
- %O<sub>2</sub> = O<sub>2</sub> concentration in FCCU regenerator or fluid coking burner exhaust, percent by volume (dry basis);
- %O<sub>oxy</sub> = O<sub>2</sub> concentration in O<sub>2</sub> enriched air stream inlet to the FCCU regenerator or fluid coking burner, percent by volume (dry basis);

 $K_1$  = Material balance and conversion factor, 0.2982 (kg-min)/(hr-dscm-%) [0.0186 (lb-min)/(hr-dscf-%)];

K<sub>2</sub> = Material balance and conversion factor, 2.088 (kg-min)/(hr-dscm) [0.1303 (lb-min)/ (hr-dscf)]; and

K<sub>3</sub> = Material balance and conversion factor, 0.0994 (kg-min)/(hr-dscm-%) [0.00624 (lb-min)/(hr-dscf-%)].

(iv) During the performance test, the volumetric flow rate of exhaust gas from catalyst regenerator  $(Q_r)$  before

any emission control or energy recovery system that burns auxiliary fuel is measured using Method 2 of appendix A-1 to part 60.

(v) For subsequent calculations of coke burn-off rates or exhaust gas flow rates, the volumetric flow rate of  $Q_r$  is calculated using average exhaust gas concentrations as measured by the monitors required in  $\S 60.105a(b)(2)$ , if applicable, using Equation 7 of this section:

$$Q_r = \frac{79 \times Q_a + (100 - \%Oxy) \times Q_{oxy}}{100 - \%CO_2 - \%CO - \%O_2}$$
 (Eq. 7)

Where:

 $Q_r$  = Volumetric flow rate of exhaust gas from FCCU regenerator or fluid coking burner before any emission control or energy recovery system that burns auxiliary fuel, dscm/min (dscf/min);

Q<sub>a</sub> = Volumetric flow rate of air to FCCU regenerator or fluid coking burner, as determined from the unit's control room instrumentation, dscm/min (dscf/min);

Q<sub>oxy</sub> = Volumetric flow rate of O<sub>2</sub> enriched air to FCCU regenerator or fluid coking unit, as determined from the unit's control room instrumentation, dscm/min (dscf/min);

%CO<sub>2</sub> = Carbon dioxide concentration in FCCU regenerator or fluid coking burner exhaust, percent by volume (dry basis);

%CO = CO concentration FCCU regenerator or fluid coking burner exhaust, percent by volume (dry basis). When no auxiliary fuel is burned and a continuous CO monitor is not required in accordance with §60.105a(h)(3), assume %CO to be zero;

%O<sub>2</sub> = O<sub>2</sub> concentration in FCCU regenerator or fluid coking burner exhaust, percent by volume (dry basis); and

%O<sub>oxy</sub> = O<sub>2</sub> concentration in O<sub>2</sub> enriched air stream inlet to the FCCU regenerator or fluid coking burner, percent by volume (dry basis).

(5) Method 6, 6A, or 6C of appendix A-4 to part 60 for moisture content and

for the concentration of  $SO_2$ ; the duration of each test run must be no less than 4 hours. The method ANSI/ASME PTC 19.10–1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see  $\S 60.17$ ) is an acceptable alternative to EPA Method 6 or 6A of appendix A-4 to part 60.

(6) Method 7, 7A, 7C, 7D, or 7E of appendix A-4 to part 60 for moisture content and for the concentration of  $NO_X$  calculated as nitrogen dioxide ( $NO_2$ ); the duration of each test run must be no less than 4 hours. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see  $\S 60.17$ ) is an acceptable alternative to EPA Method 7 or 7C of appendix A-4 to part 60.

(7) Method 10, 10A, or 10B of appendix A-4 to part 60 for moisture content and for the concentration of CO. The sampling time for each run must be 60 minutes.

(8) The owner or operator shall adjust PM,  $NO_X$ ,  $SO_2$  and CO pollutant concentrations to 0-percent excess air or 0-percent  $O_2$  using Equation 8 of this section:

$$C_{adj} = C_{meas} \left[ 20.9_{c} / (20.9 - \% O_{2}) \right]$$
 (Eq. 8)

Where

 $C_{adj}$  = pollutant concentration adjusted to 0-percent excess air or  $O_2$ , parts per million (ppm) or g/dscm;

C<sub>meas</sub> = pollutant concentration measured on a dry basis, ppm or g/dscm;

 $20.9_c = 20.9$  percent  $O_2 - 0.0$  percent  $O_2$  (defined  $O_2$  correction basis), percent;

 $20.9 = O_2$  concentration in air, percent; and  ${}^{\circ}\!\!\!/ O_2 = O_2$  concentration measured on a dry basis, percent.

- (e) The owner or operator of a FCCU or FCU that is controlled by an electrostatic precipitator or wet scrubber and that is subject to control device operating parameter limits in §60.102a(c) shall establish the limits based on the performance test results according to the following procedures:
- (1) Reduce the parameter monitoring data to hourly averages for each test run.

(2) Determine the hourly average operating limit for each required parameter as the average of the three test runs.

(f) The owner or operator of an FCCU or FCU that uses cyclones to comply with the PM limit in §60.102a(b)(1) and elects to comply with the COMS alternative monitoring option in §60.105a(d) shall establish a site-specific opacity operating limit according to the procedures in paragraphs (f)(1) through (3) of this section.

(1) Collect COMS data every 10 seconds during the entire period of the PM performance test and reduce the data to 6-minute averages.

(2) Determine and record the hourly average opacity from all the 6-minute averages.

(3) Compute the site-specific limit using Equation 9 of this section:

Opacity Limit = Opacity<sub>u</sub> 
$$x \left( \frac{1 \text{lb}/1,000 \text{ lb coke burn}}{PMEmR_u} \right)$$
 (Eq. 9)

Where:

Opacity limit = Maximum permissible 3-hour average opacity, percent, or 10 percent, whichever is greater;

Opacity<sub>st</sub> = Hourly average opacity measured during the source test, percent; and

 $PMEmR_{st} = PM$  emission rate measured during the source test, 1b/1,000 1b coke burn.

(g) The owner or operator of a FCCU or FCU that is exempt from the requirement to install and operate a CO CEMS pursuant to \$60.105a(h)(3) and that is subject to control device operating parameter limits in \$60.102a(c) shall establish the limits based on the performance test results using the procedures in paragraphs (g)(1) and (2) of this section.

- (1) Reduce the temperature and  $O_2$  concentrations from the parameter monitoring systems to hourly averages for each test run.
- (2) Determine the operating limit for temperature and  $O_2$  concentrations as the average of the average temperature and  $O_2$  concentration for the three test runs.
- (h) The owner or operator shall determine compliance with the SO<sub>2</sub> and H<sub>2</sub>S emissions limits for sulfur recovery

plants in  $\S 60.102a(f)(1)(i), 60.102a(f)(1)(iii), 60.102a(f)(2)(i), and <math display="inline">60.102a(f)(2)(iii)$  and the reduced sulfur compounds and  $H_2S$  emissions limits for sulfur recovery plants in  $\S 60.102a(f)(1)(ii)$  and  $\S 60.102a(f)(2)(ii)$  using the following methods and procedures:

(1) Method 1 of appendix A-1 to part 60 for sample and velocity traverses.

(2) Method 2 of appendix A-1 to part 60 for velocity and volumetric flow rate.

(3) Method 3, 3A, or 3B of appendix A-2 to part 60 for gas analysis. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 3B of appendix A-2 to part 60.

(4) Method 6, 6A, or 6C of appendix A-4 to part 60 to determine the SO<sub>2</sub> concentration. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 6 or 6A of appendix A-4 to part 60.

(5) Method 15 or 15A of appendix A-5 to part 60 or Method 16 of appendix A-

- 6 to part 60 to determine the reduced sulfur compounds and  $H_2S$  concentrations. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see  $\S 60.17$ ) is an acceptable alternative to EPA Method 15A of appendix A-5 to part 60.
- (i) Each run consists of 16 samples taken over a minimum of 3 hours.
- (ii) The owner or operator shall calculate the average  $\rm H_2S$  concentration after correcting for moisture and  $\rm O_2$  as the arithmetic average of the  $\rm H_2S$  concentration for each sample during the run (ppmv, dry basis, corrected to 0 percent excess air).
- (iii) The owner or operator shall calculate the  $SO_2$  equivalent for each run after correcting for moisture and  $O_2$  as the arithmetic average of the  $SO_2$  equivalent of reduced sulfur compounds for each sample during the run (ppmv, dry basis, corrected to 0 percent excess air).
- (iv) The owner or operator shall use Equation 8 of this section to adjust pollutant concentrations to 0-percent  ${\rm O_2}$  or 0-percent excess air.
- (i) The owner or operator shall determine compliance with the  $SO_2$  and  $NO_X$  emissions limits in  $\S 60.102a(g)$  for a fuel gas combustion device according to the following test methods and procedures:
- (1) Method 1 of appendix A-1 to part 60 for sample and velocity traverses;
- (2) Method 2 of appendix A-1 to part 60 for velocity and volumetric flow rate;
- (3) Method 3, 3A, or 3B of appendix A-2 to part 60 for gas analysis. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 3B of appendix A-2 to part 60;
- (4) Method 6, 6A, or 6C of appendix A-4 to part 60 to determine the  $SO_2$  concentration. The method ANSI/ASME PTC 19.10–1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 6 or 6A of appendix A-4 to part 60.
- (i) The performance test consists of 3 valid test runs; the duration of each test run must be no less than 1 hour.
- (ii) If a single fuel gas combustion device having a common source of fuel

- gas is monitored as allowed under  $\S60.107a(a)(1)(v)$ , only one performance test is required. That is, performance tests are not required when a new affected fuel gas combustion device is added to a common source of fuel gas that previously demonstrated compliance.
- (5) Method 7, 7A, 7C, 7D, or 7E of appendix A-4 to part 60 for moisture content and for the concentration of  $NO_X$  calculated as  $NO_2$ ; the duration of each test run must be no less than 4 hours. The method ANSI/ASME PTC 19.10–1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 7 or 7C of appendix A-4 to part 60.
- (6) For process heaters with a rated heat capacity between 40 and 100 MMBtu/hr that elect to demonstrate continuous compliance with a maximum excess oxygen limit as provided in  $\S 60.107a(c)(6)$  or (d)(8), the owner or operator shall establish the  $O_2$  operating limit or  $O_2$  operating curve based on the performance test results according to the requirements in paragraph (i)(6)(i) or (ii) of this section, respectively.
- (i) If a single O<sub>2</sub> operating limit will be used:
- (A) Conduct the performance test following the methods provided in paragraphs (i)(1), (2), (3) and (5) of this section when the process heater is firing at no less than 70 percent of the rated heat capacity. For co-fired process heaters, conduct at least one of the test runs while the process heater is being supplied by both fuel gas and fuel oil and conduct at least one of the test runs while the process heater is being supplied solely by fuel gas.
- (B) Each test will consist of three test runs. Calculate the  $NO_X$  concentration for the performance test as the average of the  $NO_X$  concentrations from each of the three test runs. If the  $NO_X$  concentration for the performance test is less than or equal to the numerical value of the applicable  $NO_X$  emissions limit (regardless of averaging time), then the test is considered to be a valid test.
- (C) Determine the average  ${\rm O}_2$  concentration for each test run of a valid test.

- (D) Calculate the  $O_2$  operating limit as the average  $O_2$  concentration of the three test runs from a valid test.
- (ii) If an  $O_2$  operating curve will be used:
- (A) Conduct a performance test following the methods provided in paragraphs (i)(1), (2), (3) and (5) of this section at a representative condition for each operating range for which different O2 operating limits will be established. Different operating conditions may be defined as different firing rates (e.g., above 50 percent of rated heat capacity and at or below 50 percent of rated heat capacity) and/or, for co-fired process heaters, different fuel mixtures (e.g., primarily gas fired, primarily oil fired, and equally co-fired, i.e., approximately 50 percent of the input heating value is from fuel gas and approximately 50 percent of the input heating value is from fuel oil). Performance tests for different operating ranges may be conducted at different times.
- (B) Each test will consist of three test runs. Calculate the  $NO_X$  concentration for the performance test as the average of the  $NO_X$  concentrations from each of the three test runs. If the  $NO_X$  concentration for the performance test is less than or equal to the numerical value of the applicable  $NO_X$  emissions limit (regardless of averaging time), then the test is considered to be a valid test.
- (C) If an operating curve is developed for different firing rates, conduct at least one test when the process heater is firing at no less than 70 percent of the rated heat capacity and at least one test under turndown conditions (i.e., when the process heater is firing at 50 percent or less of the rated heat capacity). If O2 operating limits are developed for co-fired process heaters based only on overall firing rates (and not by fuel mixtures), conduct at least one of the test runs for each test while the process heater is being supplied by both fuel gas and fuel oil and conduct at least one of the test runs while the process heater is being supplied solely by fuel gas.
- (D) Determine the average  $O_2$  concentration for each test run of a valid test.

- (E) Calculate the  $O_2$  operating limit for each operating range as the average  $O_2$  concentration of the three test runs from a valid test conducted at the representative conditions for that given operating range.
- (F) Identify the firing rates for which the different operating limits apply. If only two operating limits are established based on firing rates, the  $O_2$  operating limits established when the process heater is firing at no less than 70 percent of the rated heat capacity must apply when the process heater is firing above 50 percent of the rated heat capacity and the  $O_2$  operating limits established for turndown conditions must apply when the process heater is firing at 50 percent or less of the rated heat capacity.
- (G) Operating limits associated with each interval will be valid for 2 years or until another operating limit is established for that interval based on a more recent performance test specific for that interval, whichever occurs first. Owners and operators must use the operating limits determined for a given interval based on the most recent performance test conducted for that interval.
- (7) The owner or operator of a process heater complying with a NO<sub>X</sub> limit in terms of lb/MMBtu as provided in 60.102a(g)(2)(i)(B), (g)(2)(ii)(B),(g)(2)(iii)(B) or (g)(2)(iv)(B) or a process heater with a rated heat capacity between 40 and 100 MMBtu/hr that elects to demonstrate continuous compliance with a maximum excess O2 limit, as provided in  $\S60.107a(c)(6)$  or (d)(8), shall determine heat input to the process heater in MMBtu/hr during each performance test run by measuring fuel gas flow rate, fuel oil flow rate (as applicable) and heating value content according to the methods provided in 60.107a(d)(5), (d)(6), and (d)(4) or (d)(7), respectively.
- (8) The owner or operator shall use Equation 8 of this section to adjust pollutant concentrations to 0-percent  $O_2$  or 0-percent excess air.
- (j) The owner or operator shall determine compliance with the applicable  $H_2S$  emissions limit in  $\S 60.102a(g)(1)$  for a fuel gas combustion device or the concentration requirement in

§60.103a(h) for a flare according to the following test methods and procedures:

- (1) Method 1 of appendix A-1 to part 60 for sample and velocity traverses;
- (2) Method 2 of appendix A-1 to part 60 for velocity and volumetric flow rate:
- (3) Method 3, 3A, or 3B of appendix A-2 to part 60 for gas analysis. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 3B of appendix A-2 to part 60;
- (4) EPA Method 11, 15 or 15A of Appendix A-5 to part 60 or EPA Method 16 of Appendix A-6 to part 60 for determining the H2S concentration for affected facilities using an H2S monitor as specified in §60.107a(a)(2). The method ANSI/ASME PTC 19.10-1981 (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 15A of Appendix A-5 to part 60. The owner or operator may demonstrate compliance based on the mixture used in the fuel gas combustion device or flare or for each individual fuel gas stream used in the fuel gas combustion device or flare.
- (i) For Method 11 of appendix A-5 to part 60, the sampling time and sample volume must be at least 10 minutes and 0.010 dscm (0.35 dscf). Two samples of equal sampling times must be taken at about 1-hour intervals. The arithmetic average of these two samples constitutes a run. For most fuel gases, sampling times exceeding 20 minutes may result in depletion of the collection solution, although fuel gases containing low concentrations of H<sub>2</sub>S may necessitate sampling for longer periods of time.
- (ii) For Method 15 of appendix A-5 to part 60, at least three injects over a 1-hour period constitutes a run.
- (iii) For Method 15A of appendix A-5 to part 60, a 1-hour sample constitutes a run. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 15A of appendix A-5 to part 60.
- (iv) If monitoring is conducted at a single point in a common source of fuel gas as allowed under \$60.107a(a)(2)(iv), only one performance test is required.

That is, performance tests are not required when a new affected fuel gas combustion device or flare is added to a common source of fuel gas that previously demonstrated compliance.

[73 FR 35867, June 24, 2008, as amended at 77 FR 56470, Sep. 12, 2012]

# § 60.105a Monitoring of emissions and operations for fluid catalytic cracking units (FCCU) and fluid coking units (FCU).

- (a) FCCU and FCU subject to PM emissions limit. Each owner or operator subject to the provisions of this subpart shall monitor each FCCU and FCU subject to the PM emissions limit in \$60.102a(b)(1) according to the requirements in paragraph (b), (c), (d), or (e) of this section.
- (b) Control device operating parameters. Each owner or operator of a FCCU or FCU subject to the PM per coke burn-off emissions limit in \$60.102a(b)(1) that uses a control device other than fabric filter or cyclone shall comply with the requirements in paragraphs (b)(1) and (2) of this section.
- (1) The owner or operator shall install, operate and maintain continuous parameter monitor systems (CPMS) to measure and record operating parameters for each control device according to the applicable requirements in paragraphs (b)(1)(i) through (v) of this section.
- (i) For units controlled using an electrostatic precipitator, the owner or operator shall use CPMS to measure and record the hourly average total power input and secondary voltage to the entire system.
- (ii) For units controlled using a wet scrubber, the owner or operator shall use CPMS to measure and record the hourly average pressure drop, liquid feed rate, and exhaust gas flow rate. As an alternative to a CPMS, the owner or operator must comply with the requirements in either paragraph (b)(1)(ii)(A) or (B) of this section.
- (A) As an alternative to pressure drop, the owner or operator of a jet ejector type wet scrubber or other type of wet scrubber equipped with atomizing spray nozzles must conduct a daily check of the air or water pressure to the spray nozzles and record the results of each check.

- (B) As an alternative to exhaust gas flow rate, the owner or operator shall comply with the approved alternative for monitoring exhaust gas flow rate in 40 CFR 63.1573(a) of the National Emission Standards for Hazardous Air Pollutants for Petroleum Refineries: Catalytic Cracking Units, Catalytic Reforming Units, and Sulfur Recovery Units.
- (iii) The owner or operator shall install, operate, and maintain each CPMS according to the manufacturer's specifications and requirements.
- (iv) The owner or operator shall determine and record the average coke burn-off rate and hours of operation for each FCCU or FCU using the procedures in §60.104a(d)(4)(iii).
- (v) If you use a control device other than an electrostatic precipitator, wet scrubber, fabric filter, or cyclone, you may request approval to monitor parameters other than those required in paragraph (b)(1) of this section by submitting an alternative monitoring plan to the Administrator. The request must include the information in paragraphs (b)(1)(v)(A) through (E) of this section.
- (A) A description of each affected facility and the parameter(s) to be monitored to determine whether the affected facility will continuously comply with the emission limitations and an explanation of the criteria used to select the parameter(s).
- (B) A description of the methods and procedures that will be used to demonstrate that the parameter(s) can be used to determine whether the affected facility will continuously comply with the emission limitations and the schedule for this demonstration. The owner or operator must certify that an operating limit will be established for the monitored parameter(s) that represents the conditions in existence when the control device is being properly operated and maintained to meet the emission limitation.
- (C) The frequency and content of the recordkeeping, recording, and reporting, if monitoring and recording are not continuous. The owner or operator also must include the rationale for the proposed monitoring, recording, and reporting requirements.
  - (D) Supporting calculations.

- (E) Averaging time for the alternative operating parameter.
- (2) For use in determining the coke burn-off rate for an FCCU or FCU, the owner or operator shall install, operate, calibrate, and maintain an instrument for continuously monitoring the concentrations of  $CO_2$ ,  $O_2$  (dry basis), and if needed, CO in the exhaust gases prior to any control or energy recovery system that burns auxiliary fuels.
- (i) The owner or operator shall install, operate and maintain each monitor according to Performance Specifications 3 and 4 of Appendix B to part so
- (ii) The owner or operator shall conduct performance evaluations of each  $\mathrm{CO}_2$ ,  $\mathrm{O}_2$  and  $\mathrm{CO}$  monitor according to the requirements in §60.13(c) and Performance Specifications 3 and 4 of Appendix B to part 60. The owner or operator shall use EPA Method 3 of Appendix A-3 to part 60 and EPA Method 10, 10A or 10B of Appendix A-4 to part 60 for conducting the relative accuracy evaluations.
- (iii) The owner or operator shall comply with the quality assurance requirements of procedure 1 of appendix F to part 60, including quarterly accuracy determinations for  $CO_2$  and CO monitors, annual accuracy determinations for  $O_2$  monitors, and daily calibration drift tests.
- (c) Bag leak detection systems. Each owner or operator shall install, operate, and maintain a bag leak detection system for each baghouse or similar fabric filter control device that is used to comply with the PM per coke burnoff emissions limit in §60.102a(b)(1) for an FCCU or FCU according to paragraph (c)(1) of this section; prepare and operate by a site-specific monitoring plan according to paragraph (c)(2) of this section; take action according to paragraph (c)(3) of this section; and record information according to paragraph (c)(4) of this section.
- (1) Each bag leak detection system must meet the specifications and requirements in paragraphs (c)(1)(i) through (viii) of this section.
- (i) The bag leak detection system must be certified by the manufacturer to be capable of detecting PM emissions at concentrations of 0.00044 grains per actual cubic foot or less.

### §60.105a

- (ii) The bag leak detection system sensor must provide output of relative PM loadings. The owner or operator shall continuously record the output from the bag leak detection system using electronic or other means (e.g., using a strip chart recorder or a data logger).
- (iii) The bag leak detection system must be equipped with an alarm system that will sound when the system detects an increase in relative particulate loading over the alarm set point established according to paragraph (c)(1)(iv) of this section, and the alarm must be located such that it can be heard by the appropriate plant personnel.
- (iv) In the initial adjustment of the bag leak detection system, the owner or operator must establish, at a minimum, the baseline output by adjusting the sensitivity (range) and the averaging period of the device, the alarm set points, and the alarm delay time.
- (v) Following initial adjustment, the owner or operator shall not adjust the averaging period, alarm set point, or alarm delay time without approval from the Administrator or delegated authority except as provided in paragraph (c)(1)(vi) of this section.
- (vi) Once per quarter, the owner or operator may adjust the sensitivity of the bag leak detection system to account for seasonal effects, including temperature and humidity, according to the procedures identified in the site-specific monitoring plan required by paragraph (c)(2) of this section.
- (vii) The owner or operator shall install the bag leak detection sensor downstream of the baghouse and upstream of any wet scrubber.
- (viii) Where multiple detectors are required, the system's instrumentation and alarm may be shared among detectors.
- (2) The owner or operator shall develop and submit to the Administrator for approval a site-specific monitoring plan for each baghouse and bag leak detection system. The owner or operator shall operate and maintain each baghouse and bag leak detection system according to the site-specific monitoring plan at all times. Each monitoring plan must describe the items in

- paragraphs (c)(2)(i) through (vii) of this section.
- (i) Installation of the bag leak detection system;
- (ii) Initial and periodic adjustment of the bag leak detection system, including how the alarm set-point will be established:
- (iii) Operation of the bag leak detection system, including quality assurance procedures:
- (iv) How the bag leak detection system will be maintained, including a routine maintenance schedule and spare parts inventory list;
- (v) How the bag leak detection system output will be recorded and stored;
- (vi) Procedures as specified in paragraph (c)(3) of this section. In approving the site-specific monitoring plan, the Administrator or delegated authority may allow owners and operators more than 3 hours to alleviate a specific condition that causes an alarm if the owner or operator identifies in the monitoring plan this specific condition as one that could lead to an alarm, adequately explains why it is not feasible to alleviate this condition within 3 hours of the time the alarm occurs, and demonstrates that the requested time will ensure alleviation of this condition as expeditiously as practicable; and
- (vii) How the baghouse system will be operated and maintained, including monitoring of pressure drop across baghouse cells and frequency of visual inspections of the baghouse interior and baghouse components such as fans and dust removal and bag cleaning mechanisms.
- (3) For each bag leak detection system, the owner or operator shall initiate procedures to determine the cause of every alarm within 1 hour of the alarm. Except as provided in paragraph (c)(2)(vi) of this section, the owner or operator shall alleviate the cause of the alarm within 3 hours of the alarm by taking whatever action(s) are necessary. Actions may include, but are not limited to the following:
- (i) Inspecting the baghouse for air leaks, torn or broken bags or filter media, or any other condition that may cause an increase in particulate emissions;

- (ii) Sealing off defective bags or filter media:
- (iii) Replacing defective bags or filter media or otherwise repairing the control device:
- (iv) Sealing off a defective baghouse compartment;
- (v) Cleaning the bag leak detection system probe or otherwise repairing the bag leak detection system, or
- (vi) Shutting down the process producing the particulate emissions.
- (4) The owner or operator shall maintain records of the information specified in paragraphs (c)(4)(i) through (iii) of this section for each bag leak detection system.
- (i) Records of the bag leak detection system output;
- (ii) Records of bag leak detection system adjustments, including the date and time of the adjustment, the initial bag leak detection system settings, and the final bag leak detection system settings; and
- (iii) The date and time of all bag leak detection system alarms, the time that procedures to determine the cause of the alarm were initiated, the cause of the alarm, an explanation of the actions taken, the date and time the cause of the alarm was alleviated, and whether the alarm was alleviated within 3 hours of the alarm.
- (d) Continuous emissions monitoring systems (CEMS). An owner or operator subject to the PM concentration emission limit (in gr/dsef) in §60.102a(b)(1) for an FCCU or FCU shall install, operate, calibrate, and maintain an instrument for continuously monitoring and recording the concentration (0 percent excess air) of PM in the exhaust gases prior to release to the atmosphere. The monitor must include an O<sub>2</sub> monitor for correcting the data for excess air.
- (1) The owner or operator shall install, operate, and maintain each PM monitor according to Performance Specification 11 of appendix B to part 60. The span value of this PM monitor is 0.08 gr/dscf PM.
- (2) The owner or operator shall conduct performance evaluations of each PM monitor according to the requirements in \$60.13(c) and Performance Specification 11 of appendix B to part 60. The owner or operator shall use EPA Methods 5 or 5I of appendix A-3 to

- part 60 or Method 17 of appendix A-6 to part 60 for conducting the relative accuracy evaluations.
- (3) The owner or operator shall install, operate, and maintain each  $O_2$  monitor according to Performance Specification 3 of appendix B to part 60. The span value of this  $O_2$  monitor must be selected between 10 and 25 percent, inclusive.
- (4) The owner or operator shall conduct performance evaluations of each O<sub>2</sub> monitor according to the requirements in \$60.13(c) and Performance Specification 3 of appendix B to part 60. Method 3, 3A, or 3B of appendix A-2 to part 60 shall be used for conducting the relative accuracy evaluations. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see \$60.17) is an acceptable alternative to EPA Method 3B of appendix A-2 to part 60.
- (5) The owner or operator shall comply with the quality assurance requirements of Procedure 2 of appendix B to part 60 for each PM CEMS and Procedure 1 of appendix F to part 60 for each  $O_2$  monitor, including quarterly accuracy determinations for each PM monitor, annual accuracy determinations for each  $O_2$  monitor, and daily calibration drift tests.
- (e) Alternative monitoring option for FCCU and FCU—COMS. Each owner or operator of an FCCU or FCU that uses cyclones to comply with the PM emission limit in \$60.102a(b)(1) shall monitor the opacity of emissions according to the requirements in paragraphs (e)(1) through (3) of this section.
- (1) The owner or operator shall install, operate, and maintain an instrument for continuously monitoring and recording the opacity of emissions from the FCCU or the FCU exhaust vent.
- (2) The owner or operator shall install, operate, and maintain each COMS according to Performance Specification 1 of appendix B to part 60. The instrument shall be spanned at 20 to 60 percent opacity.
- (3) The owner or operator shall conduct performance evaluations of each COMS according to §60.13(c) and Performance Specification 1 of appendix B to part 60.

### § 60, 105a

- (f) FCCU and FCU subject to  $NO_X$  limit. Each owner or operator subject to the  $NO_X$  emissions limit in §60.102a(b)(2) for an FCCU or FCU shall install, operate, calibrate, and maintain an instrument for continuously monitoring and recording the concentration by volume (dry basis, 0 percent excess air) of  $NO_X$  emissions into the atmosphere. The monitor must include an  $O_2$  monitor for correcting the data for excess air.
- (1) The owner or operator shall install, operate, and maintain each  $NO_X$  monitor according to Performance Specification 2 of appendix B to part 60. The span value of this  $NO_X$  monitor is 200 ppmv  $NO_X$ .
- (2) The owner or operator shall conduct performance evaluations of each NO<sub>X</sub> monitor according to the requirements in §60.13(c) and Performance Specification 2 of appendix B to part 60. The owner or operator shall use Methods 7, 7A, 7C, 7D, or 7E of appendix A-4 to part 60 for conducting the relative accuracy evaluations. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 7 or 7C of appendix A-4 to part 60.
- (3) The owner or operator shall install, operate, and maintain each  $O_2$  monitor according to Performance Specification 3 of appendix B to part 60. The span value of this  $O_2$  monitor must be selected between 10 and 25 percent, inclusive.
- (4) The owner or operator shall conduct performance evaluations of each O<sub>2</sub> monitor according to the requirements in §60.13(c) and Performance Specification 3 of appendix B to part 60. Method 3, 3A, or 3B of appendix A-2 to part 60 shall be used for conducting the relative accuracy evaluations. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 3B of appendix A-2 to part 60.
- (5) The owner or operator shall comply with the quality assurance requirements of Procedure 1 of appendix F to part 60 for each  $NO_X$  and  $O_2$  monitor, including quarterly accuracy determinations for  $NO_X$  monitors, annual accuracy determinations for  $O_2$  monitors, and daily calibration drift tests.

- (g) FCCU and FCU subject to  $SO_2$  limit. The owner or operator subject to the  $SO_2$  emissions limit in §60.102a(b)(3) for an FCCU or an FCU shall install, operate, calibrate, and maintain an instrument for continuously monitoring and recording the concentration by volume (dry basis, corrected to 0 percent excess air) of  $SO_2$  emissions into the atmosphere. The monitor shall include an  $O_2$  monitor for correcting the data for excess air.
- (1) The owner or operator shall install, operate, and maintain each  $SO_2$  monitor according to Performance Specification 2 of appendix B to part 60. The span value of this  $SO_2$  monitor is 200 ppmv  $SO_2$ .
- (2) The owner or operator shall conduct performance evaluations of each SO<sub>2</sub> monitor according to the requirements in §60.13(c) and Performance Specification 2 of appendix B to part 60. The owner or operator shall use Methods 6, 6A, or 6C of appendix A-4 to part 60 for conducting the relative accuracy evaluations. The method ANSI / ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 6 or 6A of appendix A-4 to part 60.
- (3) The owner or operator shall install, operate, and maintain each  $O_2$  monitor according to Performance Specification 3 of appendix B to part 60. The span value of this  $O_2$  monitor must be selected between 10 and 25 percent, inclusive.
- (4) The owner or operator shall conduct performance evaluations of each O<sub>2</sub> monitor according to the requirements in §60.13(c) and Performance Specification 3 of appendix B to part 60. Method 3, 3A, or 3B of appendix A-2 to part 60 shall be used for conducting the relative accuracy evaluations. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 3B of appendix A-2 to part 60.
- (5) The owner or operator shall comply with the quality assurance requirements in Procedure 1 of appendix F to

part 60 for each  $SO_2$  and  $O_2$  monitor, including quarterly accuracy determinations for  $SO_2$  monitors, annual accuracy determinations for  $O_2$  monitors, and daily calibration drift tests.

- (h) FCCU and fluid coking units subject to CO emissions limit. Except as specified in paragraph (h)(3) of this section, the owner or operator shall install, operate, calibrate, and maintain an instrument for continuously monitoring and recording the concentration by volume (dry basis) of CO emissions into the atmosphere from each FCCU and FCU subject to the CO emissions limit in §60.102a(b)(4).
- (1) The owner or operator shall install, operate, and maintain each CO monitor according to Performance Specification 4 or 4A of appendix B to part 60. The span value for this instrument is 1,000 ppm CO.
- (2) The owner or operator shall conduct performance evaluations of each CO monitor according to the requirements in §60.13(c) and Performance Specification 4 or 4A of appendix B to part 60. The owner or operator shall use Methods 10, 10A, or 10B of appendix A-4 to part 60 for conducting the relative accuracy evaluations.
- (3) A CO CEMS need not be installed if the owner or operator demonstrates that all hourly average CO emissions are and will remain less than 50 ppmv (dry basis) corrected to 0 percent excess air. The Administrator may revoke this exemption from monitoring upon a determination that CO emissions on an hourly average basis have exceeded 50 ppmv (dry basis) corrected to 0 percent excess air, in which case a CO CEMS shall be installed within 180 days.
- (1) The demonstration shall consist of continuously monitoring CO emissions for 30 days using an instrument that meets the requirements of Performance Specification 4 or 4A of appendix B to part 60. The span value shall be 100 ppm CO instead of 1,000 ppm, and the relative accuracy limit shall be 10 percent of the average CO emissions or 5 ppm CO, whichever is greater. For instruments that are identical to Method 10 of appendix A-4 to part 60 and employ the sample conditioning system of Method 10A of appendix A-4 to part 60, the alternative relative accuracy test

procedure in section 10.1 of Performance Specification 2 of appendix B to part 60 may be used in place of the relative accuracy test.

- (ii) The owner or operator must submit the following information to the Administrator:
- (A) The measurement data specified in paragraph (h)(3)(i) of this section along with all other operating data known to affect CO emissions; and
- (B) Descriptions of the CPMS for exhaust gas temperature and  $\rm O_2$  monitor required in paragraph (h)(4) of this section and operating limits for those parameters to ensure combustion conditions remain similar to those that exist during the demonstration period.
- (iii) The effective date of the exemption from installation and operation of a CO CEMS is the date of submission of the information and data required in paragraph (h)(3)(ii) of this section.
- (4) The owner or operator of a FCCU or FCU that is exempted from the requirement to install and operate a CO CEMS in paragraph (h)(3) of this section shall install, operate, calibrate, and maintain CPMS to measure and record the operating parameters in paragraph (h)(4)(i) or (ii) of this section. The owner or operator shall install, operate, and maintain each CPMS according to the manufacturer's specifications.
- (i) For a FCCU or FCU with no postcombustion control device, the temperature and O<sub>2</sub> concentration of the exhaust gas stream exiting the unit.
- (ii) For a FCCU or FCU with a post-combustion control device, the temperature and  $O_2$  concentration of the exhaust gas stream exiting the control device.
- (i) Excess emissions. For the purpose of reports required by \$60.7(c), periods of excess emissions for a FCCU or FCU subject to the emissions limitations in \$60.102a(b) are defined as specified in paragraphs (i)(1) through (6) of this section. Note: Determine all averages, except for opacity, as the arithmetic average of the applicable 1-hour averages, e.g., determine the rolling 3-hour average as the arithmetic average of three contiguous 1-hour averages.
- (1) If a CPMS is used according to §60.105a(b)(1), all 3-hour periods during which the average PM control device

operating characteristics, as measured by the continuous monitoring systems under §60.105a(b)(1), fall below the levels established during the performance test.

- (2) If a PM CEMS is used according to \$60.105a(d), all 7-day periods during which the average PM emission rate, as measured by the continuous PM monitoring system under \$60.105a(d) exceeds 0.040 gr/dscf corrected to 0 percent excess air for a modified or reconstructed FCCU, 0.020 gr/dscf corrected to 0 percent excess air for a newly constructed FCCU, or 0.040 gr/dscf for an affected fluid coking unit.
- (3) If a COMS is used according to §60.105a(e), all 3-hour periods during which the average opacity, as measured by the COMS under §60.105a(e), exceeds the site-specific limit established during the most recent performance test.
- (4) All rolling 7-day periods during which the average concentration of  $NO_X$  as measured by the  $NO_X$  CEMS under \$60.105a(f) exceeds 80 ppmv for an affected FCCU or FCU.
- (5) All rolling 7-day periods during which the average concentration of  $SO_2$  as measured by the  $SO_2$  CEMS under  $\S 60.105a(g)$  exceeds 50 ppmv, and all rolling 365-day periods during which the average concentration of  $SO_2$  as measured by the  $SO_2$  CEMS exceeds 25 ppmv.
- (6) All 1-hour periods during which the average CO concentration as measured by the CO continuous monitoring system under §1A60.105a(h) exceeds 500 ppmv or, if applicable, all 1-hour periods during which the average temperature and O<sub>2</sub> concentration as measured by the continuous monitoring systems under §60.105a(h)(4) fall below the operating limits established during the performance test.

[73 FR 35867, June 24, 2008, as amended at 77 FR 56473, Sep. 12, 2012]

## § 60.106a Monitoring of emissions and operations for sulfur recovery plants.

- (a) The owner or operator of a sulfur recovery plant that is subject to the emissions limits in 60.102a(f)(1) or 60.102a(f)(2) shall:
- (1) For sulfur recovery plants subject to the  $SO_2$  emission limit in

 $\S60.102a(f)(1)(i)$  or  $\S60.102a(f)(2)(i)$ , the owner or operator shall install, operate, calibrate, and maintain an instrument for continuously monitoring and recording the concentration (dry basis, zero percent excess air) of any  $SO_2$  emissions into the atmosphere. The monitor shall include an oxygen monitor for correcting the data for excess air.

- (i) The span values for this monitor are two times the applicable  $SO_2$  emission limit and between 10 and 25 percent  $O_2$ , inclusive.
- (ii) The owner or operator shall install, operate, and maintain each  $SO_2$  CEMS according to Performance Specification 2 of appendix B to part 60.
- (iii) The owner or operator shall conduct performance evaluations of each SO<sub>2</sub> monitor according to the requirements in \$60.13(c) and Performance Specification 2 of appendix B to part 60. The owner or operator shall use Methods 6 or 6C of appendix A-4 to part 60 and Method 3 or 3A of appendix A-2 of part 60 for conducting the relative accuracy evaluations. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see \$60.17) is an acceptable alternative to EPA Method 6.
- (2) For sulfur recovery plants that are subject to the reduced sulfur compound and  $H_2S$  emission limit in  $\S 60.102a(f)(1)(ii)$  or  $\S 60.102a(f)(2)(ii)$ , the owner or operator shall install, operate, calibrate, and maintain an instrument for continuously monitoring and recording the concentration of reduced sulfur,  $H_2S$ , and  $O_2$  emissions into the atmosphere. The reduced sulfur emissions shall be calculated as  $SO_2$  (dry basis, zero percent excess air).
- (i) The span values for this monitor are two times the applicable reduced sulfur emission limit, two times the  $H_2S$  emission limit, and between 10 and 25 percent  $O_2$ , inclusive.
- (ii) The owner or operator shall install, operate, and maintain each reduced sulfur CEMS according to Performance Specification 5 of appendix B to part 60.
- (iii) The owner or operator shall conduct performance evaluations of each reduced sulfur monitor according to the requirements in §60.13(c) and Performance Specification 5 of appendix B

to part 60. The owner or operator shall use Methods 15 or 15A of appendix A-5 to part 60 for conducting the relative accuracy evaluations. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 15A of appendix A-5 to part 60.

- (iv) The owner or operator shall install, operate, and maintain each H<sub>2</sub>S CEMS according to Performance Specification 7 of appendix B to part 60.
- (v) The owner or operator shall conduct performance evaluations of each reduced sulfur monitor according to the requirements in §60.13(c) and Performance Specification 5 of appendix B to part 60. The owner or operator shall use Methods 11, 15, or 15A of appendix A-5 to part 60 or Method 16 of appendix A-6 to part 60 for conducting the relative accuracy evaluations. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 15A of appendix A-5 to part 60.
- (vi) The owner or operator shall install, operate, and maintain each  $O_2$  monitor according to Performance Specification 3 of appendix B to part 60.
- (vii) The span value for the  $\overline{O}_2$  monitor must be selected between 10 and 25 percent, inclusive.
- (viii) The owner or operator shall conduct performance evaluations for the O<sub>2</sub> monitor according to the requirements of §60.13(c) and Performance Specification 3 of appendix B to part 60. The owner or operator shall use Methods 3, 3A, or 3B of appendix A-2 to part 60 for conducting the relative accuracy evaluations. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 3B of appendix A-2 to part 60.
- (ix) The owner or operator shall comply with the applicable quality assurance procedures of appendix F to part 60 for each monitor, including annual accuracy determinations for each  $O_2$  monitor, and daily calibration drift determinations.
- (3) In place of the reduced sulfur monitor required in paragraph (a)(2) of this section, the owner or operator

shall install, calibrate, operate, and maintain an instrument using an air or  $O_2$  dilution and oxidation system to convert any reduced sulfur to  $SO_2$  for continuously monitoring and recording the concentration (dry basis, 0 percent excess air) of the total resultant  $SO_2$ . The monitor must include an  $O_2$  monitor for correcting the data for excess  $O_2$ .

- (i) The span value for this monitor is two times the applicable  $SO_2$  emission limit.
- (ii) The owner or operator shall conduct performance evaluations of each  $SO_2$  monitor according to the requirements in §60.13(c) and Performance Specification 5 of appendix B to part 60. The owner or operator shall use Methods 15 or 15A of appendix A-5 to part 60 for conducting the relative accuracy evaluations. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 15A of appendix A-5 to part 60.
- (iii) The owner or operator shall install, operate, and maintain each  $O_2$  monitor according to Performance Specification 3 of appendix B to part 60.
- (iv) The span value for the  $O_2$  monitor must be selected between 10 and 25 percent, inclusive.
- (v) The owner or operator shall conduct performance evaluations for the  $O_2$  monitor according to the requirements of  $\S 60.13(c)$  and Performance Specification 3 of appendix B to part 60. The owner or operator shall use Methods 3, 3A, or 3B of appendix A-2 to part 60 for conducting the relative accuracy evaluations. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see  $\S 60.17$ ) is an acceptable alternative to EPA Method 3B of appendix A-2 to part 60.
- (vi) The owner or operator shall comply with the applicable quality assurance procedures of appendix F to part 60 for each monitor, including quarterly accuracy determinations for each  $SO_2$  monitor, annual accuracy determinations for each  $O_2$  monitor, and daily calibration drift determinations.
- (b) Excess emissions. For the purpose of reports required by §60.7(c), periods of excess emissions for sulfur recovery

plants subject to the emissions limitations in §60.102a(f) are defined as specified in paragraphs (b)(1) through (3) of this section. Note: Determine all averages as the arithmetic average of the applicable 1-hour averages, e.g., determine the rolling 12-hour average as the arithmetic average of 12 contiguous 1-hour averages.

- (1) All 12-hour periods during which the average concentration of  $SO_2$  as measured by the  $SO_2$  continuous monitoring system required under paragraph (a)(1) of this section exceeds the applicable emission limit (dry basis, zero percent excess air); or
- (2) All 12-hour periods during which the average concentration of reduced sulfur (as  $SO_2$ ) as measured by the reduced sulfur continuous monitoring system required under paragraph (a)(2) of this section exceeds the applicable emission limit; or
- (3) All 12-hour periods during which the average concentration of  $H_2S$  as measured by the  $H_2S$  continuous monitoring system required under paragraph (a)(2) of this section exceeds the applicable emission limit (dry basis, 0 percent excess air).

## § 60.107a Monitoring of emissions and operations for fuel gas combustion devices and flares.

- (a) Fuel gas combustion devices subject to SO2 or H2S limit and flares subject to H<sub>2</sub>S concentration requirements. The owner or operator of a fuel gas combustion device that is subject to §60.102a(g)(1) and elects to comply with emission limits  $SO_2$ §60.102a(g)(1)(i) shall comply with the requirements in paragraph (a)(1) of this section. The owner or operator of a fuel gas combustion device that is subject to §60.102a(g)(1) and elects to comply with the H2S concentration limits in §60.102a(g)(1)(ii) or a flare that is subject to the H2S concentration requirement in §60.103a(h) shall comply with paragraph (a)(2) of this section.
- (1) The owner or operator of a fuel gas combustion device that elects to comply with the SO<sub>2</sub> emissions limits in §60.102a(g)(1)(i) shall install, operate, calibrate and maintain an instrument for continuously monitoring and recording the concentration (dry basis, 0-percent excess air) of SO<sub>2</sub> emissions

into the atmosphere. The monitor must include an  $O_2$  monitor for correcting the data for excess air.

- (i) The owner or operator shall install, operate, and maintain each  $SO_2$  monitor according to Performance Specification 2 of appendix B to part 60. The span value for the  $SO_2$  monitor is 50 ppm  $SO_2$ .
- (ii) The owner or operator shall conduct performance evaluations for the SO<sub>2</sub> monitor according to the requirements of §60.13(c) and Performance Specification 2 of appendix B to part 60. The owner or operator shall use Methods 6, 6A, or 6C of appendix A-4 to part 60 for conducting the relative accuracy evaluations. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by erence-see §60.17) is an acceptable alternative to EPA Method 6 or 6A of appendix A-4 to part 60. Samples taken by Method 6 of appendix A-4 to part 60 shall be taken at a flow rate of approximately 2 liters/min for at least 30 minutes. The relative accuracy limit shall be 20 percent or 4 ppm, whichever is greater, and the calibration drift limit shall be 5 percent of the established span value.
- (iii) The owner or operator shall install, operate, and maintain each  $O_2$  monitor according to Performance Specification 3 of appendix B to part 60. The span value for the  $O_2$  monitor must be selected between 10 and 25 percent, inclusive.
- (iv) The owner or operator shall conduct performance evaluations for the  $O_2$  monitor according to the requirements of  $\S 60.13(c)$  and Performance Specification 3 of appendix B to part 60. The owner or operator shall use Methods 3, 3A, or 3B of appendix A-2 to part 60 for conducting the relative accuracy evaluations. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see  $\S 60.17$ ) is an acceptable alternative to EPA Method 3B of appendix A-2 to part 60.
- (v) The owner or operator shall comply with the applicable quality assurance procedures in appendix F to part 60, including quarterly accuracy determinations for  $SO_2$  monitors, annual accuracy determinations for  $O_2$  monitors, and daily calibration drift tests.

- (vi) Fuel gas combustion devices having a common source of fuel gas may be monitored at only one location (i.e., after one of the combustion devices), if monitoring at this location accurately represents the  $SO_2$  emissions into the atmosphere from each of the combustion devices.
- (2) The owner or operator of a fuel gas combustion device that elects to comply with the  $\rm H_2S$  concentration limits in  $\S 60.102a(g)(1)(ii)$  or a flare that is subject to the  $\rm H_2S$  concentration requirement in  $\S 60.103a(h)$  shall install, operate, calibrate and maintain an instrument for continuously monitoring and recording the concentration by volume (dry basis) of  $\rm H_2S$  in the fuel gases before being burned in any fuel gas combustion device or flare.
- (i) The owner or operator shall install, operate and maintain each H<sub>2</sub>S monitor according to Performance Specification 7 of Appendix B to part 60. The span value for this instrument is 300 ppmy H<sub>2</sub>S.
- (ii) The owner or operator shall conduct performance evaluations for each  $\rm H_2S$  monitor according to the requirements of §60.13(c) and Performance Specification 7 of appendix B to part 60. The owner or operator shall use Method 11, 15, or 15A of appendix A-5 to part 60 or Method 16 of appendix A-6 to part 60 for conducting the relative accuracy evaluations. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 15A of appendix A-5 to part 60.
- (iii) The owner or operator shall comply with the applicable quality assurance procedures in appendix F to part 60 for each H₂S monitor.
- (iv) Fuel gas combustion devices or flares having a common source of fuel gas may be monitored at only one location, if monitoring at this location accurately represents the concentration of  $\rm H_2S$  in the fuel gas being burned in the respective fuel gas combustion devices or flares.
- (v) The owner or operator of a flare subject to \$60.103a(c) through (e) may use the instrument required in paragraph (e)(1) of this section to demonstrate compliance with the  $H_2S$  concentration requirement in \$60.103a(h) if

- the owner or operator complies with the requirements of paragraph (e)(1)(i) through (iv) and if the instrument has a span (or dual span, if necessary) capable of accurately measuring concentrations between 20 and 300 ppmv. If the instrument required in paragraph (e)(1) of this section is used to demonstrate compliance with the  $\rm H_2S$  concentration requirement, the concentration directly measured by the instrument must meet the numeric concentration in §60.103a(h).
- (vi) The owner or operator of modified flare that meets all three criteria in paragraphs (a)(2)(vi)(A) through (C) of this section shall comply with the requirements of paragraphs (a)(2)(i) through (v) of this section no later than November 11, 2015. The owner or operator shall comply with the approved alternative monitoring plan or plans pursuant to §60.13(i) until the flare is in compliance with requirements of paragraphs (a)(2)(i) through (v) of this section.
- (A) The flare was an affected facility subject to subpart J of this part prior to becoming an affected facility under \$60.100a.
- (B) The owner or operator had an approved alternative monitoring plan or plans pursuant to §60.13(i) for all fuel gases combusted in the flare.
- (C) The flare did not have in place on or before September 12, 2012 an instrument for continuously monitoring and recording the concentration by volume (dry basis) of  $H_2S$  in the fuel gases that is capable of complying with the requirements of paragraphs (a)(2)(i) through (v) of this section.
- (3) The owner or operator of a fuel gas combustion device or flare is not required to comply with paragraph (a)(1) or (2) of this section for fuel gas streams that are exempt under §§60.102a(g)(1)(iii) or 60.103a(h) or, for fuel gas streams combusted in a process heater, other fuel gas combustion device or flare that are inherently low in sulfur content. Fuel gas streams meeting one of the requirements in paragraphs (a)(3)(i) through (iv) of this section will be considered inherently low in sulfur content.
  - (i) Pilot gas for heaters and flares.

### § 60.107a

- (ii) Fuel gas streams that meet a commercial-grade product specification for sulfur content of 30 ppmv or less. In the case of a liquefied petroleum gas (LPG) product specification in the pressurized liquid state, the gas phase sulfur content should be evaluated assuming complete vaporization of the LPG and sulfur containing-compounds at the product specification concentration.
- (iii) Fuel gas streams produced in process units that are intolerant to sulfur contamination, such as fuel gas streams produced in the hydrogen plant, catalytic reforming unit, isomerization unit, and HF alkylation process units.
- (iv) Other fuel gas streams that an owner or operator demonstrates are low-sulfur according to the procedures in paragraph (b) of this section.
- (4) If the composition of an exempt fuel gas stream changes, the owner or operator must follow the procedures in paragraph (b)(3) of this section.
- (b) Exemption from H<sub>2</sub>S monitoring requirements for low-sulfur fuel gas streams. The owner or operator of a fuel gas combustion device or flare may apply for an exemption from the H<sub>2</sub>S monitoring requirements in paragraph (a)(2) of this section for a fuel gas stream that is inherently low in sulfur content. A fuel gas stream that is demonstrated to be low-sulfur is exempt from the monitoring requirements of paragraphs (a)(1) and (2) of this section until there are changes in operating conditions or stream composition.
- (1) The owner or operator shall submit to the Administrator a written application for an exemption from monitoring. The application must contain the following information:
- (i) A description of the fuel gas stream/system to be considered, including submission of a portion of the appropriate piping diagrams indicating the boundaries of the fuel gas stream/system and the affected fuel gas combustion device(s) or flare(s) to be considered:
- (ii) A statement that there are no crossover or entry points for sour gas (high H<sub>2</sub>S content) to be introduced into the fuel gas stream/system (this should be shown in the piping diagrams):

- (iii) An explanation of the conditions that ensure low amounts of sulfur in the fuel gas stream (i.e., control equipment or product specifications) at all times;
- (iv) The supporting test results from sampling the requested fuel gas stream/ system demonstrating that the sulfur content is less than 5 ppm H2S. Sampling data must include, at minimum, 2 weeks of daily monitoring (14 grab samples) for frequently operated fuel gas streams/systems; for infrequently operated fuel gas streams/systems, seven grab samples must be collected unless other additional information would support reduced sampling. The owner or operator shall use detector tubes ("length-of-stain tube" type measurement) following the "Gas Processors Association Standard 2377-86, Test for Hydrogen Sulfide and Carbon Dioxide in Natural Gas Using Length of Stain Tubes," 1986 Revision (incorporated by reference—see §60.17), with ranges 0-10/0-100 ppm (N = 10/1) to test the applicant fuel gas stream for H<sub>2</sub>S:
- (v) A description of how the 2 weeks (or seven samples for infrequently operated fuel gas streams/systems) of monitoring results compares to the typical range of H<sub>2</sub>S concentration (fuel quality) expected for the fuel gas stream/system going to the affected fuel gas combustion device or flare (e.g., the 2 weeks of daily detector tube results for a frequently operated loading rack included the entire range of products loaded out and, therefore, should be representative of typical operating conditions affecting H2S content in the fuel gas stream going to the loading rack flare).
- (2) The effective date of the exemption is the date of submission of the information required in paragraph (b)(1) of this section.
- (3) No further action is required unless refinery operating conditions change in such a way that affects the exempt fuel gas stream/system (e.g., the stream composition changes). If such a change occurs, the owner or operator shall follow the procedures in paragraph (b)(3)(i), (b)(3)(ii), or (b)(3)(iii) of this section.
- (i) If the operation change results in a sulfur content that is still within the

range of concentrations included in the original application, the owner or operator shall conduct an H<sub>2</sub>S test on a grab sample and record the results as proof that the concentration is still within the range.

(ii) If the operation change results in a sulfur content that is outside the range of concentrations included in the original application, the owner or operator may submit new information following the procedures of paragraph (b)(1) of this section within 60 days (or within 30 days after the seventh grab sample is tested for infrequently operated process units).

(iii) If the operation change results in a sulfur content that is outside the range of concentrations included in the original application and the owner or operator chooses not to submit new information to support an exemption, the owner or operator must begin H2S monitoring using daily stain sampling to demonstrate compliance. The owner or operator must begin monitoring according to the requirements in paragraphs (a)(1) or (a)(2) of this section as soon as practicable, but in no case later than 180 days after the operation change. During daily stain tube sampling, a daily sample exceeding 162 ppmv is an exceedance of the 3-hour H<sub>2</sub>S concentration limit. The owner or operator of a fuel gas combustion device must also determine a rolling 365day average using the stain sampling results; an average H2S concentration of 5 ppmv must be used for days within the rolling 365-day period prior to the operation change.

(c) Process heaters complying with the NOx concentration-based limit. The owner or operator of a process heater subject to the NO<sub>X</sub> emissions limit in §60.102a(g)(2) and electing to comply with the applicable emissions limit in  $\S60.102a(g)(2)(i)(A),$ (g)(2)(ii)(A),(g)(2)(iii)(A) or (g)(2)(iv)(A) shall install, operate, calibrate and maintain an instrument for continuously monitoring and recording the concentration (dry basis, 0-percent excess air) of NOx emissions into the atmosphere according to the requirements in paragraphs (c)(1) through (5) of this section, except as provided in paragraph (c)(6) of this section. The monitor must include an  $O_2$  monitor for correcting the data for excess air.

- (1) Except as provided in paragraph (c)(6) of this section, the owner or operator shall install, operate and maintain each  $NO_{\rm x}$  monitor according to Performance Specification 2 of Appendix B to part 60. The span value of this  $NO_{\rm x}$  monitor must be between 2 and 3 times the applicable emissions limit, inclusive.
- (2) The owner or operator shall conduct performance evaluations of each  $NO_X$  monitor according to the requirements in §60.13(c) and Performance Specification 2 of appendix B to part 60. The owner or operator shall use Methods 7, 7A, 7C, 7D, or 7E of appendix A-4 to part 60 for conducting the relative accuracy evaluations. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 7 or 7C of appendix A-4 to part 60.
- (3) The owner or operator shall install, operate, and maintain each  $O_2$  monitor according to Performance Specification 3 of appendix B to part 60. The span value of this  $O_2$  monitor must be selected between 10 and 25 percent, inclusive.
- (4) The owner or operator shall conduct performance evaluations of each  $O_2$  monitor according to the requirements in  $\S 60.13(c)$  and Performance Specification 3 of appendix B to part 60. Method 3, 3A, or 3B of appendix A-2 to part 60 shall be used for conducting the relative accuracy evaluations. The method ANSI/ASME PTC 19.10-1981, "Flue and Exhaust Gas Analyses," (incorporated by reference—see  $\S 60.17$ ) is an acceptable alternative to EPA Method 3B of appendix A-2 to part 60.
- (5) The owner or operator shall comply with the quality assurance requirements in Procedure 1 of appendix F to part 60 for each  $NO_X$  and  $O_2$  monitor, including quarterly accuracy determinations for  $NO_X$  monitors, annual accuracy determinations for  $O_2$  monitors, and daily calibration drift tests.

(6) The owner or operator of a process heater that has a rated heating capacity of less than 100 MMBtu and is equipped with combustion modification-based technology to reduce  $NO_X$  emissions (i.e., low- $NO_X$  burners, ultra-

### §60.107a

low-NO<sub>x</sub> burners) may elect to comply with the monitoring requirements in paragraphs (c)(1) through (5) of this section or, alternatively, the owner or operator of such a process heater shall conduct biennial performance tests according to the requirements in §60.104a(i), establish a maximum excess O2 operating limit or operating curve according to the requirements in  $\S60.104a(i)(6)$  and comply with the  $O_2$ monitoring requirements in paragraphs (c)(3) through (5) of this section to demonstrate compliance. If an O2 operating curve is used (i.e., if different O2 operating limits are established for different operating ranges), the owner or operator of the process heater must also monitor fuel gas flow rate, fuel oil flow rate (as applicable) and heating value content according to the methods provided in paragraphs (d)(5), (d)(6), and (d)(4) or (d)(7) of this section, respectively.

(d) Process heaters complying with the NOx heating value-based or mass-based limit The owner or operator of a process heater subject to the NO<sub>X</sub> emissions limit in §60.102a(g)(2) and electing to comply with the applicable emissions §60.102a(g)(2)(i)(B) limit in (g)(2)(ii)(B) shall install, operate, calibrate and maintain an instrument for continuously monitoring and recording the concentration (dry basis, 0-percent excess air) of NOx emissions into the atmosphere and shall determine the F factor of the fuel gas stream no less frequently than once per day according to the monitoring requirements in paragraphs (d)(1) through (4) of this section. The owner or operator of a cofired process heater subject to the NOx emissions limit in §60.102a(g)(2) and electing to comply with the heating limit value-based §60.102a(g)(2)(iii)(B) (g)(2)(iv)(B)orshall install, operate, calibrate and maintain an instrument for continuously monitoring and recording the concentration (dry basis, 0-percent excess air) of NOx emissions into the atmosphere according to the monitoring requirements in paragraph (d)(1) of this

section; install, operate, calibrate and maintain an instrument for continuously monitoring and recording the flow rate of the fuel gas and fuel oil fed to the process heater according to the monitoring requirements in paragraph (d)(5) and (6) of this section; for fuel gas streams, determine gas composition according to the requirements in paragraph (d)(4) of this section or the higher heating value according to the requirements in paragraph (d)(7) of this section; and for fuel oil streams, determine the heating value according to the monitoring requirements in paragraph (d)(7) of this section.

- (1) Except as provided in paragraph (d)(8) of this section, the owner or operator shall install, operate and maintain each  $NO_X$  monitor according to the requirements in paragraphs (c)(1) through (5) of this section. The monitor must include an  $O_2$  monitor for correcting the data for excess air.
- (2) Except as provided in paragraph (d)(3) of this section, the owner or operator shall sample and analyze each fuel stream fed to the process heater using the methods and equations in section 12.3.2 of EPA Method 19 of Appendix A-7 to part 60 to determine the F factor on a dry basis. If a single fuel gas system provides fuel gas to several process heaters, the F factor may be determined at a single location in the fuel gas system provided it is representative of the fuel gas fed to the affected process heater(s).
- (3) As an alternative to the requirements in paragraph (d)(2) of this section, the owner or operator of a gasfired process heater shall install, operate and maintain a gas composition analyzer and determine the average F factor of the fuel gas using the factors in Table 1 of this subpart and Equation 10 of this section. If a single fuel gas system provides fuel gas to several process heaters, the F factor may be determined at a single location in the fuel gas system provided it is representative of the fuel gas fed to the affected process heater(s).

$$F_d = \frac{1,000,000 \times \sum (X_i \times MEV_i)}{\sum (X_i \times MHC_i)}$$
 (Eq. 10)

Where:

 $F_d$  = F factor on dry basis at 0-percent excess air, dscf/MMBtu.

 $X_i$  = mole or volume fraction of each component in the fuel gas.

MEV; = molar exhaust volume, dry standard cubic feet per mole (dscf/mol).

MHC; = molar heat content, Btu per mole (Btu/mol).

1,000,000 = unit conversion, Btu per MMBtu.

- (4) The owner or operator shall conduct performance evaluations of each compositional monitor according to the requirements in Performance Specification 9 of Appendix B to part 60. Any of the following methods shall be used for conducting the relative accuracy evaluations:
- (i) EPA Method 18 of Appendix A-6 to part 60;
- (ii) ASTM D1945-03 (Reapproved 2010)(incorporated by reference-see §60.17);
- (iii) ASTM D1946-90 (Reapproved 2006)(incorporated by reference-see §60.17);
- (iv) ASTM D6420-99 (Reapproved 2004)(incorporated by reference-see \$60.17):
- (v) GPA 2261-00 (incorporated by reference-see § 60.17); or
- (vi) ASTM UOP539-97 (incorporated by reference-see §60.17).
- (5) The owner or operator shall install, operate and maintain fuel gas flow monitors according to the manufacturer's recommendations. For volumetric flow meters, temperature and pressure monitors must be installed in conjunction with the flow meter or in a representative location to correct the measured flow to standard conditions (i.e., 68 °F and 1 atmosphere). For mass flow meters, use gas compositions determined according to paragraph (d)(4) of this section to determine the average molecular weight of the fuel gas and convert the mass flow to a volumetric flow at standard conditions (i.e., 68 °F and 1 atmosphere). The owner or operator shall conduct performance evaluations of each fuel gas flow mon-

itor according to the requirements in §60.13 and Performance Specification 6 of Appendix B to part 60. Any of the following methods shall be used for conducting the relative accuracy evaluations:

- (i) EPA Method 2, 2A, 2B, 2C or 2D of Appendix A-2 to part 60;
- (ii) ASME MFC-3M-2004 (incorporated by reference-see §60.17);
- (iii) ANSI/ASME MFC-4M-1986 (Reaffirmed 2008) (incorporated by reference-see § 60.17);
- (iv) ASME MFC-6M-1998 (Reaffirmed 2005) (incorporated by reference-see \$60.17);
- (v) ASME/ANSI MFC-7M-1987 (Reaffirmed 2006) (incorporated by reference-see § 60.17);
- (vi) ASME MFC-11M-2006 (incorporated by reference-see § 60.17);
- (vii) ASME MFC-14M-2003 (incorporated by reference-see § 60.17);
- (viii) ASME MFC-18M-2001 (incorporated by reference-see § 60.17);
- (ix) AGA Report No. 3, Part 1 (incorporated by reference-see § 60.17);
- (x) AGA Report No. 3, Part 2 (incorporated by reference-see §60.17);
- (xi) AGA Report No. 11 (incorporated by reference-see §60.17);
- (xii) AGA Report No. 7 (incorporated by reference-see §60.17); and
- (xiii) API Manual of Petroleum Measurement Standards, Chapter 22, Section 2 (incorporated by reference-see §60.17).
- (6) The owner or operator shall install, operate and maintain each fuel oil flow monitor according to the manufacturer's recommendations. The owner or operator shall conduct performance evaluations of each fuel oil flow monitor according to the requirements in §60.13 and Performance Specification 6 of Appendix B to part 60. Any of the following methods shall be used for conducting the relative accuracy evaluations:
- (i) Any one of the methods listed in paragraph (d)(5) of this section that are applicable to fuel oil (i.e., "fluids");

### §60.107a

- (ii) ANSI/ASME-MFC-5M-1985 (Reaffirmed 2006) (incorporated by reference-see § 60.17):
- (iii) ASME/ANSI MFC-9M-1988 (Reaffirmed 2006) (incorporated by reference-see § 60.17);
- (iv) ASME MFC-16-2007 (incorporated by reference-see § 60.17);
- (v) ASME MFC-22-2007 (incorporated by reference-see §60.17); or
- (vi) ISO 8316 (incorporated by reference-see §60.17).
- (7) The owner or operator shall determine the higher heating value of each fuel fed to the process heater using any of the applicable methods included in paragraphs (d)(7)(i) through (ix) of this section. If a common fuel supply system provides fuel gas or fuel oil to several process heaters, the higher heating value of the fuel in each fuel supply system may be determined at a single location in the fuel supply system provided it is representative of the fuel fed to the affected process heater(s). The higher heating value of each fuel fed to the process heater must be determined no less frequently than once per day except as provided in paragraph (d)(7)(x) of this section.
- (i) ASTM D240-02 (Reapproved 2007) (incorporated by reference-see § 60.17).
- (ii) ASTM D1826-94 (Reapproved 2003) (incorporated by reference-see § 60.17).
- (iii) ASTM D1945-03 (Reapproved 2010) (incorporated by reference-see §60.17).
- (iv) ASTM D1946-90 (Reapproved 2006) (incorporated by reference-see § 60.17).
- (v) ASTM D3588-98 (Reapproved 2003) (incorporated by reference-see §60.17).
- (vi) ASTM D4809-06 (incorporated by reference-see §60.17).
- (vii) ASTM D4891-89 (Reapproved 2006) (incorporated by reference-see \$60.17).
- (viii) GPA 2172-09 (incorporated by reference-see §60.17).
- (ix) Any of the methods specified in section 2.2.7 of Appendix D to part 75.
- (x) If the fuel oil supplied to the affected co-fired process heater originates from a single storage tank, the owner or operator may elect to use the storage tank sampling method in section 2.2.4.2 of Appendix D to part 75 instead of daily sampling, except that the most recent value for heating content must be used.

- (8) The owner or operator of a process heater that has a rated heating capacity of less than 100 MMBtu and is equipped with combustion modification based technology to reduce NOx emissions (i.e., low-NOx burners or ultralow NO<sub>X</sub> burners) may elect to comply with the monitoring requirements in paragraphs (d)(1) through (7) of this section or, alternatively, the owner or operator of such a process heater shall conduct biennial performance tests acto the requirements in cording §60.104a(i), establish a maximum excess O<sub>2</sub> operating limit or operating curve according to the requirements in  $\S60.104a(i)(6)$  and comply with the  $O_2$ monitoring requirements in paragraphs (c)(3) through (5) of this section to demonstrate compliance. If an O2 operating curve is used (i.e., if different O2 operating limits are established for different operating ranges), the owner or operator of the process heater must also monitor fuel gas flow rate, fuel oil flow rate (as applicable) and heating value content according to the methods provided in paragraphs (d)(5), (d)(6), and (d)(4) or (d)(7) of this section, respectively.
- (e) Sulfur monitoring for assessing root cause analysis threshold for affected flares. Except as described in paragraphs (e)(4) and (h) of this section, the owner or operator of an affected flare subject to §60.103a(c) through (e) shall determine the total reduced sulfur concentration for each gas line directed to the affected flare in accordance with either paragraph (e)(1), (e)(2) or (e)(3) of this section. Different options may be elected for different gas lines. If a monitoring system is in place that is capable of complying with the requirements related to either paragraph (e)(1), (e)(2) or (e)(3) of this section, the owner or operator of a modified flare must comply with the requirements related to either paragraph (e)(1), (e)(2) or (e)(3) of this section upon startup of the modified flare. If a monitoring system is not in place that is capable of complying with the requirements related to either paragraph (e)(1), (e)(2) or (e)(3) of this section, the owner or operator of a modified flare must comply with the requirements related to either paragraph (e)(1), (e)(2) or (e)(3) of this section no later than November 11, 2015 or

upon startup of the modified flare, whichever is later.

- (1) Total reduced sulfur monitoring requirements. The owner or operator shall install, operate, calibrate and maintain an instrument for continuously monitoring and recording the concentration of total reduced sulfur in gas discharged to the flare.
- (i) The owner or operator shall install, operate and maintain each total reduced sulfur monitor according to Performance Specification 5 of Appendix B to part 60. The span value should be determined based on the maximum sulfur content of gas that can be discharged to the flare (e.g., roughly 1.1 to 1.3 times the maximum anticipated sulfur concentration), but may be no less than 5,000 ppmv. A single dual range monitor may be used to comply with the requirements of this paragraph and paragraph (a)(2) of this section provided the applicable span specifications are met.
- (ii) The owner or operator shall conduct performance evaluations of each total reduced sulfur monitor according to the requirements in §60.13(c) and Performance Specification 5 of Appendix B to part 60. For flares that routinely have flow, the owner or operator of each total reduced sulfur monitor shall use EPA Method 15A of Appendix A-5 to part 60 for conducting the relative accuracy evaluations. The method ANSI/ASME PTC 19.10-1981 (incorporated by reference-see §60.17) is an acceptable alternative to EPA Method 15A of Appendix A-5 to part 60. The alternative relative accuracy procedures described in section 16.0 of Performance Specification 2 of Appendix B to part 60 (cylinder gas audits) may be used for conducting the relative accuracy evaluations. For flares that do not receive routine flow, the alternative relative accuracy procedures described in section 16.0 of Performance Specification 2 of Appendix B to part 60 (cylinder gas audits) may be used for conducting the relative accuracy evaluations, except that it is not necessary to include as much of the sampling probe or sampling line as practical.
- (iii) The owner or operator shall comply with the applicable quality assurance procedures in Appendix F to part

- 60 for each total reduced sulfur monitor.
- (2)  $H_2$ S monitoring requirements. The owner or operator shall install, operate, calibrate, and maintain an instrument for continuously monitoring and recording the concentration of  $H_2$ S in gas discharged to the flare according to the requirements in paragraphs (e)(2)(i) through (iii) of this section and shall collect and analyze samples of the gas and calculate total sulfur concentrations as specified in paragraphs (e)(2)(iv) through (ix) of this section.
- (i) The owner or operator shall install, operate and maintain each H2S monitor according to Performance Specification 7 of Appendix B to part 60. The span value should be determined based on the maximum sulfur content of gas that can be discharged to the flare (e.g., roughly 1.1 to 1.3)times the maximum anticipated sulfur concentration), but may be no less than 5,000 ppmv. A single dual range H<sub>2</sub>S monitor may be used to comply with the requirements of this paragraph and paragraph (a)(2) of this section provided the applicable span specifications are met.
- (ii) The owner or operator shall conduct performance evaluations of each H<sub>2</sub>S monitor according to the requirements in §60.13(c) and Performance Specification 7 of Appendix B to part 60. For flares that routinely have flow, the owner or operator shall use EPA Method 11, 15 or 15A of Appendix A-5 to part 60 for conducting the relative accuracy evaluations. The method ANSI/ ASME PTC 19.10-1981 (incorporated by reference—see §60.17) is an acceptable alternative to EPA Method 15A of Appendix A-5 to part 60. The alternative relative accuracy procedures described in section 16.0 of Performance Specification 2 of Appendix B to part 60 (cylinder gas audits) may be used for conducting the relative accuracy evaluations. For flares that do not receive routine flow, the alternative relative accuracy procedures described in section 16.0 of Performance Specification 2 of Appendix B to part 60 (cylinder gas audits) may be used for conducting the relative accuracy evaluations, except that it is not necessary to include as much of the sampling probe or sampling line as practical.

### §60.107a

(iii) The owner or operator shall comply with the applicable quality assurance procedures in Appendix F to part 60 for each  $\rm H_2S$  monitor.

(iv) In the first 10 operating days after the date the flare must begin to comply with §60.103a(c)(1), the owner or operator shall collect representative daily samples of the gas discharged to the flare. The samples may be grab samples or integrated samples. The owner or operator shall take subsequent representative daily samples at least once per week or as required in paragraph (e)(2)(ix) of this section.

(v) The owner or operator shall analyze each daily sample for total sulfur using either EPA Method 15A of Appendix A-5 to part 60, EPA Method 16A of Appendix A-6 to part 60, ASTM Method D4468-85 (Reapproved 2006) (incorporated by reference—see §60.17) or

ASTM Method D5504-08 (incorporated by reference—see §60.17).

(vi) The owner or operator shall develop a 10-day average total sulfur-to-  $\rm H_2S$  ratio and 95-percent confidence interval as follows:

(A) Calculate the ratio of the total sulfur concentration to the  $\rm H_2S$  concentration for each day during which samples are collected.

(B) Determine the 10-day average total sulfur-to-H<sub>2</sub>S ratio as the arithmetic average of the daily ratios calculated in paragraph (e)(2)(vi)(A) of this section.

(C) Determine the acceptable range for subsequent weekly samples based on the 95-percent confidence interval for the distribution of daily ratios basedon the 10 individual daily ratios using Equation 11 of this section.

### $AR = Ratio_{Avg} \pm 2.262 \times SDev$ (Eq. 11)

Where:

AR = Acceptable range of subsequent ratio determinations, unitless.

 $Ratio_{Avg} = 10$ -day average total sulfur-to- $H_2S$  concentration ratio, unitless.

2.262 = t-distribution statistic for 95-percent 2-sided confidence interval for 10 samples (9 degrees of freedom).

SDev = Standard deviation of the 10 daily average total sulfur-to-H<sub>2</sub>S concentration ratios used to develop the 10-day average total sulfur-to-H<sub>2</sub>S concentration ratio, unitless.

(vii) For each day during the period when data are being collected to develop a 10-day average, the owner or operator shall estimate the total sulfur concentration using the measured total sulfur concentration measured for that day.

(viii) For all days other than those during which data are being collected to develop a 10-day average, the owner or operator shall multiply the most recent 10-day average total sulfur-to- $H_2S$  ratio by the daily average  $H_2S$  concentrations obtained using the monitor as required by paragraph (e)(2)(i) through (iii) of this section to estimate total sulfur concentrations.

(ix) If the total sulfur-to- $H_2S$  ratio for a subsequent weekly sample is outside the acceptable range for the most recent distribution of daily ratios, the owner or operator shall develop a new 10-day average ratio and acceptable range based on data for the outlying weekly sample plus data collected over the following 9 operating days.

(3) SO<sub>2</sub> monitoring requirements. The owner or operator shall install, operate, calibrate and maintain an instrument for continuously monitoring and recording the concentration of SO<sub>2</sub> from a process heater or other fuel gas combustion device that is combusting gas representative of the fuel gas in the flare gas line according to the requirements in paragraph (a)(1) of this section, determine the F factor of the fuel gas at least daily according to the requirements in paragraphs through (4) of this section, determine the higher heating value of the fuel gas at least daily according to the requirements in paragraph (d)(7) of this section and calculate the total sulfur content (as SO<sub>2</sub>) in the fuel gas using Equation 12 of this section.

$$TS_{FG} = C_{SO_3} \times F_d \times HHV_{FG}$$
 (Eq. 12)

Where:

 $TS_{FG} = Total sulfur concentration, as <math>SO_2$ , in the fuel gas, ppmv.

C<sub>SO2</sub> = Concentration of SO<sub>2</sub> in the exhaust gas, ppmv (dry basis at 0-percent excess air).

 $F_d = F$  factor gas on dry basis at 0-percent excess air, dscf/MMBtu.

 $\mathrm{HHV}_{\mathrm{FG}}=\mathrm{Higher}$  heating value of the fuel gas, MMBtu/scf.

- (4) Exemptions from sulfur monitoring requirements. Flares identified in paragraphs (e)(4)(i) through (iv) of this section are exempt from the requirements in paragraphs (e)(1) through (3) of this section. For each such flare, except as provided in paragraph (e)(4)(iv), engineering calculations shall be used to calculate the SO<sub>2</sub> emissions in the event of a discharge that may trigger a root cause analysis under \$60.103a(c)(1).
  - (i) Flares that can only receive:
- (A) Fuel gas streams that are inherently low in sulfur content as described in paragraph (a)(3)(i) through (iv) of this section; and/or
- (B) Fuel gas streams that are inherently low in sulfur content for which the owner or operator has applied for an exemption from the H<sub>2</sub>S monitoring requirements as described in paragraph (b) of this section.
- (ii) Emergency flares, provided that for each such flare, the owner or operator complies with the monitoring alternative in paragraph (g) of this section.
- (iii) Flares equipped with flare gas recovery systems designed, sized and operated to capture all flows except those resulting from startup, shutdown or malfunction, provided that for each such flare, the owner or operator complies with the monitoring alternative in paragraph (g) of this section.
- (iv) Secondary flares that receive gas diverted from the primary flare. In the event of a discharge from the secondary flare, the sulfur content measured by the sulfur monitor on the primary flare should be used to calculate SO<sub>2</sub> emissions, regardless of whether or not the monitoring alternative in paragraph (g) of this section is selected for the secondary flare.

- (f) Flow monitoring for flares. Except as provided in paragraphs (f)(2) and (h) of this section, the owner or operator of an affected flare subject to \$60.103a(c) through (e) shall install, operate, calibrate and maintain, in accordance with the specifications in paragraph (f)(1) of this section, a CPMS to measure and record the flow rate of gas discharged to the flare. If a flow monitor is not already in place, the owner or operator of a modified flare shall comply with the requirements of this paragraph by no later than November 11, 2015 or upon startup of the modified flare, whichever is later.
- (1) The owner or operator shall install, calibrate, operate and maintain each flow monitor according to the manufacturer's procedures and specifications and the following requirements.
- (i) Locate the monitor in a position that provides a representative measurement of the total gas flow rate.
- (ii) Use a flow sensor with a measurement sensitivity of no more than 5 percent of the flow rate or 10 cubic feet per minute, whichever is greater.
- (iii) Use a flow monitor that is maintainable online, is able to continuously correct for temperature and pressure and is able to record flow in standard conditions (as defined in §60.2) over one-minute averages.
- (iv) At least quarterly, perform a visual inspection of all components of the monitor for physical and operational integrity and all electrical connections for oxidation and galvanic corrosion if the flow monitor is not equipped with a redundant flow sensor.
- (v) Recalibrate the flow monitor in accordance with the manufacturer's procedures and specifications biennially (every two years) or at the frequency specified by the manufacturer.
- (2) Emergency flares, secondary flares and flares equipped with flare gas recovery systems designed, sized and operated to capture all flows except those resulting from startup, shutdown or malfunction are not required to install continuous flow monitors; provided, however, that for any

### §60.107a

such flare, the owner or operator shall comply with the monitoring alternative in paragraph (g) of this section.

- (g) Alternative monitoring for certain flares equipped with water seals. The owner or operator of an affected flare subject to §60.103a(c) through (e) that can be classified as either an emergency flare, a secondary flare or a flare equipped with a flare gas recovery system designed, sized and operated to capture all flows except those resulting from startup, shutdown or malfunction may, as an alternative to the sulfur and flow monitoring requirements of paragraphs (e) and (f) of this section, install, operate, calibrate and maintain, in accordance with the requirements in paragraphs (g)(1) through (7) of this section, a CPMS to measure and record the pressure in the flare gas header between the knock-out pot and water seal and to measure and record the water seal liquid level. If the required monitoring systems are not already in place, the owner or operator of a modified flare shall comply with the requirements of this paragraph by no later than November 11, 2015 or upon startup of the modified flare, whichever is later.
- (1) Locate the pressure sensor(s) in a position that provides a representative measurement of the pressure and locate the liquid seal level monitor in a position that provides a representative measurement of the water column height.
- (2) Minimize or eliminate pulsating pressure, vibration and internal and external corrosion.
- (3) Use a pressure sensor and level monitor with a minimum tolerance of 1.27 centimeters of water.
- (4) Using a manometer, check pressure sensor calibration quarterly.
- (5) Conduct calibration checks any time the pressure sensor exceeds the manufacturer's specified maximum operating pressure range or install a new pressure sensor.
- (6) In a cascaded flare system that employs multiple secondary flares, pressure and liquid level monitoring is required only on the first secondary flare in the system (i.e., the secondary flare with the lowest pressure release set point).

- (7) This alternative monitoring option may be elected only for flares with four or fewer pressure exceedances reauired to be reported under §60.108a(d)(5) ("reportable pressure exceedances") in any 365 consecutive calendar days. Following the fifth reportable pressure exceedance in a 365day period, the owner or operator must comply with the sulfur and flow monitoring requirements of paragraphs (e) and (f) of this section as soon as practical, but no later than 180 days after the fifth reportable pressure exceedance in a 365-day period.
- (h) Alternative monitoring for flares located in the BAAQMD or SCAQMD. An affected flare subject to this subpart located in the BAAQMD may elect to comply with the monitoring requirements in both BAAQMD Regulation 12. Rule 11 and BAAQMD Regulation 12, Rule 12 as an alternative to complying with the requirements of paragraphs (e) and (f) of this section. An affected flare subject to this subpart located in the SCAQMD may elect to comply with monitoring requirements SCAOMD Rule 1118 as an alternative to complying with the requirements of paragraphs (e) and (f) of this section.
- (i) Excess emissions. For the purpose of reports required by §60.7(c), periods of excess emissions for fuel gas combustion devices subject to the emissions limitations in §60.102a(g) and flares subject to the concentration requirement in §60.103a(h) are defined as specified in paragraphs (i)(1) through (5) of this section. Determine a rolling 3-hour or a rolling daily average as the arithmetic average of the applicable 1-hour averages (e.g., a rolling 3-hour average is the arithmetic average of three contiguous 1-hour averages). Determine a rolling 30-day or a rolling 365-day average as the arithmetic average of the applicable daily averages (e.g., a rolling 30-day average is the arithmetic average of 30 contiguous daily averages).
- (1)  $SO_2$  or  $H_2S$  limits for fuel gas combustion devices. (i) If the owner or operator of a fuel gas combustion device elects to comply with the  $SO_2$  emission limits in  $\S60.102a(g)(1)(i)$ , each rolling 3-hour period during which the average concentration of  $SO_2$  as measured by the  $SO_2$  continuous monitoring system required under paragraph (a)(1) of this

section exceeds 20 ppmv, and each rolling 365-day period during which the average concentration of SO<sub>2</sub> as measured by the SO<sub>2</sub> continuous monitoring system required under paragraph (a)(1) of this section exceeds 8 ppmv.

(ii) If the owner or operator of a fuel gas combustion device elects to comply with the  $\rm H_2S$  concentration limits in  $\S60.102a(\rm g)(1)(ii)$ , each rolling 3-hour period during which the average concentration of  $\rm H_2S$  as measured by the  $\rm H_2S$  continuous monitoring system required under paragraph (a)(2) of this section exceeds 162 ppmv and each rolling 365-day period during which the average concentration as measured by the  $\rm H_2S$  continuous monitoring system under paragraph (a)(2) of this section exceeds 60 ppmv.

(iii) If the owner or operator of a fuel gas combustion device becomes subject to the requirements of daily stain tube sampling in paragraph (b)(3)(iii) of this section, each day during which the daily concentration of  $H_2S$  exceeds 162 ppmv and each rolling 365-day period during which the average concentration of  $H_2S$  exceeds 60 ppmv.

(2) H<sub>2</sub>S concentration limits for flares. (i) Each rolling 3-hour period during which the average concentration of H<sub>2</sub>S as measured by the H<sub>2</sub>S continuous monitoring system required under paragraph (a)(2) of this section exceeds 162 ppmv.

(ii) If the owner or operator of a flare becomes subject to the requirements of daily stain tube sampling in paragraph (b)(3)(iii) of this section, each day during which the daily concentration of  $\rm H_2S$  exceeds 162 ppmv.

(3) Rolling 30-day average  $NO_X$  limits for fuel gas combustion devices. Each rolling 30-day period during which the average concentration of  $NO_X$  as measured by the  $NO_X$  continuous monitoring system required under paragraph (c) or (d) of this section exceeds:

(i) For a natural draft process heater, 40 ppmv and, if monitored according to §60.107a(d), 0.040 lb/MMBtu;

(ii) For a forced draft process heater, 60 ppmv and, if monitored according to \$60.107a(d), 0.060 lb/MMBtu; and

(iii) For a co-fired process heater electing to comply with the  $NO_X$  limit in  $\S 60.102a(g)(2)(iii)(A)$  or (g)(2)(iv)(A), 150 ppmv.

- (iv) The site-specific limit determined by the Administrator under §60.102a(i).
- (4) Daily  $NO_X$  limits for fuel gas combustion devices. Each day during which the concentration of  $NO_X$  as measured by the  $NO_X$  continuous monitoring system required under paragraph (d) of this section exceeds the daily average emissions limit calculated using Equation 3 in  $\S60.102a(g)(2)(iii)(B)$  or Equation 4 in  $\S60.102a(g)(2)(iii)(B)$ .
- (5) Daily  $O_2$  limits for fuel gas combustion devices. Each day during which the concentration of  $O_2$  as measured by the  $O_2$  continuous monitoring system required under paragraph (c)(6) of this section exceeds the  $O_2$  operating limit or operating curve determined during the most recent biennial performance test.

[73 FR 35867, June 24, 2008, as amended at 77 FR 56473, Sep. 12, 2012]

## § 60.108a Recordkeeping and reporting requirements.

(a) Each owner or operator subject to the emissions limitations in §60.102a shall comply with the notification, recordkeeping, and reporting requirements in §60.7 and other requirements as specified in this section.

(b) Each owner or operator subject to an emissions limitation in §60.102a shall notify the Administrator of the specific monitoring provisions §§ 60,105a, 60,106a and 60,107a with which the owner or operator intends to comply. Each owner or operator of a cofired process heater subject to an emissions limitation in §60.102a(g)(2)(iii) or (iv) shall submit to the Administrator documentation showing that the process heater meets the definition of a cofired process heater in §60.101a. Notifications required by this paragraph shall be submitted with the notification of initial startup required by  $\S60.7(a)(3)$ .

- (c) The owner or operator shall maintain the following records:
- (1) A copy of the flare management
- (2) Records of information to document conformance with bag leak detection system operation and maintenance requirements in §60.105a(c).

### § 60.108a

- (3) Records of bag leak detection system alarms and actions according to §60.105a(c).
- (4) For each FCCU and fluid coking unit subject to the monitoring requirements in §60.105a(b)(1), records of the average coke burn-off rate and hours of operation.
- (5) For each fuel gas stream to which one of the exemptions listed in §60.107a(a)(3) applies, records of the specific exemption determined to apply for each fuel stream. If the owner or operator applies for the exemption described in §60.107a(a)(3)(iv), the owner or operator must keep a copy of the application as well as the letter from the Administrator granting approval of the application.
- (6) Records of discharges greater than 500 lb SO2 in any 24-hour period from any affected flare, discharges greater than 500 lb SO2 in excess of the allowable limits from a fuel gas combustion device or sulfur recovery plant and discharges to an affected flare in excess of 500,000 scf above baseline in any 24-hour period as required by §60.103a(c). If the monitoring alternative provided in §60.107a(g) is selected, the owner or operator shall record any instance when the flare gas line pressure exceeds the water seal liquid depth, except for periods attributable to compressor staging that do not exceed the staging time specified in  $\S60.103a(a)(3)(vii)(C)$ . The following information shall be recorded no later than 45 days following the end of a discharge exceeding the thresholds:
  - (i) A description of the discharge.
- (ii) The date and time the discharge was first identified and the duration of the discharge.
- (iii) The measured or calculated cumulative quantity of gas discharged over the discharge duration. If the discharge duration exceeds 24 hours, record the discharge quantity for each 24-hour period. For a flare, record the measured or calculated cumulative quantity of gas discharged to the flare over the discharge duration. If the discharge duration exceeds 24 hours, record the quantity of gas discharged to the flare for each 24-hour period. Engineering calculations are allowed for fuel gas combustion devices, but are not allowed for flares, except for those

- complying with the alternative monitoring requirements in §60.107a(g).
- (iv) For each discharge greater than 500 lb  $SO_2$  in any 24-hour period from a flare, the measured total sulfur concentration or both the measured  $H_2S$  concentration and the estimated total sulfur concentration in the fuel gas at a representative location in the flare inlet.
- (v) For each discharge greater than 500 lb  $SO_2$  in excess of the applicable short-term emissions limit in  $\S 60.102a(g)(1)$  from a fuel gas combustion device, either the measured concentration of  $H_2S$  in the fuel gas or the measured concentration of  $SO_2$  in the stream discharged to the atmosphere. Process knowledge can be used to make these estimates for fuel gas combustion devices, but cannot be used to make these estimates for flares, except as provided in  $\S 60.107a(e)(4)$ .
- (vi) For each discharge greater than 500 lb  $SO_2$  in excess of the allowable limits from a sulfur recovery plant, either the measured concentration of reduced sulfur or  $SO_2$  discharged to the atmosphere.
- (vii) For each discharge greater than 500 lb  $SO_2$  in any 24-hour period from any affected flare or discharge greater than 500 lb  $SO_2$  in excess of the allowable limits from a fuel gas combustion device or sulfur recovery plant, the cumulative quantity of  $H_2S$  and  $SO_2$  released into the atmosphere. For releases controlled by flares, assume 99-percent conversion of reduced sulfur or total sulfur to  $SO_2$ . For fuel gas combustion devices, assume 99-percent conversion of  $H_2S$  to  $SO_2$ .
- (viii) The steps that the owner or operator took to limit the emissions during the discharge.
- (ix) The root cause analysis and corrective action analysis conducted as required in §60.103a(d), including an identification of the affected facility, the date and duration of the discharge, a statement noting whether the discharge resulted from the same root cause(s) identified in a previous analysis and either a description of the recommended corrective action(s) or an explanation of why corrective action is not necessary under §60.103a(e).

- (x) For any corrective action analysis for which corrective actions are required in §60.103a(e), a description of the corrective action(s) completed within the first 45 days following the discharge and, for action(s) not already completed, a schedule for implementation, including proposed commencement and completion dates.
- (xi) For each discharge from any affected flare that is the result of a planned startup or shutdown of a refinery process unit or ancillary equipment connected to the affected flare, a statement that a root cause analysis and corrective action analysis are not necessary because the owner or operator followed the flare management plan.
- (7) If the owner or operator elects to comply with \$60.107a(e)(2) for a flare, records of the H<sub>2</sub>S and total sulfur analyses of each grab or integrated sample, the calculated daily total sulfur-to-H<sub>2</sub>S ratios, the calculated 10-day average total sulfur-to-H<sub>2</sub>S ratios and the 95-percent confidence intervals for each 10-day average total sulfur-to-H<sub>2</sub>S ratio
- (d) Each owner or operator subject to this subpart shall submit an excess emissions report for all periods of excess emissions according to the requirements of §60.7(c) except that the report shall contain the information specified in paragraphs (d)(1) through (7) of this section.
- (1) The date that the exceedance occurred;
- (2) An explanation of the exceedance:
- (3) Whether the exceedance was concurrent with a startup, shutdown, or malfunction of an affected facility or control system; and
- (4) A description of the action taken, if any.
- (5) The information described in paragraph (c)(6) of this section for all discharges listed in paragraph (c)(6) of this section. For a flare complying with the monitoring alternative under \$60.107a(g), following the fifth discharge required to be recorded under paragraph (c)(6) of this section and reported under this paragraph, the owner or operator shall include notification that monitoring systems will be installed according to \$60.107a(e) and (f) within 180 days following the fifth discharge.

- (6) For any periods for which monitoring data are not available, any changes made in operation of the emission control system during the period of data unavailability which could affect the ability of the system to meet the applicable emission limit. Operations of the control system and affected facility during periods of data unavailability are to be compared with operation of the control system and affected facility before and following the period of data unavailability.
- (7) A written statement, signed by a responsible official, certifying the accuracy and completeness of the information contained in the report.

[73 FR 35867, June 24, 2008, as amended at 77 FR 56479, Sep. 12, 2012]

### § 60.109a Delegation of authority.

- (a) This subpart can be implemented and enforced by the U.S. EPA or a delegated authority such as a State, local, or tribal agency. You should contact your U.S. EPA Regional Office to find out if this subpart is delegated to a State, local, or tribal agency within your State.
- (b) In delegating implementation and enforcement authority of this subpart to a state, local or tribal agency, the approval authorities contained in paragraphs (b)(1) through (4) of this section are retained by the Administrator of the U.S. EPA and are not transferred to the state, local or tribal agency.
- (1) Approval of a major change to test methods under §60.8(b). A "major change to test method" is defined in 40 CFR 63.90.
- (2) Approval of a major change to monitoring under §60.13(i). A "major change to monitoring" is defined in 40 CFR 63.90
- (3) Approval of a major change to recordkeeping/reporting under §60.7(b) through (f). A "major change to recordkeeping/reporting" is defined in 40 CFR 63 90
- (4) Approval of an application for an alternative means of emission limitation under §60.103a(j) of this subpart.
- [73 FR 35867, June 24, 2008, as amended at 77 FR 56480, Sep. 12, 2012]

### Pt. 60, Subpt. Ja, Table 1

TABLE 1 TO SUBPART JA OF PART 60— MOLAR EXHAUST VOLUMES AND MOLAR HEAT CONTENT OF FUEL GAS CONSTITUENTS

Constituent	MEV <sup>a</sup> dscf/mol	MHC <sup>b</sup> Btu/mol
Methane (CH <sub>4</sub> )	7.29	842
Ethane (C <sub>2</sub> H <sub>6</sub> )	12.96	1,475
Hydrogen (H <sub>2</sub> )	1.61	269
Ethene (C <sub>2</sub> H <sub>4</sub> )	11.34	1,335
Propane (C <sub>3</sub> H <sub>8</sub> )	18.62	2,100
Propene (C <sub>3</sub> H <sub>6</sub> )	17.02	1,947
Butane (C <sub>4</sub> H <sub>10</sub> )	24.30	2,717
Butene (C <sub>4</sub> H <sub>8</sub> )	22.69	2,558
Inerts	0.85	0

a MEV = molar exhaust volume, dry standard cubic feet per gram-mole (dscf/g-mol) at standard conditions of 68 °F and 1 atmosphere.

bMHC = molar heat content (higher heating value basis), Btu per gram-mole (Btu/g-mol).

[77 FR 56480, Sep. 12, 2012]

Subpart K—Standards of Performance for Storage Vessels for Petroleum Liquids for Which Construction, Reconstruction, or Modification Commenced After June 11, 1973, and Prior to May 19, 1978

## § 60.110 Applicability and designation of affected facility.

- (a) Except as provided in §60.110(b), the affected facility to which this subpart applies is each storage vessel for petroleum liquids which has a storage capacity greater than 151,412 liters (40,000 gallons).
- (b) This subpart does not apply to storage vessels for petroleum or condensate stored, processed, and/or treated at a drilling and production facility prior to custody transfer.
- (c) Subject to the requirements of this subpart is any facility under paragraph (a) of this section which:
- (1) Has a capacity greater than 151, 416 liters (40,000 gallons), but not exceeding 246,052 liters (65,000 gallons), and commences construction or modification after March 8, 1974, and prior to May 19, 1978.
- (2) Has a capacity greater than 246,052 liters (65,000 gallons) and commences construction or modification after June 11, 1973, and prior to May 19, 1978.

[42 FR 37937, July 25, 1977, as amended at 45 FR 23379, Apr. 4, 1980]

#### § 60.111 Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the Act and in subpart A of this part.

- (a) Storage vessel means any tank, reservoir, or container used for the storage of petroleum liquids, but does not include:
- (1) Pressure vessels which are designed to operate in excess of 15 pounds per square inch gauge without emissions to the atmosphere except under emergency conditions,
- (2) Subsurface caverns or porous rock reservoirs, or
- (3) Underground tanks if the total volume of petroleum liquids added to and taken from a tank annually does not exceed twice the volume of the tank.
- (b) Petroleum liquids means petroleum, condensate, and any finished or intermediate products manufactured in a petroleum refinery but does not mean Nos. 2 through 6 fuel oils as specified in ASTM D396-78, 89, 90, 92, 96, or 98, gas turbine fuel oils Nos. 2-GT through 4-GT as specified in ASTM D2880-78 or 96, or diesel fuel oils Nos. 2-D and 4-D as specified in ASTM D975-78, 96, or 98a. (These three methods are incorporated by reference—see §60.17.)
- (c) Petroleum refinery means each facility engaged in producing gasoline, kerosene, distillate fuel oils, residual fuel oils, lubricants, or other products through distillation of petroleum or through redistillation, cracking, extracting, or reforming of unfinished petroleum derivatives.
- (d) *Petroleum* means the crude oil removed from the earth and the oils derived from tar sands, shale, and coal.
- (e) *Hydrocarbon* means any organic compound consisting predominantly of carbon and hydrogen.
- (f) Condensate means hydrocarbon liquid separated from natural gas which condenses due to changes in the temperature and/or pressure and remains liquid at standard conditions.
- (g) Custody transfer means the transfer of produced petroleum and/or condensate, after processing and/or treating in the producing operations, from storage tanks or automatic transfer facilities to pipelines or any other forms of transportation.

Appendix C 40 CFR Part 60, Subpart Kb

Standards of Performance for Volatile Organic Liquid Storage Vessels (Including Petroleum Liquid Storage Vessels) for Which Construction, Reconstruction, or Modification Commenced After July 23, 1984

## §60.114a

## § 60.114a Alternative means of emission limitation.

- (a) If, in the Administrator's judgment, an alternative means of emission limitation will achieve a reduction in emissions at least equivalent to the reduction in emissions achieved by any requirement in §60.112a, the Administrator will publish in the FEDERAL REGISTER a notice permitting the use of the alternative means for purposes of compliance with that requirement.
- (b) Any notice under paragraph (a) of this section will be published only after notice and an opportunity for a hearing.
- (c) Any person seeking permission under this section shall submit to the Administrator a written application including:
- (1) An actual emissions test that uses a full-sized or scale-model storage vessel that accurately collects and measures all VOC emissions from a given control device and that accurately simulates wind and accounts for other emission variables such as temperature and barometric pressure.
- (2) An engineering evaluation that the Administrator determines is an accurate method of determining equivalence.
- (d) The Administrator may condition the permission on requirements that may be necessary to ensure operation and maintenance to achieve the same emissions reduction as specified in §60.112a.
- (e) The primary vapor-mounted seal in the "Volume-Maximizing Seal" manufactured by R.F.I. Services Corporation is approved as equivalent to the vapor-mounted seal required by §60.112a(a)(1)(i) and must meet the gap criteria specified in §60.112a(a)(1)(i)(B). There shall be no gaps between the tank wall and any secondary seal used in conjunction with the primary seal in the "Volume-Maximizing Seal".

[52 FR 11429, Apr. 8, 1987]

#### § 60.115a Monitoring of operations.

(a) Except as provided in paragraph (d) of this section, the owner or operator subject to this subpart shall maintain a record of the petroleum liquid stored, the period of storage, and the maximum true vapor pressure of that

## 40 CFR Ch. I (7-1-13 Edition)

liquid during the respective storage period.

- (b) Available data on the typical Reid vapor pressure and the maximum expected storage temperature of the stored product may be used to determine the maximum true vapor pressure from nomographs contained in API Bulletin 2517, unless the Administrator specifically requests that the liquid be sampled, the actual storage temperature determined, and the Reid vapor pressure determined from the sample(s).
- (c) The true vapor pressure of each type of crude oil with a Reid vapor pressure less than 13.8 kPa (2.0 psia) or whose physical properties preclude determination by the recommended method is to be determined from available data and recorded if the estimated true vapor pressure is greater than 6.9 kPa (1.0 psia).
- (d) The following are exempt from the requirements of this section:
- (1) Each owner or operator of each storage vessel storing a petroleum liquid with a Reid vapor pressure of less than 6.9 kPa (1.0 psia) provided the maximum true vapor pressure does not exceed 6.9 kPa (1.0 psia).
- (2) The owner or operator of each storage vessel equipped with a vapor recovery and return or disposal system in accordance with the requirements of \$60.112a(a)(3) and (b), or a closed vent system and control device meeting the specifications of 40 CFR 65.42(b)(4), (b)(5), or (c).

[45 FR 23379, Apr. 4, 1980, as amended at 65 FR 78275, Dec. 14, 2000]

Subpart Kb—Standards of Performance for Volatile Organic Liquid Storage Vessels (Including Petroleum Liquid Storage Vessels) for Which Construction, Reconstruction, or Modification Commenced After July 23, 1984

SOURCE: 52 FR 11429, Apr. 8, 1987, unless otherwise noted.

# § 60.110b Applicability and designation of affected facility.

(a) Except as provided in paragraph (b) of this section, the affected facility

to which this subpart applies is each storage vessel with a capacity greater than or equal to 75 cubic meters (m³) that is used to store volatile organic liquids (VOL) for which construction, reconstruction, or modification is commenced after July 23, 1984.

- (b) This subpart does not apply to storage vessels with a capacity greater than or equal to 151 m³ storing a liquid with a maximum true vapor pressure less than 3.5 kilopascals (kPa) or with a capacity greater than or equal to 75 m³ but less than 151 m³ storing a liquid with a maximum true vapor pressure less than 15.0 kPa.
  - (c) [Reserved]
- (d) This subpart does not apply to the following:
- (1) Vessels at coke oven by-product plants.
- (2) Pressure vessels designed to operate in excess of 204.9 kPa and without emissions to the atmosphere.
- (3) Vessels permanently attached to mobile vehicles such as trucks, railcars, barges, or ships.
- (4) Vessels with a design capacity less than or equal to 1,589.874 m³ used for petroleum or condensate stored, processed, or treated prior to custody transfer.
- (5) Vessels located at bulk gasoline plants.
- (6) Storage vessels located at gasoline service stations.
- (7) Vessels used to store beverage alcohol.
- (8) Vessels subject to subpart GGGG of 40 CFR part 63.
- (e) Alternative means of compliance—(1) Option to comply with part 65. Owners or operators may choose to comply with 40 CFR part 65, subpart C, to satisfy the requirements of §§ 60.112b through 60.117b for storage vessels that are subject to this subpart that meet the specifications in paragraphs (e)(1)(i) and (ii) of this section. When choosing to comply with 40 CFR part 65, subpart C, the monitoring requirements of §60.116b(c), (e), (f)(1), and (g) still apply. Other provisions applying to owners or operators who choose to comply with 40 CFR part 65 are provided in 40 CFR 65.1.
- (i) A storage vessel with a design capacity greater than or equal to 151 m<sup>3</sup> containing a VOL that, as stored, has a

maximum true vapor pressure equal to or greater than 5.2 kPa; or

- (ii) A storage vessel with a design capacity greater than 75 m³ but less than 151 m³ containing a VOL that, as stored, has a maximum true vapor pressure equal to or greater than 27.6 kPa.
- (2) Part 60, subpart A. Owners or operators who choose to comply with 40 CFR part 65, subpart C, must also comply with §§ 60.1, 60.2, 60.5, 60.6, 60.7(a)(1) and (4), 60.14, 60.15, and 60.16 for those storage vessels. All sections and paragraphs of subpart A of this part that are not mentioned in this paragraph (e)(2) do not apply to owners or operators of storage vessels complying with 40 CFR part 65, subpart C, except that provisions required to be met prior to implementing 40 CFR part 65 still apply. Owners and operators who choose to comply with 40 CFR part 65, subpart C, must comply with 40 CFR part 65, subpart A.
- (3) Internal floating roof report. If an owner or operator installs an internal floating roof and, at initial startup, chooses to comply with 40 CFR part 65, subpart C, a report shall be furnished to the Administrator stating that the control equipment meets the specifications of 40 CFR 65.43. This report shall be an attachment to the notification required by 40 CFR 65.5(b).
- (4) External floating roof report. If an owner or operator installs an external floating roof and, at initial startup, chooses to comply with 40 CFR part 65, subpart C, a report shall be furnished to the Administrator stating that the control equipment meets the specifications of 40 CFR 65.44. This report shall be an attachment to the notification required by 40 CFR 65.5(b).

[52 FR 11429, Apr. 8, 1987, as amended at 54 FR 32973, Aug. 11, 1989; 65 FR 78275, Dec. 14, 2000; 68 FR 59332. Oct. 15, 2003]

## § 60.111b Definitions.

Terms used in this subpart are defined in the Act, in subpart A of this part, or in this subpart as follows:

Bulk gasoline plant means any gasoline distribution facility that has a gasoline throughput less than or equal to 75,700 liters per day. Gasoline throughput shall be the maximum calculated design throughput as may be

limited by compliance with an enforceable condition under Federal requirement or Federal, State or local law, and discoverable by the Administrator and any other person.

Condensate means hydrocarbon liquid separated from natural gas that condenses due to changes in the temperature or pressure, or both, and remains liquid at standard conditions.

Custody transfer means the transfer of produced petroleum and/or condensate, after processing and/or treatment in the producing operations, from storage vessels or automatic transfer facilities to pipelines or any other forms of transportation.

Fill means the introduction of VOL into a storage vessel but not necessarily to complete capacity.

Gasoline service station means any site where gasoline is dispensed to motor vehicle fuel tanks from stationary storage tanks.

Maximum true vapor pressure means the equilibrium partial pressure exerted by the volatile organic compounds (as defined in 40 CFR 51.100) in the stored VOL at the temperature equal to the highest calendar-month average of the VOL storage temperature for VOL's stored above or below the ambient temperature or at the local maximum monthly average temperature as reported by the National Weather Service for VOL's stored at the ambient temperature, as determined:

- (1) In accordance with methods described in American Petroleum institute Bulletin 2517, Evaporation Loss From External Floating Roof Tanks, (incorporated by reference—see §60.17); or
- (2) As obtained from standard reference texts; or
- (3) As determined by ASTM D2879-83, 96, or 97 (incorporated by reference—see §60.17);
- (4) Any other method approved by the Administrator.

Petroleum means the crude oil removed from the earth and the oils derived from tar sands, shale, and coal.

Petroleum liquids means petroleum, condensate, and any finished or intermediate products manufactured in a petroleum refinery.

Process tank means a tank that is used within a process (including a solvent or raw material recovery process) to collect material discharged from a feedstock storage vessel or equipment within the process before the material is transferred to other equipment within the process, to a product or by-product storage vessel, or to a vessel used to store recovered solvent or raw material. In many process tanks, unit operations such as reactions and blending are conducted. Other process tanks, such as surge control vessels and bottoms receivers, however, may not involve unit operations.

Reid vapor pressure means the absolute vapor pressure of volatile crude oil and volatile nonviscous petroleum liquids except liquified petroleum gases, as determined by ASTM D323-82 or 94 (incorporated by reference—see §60.17).

Storage vessel means each tank, reservoir, or container used for the storage of volatile organic liquids but does not include:

- (1) Frames, housing, auxiliary supports, or other components that are not directly involved in the containment of liquids or vapors;
- (2) Subsurface caverns or porous rock reservoirs; or
  - (3) Process tanks.

Volatile organic liquid (VOL) means any organic liquid which can emit volatile organic compounds (as defined in 40 CFR 51.100) into the atmosphere.

Waste means any liquid resulting from industrial, commercial, mining or agricultural operations, or from community activities that is discarded or is being accumulated, stored, or physically, chemically, or biologically treated prior to being discarded or recycled.

[52 FR 11429, Apr. 8, 1987, as amended at 54 FR 32973, Aug. 11, 1989; 65 FR 61756, Oct. 17, 2000; 68 FR 59333, Oct. 15, 2003]

## § 60.112b Standard for volatile organic compounds (VOC).

(a) The owner or operator of each storage vessel either with a design capacity greater than or equal to 151 m³ containing a VOL that, as stored, has a maximum true vapor pressure equal to or greater than 5.2 kPa but less than 76.6 kPa or with a design capacity greater than or equal to 75 m³ but less

than 151 m³ containing a VOL that, as stored, has a maximum true vapor pressure equal to or greater than 27.6 kPa but less than 76.6 kPa, shall equip each storage vessel with one of the following:

- (1) A fixed roof in combination with an internal floating roof meeting the following specifications:
- (i) The internal floating roof shall rest or float on the liquid surface (but not necessarily in complete contact with it) inside a storage vessel that has a fixed roof. The internal floating roof shall be floating on the liquid surface at all times, except during initial fill and during those intervals when the storage vessel is completely emptied or subsequently emptied and refilled. When the roof is resting on the leg supports, the process of filling, emptying, or refilling shall be continuous and shall be accomplished as rapidly as possible.
- (ii) Each internal floating roof shall be equipped with one of the following closure devices between the wall of the storage vessel and the edge of the internal floating roof:
- (A) A foam- or liquid-filled seal mounted in contact with the liquid (liquid-mounted seal). A liquid-mounted seal means a foam- or liquid-filled seal mounted in contact with the liquid between the wall of the storage vessel and the floating roof continuously around the circumference of the tank.
- (B) Two seals mounted one above the other so that each forms a continuous closure that completely covers the space between the wall of the storage vessel and the edge of the internal floating roof. The lower seal may be vapor-mounted, but both must be continuous.
- (C) A mechanical shoe seal. A mechanical shoe seal is a metal sheet held vertically against the wall of the storage vessel by springs or weighted levers and is connected by braces to the floating roof. A flexible coated fabric (envelope) spans the annular space between the metal sheet and the floating roof.
- (iii) Each opening in a noncontact internal floating roof except for automatic bleeder vents (vacuum breaker vents) and the rim space vents is to provide a projection below the liquid surface.

- (iv) Each opening in the internal floating roof except for leg sleeves, automatic bleeder vents, rim space vents, column wells, ladder wells, sample wells, and stub drains is to be equipped with a cover or lid which is to be maintained in a closed position at all times (i.e., no visible gap) except when the device is in actual use. The cover or lid shall be equipped with a gasket. Covers on each access hatch and automatic gauge float well shall be bolted except when they are in use.
- (v) Automatic bleeder vents shall be equipped with a gasket and are to be closed at all times when the roof is floating except when the roof is being floated off or is being landed on the roof leg supports.
- (vi) Rim space vents shall be equipped with a gasket and are to be set to open only when the internal floating roof is not floating or at the manufacturer's recommended setting.
- (vii) Each penetration of the internal floating roof for the purpose of sampling shall be a sample well. The sample well shall have a slit fabric cover that covers at least 90 percent of the opening.
- (viii) Each penetration of the internal floating roof that allows for passage of a column supporting the fixed roof shall have a flexible fabric sleeve seal or a gasketed sliding cover.
- (ix) Each penetration of the internal floating roof that allows for passage of a ladder shall have a gasketed sliding cover
- (2) An external floating roof. An external floating roof means a pontoon-type or double-deck type cover that rests on the liquid surface in a vessel with no fixed roof. Each external floating roof must meet the following specifications:
- (i) Each external floating roof shall be equipped with a closure device between the wall of the storage vessel and the roof edge. The closure device is to consist of two seals, one above the other. The lower seal is referred to as the primary seal, and the upper seal is referred to as the secondary seal.
- (A) The primary seal shall be either a mechanical shoe seal or a liquid-mounted seal. Except as provided in §60.113b(b)(4), the seal shall completely

cover the annular space between the edge of the floating roof and tank wall.

- (B) The secondary seal shall completely cover the annular space between the external floating roof and the wall of the storage vessel in a continuous fashion except as allowed in §60.113b(b)(4).
- (ii) Except for automatic bleeder vents and rim space vents, each opening in a noncontact external floating roof shall provide a projection below the liquid surface. Except for automatic bleeder vents, rim space vents, roof drains, and leg sleeves, each opening in the roof is to be equipped with a gasketed cover, seal, or lid that is to be maintained in a closed position at all times (i.e., no visible gap) except when the device is in actual use. Automatic bleeder vents are to be closed at all times when the roof is floating except when the roof is being floated off or is being landed on the roof leg supports. Rim vents are to be set to open when the roof is being floated off the roof legs supports or at the manufacturer's recommended setting. Automatic bleeder vents and rim space vents are to be gasketed. Each emergency roof drain is to be provided with a slotted membrane fabric cover that covers at least 90 percent of the area of the opening.
- (iii) The roof shall be floating on the liquid at all times (i.e., off the roof leg supports) except during initial fill until the roof is lifted off leg supports and when the tank is completely emptied and subsequently refilled. The process of filling, emptying, or refilling when the roof is resting on the leg supports shall be continuous and shall be accomplished as rapidly as possible.
- (3) A closed vent system and control device meeting the following specifications:
- (i) The closed vent system shall be designed to collect all VOC vapors and gases discharged from the storage vessel and operated with no detectable emissions as indicated by an instrument reading of less than 500 ppm above background and visual inspections, as determined in part 60, subpart VV, § 60.485(b).
- (ii) The control device shall be designed and operated to reduce inlet VOC emissions by 95 percent or great-

- er. If a flare is used as the control device, it shall meet the specifications described in the general control device requirements (§60.18) of the General Provisions.
- (4) A system equivalent to those described in paragraphs (a)(1), (a)(2), or (a)(3) of this section as provided in §60.114b of this subpart.
- (b) The owner or operator of each storage vessel with a design capacity greater than or equal to 75 m³ which contains a VOL that, as stored, has a maximum true vapor pressure greater than or equal to 76.6 kPa shall equip each storage vessel with one of the following:
- (1) A closed vent system and control device as specified in §60.112b(a)(3).
- (2) A system equivalent to that described in paragraph (b)(1) as provided in §60.114b of this subpart.
- (c) Site-specific standard for Merck & Co., Inc.'s Stonewall Plant in Elkton, Virginia. This paragraph applies only to the pharmaceutical manufacturing facility, commonly referred to as the Stonewall Plant, located at Route 340 South, in Elkton, Virginia ("site").
- (1) For any storage vessel that otherwise would be subject to the control technology requirements of paragraphs (a) or (b) of this section, the site shall have the option of either complying directly with the requirements of this subpart, or reducing the site-wide total criteria pollutant emissions cap (total emissions cap) in accordance with the procedures set forth in a permit issued pursuant to 40 CFR 52.2454. If the site chooses the option of reducing the total emissions cap in accordance with the procedures set forth in such permit, the requirements of such permit shall apply in lieu of the otherwise applicable requirements of this subpart for such storage vessel.
- (2) For any storage vessel at the site not subject to the requirements of 40 CFR 60.112b (a) or (b), the requirements of 40 CFR 60.116b (b) and (c) and the General Provisions (subpart A of this part) shall not apply.

[52 FR 11429, Apr. 8, 1987, as amended at 62 FR 52641, Oct. 8, 1997]

#### § 60.113b Testing and procedures.

The owner or operator of each storage vessel as specified in §60.112b(a)

shall meet the requirements of paragraph (a), (b), or (c) of this section. The applicable paragraph for a particular storage vessel depends on the control equipment installed to meet the requirements of §60.112b.

- (a) After installing the control equipment required to meet \$60.112b(a)(1) (permanently affixed roof and internal floating roof), each owner or operator shall:
- (1) Visually inspect the internal floating roof, the primary seal, and the secondary seal (if one is in service), prior to filling the storage vessel with VOL. If there are holes, tears, or other openings in the primary seal, the secondary seal, or the seal fabric or defects in the internal floating roof, or both, the owner or operator shall repair the items before filling the storage vessel.
- (2) For Vessels equipped with a liquid-mounted or mechanical shoe primary seal, visually inspect the internal floating roof and the primary seal or the secondary seal (if one is in service) through manholes and roof hatches on the fixed roof at least once every 12 months after initial fill. If the internal floating roof is not resting on the surface of the VOL inside the storage vessel, or there is liquid accumulated on the roof, or the seal is detached, or there are holes or tears in the seal fabric, the owner or operator shall repair the items or empty and remove the storage vessel from service within 45 days. If a failure that is detected during inspections required in this paragraph cannot be repaired within 45 days and if the vessel cannot be emptied within 45 days, a 30-day extension may be requested from the Administrator in the inspection report required in §60.115b(a)(3). Such a request for an extension must document that alternate storage capacity is unavailable and specify a schedule of actions the company will take that will assure that the control equipment will be repaired or the vessel will be emptied as soon as possible.
- (3) For vessels equipped with a double-seal system as specified in §60.112b(a)(1)(ii)(B):
- (i) Visually inspect the vessel as specified in paragraph (a)(4) of this section at least every 5 years; or

- (ii) Visually inspect the vessel as specified in paragraph (a)(2) of this section.
- Visually inspect the internal floating roof, the primary seal, the secondary seal (if one is in service), gaskets, slotted membranes and sleeve seals (if any) each time the storage vessel is emptied and degassed. If the internal floating roof has defects, the primary seal has holes, tears, or other openings in the seal or the seal fabric. or the secondary seal has holes, tears, or other openings in the seal or the seal fabric, or the gaskets no longer close off the liquid surfaces from the atmosphere, or the slotted membrane has more than 10 percent open area, the owner or operator shall repair the items as necessary so that none of the conditions specified in this paragraph exist before refilling the storage vessel with VOL. In no event shall inspections conducted in accordance with this provision occur at intervals greater than 10 years in the case of vessels conducting the annual visual inspection as specified in paragraphs (a)(2) and (a)(3)(ii) of this section and at intervals no greater than 5 years in the case of vessels specified in paragraph (a)(3)(i) of this section.
- (5) Notify the Administrator in writing at least 30 days prior to the filling or refilling of each storage vessel for which an inspection is required by paragraphs (a)(1) and (a)(4) of this section to afford the Administrator the opportunity to have an observer present. If the inspection required by paragraph (a)(4) of this section is not planned and the owner or operator could not have known about the inspection 30 days in advance or refilling the tank, the owner or operator shall notify the Administrator at least 7 days prior to the refilling of the storage vessel. Notification shall be made by telephone immediately followed by written documentation demonstrating why the inspection was unplanned. Alternatively, this notification including the written documentation may be made in writing and sent by express mail so that it is received by the Administrator at least 7 days prior to the refilling.
- (b) After installing the control equipment required to meet §60.112b(a)(2)

(external floating roof), the owner or operator shall:

- (1) Determine the gap areas and maximum gap widths, between the primary seal and the wall of the storage vessel and between the secondary seal and the wall of the storage vessel according to the following frequency.
- (i) Measurements of gaps between the tank wall and the primary seal (seal gaps) shall be performed during the hydrostatic testing of the vessel or within 60 days of the initial fill with VOL and at least once every 5 years thereafter.
- (ii) Measurements of gaps between the tank wall and the secondary seal shall be performed within 60 days of the initial fill with VOL and at least once per year thereafter.
- (iii) If any source ceases to store VOL for a period of 1 year or more, subsequent introduction of VOL into the vessel shall be considered an initial fill for the purposes of paragraphs (b)(1)(i) and (b)(1)(ii) of this section.
- (2) Determine gap widths and areas in the primary and secondary seals individually by the following procedures:
- (i) Measure seal gaps, if any, at one or more floating roof levels when the roof is floating off the roof leg supports.
- (ii) Measure seal gaps around the entire circumference of the tank in each place where a 0.32-cm diameter uniform probe passes freely (without forcing or binding against seal) between the seal and the wall of the storage vessel and measure the circumferential distance of each such location.
- (iii) The total surface area of each gap described in paragraph (b)(2)(ii) of this section shall be determined by using probes of various widths to measure accurately the actual distance from the tank wall to the seal and multiplying each such width by its respective circumferential distance.
- (3) Add the gap surface area of each gap location for the primary seal and the secondary seal individually and divide the sum for each seal by the nominal diameter of the tank and compare each ratio to the respective standards in paragraph (b)(4) of this section.
- (4) Make necessary repairs or empty the storage vessel within 45 days of identification in any inspection for seals not meeting the requirements

- listed in (b)(4) (i) and (ii) of this section:
- (i) The accumulated area of gaps between the tank wall and the mechanical shoe or liquid-mounted primary seal shall not exceed 212 Cm<sup>2</sup> per meter of tank diameter, and the width of any portion of any gap shall not exceed 3.81 cm.
- (A) One end of the mechanical shoe is to extend into the stored liquid, and the other end is to extend a minimum vertical distance of 61 cm above the stored liquid surface.
- (B) There are to be no holes, tears, or other openings in the shoe, seal fabric, or seal envelope.
- (ii) The secondary seal is to meet the following requirements:
- (A) The secondary seal is to be installed above the primary seal so that it completely covers the space between the roof edge and the tank wall except as provided in paragraph (b)(2)(iii) of this section.
- (B) The accumulated area of gaps between the tank wall and the secondary seal shall not exceed 21.2 cm<sup>2</sup> per meter of tank diameter, and the width of any portion of any gap shall not exceed 1.27 cm.
- (C) There are to be no holes, tears, or other openings in the seal or seal fabric.
- (iii) If a failure that is detected during inspections required in paragraph (b)(1) of §60.113b(b) cannot be repaired within 45 days and if the vessel cannot be emptied within 45 days, a 30-day extension may be requested from the Administrator in the inspection report required in §60.115b(b)(4). Such extension request must include a demonstration of unavailability of alternate storage capacity and a specification of a schedule that will assure that the control equipment will be repaired or the vessel will be emptied as soon as possible.
- (5) Notify the Administrator 30 days in advance of any gap measurements required by paragraph (b)(1) of this section to afford the Administrator the opportunity to have an observer present.
- (6) Visually inspect the external floating roof, the primary seal, secondary seal, and fittings each time the vessel is emptied and degassed.

- (i) If the external floating roof has defects, the primary seal has holes, tears, or other openings in the seal or the seal fabric, or the secondary seal has holes, tears, or other openings in the seal or the seal fabric, the owner or operator shall repair the items as necessary so that none of the conditions specified in this paragraph exist before filling or refilling the storage vessel with VOL.
- (ii) For all the inspections required by paragraph (b)(6) of this section, the owner or operator shall notify the Administrator in writing at least 30 days prior to the filling or refilling of each storage vessel to afford the Administrator the opportunity to inspect the storage vessel prior to refilling. If the inspection required by paragraph (b)(6) of this section is not planned and the owner or operator could not have known about the inspection 30 days in advance of refilling the tank, the owner or operator shall notify the Administrator at least 7 days prior to the refilling of the storage vessel. Notification shall be made by telephone immediately followed by written documentation demonstrating why the inspection was unplanned. Alternatively, this notification including the written documentation may be made in writing and sent by express mail so that it is received by the Administrator at least 7 days prior to the refilling.
- (c) The owner or operator of each source that is equipped with a closed vent system and control device as required in §60.112b (a)(3) or (b)(2) (other than a flare) is exempt from §60.8 of the General Provisions and shall meet the following requirements.
- (1) Submit for approval by the Administrator as an attachment to the notification required by §60.7(a)(1) or, if the facility is exempt from §60.7(a)(1), as an attachment to the notification required by §60.7(a)(2), an operating plan containing the information listed below.
- (i) Documentation demonstrating that the control device will achieve the required control efficiency during maximum loading conditions. This documentation is to include a description of the gas stream which enters the control device, including flow and VOC content under varying liquid level con-

- ditions (dynamic and static) and manufacturer's design specifications for the control device. If the control device or the closed vent capture system receives vapors, gases, or liquids other than fuels from sources that are not designated sources under this subpart, the efficiency demonstration is to include consideration of all vapors, gases, and liquids received by the closed vent capture system and control device. If an enclosed combustion device with a minimum residence time of 0.75 seconds and a minimum temperature of 816 °C is used to meet the 95 percent requirement, documentation that those conditions will exist is sufficient to meet the requirements of this paragraph.
- (ii) A description of the parameter or parameters to be monitored to ensure that the control device will be operated in conformance with its design and an explanation of the criteria used for selection of that parameter (or parameters).
- (2) Operate the closed vent system and control device and monitor the parameters of the closed vent system and control device in accordance with the operating plan submitted to the Administrator in accordance with paragraph (c)(1) of this section, unless the plan was modified by the Administrator during the review process. In this case, the modified plan applies.
- (d) The owner or operator of each source that is equipped with a closed vent system and a flare to meet the requirements in §60.112b (a)(3) or (b)(2) shall meet the requirements as specified in the general control device requirements, §60.18 (e) and (f).

[52 FR 11429, Apr. 8, 1987, as amended at 54 FR 32973, Aug. 11, 1989]

## § 60.114b Alternative means of emission limitation.

(a) If, in the Administrator's judgment, an alternative means of emission limitation will achieve a reduction in emissions at least equivalent to the reduction in emissions achieved by any requirement in §60.112b, the Administrator will publish in the FEDERAL REGISTER a notice permitting the use of the alternative means for purposes of compliance with that requirement.

### §60.115b

- (b) Any notice under paragraph (a) of this section will be published only after notice and an opportunity for a hearing.
- (c) Any person seeking permission under this section shall submit to the Administrator a written application including:
- (1) An actual emissions test that uses a full-sized or scale-model storage vessel that accurately collects and measures all VOC emissions from a given control device and that accurately simulates wind and accounts for other emission variables such as temperature and barometric pressure.
- (2) An engineering evaluation that the Administrator determines is an accurate method of determining equivalence.
- (d) The Administrator may condition the permission on requirements that may be necessary to ensure operation and maintenance to achieve the same emissions reduction as specified in §60.112b.

## § 60.115b Reporting and recordkeeping requirements.

The owner or operator of each storage vessel as specified in §60.112b(a) shall keep records and furnish reports as required by paragraphs (a), (b), or (c) of this section depending upon the control equipment installed to meet the requirements of §60.112b. The owner or operator shall keep copies of all reports and records required by this section, except for the record required by (c)(1), for at least 2 years. The record required by (c)(1) will be kept for the life of the control equipment.

- (a) After installing control equipment in accordance with §60.112b(a)(1) (fixed roof and internal floating roof), the owner or operator shall meet the following requirements.
- (1) Furnish the Administrator with a report that describes the control equipment and certifies that the control equipment meets the specifications of §60.112b(a)(1) and §60.113b(a)(1). This report shall be an attachment to the notification required by §60.7(a)(3).
- (2) Keep a record of each inspection performed as required by 60.113b(a)(1), (a)(2), (a)(3), and (a)(4). Each record shall identify the storage vessel on which the inspection was performed

- and shall contain the date the vessel was inspected and the observed condition of each component of the control equipment (seals, internal floating roof, and fittings).
- (3) If any of the conditions described in §60.113b(a)(2) are detected during the annual visual inspection required by §60.113b(a)(2), a report shall be furnished to the Administrator within 30 days of the inspection. Each report shall identify the storage vessel, the ature of the defects, and the date the storage vessel was emptied or the nature of and date the repair was made.
- (4) After each inspection required by \$60.113b(a)(3) that finds holes or tears in the seal or seal fabric, or defects in the internal floating roof, or other control equipment defects listed in \$60.113b(a)(3)(ii), a report shall be furnished to the Administrator within 30 days of the inspection. The report shall identify the storage vessel and the reason it did not meet the specifications of \$61.112b(a)(1) or \$60.113b(a)(3) and list each repair made.
- (b) After installing control equipment in accordance with §61.112b(a)(2) (external floating roof), the owner or operator shall meet the following requirements.
- (1) Furnish the Administrator with a report that describes the control equipment and certifies that the control equipment meets the specifications of §60.112b(a)(2) and §60.113b(b)(2), (b)(3), and (b)(4). This report shall be an attachment to the notification required by §60.7(a)(3).
- (2) Within 60 days of performing the seal gap measurements required by §60.113b(b)(1), furnish the Administrator with a report that contains:
  - (i) The date of measurement.
- (ii) The raw data obtained in the measurement.
- (iii) The calculations described in  $\S 60.113b (b)(2)$  and (b)(3).
- (3) Keep a record of each gap measurement performed as required by §60.113b(b). Each record shall identify the storage vessel in which the measurement was performed and shall contain:
  - (i) The date of measurement.
- (ii) The raw data obtained in the measurement.

- (iii) The calculations described in  $\S 60.113b$  (b)(2) and (b)(3).
- (4) After each seal gap measurement that detects gaps exceeding the limitations specified by §60.113b(b)(4), submit a report to the Administrator within 30 days of the inspection. The report will identify the vessel and contain the information specified in paragraph (b)(2) of this section and the date the vessel was emptied or the repairs made and date of repair.
- (c) After installing control equipment in accordance with §60.112b (a)(3) or (b)(1) (closed vent system and control device other than a flare), the owner or operator shall keep the following records.
  - (1) A copy of the operating plan.
- (2) A record of the measured values of the parameters monitored in accordance with §60.113b(c)(2).
- (d) After installing a closed vent system and flare to comply with §60.112b, the owner or operator shall meet the following requirements
- (1) A report containing the measurements required by §60.18(f) (1), (2), (3), (4), (5), and (6) shall be furnished to the Administrator as required by §60.8 of the General Provisions. This report shall be submitted within 6 months of the initial start-up date.
- (2) Records shall be kept of all periods of operation during which the flare pilot flame is absent.
- (3) Semiannual reports of all periods recorded under §60.115b(d)(2) in which the pilot flame was absent shall be furnished to the Administrator.

### § 60.116b Monitoring of operations.

- (a) The owner or operator shall keep copies of all records required by this section, except for the record required by paragraph (b) of this section, for at least 2 years. The record required by paragraph (b) of this section will be kept for the life of the source.
- (b) The owner or operator of each storage vessel as specified in §60.110b(a) shall keep readily accessible records showing the dimension of the storage vessel and an analysis showing the capacity of the storage vessel.
- (c) Except as provided in paragraphs (f) and (g) of this section, the owner or operator of each storage vessel either with a design capacity greater than or

- equal to 151 m³ storing a liquid with a maximum true vapor pressure greater than or equal to 3.5 kPa or with a design capacity greater than or equal to 75 m³ but less than 151 m³ storing a liquid with a maximum true vapor pressure greater than or equal to 15.0 kPa shall maintain a record of the VOL stored, the period of storage, and the maximum true vapor pressure of that VOL during the respective storage period.
- (d) Except as provided in paragraph (g) of this section, the owner or operator of each storage vessel either with a design capacity greater than or equal to 151 m3 storing a liquid with a maximum true vapor pressure that is normally less than 5.2 kPa or with a design capacity greater than or equal to 75 m³ but less than 151 m³ storing a liquid with a maximum true vapor pressure that is normally less than 27.6 kPa shall notify the Administrator within 30 days when the maximum true vapor pressure of the liquid exceeds the respective maximum true vapor vapor pressure values for each volume range.
- (e) Available data on the storage temperature may be used to determine the maximum true vapor pressure as determined below.
- (1) For vessels operated above or below ambient temperatures, the maximum true vapor pressure is calculated based upon the highest expected calendar-month average of the storage temperature. For vessels operated at ambient temperatures, the maximum true vapor pressure is calculated based upon the maximum local monthly average ambient temperature as reported by the National Weather Service.
- (2) For crude oil or refined petroleum products the vapor pressure may be obtained by the following:
- (i) Available data on the Reid vapor pressure and the maximum expected storage temperature based on the highest expected calendar-month average temperature of the stored product may be used to determine the maximum true vapor pressure from nomographs contained in API Bulletin 2517 (incorporated by reference—see §60.17), unless the Administrator specifically requests that the liquid be sampled, the

#### §60.117b

actual storage temperature determined, and the Reid vapor pressure determined from the sample(s).

- (ii) The true vapor pressure of each type of crude oil with a Reid vapor pressure less than 13.8 kPa or with physical properties that preclude determination by the recommended method is to be determined from available data and recorded if the estimated maximum true vapor pressure is greater than 3.5 kPa.
- (3) For other liquids, the vapor pressure:
- (i) May be obtained from standard reference texts, or
- (ii) Determined by ASTM D2879-83, 96, or 97 (incorporated by reference—see §60.17); or
- (iii) Measured by an appropriate method approved by the Administrator; or
- (iv) Calculated by an appropriate method approved by the Administrator.
- (f) The owner or operator of each vessel storing a waste mixture of indeterminate or variable composition shall be subject to the following requirements.
- (1) Prior to the initial filling of the vessel, the highest maximum true vapor pressure for the range of anticipated liquid compositions to be stored will be determined using the methods described in paragraph (e) of this section.
- (2) For vessels in which the vapor pressure of the anticipated liquid composition is above the cutoff for monitoring but below the cutoff for controls as defined in §60.112b(a), an initial physical test of the vapor pressure is required; and a physical test at least once every 6 months thereafter is required as determined by the following methods:
- (i) ASTM D2879-83, 96, or 97 (incorporated by reference—see §60.17); or
- (ii) ASTM D323-82 or 94 (incorporated by reference—see §60.17); or
- (iii) As measured by an appropriate method as approved by the Administrator.
- (g) The owner or operator of each vessel equipped with a closed vent system and control device meeting the specification of §60.112b or with emissions reductions equipment as specified

in 40 CFR 65.42(b)(4), (b)(5), (b)(6), or (c) is exempt from the requirements of paragraphs (c) and (d) of this section.

[52 FR 11429, Apr. 8, 1987, as amended at 65 FR 61756, Oct. 17, 2000; 65 FR 78276, Dec. 14, 2000; 68 FR 59333, Oct. 15, 2003]

### § 60.117b Delegation of authority.

- (a) In delegating implementation and enforcement authority to a State under section 111(c) of the Act, the authorities contained in paragraph (b) of this section shall be retained by the Administrator and not transferred to a State.
- (b) Authorities which will not be delegated to States:  $\S 60.111b(f)(4)$ , 60.114b, 60.116b(e)(3)(iii), 60.116b(e)(3)(iv), and 60.116b(f)(2)(iii).

[52 FR 11429, Apr. 8, 1987, as amended at 52 FR 22780, June 16, 1987]

## Subpart L—Standards of Performance for Secondary Lead Smelters

## § 60.120 Applicability and designation of affected facility.

- (a) The provisions of this subpart are applicable to the following affected facilities in secondary lead smelters: Pot furnaces of more than 250 kg (550 lb) charging capacity, blast (cupola) furnaces, and reverberatory furnaces.
- (b) Any facility under paragraph (a) of this section that commences construction or modification after June 11, 1973, is subject to the requirements of this subpart.

[42 FR 37937, July 25, 1977]

#### § 60.121 Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the Act and in subpart A of this part.

- (a) Reverberatory furnace includes the following types of reverberatory furnaces: stationary, rotating, rocking, and tilting.
- (b) Secondary lead smelter means any facility producing lead from a leadbearing scrap material by smelting to the metallic form.
- (c) Lead means elemental lead or alloys in which the predominant component is lead.

Appendix D 40 CFR Part 60, Subpart GG Standards of Performance for Stationary Gas Turbines

- (3) For thermal and catalytic incinerators, if no such periods as described in paragraphs (c)(1) and (c)(2) of this section occur, the owner or operator shall state this in the report.
- (d) Each owner or operator subject to the provisions of this subpart shall maintain at the source, for a period of at least 2 years, records of all data and calculations used to determine VOC emissions from each affected facility. Where compliance is achieved through the use of thermal incineration, each owner or operator shall maintain, at the source, daily records of the incinerator combustion chamber temperature. If catalytic incineration is used, the owner or operator shall maintain at the source daily records of the gas temperature, both upstream and downstream of the incinerator catalyst bed. Where compliance is achieved through the use of a solvent recovery system. the owner or operator shall maintain at the source daily records of the amount of solvent recovered by the system for each affected facility.

[47 FR 49287, Oct. 29, 1982, as amended at 55 FR 51383, Dec. 13, 1990; 65 FR 61759, Oct. 17, 2000]

#### § 60.316 Test methods and procedures.

- (a) The reference methods in appendix A to this part except as provided under §60.8(b) shall be used to determine compliance with §60.312 as follows:
- (1) Method 24, or coating manufacturer's formulation data, for use in the determination of VOC content of each batch of coating as applied to the surface of the metal parts. In case of an inconsistency between the Method 24 results and the formulation data, the Method 24 results will govern.
- (2) Method 25 for the measurement of VOC concentration.
- (3) Method 1 for sample and velocity traverses.
- (4) Method 2 for velocity and volumetric flow rate.
  - (5) Method 3 for gas analysis.
  - (6) Method 4 for stack gas moisture.
- (b) For Method 24, the coating sample must be at least a 1 liter sample in a 1 liter container taken at a point where the sample will be representative of the coating material as applied to the surface of the metal part.

## 40 CFR Ch. I (7-1-13 Edition)

- (c) For Method 25, the minimum sampling time for each of 3 runs is 60 minutes and the minimum sample volume is 0.003 dry standard cubic meters except that shorter sampling times or smaller volumes, when necessitated by process variables or other factors, may be approved by the Administrator.
- (d) The Administrator will approve testing of representative stacks on a case-by-case basis if the owner or operator can demonstrate to the satisfaction of the Administrator that testing of representative stacks yields results comparable to those that would be obtained by testing all stacks.

## Subpart FF [Reserved]

### Subpart GG—Standards of Performance for Stationary Gas Turbines

# § 60.330 Applicability and designation of affected facility.

- (a) The provisions of this subpart are applicable to the following affected facilities: All stationary gas turbines with a heat input at peak load equal to or greater than 10.7 gigajoules (10 million Btu) per hour, based on the lower heating value of the fuel fired.
- (b) Any facility under paragraph (a) of this section which commences construction, modification, or reconstruction after October 3, 1977, is subject to the requirements of this part except as provided in paragraphs (e) and (j) of \$60.332.

[44 FR 52798, Sept. 10, 1979, as amended at 52 FR 42434, Nov. 5, 1987; 65 FR 61759, Oct. 17, 2000]

## § 60.331 Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the Act and in subpart A of this part.

- (a) Stationary gas turbine means any simple cycle gas turbine, regenerative cycle gas turbine or any gas turbine portion of a combined cycle steam/electric generating system that is not self propelled. It may, however, be mounted on a vehicle for portability.
- (b) Simple cycle gas turbine means any stationary gas turbine which does not

recover heat from the gas turbine exhaust gases to preheat the inlet combustion air to the gas turbine, or which does not recover heat from the gas turbine exhaust gases to heat water or generate steam.

- (c) Regenerative cycle gas turbine means any stationary gas turbine which recovers heat from the gas turbine exhaust gases to preheat the inlet combustion air to the gas turbine.
- (d) Combined cycle gas turbine means any stationary gas turbine which recovers heat from the gas turbine exhaust gases to heat water or generate steam.
- (e) Emergency gas turbine means any stationary gas turbine which operates as a mechanical or electrical power source only when the primary power source for a facility has been rendered inoperable by an emergency situation.
- (f) *Ice fog* means an atmospheric suspension of highly reflective ice crystals.
- (g) ISO standard day conditions means 288 degrees Kelvin, 60 percent relative humidity and 101.3 kilopascals pressure.
- (h) Efficiency means the gas turbine manufacturer's rated heat rate at peak load in terms of heat input per unit of power output based on the lower heating value of the fuel.
- (i) Peak load means 100 percent of the manufacturer's design capacity of the gas turbine at ISO standard day conditions.
- (j) Base load means the load level at which a gas turbine is normally operated.
- (k) Fire-fighting turbine means any stationary gas turbine that is used solely to pump water for extinguishing fires.
- (1) Turbines employed in oil/gas production or oil/gas transportation means any stationary gas turbine used to provide power to extract crude oil/natural gas from the earth or to move crude oil/natural gas, or products refined from these substances through pipelines.
- (m) A Metropolitan Statistical Area or MSA as defined by the Department of Commerce.
- (n) Offshore platform gas turbines means any stationary gas turbine located on a platform in an ocean.

- (o) Garrison facility means any permanent military installation.
- (p) Gas turbine model means a group of gas turbines having the same nominal air flow, combuster inlet pressure, combuster inlet temperature, firing temperature, turbine inlet temperature and turbine inlet pressure.
- (q) Electric utility stationary gas turbine means any stationary gas turbine constructed for the purpose of supplying more than one-third of its potential electric output capacity to any utility power distribution system for sale.
- (r) Emergency fuel is a fuel fired by a gas turbine only during circumstances, such as natural gas supply curtailment or breakdown of delivery system, that make it impossible to fire natural gas in the gas turbine.
- (s) Unit operating hour means a clock hour during which any fuel is combusted in the affected unit. If the unit combusts fuel for the entire clock hour, it is considered to be a full unit operating hour. If the unit combusts fuel for only part of the clock hour, it is considered to be a partial unit operating hour.
- (t) Excess emissions means a specified averaging period over which either:
- (1) The  $NO_X$  emissions are higher than the applicable emission limit in  $\S 60.332;$
- (2) The total sulfur content of the fuel being combusted in the affected facility exceeds the limit specified in §60.333; or
- (3) The recorded value of a particular monitored parameter is outside the acceptable range specified in the parameter monitoring plan for the affected unit.
- (u) Natural gas means a naturally occurring fluid mixture of hydrocarbons (e.g., methane, ethane, or propane) produced in geological formations beneath the Earth's surface that maintains a gaseous state at standard atmospheric temperature and pressure under ordinary conditions. Natural gas contains 20.0 grains or less of total sulfur per 100 standard cubic feet. Equivalents of this in other units are as follows: 0.068 weight percent total sulfur, 680 parts per million by weight (ppmw) total sulfur, and 338 parts per million by volume (ppmv) at 20 degrees Celsius total

sulfur. Additionally, natural gas must either be composed of at least 70 percent methane by volume or have a gross calorific value between 950 and 1100 British thermal units (Btu) per standard cubic foot. Natural gas does not include the following gaseous fuels: landfill gas, digester gas, refinery gas, sour gas, blast furnace gas, coal-derived gas, producer gas, coke oven gas, or any gaseous fuel produced in a process which might result in highly variable sulfur content or heating value.

(v) Duct burner means a device that combusts fuel and that is placed in the exhaust duct from another source, such as a stationary gas turbine, internal combustion engine, kiln, etc., to allow the firing of additional fuel to heat the exhaust gases before the exhaust gases enter a heat recovery steam generating unit.

(w) Lean premix stationary combustion turbine means any stationary combustion turbine where the air and fuel are thoroughly mixed to form a lean mixture for combustion in the combustor. Mixing may occur before or in the combustion chamber. A unit which is capable of operating in both lean premix and diffusion flame modes is considered a lean premix stationary combustion turbine when it is in the lean premix mode, and it is considered a diffusion flame stationary combustion turbine when it is in the diffusion flame mode.

(x) Diffusion flame stationary combustion turbine means any stationary combustion turbine where fuel and air are injected at the combustor and are mixed only by diffusion prior to ignition. A unit which is capable of operating in both lean premix and diffusion flame modes is considered a lean premix stationary combustion turbine when it is in the lean premix mode, and it is considered a diffusion flame stationary combustion turbine when it is in the diffusion flame mode.

(y) Unit operating day means a 24-hour period between 12:00 midnight and the following midnight during which any fuel is combusted at any time in the unit. It is not necessary for fuel to be combusted continuously for the entire 24-hour period.

[44 FR 52798, Sept. 10, 1979, as amended at 47 FR 3770, Jan. 27, 1982; 65 FR 61759, Oct. 17, 2000; 69 FR 41359, July 8, 2004]

#### § 60.332 Standard for nitrogen oxides.

(a) On and after the date on which the performance test required by §60.8 is completed, every owner or operator subject to the provisions of this subpart as specified in paragraphs (b), (c), and (d) of this section shall comply with one of the following, except as provided in paragraphs (e), (f), (g), (h), (i), (j), (k), and (l) of this section.

(1) No owner or operator subject to the provisions of this subpart shall cause to be discharged into the atmosphere from any stationary gas turbine, any gases which contain nitrogen oxides in excess of:

$$STD = 0.0075 \frac{(14.4)}{V} + F$$

where:

STD = allowable ISO corrected (if required as given in §60.335(b)(1)) NO<sub>X</sub> emission concentration (percent by volume at 15 percent oxygen and on a dry basis),

Y = manufacturer's rated heat rate at manufacturer's rated load (kilojoules per watt hour) or, actual measured heat rate based on lower heating value of fuel as measured at actual peak load for the facility. The value of Y shall not exceed 14.4 kilojoules per watt hour, and

F = NO<sub>X</sub> emission allowance for fuel-bound nitrogen as defined in paragraph (a)(4) of this section.

(2) No owner or operator subject to the provisions of this subpart shall cause to be discharged into the atmosphere from any stationary gas turbine, any gases which contain nitrogen oxides in excess of:

$$STD = 0.0150 \frac{(14.4)}{Y} + F$$

where:

STD = allowable ISO corrected (if required as given in \$60.335(b)(1)) NO<sub>X</sub> emission concentration (percent by volume at 15 percent oxygen and on a dry basis).

Y = manufacturer's rated heat rate at manufacturer's rated peak load (kilojoules per watt hour), or actual measured heat rate based on lower heating value of fuel as measured at actual peak load for the facility. The value of Y shall not exceed 14.4 kilojoules per watt hour, and

F = NO<sub>x</sub> emission allowance for fuel-bound nitrogen as defined in paragraph (a)(4) of this section.

### **Environmental Protection Agency**

- (3) The use of F in paragraphs (a)(1) and (2) of this section is optional. That is, the owner or operator may choose to apply a  $NO_X$  allowance for fuel-bound nitrogen and determine the appropriate F-value in accordance with paragraph (a)(4) of this section or may accept an F-value of zero.
- (4) If the owner or operator elects to apply a  $NO_X$  emission allowance for fuel-bound nitrogen, F shall be defined according to the nitrogen content of the fuel during the most recent performance test required under 60.8 as follows:

Fuel-bound nitrogen (percent by weight)	F (NO <sub>x</sub> percent by volume)
N ≤ .015 0.015 <n≤0.1 0.1<n≤0.25 N &gt;0.25</n≤0.25 </n≤0.1 	0 0.04 (N) 0.004+0.0067(N-0.1)

#### Where:

N = the nitrogen content of the fuel (percent by weight).

or

Manufacturers may develop and submit to EPA custom fuel-bound nitrogen allowances for each gas turbine model they manufacture. These fuel-bound nitrogen allowances shall be substantiated with data and must be approved for use by the Administrator before the initial performance test required by \$60.8. Notices of approval of custom fuel-bound nitrogen allowances will be published in the FEDERAL REGISTER.

- (b) Electric utility stationary gas turbines with a heat input at peak load greater than 107.2 gigajoules per hour (100 million Btu/hour) based on the lower heating value of the fuel fired shall comply with the provisions of paragraph (a)(1) of this section.
- (c) Stationary gas turbines with a heat input at peak load equal to or greater than 10.7 gigajoules per hour (10 million Btu/hour) but less than or equal to 107.2 gigajoules per hour (100 million Btu/hour) based on the lower heating value of the fuel fired, shall comply with the provisions of paragraph (a)(2) of this section.
- (d) Stationary gas turbines with a manufacturer's rated base load at ISO conditions of 30 megawatts or less except as provided in §60.332(b) shall com-

ply with paragraph (a)(2) of this section.

- (e) Stationary gas turbines with a heat input at peak load equal to or greater than 10.7 gigajoules per hour (10 million Btwhour) but less than or equal to 107.2 gigajoules per hour (100 million Btwhour) based on the lower heating value of the fuel fired and that have commenced construction prior to October 3, 1982 are exempt from paragraph (a) of this section.
- (f) Stationary gas turbines using water or steam injection for control of  $NO_X$  emissions are exempt from paragraph (a) when ice fog is deemed a traffic hazard by the owner or operator of the gas turbine
- (g) Emergency gas turbines, military gas turbines for use in other than a garrison facility, military gas turbines installed for use as military training facilities, and fire fighting gas turbines are exempt from paragraph (a) of this section.
- (h) Stationary gas turbines engaged by manufacturers in research and development of equipment for both gas turbine emission control techniques and gas turbine efficiency improvements are exempt from paragraph (a) on a case-by-case basis as determined by the Administrator.
- (i) Exemptions from the requirements of paragraph (a) of this section will be granted on a case-by-case basis as determined by the Administrator in specific geographical areas where mandatory water restrictions are required by governmental agencies because of drought conditions. These exemptions will be allowed only while the mandatory water restrictions are in effect.
- (j) Stationary gas turbines with a heat input at peak load greater than 107.2 gigajoules per hour that commenced construction, modification, or reconstruction between the dates of October 3, 1977, and January 27, 1982, and were required in the September 10, 1979, FEDERAL REGISTER (44 FR 52792) to comply with paragraph (a)(1) of this section, except electric utility stationary gas turbines, are exempt from paragraph (a) of this section.
- (k) Stationary gas turbines with a heat input greater than or equal to 10.7 gigajoules per hour (10 million Btu/hour) when fired with natural gas are

exempt from paragraph (a)(2) of this section when being fired with an emergency fuel.

(1) Regenerative cycle gas turbines with a heat input less than or equal to 107.2 gigajoules per hour (100 million Btu/hour) are exempt from paragraph (a) of this section.

[44 FR 52798, Sept. 10, 1979, as amended at 47 FR 3770, Jan. 27, 1982; 65 FR 61759, Oct. 17, 2000; 69 FR 41359, July 8, 2004]

## § 60.333 Standard for sulfur dioxide.

On and after the date on which the performance test required to be conducted by §60.8 is completed, every owner or operator subject to the provision of this subpart shall comply with one or the other of the following conditions:

- (a) No owner or operator subject to the provisions of this subpart shall cause to be discharged into the atmosphere from any stationary gas turbine any gases which contain sulfur dioxide in excess of 0.015 percent by volume at 15 percent oxygen and on a dry basis.
- (b) No owner or operator subject to the provisions of this subpart shall burn in any stationary gas turbine any fuel which contains total sulfur in excess of 0.8 percent by weight (8000 ppmw).

[44 FR 52798, Sept. 10, 1979, as amended at 69 FR 41360, July 8, 2004]

#### § 60.334 Monitoring of operations.

- (a) Except as provided in paragraph (b) of this section, the owner or operator of any stationary gas turbine subject to the provisions of this subpart and using water or steam injection to control  $NO_X$  emissions shall install, calibrate, maintain and operate a continuous monitoring system to monitor and record the fuel consumption and the ratio of water or steam to fuel being fired in the turbine.
- (b) The owner or operator of any stationary gas turbine that commenced construction, reconstruction or modification after October 3, 1977, but before July 8, 2004, and which uses water or steam injection to control NO $_{\rm X}$  emissions may, as an alternative to operating the continuous monitoring system described in paragraph (a) of this section, install, certify, maintain, op-

erate, and quality-assure a continuous emission monitoring system (CEMS) consisting of  $NO_X$  and  $O_2$  monitors. As an alternative, a CO2 monitor may be used to adjust the measured NOx concentrations to 15 percent O2 by either converting the CO2 hourly averages to equivalent O2 concentrations using Equation F-14a or F-14b in appendix F to part 75 of this chapter and making the adjustments to 15 percent  $O_2$ , or by using the CO2 readings directly to make the adjustments, as described in Method 20. If the option to use a CEMS is chosen, the CEMS shall be installed, certified, maintained and operated as follows:

- (1) Each CEMS must be installed and certified according to PS 2 and 3 (for diluent) of 40 CFR part 60, appendix B, except the 7-day calibration drift is based on unit operating days, not calendar days. Appendix F, Procedure 1 is not required. The relative accuracy test audit (RATA) of the NO<sub>X</sub> and diluent monitors may be performed individually or on a combined basis, *i.e.*, the relative accuracy tests of the CEMS may be performed either:
- (i) On a ppm basis (for  $NO_X$ ) and a percent  $O_2$  basis for oxygen; or
- (ii) On a ppm at 15 percent O<sub>2</sub> basis;
- (iii) On a ppm basis (for  $NO_X$ ) and a percent  $CO_2$  basis (for a  $CO_2$  monitor that uses the procedures in Method 20 to correct the  $NO_X$  data to 15 percent  $O_2$ ).
- (2) As specified in §60.13(e)(2), during each full unit operating hour, each monitor must complete a minimum of one cycle of operation (sampling, analyzing, and data recording) for each 15minute quadrant of the hour, to validate the hour. For partial unit operating hours, at least one valid data point must be obtained for each quadrant of the hour in which the unit operates. For unit operating hours in which required quality assurance and maintenance activities are performed on the CEMS, a minimum of two valid data points (one in each of two quadrants) are required to validate the hour.
- (3) For purposes of identifying excess emissions, CEMS data must be reduced to hourly averages as specified in §60.13(h).

### **Environmental Protection Agency**

- (i) For each unit operating hour in which a valid hourly average, as described in paragraph (b)(2) of this section, is obtained for both NOx and diluent, the data acquisition and handling system must calculate and record the hourly NO<sub>x</sub> emissions in the units of the applicable NO<sub>X</sub> emission standard under §60.332(a), i.e., percent NO<sub>X</sub> by volume, dry basis, corrected to 15 percent O2 and International Organization for Standardization (ISO) standard conditions (if required as given in  $\S60.335(b)(1)$ ). For any hour in which the hourly average O2 concentration exceeds 19.0 percent O2, a diluent cap value of 19.0 percent O2 may be used in the emission calculations.
- (ii) A worst case ISO correction factor may be calculated and applied using historical ambient data. For the purpose of this calculation, substitute the maximum humidity of ambient air (Ho), minimum ambient temperature ( $T_a$ ), and minimum combustor inlet absolute pressure ( $P_o$ ) into the ISO correction equation.
- (iii) If the owner or operator has installed a NOx CEMS to meet the requirements of part 75 of this chapter. and is continuing to meet the ongoing requirements of part 75 of this chapter, the CEMS may be used to meet the requirements of this section, except that the missing data substitution methodology provided for at 40 CFR part 75, subpart D. is not required for purposes of identifying excess emissions. Instead, periods of missing CEMS data are to be reported as monitor downtime in the excess emissions and monitoring performance report required in §60.7(c).
- (c) For any turbine that commenced construction, reconstruction or modification after October 3, 1977, but before July 8, 2004, and which does not use steam or water injection to control NO<sub>X</sub> emissions, the owner or operator may, but is not required to, for purposes of determining excess emissions. use a CEMS that meets the requirements of paragraph (b) of this section. Also, if the owner or operator has previously submitted and received EPA, State, or local permitting authority approval of a procedure for monitoring compliance with the applicable NOx emission limit under §60.332, that ap-

proved procedure may continue to be used.

- (d) The owner or operator of any new turbine constructed after July 8, 2004, and which uses water or steam injection to control  $NO_X$  emissions may elect to use either the requirements in paragraph (a) of this section for continuous water or steam to fuel ratio monitoring or may use a  $NO_X$  CEMS installed, certified, operated, maintained, and quality-assured as described in paragraph (b) of this section.
- (e) The owner or operator of any new turbine that commences construction after July 8, 2004, and which does not use water or steam injection to control  $NO_X$  emissions, may, but is not required to, elect to use a  $NO_X$  CEMS installed, certified, operated, maintained, and quality-assured as described in paragraph (b) of this section. Other acceptable monitoring approaches include periodic testing approved by EPA or the State or local permitting authority or continuous parameter monitoring as described in paragraph (f) of this section.
- (f) The owner or operator of a new turbine that commences construction after July 8, 2004, which does not use water or steam injection to control NO<sub>X</sub> emissions may, but is not required to, perform continuous parameter monitoring as follows:
- (1) For a diffusion flame turbine without add-on selective catalytic reduction controls (SCR), the owner or operator shall define at least four parameters indicative of the unit's  $NO_X$  formation characteristics and shall monitor these parameters continuously.
- (2) For any lean premix stationary combustion turbine, the owner or operator shall continuously monitor the appropriate parameters to determine whether the unit is operating in low-NO<sub>x</sub> mode.
- (3) For any turbine that uses SCR to reduce  $NO_X$  emissions, the owner or operator shall continuously monitor appropriate parameters to verify the proper operation of the emission controls.
- (4) For affected units that are also regulated under part 75 of this chapter,

if the owner or operator elects to monitor  $NO_X$  emission rate using the methodology in appendix E to part 75 of this chapter, or the low mass emissions methodology in §75.19 of this chapter, the requirements of this paragraph (f) may be met by performing the parametric monitoring described in section 2.3 of appendix E or in §75.19(c)(1)(iv)(H) of this chapter.

(g) The steam or water to fuel ratio or other parameters that are continuously monitored as described in paragraphs (a), (d) or (f) of this section shall be monitored during the performance test required under §60.8, to establish acceptable values and ranges. The owner or operator may supplement the performance test data with engineering analyses, design specifications, manufacturer's recommendations and other relevant information to define the acceptable parametric ranges more precisely. The owner or operator shall develop and keep on-site a parameter monitoring plan which explains the procedures used to document proper operation of the NO<sub>X</sub> emission controls. The plan shall include the parameter(s) monitored and the acceptable range(s) of the parameter(s) as well as the basis for designating the parameter(s) and acceptable range(s). Any supplemental data such as engineering analyses, design specifications, manufacturer's recommendations and other relevant information shall be included in the monitoring plan. For affected units that are also subject to part 75 of this chapter and that use the low mass emissions methodology in §75.19 of this chapter or the NOx emission measurement methodology in appendix E to part 75, the owner or operator may meet the requirements of this paragraph by developing and keeping onsite (or at a central location for unmanned facilities) a quality-assurance plan, as described in §75.19 (e)(5) or in section 2.3 of appendix E and section 1.3.6 of appendix B to part 75 of this chapter.

- (h) The owner or operator of any stationary gas turbine subject to the provisions of this subpart:
- (1) Shall monitor the total sulfur content of the fuel being fired in the turbine, except as provided in paragraph (h)(3) of this section. The sulfur

content of the fuel must be determined using total sulfur methods described in §60.335(b)(10). Alternatively, if the total sulfur content of the gaseous fuel during the most recent performance test was less than 0.4 weight percent (4000 ppmw), ASTM D4084-82, 94, D5504-01, D6228-98, or Gas Processors Association Standard 2377-86 (all of which are incorporated by reference-see §60.17), which measure the major sulfur compounds may be used; and

- (2) Shall monitor the nitrogen content of the fuel combusted in the turbine, if the owner or operator claims an allowance for fuel bound nitrogen (i.e., if an F-value greater than zero is being or will be used by the owner or operator to calculate STD in §60.332). The nitrogen content of the fuel shall be determined using methods described in §60.335(b)(9) or an approved alternative.
- (3) Notwithstanding the provisions of paragraph (h)(1) of this section, the owner or operator may elect not to monitor the total sulfur content of the gaseous fuel combusted in the turbine, if the gaseous fuel is demonstrated to meet the definition of natural gas in §60.331(u), regardless of whether an existing custom schedule approved by the administrator for subpart GG requires such monitoring. The owner or operator shall use one of the following sources of information to make the required demonstration:
- (i) The gas quality characteristics in a current, valid purchase contract, tariff sheet or transportation contract for the gaseous fuel, specifying that the maximum total sulfur content of the fuel is 20.0 grains/100 scf or less; or
- (ii) Representative fuel sampling data which show that the sulfur content of the gaseous fuel does not exceed 20 grains/100 scf. At a minimum, the amount of fuel sampling data specified in section 2.3.1.4 or 2.3.2.4 of appendix D to part 75 of this chapter is required.
- (4) For any turbine that commenced construction, reconstruction or modification after October 3, 1977, but before July 8, 2004, and for which a custom fuel monitoring schedule has previously been approved, the owner or operator may, without submitting a special petition to the Administrator, continue monitoring on this schedule.

- (i) The frequency of determining the sulfur and nitrogen content of the fuel shall be as follows:
- (1) Fuel oil. For fuel oil, use one of the total sulfur sampling options and the associated sampling frequency described in sections 2.2.3, 2.2.4.1, 2.2.4.2, and 2.2.4.3 of appendix D to part 75 of this chapter (i.e., flow proportional sampling, daily sampling, sampling from the unit's storage tank after each addition of fuel to the tank, or sampling each delivery prior to combining it with fuel oil already in the intended storage tank). If an emission allowance is being claimed for fuel-bound nitrogen, the nitrogen content of the oil shall be determined and recorded once per unit operating day.
- (2) Gaseous fuel. Any applicable nitrogen content value of the gaseous fuel shall be determined and recorded once per unit operating day. For owners and operators that elect not to demonstrate sulfur content using options in paragraph (h)(3) of this section, and for which the fuel is supplied without intermediate bulk storage, the sulfur content value of the gaseous fuel shall be determined and recorded once per unit operating day
- (3) Custom schedules. Notwithstanding the requirements of paragraph (i)(2) of this section, operators or fuel vendors may develop custom schedules for determination of the total sulfur content of gaseous fuels, based on the design and operation of the affected facility and the characteristics of the fuel supply. Except as provided in paragraphs (i)(3)(i) and (i)(3)(ii) of this section, custom schedules shall be substantiated with data and shall be approved by the Administrator before they can be used to comply with the standard in §60.333.
- (i) The two custom sulfur monitoring schedules set forth in paragraphs (i)(3)(i)(A) through (D) and in paragraph (i)(3)(ii) of this section are acceptable, without prior Administrative approval:
- (A) The owner or operator shall obtain daily total sulfur content measurements for 30 consecutive unit operating days, using the applicable methods specified in this subpart. Based on the results of the 30 daily samples, the required frequency for subsequent monitoring of the fuel's total sulfur con-

- tent shall be as specified in paragraph (i)(3)(i)(B), (C), or (D) of this section, as applicable.
- (B) If none of the 30 daily measurements of the fuel's total sulfur content exceeds 0.4 weight percent (4000 ppmw), subsequent sulfur content monitoring may be performed at 12 month intervals. If any of the samples taken at 12month intervals has a total sulfur content between 0.4 and 0.8 weight percent (4000 and 8000 ppmw), follow the procedures in paragraph (i)(3)(i)(C) of this section. If any measurement exceeds 0.8 weight percent (8000 ppmw), follow the procedures in paragraph (i)(3)(i)(D) of this section.
- (C) If at least one of the 30 daily measurements of the fuel's total sulfur content is between 0.4 and 0.8 weight percent (4000 and 8000 ppmw), but none exceeds 0.8 weight percent (8000 ppmw), then:
- (1) Collect and analyze a sample every 30 days for three months. If any sulfur content measurement exceeds 0.8 weight percent (8000 ppmw), follow the procedures in paragraph (i)(3)(i)(D) of this section. Otherwise, follow the procedures in paragraph (i)(3)(i)(C)(2) of this section.
- (2) Begin monitoring at 6-month intervals for 12 months. If any sulfur content measurement exceeds 0.8 weight percent (8000 ppmw), follow the procedures in paragraph (i)(3)(i)(D) of this section. Otherwise, follow the procedures in paragraph (i)(3)(i)(C)(3) of this section.
- (3) Begin monitoring at 12-month intervals. If any sulfur content measurement exceeds 0.8 weight percent (8000 ppmw), follow the procedures in paragraph (i)(3)(i)(D) of this section. Otherwise, continue to monitor at this frequency.
- (D) If a sulfur content measurement exceeds 0.8 weight percent (8000 ppmw), immediately begin daily monitoring according to paragraph (i)(3)(i)(A) of this section. Daily monitoring shall continue until 30 consecutive daily samples, each having a sulfur content no greater than 0.8 weight percent (8000 ppmw), are obtained. At that point, the applicable procedures of paragraph (i)(3)(i)(B) or (C) of this section shall be followed.

- (ii) The owner or operator may use the data collected from the 720-hour sulfur sampling demonstration described in section 2.3.6 of appendix D to part 75 of this chapter to determine a custom sulfur sampling schedule, as follows:
- (A) If the maximum fuel sulfur content obtained from the 720 hourly samples does not exceed 20 grains/100 scf (i.e., the maximum total sulfur content of natural gas as defined in \$60.331(u)), no additional monitoring of the sulfur content of the gas is required, for the purposes of this subpart.
- (B) If the maximum fuel sulfur content obtained from any of the 720 hourly samples exceeds 20 grains/100 scf, but none of the sulfur content values (when converted to weight percent sulfur) exceeds 0.4 weight percent (4000 ppmw), then the minimum required sampling frequency shall be one sample at 12 month intervals.
- (C) If any sample result exceeds 0.4 weight percent sulfur (4000 ppmw), but none exceeds 0.8 weight percent sulfur (8000 ppmw), follow the provisions of paragraph (i)(3)(i)(C) of this section.
- (D) If the sulfur content of any of the 720 hourly samples exceeds 0.8 weight percent (8000 ppmw), follow the provisions of paragraph (i)(3)(i)(D) of this section.
- (j) For each affected unit that elects to continuously monitor parameters or emissions, or to periodically determine the fuel sulfur content or fuel nitrogen content under this subpart, the owner or operator shall submit reports of excess emissions and monitor downtime, in accordance with §60.7(c). Excess emissions shall be reported for all periods of unit operation, including startup, shutdown and malfunction. For the purpose of reports required under §60.7(c), periods of excess emissions and monitor downtime that shall be reported are defined as follows:
  - (1) Nitrogen oxides.
- (i) For turbines using water or steam to fuel ratio monitoring:
- (A) An excess emission shall be any unit operating hour for which the average steam or water to fuel ratio, as measured by the continuous monitoring system, falls below the acceptable steam or water to fuel ratio needed to demonstrate compliance with

- §60.332, as established during the performance test required in §60.8. Any unit operating hour in which no water or steam is injected into the turbine shall also be considered an excess emission.
- (B) A period of monitor downtime shall be any unit operating hour in which water or steam is injected into the turbine, but the essential parametric data needed to determine the steam or water to fuel ratio are unavailable or invalid.
- (C) Each report shall include the average steam or water to fuel ratio, average fuel consumption, ambient conditions (temperature, pressure, and humidity), gas turbine load, and (if applicable) the nitrogen content of the fuel during each excess emission. You do not have to report ambient conditions if you opt to use the worst case ISO correction factor as specified in §60.334(b)(3)(ii), or if you are not using the ISO correction equation under the provisions of §60.335(b)(1).
- (ii) If the owner or operator elects to take an emission allowance for fuel bound nitrogen, then excess emissions and periods of monitor downtime are as described in paragraphs (j)(1)(ii)(A) and (B) of this section.
- (A) An excess emission shall be the period of time during which the fuel-bound nitrogen (N) is greater than the value measured during the performance test required in §60.8 and used to determine the allowance. The excess emission begins on the date and hour of the sample which shows that N is greater than the performance test value, and ends with the date and hour of a subsequent sample which shows a fuel nitrogen content less than or equal to the performance test value.
- (B) A period of monitor downtime begins when a required sample is not taken by its due date. A period of monitor downtime also begins on the date and hour that a required sample is taken, if invalid results are obtained. The period of monitor downtime ends on the date and hour of the next valid sample.
- (iii) For turbines using  $NO_{\mathbf{X}}$  and diluent CEMS:
- (A) An hour of excess emissions shall be any unit operating hour in which

the 4-hour rolling average  $NO_X$  concentration exceeds the applicable emission limit in  $\S60.332(a)(1)$  or (2). For the purposes of this subpart, a "4-hour rolling average  $NO_X$  concentration" is the arithmetic average of the average  $NO_X$  concentration measured by the CEMS for a given hour (corrected to 15 percent  $O_2$  and, if required under  $\S60.335(b)(1)$ , to ISO standard conditions) and the three unit operating hour average  $NO_X$  concentrations immediately preceding that unit operating hour.

- (B) A period of monitor downtime shall be any unit operating hour in which sufficient data are not obtained to validate the hour, for either NO<sub>X</sub> concentration or diluent (or both).
- (C) Each report shall include the ambient conditions (temperature, pressure, and humidity) at the time of the excess emission period and (if the owner or operator has claimed an emission allowance for fuel bound nitrogen) the nitrogen content of the fuel during the period of excess emissions. You do not have to report ambient conditions if you opt to use the worst case ISO correction factor as specified in \$60.334(b)(3)(ii), or if you are not using the ISO correction equation under the provisions of \$60.335(b)(1).
- (iv) For owners or operators that elect, under paragraph (f) of this section, to monitor combustion parameters or parameters that document proper operation of the  $NO_X$  emission controls:
- (A) An excess emission shall be a 4-hour rolling unit operating hour average in which any monitored parameter does not achieve the target value or is outside the acceptable range defined in the parameter monitoring plan for the unit.
- (B) A period of monitor downtime shall be a unit operating hour in which any of the required parametric data are either not recorded or are invalid.
- (2) Sulfur dioxide. If the owner or operator is required to monitor the sulfur content of the fuel under paragraph (h) of this section:
- (i) For samples of gaseous fuel and for oil samples obtained using daily sampling, flow proportional sampling, or sampling from the unit's storage tank, an excess emission occurs each

unit operating hour included in the period beginning on the date and hour of any sample for which the sulfur content of the fuel being fired in the gas turbine exceeds 0.8 weight percent and ending on the date and hour that a subsequent sample is taken that demonstrates compliance with the sulfur limit.

- (ii) If the option to sample each delivery of fuel oil has been selected, the owner or operator shall immediately switch to one of the other oil sampling options (i.e., daily sampling, flow proportional sampling, or sampling from the unit's storage tank) if the sulfur content of a delivery exceeds 0.8 weight percent. The owner or operator shall continue to use one of the other sampling options until all of the oil from the delivery has been combusted, and shall evaluate excess emissions according to paragraph (j)(2)(i) of this section. When all of the fuel from the delivery has been burned, the owner or operator may resume using the as-delivered sampling option.
- (iii) A period of monitor downtime begins when a required sample is not taken by its due date. A period of monitor downtime also begins on the date and hour of a required sample, if invalid results are obtained. The period of monitor downtime shall include only unit operating hours, and ends on the date and hour of the next valid sample.
- (3) Ice fog. Each period during which an exemption provided in \$60.332(f) is in effect shall be reported in writing to the Administrator quarterly. For each period the ambient conditions existing during the period, the date and time the air pollution control system was deactivated, and the date and time the air pollution control system was reactivated shall be reported. All quarterly reports shall be postmarked by the 30th day following the end of each calendar quarter.
- (4) Emergency fuel. Each period during which an exemption provided in §60.332(k) is in effect shall be included in the report required in §60.7(c). For each period, the type, reasons, and duration of the firing of the emergency fuel shall be reported.
- (5) All reports required under §60.7(c) shall be postmarked by the 30th day

following the end of each 6-month period.

[44 FR 52798, Sept. 10, 1979, as amended at 47 FR 3770, Jan. 27, 1982; 65 FR 61759, Oct. 17, 2000; 69 FR 41360, July 8, 2004; 71 FR 9457, Feb. 24, 2006]

## $\S 60.335$ Test methods and procedures.

- (a) The owner or operator shall conduct the performance tests required in §60.8, using either
  - (1) EPA Method 20.
- (2) ASTM D6522-00 (incorporated by reference, see § 60.17), or
- (3) EPA Method 7E and either EPA Method 3 or 3A in appendix A to this part, to determine  $NO_X$  and diluent concentration.
- (4) Sampling traverse points are to be selected following Method 20 or Method 1, (non-particulate procedures) and sampled for equal time intervals. The sampling shall be performed with a traversing single-hole probe or, if feasible, with a stationary multi-hole probe that samples each of the points sequentially. Alternatively, a multi-hole probe designed and documented to sample equal volumes from each hole may be used to sample simultaneously at the required points.
- (5) Notwithstanding paragraph (a)(4) of this section, the owner or operator may test at few points than are specified in Method 1 or Method 20 if the following conditions are met:
- (i) You may perform a stratification test for NOx and diluent pursuant to
  - (A) [Reserved]
- (B) The procedures specified in section 6.5.6.1(a) through (e) appendix A to part 75 of this chapter.
- (ii) Once the stratification sampling is completed, the owner or operator may use the following alternative sample point selection criteria for the performance test:
- (A) If each of the individual traverse point NO<sub>v</sub> concentrations, normalized to 15 percent  $O_2$ , is within  $\pm 10$  percent of the mean normalized concentration for all traverse points, then you may use 3 points (located either 16.7, 50.0, and 83.3 percent of the way across the stack or duct, or, for circular stacks or ducts greater than 2.4 meters (7.8 feet) in diameter, at 0.4, 1.2, and 2.0 meters from the wall). The 3 points shall be located along the measurement line that

exhibited the highest average normalized NOx concentration during the stratification test; or

- (B) If each of the individual traverse point NOx concentrations, normalized to 15 percent O2, is within ±5 percent of the mean normalized concentration for all traverse points, then you may sample at a single point, located at least 1 meter from the stack wall or at the stack centroid.
- (6) Other acceptable alternative reference methods and procedures are given in paragraph (c) of this section.
- (b) The owner or operator shall determine compliance with the applicable nitrogen oxides emission limitation in §60.332 and shall meet the performance test requirements of § 60.8 as follows:
- (1) For each run of the performance test, the mean nitrogen oxides emission concentration (NO<sub>Xo</sub>) corrected to 15 percent O2 shall be corrected to ISO standard conditions using the following equation. Notwithstanding this requirement, use of the ISO correction equation is optional for: Lean premix stationary combustion turbines; units used in association with heat recovery steam generators (HRSG) equipped with duct burners; and units equipped with add-on emission control devices:

 $NO_X = (NO_{Xo})(P_r/P_o)^{0.5}$   $e^{19}$  (Ho - 0.00633) (288)  $^{\circ} \text{K/T}_{a})^{1.53}$ 

Where:

 $NO_x$  = emission concentration of  $NO_x$  at 15 percent O2 and ISO standard ambient conditions, ppm by volume, dry basis,

 $NO_{Xo}$  = mean observed  $NO_{X}$  concentration, ppm by volume, dry basis, at 15 percent Ο2,

- Pr = reference combustor inlet absolute pressure at 101.3 kilopascals ambient pressure, mm Hg.
- $P_o$  = observed combustor inlet absolute pressure at test, mm Hg,
- = observed humidity of ambient air g H<sub>2</sub>O/g air.
- e = transcendental constant, 2.718, and
- Ta = ambient temperature, °K.
- (2) The 3-run performance test required by §60.8 must be performed within ±5 percent at 30, 50, 75, and 90to-100 percent of peak load or at four evenly-spaced load points in the normal operating range of the gas turbine, including the minimum point in the operating range and 90-to-100 percent of peak load, or at the highest achievable

load point if 90-to-100 percent of peak load cannot be physically achieved in practice. If the turbine combusts both oil and gas as primary or backup fuels, separate performance testing is required for each fuel. Notwithstanding these requirements, performance testing is not required for any emergency fuel (as defined in §60.331).

- (3) For a combined cycle turbine system with supplemental heat (duct burner), the owner or operator may elect to measure the turbine  $NO_X$  emissions after the duct burner rather than directly after the turbine. If the owner or operator elects to use this alternative sampling location, the applicable  $NO_X$  emission limit in §60.332 for the combustion turbine must still be met.
- (4) If water or steam injection is used to control  $NO_X$  with no additional post-combustion  $NO_X$  control and the owner or operator chooses to monitor the steam or water to fuel ratio in accordance with  $\S60.334(a)$ , then that monitoring system must be operated concurrently with each EPA Method 20, ASTM D6522-00 (incorporated by reference, see  $\S60.17$ ), or EPA Method 7E run and shall be used to determine the fuel consumption and the steam or water to fuel ratio necessary to comply with the applicable  $\S60.332$   $NO_X$  emission limit.
- (5) If the owner operator elects to claim an emission allowance for fuel bound nitrogen as described in §60.332, then concurrently with each reference method run, a representative sample of the fuel used shall be collected and analyzed, following the applicable procedures described in §60.335(b)(9). These data shall be used to determine the maximum fuel nitrogen content for which the established water (or steam) to fuel ratio will be valid.
- (6) If the owner or operator elects to install a CEMS, the performance evaluation of the CEMS may either be conducted separately (as described in paragraph (b)(7) of this section) or as part of the initial performance test of the affected unit.
- (7) If the owner or operator elects to install and certify a  $NO_X$  CEMS under  $\S60.334(e)$ , then the initial performance test required under  $\S60.8$  may be done in the following alternative manner:

- (i) Perform a minimum of 9 reference method runs, with a minimum time per run of 21 minutes, at a single load level, between 90 and 100 percent of peak (or the highest physically achievable) load.
- (ii) Use the test data both to demonstrate compliance with the applicable  $NO_X$  emission limit under §60.332 and to provide the required reference method data for the RATA of the CEMS described under §60.334(b).
- (iii) The requirement to test at three additional load levels is waived.
- (8) If the owner or operator elects under  $\S 60.334(f)$  to monitor combustion parameters or parameters indicative of proper operation of  $NO_X$  emission controls, the appropriate parameters shall be continuously monitored and recorded during each run of the initial performance test, to establish acceptable operating ranges, for purposes of the parameter monitoring plan for the affected unit, as specified in  $\S 60.334(g)$ .
- (9) To determine the fuel bound nitrogen content of fuel being fired (if an emission allowance is claimed for fuel bound nitrogen), the owner or operator may use equipment and procedures meeting the requirements of:
- (i) For liquid fuels, ASTM D2597-94 (Reapproved 1999), D6366-99, D4629-02, D5762-02 (all of which are incorporated by reference, see §60.17); or
- (ii) For gaseous fuels, shall use analytical methods and procedures that are accurate to within 5 percent of the instrument range and are approved by the Administrator.
- (10) If the owner or operator is required under §60.334(i)(1) or (3) to periodically determine the sulfur content of the fuel combusted in the turbine, a minimum of three fuel samples shall be collected during the performance test. Analyze the samples for the total sulfur content of the fuel using:
- (i) For liquid fuels, ASTM D129-00, D2622-98, D4294-02, D1266-98, D5453-00 or D1552-01 (all of which are incorporated by reference, see §60.17); or
- (ii) For gaseous fuels, ASTM D1072-80, 90 (Reapproved 1994); D3246-81, 92, 96; D4468-85 (Reapproved 2000); or D6667-01 (all of which are incorporated by reference, see §60.17). The applicable

ranges of some ASTM methods mentioned above are not adequate to measure the levels of sulfur in some fuel gases. Dilution of samples before analysis (with verification of the dilution ratio) may be used, subject to the prior approval of the Administrator.

- (11) The fuel analyses required under paragraphs (b)(9) and (b)(10) of this section may be performed by the owner or operator, a service contractor retained by the owner or operator, the fuel vendor, or any other qualified agency.
- (c) The owner or operator may use the following as alternatives to the reference methods and procedures specified in this section:
- (1) Instead of using the equation in paragraph (b)(1) of this section, manufacturers may develop ambient condition correction factors to adjust the nitrogen oxides emission level measured by the performance test as provided in §60.8 to ISO standard day conditions.

[69 FR 41363, July 8, 2004, as amended at 71 FR 9458, Feb. 24, 2006]

## Subpart HH—Standards of Performance for Lime Manufacturing Plants

Source: 49 FR 18080, Apr. 26, 1984, unless otherwise noted.

# § 60.340 Applicability and designation of affected facility.

- (a) The provisions of this subpart are applicable to each rotary lime kiln used in the manufacture of lime.
- (b) The provisions of this subpart are not applicable to facilities used in the manufacture of lime at kraft pulp mills.
- (c) Any facility under paragraph (a) of this section that commences construction or modification after May 3, 1977, is subject to the requirements of this subpart.

## § 60.341 Definitions.

As used in this subpart, all terms not defined herein shall have the same meaning given them in the Act and in the General Provisions.

(a) Lime manufacturing plant means any plant which uses a rotary lime kiln to produce lime product from limestone by calcination.

- (b) Lime product means the product of the calcination process including, but not limited to, calcitic lime, dolomitic lime, and dead-burned dolomite.
- (c) Positive-pressure fabric filter means a fabric filter with the fans on the upstream side of the filter bags.
- (d) Rotary lime kiln means a unit with an inclined rotating drum that is used to produce a lime product from limestone by calcination.
- (e) Stone feed means limestone feedstock and millscale or other iron oxide additives that become part of the product.

## § 60.342 Standard for particulate matter.

- (a) On and after the date on which the performance test required to be conducted by §60.8 is completed, no owner or operator subject to the provisions of this subpart shall cause to be discharged into the atmosphere from any rotary lime kiln any gases which:
- (1) Contain particulate matter in excess of 0.30 kilogram per megagram (0.60 lb/ton) of stone feed.
- (2) Exhibit greater than 15 percent opacity when exiting from a dry emission control device.

## § 60.343 Monitoring of emissions and operations.

- (a) The owner or operator of a facility that is subject to the provisions of this subpart shall install, calibrate, maintain, and operate a continuous monitoring system, except as provided in paragraphs (b) and (c) of this section, to monitor and record the opacity of a representative portion of the gases discharged into the atmosphere from any rotary lime kiln. The span of this system shall be set at 40 percent opacity.
- (b) The owner or operator of any rotary lime kiln having a control device with a multiple stack exhaust or a roof monitor may, in lieu of the continuous opacity monitoring requirement of \$60.343(a), monitor visible emissions at least once per day of operation by using a certified visible emissions observer who, for each site where visible emissions are observed, will perform three Method 9 tests and record the results. Visible emission observations shall occur during normal operation of

Appendix E

40 CFR Part 60, Subpart UU

Standards of Performance for Asphalt Processing and Asphalt Roofing Manufacture

- (c) Following the initial performance test, the owner or operator of an affected facility shall identify, record, and submit a written report to the Administrator every calendar quarter of each instance in which the volume-weighted average of the local mass of VOC's emitted to the atmosphere per volume of applied coating solids (N) is greater than the limit specified under \$60.462. If no such instances have occurred during a particular quarter, a report stating this shall be submitted to the Administrator semiannually.
- (d) The owner or operator of each affected facility shall also submit reports at the frequency specified in §60.7(c) when the incinerator temperature drops as defined under §60.464(c). If no such periods occur, the owner or operator shall state this in the report.
- (e) Each owner or operator subject to the provisions of this subpart shall maintain at the source, for a period of at least 2 years, records of all data and calculations used to determine monthly VOC emissions from each affected facility and to determine the monthly emission limit, where applicable. Where compliance is achieved through the use of thermal incineration, each owner or operator shall maintain, at the source, daily records of the incinerator combustion temperature. If catalytic incineration is used, the owner or operator shall maintain at the source daily records of the gas temperature, both upstream and downstream of the incinerator catalyst bed.

[47 FR 49612, Nov. 1, 1982, as amended at 55 FR 51383, Dec. 13, 1990; 56 FR 20497, May 3, 1991; 65 FR 61761, Oct. 17, 2000]

#### § 60.466 Test methods and procedures.

- (a) The reference methods in appendix A to this part, except as provided under \$60.8(b), shall be used to determine compliance with \$60.462 as follows:
- (1) Method 24, or data provided by the formulator of the coating, shall be used for determining the VOC content of each coating as applied to the surface of the metal coil. In the event of a dispute, Method 24 shall be the reference method. When VOC content of waterborne coatings, determined by Method 24, is used to determine compliance of affected facilities, the results of the

### 40 CFR Ch. I (7-1-13 Edition)

Method 24 analysis shall be adjusted as described in Section 12.6 of Method 24;

- (2) Method 25, both for measuring the VOC concentration in each gas stream entering and leaving the control device on each stack equipped with an emission control device and for measuring the VOC concentration in each gas stream emitted directly to the atmosphere;
- (3) Method 1 for sample and velocity traverses:
- (4) Method 2 for velocity and volumetric flow rate:
  - (5) Method 3 for gas analysis; and
  - (6) Method 4 for stack gas moisture.
- (b) For Method 24, the coating sample must be at least a 1-liter sample taken at a point where the sample will be representative of the coating as applied to the surface of the metal coil.
- (c) For Method 25, the sampling time for each of three runs is to be at least 60 minutes, and the minimum sampling volume is to be at least 0.003 dscm (0.11 dscf); however, shorter sampling times or smaller volumes, when necessitated by process variables or other factors, may be approved by the Administrator.
- (d) The Administrator will approve testing of representative stacks on a case-by-case basis if the owner or operator can demonstrate to the satisfaction of the Administrator that testing of representative stacks yields results comparable to those that would be obtained by testing all stacks.

[47 FR 49612, Nov. 1, 1982, as amended at 51 FR 22938, June 24, 1986; 65 FR 61761, Oct. 17, 2000]

## Subpart UU—Standards of Performance for Asphalt Processing and Asphalt Roofing Manufacture

SOURCE: 47 FR 34143, Aug. 6, 1982, unless otherwise noted.

# § 60.470 Applicability and designation of affected facilities.

(a) The affected facilities to which this subpart applies are each saturator and each mineral handling and storage facility at asphalt roofing plants; and each asphalt storage tank and each blowing still at asphalt processing plants, petroleum refineries, and asphalt roofing plants.

(b) Any saturator or mineral handling and storage facility under paragraph (a) of this section that commences construction or modification after November 18, 1980, is subject to the requirements of this subpart. Any asphalt storage tank or blowing still that processes and/or stores asphalt used for roofing only or for roofing and other purposes, and that commences construction or modification after November 18, 1980, is subject to the requirements of this subpart.

Any asphalt storage tank or blowing still that processes and/or stores only nonroofing asphalts and that commences construction or modification after May 26, 1981, is subject to the requirements of this subpart.

#### § 60.471 Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the Act and in subpart A of this part.

Afterburner (A/B) means an exhaust gas incinerator used to control emissions of particulate matter.

Asphalt processing means the storage and blowing of asphalt.

Asphalt processing plant means a plant which blows asphalt for use in the manufacture of asphalt products.

Asphalt roofing plant means a plant which produces asphalt roofing products (shingles, roll roofing, siding, or saturated felt).

Asphalt storage tank means any tank used to store asphalt at asphalt roofing plants, petroleum refineries, and asphalt processing plants. Storage tanks containing cutback asphalts (asphalts diluted with solvents to reduce viscosity for low temperature applications) and emulsified asphalts (asphalts dispersed in water with an emulsifying agent) are not subject to this regulation.

Blowing still means the equipment in which air is blown through asphalt flux to change the softening point and penetration rate.

Catalyst means a substance which, when added to asphalt flux in a blowing still, alters the penetrating-softening point relationship or increases the rate of oxidation of the flux.

Coating blow means the process in which air is blown through hot asphalt flux to produce coating asphalt. The coating blow starts when the air is turned on and stops when the air is turned off.

Electrostatic precipitator (ESP) means an air pollution control device in which solid or liquid particulates in a gas stream are charged as they pass through an electric field and precipitated on a collection suface.

High velocity air filter (HVAF) means an air pollution control filtration device for the removal of sticky, oily, or liquid aerosol particulate matter from exhaust gas streams.

Mineral handling and storage facility means the areas in asphalt roofing plants in which minerals are unloaded from a carrier, the conveyor transfer points between the carrier and the storage silos, and the storage silos.

Saturator means the equipment in which asphalt is applied to felt to make asphalt roofing products. The term saturator includes the saturator, wet looper, and coater.

[47 FR 34143, Aug. 6, 1982, as amended at 65 FR 61762, Oct. 17, 2000]

## § 60.472 Standards for particulate matter.

- (a) On and after the date on which §60.8(b) requires a performance test to be completed, no owner or operator subject to the provisions of this subpart shall cause to be discharged into the atmosphere from any saturator:
  - (1) Particulate matter in excess of:
- (i) 0.04 kg/Mg (0.08 lb/ton) of asphalt shingle or mineral-surfaced roll roofing produced, or
- (ii) 0.04 kg/Mg (0.08 lb/ton) of saturated felt or smooth-surfaced roll roofing produced:
- (2) Exhaust gases with opacity greater than 20 percent; and
- (3) Any visible emissions from a saturator capture system for more than 20 percent of any period of consecutive valid observations totaling 60 minutes. Saturators that were constructed before November 18, 1980, and that have not been reconstructed since that date and that become subject to these standards through modification are exempt from the visible emissions standard. Saturators that have been newly

constructed or reconstructed since November 18, 1980 are subject to the visible emissions standard.

- (b) On and after the date on which §60.8(b) requires a performance test to be completed, no owner or operator subject to the provisions of this subject shall cause to be discharged into the atmosphere from any blowing still:
- (1) Particulate matter in excess of 0.67 kg/Mg (1.3 lb/ton) of asphalt charged to the still when a catalyst is added to the still: and
- (2) Particulate matter in excess of 0.71 kg/Mg (1.4 lb/ton) of asphalt charged to the still when a catalyst is added to the still and when No. 6 fuel oil is fired in the afterburner; and
- (3) Particulate matter in excess of 0.60 kg/Mg (1.2 lb/ton) of asphalt charged to the still during blowing without a catalyst; and
- (4) Particulate matter in excess of 0.64 kg/Mg (1.3 lb/ton) of asphalt charged to the still during blowing without a catalyst and when No. 6 fuel oil is fired in the afterburner; and
- (5) Exhaust gases with an opacity greater than 0 percent unless an opacity limit for the blowing still when fuel oil is used to fire the afterburner has been established by the Administrator in accordance with the procedures in §60.474(g).
- (c) Within 60 days after achieving the maximum production rate at which the affected facility will be operated, but not later than 180 days after initial startup of such facility, no owner or operator subject to the provisions of this subpart shall cause to be discharged into the atmosphere from any asphalt storage tank exhaust gases with opacity greater than 0 percent, except for one consecutive 15-minute period in any 24-hour period when the transfer lines are being blown for clearing. The control device shall not be bypassed during this 15-minute period. If, however, the emissions from any asphalt storage tank(s) are ducted to a control device for a saturator, the combined emissions shall meet the emission limit contained in paragraph (a) of this section during the time the saturator control device is operating. At any other time the asphalt storage tank(s) must meet the opacity limit specified above for storage tanks.

(d) Within 60 days after achieving the maximum production rate at which the affected facility will be operated, but not later than 180 days after initial startup of such facility, no owner or operator subject to the provisions of this subpart shall cause to be discharged into the atmosphere from any mineral handling and storage facility emissions with opacity greater than 1 percent.

[47 FR 34143, Aug. 6, 1982, as amended at 65 FR 61762, Oct. 17, 2000]

## § 60.473 Monitoring of operations.

- (a) The owner or operator subject to the provisions of this subpart, and using either an electrostatic precipitator or a high velocity air filter to meet the emission limit in  $\S 60.472(a)(1)$  and/or (b)(1) shall continuously monitor and record the temperature of the gas at the inlet of the control device. The temperature monitoring instrument shall have an accuracy of  $\pm 15$  °C ( $\pm 25$  °F) over its range.
- (b) The owner or operator subject to the provisions of this subpart and using an afterburner to meet the emission limit in 60.472(a)(1) and/or (b)(1) shall continuously monitor and record the temperature in the combustion zone of the afterburner. The monitoring instrument shall have an accuracy of  $\pm 10$  °C ( $\pm 18$  °F) over its range.
- (c) An owner or operator subject to the provisions of this subpart and using a control device not mentioned in paragraphs (a) or (b) of this section shall provide to the Administrator information describing the operation of the control device and the process parameter(s) which would indicate proper operation and maintenance of the device. The Administrator may require continuous monitoring and will determine the process parameters to be monitored.
- (d) The industry is exempted from the quarterly reports required under §60.7(c). The owner/operator is required to record and report the operating temperature of the control device during the performance test and, as required

## **Environmental Protection Agency**

by §60.7(d), maintain a file of the temperature monitoring results for at least two years.

[47 FR 34143, Aug. 6, 1982, as amended at 65 FR 61762, Oct. 17, 2000]

#### § 60.474 Test methods and procedures.

- (a) For saturators, the owner or operator shall conduct performance tests required in §60.8 as follows:
- (1) If the final product is shingle or mineral-surfaced roll roofing, the tests shall be conducted while 106.6-kg (235-1b) shingle is being produced.
- (2) If the final product is saturated felt or smooth-surfaced roll roofing, the tests shall be conducted while 6.8-kg (15-lb) felt is being produced.
- (3) If the final product is fiberglass shingle, the test shall be conducted while a nominal 100-kg (220-lb) shingle is being produced.
- (b) In conducting the performance tests required in §60.8, the owner or operator shall use as reference methods and procedures the test methods in appendix A of this part or other methods and procedures as specified in this section, except as provided in §60.8(b).
- (c) The owner or operator shall determine compliance with the particulate matter standards in §60.472 as follows:
- (1) The emission rate (E) of particulate matter shall be computed for each run using the following equation:

 $E=(c_s Q_{sd})/(PK)$ 

where:

E=emission rate of particulate matter, kg/Mg (lb/ton).

 $c_s$ =concentration of particulate matter, g/dscm (gr/dscf).

 $Q_{sd}$ =volumetric flow rate of effluent gas, dscm/hr (dscf/hr).

P=asphalt roofing production rate or asphalt charging rate, Mg/hr (ton/hr).

K=conversion factor, 1000 g/kg [7000 (gr/lb)].

(2) Method 5A shall be used to determine the particulate matter concentration  $(c_s)$  and volumetric flow rate  $(Q_{sd})$  of the effluent gas. For a saturator, the sampling time and sample volume for each run shall be at least 120 minutes and 3.00 dscm (106 dscf), and for the blowing still, at least 90 minutes or the duration of the coating blow or noncoating blow, whichever is greater, and 2.25 dscm (79.4 dscf).

- (3) For the saturator, the asphalt roofing production rate (P) for each run shall be determined as follows: The amount of asphalt roofing produced on the shingle or saturated felt process lines shall be obtained by direct measurement. The asphalt roofing production rate is the amount produced divided by the time taken for the run.
- (4) For the blowing still, the asphalt charging rate (P) shall be computed for each run using the following equation:

 $P=(Vd)/(K'\theta)$ 

where:

P=asphalt charging rate to blowing still, Mg/hr (ton/hr).

V=volume of asphalt charged, m3 (ft3).

d=density of asphalt, kg/m³ (lb/ft³).

K'=conversion factor, 1000 kg/Mg (2000 lb/ton).

 $\theta$ =duration of test run, hr.

- (i) The volume (V) of asphalt charged shall be measured by any means accurate to within 10 percent.
- (ii) The density (d) of the asphalt shall be computed using the following equation:

$$d = K_1 - K_2 T_i$$

Where:

d = Density of the asphalt,  $kg/m^3$  ( $lb/ft^3$ )

 $K_1 = 1056.1 \text{ kg/m}^3 \text{ (metric units)}$ 

= 64.70 lb/ft3 (English Units)

 $K_2 = 0.6176 \text{ kg/(m}^3 \text{ °C) (metric units)}$ 

= 0.0694 lb/(ft $^3$  °F) (English Units)

 $T_i$  = temperature at the start of the blow, °C (( °deg;F)

- (5) Method 9 and the procedures in §60.11 shall be used to determine opacity.
- (d) The Administrator will determine compliance with the standards in §60.472(a)(3) by using Method 22, modified so that readings are recorded every 15 seconds for a period of consecutive observations during representative conditions (in accordance with §60.8(c)) totaling 60 minutes. A performance test shall consist of one run.
- (e) The owner or operator shall use the monitoring device in §60.473 (a) or (b) to monitor and record continuously the temperature during the particulate matter run and shall report the results to the Administrator with the performance test results.

- (f) If at a later date the owner or operator believes that the emission limits in §60.472(a) and (b) are being met even though one of the conditions listed in this paragraph exist, he may submit a written request to the Administrator to repeat the performance test and procedure outlined in paragraph (c) of this section.
- (1) The temperature measured in accordance with §60.473(a) is exceeding that measured during the performance test.
- (2) The temperature measured in accordance with §60.473(b) is lower than that measured during the performance test.
- (g) If fuel oil is to be used to fire an afterburner used to control emissions from a blowing still, the owner or operator may petition the Administrator in accordance with §60.11(e) of the General Provisions to establish an opacity standard for the blowing still that will be the opacity standard when fuel oil is used to fire the afterburner. To obtain this opacity standard, the owner or operator must request the Administrator to determine opacity during an initial, or subsequent, performance test when fuel oil is used to fire the afterburner. Upon receipt of the results of the performance test, the Administrator will make a finding concerning compliance with the mass standard for the blowing still. If the Administrator finds that the facility was in compliance with the mass standard during the performance test but failed to meet the zero opacity standard, the Administrator will establish and promulgate in the FEDERAL REGISTER an opacity standard for the blowing still that will be the opacity standard when fuel oil is used to fire the afterburner. When the afterburner is fired with natural gas, the zero percent opacity remains the applicable opacity standard.
- [54 FR 6677, Feb. 14, 1989, as amended 54 FR 27016, June 27, 1989; 65 FR 61762, Oct. 17, 2000]

Subpart VV—Standards of Performance for Equipment Leaks of VOC in the Synthetic Organic Chemicals Manufacturing Industry for which Construction, Reconstruction, or Modification Commenced After January 5, 1981, and on or Before November 7, 2006

SOURCE: 48 FR 48335, Oct. 18, 1983, unless otherwise noted.

# § 60.480 Applicability and designation of affected facility.

- (a)(1) The provisions of this subpart apply to affected facilities in the synthetic organic chemicals manufacturing industry.
- (2) The group of all equipment (defined in §60.481) within a process unit is an affected facility.
- (b) Any affected facility under paragraph (a) of this section that commences construction, reconstruction, or modification after January 5, 1981, and on or before November 7, 2006, shall be subject to the requirements of this subpart.
- (c) Addition or replacement of equipment for the purpose of process improvement which is accomplished without a capital expenditure shall not by itself be considered a modification under this subpart.
- (d)(1) If an owner or operator applies for one or more of the exemptions in this paragraph, then the owner or operator shall maintain records as required in §60.486(i).
- (2) Any affected facility that has the design capacity to produce less than 1,000 Mg/yr (1,102 ton/yr) of a chemical listed in §60.489 is exempt from §§60.482-1 through 60.482-10.
- (3) If an affected facility produces heavy liquid chemicals only from heavy liquid feed or raw materials, then it is exempt from §§ 60.482-1 through 60.482-10.
- (4) Any affected facility that produces beverage alcohol is exempt from §§ 60.482–1 through 60.482–10.
- (5) Any affected facility that has no equipment in volatile organic compounds (VOC) service is exempt from §§ 60.482-1 through 60.482-10.

Appendix F 40 CFR Part 60, Subpart GGGa

Standards of Performance for Equipment Leaks for VOC in Petroleum Refineries for Which Construction, Reconstruction, or Modification Commenced After November 7, 2006

## §60.590a

- (ii) If an owner or operator determines that a piece of equipment is in hydrogen service, the determination can be revised only after following the procedures in paragraph (b)(2).
- (c) Any existing reciprocating compressor that becomes an affected facility under provisions of §60.14 or §60.15 is exempt from §60.482-3(a), (b), (c), (d), (e), and (h) provided the owner or operator demonstrates that recasting the distance piece or replacing the compressor are the only options available to bring the compressor into compliance with the provisions of §60.482-3(a), (b), (c), (d), (e), and (h).
- (d) An owner or operator may use the following provision in addition to  $\S60.485$ (e): Equipment is in light liquid service if the percent evaporated is greater than 10 percent at 150 °C as determined by ASTM Method D86-78, 82, 90, 95, or 96 (incorporated by reference as specified in  $\S60.17$ ).
- (e) Pumps in light liquid service and valves in gas/vapor and light liquid service within a process unit that is located in the Alaskan North Slope are exempt from the requirements of §60.482–2 and §60.482–7.
- (f) Open-ended valves or lines containing asphalt as defined in §60.591 are exempt from the requirements of §60.482-6(a) through (c).

[49 FR 22606, May 30, 1984, as amended at 65 FR 61768, Oct. 17, 2000; 72 FR 64896, Nov. 16, 2007]

Subpart GGGa—Standards of Performance for Equipment Leaks of VOC in Petroleum Refineries for Which Construction, Reconstruction, or Modification Commenced After November 7, 2006

SOURCE: 72 FR 64896, Nov. 16, 2007, unless otherwise noted.

## § 60.590a Applicability and designation of affected facility.

- (a)(1) The provisions of this subpart apply to affected facilities in petroleum refineries.
- (2) A compressor is an affected facility.

### 40 CFR Ch. I (7-1-13 Edition)

- (3) The group of all the equipment (defined in §60.591a) within a process unit is an affected facility.
- (b) Any affected facility under paragraph (a) of this section that commences construction, reconstruction, or modification after November 7, 2006, is subject to the requirements of this subpart.
- (c) Addition or replacement of equipment (defined in §60.591a) for the purpose of process improvement which is accomplished without a capital expenditure shall not by itself be considered a modification under this subpart.
- (d) Facilities subject to subpart VV, subpart VVa, subpart GGG, or subpart KKK of this part are excluded from this subpart.
- (e) Stay of standards. Owners or operators are not required to comply with the definition of "process unit" in §60.590 of this subpart until the EPA takes final action to require compliance and publishes a document in the FEDERAL REGISTER. While the definition of "process unit" is stayed, owners or operators should use the following definition:

Process unit means components assembled to produce intermediate or final products from petroleum, unfinished petroleum derivatives, or other intermediates; a process unit can operate independently if supplied with sufficient feed or raw materials and sufficient storage facilities for the product.

[49 FR 22606, May 30, 1984, as amended at 73 FR 31376, June 2, 2008]

### § 60.591a Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the Clean Air Act, in subpart A of part 60, or in subpart VVa of this part, and the following terms shall have the specific meanings given them.

Alaskan North Slope means the approximately 69,000 square mile area extending from the Brooks Range to the Arctic Ocean.

Asphalt (also known as Bitumen) is a black or dark brown solid or semi-solid thermo-plastic material possessing waterproofing and adhesive properties. It is a complex combination of higher molecular weight organic compounds containing a relatively high proportion of

#### **Environmental Protection Agency**

hydrocarbons having carbon numbers greater than C25 with a high carbon to hydrogen ratio. It is essentially non-volatile at ambient temperatures with closed cup flash point of 445 °F (230 °C) or greater.

Equipment means each valve, pump, pressure relief device, sampling connection system, open-ended valve or line, and flange or other connector in VOC service. For the purposes of recordkeeping and reporting only, compressors are considered equipment.

In hydrogen service means that a compressor contains a process fluid that meets the conditions specified in \$60.593a(b).

In light liquid service means that the piece of equipment contains a liquid that meets the conditions specified in §60.593a(c).

Petroleum means the crude oil removed from the earth and the oils derived from tar sands, shale, and coal.

Petroleum refinery means any facility engaged in producing gasoline, kerosene, distillate fuel oils, residual fuel oils, lubricants, or other products through the distillation of petroleum, or through the redistillation, cracking, or reforming of unfinished petroleum derivatives.

Process unit means the components assembled and connected by pipes or ducts to process raw materials and to produce intermediate or final products from petroleum, unfinished petroleum derivatives, or other intermediates. A process unit can operate independently if supplied with sufficient feed or raw materials and sufficient storage facilities for the product. For the purpose of this subpart, process unit includes any feed, intermediate and final product storage vessels (except as specified in §60.482-1a(g)), product transfer racks, and connected ducts and piping. A process unit includes all equipment as defined in this subpart.

EFFECTIVE DATE NOTE: At 73 FR 31376, June 2, 2008, §60.591a, the definition of "process unit" was stayed until further notice.

## § 60.592a Standards.

(a) Each owner or operator subject to the provisions of this subpart shall comply with the requirements of §§ 60.482-1a to 60.482-10a as soon as prac-

ticable, but no later than 180 days after initial startup.

- (b) For a given process unit, an owner or operator may elect to comply with the requirements of paragraphs (b)(1), (2), or (3) of this section as an alternative to the requirements in §60.482–7a.
  - (1) Comply with §60.483-1a.
  - (2) Comply with §60.483-2a.
- (3) Comply with the Phase III provisions in §63.168, except an owner or operator may elect to follow the provisions in §60.482-7a(f) instead of §63.168 for any valve that is designated as being leakless.
- (c) An owner or operator may apply to the Administrator for a determination of equivalency for any means of emission limitation that achieves a reduction in emissions of VOC at least equivalent to the reduction in emissions of VOC achieved by the controls required in this subpart. In doing so, the owner or operator shall comply with requirements of §60.484a.
- (d) Each owner or operator subject to the provisions of this subpart shall comply with the provisions of §60.485a except as provided in §60.593a.
- (e) Each owner or operator subject to the provisions of this subpart shall comply with the provisions of §§ 60.486a and 60.487a.

## § 60.593a Exceptions.

- (a) Each owner or operator subject to the provisions of this subpart may comply with the following exceptions to the provisions of subpart VVa of this part.
- (b)(1) Compressors in hydrogen service are exempt from the requirements of §60.592a if an owner or operator demonstrates that a compressor is in hydrogen service.
- (2) Each compressor is presumed not to be in hydrogen service unless an owner or operator demonstrates that the piece of equipment is in hydrogen service. For a piece of equipment to be considered in hydrogen service, it must be determined that the percent hydrogen content can be reasonably expected always to exceed 50 percent by volume. For purposes of determining the percent hydrogen content in the process fluid that is contained in or contacts a compressor, procedures that conform

to the general method described in ASTM E260-73, 91, or 96, E168-67, 77, or 92, or E169-63, 77, or 93 (incorporated by reference as specified in §60.17) shall be used

(3)(i) An owner or operator may use engineering judgment rather than procedures in paragraph (b)(2) of this section to demonstrate that the percent content exceeds 50 percent by volume, provided the engineering judgment demonstrates that the content clearly exceeds 50 percent by volume. When an owner or operator and the Administrator do not agree on whether a piece of equipment is in hydrogen service, however, the procedures in paragraph (b)(2) of this section shall be used to resolve the disagreement.

- (ii) If an owner or operator determines that a piece of equipment is in hydrogen service, the determination can be revised only after following the procedures in paragraph (b)(2).
- (c) Any existing reciprocating compressor that becomes an affected facility under provisions of §60.14 or §60.15 is exempt from §60.482–3a(a), (b), (c), (d), (e), and (h) provided the owner or operator demonstrates that recasting the distance piece or replacing the compressor are the only options available to bring the compressor into compliance with the provisions of §60.482–3a(a), (b), (c), (d), (e), and (h).
- (d) An owner or operator may use the following provision in addition to §60.485a(e): Equipment is in light liquid service if the percent evaporated is greater than 10 percent at 150 °C as determined by ASTM Method D86-78, 82, 90, 93, 95, or 96 (incorporated by reference as specified in §60.17).
- (e) Pumps in light liquid service and valves in gas/vapor and light liquid service within a process unit that is located in the Alaskan North Slope are exempt from the requirements of §§ 60.482-2a and 60.482-7a.
- (f) Open-ended valves or lines containing asphalt as defined in §60.591a are exempt from the requirements of §60.482-6a(a) through (c).
- (g) Connectors in gas/vapor or light liquid service are exempt from the requirements in §60.482-11a, provided the owner or operator complies with §60.482-8a for all connectors, not just those in heavy liquid service.

#### Subpart HHH—Standards of Performance for Synthetic Fiber Production Facilities

SOURCE: 49 FR 13651, Apr. 5, 1984, unless otherwise noted.

## § 60.600 Applicability and designation of affected facility.

- (a) Except as provided in paragraph (b) of this section, the affected facility to which the provisions of this subpart apply is each solvent-spun synthetic fiber process that produces more than 500 Mg (551 ton) of fiber per year.
- (b) The provisions of this subpart do not apply to any facility that uses the reaction spinning process to produce spandex fiber or the viscose process to produce rayon fiber.
- (c) The provisions of this subpart apply to each facility as identified in paragraph (a) of this section and that commences construction or reconstruction after November 23, 1982. The provisions of this subpart do not apply to facilities that commence modification but not reconstruction after November 23, 1982.

[49 FR 22606, May 30, 1984, as amended at 65 FR 61768, Oct. 17, 2000]

#### § 60.601 Definitions.

All terms that are used in this subpart and are not defined below are given the same meaning as in the Act and in subpart A of this part.

Acrylic fiber means a manufactured synthetic fiber in which the fiber-forming substance is any long-chain synthetic polymer composed of at least 85 percent by weight of acrylonitrile units.

Makeup solvent means the solvent introduced into the affected facility that compensates for solvent lost from the affected facility during the manufacturing process.

Nongaseous losses means the solvent that is not volatilized during fiber production, and that escapes the process and is unavailable for recovery, or is in a form or concentration unsuitable for economical recovery.

Polymer means any of the natural or synthetic compounds of usually high molecular weight that consist of many

Appendix G
40 CFR Part 60, Subpart QQQ
Standards of Performance for COC Emissions From Petroleum Refinery Wastewater Systems



#### **Environmental Protection Agency**

within the State will be relieved of the obligation to comply with this section, provided that they comply with the requirements established by the State.

#### § 60.685 Test methods and procedures.

- (a) In conducting the performance tests required in §60.8, the owner or operator shall use as reference methods and procedures the test methods in appendix A of this part or other methods and procedures as specified in this section, except as provided in §60.8(b).
- (b) The owner or operator shall conduct performance tests while the product with the highest loss on ignition (LOI) expected to be produced by the affected facility is being manufactured.
- (c) The owner or operator shall determine compliance with the particulate matter standard in §60.682 as follows:
- (1) The emission rate (E) of particulate matter shall be computed for each run using the following equation:

 $E = (C_t \ Q_{sd})/(P_{avg} \ K)$ 

where:

- E = emission rate of particulate matter, kg/Mg (lb/ton).
- C<sub>1</sub> = concentration of particulate matter, g/dscm (gr/dscf).
- Q<sub>sd</sub> = volumetric flow rate of effluent gas, dsem/hr (dsef/hr).
- $\begin{array}{l} dscm/hr \; (dscf/hr). \\ P_{avg} = average \; glass \; pull \; rate, \; Mg/hr \; (ton/hr). \\ K = 1,000 \; g/kg \; (7,000 \; gr/lb). \end{array}$
- (2) Method 5E shall be used to determine the particulate matter concentration ( $C_t$ ) and the volumetric flow rate ( $Q_{sd}$ ) of the effluent gas. The sampling time and sample volume shall be at least 120 minutes and 2.55 dscm (90.1
- dscf). (3) The average glass pull rate  $(P_{avg})$  for the manufacturing line shall be the arithmetic average of three glass pull rate  $(P_i)$  determinations taken at intervals of at least 30 minutes during each run.

The individual glass pull rates  $(P_i)$  shall be computed using the following equation:

 $P_i = K' L_s W_m M [1.0 - (LOI/100)]$ 

where:

P<sub>i</sub>=glass pull rate at interval "i", Mg/hr (ton/hr).

L<sub>s</sub>=line speed, m/min (ft/min). W<sub>m</sub>=trimmed mat width, m (ft). M=mat gram weight, g/m<sup>2</sup> (lb/ft<sup>2</sup>). LOI=loss on ignition, weight percent K'=conversion factor,  $6\times10^{-5}$  (min-Mg)/(hr-g)  $[3\times10^{-2}$  (min-ton)/(hr-lb)].

- (i) ASTM D2584-68 (Reapproved 1985) or 94 (incorporated by reference—see §60.17), shall be used to determine the LOI for each run.
- (ii) Line speed  $(L_x)$ , trimmed mat width  $(W_m)$ , and mat gram weight (M) shall be determined for each run from the process information or from direct measurements.
- (d) To comply with §60.684(d), the owner or operator shall record measurements as required in §60.684 (a) and (b) using the monitoring devices in §60.683 (a) and (b) during the particulate matter runs.

[54 FR 6680, Feb. 14, 1989, as amended at 65 FR 61778, Oct. 17, 2000]

#### Subpart QQQ—Standards of Performance for VOC Emissions From Petroleum Refinery Wastewater Systems

SOURCE: 53 FR 47623, Nov. 23, 1988, unless otherwise noted.

## § 60.690 Applicability and designation of affected facility.

- (a)(1) The provisions of this subpart apply to affected facilities located in petroleum refineries for which construction, modification, or reconstruction is commenced after May 4, 1987.
- (2) An individual drain system is a separate affected facility.
- (3) An oil-water separator is a separate affected facility.
- (4) An aggregate facility is a separate affected facility.
- (b) Notwithstanding the provisions of 40 CFR 60.14(e)(2), the construction or installation of a new individual drain system shall constitute a modification to an affected facility described in \$60.690(a)(4). For purposes of this paragraph, a new individual drain system shall be limited to all process drains and the first common junction box.

#### § 60.691 Definitions.

As used in this subpart, all terms not defined herein shall have the meaning given them in the Act or in subpart A of 40 CFR part 60, and the following terms shall have the specific meanings given them.

#### § 60.691

Active service means that a drain is receiving refinery wastewater from a process unit that will continuously maintain a water seal.

Aggregate facility means an individual drain system together with ancillary downstream sewer lines and oil-water separators, down to and including the secondary oil-water separator, as applicable.

Catch basin means an open basin which serves as a single collection point for stormwater runoff received directly from refinery surfaces and for refinery wastewater from process drains.

Closed vent system means a system that is not open to the atmosphere and that is composed of piping, connections, and, if necessary, flow-inducing devices that transport gas or vapor from an emission source to a control device. If gas or vapor from regulated equipment are routed to a process (e.g., to a petroleum refinery fuel gas system), the process shall not be considered a closed vent system and is not subject to the closed vent system standards.

Completely closed drain system means an individual drain system that is not open to the atmosphere and is equipped and operated with a closed vent system and control device complying with the requirements of §60.692-5.

Control device means an enclosed combustion device, vapor recovery system or flare.

Fixed roof means a cover that is mounted to a tank or chamber in a stationary manner and which does not move with fluctuations in wastewater levels.

Floating roof means a pontoon-type or double-deck type cover that rests on the liquid surface.

Gas-tight means operated with no detectable emissions.

Individual drain system means all process drains connected to the first common downstream junction box. The term includes all such drains and common junction box, together with their associated sewer lines and other junction boxes, down to the receiving oilwater separator.

Junction box means a manhole or access point to a wastewater sewer system line.

No detectable emissions means less than 500 ppm above background levels, as measured by a detection instrument in accordance with Method 21 in appendix A of 40 CFR part 60.

Non-contact cooling water system means a once-through drain, collection and treatment system designed and operated for collecting cooling water which does not come into contact with hydrocarbons or oily wastewater and which is not recirculated through a cooling tower.

Oil-water separator means wastewater treatment equipment used to separate oil from water consisting of a separation tank, which also includes the forebay and other separator basins, skimmers, weirs, grit chambers, and sludge hoppers. Slop oil facilities, including tanks, are included in this term along with storage vessels and auxiliary equipment located between individual drain systems and the oil-water separator. This term does not include storage vessels or auxiliary equipment which do not come in contact with or store oily wastewater.

Oily wastewater means wastewater generated during the refinery process which contains oil, emulsified oil, or other hydrocarbons. Oily wastewater originates from a variety of refinery processes including cooling water, condensed stripping steam, tank draw-off, and contact process water.

Petroleum means the crude oil removed from the earth and the oils derived from tar sands, shale, and coal.

Petroleum refinery means any facility engaged in producing gasoline, kerosene, distillate fuel oils, residual fuel oils, lubricants, or other products through the distillation of petroleum, or through the redistillation of petroleum, cracking, or reforming unfinished petroleum derivatives.

Sewer line means a lateral, trunk line, branch line, ditch, channel, or other conduit used to convey refinery wastewater to downstream components of a refinery wastewater treatment system. This term does not include buried, below-grade sewer lines.

Slop oil means the floating oil and solids that accumulate on the surface of an oil-water separator.

Storage vessel means any tank, reservoir, or container used for the storage of petroleum liquids, including oily wastewater.

Stormwater sewer system means a drain and collection system designed and operated for the sole purpose of collecting stormwater and which is segregated from the process wastewater collection system.

Wastewater system means any component, piece of equipment, or installation that receives, treats, or processes oily wastewater from petroleum refinery process units.

Water seal controls means a seal pot, p-leg trap, or other type of trap filled with water that has a design capability to create a water barrier between the sewer and the atmosphere.

[53 FR 47623, Nov. 23, 1985, as amended at 60 FR 43259, Aug. 18, 1995]

#### § 60.692-1 Standards: General.

- (a) Each owner or operator subject to the provisions of this subpart shall comply with the requirements of  $\S 60.692-1$  to 60.692-5 and with  $\S 60.693-1$  and 60.693-2, except during periods of startup, shutdown, or malfunction.
- (b) Compliance with §§ 60.692-1 to 60.692-5 and with §§ 60.693-1 and 60.693-2 will be determined by review of records and reports, review of performance test results, and inspection using the methods and procedures specified in § 60.696.
- (c) Permission to use alternative means of emission limitation to meet the requirements of §§60.692-2 through 60.692-4 may be granted as provided in §60.694.
- (d)(1) Stormwater sewer systems are not subject to the requirements of this subpart.
- (2) Ancillary equipment, which is physically separate from the wastewater system and does not come in contact with or store oily wastewater, is not subject to the requirements of this subpart.
- (3) Non-contact cooling water systems are not subject to the requirements of this subpart.
- (4) An owner or operator shall demonstrate compliance with the exclusions in paragraphs (d)(1), (2), and (3) of this section as provided in §60.697 (h), (i), and (j).

## § 60.692-2 Standards: Individual drain systems.

- (a)(1) Each drain shall be equipped with water seal controls.
- (2) Each drain in active service shall be checked by visual or physical inspection initially and monthly thereafter for indications of low water levels or other conditions that would reduce the effectiveness of the water seal controls.
- (3) Except as provided in paragraph (a)(4) of this section, each drain out of active service shall be checked by visual or physical inspection initially and weekly thereafter for indications of low water levels or other problems that could result in VOC emissions.
- (4) As an alternative to the requirements in paragraph (a)(3) of this section, if an owner or operator elects to install a tightly sealed cap or plug over a drain that is out of service, inspections shall be conducted initially and semiannually to ensure caps or plugs are in place and properly installed.
- (5) Whenever low water levels or missing or improperly installed caps or plugs are identified, water shall be added or first efforts at repair shall be made as soon as practicable, but not later than 24 hours after detection, except as provided in §60.692-6.
- (b)(1) Junction boxes shall be equipped with a cover and may have an open vent pipe. The vent pipe shall be at least 90 cm (3 ft) in length and shall not exceed 10.2 cm (4 in) in diameter.
- (2) Junction box covers shall have a tight seal around the edge and shall be kept in place at all times, except during inspection and maintenance.
- (3) Junction boxes shall be visually inspected initially and semiannually thereafter to ensure that the cover is in place and to ensure that the cover has a tight seal around the edge.
- (4) If a broken seal or gap is identified, first effort at repair shall be made as soon as practicable, but not later than 15 calendar days after the broken seal or gap is identified, except as provided in § 60.692-6.
- (c)(1) Sewer lines shall not be open to the atmosphere and shall be covered or enclosed in a manner so as to have no visual gaps or cracks in joints, seals, or other emission interfaces.

#### § 60.692-3

- (2) The portion of each unburied sewer line shall be visually inspected initially and semiannually thereafter for indication of cracks, gaps, or other problems that could result in VOC emissions.
- (3) Whenever cracks, gaps, or other problems are detected, repairs shall be made as soon as practicable, but not later than 15 calendar days after identification, except as provided in \$60.692-6.
- (d) Except as provided in paragraph (e) of this section, each modified or reconstructed individual drain system that has a catch basin in the existing configuration prior to May 4, 1987 shall be exempt from the provisions of this section.
- (e) Refinery wastewater routed through new process drains and a new first common downstream junction box, either as part of a new individual drain system or an existing individual drain system, shall not be routed through a downstream catch basin.

### § 60.692-3 Standards: Oil-water separators.

- (a) Each oil-water separator tank, slop oil tank, storage vessel, or other auxiliary equipment subject to the requirements of this subpart shall be equipped and operated with a fixed roof, which meets the following specifications, except as provided in paragraph (d) of this section or in §60.693-2.
- (1) The fixed roof shall be installed to completely cover the separator tank, slop oil tank, storage vessel, or other auxiliary equipment with no separation between the roof and the wall.
- (2) The vapor space under a fixed roof shall not be purged unless the vapor is directed to a control device.
- (3) If the roof has access doors or openings, such doors or openings shall be gasketed, latched, and kept closed at all times during operation of the separator system, except during inspection and maintenance.
- (4) Roof seals, access doors, and other openings shall be checked by visual inspection initially and semiannually thereafter to ensure that no cracks or gaps occur between the roof and wall and that access doors and other openings are closed and gasketed properly.

- (5) When a broken seal or gasket or other problem is identified, first efforts at repair shall be made as soon as practicable, but not later than 15 calendar days after it is identified, except as provided in §60.692-6.
- (b) Each oil-water separator tank or auxiliary equipment with a design capacity to treat more than 16 liters per second (250 gallons per minute (gpm)) of refinery wastewater shall, in addition to the requirements in paragraph (a) of this section, be equipped and operated with a closed vent system and control device, which meet the requirements of §60.692-5, except as provided in paragraph (c) of this section or in §60.693-2.
- (c)(1) Each modified or reconstructed oil-water separator tank with a maximum design capacity to treat less than 38 liters per second (600 gpm) of refinery wastewater which was equipped and operated with a fixed roof covering the entire separator tank or a portion of the separator tank prior to May 4, 1987 shall be exempt from the requirements of paragraph (b) of this section, but shall meet the requirements of paragraph (a) of this section, or may elect to comply with paragraph (c)(2) of this section.
- (2) The owner or operator may elect to comply with the requirements of paragraph (a) of this section for the existing fixed roof covering a portion of the separator tank and comply with the requirements for floating roofs in §60.693-2 for the remainder of the separator tank
- (d) Storage vessels, including slop oil tanks and other auxiliary tanks that are subject to the standards in §§ 60.112, 60.112a, and 60.112b and associated requirements, 40 CFR part 60, subparts K, Ka, or Kb are not subject to the requirements of this section.
- (e) Slop oil from an oil-water separator tank and oily wastewater from slop oil handling equipment shall be collected, stored, transported, recycled, reused, or disposed of in an enclosed system. Once slop oil is returned to the process unit or is disposed of, it is no longer within the scope of this subpart. Equipment used in handling slop oil shall be equipped with a fixed roof meeting the requirements of paragraph (a) of this section.

(f) Each oil-water separator tank, slop oil tank, storage vessel, or other auxiliary equipment that is required to comply with paragraph (a) of this section, and not paragraph (b) of this section, may be equipped with a pressure control valve as necessary for proper system operation. The pressure control valve shall be set at the maximum pressure necessary for proper system operation, but such that the value will not vent continuously.

[53 FR 47623, Nov. 23, 1985, as amended at 60 FR 43259, Aug. 18, 1995; 65 FR 61778, Oct. 17, 20001

### § 60.692-4 Standards: Aggregate facility.

A new, modified, or reconstructed aggregate facility shall comply with the requirements of §§ 60.692–2 and 60.692–3.

### § 60.692-5 Standards: Closed vent systems and control devices.

- (a) Enclosed combustion devices shall be designed and operated to reduce the VOC emissions vented to them with an efficiency of 95 percent or greater or to provide a minimum residence time of 0.75 seconds at a minimum temperature of 816 °C (1,500 °F).
- (b) Vapor recovery systems (for example, condensers and adsorbers) shall be designed and operated to recover the VOC emissions vented to them with an efficiency of 95 percent or greater.
- (c) Flares used to comply with this subpart shall comply with the requirements of 40 CFR 60.18.
- (d) Closed vent systems and control devices used to comply with provisions of this subpart shall be operated at all times when emissions may be vented to them.
- (e)(1) Closed vent systems shall be designed and operated with no detectable emissions, as indicated by an instrument reading of less than 500 ppm above background, as determined during the initial and semiannual inspections by the methods specified in \$60.696.
- (2) Closed vent systems shall be purged to direct vapor to the control device.
- (3) A flow indicator shall be installed on a vent stream to a control device to ensure that the vapors are being routed to the device.

- (4) All gauging and sampling devices shall be gas-tight except when gauging or sampling is taking place.
- (5) When emissions from a closed system are detected, first efforts at repair to eliminate the emissions shall be made as soon as practicable, but not later than 30 calendar days from the date the emissions are detected, except as provided in §60.692-6.

#### § 60.692-6 Standards: Delay of repair.

- (a) Delay of repair of facilities that are subject to the provisions of this subpart will be allowed if the repair is technically impossible without a complete or partial refinery or process unit shutdown
- (b) Repair of such equipment shall occur before the end of the next refinery or process unit shutdown.

### § 60.692-7 Standards: Delay of compliance.

- (a) Delay of compliance of modified individual drain systems with ancillary downstream treatment components will be allowed if compliance with the provisions of this subpart cannot be achieved without a refinery or process unit shutdown.
- (b) Installation of equipment necessary to comply with the provisions of this subpart shall occur no later than the next scheduled refinery or process unit shutdown.

### § 60.693-1 Alternative standards for individual drain systems.

- (a) An owner or operator may elect to construct and operate a completely closed drain system.
- (b) Each completely closed drain system shall be equipped and operated with a closed vent system and control device complying with the requirements of §60.692-5.
- (c) An owner or operator must notify the Administrator in the report required in 40 CFR 60.7 that the owner or operator has elected to construct and operate a completely closed drain system.
- (d) If an owner or operator elects to comply with the provisions of this section, then the owner or operator does not need to comply with the provisions of §60.692-2 or §60.694.

#### § 60.693-2

- (e)(1) Sewer lines shall not be open to the atmosphere and shall be covered or enclosed in a manner so as to have no visual gaps or cracks in joints, seals, or other emission interfaces.
- (2) The portion of each unburied sewer line shall be visually inspected initially and semiannually thereafter for indication of cracks, gaps, or other problems that could result in VOC emissions.
- (3) Whenever cracks, gaps, or other problems are detected, repairs shall be made as soon as practicable, but not later than 15 calendar days after identification, except as provided in §60.692-6.

## § 60.693-2 Alternative standards for oil-water separators.

- (a) An owner or operator may elect to construct and operate a floating roof on an oil-water separator tank, slop oil tank, storage vessel, or other auxiliary equipment subject to the requirements of this subpart which meets the following specifications.
- (1) Each floating roof shall be equipped with a closure device between the wall of the separator and the roof edge. The closure device is to consist of a primary seal and a secondary seal.
- (i) The primary seal shall be a liquidmounted seal or a mechanical shoe seal
- (A) A liquid-mounted seal means a foam- or liquid-filled seal mounted in contact with the liquid between the wall of the separator and the floating roof. A mechanical shoe seal means a metal sheet held vertically against the wall of the separator by springs or weighted levers and is connected by braces to the floating roof. A flexible coated fabric (envelope) spans the anular space between the metal sheet and the floating roof.
- (B) The gap width between the primary seal and the separator wall shall not exceed 3.8 cm (1.5 in.) at any point.
- (C) The total gap area between the primary seal and the separator wall shall not exceed 67 cm<sup>2</sup>/m (3.2 in.<sup>2</sup>/ft) of separator wall perimeter.
- (ii) The secondary seal shall be above the primary seal and cover the annular space between the floating roof and the wall of the separator.

- (A) The gap width between the secondary seal and the separator wall shall not exceed 1.3 cm (0.5 in.) at any point.
- (B) The total gap area between the secondary seal and the separator wall shall not exceed 6.7 cm<sup>2</sup>/m (0.32 in.<sup>2</sup>/ft) of separator wall perimeter.
- (iii) The maximum gap width and total gap area shall be determined by the methods and procedures specified in §60.696(d).
- (A) Measurement of primary seal gaps shall be performed within 60 calendar days after initial installation of the floating roof and introduction of refinery wastewater and once every 5 years thereafter.
- (B) Measurement of secondary seal gaps shall be performed within 60 calendar days of initial introduction of refinery wastewater and once every year thereafter.
- (iv) The owner or operator shall make necessary repairs within 30 calendar days of identification of seals not meeting the requirements listed in paragraphs (a)(1) (i) and (ii) of this section.
- (2) Except as provided in paragraph (a)(4) of this section, each opening in the roof shall be equipped with a gasketed cover, seal, or lid, which shall be maintained in a closed position at all times, except during inspection and maintenance.
- (3) The roof shall be floating on the liquid (i.e., off the roof supports) at all times except during abnormal conditions (i.e., low flow rate).
- (4) The floating roof may be equipped with one or more emergency roof drains for removal of stormwater. Each emergency roof drain shall be fitted with a slotted membrane fabric cover that covers at least 90 percent of the drain opening area or a flexible fabric sleeve seal.
- (5)(i) Access doors and other openings shall be visually inspected initially and semiannually thereafter to ensure that there is a tight fit around the edges and to identify other problems that could result in VOC emissions.
- (ii) When a broken seal or gasket on an access door or other opening is identified, it shall be repaired as soon as

practicable, but not later than 30 calendar days after it is identified, except as provided in §60.692-6.

- (b) An owner or operator must notify the Administrator in the report required by 40 CFR 60.7 that the owner or operator has elected to construct and operate a floating roof under paragraph (a) of this section.
- (c) For portions of the oil-water separator tank where it is infeasible to construct and operate a floating roof, such as the skimmer mechanism and weirs, a fixed roof meeting the requirements of §60.692–3(a) shall be installed.
- (d) Except as provided in paragraph (c) of this section, if an owner or operator elects to comply with the provisions of this section, then the owner or operator does not need to comply with the provisions of §§60.692-3 or 60.694 applicable to the same facilities.

[53 FR 47623, Nov. 23, 1985, as amended at 60 FR 43259, Aug. 18, 1995]

## § 60.694 Permission to use alternative means of emission limitation.

- (a) If, in the Administrator's judgment, an alternative means of emission limitation will achieve a reduction in VOC emissions at least equivalent to the reduction in VOC emissions achieved by the applicable requirement in §60.692, the Administrator will publish in the FEDERAL REGISTER a notice permitting the use of the alternative means for purposes of compliance with that requirement. The notice may condition the permission on requirements related to the operation and maintenance of the alternative means.
- (b) Any notice under paragraph (a) of this section shall be published only after notice and an opportunity for a hearing.
- (c) Any person seeking permission under this section shall collect, verify, and submit to the Administrator information showing that the alternative means achieves equivalent emission reductions.

#### § 60.695 Monitoring of operations.

(a) Each owner or operator subject to the provisions of this subpart shall install, calibrate, maintain, and operate according to manufacturer's specifications the following equipment, unless alternative monitoring procedures or requirements are approved for that facility by the Administrator.

- (1) Where a thermal incinerator is used for VOC emission reduction, a temperature monitoring device equipped with a continuous recorder shall be used to measure the temperature of the gas stream in the combustion zone of the incinerator. The temperature monitoring device shall have an accuracy of  $\pm 1$  percent of the temperature being measured, expressed in °C, or  $\pm 0.5$  °C (0.9 °F), whichever is greater.
- (2) Where a catalytic incinerator is used for VOC emission reduction, temperature monitoring devices, each equipped with a continuous recorder shall be used to measure the temperature in the gas stream immediately before and after the catalyst bed of the incinerator. The temperature monitoring devices shall have an accuracy of ±1 percent of the temperature being measured, expressed in °C, or ±0.5 °C (0.9 °F), whichever is greater.
- (3) Where a carbon adsorber is used for VOC emissions reduction, a monitoring device that continuously indicates and records the VOC concentration level or reading of organics in the exhaust gases of the control device outlet gas stream or inlet and outlet gas stream shall be used.
- (i) For a carbon adsorption system that regenerates the carbon bed directly onsite, a monitoring device that continuously indicates and records the volatile organic compound concentration level or reading of organics in the exhaust gases of the control device outlet gas stream or inlet and outlet gas stream shall be used.
- (ii) For a carbon adsorption system that does not regenerate the carbon bed directly onsite in the control device (e.g., a carbon canister), the concentration level of the organic compounds in the exhaust vent stream from the carbon adsorption system shall be monitored on a regular schedule, and the existing carbon shall be replaced with fresh carbon immediately when carbon breakthrough is indicated. The device shall be monitored on a daily basis or at intervals no greater than 20 percent of the design carbon replacement interval, whichever is greater. As an alternative to

#### § 60.696

conducting this monitoring, an owner or operator may replace the carbon in the carbon adsorption system with fresh carbon at a regular predetermined time interval that is less than the carbon replacement interval that is determined by the maximum design flow rate and organic concentration in the gas stream vented to the carbon adsorption system.

- (4) Where a flare is used for VOC emission reduction, the owner or operator shall comply with the monitoring requirements of 40 CFR 60.18(f)(2).
- (b) Where a VOC recovery device other than a carbon adsorber is used to meet the requirements specified in \$60.692-5(a), the owner or operator shall provide to the Administrator information describing the operation of the control device and the process parameter(s) that would indicate proper operation and maintenance of the device. The Administrator may request further information and will specify appropriate monitoring procedures or requirements.
- (c) An alternative operational or process parameter may be monitored if it can be demonstrated that another parameter will ensure that the control device is operated in conformance with these standards and the control device's design specifications.

[53 FR 47623, Nov. 23, 1985, as amended at 60 FR 43259, Aug. 18, 1995; 65 FR 61778, Oct. 17, 2000]

## § 60.696 Performance test methods and procedures and compliance provisions.

- (a) Before using any equipment installed in compliance with the requirements of §60.692-2, §60.692-3, §60.692-4, §60.692-5, or §60.693, the owner or operator shall inspect such equipment for indications of potential emissions, defects, or other problems that may cause the requirements of this subpart not to be met. Points of inspection shall include, but are not limited to, seals, flanges, joints, gaskets, hatches, caps, and plugs.
- (b) The owner or operator of each source that is equipped with a closed vent system and control device as required in §60.692-5 (other than a flare) is exempt from §60.8 of the General Provisions and shall use Method 21 to

measure the emission concentrations, using 500 ppm as the no detectable emission limit. The instrument shall be calibrated each day before using. The calibration gases shall be:

- (1) Zero air (less than 10 ppm of hydrocarbon in air), and
- (2) A mixture of either methane or nhexane and air at a concentration of approximately, but less than, 10,000 ppm methane or n-hexane.
- (c) The owner or operator shall conduct a performance test initially, and at other times as requested by the Administrator, using the test methods and procedures in §60.18(f) to determine compliance of flares.
- (d) After installing the control equipment required to meet §60.693-2(a) or whenever sources that have ceased to treat refinery wastewater for a period of 1 year or more are placed back into service, the owner or operator shall determine compliance with the standards in §60.693-2(a) as follows:
- (1) The maximum gap widths and maximum gap areas between the primary seal and the separator wall and between the secondary seal and the separator wall shall be determined individually within 60 calendar days of the initial installation of the floating roof and introduction of refinery wastewater or 60 calendar days after the equipment is placed back into service using the following procedure when the separator is filled to the design operating level and when the roof is floating off the roof supports.
- (i) Measure seal gaps around the entire perimeter of the separator in each place where a 0.32 cm (0.125 in.) diameter uniform probe passes freely (without forcing or binding against seal) between the seal and the wall of the separator and measure the gap width and perimetrical distance of each such location.
- (ii) The total surface area of each gap described in (d)(1)(i) of this section shall be determined by using probes of various widths to measure accurately the actual distance from the wall to the seal and multiplying each such width by its respective perimetrical distance.
- (iii) Add the gap surface area of each gap location for the primary seal and the secondary seal individually, divide

the sum for each seal by the nominal perimeter of the separator basin and compare each to the maximum gap area as specified in §60.693-2.

- (2) The gap widths and total gap area shall be determined using the procedure in paragraph (d)(1) of this section according to the following frequency:
- (i) For primary seals, once every 5 years.
- (ii) For secondary seals, once every year.

#### § 60.697 Recordkeeping requirements.

- (a) Each owner or operator of a facility subject to the provisions of this subpart shall comply with the record-keeping requirements of this section. All records shall be retained for a period of 2 years after being recorded unless otherwise noted.
- (b)(1) For individual drain systems subject to §60.692-2, the location, date, and corrective action shall be recorded for each drain when the water seal is dry or otherwise breached, when a drain cap or plug is missing or improperly installed, or other problem is identified that could result in VOC emissions, as determined during the initial and periodic visual or physical inspection.
- (2) For junction boxes subject to §60.692-2, the location, date, and corrective action shall be recorded for inspections required by §60.692-2(b) when a broken seal, gap, or other problem is identified that could result in VOC emissions.
- (3) For sewer lines subject to §§ 60.692-2 and 60.693-1(e), the location, date, and corrective action shall be recorded for inspections required by §§ 60.692-2(c) and 60.693-1(e) when a problem is identified that could result in VOC emissions.
- (c) For oil-water separators subject to §60.692-3, the location, date, and corrective action shall be recorded for inspections required by §60.692-3(a) when a problem is identified that could result in VOC emissions.
- (d) For closed vent systems subject to \$60.692-5 and completely closed drain systems subject to \$60.693-1, the location, date, and corrective action shall be recorded for inspections required by \$60.692-5(e) during which detectable emissions are measured or a

problem is identified that could result in VOC emissions.

- (e)(1) If an emission point cannot be repaired or corrected without a process unit shutdown, the expected date of a successful repair shall be recorded.
- (2) The reason for the delay as specified in §60.692-6 shall be recorded if an emission point or equipment problem is not repaired or corrected in the specified amount of time.
- (3) The signature of the owner or operator (or designee) whose decision it was that repair could not be effected without refinery or process shutdown shall be recorded.
- (4) The date of successful repair or corrective action shall be recorded.
- (f)(1) A copy of the design specifications for all equipment used to comply with the provisions of this subpart shall be kept for the life of the source in a readily accessible location.
- (2) The following information pertaining to the design specifications shall be kept.
- (i) Detailed schematics, and piping and instrumentation diagrams.
- (ii) The dates and descriptions of any changes in the design specifications.
- (3) The following information pertaining to the operation and maintenance of closed drain systems and closed vent systems shall be kept in a readily accessible location.
- (i) Documentation demonstrating that the control device will achieve the required control efficiency during maximum loading conditions shall be kept for the life of the facility. This documentation is to include a general description of the gas streams that enter the control device, including flow and volatile organic compound content under varying liquid level conditions (dynamic and static) and manufacturer's design specifications for the control device. If an enclosed combustion device with a minimum residence time of 0.75 seconds and a minimum temperature of 816 °C (1,500 °F) is used to meet the 95-percent requirement, documentation that those conditions exist is sufficient to meet the requirements of this paragraph.
- (ii) For a carbon adsorption system that does not regenerate the carbon

#### § 60.697

bed directly onsite in the control device such as a carbon canister, the design analysis shall consider the vent stream composition, constituent concentrations, flow rate, relative humidity, and temperature. The design analysis shall also establish the design exhaust vent stream organic compound concentration level, capacity of carbon bed, type and working capacity of activated carbon used for carbon bed, and design carbon replacement interval based on the total carbon working capacity of the control device and source operating schedule.

(iii) Periods when the closed vent systems and control devices required in §60.692 are not operated as designed, including periods when a flare pilot does not have a flame shall be recorded and kept for 2 years after the information is recorded.

(iv) Dates of startup and shutdown of the closed vent system and control devices required in §60.692 shall be recorded and kept for 2 years after the information is recorded.

(v) The dates of each measurement of detectable emissions required in §§ 60.692, 60.693, or 60.692-5 shall be recorded and kept for 2 years after the information is recorded.

(vi) The background level measured during each detectable emissions measurement shall be recorded and kept for 2 years after the information is recorded.

(vii) The maximum instrument reading measured during each detectable emission measurement shall be recorded and kept for 2 years after the information is recorded.

(viii) Each owner or operator of an affected facility that uses a thermal incinerator shall maintain continuous records of the temperature of the gas stream in the combustion zone of the incinerator and records of all 3-hour periods of operation during which the average temperature of the gas stream in the combustion zone is more than 28 °C (50 °F) below the design combustion zone temperature, and shall keep such records for 2 years after the information is recorded.

(ix) Each owner or operator of an affected facility that uses a catalytic incinerator shall maintain continuous records of the temperature of the gas

stream both upstream and downstream of the catalyst bed of the incinerator, records of all 3-hour periods of operation during which the average temperature measured before the catalyst bed is more than 28 °C (50 °F) below the design gas stream temperature, and records of all 3-hour periods during which the average temperature difference across the catalyst bed is less than 80 percent of the design temperature difference, and shall keep such records for 2 years after the information is recorded.

(x) Each owner or operator of an affected facility that uses a carbon adsorber shall maintain continuous records of the VOC concentration level or reading of organics of the control device outlet gas stream or inlet and outlet gas stream and records of all 3-hour periods of operation during which the average VOC concentration level or reading of organics in the exhaust gases, or inlet and outlet gas stream, is more than 20 percent greater than the design exhaust gas concentration level, and shall keep such records for 2 years after the information is recorded.

(A) Each owner or operator of an affected facility that uses a carbon adsorber which is regenerated directly onsite shall maintain continuous records of the volatile organic compound concentration level or reading of organics of the control device outlet gas stream or inlet and outlet gas stream and records of all 3-hour periods of operation during which the average volatile organic compound concentration level or reading of organics in the exhaust gases, or inlet and outlet gas stream, is more than 20 percent greater than the design exhaust gas concentration level, and shall keep such records for 2 years after the information is recorded.

(B) If a carbon adsorber that is not regenerated directly onsite in the control device is used, then the owner or operator shall maintain records of dates and times when the control device is monitored, when breakthrough is measured, and shall record the date and time that the existing carbon in the control device is replaced with fresh carbon.

(g) If an owner or operator elects to install a tightly sealed cap or plug over

- a drain that is out of active service, the owner or operator shall keep for the life of a facility in a readily accessible location, plans or specifications which indicate the location of such drains.
- (h) For stormwater sewer systems subject to the exclusion in §60.692-1(d)(1), an owner or operator shall keep for the life of the facility in a readily accessible location, plans or specifications which demonstrate that no wastewater from any process units or equipment is directly discharged to the stormwater sewer system.
- (i) For ancillary equipment subject to the exclusion in §60.692-1(d)(2), an owner or operator shall keep for the life of a facility in a readily accessible location, plans or specifications which demonsrate that the ancillary equipment does not come in contact with or store oily wastewater.
- (j) For non-contact cooling water systems subject to the exclusion in §60.692-1(d)(3), an owner or operator shall keep for the life of the facility in a readily accessible location, plans or specifications which demonstrate that the cooling water does not contact hydrocarbons or oily wastewater and is not recirculated through a cooling tower.
- (k) For oil-water separators subject to \$60.693-2, the location, date, and corrective action shall be recorded for inspections required by \$\$60.693-2(a)(1)(iii)(A) and (B), and shall be maintained for the time period specified in paragraphs (k)(1) and (2) of this section.
- (1) For inspections required by  $\S60.693-2(a)(1)(iii)(A)$ , ten years after the information is recorded.
- (2) For inspections required by §60.693-2(a)(1)(iii)(B), two years after the information is recorded.

[53 FR 47623, Nov. 23, 1985, as amended at 60 FR 43259, Aug. 18, 1995; 65 FR 61778, Oct. 17, 2000]

#### $\S 60.698$ Reporting requirements.

- (a) An owner or operator electing to comply with the provisions of §60.693 shall notify the Administrator of the alternative standard selected in the report required in §60.7.
- (b)(1) Each owner or operator of a facility subject to this subpart shall sub-

- mit to the Administrator within 60 days after initial startup a certification that the equipment necessary to comply with these standards has been installed and that the required initial inspections or tests of process drains, sewer lines, junction boxes, oil-water separators, and closed vent systems and control devices have been carried out in accordance with these standards. Thereafter, the owner or operator shall submit to the Administrator semiannually a certification that all of the required inspections have been carried out in accordance with these standards.
- (2) Each owner or operator of an affected facility that uses a flare shall submit to the Administrator within 60 days after initial startup, as required under §60.8(a), a report of the results of the performance test required in §60.696(c).
- (c) A report that summarizes all inspections when a water seal was dry or otherwise breached, when a drain cap or plug was missing or improperly installed, or when cracks, gaps, or other problems were identified that could result in VOC emissions, including information about the repairs or corrective action taken, shall be submitted initially and semiannually thereafter to the Administrator.
- (d) As applicable, a report shall be submitted semiannually to the Administrator that indicates:
- (1) Each 3-hour period of operation during which the average temperature of the gas stream in the combustion zone of a thermal incinerator, as measured by the temperature monitoring device, is more than 28 °C (50 °F) below the design combustion zone temperature.
- (2) Each 3-hour period of operation during which the average temperature of the gas stream immediately before the catalyst bed of a catalytic incinerator, as measured by the temperature monitoring device, is more than 28 °C (50 °F) below the design gas stream temperature, and any 3-hour period during which the average temperature difference across the catalyst bed (i.e., the difference between the temperatures of the gas stream immediately before and after the catalyst bed), as

#### §60.699

measured by the temperature monitoring device, is less than 80 percent of the design temperature difference, or,

- (3) Each 3-hour period of operation during which the average VOC concentration level or reading of organics in the exhaust gases from a carbon adsorber is more than 20 percent greater than the design exhaust gas concentration level or reading.
- (i) Each 3-hour period of operation during which the average volatile organic compound concentration level or reading of organics in the exhaust gases from a carbon adsorber which is regenerated directly onsite is more than 20 percent greater than the design exhaust gas concentration level or reading.
- (ii) Each occurrence when the carbon in a carbon adsorber system that is not regenerated directly onsite in the control device is not replaced at the predetermined interval specified in §60.695(a)(3)(ii).
- (e) If compliance with the provisions of this subpart is delayed pursuant to \$60.692-7, the notification required under 40 CFR 60.7(a)(4) shall include the estimated date of the next scheduled refinery or process unit shutdown after the date of notification and the reason why compliance with the standards is technically impossible without a refinery or process unit shutdown.

[53 FR 47623, Nov. 23, 1988, as amended at 60 FR 43260, Aug. 18, 1995]

#### § 60.699 Delegation of authority.

- (a) In delegating implementation and enforcement authority to a State under section 111(c) of the Act, the authorities contained in paragraph (b) of this section shall be retained by the Administrator and not transferred to a State.
- (b) Authorities which will not be delegated to States:

§60.694 Permission to use alternative means of emission limitations.

[53 FR 47623, Nov. 23, 1985]

Subpart RRR—Standards of Performance for Volatile Organic Compound Emissions From Synthetic Organic Chemical Manufacturing Industry (SOCMI) Reactor Processes

SOURCE: 58 FR 45962, Aug. 31, 1993, unless otherwise noted.

## § 60.700 Applicability and designation of affected facility.

- (a) The provisions of this subpart apply to each affected facility designated in paragraph (b) of this section that is part of a process unit that produces any of the chemicals listed in §60.707 as a product, co-product, byproduct, or intermediate, except as provided in paragraph (c) of this section.
- (b) The affected facility is any of the following for which construction, modification, or reconstruction commenced after June 29, 1990:
- (1) Each reactor process not discharging its vent stream into a recovery system.
- (2) Each combination of a reactor process and the recovery system into which its vent stream is discharged.
- (3) Each combination of two or more reactor processes and the common recovery system into which their vent streams are discharged.
- (c) Exemptions from the provisions of paragraph (a) of this section are as follows:
- (1) Any reactor process that is designed and operated as a batch operation is not an affected facility.
- (2) Each affected facility that has a total resource effectiveness (TRE) index value greater than 8.0 is exempt from all provisions of this subpart except for §§60.702(c); 60.704 (d), (e), and (f); and 60.705 (g), (1)(1), (1)(6), and (t).
- (3) Each affected facility in a process unit with a total design capacity for all chemicals produced within that unit of less than 1 gigagram per year (1,100 tons per year) is exempt from all provisions of this subpart except for the recordkeeping and reporting requirements in §60.705 (i), (1)(5), and (n).
- (4) Each affected facility operated with a vent stream flow rate less than

Appendix H
40 CFR Part 60 Subpart IIII
Standards of Performance for Stationary Compression Ignition Internal Combustion Engines

#### **Environmental Protection Agency**

#### § 60.4200

Report	Due date	Contents	Reference
		ii. Description of efforts to have an accessible quali- fied operator; and.	§ 60.3054(a)(1).
		iii. The date a qualified operator will be accessible.	§ 60.3054(a)(1).
<ol><li>Qualified operation deviation status report.</li></ol>	Every 4 weeks following deviation	Description of efforts to     have an accessible qualified operator;.	§ 60.3054(a)(2).
	j	ii. The date a qualified oper- ator will be accessible; and.	§ 60.3054(a)(2).
		iii. Request to continue oper- ation.	§ 60.3054(a)(2).
<ol> <li>Qualified operator deviation notification of resumed oper- ation.</li> </ol>	a. Prior to resuming operation	Notification that you are re- surning operation.	§ 60.3054(b).

Note: This table is only a summary, see the referenced sections of the rule for the complete requirements.

[70 FR 74907, Dec. 16, 2005, as amended at 71 FR 67806, Nov. 24, 2006]

#### Subparts GGGG-HHHH [Reserved]

#### Subpart IIII—Standards of Performance for Stationary Compression Ignition Internal Combustion Engines

SOURCE: 71 FR 39172, July 11, 2006, unless otherwise noted.

WHAT THIS SUBPART COVERS

#### § 60.4200 Am I subject to this subpart?

- (a) The provisions of this subpart are applicable to manufacturers, owners, and operators of stationary compression ignition (CI) internal combustion engines (ICE) and other persons as specified in paragraphs (a)(1) through (4) of this section. For the purposes of this subpart, the date that construction commences is the date the engine is ordered by the owner or operator.
- (1) Manufacturers of stationary CI ICE with a displacement of less than 30 liters per cylinder where the model year is:
- (i) 2007 or later, for engines that are not fire pump engines:
- (ii) The model year listed in Table 3 to this subpart or later model year, for fire pump engines.
- (2) Owners and operators of stationary CI ICE that commence construction after July 11, 2005, where the stationary CI ICE are:
- (i) Manufactured after April 1, 2006, and are not fire pump engines, or
- (ii) Manufactured as a certified National Fire Protection Association

(NFPA) fire pump engine after July 1, 2006

- (3) Owners and operators of any stationary CI ICE that are modified or reconstructed after July 11, 2005 and any person that modifies or reconstructs any stationary CI ICE after July 11, 2005.
- (4) The provisions of §60.4208 of this subpart are applicable to all owners and operators of stationary CI ICE that commence construction after July 11, 2005.
- (b) The provisions of this subpart are not applicable to stationary CI ICE being tested at a stationary CI ICE test cell/stand.
- (c) If you are an owner or operator of an area source subject to this subpart, you are exempt from the obligation to obtain a permit under 40 CFR part 70 or 40 CFR part 71, provided you are not required to obtain a permit under 40 CFR 70.3(a) or 40 CFR 71.3(a) for a reason other than your status as an area source under this subpart. Notwithstanding the previous sentence, you must continue to comply with the provisions of this subpart applicable to area sources.
- (d) Stationary CI ICE may be eligible for exemption from the requirements of this subpart as described in 40 CFR part 1068, subpart C (or the exemptions described in 40 CFR part 89, subpart J and 40 CFR part 94, subpart J, for engines that would need to be certified to standards in those parts), except that owners and operators, as well as manufacturers, may be eligible to request an exemption for national security.

#### § 60.4201

(e) Owners and operators of facilities with CI ICE that are acting as temporary replacement units and that are located at a stationary source for less than 1 year and that have been properly certified as meeting the standards that would be applicable to such engine under the appropriate nonroad engine provisions, are not required to meet any other provisions under this subpart with regard to such engines.

[71 FR 39172, July 11, 2006, as amended at 76 FR 37967, June 28, 2011]

EMISSION STANDARDS FOR MANUFACTURERS

#### § 60.4201 What emission standards must I meet for non-emergency engines if I am a stationary CI internal combustion engine manufacturer?

- (a) Stationary CI internal combustion engine manufacturers must certify their 2007 model year and later nonemergency stationary CI ICE with a maximum engine power less than or equal to 2,237 kilowatt (KW) (3,000 horsepower (HP)) and a displacement of less than 10 liters per cylinder to the certification emission standards for new nonroad CI engines in 40 CFR 89.112, 40 CFR 89.113, 40 CFR 1039.101, 40 CFR 1039.105, 40 CFR 1039.107, and 40 CFR 1039.115, as applicable, for all pollutants, for the same model year and maximum engine power.
- (b) Stationary CI internal combustion engine manufacturers must certify their 2007 through 2010 model year nonemergency stationary CI ICE with a maximum engine power greater than 2,237 KW (3,000 HP) and a displacement of less than 10 liters per cylinder to the emission standards in table 1 to this subpart, for all pollutants, for the same maximum engine power.
- (c) Stationary CI internal combustion engine manufacturers must certify their 2011 model year and later non-emergency stationary CI ICE with a maximum engine power greater than 2,237 KW (3,000 HP) and a displacement of less than 10 liters per cylinder to the certification emission standards for new nonroad CI engines in 40 CFR 1039.101, 40 CFR 1039.102, 40 CFR 1039.104, 40 CFR 1039.105, 40 CFR

1039.107, and 40 CFR 1039.115, as applicable, for all pollutants, for the same maximum engine power.

- (d) Stationary CI internal combustion engine manufacturers must certify the following non-emergency stationary CI ICE to the certification emission standards for new marine CI engines in 40 CFR 94.8, as applicable, for all pollutants, for the same displacement and maximum engine power:
- (1) Their 2007 model year through 2012 non-emergency stationary CI ICE with a displacement of greater than or equal to 10 liters per cylinder and less than 30 liters per cylinder:
- (2) Their 2013 model year non-emergency stationary CI ICE with a maximum engine power greater than or equal to 3,700 KW (4,958 HP) and a displacement of greater than or equal to 10 liters per cylinder and less than 15 liters per cylinder; and
- (3) Their 2013 model year non-emergency stationary CI ICE with a displacement of greater than or equal to 15 liters per cylinder and less than 30 liters per cylinder.
- (e) Stationary CI internal combustion engine manufacturers must certify the following non-emergency stationary CI ICE to the certification emission standards and other requirements for new marine CI engines in 40 CFR 1042.101, 40 CFR 1042.107, 40 CFR 1042.110, 40 CFR 1042.115, 40 CFR 1042.120, and 40 CFR 1042.145, as applicable, for all pollutants, for the same displacement and maximum engine power:
- (1) Their 2013 model year non-emergency stationary CI ICE with a maximum engine power less than 3,700 KW (4,958 HP) and a displacement of greater than or equal to 10 liters per cylinder and less than 15 liters per cylinder; and
- (2) Their 2014 model year and later non-emergency stationary CI ICE with a displacement of greater than or equal to 10 liters per cylinder and less than 30 liters per cylinder.
- (f) Notwithstanding the requirements in paragraphs (a) through (c) of this section, stationary non-emergency CI ICE identified in paragraphs (a) and (c) may be certified to the provisions of 40 CFR part 94 or, if Table 1 to 40 CFR 1042.1 identifies 40 CFR part 1042 as being applicable, 40 CFR part 1042, if

the engines will be used solely in either or both of the following locations:

- (1) Areas of Alaska not accessible by the Federal Aid Highway System (FAHS); and
  - (2) Marine offshore installations.
- (g) Notwithstanding the requirements in paragraphs (a) through (f) of this section, stationary CI internal combustion engine manufacturers are not required to certify reconstructed engines; however manufacturers may elect to do so. The reconstructed engine must be certified to the emission standards specified in paragraphs (a) through (e) of this section that are applicable to the model year, maximum engine power, and displacement of the reconstructed stationary CI ICE.

[71 FR 39172, July 11, 2006, as amended at 76 FR 37967, June 28, 2011]

#### § 60.4202 What emission standards must I meet for emergency engines if I am a stationary CI internal combustion engine manufacturer?

- (a) Stationary CI internal combustion engine manufacturers must certify their 2007 model year and later emergency stationary CI ICE with a maximum engine power less than or equal to 2,237 KW (3,000 HP) and a displacement of less than 10 liters per cylinder that are not fire pump engines to the emission standards specified in paragraphs (a)(1) through (2) of this section.
- (1) For engines with a maximum engine power less than 37 KW (50 HP):
- (i) The certification emission standards for new nonroad CI engines for the same model year and maximum engine power in 40 CFR 89.112 and 40 CFR 89.113 for all pollutants for model year 2007 engines, and
- (ii) The certification emission standards for new nonroad CI engines in 40 CFR 1039.104, 40 CFR 1039.105, 40 CFR 1039.107, 40 CFR 1039.115, and table 2 to this subpart, for 2008 model year and later engines.
- (2) For engines with a maximum engine power greater than or equal to 37 KW (50 HP), the certification emission standards for new nonroad CI engines for the same model year and maximum engine power in 40 CFR 89.112 and 40 CFR 89.113 for all pollutants beginning in model year 2007.

- (b) Stationary CI internal combustion engine manufacturers must certify their 2007 model year and later emergency stationary CI ICE with a maximum engine power greater than 2,237 KW (3,000 HP) and a displacement of less than 10 liters per cylinder that are not fire pump engines to the emission standards specified in paragraphs (b)(1) through (2) of this section.
- (1) For 2007 through 2010 model years, the emission standards in table 1 to this subpart, for all pollutants, for the same maximum engine power.
- (2) For 2011 model year and later, the certification emission standards for new nonroad CI engines for engines of the same model year and maximum engine power in 40 CFR 89.112 and 40 CFR 89.113 for all pollutants.
  - (c) [Reserved]
- (d) Beginning with the model years in table 3 to this subpart, stationary CI internal combustion engine manufacturers must certify their fire pump stationary CI ICE to the emission standards in table 4 to this subpart, for all pollutants, for the same model year and NFPA nameplate power.
- (e) Stationary CI internal combustion engine manufacturers must certify the following emergency stationary CI ICE that are not fire pump engines to the certification emission standards for new marine CI engines in 40 CFR 94.8, as applicable, for all pollutants, for the same displacement and maximum engine power:
- (1) Their 2007 model year through 2012 emergency stationary CI ICE with a displacement of greater than or equal to 10 liters per cylinder and less than 30 liters per cylinder;
- (2) Their 2013 model year and later emergency stationary CI ICE with a maximum engine power greater than or equal to 3,700 KW (4,958 HP) and a displacement of greater than or equal to 10 liters per cylinder and less than 15 liters per cylinder;
- (3) Their 2013 model year emergency stationary CI ICE with a displacement of greater than or equal to 15 liters per cylinder and less than 30 liters per cylinder; and
- (4) Their 2014 model year and later emergency stationary CI ICE with a maximum engine power greater than or

#### §60.4203

equal to 2,000 KW (2,682 HP) and a displacement of greater than or equal to 15 liters per cylinder and less than 30 liters per cylinder.

- (f) Stationary CI internal combustion engine manufacturers must certify the following emergency stationary CI ICE to the certification emission standards and other requirements applicable to Tier 3 new marine CI engines in 40 CFR 1042.101, 40 CFR 1042.107, 40 CFR 1042.115, 40 CFR 1042.120, and 40 CFR 1042.145, for all pollutants, for the same displacement and maximum engine power:
- (1) Their 2013 model year and later emergency stationary CI ICE with a maximum engine power less than 3,700 KW (4,958 HP) and a displacement of greater than or equal to 10 liters per cylinder and less than 15 liters per cylinder; and
- (2) Their 2014 model year and later emergency stationary CI ICE with a maximum engine power less than 2,000 KW (2,682 HP) and a displacement of greater than or equal to 15 liters per cylinder and less than 30 liters per cylinder.
- (g) Notwithstanding the requirements in paragraphs (a) through (d) of this section, stationary emergency CI internal combustion engines identified in paragraphs (a) and (c) may be certified to the provisions of 40 CFR part 94 or, if Table 2 to 40 CFR 1042.101 identifies Tier 3 standards as being applicable, the requirements applicable to Tier 3 engines in 40 CFR part 1042, if the engines will be used solely in either or both of the following locations:
- (1) Areas of Alaska not accessible by the FAHS; and
  - (2) Marine offshore installations.
- (h) Notwithstanding the requirements in paragraphs (a) through (f) of this section, stationary CI internal combustion engine manufacturers are not required to certify reconstructed engines; however manufacturers may elect to do so. The reconstructed engine must be certified to the emission standards specified in paragraphs (a) through (f) of this section that are applicable to the model year, maximum engine power and displacement of the

reconstructed emergency stationary CI ICE.

[71 FR 39172, July 11, 2006, as amended at 76 FR 37968, June 28, 2011]

#### § 60.4203 How long must my engines meet the emission standards if I am a manufacturer of stationary CI internal combustion engines?

Engines manufactured by stationary CI internal combustion engine manufacturers must meet the emission standards as required in §§60.4201 and 60.4202 during the certified emissions life of the engines.

[76 FR 37968, June 28, 2011]

EMISSION STANDARDS FOR OWNERS AND OPERATORS

#### § 60.4204 What emission standards must I meet for non-emergency engines if I am an owner or operator of a stationary CI internal combustion engine?

- (a) Owners and operators of pre-2007 model year non-emergency stationary CI ICE with a displacement of less than 10 liters per cylinder must comply with the emission standards in table 1 to this subpart. Owners and operators of pre-2007 model year non-emergency stationary CI ICE with a displacement of greater than or equal to 10 liters per cylinder and less than 30 liters per cylinder must comply with the emission standards in 40 CFR 94.8(a)(1).
- (b) Owners and operators of 2007 model year and later non-emergency stationary CI ICE with a displacement of less than 30 liters per cylinder must comply with the emission standards for new CI engines in §60.4201 for their 2007 model year and later stationary CI ICE, as applicable.
- (c) Owners and operators of nonemergency stationary CI engines with a displacement of greater than or equal to 30 liters per cylinder must meet the following requirements:
- (1) For engines installed prior to January 1, 2012, limit the emissions of  $NO_X$  in the stationary CI internal combustion engine exhaust to the following:
- (i) 17.0 grams per kilowatt-hour (g/KW-hr) (12.7 grams per horsepower-hr (g/HP-hr)) when maximum engine speed is less than 130 revolutions per minute (rpm);

#### **Environmental Protection Agency**

- (ii)  $45 \cdot n^{-0.2}$  g/KW-hr ( $34 \cdot n^{-0.2}$  g/HP-hr) when maximum engine speed is 130 or more but less than 2,000 rpm, where n is maximum engine speed; and
- (iii) 9.8 g/KW-hr (7.3 g/HP-hr) when maximum engine speed is 2,000 rpm or more.
- (2) For engines installed on or after January 1, 2012 and before January 1, 2016, limit the emissions of  $NO_X$  in the stationary CI internal combustion engine exhaust to the following:
- (i) 14.4 g/KW-hr (10.7 g/HP-hr) when maximum engine speed is less than 130 rpm;
- (ii)  $44 \cdot n^{-0.23}$  g/KW-hr ( $33 \cdot n^{-0.23}$  g/HP-hr) when maximum engine speed is greater than or equal to 130 but less than 2,000 rpm and where n is maximum engine speed; and
- (iii) 7.7 g/KW-hr (5.7 g/HP-hr) when maximum engine speed is greater than or equal to 2,000 rpm.
- (3) For engines installed on or after January 1, 2016, limit the emissions of  $NO_X$  in the stationary CI internal combustion engine exhaust to the following:
- (i) 3.4 g/KW-hr (2.5 g/HP-hr) when maximum engine speed is less than 130 rpm:
- (ii) 9.0  $\cdot$  n<sup>-0.20</sup> g/KW-hr (6.7  $\cdot$  n<sup>-0.20</sup> g/HP-hr) where n (maximum engine speed) is 130 or more but less than 2,000 rpm; and
- (iii) 2.0 g/KW-hr (1.5 g/HP-hr) where maximum engine speed is greater than or equal to 2.000 rpm.
- (4) Reduce particulate matter (PM) emissions by 60 percent or more, or limit the emissions of PM in the stationary CI internal combustion engine exhaust to 0.15 g/KW-hr (0.11 g/HP-hr).
- (d) Owners and operators of nonemergency stationary CI ICE with a displacement of less than 30 liters per cylinder who conduct performance tests in-use must meet the not-to-exceed (NTE) standards as indicated in §60.4212.
- (e) Owners and operators of any modified or reconstructed non-emergency stationary CI ICE subject to this subpart must meet the emission standards applicable to the model year, maximum engine power, and displacement of the modified or reconstructed non-emergency stationary CI ICE that are

specified in paragraphs (a) through (d) of this section.

[71 FR 39172, July 11, 2006, as amended at 76 FR 37968, June 28, 2011]

#### § 60.4205 What emission standards must I meet for emergency engines if I am an owner or operator of a stationary CI internal combustion engine?

- (a) Owners and operators of pre-2007 model year emergency stationary CI ICE with a displacement of less than 10 liters per cylinder that are not fire pump engines must comply with the emission standards in Table 1 to this subpart. Owners and operators of pre-2007 model year emergency stationary CI ICE with a displacement of greater than or equal to 10 liters per cylinder and less than 30 liters per cylinder that are not fire pump engines must comply with the emission standards in 40 CFR 94.8(a)(1).
- (b) Owners and operators of 2007 model year and later emergency stationary CI ICE with a displacement of less than 30 liters per cylinder that are not fire pump engines must comply with the emission standards for new nonroad CI engines in §60.4202, for all pollutants, for the same model year and maximum engine power for their 2007 model year and later emergency stationary CI ICE.
- (c) Owners and operators of fire pump engines with a displacement of less than 30 liters per cylinder must comply with the emission standards in table 4 to this subpart, for all pollutants.
- (d) Owners and operators of emergency stationary CI engines with a displacement of greater than or equal to 30 liters per cylinder must meet the requirements in this section.
- (1) For engines installed prior to January 1, 2012, limit the emissions of  $NO_X$  in the stationary CI internal combustion engine exhaust to the following:
- (i)  $17.0~{\rm g/KW\text{-}hr}$  ( $12.7~{\rm g/HP\text{-}hr}$ ) when maximum engine speed is less than 130 rpm;
- (ii)  $45 \cdot n^{-0.2}$  g/KW-hr ( $34 \cdot n^{-0.2}$  g/HP-hr) when maximum engine speed is 130 or more but less than 2,000 rpm, where n is maximum engine speed; and
- (iii) 9.8 g/kW-hr (7.3 g/HP-hr) when maximum engine speed is 2,000 rpm or more.

#### § 60.4206

- (2) For engines installed on or after January 1, 2012, limit the emissions of  $NO_{\rm X}$  in the stationary CI internal combustion engine exhaust to the following:
- (i) 14.4 g/KW-hr (10.7 g/HP-hr) when maximum engine speed is less than 130 rpm;
- (ii)  $44 \cdot n^{-0.23}$  g/KW-hr ( $33 \cdot n^{-0.23}$  g/HP-hr) when maximum engine speed is greater than or equal to 130 but less than 2,000 rpm and where n is maximum engine speed; and
- (iii) 7.7 g/KW-hr (5.7 g/HP-hr) when maximum engine speed is greater than or equal to 2,000 rpm.
- (3) Limit the emissions of PM in the stationary CI internal combustion engine exhaust to 0.40 g/KW-hr (0.30 g/HP-hr).
- (e) Owners and operators of emergency stationary CI ICE with a displacement of less than 30 liters per cylinder who conduct performance tests in-use must meet the NTE standards as indicated in §60.4212.
- (f) Owners and operators of any modified or reconstructed emergency stationary CI ICE subject to this subpart must meet the emission standards applicable to the model year, maximum engine power, and displacement of the modified or reconstructed CI ICE that are specified in paragraphs (a) through (e) of this section.

[71 FR 39172, July 11, 2006, as amended at 76 FR 37969, June 28, 2011]

#### § 60.4206 How long must I meet the emission standards if I am an owner or operator of a stationary CI internal combustion engine?

Owners and operators of stationary CI ICE must operate and maintain stationary CI ICE that achieve the emission standards as required in §§ 60.4204 and 60.4205 over the entire life of the engine.

[76 FR 37969, June 28, 2011]

FUEL REQUIREMENTS FOR OWNERS AND OPERATORS

## § 60.4207 What fuel requirements must I meet if I am an owner or operator of a stationary CI internal combustion engine subject to this subpart?

(a) Beginning October 1, 2007, owners and operators of stationary CI ICE sub-

ject to this subpart that use diesel fuel must use diesel fuel that meets the requirements of 40 CFR 80.510(a).

- (b) Beginning October 1, 2010, owners and operators of stationary CI ICE subject to this subpart with a displacement of less than 30 liters per cylinder that use diesel fuel must use diesel fuel that meets the requirements of 40 CFR 80.510(b) for nonroad diesel fuel, except that any existing diesel fuel purchased (or otherwise obtained) prior to October 1, 2010, may be used until depleted.
  - (c) [Reserved]
- (d) Beginning June 1, 2012, owners and operators of stationary CI ICE subject to this subpart with a displacement of greater than or equal to 30 liters per cylinder are no longer subject to the requirements of paragraph (a) of this section, and must use fuel that meets a maximum per-gallon sulfur content of 1,000 parts per million (ppm).
- (e) Stationary CI ICE that have a national security exemption under §60.4200(d) are also exempt from the fuel requirements in this section.

[71 FR 39172, July 11, 2006, as amended at 76 FR 37969, June 28, 2011; 78 FR 6695, Jan. 30, 2013]

OTHER REQUIREMENTS FOR OWNERS AND OPERATORS

## § 60.4208 What is the deadline for importing or installing stationary CI ICE produced in previous model years?

- (a) After December 31, 2008, owners and operators may not install stationary CI ICE (excluding fire pump engines) that do not meet the applicable requirements for 2007 model year engines.
- (b) After December 31, 2009, owners and operators may not install stationary CI ICE with a maximum engine power of less than 19 KW (25 HP) (excluding fire pump engines) that do not meet the applicable requirements for 2008 model year engines.
- (c) After December 31, 2014, owners and operators may not install non-emergency stationary CI ICE with a maximum engine power of greater than or equal to 19 KW (25 HP) and less than

56 KW (75 HP) that do not meet the applicable requirements for 2013 model year non-emergency engines.

- (d) After December 31, 2013, owners and operators may not install non-emergency stationary CI ICE with a maximum engine power of greater than or equal to 56 KW (75 HP) and less than 130 KW (175 HP) that do not meet the applicable requirements for 2012 model year non-emergency engines.
- (e) After December 31, 2012, owners and operators may not install non-emergency stationary CI ICE with a maximum engine power of greater than or equal to 130 KW (175 HP), including those above 560 KW (750 HP), that do not meet the applicable requirements for 2011 model year non-emergency engines.
- (f) After December 31, 2016, owners and operators may not install non-emergency stationary CI ICE with a maximum engine power of greater than or equal to 560 KW (750 HP) that do not meet the applicable requirements for 2015 model year non-emergency engines.
- (g) After December 31, 2018, owners and operators may not install non-emergency stationary CI ICE with a maximum engine power greater than or equal to 600 KW (804 HP) and less than 2,000 KW (2,680 HP) and a displacement of greater than or equal to 10 liters per cylinder and less than 30 liters per cylinder that do not meet the applicable requirements for 2017 model year non-emergency engines.
- (h) In addition to the requirements specified in §§60.4201, 60.4202, 60.4204, and 60.4205, it is prohibited to import stationary CI ICE with a displacement of less than 30 liters per cylinder that do not meet the applicable requirements specified in paragraphs (a) through (g) of this section after the dates specified in paragraphs (a) through (g) of this section.
- (i) The requirements of this section do not apply to owners or operators of stationary CI ICE that have been modified, reconstructed, and do not apply to engines that were removed from one existing location and reinstalled at a new location.

[71 FR 39172, July 11, 2006, as amended at 76 FR 37969, June 28, 2011]

## § 60.4209 What are the monitoring requirements if I am an owner or operator of a stationary CI internal combustion engine?

If you are an owner or operator, you must meet the monitoring requirements of this section. In addition, you must also meet the monitoring requirements specified in § 60.4211.

- (a) If you are an owner or operator of an emergency stationary CI internal combustion engine that does not meet the standards applicable to non-emergency engines, you must install a nonresettable hour meter prior to startup of the engine.
- (b) If you are an owner or operator of a stationary CI internal combustion engine equipped with a diesel particulate filter to comply with the emission standards in §60.4204, the diesel particulate filter must be installed with a backpressure monitor that notifies the owner or operator when the high backpressure limit of the engine is approached.

[71 FR 39172, July 11, 2006, as amended at 76 FR 37969, June 28, 2011]

#### COMPLIANCE REQUIREMENTS

## § 60.4210 What are my compliance requirements if I am a stationary CI internal combustion engine manufacturer?

(a) Stationary CI internal combustion engine manufacturers must certify their stationary CI ICE with a displacement of less than 10 liters per cylinder to the emission standards specified in §60.4201(a) through (c) and §60.4202(a), (b) and (d) using the certification procedures required in 40 CFR part 89, subpart B, or 40 CFR part 1039, subpart C, as applicable, and must test their engines as specified in those parts. For the purposes of this subpart, engines certified to the standards in table 1 to this subpart shall be subject to the same requirements as engines certified to the standards in 40 CFR part 89. For the purposes of this subpart, engines certified to the standards in table 4 to this subpart shall be subject to the same requirements as engines certified to the standards in 40 CFR part 89, except that engines with NFPA nameplate power of less than 37 KW (50 HP) certified to model year 2011 or later standards shall be subject to the same requirements as engines certified to the standards in 40 CFR part 1039.

- (b) Stationary CI internal combustion engine manufacturers must certify their stationary CI ICE with a displacement of greater than or equal to 10 liters per cylinder and less than 30 liters per cylinder to the emission standards specified in §60.4201(d) and (e) and §60.4202(e) and (f) using the certification procedures required in 40 CFR part 94, subpart C, or 40 CFR part 1042, subpart C, as applicable, and must test their engines as specified in 40 CFR part 94 or 1042, as applicable.
- (c) Stationary CI internal combustion engine manufacturers must meet the requirements of 40 CFR 1039.120, 1039.125, 1039.130, and 1039.135, and 40 CFR part 1068 for engines that are certified to the emission standards in 40 CFR part 1039. Stationary CI internal combustion engine manufacturers must meet the corresponding provisions of 40 CFR part 89, 40 CFR part 94 or 40 CFR part 1042 for engines that would be covered by that part if they were nonroad (including marine) engines. Labels on such engines must refer to stationary engines, rather than or in addition to nonroad or marine engines, as appropriate. Stationary CI internal combustion engine manufacturers must label their engines according to paragraphs (c)(1) through (3) of this section.
- (1) Stationary CI internal combustion engines manufactured from January 1, 2006 to March 31, 2006 (January 1, 2006 to June 30, 2006 for fire pump engines), other than those that are part of certified engine families under the nonroad CI engine regulations, must be labeled according to 40 CFR 1039.20.
- (2) Stationary CI internal combustion engines manufactured from April 1, 2006 to December 31, 2006 (or, for fire pump engines, July 1, 2006 to December 31 of the year preeding the year listed in table 3 to this subpart) must be labeled according to paragraphs (c)(2)(i) through (iii) of this section:
- (i) Stationary CI internal combustion engines that are part of certified engine families under the nonroad regulations must meet the labeling requirements for nonroad CI engines, but do

not have to meet the labeling requirements in 40 CFR 1039.20.

- (ii) Stationary CI internal combustion engines that meet Tier 1 requirements (or requirements for fire pumps) under this subpart, but do not meet the requirements applicable to nonroad CI engines must be labeled according to 40 CFR 1039.20. The engine manufacturer may add language to the label clarifying that the engine meets Tier 1 requirements (or requirements for fire pumps) of this subpart.
- (iii) Stationary CI internal combustion engines manufactured after April 1, 2006 that do not meet Tier 1 requirements of this subpart, or fire pumps engines manufactured after July 1, 2006 that do not meet the requirements for fire pumps under this subpart, may not be used in the U.S. If any such engines are manufactured in the U.S. after April 1, 2006 (July 1, 2006 for fire pump engines), they must be exported or must be brought into compliance with the appropriate standards prior to initial operation. The export provisions of 40 CFR 1068.230 would apply to engines for export and the manufacturers must label such engines according to 40 CFR 1068,230.
- (3) Stationary CI internal combustion engines manufactured after January 1, 2007 (for fire pump engines, after January 1 of the year listed in table 3 to this subpart, as applicable) must be labeled according to paragraphs (c)(3)(i) through (iii) of this section.
- (i) Stationary CI internal combustion engines that meet the requirements of this subpart and the corresponding requirements for nonroad (including marine) engines of the same model year and HP must be labeled according to the provisions in 40 CFR parts 89, 94, 1039 or 1042, as appropriate.
- (ii) Stationary CI internal combustion engines that meet the requirements of this subpart, but are not certified to the standards applicable to nonroad (including marine) engines of the same model year and HP must be labeled according to the provisions in 40 CFR parts 89, 94, 1039 or 1042, as appropriate, but the words "stationary" must be included instead of "nonroad" or "marine" on the label. In addition, such engines must be labeled according to 40 CFR 1039.20.

- (iii) Stationary CI internal combustion engines that do not meet the requirements of this subpart must be labeled according to 40 CFR 1068.230 and must be exported under the provisions of 40 CFR 1068.230.
- (d) An engine manufacturer certifying an engine family or families to standards under this subpart that are identical to standards applicable under 40 CFR parts 89, 94, 1039 or 1042 for that model year may certify any such family that contains both nonroad (including marine) and stationary engines as a single engine family and/or may include any such family containing stationary engines in the averaging, banking and trading provisions applicable for such engines under those parts.
- (e) Manufacturers of engine families discussed in paragraph (d) of this section may meet the labeling requirements referred to in paragraph (c) of this section for stationary CI ICE by either adding a separate label containing the information required in paragraph (c) of this section or by adding the words "and stationary" after the word "nonroad" or "marine," as appropriate, to the label.
- (f) Starting with the model years shown in table 5 to this subpart, stationary CI internal combustion engine manufacturers must add a permanent label stating that the engine is for stationary emergency use only to each new emergency stationary CI internal combustion engine greater than or equal to 19 KW (25 HP) that meets all the emission standards for emergency engines in §60.4202 but does not meet all the emission standards for nonemergency engines in §60.4201. The label must be added according to the labeling requirements specified in 40 CFR 1039.135(b). Engine manufacturers must specify in the owner's manual that operation of emergency engines is limited to emergency operations and required maintenance and testing.
- (g) Manufacturers of fire pump engines may use the test cycle in table 6 to this subpart for testing fire pump engines and may test at the NFPA certified nameplate HP, provided that the engine is labeled as "Fire Pump Applications Only".
- (h) Engine manufacturers, including importers, may introduce into com-

- merce uncertified engines or engines certified to earlier standards that were manufactured before the new or changed standards took effect until inventories are depleted, as long as such engines are part of normal inventory. For example, if the engine manufacturers' normal industry practice is to keep on hand a one-month supply of engines based on its projected sales, and a new tier of standards starts to apply for the 2009 model year, the engine manufacturer may manufacture engines based on the normal inventory requirements late in the 2008 model year, and sell those engines for installation. The engine manufacturer may not circumvent the provisions of §§ 60.4201 or 60.4202 by stockpiling engines that are built before new or changed standards take effect. Stockpiling of such engines beyond normal industry practice is a violation of this subpart.
- (i) The replacement engine provisions of 40 CFR 89.1003(b)(7), 40 CFR 94.1103(b)(3), 40 CFR 94.1103(b)(4) and 40 CFR 1068.240 are applicable to stationary CI engines replacing existing equipment that is less than 15 years old.

[71 FR 39172, July 11, 2006, as amended at 76 FR 37969, June 28, 2011]

## § 60.4211 What are my compliance requirements if I am an owner or operator of a stationary CI internal combustion engine?

- (a) If you are an owner or operator and must comply with the emission standards specified in this subpart, you must do all of the following, except as permitted under paragraph (g) of this section:
- (1) Operate and maintain the stationary CI internal combustion engine and control device according to the manufacturer's emission-related written instructions;
- (2) Change only those emission-related settings that are permitted by the manufacturer; and
- (3) Meet the requirements of 40 CFR parts 89, 94 and/or 1068, as they apply to you.
- (b) If you are an owner or operator of a pre-2007 model year stationary CI internal combustion engine and must comply with the emission standards specified in §§ 60.4204(a) or 60.4205(a), or

#### §60.4211

if you are an owner or operator of a CI fire pump engine that is manufactured prior to the model years in table 3 to this subpart and must comply with the emission standards specified in §60.4205(c), you must demonstrate compliance according to one of the methods specified in paragraphs (b)(1) through (5) of this section.

- (1) Purchasing an engine certified according to 40 CFR part 89 or 40 CFR part 94, as applicable, for the same model year and maximum engine power. The engine must be installed and configured according to the manufacturer's specifications.
- (2) Keeping records of performance test results for each pollutant for a test conducted on a similar engine. The test must have been conducted using the same methods specified in this subpart and these methods must have been followed correctly.
- (3) Keeping records of engine manufacturer data indicating compliance with the standards.
- (4) Keeping records of control device vendor data indicating compliance with the standards.
- (5) Conducting an initial performance test to demonstrate compliance with the emission standards according to the requirements specified in §60.4212, as applicable.
- (c) If you are an owner or operator of a 2007 model year and later stationary CI internal combustion engine and must comply with the emission standspecified in §60.4204(b) or §60.4205(b), or if you are an owner or operator of a CI fire pump engine that is manufactured during or after the model year that applies to your fire pump engine power rating in table 3 to this subpart and must comply with the emission standards specified §60.4205(c), you must comply by purchasing an engine certified to the emisstandards in  $\S60.4204(b)$ , §60.4205(b) or (c), as applicable, for the same model year and maximum (or in the case of fire pumps, NFPA nameplate) engine power. The engine must be installed and configured according to the manufacturer's emission-related specifications, except as permitted in paragraph (g) of this section.
- (d) If you are an owner or operator and must comply with the emission

standards specified in §60.4204(c) or §60.4205(d), you must demonstrate compliance according to the requirements specified in paragraphs (d)(1) through (3) of this section.

- (1) Conducting an initial performance test to demonstrate initial compliance with the emission standards as specified in  $\S 60.4213$ .
- (2) Establishing operating parameters to be monitored continuously to ensure the stationary internal combustion engine continues to meet the emission standards. The owner or operator must petition the Administrator for approval of operating parameters to be monitored continuously. The petition must include the information described in paragraphs (d)(2)(i) through (v) of this section.
- (i) Identification of the specific parameters you propose to monitor continuously:
- (ii) A discussion of the relationship between these parameters and  $NO_X$  and PM emissions, identifying how the emissions of these pollutants change with changes in these parameters, and how limitations on these parameters will serve to limit  $NO_X$  and PM emissions;
- (iii) A discussion of how you will establish the upper and/or lower values for these parameters which will establish the limits on these parameters in the operating limitations;
- (iv) A discussion identifying the methods and the instruments you will use to monitor these parameters, as well as the relative accuracy and precision of these methods and instruments;
- (v) A discussion identifying the frequency and methods for recalibrating the instruments you will use for monitoring these parameters.
- (3) For non-emergency engines with a displacement of greater than or equal to 30 liters per cylinder, conducting annual performance tests to demonstrate continuous compliance with the emission standards as specified in §60.4213.
- (e) If you are an owner or operator of a modified or reconstructed stationary CI internal combustion engine and must comply with the emission standards specified in \$60.4204(e) or

 $\S60.4205(f)$ , you must demonstrate compliance according to one of the methods specified in paragraphs (e)(1) or (2) of this section.

- (1) Purchasing, or otherwise owning or operating, an engine certified to the emission standards in §60.4204(e) or §60.4205(f), as applicable.
- (2) Conducting a performance test to demonstrate initial compliance with the emission standards according to the requirements specified in §60.4212 or §60.4213, as appropriate. The test must be conducted within 60 days after the engine commences operation after the modification or reconstruction.
- (f) If you own or operate an emergency stationary ICE, you must operate the emergency stationary ICE according to the requirements in paragraphs (f)(1) through (3) of this section. In order for the engine to be considered an emergency stationary ICE under this subpart, any operation other than emergency operation, maintenance and testing, emergency demand response, and operation in non-emergency situations for 50 hours per year, as described in paragraphs (f)(1) through (3) of this section, is prohibited. If you do not operate the engine according to the requirements in paragraphs (f)(1) through (3) of this section, the engine will not be considered an emergency engine under this subpart and must meet all requirements for non-emergency engines.
- (1) There is no time limit on the use of emergency stationary ICE in emergency situations.
- (2) You may operate your emergency stationary ICE for any combination of the purposes specified in paragraphs (f)(2)(i) through (iii) of this section for a maximum of 100 hours per calendar year. Any operation for non-emergency situations as allowed by paragraph (f)(3) of this section counts as part of the 100 hours per calendar year allowed by this paragraph (f)(2).
- (i) Emergency stationary ICE may be operated for maintenance checks and readiness testing, provided that the tests are recommended by federal, state or local government, the manufacturer, the vendor, the regional transmission organization or equivalent balancing authority and transmission operator, or the insurance

company associated with the engine. The owner or operator may petition the Administrator for approval of additional hours to be used for maintenance checks and readiness testing, but a petition is not required if the owner or operator maintains records indicating that federal, state, or local standards require maintenance and testing of emergency ICE beyond 100 hours per calendar year.

- (ii) Emergency stationary ICE may be operated for emergency demand response for periods in which the Reliability Coordinator under the North American Electric Reliability Corporation (NERC) Reliability Standard EOP-002-3, Capacity and Energy Emergencies (incorporated by reference, see §60.17), or other authorized entity as determined by the Reliability Coordinator, has declared an Energy Emergency Alert Level 2 as defined in the NERC Reliability Standard EOP-002-3.
- (iii) Emergency stationary ICE may be operated for periods where there is a deviation of voltage or frequency of 5 percent or greater below standard voltage or frequency.
- (3) Emergency stationary ICE may be operated for up to 50 hours per calendar year in non-emergency situations. The 50 hours of operation in non-emergency situations are counted as part of the 100 hours per calendar year for maintenance and testing and emergency demand response provided in paragraph (f)(2) of this section. Except as provided in paragraph (f)(3)(i) of this section, the 50 hours per calendar year for nonemergency situations cannot be used for peak shaving or non-emergency demand response, or to generate income for a facility to an electric grid or otherwise supply power as part of a financial arrangement with another entity.
- (i) The 50 hours per year for nonemergency situations can be used to supply power as part of a financial arrangement with another entity if all of the following conditions are met:
- (A) The engine is dispatched by the local balancing authority or local transmission and distribution system operator;
- (B) The dispatch is intended to mitigate local transmission and/or distribution limitations so as to avert potential voltage collapse or line overloads

that could lead to the interruption of power supply in a local area or region.

- (C) The dispatch follows reliability, emergency operation or similar protocols that follow specific NERC, regional, state, public utility commission or local standards or guidelines.
- (D) The power is provided only to the facility itself or to support the local transmission and distribution system.
- (E) The owner or operator identifies and records the entity that dispatches the engine and the specific NERC, regional, state, public utility commission or local standards or guidelines that are being followed for dispatching the engine. The local balancing authority or local transmission and distribution system operator may keep these records on behalf of the engine owner or operator.
  - (ii) [Reserved]
- (g) If you do not install, configure, operate, and maintain your engine and control device according to the manufacturer's emission-related written instructions, or you change emission-related settings in a way that is not permitted by the manufacturer, you must demonstrate compliance as follows:
- (1) If you are an owner or operator of a stationary CI internal combustion engine with maximum engine power less than 100 HP, you must keep a maintenance plan and records of conducted maintenance to demonstrate compliance and must, to the extent practicable, maintain and operate the engine in a manner consistent with good air pollution control practice for minimizing emissions. In addition, if you do not install and configure the engine and control device according to the manufacturer's emission-related written instructions, or you change the emission-related settings in a way that is not permitted by the manufacturer, you must conduct an initial performance test to demonstrate compliance with the applicable emission standards within 1 year of such action.
- (2) If you are an owner or operator of a stationary CI internal combustion engine greater than or equal to 100 HP and less than or equal to 500 HP, you must keep a maintenance plan and records of conducted maintenance and must, to the extent practicable, maintain and operate the engine in a man-

ner consistent with good air pollution control practice for minimizing emissions. In addition, you must conduct an initial performance test to demonstrate compliance with the applicable emission standards within 1 year of startup, or within 1 year after an engine and control device is no longer installed, configured, operated, and maintained in accordance with the manufacturer's emission-related written instructions, or within 1 year after you change emission-related settings in a way that is not permitted by the manufacturer.

(3) If you are an owner or operator of a stationary CI internal combustion engine greater than 500 HP, you must keep a maintenance plan and records of conducted maintenance and must, to the extent practicable, maintain and operate the engine in a manner consistent with good air pollution control practice for minimizing emissions. In addition, you must conduct an initial performance test to demonstrate compliance with the applicable emission standards within 1 year of startup, or within 1 year after an engine and control device is no longer installed, configured, operated, and maintained in accordance with the manufacturer's emission-related written instructions. or within 1 year after you change emission-related settings in a way that is not permitted by the manufacturer. You must conduct subsequent performance testing every 8,760 hours of engine operation or 3 years, whichever comes first, thereafter to demonstrate compliance with the applicable emission standards.

[71 FR 39172, July 11, 2006, as amended at 76 FR 37970, June 28, 2011; 78 FR 6695, Jan. 30, 2013]

TESTING REQUIREMENTS FOR OWNERS AND OPERATORS

§ 60.4212 What test methods and other procedures must I use if I am an owner or operator of a stationary CI internal combustion engine with a displacement of less than 30 liters per cylinder?

Owners and operators of stationary CI ICE with a displacement of less than

#### **Environmental Protection Agency**

30 liters per cylinder who conduct performance tests pursuant to this subpart must do so according to paragraphs (a) through (e) of this section.

- (a) The performance test must be conducted according to the in-use testing procedures in 40 CFR part 1039, subpart F, for stationary CI ICE with a displacement of less than 10 liters per cylinder, and according to 40 CFR part 1042, subpart F, for stationary CI ICE with a displacement of greater than or equal to 10 liters per cylinder and less than 30 liters per cylinder.
- (b) Exhaust emissions from stationary CI ICE that are complying with the emission standards for new CI engines in 40 CFR part 1039 must not ex-

ceed the not-to-exceed (NTE) standards for the same model year and maximum engine power as required in 40 CFR 1039.101(e) and 40 CFR 1039.102(g)(1), except as specified in 40 CFR 1039.104(d). This requirement starts when NTE requirements take effect for nonroad diesel engines under 40 CFR part 1039.

(c) Exhaust emissions from stationary CI ICE that are complying with the emission standards for new CI engines in 40 CFR 89.112 or 40 CFR 94.8, as applicable, must not exceed the NTE numerical requirements, rounded to the same number of decimal places as the applicable standard in 40 CFR 89.112 or 40 CFR 94.8, as applicable, determined from the following equation:

NTE requirement for each pollutant =  $(1.25) \times (STD)$  (Eq. 1)

Where:

STD = The standard specified for that pollutant in 40 CFR 89.112 or 40 CFR 94.8, as applicable.

Alternatively, stationary CI ICE that are complying with the emission standards for new CI engines in 40 CFR 89.112 or 40 CFR 94.8 may follow the testing procedures specified in §60.4213 of this subpart, as appropriate.

(d) Exhaust emissions from stationary CI ICE that are complying with the emission standards for pre-2007 model year engines in §60.4204(a), §60.4205(a), or §60.4205(c) must not exceed the NTE numerical requirements, rounded to the same number of decimal places as the applicable standard in §60.4204(a), §60.4205(a), or §60.4205(c), determined from the equation in paragraph (c) of this section.

Where:

Alternatively, stationary CI ICE that are complying with the emission standards for pre-2007 model year engines in §60.4204(a), §60.4205(a), or §60.4205(c) may follow the testing procedures specified in §60.4213, as appropriate.

(e) Exhaust emissions from stationary CI ICE that are complying with the emission standards for new CI en-

gines in 40 CFR part 1042 must not exceed the NTE standards for the same model year and maximum engine power as required in 40 CFR 1042.101(c).

[71 FR 39172, July 11, 2006, as amended at 76 FR 37971, June 28, 2011]

§ 60.4213 What test methods and other procedures must I use if I am an owner or operator of a stationary CI internal combustion engine with a displacement of greater than or equal to 30 liters per cylinder?

Owners and operators of stationary CI ICE with a displacement of greater than or equal to 30 liters per cylinder must conduct performance tests according to paragraphs (a) through (f) of this section.

- (a) Each performance test must be conducted according to the requirements in \$60.8 and under the specific conditions that this subpart specifies in table 7. The test must be conducted within 10 percent of 100 percent peak (or the highest achievable) load.
- (b) You may not conduct performance tests during periods of startup, shutdown, or malfunction, as specified in §60.8(c).
- (c) You must conduct three separate test runs for each performance test required in this section, as specified in §60.8(f). Each test run must last at least 1 hour.

#### §60.4213

- (d) To determine compliance with the percent reduction requirement, you must follow the requirements as specified in paragraphs (d)(1) through (3) of this section.
- (1) You must use Equation 2 of this section to determine compliance with the percent reduction requirement:

$$\frac{C_i - C_o}{C_i} \times 100 = R \qquad (Eq. 2)$$

Where:

- $C_i$  = concentration of  $NO_X$  or PM at the control device inlet,
- $C_o = \text{concentration of NO}_X \text{ or PM at the control device outlet, and}$
- $R = percent reduction of NO_X or PM emissions.$
- (2) You must normalize the  $NO_X$  or PM concentrations at the inlet and outlet of the control device to a dry basis and to 15 percent oxygen  $(O_2)$  using Equation 3 of this section, or an equivalent percent carbon dioxide  $(CO_2)$  using the procedures described in paragraph (d)(3) of this section.

$$C_{adj} = C_d \frac{5.9}{20.9 - \% O_2}$$
 (Eq. 3)

Where:

- $C_{adj}$  = Calculated NO<sub>X</sub> or PM concentration adjusted to 15 percent O<sub>2</sub>.  $C_d$  = Measured concentration of NO<sub>X</sub> or PM,
- $C_d$  = Measured concentration of  $NO_X$  or PM, uncorrected.
- 5.9 = 20.9 percent  $O_2-15$  percent  $O_2$ , the defined  $O_2$  correction value, percent.
- $%O_2$  = Measured  $O_2$  concentration, dry basis, percent.
- (3) If pollutant concentrations are to be corrected to 15 percent  $O_2$  and  $CO_2$  concentration is measured in lieu of  $O_2$  concentration measurement, a  $CO_2$  correction factor is needed. Calculate the  $CO_2$  correction factor as described in paragraphs (d)(3)(i) through (iii) of this section.
- (i) Calculate the fuel-specific  $F_{\rm o}$  value for the fuel burned during the

test using values obtained from Method 19, Section 5.2, and the following equation:

$$F_o = \frac{0.209_{F_d}}{F_c}$$
 (Eq. 4)

Where:

- $F_{\rm o}$  = Fuel factor based on the ratio of  $O_2$  volume to the ultimate  $CO_2$  volume produced by the fuel at zero percent excess air.
- 0.209 = Fraction of air that is  $O_2$ , percent/100.  $F_d$  = Ratio of the volume of dry effluent gas to the gross calorific value of the fuel from Method 19, dsm<sup>3</sup>/J (dscf/10<sup>6</sup> Btu).
- $F_c$  = Ratio of the volume of  $CO_2$  produced to the gross calorific value of the fuel from Method 19, dsm<sup>3</sup>/J (dscf/10<sup>6</sup> Btu).
- (ii) Calculate the  $CO_2$  correction factor for correcting measurement data to 15 percent  $O_2$ , as follows:

$$X_{co_2} = \frac{5.9}{F_0}$$
 (Eq. 5)

Where:

- $X_{CO2}$  =  $CO_2$  correction factor, percent. 5.9 = 20.9 percent  $O_2$ -15 percent  $O_2$ , the defined  $O_2$  correction value, percent.
- (iii) Calculate the  $NO_X$  and PM gas concentrations adjusted to 15 percent  $O_2$  using  $CO_2$  as follows:

$$C_{adj} = C_d \frac{X_{CO_2}}{{}^{0}\!\!/\!\!\!/ CO_2}$$
 (Eq. 6)

Where:

- $C_{adj}$  = Calculated  $NO_{\rm X}$  or PM concentration adjusted to 15 percent  $O_2$ .
- $C_d$  = Measured concentration of  $NO_X$  or PM, uncorrected.
- %CO<sub>2</sub> = Measured CO<sub>2</sub> concentration, dry basis, percent.
- (e) To determine compliance with the  $NO_X$  mass per unit output emission limitation, convert the concentration of  $NO_X$  in the engine exhaust using Equation 7 of this section:

$$ER = \frac{C_d \times 1.912 \times 10^{-3} \times Q \times T}{KW\text{-hour}}$$
 (Eq. 7)

Where

ER = Emission rate in grams per KW-hour.  $C_d$  = Measured NO<sub>X</sub> concentration in ppm.

1.912x10<sup>-3</sup> = Conversion constant for ppm NO<sub>x</sub> to grams per standard cubic meter at 25 degrees Celsius.

Q = Stack gas volumetric flow rate, in standard cubic meter per hour.

T = Time of test run, in hours.

KW-hour = Brake work of the engine, in KW-hour.

(f) To determine compliance with the PM mass per unit output emission limitation, convert the concentration of PM in the engine exhaust using Equation 8 of this section:

$$ER = \frac{C_{adj} \times Q \times T}{KW-hour}$$
 (Eq. 8)

Where:

ER = Emission rate in grams per KW-hour.

C<sub>adj</sub> = Calculated PM concentration in grams
per standard cubic meter.

Q = Stack gas volumetric flow rate, in standard cubic meter per hour.

T = Time of test run, in hours.

KW-hour = Energy output of the engine, in KW.

[71 FR 39172, July 11, 2006, as amended at 76 FR 37971, June 28, 2011]

NOTIFICATION, REPORTS, AND RECORDS FOR OWNERS AND OPERATORS

# § 60.4214 What are my notification, reporting, and recordkeeping requirements if I am an owner or operator of a stationary CI internal combustion engine?

- (a) Owners and operators of nonemergency stationary CI ICE that are greater than 2,237 KW (3,000 HP), or have a displacement of greater than or equal to 10 liters per cylinder, or are pre-2007 model year engines that are greater than 130 KW (175 HP) and not certified, must meet the requirements of paragraphs (a)(1) and (2) of this section.
- (1) Submit an initial notification as required in  $\S 60.7(a)(1)$ . The notification must include the information in paragraphs (a)(1)(i) through (v) of this section.
- (i) Name and address of the owner or operator;
- (ii) The address of the affected source;
- (iii) Engine information including make, model, engine family, serial

number, model year, maximum engine power, and engine displacement;

- (iv) Emission control equipment; and
- (v) Fuel used.
- (2) Keep records of the information in paragraphs (a)(2)(i) through (iv) of this section.
- (i) All notifications submitted to comply with this subpart and all documentation supporting any notification.
- (ii) Maintenance conducted on the engine.
- (iii) If the stationary CI internal combustion is a certified engine, documentation from the manufacturer that the engine is certified to meet the emission standards.
- (iv) If the stationary CI internal combustion is not a certified engine, documentation that the engine meets the emission standards.
- (b) If the stationary CI internal combustion engine is an emergency stationary internal combustion engine, the owner or operator is not required to submit an initial notification. Starting with the model years in table 5 to this subpart, if the emergency engine does not meet the standards applicable to non-emergency engines in the applicable model year, the owner or operator must keep records of the operation of the engine in emergency and non-emergency service that are recorded through the non-resettable hour meter. The owner must record the time of operation of the engine and the reason the engine was in operation during
- (c) If the stationary CI internal combustion engine is equipped with a diesel particulate filter, the owner or operator must keep records of any corrective action taken after the backpressure monitor has notified the owner or operator that the high backpressure limit of the engine is approached.
- (d) If you own or operate an emergency stationary CI ICE with a maximum engine power more than 100 HP that operates or is contractually obligated to be available for more than 15 hours per calendar year for the purposes specified in §60.4211(f)(2)(ii) and (iii) or that operates for the purposes specified in §60.4211(f)(3)(i), you must submit an annual report according to

#### § 60.4215

the requirements in paragraphs (d)(1) through (3) of this section.

- (1) The report must contain the following information:
- (i) Company name and address where the engine is located.
- (ii) Date of the report and beginning and ending dates of the reporting period.
- (iii) Engine site rating and model year.
- (iv) Latitude and longitude of the engine in decimal degrees reported to the fifth decimal place.
- (v) Hours operated for the purposes specified in  $\S 60.4211(f)(2)(ii)$  and (iii), including the date, start time, and end time for engine operation for the purposes specified in  $\S 60.4211(f)(2)(ii)$  and (iii).
- (vi) Number of hours the engine is contractually obligated to be available for the purposes specified in §60.4211(f)(2)(ii) and (iii).
- (vii) Hours spent for operation for the purposes specified in §60.4211(f)(3)(i), including the date, start time, and end time for engine operation for the purposes specified in §60.4211(f)(3)(i). The report must also identify the entity that dispatched the engine and the situation that necessitated the dispatch of the engine.
- (2) The first annual report must cover the calendar year 2015 and must be submitted no later than March 31, 2016. Subsequent annual reports for each calendar year must be submitted no later than March 31 of the following calendar year.
- (3) The annual report must be submitted electronically using the subpart specific reporting form in the Compliance and Emissions Data Reporting Interface (CEDRI) that is accessed through EPA's Central Data Exchange (CDX) (www.epa.gov/cdx). However, if the reporting form specific to this subpart is not available in CEDRI at the time that the report is due, the written report must be submitted to the Administrator at the appropriate address listed in §60.4.

[71 FR 39172, July 11, 2006, as amended at 78 FR 6696, Jan. 30, 2013]

SPECIAL REQUIREMENTS

# § 60.4215 What requirements must I meet for engines used in Guam, American Samoa, or the Commonwealth of the Northern Mariana Islands?

- (a) Stationary CI ICE with a displacement of less than 30 liters per cylinder that are used in Guam, American Samoa, or the Commonwealth of the Northern Mariana Islands are required to meet the applicable emission standards in §§ 60.4202 and 60.4205.
- (b) Stationary CI ICE that are used in Guam, American Samoa, or the Commonwealth of the Northern Mariana Islands are not required to meet the fuel requirements in §60.4207.
- (c) Stationary CI ICE with a displacement of greater than or equal to 30 liters per cylinder that are used in Guam, American Samoa, or the Commonwealth of the Northern Mariana Islands are required to meet the following emission standards:
- (1) For engines installed prior to January 1, 2012, limit the emissions of  $NO_X$  in the stationary CI internal combustion engine exhaust to the following:
- (i) 17.0 g/KW-hr (12.7 g/HP-hr) when maximum engine speed is less than 130 rpm;
- (ii)  $45 \cdot n^{-0.2}$  g/KW-hr ( $34 \cdot n^{-0.2}$  g/HP-hr) when maximum engine speed is 130 or more but less than 2,000 rpm, where n is maximum engine speed; and
- (iii) 9.8 g/KW-hr (7.3 g/HP-hr) when maximum engine speed is 2,000 rpm or more.
- (2) For engines installed on or after January 1, 2012, limit the emissions of  $NO_{\rm X}$  in the stationary CI internal combustion engine exhaust to the following:
- (i) 14.4 g/KW-hr (10.7 g/HP-hr) when maximum engine speed is less than 130 rpm:
- (ii)  $44 \cdot n^{-0.23}$  g/KW-hr ( $33 \cdot n^{-0.23}$  g/HP-hr) when maximum engine speed is greater than or equal to 130 but less than 2,000 rpm and where n is maximum engine speed; and
- (iii) 7.7 g/KW-hr (5.7 g/HP-hr) when maximum engine speed is greater than or equal to 2,000 rpm.

(3) Limit the emissions of PM in the stationary CI internal combustion engine exhaust to 0.40 g/KW-hr (0.30 g/HP-hr)

[71 FR 39172, July 11, 2006, as amended at 76 FR 37971, June 28, 2011]

## § 60.4216 What requirements must I meet for engines used in Alaska?

- (a) Prior to December 1, 2010, owners and operators of stationary CI ICE with a displacement of less than 30 liters per cylinder located in areas of Alaska not accessible by the FAHS should refer to 40 CFR part 69 to determine the diesel fuel requirements applicable to such engines.
- (b) Except as indicated in paragraph (c) of this section, manufacturers, owners and operators of stationary CI ICE with a displacement of less than 10 liters per cylinder located in areas of Alaska not accessible by the FAHS may meet the requirements of this subpart by manufacturing and installing engines meeting the requirements of 40 CFR parts 94 or 1042, as appropriate, rather than the otherwise applicable requirements of 40 CFR parts 89 and 1039, as indicated in sections §§ 60.4201(f) and 60.4202(g) of this subpart.
- (c) Manufacturers, owners and operators of stationary CI ICE that are located in areas of Alaska not accessible by the FAHS may choose to meet the applicable emission standards for emergency engines in §60.4202 and §60.4205, and not those for non-emergency engines in §60.4201 and §60.4204, except that for 2014 model year and later non-emergency CI ICE, the owner or operator of any such engine that was not certified as meeting Tier 4 PM standards, must meet the applicable requirements for PM in §60.4201 and §60.4204 or install a PM emission control device that achieves PM emission reductions of 85 percent, or 60 percent for engines with a displacement of greater than or equal to 30 liters per cylinder, compared to engine-out emissions.
- (d) The provisions of §60.4207 do not apply to owners and operators of pre-2014 model year stationary CI ICE subject to this subpart that are located in areas of Alaska not accessible by the FAHS.

- (e) The provisions of §60.4208(a) do not apply to owners and operators of stationary CI ICE subject to this subpart that are located in areas of Alaska not accessible by the FAHS until after December 31, 2009.
- (f) The provisions of this section and \$60.4207 do not prevent owners and operators of stationary CI ICE subject to this subpart that are located in areas of Alaska not accessible by the FAHS from using fuels mixed with used lubricating oil, in volumes of up to 1.75 percent of the total fuel. The sulfur content of the used lubricating oil must be less than 200 parts per million. The used lubricating oil must meet the onspecification levels and properties for used oil in 40 CFR 279.11.

[76 FR 37971, June 28, 2011]

#### § 60.4217 What emission standards must I meet if I am an owner or operator of a stationary internal combustion engine using special fuels?

Owners and operators of stationary CI ICE that do not use diesel fuel may petition the Administrator for approval of alternative emission standards, if they can demonstrate that they use a fuel that is not the fuel on which the manufacturer of the engine certified the engine and that the engine cannot meet the applicable standards required in §60.4204 or §60.4205 using such fuels and that use of such fuel is appropriate and reasonably necessary. considering cost, energy, technical feasibility, human health and environmental, and other factors, for the operation of the engine.

[76 FR 37972, June 28, 2011]

#### GENERAL PROVISIONS

## § 60.4218 What parts of the General Provisions apply to me?

Table 8 to this subpart shows which parts of the General Provisions in §§ 60.1 through 60.19 apply to you.

#### DEFINITIONS

## § 60.4219 What definitions apply to this subpart?

As used in this subpart, all terms not defined herein shall have the meaning given them in the CAA and in subpart A of this part.

#### § 60.4219

Certified emissions life means the period during which the engine is designed to properly function in terms of reliability and fuel consumption, without being remanufactured, specified as a number of hours of operation or calendar years, whichever comes first. The values for certified emissions life for stationary CI ICE with a displacement of less than 10 liters per cylinder are given in 40 CFR 1039.101(g). The values for certified emissions life for stationary CI ICE with a displacement of greater than or equal to 10 liters per cylinder and less than 30 liters per cylinder are given in 40 CFR 94.9(a).

Combustion turbine means all equipment, including but not limited to the turbine, the fuel, air, lubrication and exhaust gas systems, control systems (except emissions control equipment), and any ancillary components and subcomponents comprising any simple cycle combustion turbine, any regenerative/recuperative cycle combustion turbine, the combustion turbine portion of any cogeneration cycle combustion system, or the combustion turbine portion of any combined cycle steam/electric generating system.

Compression ignition means relating to a type of stationary internal combustion engine that is not a spark ignition engine.

Date of manufacture means one of the following things:

- (1) For freshly manufactured engines and modified engines, date of manufacture means the date the engine is originally produced.
- (2) For reconstructed engines, date of manufacture means the date the engine was originally produced, except as specified in paragraph (3) of this definition
- (3) Reconstructed engines are assigned a new date of manufacture if the fixed capital cost of the new and refurbished components exceeds 75 percent of the fixed capital cost of a comparable entirely new facility. An engine that is produced from a previously used engine block does not retain the date of manufacture of the engine in which the engine block was previously used if the engine is produced using all new components except for the engine block. In these cases, the date of manu-

facture is the date of reconstruction or the date the new engine is produced.

Diesel fuel means any liquid obtained from the distillation of petroleum with a boiling point of approximately 150 to 360 degrees Celsius. One commonly used form is number 2 distillate oil.

Diesel particulate filter means an emission control technology that reduces PM emissions by trapping the particles in a flow filter substrate and periodically removes the collected particles by either physical action or by oxidizing (burning off) the particles in a process called regeneration.

Emergency stationary internal combustion engine means any stationary reciprocating internal combustion engine that meets all of the criteria in paragraphs (1) through (3) of this definition. All emergency stationary ICE must comply with the requirements specified in §60.4211(f) in order to be considered emergency stationary ICE. If the engine does not comply with the requirements specified in §60.4211(f), then it is not considered to be an emergency stationary ICE under this subpart.

- (1) The stationary ICE is operated to provide electrical power or mechanical work during an emergency situation. Examples include stationary ICE used to produce power for critical networks or equipment (including power supplied to portions of a facility) when electric power from the local utility (or the normal power source, if the facility runs on its own power production) is interrupted, or stationary ICE used to pump water in the case of fire or flood, etc.
- (2) The stationary ICE is operated under limited circumstances for situations not included in paragraph (1) of this definition, as specified in §60.4211(f).
- (3) The stationary ICE operates as part of a financial arrangement with another entity in situations not included in paragraph (1) of this definition only as allowed in §60.4211(f)(2)(ii) or (iii) and §60.4211(f)(3)(i).

Engine manufacturer means the manufacturer of the engine. See the definition of "manufacturer" in this section.

Fire pump engine means an emergency stationary internal combustion engine certified to NFPA requirements that is

used to provide power to pump water for fire suppression or protection.

Freshly manufactured engine means an engine that has not been placed into service. An engine becomes freshly manufactured when it is originally produced.

Installed means the engine is placed and secured at the location where it is intended to be operated.

Manufacturer has the meaning given in section 216(1) of the Act. In general, this term includes any person who manufactures a stationary engine for sale in the United States or otherwise introduces a new stationary engine into commerce in the United States. This includes importers who import stationary engines for sale or resale.

Maximum engine power means maximum engine power as defined in 40 CFR 1039.801.

Model year means the calendar year in which an engine is manufactured (see "date of manufacture"), except as follows:

- (1) Model year means the annual new model production period of the engine manufacturer in which an engine is manufactured (see "date of manufacture"), if the annual new model production period is different than the calendar year and includes January 1 of the calendar year for which the model year is named. It may not begin before January 2 of the previous calendar year and it must end by December 31 of the named calendar year.
- (2) For an engine that is converted to a stationary engine after being placed into service as a nonroad or other non-stationary engine, model year means the calendar year or new model production period in which the engine was manufactured (see "date of manufacture").

Other internal combustion engine means any internal combustion engine, except combustion turbines, which is not a reciprocating internal combus-

tion engine or rotary internal combustion engine.

Reciprocating internal combustion engine means any internal combustion engine which uses reciprocating motion to convert heat energy into mechanical work.

Rotary internal combustion engine means any internal combustion engine which uses rotary motion to convert heat energy into mechanical work.

Spark ignition means relating to a gasoline, natural gas, or liquefied petroleum gas fueled engine or any other type of engine with a spark plug (or other sparking device) and with operating characteristics significantly similar to the theoretical Otto combustion cycle. Spark ignition engines usually use a throttle to regulate intake air flow to control power during normal operation. Dual-fuel engines in which a liquid fuel (typically diesel fuel) is used for CI and gaseous fuel (typically natural gas) is used as the primary fuel at an annual average ratio of less than 2 parts diesel fuel to 100 parts total fuel on an energy equivalent basis are spark ignition engines.

Stationary internal combustion engine means any internal combustion engine, except combustion turbines, that converts heat energy into mechanical work and is not mobile. Stationary ICE differ from mobile ICE in that a stationary internal combustion engine is not a nonroad engine as defined at 40 CFR 1068.30 (excluding paragraph (2)(ii) of that definition), and is not used to propel a motor vehicle, aircraft, or a vehicle used solely for competition. Stationary ICE include reciprocating ICE, rotary ICE, and other ICE, except combustion turbines.

Subpart means 40 CFR part 60, subpart IIII.

[71 FR 39172, July 11, 2006, as amended at 76 FR 37972, June 28, 2011; 78 FR 6696, Jan. 30, 2013]

#### Pt. 60, Subpt. IIII, Table 1

Table 1 to Subpart IIII of Part 60—Emission Standards for Stationary Pre-2007 Model Year Engines With a Displacement of <10 Liters per Cylinder and 2007–2010 Model Year Engines >2,237 KW (3,000 HP) and With a Displacement of <10 Liters per Cylinder

[As stated in §§ 60.4201(b), 60.4202(b), 60.4204(a), and 60.4205(a), you must comply with the following emission standards]

Maximum engine power	Emission standards for stationary pre-2007 model year engines with a displacement of <10 liters per cylinder and 2007–2010 model year engines >2,237 KW (3,000 HP) and with a displacement of <10 liters per cylinder in g/KW-hr (g/HP-hr)					
	NMHC + NO <sub>X</sub>	нс	NO <sub>x</sub>	со	PM	
KW<8 (HP<11)	10.5 (7.8)			8.0 (6.0)	1.0 (0.75)	
8≤KW<19 (11≤HP<25)	9.5 (7.1)			6.6 (4.9)	0.80 (0.60)	
19≤KW<37 (25≤HP<50)	9.5 (7.1)			5.5 (4.1)	0.80 (0.60)	
37≤KW<56 (50≤HP<75)			9.2 (6.9)			
56≤KW<75 (75≤HP<100)			9.2 (6.9)			
75≤KW<130 (100≤HP<175)			9.2 (6.9)		***************************************	
130≤KW<225 (175≤HP<300)		1.3 (1.0)	9.2 (6.9)	11.4 (8.5)	0.54 (0.40)	
225≤KW<450 (300≤HP<600)		1.3 (1.0)	9.2 (6.9)	11.4 (8.5)	0.54 (0.40)	
450≤KW≤560 (600≤HP≤750)		1.3 (1.0)	9.2 (6.9)	11.4 (8.5)	0.54 (0.40)	
KW>560 (HP>750)		1.3 (1.0)	9.2 (6.9)	11.4 (8.5)	0.54 (0.40)	

Table 2 to Subpart IIII of Part 60—Emission Standards for 2008 Model Year and Later Emergency Stationary CI ICE <37 KW (50 HP) With a Displacement of <10 Liters per Cylinder

[As stated in § 60.4202(a)(1), you must comply with the following emission standards]

Engine power	Emission standards for 2008 model year and later emergency stationary CI ICE <37 KW (50 HP) with a displacement of <10 liters per cylinder in g/KW-hr (g/ HP-hr)			
	Model year(s)	NO <sub>x</sub> + NMHC	со	PM
KW<8 (HP<11)	2008+ 2008+ 2008+	7.5 (5.6) 7.5 (5.6) 7.5 (5.6)	8.0 (6.0) 6.6 (4.9) 5.5 (4.1)	0.40 (0.30) 0.40 (0.30) 0.30 (0.22)

Table 3 to Subpart IIII of Part 60—Certification Requirements for Stationary Fire Pump Engines

As stated in §60.4202(d), you must certify new stationary fire pump engines beginning with the following model years:

As stated in § 60.4202(d), you must certify new stationary fire pump engines beginning with the following model years:

Engine power	Starting model year engine manufacturers must certify new stationary fire pump engines according to § 60.4202(d)¹
KW<75 (HP<100) 75≤KW<130	2011
(100≤HP<175) 130≤KW≤560	2010
(175≤HP≤750)	2009

Engine power	Starting model year engine manufacturers must certify new stationary fire pump engines according to § 60.4202(d)¹
KW>560 (HP>750)	2008

¹Manufacturers of fire pump stationary CI ICE with a maximum engine power greater than or equal to 37 kW (50 HP) and less than 450 KW (600 HP) and a rated speed of greater than 2,650 revolutions per minute (rpm) are not required to certify such engines until three model years following the model year indicated in this Table 3 for engines in the applicable engine power category.

[71 FR 39172, July 11, 2006, as amended at 76 FR 37972, June 28, 2011]

#### **Environmental Protection Agency**

Pt. 60, Subpt. IIII, Table 6

#### TABLE 4 TO SUBPART IIII OF PART 60—EMISSION STANDARDS FOR STATIONARY FIRE PUMP ENGINES

[As stated in §§ 60.4202(d) and 60.4205(c), you must comply with the following emission standards for stationary fire pump engines]

Maximum engine power	Model year(s)	NMHC + NO <sub>x</sub>	со	PM
KW<8 (HP<11)	1	10.5 (7.8)	8.0 (6.0)	1.0 (0.75)
		7.5 (5.6)		0.40 (0.30)
8≤KW<19 (11≤HP<25)	2010 and earlier	9.5 (7.1)	6.6 (4.9)	0.80 (0.60)
	2011+	7.5 (5.6)		0.40 (0.30)
19≤KW<37 (25≤HP<50)	2010 and earlier	9.5 (7.1)	5.5 (4.1)	0.80 (0.60)
	2011+	7.5 (5.6)		0.30 (0.22)
37≤KW<56 (50≤HP<75)		10.5 (7.8)	5.0 (3.7)	0.80 (0.60)
	2011+1	4.7 (3.5)		0.40 (0.30)
56≤KW<75 (75≤HP<100)	2010 and earlier	10.5 (7.8)	5.0 (3.7)	0.80 (0.60)
	2011+1	4.7 (3.5)		0.40 (0.30)
75≤KW<130 (100≤HP<175)	2009 and earlier	10.5 (7.8)	5.0 (3.7)	0.80 (0.60)
***************************************	2010+2	4.0 (3.0)	íí l	0.30 (0.22)
130≤KW<225 (175≤HP<300)	2008 and earlier	10.5 (7.8)	3.5 (2.6)	0.54 (0.40)
	2009+3	4.0 (3.0)	]	0.20 (0.15)
225≤KW<450 (300≤HP<600)	2008 and earlier	10.5 (7.8)	3.5 (2.6)	0.54 (0.40)
***************************************	2009+3	4.0 (3.0)		0.20 (0.15)
450≤KW≤560 (600≤HP≤750)		10.5 (7.8)		0.54 (0.40)
***************************************	2009+	4.0 (3.0)		0.20 (0.15)
KW>560 (HP>750)		10.5 (7.8)	f /	0.54 (0.40)
	2008+	6.4 (4.8)		0.20 (0.15)

#### TABLE 5 TO SUBPART IIII OF PART 60—LABELING AND RECORDKEEPING REQUIREMENTS FOR NEW STATIONARY EMERGENCY ENGINES

[You must comply with the labeling requirements in § 60.4210(f) and the recordkeeping requirements in § 60.4214(b) for new emergency stationary CI ICE beginning in the following model years:]

Engine power	Starting model year
19≤KW<56 (25≤HP<75)	2013
56≤KW<130 (75≤HP<175)	2012
KW≥130 (HP≥175)	2011

#### TABLE 6 TO SUBPART IIII OF PART 60—OPTIONAL 3-MODE TEST CYCLE FOR STATIONARY FIRE PUMP ENGINES

[As stated in §60.4210(g), manufacturers of fire pump engines may use the following test cycle for testing fire pump engines:]

Mode No.	Engine speed <sup>1</sup>	Torque (percent) 2	Weighting factors
1	Rated	100	0.30
2	Rated	75	0.50
3	Rated	50	0.20

¹For model years 2011–2013, manufacturers, owners and operators of fire pump stationary CI ICE in this engine power category with a rated speed of greater than 2,650 revolutions per minute (rpm) may comply with the emission limitations for 2010 model year engines.

²For model years 2010–2012, manufacturers, owners and operators of fire pump stationary CI ICE in this engine power category with a rated speed of greater than 2,650 rpm may comply with the emission limitations for 2009 model year engines.
³In model years 2009–2011, manufacturers of fire pump stationary CI ICE in this engine power category with a rated speed of greater than 2,650 rpm may comply with the emission limitations for 2008 model year engines.

<sup>&</sup>lt;sup>1</sup> Engine speed: ±2 percent of point.

<sup>2</sup> Torque: NFPA certified nameplate HP for 100 percent point. All points should be ±2 percent of engine percent load value.

# Pt. 60, Subpt. IIII, Table 7

TABLE 7 TO SUBPART IIII OF PART 60—REQUIREMENTS FOR PERFORMANCE TESTS FOR STATIONARY CI ICE WITH A DISPLACEMENT OF ≥30 LITERS PER CYLINDER
[As stated in § 60.4213, you must comply with the following requirements for performance tests for stationary CI ICE with a displacement of ≥30 liters per cylinder.]

For each	Complying with the requirement to	You must	Using	According to the fol- lowing requirements
1. Stationary CI internal combustion engine with a displacement of ≥30 liters per cylinder.	a. Reduce NO <sub>x</sub> emissions by 90 percent or more.	i. Select the sampling port location and the number of traverse points;	(1) Method 1 or 1A of 40 CFR part 60, ap- pendix A.	(a) Sampling sites mus be located at the inle and outlet of the con- trol device.
		ii. Measure O <sub>2</sub> at the inlet and outlet of the control device;	(2) Method 3, 3A, or 3B of 40 CFR part 60, appendix A.	(b) Measurements to determine O <sub>2</sub> concentration must be made at the same time as the measurements for NO <sub>X</sub> concentration.
		iii, If necessary, meas- ure moisture content at the inlet and outlet of the control device; and,	(3) Method 4 of 40 CFR part 60, appendix A, Method 320 of 40 CFR part 63, appendix A, or ASTM D 6348–03 (incorporated by reference, see § 60.17).	(c) Measurements to determine moisture content must be made at the same time as the measurements for NO <sub>X</sub> concentration.
		iv. Measure NO <sub>X</sub> at the inlet and outlet of the control device.	(4) Method 7E of 40 CFR part 60, appendix A, Method 320 of 40 CFR part 63, appendix A, or ASTM D 6348–03 (incorporated by reference, see § 60.17).	(d) NO <sub>X</sub> concentration must be at 15 per- cent O <sub>2</sub> , dry basis. Results of this test consist of the aver- age of the three 1- hour or longer runs.
	b. Limit the concentra- tion of NO <sub>X</sub> in the stationary CI internal combustion engine exhaust.	Select the sampling port location and the number of traverse points;	(1) Method 1 or 1A of 40 CFR part 60, ap- pendix A.	(a) If using a control device, the sampling site must be located at the outlet of the control device.
		ii. Determine the O <sub>2</sub> concentration of the stationary internal combustion engine exhaust at the sam- pling port location; and,	(2) Method 3, 3A, or 3B of 40 CFR part 60, appendix A.	(b) Measurements to determine O <sub>2</sub> concentration must be made at the same time as the measurement for NO <sub>X</sub> concentration.
		iii. If necessary, meas- ure moisture content of the stationary in- ternal combustion en- gine exhaust at the sampling port loca- tion; and,	(3) Method 4 of 40 CFR part 60, appendix A, Method 320 of 40 CFR part 63, appendix A, or ASTM D 6348–03 (incorporated by reference, see § 60.17).	(c) Measurements to determine moisture content must be made at the same time as the measurement for NO <sub>X</sub> concentration.
		iv. Measure NO <sub>x</sub> at the exhaust of the stationary internal combustion engine.	(4) Method 7E of 40 CFR part 60, appendix A, Method 320 of 40 CFR part 63, appendix A, or ASTM D 6348–03 (incorporated by reference, see § 60.17).	(d) NO <sub>x</sub> concentration must be at 15 per- cent O <sub>2</sub> , dry basis. Results of this test consist of the aver- age of the three 1- hour or longer runs.
	c. Reduce PM emissions by 60 percent or more.	i. Select the sampling port location and the number of traverse points;	(1) Method 1 or 1A of 40 CFR part 60, ap- pendix A.	(a) Sampling sites mus be located at the inle and outlet of the con- trol device.
		ii. Measure O <sub>2</sub> at the Inlet and outlet of the control device;	(2) Method 3, 3A, or 3B of 40 CFR part 60, appendix A.	(b) Measurements to determine O <sub>2</sub> con- centration must be made at the same time as the measure- ments for PM con- centration.

# Pt. 60, Subpt. IIII, Table 8

[As stated in §60.4213, you must comply with the following requirements for performance tests for stationary CI ICE with a displacement of ≥30 liters per cylinder:]

For each	Complying with the requirement to	You must	Using	According to the fol- lowing requirements
		iii. If necessary, meas- ure moisture content at the inlet and outlet of the control device; and	(3) Method 4 of 40 CFR part 60, appendix A.	(c) Measurements to determine and mois- ture content must be made at the same time as the measure- ments for PM con- centration.
		iv. Measure PM at the inlet and outlet of the control device.	(4) Method 5 of 40 CFR part 60, appendix A.	(d) PM concentration must be at 15 per- cent O <sub>2</sub> , dry basis. Results of this test consist of the aver- age of the three 1- hour or longer runs.
	d. Limit the concentra- tion of PM in the sta- tionary CI internal combustion engine exhaust.	i. Select the sampling port location and the number of traverse points;	(1) Method 1 or 1A of 40 CFR part 60, ap- pendix A.	(a) If using a control device, the sampling site must be located at the outlet of the control device.
		ii. Determine the O <sub>2</sub> concentration of the stationary internal combustion engine exhaust at the sampling port location; and	(2) Method 3, 3A, or 3B of 40 CFR part 60, appendix A.	(b) Measurements to determine O <sub>2</sub> con- centration must be made at the same time as the measure- ments for PM con- centration.
		iii. If necessary, meas- ure moisture content of the stationary in- ternal combustion en- gine exhaust at the sampling port loca- tion; and	(3) Method 4 of 40 CFR part 60, appendix A.	(c) Measurements to determine moisture content must be made at the same time as the measure- ments for PM con- centration.
		iv. Measure PM at the exhaust of the sta- tionary internal com- bustion engine.	(4) Method 5 of 40 CFR part 60, appendix A.	(d) PM concentration must be at 15 per- cent O <sub>2</sub> , dry basis. Results of this test consist of the aver- age of the three 1- hour or longer runs.

# TABLE 8 TO SUBPART IIII OF PART 60—APPLICABILITY OF GENERAL PROVISIONS TO SUBPART IIII [As stated in §60.4218, you must comply with the following applicable General Provisions:]

General Provisions citation	Subject of citation	Applies to subpart	Explanation
§ 60.1	General applicability of the General Provisions.	Yes.	
§ 60.2	Definitions	Yes	Additional terms defined in § 60.4219.
§ 60.3	Units and abbreviations	Yes.	_
§ 60.4	Address	Yes.	
§ 60.5	Determination of construction or modifica- tion.	Yes.	,
§ 60.6	Review of plans	Yes.	
§ 60.7	Notification and Recordkeeping	Yes	Except that § 60.7 only applies as specified in § 60.4214(a).
§ 60.8	Performance tests	Yes	Except that §60.8 only applies to stationary CI ICE with a displacement o (≥30 liters per cylinder and engines that are not certified.
§ 60.9	Availability of information	Yes.	
§ 60.10	State Authority	Yes.	
§ 60.11	Compliance with standards and mainte- nance requirements.		Requirements are specified in subpart IIII.
§ 60.12	Circumvention	Yes.	J

[As stated in § 60.4218, you must comply with the following applicable General Provisions:]

General Provisions citation	Subject of citation	Applies to subpart	Explanation
§ 60.13	Monitoring requirements	Yes	Except that §60.13 only applies to stationary CI ICE with a displacement of (230 liters per cylinder.
§ 60.14	Modification	Yes.	, , ,
§ 60.15	Reconstruction	Yes.	
§ 60.16	Priority list	Yes.	
§ 60.17	Incorporations by reference	Yes.	
§ 60.18	General control device requirements	No.	
§ 60.19	General notification and reporting requirements.	Yes.	

## Subpart JJJJ—Standards of Performance for Stationary Spark Ignition Internal Combustion Engines

Source: 73 FR 3591, Jan. 18, 2008, unless otherwise noted.

WHAT THIS SUBPART COVERS

## § 60.4230 Am I subject to this subpart?

- (a) The provisions of this subpart are applicable to manufacturers, owners, and operators of stationary spark ignition (SI) internal combustion engines (ICE) as specified in paragraphs (a)(1) through (6) of this section. For the purposes of this subpart, the date that construction commences is the date the engine is ordered by the owner or operator.
- (1) Manufacturers of stationary SI ICE with a maximum engine power less than or equal to 19 kilowatt (KW) (25 horsepower (HP)) that are manufactured on or after July 1, 2008.
- (2) Manufacturers of stationary SI ICE with a maximum engine power greater than 19 KW (25 HP) that are gasoline fueled or that are rich burn engines fueled by liquefied petroleum gas (LPG), where the date of manufacture is:
  - (i) On or after July 1, 2008; or
- (ii) On or after January 1, 2009, for emergency engines.
- (3) Manufacturers of stationary SI ICE with a maximum engine power greater than 19 KW (25 HP) that are not gasoline fueled and are not rich burn engines fueled by LPG, where the manufacturer participates in the voluntary manufacturer certification program de-

scribed in this subpart and where the date of manufacture is:

- (i) On or after July 1, 2007, for engines with a maximum engine power greater than or equal to 500 HP (except lean burn engines with a maximum engine power greater than or equal to 500 HP and less than 1,350 HP);
- (ii) On or after January 1, 2008, for lean burn engines with a maximum engine power greater than or equal to 500 HP and less than 1,350 HP;
- (iii) On or after July 1, 2008, for engines with a maximum engine power less than 500 HP; or
- (iv) On or after January 1, 2009, for emergency engines.
- (4) Owners and operators of stationary SI ICE that commence construction after June 12, 2006, where the stationary SI ICE are manufactured:
- (i) On or after July 1, 2007, for engines with a maximum engine power greater than or equal to 500 HP (except lean burn engines with a maximum engine power greater than or equal to 500 HP and less than 1,350 HP);
- (ii) on or after January 1, 2008, for lean burn engines with a maximum engine power greater than or equal to 500 HP and less than 1,350 HP;
- (iii) on or after July 1, 2008, for engines with a maximum engine power less than 500 HP; or
- (iv) on or after January 1, 2009, for emergency engines with a maximum engine power greater than 19 KW (25 HP).
- (5) Owners and operators of stationary SI ICE that are modified or reconstructed after June 12, 2006, and any person that modifies or reconstructs any stationary SI ICE after June 12, 2006.

Appendix I 40 CFR Part 61, Subpart FF National Emission Standards for Hazardous Air Pollutants, Benzene Waste Operations

for a source that has an initial startup date after the effective date.

- (1) Periods of operation where there were exceedances of monitored parameters recorded under §61.305(b).
- (2) All periods recorded under §61.305(c)(1) when the vent stream is diverted from the control device.
- (3) All periods recorded under §61.305(d) when the steam generating unit or process heater was not operating.
- (4) All periods recorded under §61.305(e) in which the pilot flame of the flare was absent.
- (5) All times recorded under §61.305(c)(2) when maintenance is performed on car-sealed valves, when the car seal is broken, and when the valve position is changed.
- (g) The owner or operator of an affected facility shall keep the vaportightness documentation required under §61.302 (d) and (e) on file at the affected facility in a permanent form available for inspection.
- (h) The owner or operator of an affected facility shall update the documentation file required under §61.302 (d) and (e) for each tank truck, railcar, or marine vessel at least once per year to reflect current test results as determined by the appropriate method. The owner or operator shall include, as a minimum, the following information in this documentation:
  - (1) Test title;
- (2) Tank truck, railcar, or marine vessel owner and address;
- (3) Tank truck, railcar, or marine vessel identification number;
  - (4) Testing location:
  - (5) Date of test:
  - (6) Tester name and signature;
- (7) Witnessing inspector: name, signature, and affiliation; and
- (8) Test results, including, for railcars and tank trucks, the initial pressure up to which the tank was pressured at the start of the test.
- (i) Each owner or operator of an affected facility complying with §61.300(b) or §61.300(d) shall record the following information. The first year after promulgation the owner or operator shall submit a report containing the requested information to the Director of the Emission Standards Division, (MD-13), U.S. Environmental Protec-

### 40 CFR Ch. I (7-1-13 Edition)

tion Agency, Research Triangle Park, North Carolina 27711. After the first year, the owner or operator shall continue to record; however, no reporting is required. The information shall be made available if requested. The information shall include, as a minimum:

- (1) The affected facility's name and address:
- (2) The weight percent of the benzene loaded;
- (3) The type of vessel loaded (i.e., tank truck, railcar, or marine vessel); and
- (4) The annual amount of benzene loaded into each type of vessel.

[55 FR 8341, Mar. 7, 1990, as amended at 65 FR 62159, Oct. 17, 2000]

#### § 61.306 Delegation of authority.

- (a) In delegating implementation and enforcement authority to a State under section 112(d) of the Act, the authorities contained in paragraph (b) of this section shall be retained by the Administrator and not transferred to a State.
- (b) Authorities which will not be delegated to States: No restrictions.

## Subparts CC-EE [Reserved]

### Subpart FF—National Emission Standard for Benzene Waste Operations

SOURCE: 55 FR 8346, Mar. 7, 1990, unless otherwise noted.

## § 61.340 Applicability.

- (a) The provisions of this subpart apply to owners and operators of chemical manufacturing plants, coke byproduct recovery plants, and petroleum refineries.
- (b) The provisions of this subpart apply to owners and operators of hazardous waste treatment, storage, and disposal facilities that treat, store, or dispose of hazardous waste generated by any facility listed in paragraph (a) of this section. The waste streams at hazardous waste treatment, storage, and disposal facilities subject to the provisions of this subpart are the benzene-containing hazardous waste from any facility listed in paragraph (a) of

this section. A hazardous waste treatment, storage, and disposal facility is a facility that must obtain a hazardous waste management permit under subtitle C of the Solid Waste Disposal Act.

- (c) At each facility identified in paragraph (a) or (b) of this section, the following waste is exempt from the requirements of this subpart:
- (1) Waste in the form of gases or vapors that is emitted from process fluids:
- (2) Waste that is contained in a segregated stormwater sewer system.
- (d) At each facility identified in paragraph (a) or (b) of this section, any gaseous stream from a waste management unit, treatment process, or wastewater treatment system routed to a fuel gas system, as defined in §61.341, is exempt from this subpart. No testing, monitoring, recordkeeping, or reporting is required under this subpart for any gaseous stream from a waste management unit, treatment process, or wastewater treatment unit routed to a fuel gas system.

[55 FR 8346, Mar. 7, 1990, as amended at 55 FR 37231, Sept. 10, 1990; 58 FR 3095, Jan. 7, 1993; 67 FR 68531, Nov. 12, 2002]

## § 61.341 Definitions.

Benzene concentration means the fraction by weight of benzene in a waste as determined in accordance with the procedures specified in §61.355 of this subpart.

Car-seal means a seal that is placed on a device that is used to change the position of a valve (e.g., from opened to closed) in such a way that the position of the valve cannot be changed without breaking the seal.

Chemical manufacturing plant means any facility engaged in the production of chemicals by chemical, thermal, physical, or biological processes for use as a product, co-product, by-product, or intermediate including but not limited to industrial organic chemicals, organic pesticide products, pharmaceutical preparations, paint and allied products, fertilizers, and agricultural chemicals. Examples of chemical manufacturing plants include facilities at which process units are operated to produce one or more of the following chemicals: benzenesulfonic acid, benchlorobenzene zene. cumene.

cyclohexane, ethylene, ethylbenzene, hydroquinone, linear alklylbenzene, nitrobenzene, resorcinol, sulfolane, or styrene.

Closed-vent system means a system that is not open to the atmosphere and is composed of piping, ductwork, connections, and, if necessary, flow inducing devices that transport gas or vapor from an emission source to a control device.

Coke by-product recovery plant means any facility designed and operated for the separation and recovery of coal tar derivatives (by-products) evolved from coal during the coking process of a coke oven battery.

Container means any portable waste management unit in which a material is stored, transported, treated, or otherwise handled. Examples of containers are drums, barrels, tank trucks, barges, dumpsters, tank cars, dump trucks, and ships.

Control device means an enclosed combustion device, vapor recovery system, or flare.

Cover means a device or system which is placed on or over a waste placed in a waste management unit so that the entire waste surface area is enclosed and sealed to minimize air emissions. A cover may have openings necessary for operation, inspection, and maintenance of the waste management unit such as access hatches, sampling ports, and gauge wells provided that each opening is closed and sealed when not in use. Example of covers include a fixed roof installed on a tank, a lid installed on a container, and an airsupported enclosure installed over a waste management unit.

External floating roof means a pontoon-type or double-deck type cover with certain rim sealing mechanisms that rests on the liquid surface in a waste management unit with no fixed roof.

Facility means all process units and product tanks that generate waste within a stationary source, and all waste management units that are used for waste treatment, storage, or disposal within a stationary source.

Fixed roof means a cover that is mounted on a waste management unit in a stationary manner and that does

not move with fluctuations in liquid level.

Floating roof means a cover with certain rim sealing mechanisms consisting of a double deck, pontoon single deck, internal floating cover or covered floating roof, which rests upon and is supported by the liquid being contained, and is equipped with a closure seal or seals to close the space between the roof edge and unit wall.

Flow indicator means a device which indicates whether gas flow is present in a line or vent system.

Fuel gas system means the offsite and onsite piping and control system that gathers gaseous streams generated by facility operations, may blend them with sources of gas, if available, and transports the blended gaseous fuel at suitable pressures for use as fuel in heaters, furnaces, boilers, incinerators, gas turbines, and other combustion devices located within or outside the facility. The fuel is piped directly to each individual combustion device, and the system typically operates at pressures over atmospheric.

Individual drain system means the system used to convey waste from a process unit, product storage tank, or waste management unit to a waste management unit. The term includes all process drains and common junction boxes, together with their associated sewer lines and other junction boxes, down to the receiving waste management unit.

Internal floating roof means a cover that rests or floats on the liquid surface inside a waste management unit that has a fixed roof.

Liquid-mounted seal means a foam or liquid-filled primary seal mounted in contact with the liquid between the waste management unit wall and the floating roof continuously around the circumference.

Loading means the introduction of waste into a waste management unit but not necessarily to complete capacity (also referred to as filling).

Maximum organic vapor pressure means the equilibrium partial pressure exerted by the waste at the temperature equal to the highest calendarmonth average of the waste storage temperature for waste stored above or below the ambient temperature or at

the local maximum monthly average temperature as reported by the National Weather Service for waste stored at the ambient temperature, as determined:

- (1) In accordance with §60.17(c); or
- (2) As obtained from standard reference texts; or
- (3) In accordance with §60.17(a)(37); or (4) Any other method approved by the Administrator.

No detectable emissions means less than 500 parts per million by volume (ppmv) above background levels, as measured by a detection instrument reading in accordance with the procedures specified in §61.355(h) of this subpart.

Oil-water separator means a waste management unit, generally a tank or surface impoundment, used to separate oil from water. An oil-water separator consists of not only the separation unit but also the forebay and other separator basins, skimmers, weirs, grit chambers, sludge hoppers, and bar screens that are located directly after the individual drain system and prior to additional treatment units such as an air flotation unit, clarifier, or biological treatment unit. Examples of an oil-water separator incude an API separator, parallel-plate interceptor, and corrugated-plate interceptor with the associated ancillary equipment.

Petroleum refinery means any facility engaged in producing gasoline, kerosene, distillate fuel oils, residual fuel oils, lubricants, or other products through the distillation of petroleum, or through the redistillation, cracking, or reforming of unfinished petroleum derivatives.

Petroleum means the crude oil removed from the earth and the oils derived from tar sands, shale, and coal.

Point of waste generation means the location where the waste stream exits the process unit component or storage tank prior to handling or treatment in an operation that is not an integral part of the production process, or in the case of waste management units that generate new wastes after treatment, the location where the waste stream exits the waste management unit component.

Process unit means equipment assembled and connected by pipes or ducts to

produce intermediate or final products. A process unit can be operated independently if supplied with sufficient fuel or raw materials and sufficient product storage facilities.

Process unit turnaround means the shutting down of the operations of a process unit, the purging of the contents of the process unit, the maintenance or repair work, followed by restarting of the process.

Process unit turnaround waste means a waste that is generated as a result of a process unit turnaround.

Process wastewater means water which comes in contact with benzene during manufacturing or processing operations conducted within a process unit. Process wastewater is not organic wastes, process fluids, product tank drawdown, cooling tower blowdown, steam trap condensate, or landfill leachate.

Process wastewater stream means a waste stream that contains only process wastewater.

Product tank means a stationary unit that is designed to contain an accumulation of materials that are fed to or produced by a process unit, and is constructed primarily of non-earthen materials (e.g., wood, concrete, steel, plastic) which provide structural support.

Product tank drawdown means any material or mixture of materials discharged from a product tank for the purpose of removing water or other contaminants from the product tank.

Safety device means a closure device such as a pressure relief valve, frangible disc, fusible plug, or any other type of device which functions exclusively to prevent physical damage or permanent deformation to a unit or its air emission control equipment by venting gases or vapors directly to the atmosphere during unsafe conditions resulting from an unplanned, accidental, or emergency event. For the purpose of this subpart, a safety device is not used for routine venting of gases or vapors from the vapor headspace underneath a cover such as during filling of the unit or to adjust the pressure in this vapor headspace in response to normal daily diurnal ambient temperature fluctuations. A safety device is designed to remain in a closed position during normal operations and open

only when the internal pressure, or another relevant parameter, exceeds the device threshold setting applicable to the air emission control equipment as determined by the owner or operator based on manufacturer recommendations, applicable regulations, fire protection and prevention codes, standard engineering codes and practices, or other requirements for the safe handling of flammable, ignitable, explosive, reactive, or hazardous materials.

Segregated stormwater sewer system means a drain and collection system designed and operated for the sole purpose of collecting rainfall runoff at a facility, and which is segregated from all other individual drain systems.

Sewer line means a lateral, trunk line, branch line, or other enclosed conduit used to convey waste to a downstream waste management unit.

Slop oil means the floating oil and solids that accumulate on the surface of an oil-water separator.

Sour water stream means a stream that:

- (1) Contains ammonia or sulfur compounds (usually hydrogen sulfide) at concentrations of 10 ppm by weight or more;
- (2) Is generated from separation of water from a feed stock, intermediate, or product that contained ammonia or sulfur compounds; and
- (3) Requires treatment to remove the ammonia or sulfur compounds.

Sour water stripper means a unit that:

- (1) Is designed and operated to remove ammonia or sulfur compounds (usually hydrogen sulfide) from sour water streams;
- (2) Has the sour water streams transferred to the stripper through hard piping or other enclosed system; and
- (3) Is operated in such a manner that the offgases are sent to a sulfur recovery unit, processing unit, incinerator, flare, or other combustion device.

Surface impoundment means a waste management unit which is a natural topographic depression, man-made excavation, or diked area formed primarily of earthen materials (although it may be lined with man-made materials), which is designed to hold an accumulation of liquid wastes or waste containing free liquids, and which is

not an injection well. Examples of surface impoundments are holding, storage, settling, and aeration pits, ponds, and lagoons.

Tank means a stationary waste management unit that is designed to contain an accumulation of waste and is constructed primarily of nonearthen materials (e.g., wood, concrete, steel, plastic) which provide structural support.

Treatment process means a stream stripping unit, thin-film evaporation unit, waste incinerator, or any other process used to comply with §61.348 of this subpart.

Vapor-mounted seal means a foamfilled primary seal mounted continuously around the perimeter of a waste management unit so there is an annular vapor space underneath the seal. The annular vapor space is bounded by the bottom of the primary seal, the unit wall, the liquid surface, and the floating roof.

Waste means any material resulting from industrial, commercial, mining or agricultural operations, or from community activities that is discarded or is being accumulated, stored, or physically, chemically, thermally, or biologically treated prior to being discarded, recycled, or discharged.

Waste management unit means a piece of equipment, structure, or transport mechanism used in handling, storage, treatment, or disposal of waste. Examples of a waste management unit include a tank, surface impoundment, container, oil-water separator, individual drain system, steam stripping unit, thin-film evaporation unit, waste incinerator, and landfill.

Waste stream means the waste generated by a particular process unit, product tank, or waste management unit. The characteristics of the waste stream (e.g., flow rate, benzene concentration, water content) are determined at the point of waste generation. Examples of a waste stream include process wastewater, product tank drawdown, sludge and slop oil removed from waste management units, and landfill leachate.

Wastewater treatment system means any component, piece of equipment, or installation that receives, manages, or treats process wastewater, product tank drawdown, or landfill leachate prior to direct or indirect discharge in accordance with the National Pollutant Discharge Elimination System permit regulations under 40 CFR part 122. These systems typically include individual drain systems, oil-water separators, air flotation units, equalization tanks, and biological treatment units.

Water seal controls means a seal pot, p-leg trap, or other type of trap filled with water (e.g., flooded sewers that maintain water levels adequate to prevent air flow through the system) that creates a water barrier between the sewer line and the atmosphere. The water level of the seal must be maintained in the vertical leg of a drain in order to be considered a water seal.

[55 FR 8346, Mar. 7, 1990; 55 FR 12444, Apr. 3, 1990, as amended at 58 FR 3095, Jan. 7, 1993; 67 FR 68531, Nov. 12, 2002]

#### §61.342 Standards: General.

(a) An owner or operator of a facility at which the total annual benzene quantity from facility waste is less than 10 megagrams per year (Mg/yr) (11 ton/yr) shall be exempt from the requirements of paragraphs (b) and (c) of this section. The total annual benzene quantity from facility waste is the sum of the annual benzene quantity for each waste stream at the facility that has a flow-weighted annual average water content greater than 10 percent or that is mixed with water, or other wastes, at any time and the mixture has an annual average water content greater than 10 percent. The benzene quantity in a waste stream is to be counted only once without multiple counting if other waste streams are mixed with or generated from the original waste stream. Other specific requirements for calculating the total annual benzene waste quantity are as follows:

(1) Wastes that are exempted from control under §§61.342(c)(2) and 61.342(c)(3) are included in the calculation of the total annual benzene quantity if they have an annual average water content greater than 10 percent, or if they are mixed with water or other wastes at any time and the mixture has an annual average water content greater than 10 percent.

- (2) The benzene in a material subject to this subpart that is sold is included in the calculation of the total annual benzene quantity if the material has an annual average water content greater than 10 percent.
- (3) Benzene in wastes generated by remediation activities conducted at the facility, such as the excavation of contaminated soil, pumping and treatment of groundwater, and the recovery of product from soil or groundwater, are not included in the calculation of total annual benzene quantity for that facility. If the facility's total annual benzene quantity is 10 Mg/yr (11 ton/yr) or more, wastes generated by remediation activities are subject to the requirements of paragraphs (c) through (h) of this section. If the facility is managing remediation waste generated offsite, the benzene in this waste shall be included in the calculation of total annual benzene quantity in facility waste, if the waste streams have an annual average water content greater than 10 percent, or if they are mixed with water or other wastes at any time and the mixture has an annual average water content greater than 10 percent.
- (4) The total annual benzene quantity is determined based upon the quantity of benzene in the waste before any waste treatment occurs to remove the benzene except as specified in §61.355(c)(1)(i) (A) through (C).
- (b) Each owner or operator of a facility at which the total annual benzene quantity from facility waste is equal to or greater than 10 Mg/yr (11 ton/yr) as determined in paragraph (a) of this section shall be in compliance with the requirements of paragraphs (c) through (h) of this section no later than 90 days following the effective date, unless a waiver of compliance has been obtained under §61.11, or by the initial startup for a new source with an initial startup after the effective date.
- (1) The owner or operator of an existing source unable to comply with the rule within the required time may request a waiver of compliance under §61.10.
- (2) As part of the waiver application, the owner or operator shall submit to the Administrator a plan under §61.10(b)(3) that is an enforceable commitment to obtain environmental ben-

- efits to mitigate the benzene emissions that result from extending the compliance date. The plan shall include the following information:
- (i) A description of the method of compliance, including the control approach, schedule for installing controls, and quantity of the benzene emissions that result from extending the compliance date;
- (ii) If the control approach involves a compliance strategy designed to obtain integrated compliance with multiple regulatory requirements, a description of the other regulations involved and their effective dates; and
- (iii) A description of the actions to be taken at the facility to obtain mitigating environmental benefits, including how the benefits will be obtained, the schedule for these actions, and an estimate of the quantifiable benefits that directly result from these actions.
- (c) Each owner or operator of a facility at which the total annual benzene quantity from facility waste is equal to or greater than 10 Mg/yr (11 ton/yr) as determined in paragraph (a) of this section shall manage and treat the facility waste as follows:
- (1) For each waste stream that contains benzene, including (but not limited to) organic waste streams that contain less than 10 percent water and aqueous waste streams, even if the wastes are not discharged to an individual drain system, the owner or operator shall:
- (i) Remove or destroy the benzene contained in the waste using a treatment process or wastewater treatment system that complies with the standards specified in §61.348 of this subpart.
- (ii) Comply with the standards specified in §§61.343 through 61.347 of this subpart for each waste management unit that receives or manages the waste stream prior to and during treatment of the waste stream in accordance with paragraph (c)(1)(i) of this section.
- (iii) Each waste management unit used to manage or treat waste streams that will be recycled to a process shall comply with the standards specified in §§ 61.343 through 61.347. Once the waste stream is recycled to a process, including to a tank used for the storage of production process feed, product, or

- product intermediates, unless this tank is used primarily for the storage of wastes, the material is no longer subject to paragraph (c) of this section.
- (2) A waste stream is exempt from paragraph (c)(1) of this section provided that the owner or operator demonstrates initially and, thereafter, at least once per year that the flow-weighted annual average benzene concentration for the waste stream is less than 10 ppmw as determined by the procedures specified in §61.355(c)(2) or §61.355(c)(3).
- (3) A waste stream is exempt from paragraph (c)(1) of this section provided that the owner or operator demonstrates initially and, thereafter, at least once per year that the conditions specified in either paragraph (c)(3)(i) or (c)(3)(ii) of this section are met.
- (i) The waste stream is process wastewater that has a flow rate less than 0.02 liters per minute (0.005 gallons per minute) or an annual wastewater quantity of less than 10 Mg/yr (11 ton/yr); or
- (ii) All of the following conditions are met:
- (A) The owner or operator does not choose to exempt process wastewater under paragraph (c)(3)(i) of this section.
- (B) The total annual benzene quantity in all waste streams chosen for exemption in paragraph (c)(3)(ii) of this section does not exceed 2.0 Mg/yr (2.2 ton/yr) as determined in the procedures in  $\S 61.355(j)$ , and
- (C) The total annual benzene quantity in a waste stream chosen for exemption, including process unit turnaround waste, is determined for the year in which the waste is generated.
- (d) As an alternative to the requirements specified in paragraphs (c) and (e) of this section, an owner or operator of a facility at which the total annual benzene quantity from facility waste is equal to or greater than 10 Mg/yr (11 ton/yr) as determined in paragraph (a) of this section may elect to manage and treat the facility waste as follows:
- (1) The owner or operator shall manage and treat facility waste other than process wastewater in accordance with the requirements of paragraph (c)(1) of this section.

- (2) The owner or operator shall manage and treat process wastewater in accordance with the following requirements:
- (i) Process wastewater shall be treated to achieve a total annual benzene quantity from facility process wastewater less than 1 Mg/yr (1.1 ton/yr). Total annual benzene from facility process wastewater shall be determined by adding together the annual benzene quantity at the point of waste generation for each untreated process wastewater stream plus the annual benzene quantity exiting the treatment process for each process wastewater stream treated in accordance with the requirements of paragraph (c)(1)(i) of this section.
- (ii) Each treated process wastewater stream identified in paragraph (d)(2)(i) of this section shall be managed and treated in accordance with paragraph (c)(1) of this section.
- (iii) Each untreated process wastewater stream identified in paragraph (d)(2)(i) of this section is exempt from the requirements of paragraph (c)(1) of this section.
- (e) As an alternative to the requirements specified in paragraphs (c) and (d) of this section, an owner or operator of a facility at which the total annual benzene quantity from facility waste is equal to or greater than 10 Mg/yr (11 ton/yr) as determined in paragraph (a) of this section may elect to manage and treat the facility waste as follows:
- (1) The owner or operator shall manage and treat facility waste with a flow-weighted annual average water content of less than 10 percent in accordance with the requirements of paragraph (c)(1) of this section; and
- (2) The owner or operator shall manage and treat facility waste (including remediation and process unit turnaround waste) with a flow-weighted annual average water content of 10 percent or greater, on a volume basis as total water, and each waste stream that is mixed with water or wastes at any time such that the resulting mixture has an annual water content greater than 10 percent, in accordance with the following:
- (i) The benzene quantity for the wastes described in paragraph (e)(2) of

this section must be equal to or less than 6.0 Mg/yr (6.6 ton/yr), as determined in §61.355(k). Wastes as described in paragraph (e)(2) of this section that are transferred offsite shall be included in the determination of benzene quantity as provided in §61.355(k). The provisions of paragraph (f) of this section shall not apply to any owner or operator who elects to comply with the provisions of paragraph (e) of this section.

- (ii) The determination of benzene quantity for each waste stream defined in paragraph (e)(2) of this section shall be made in accordance with §61.355(k).
- (f) Rather than treating the waste onsite, an owner or operator may elect to comply with paragraph (c)(1)(i) of this section by transferring the waste offsite to another facility where the waste is treated in accordance with the requirements of paragraph (c)(1)(i) of this section. The owner or operator transferring the waste shall:
- (1) Comply with the standards specified in §§61.343 through 61.347 of this subpart for each waste management unit that receives or manages the waste prior to shipment of the waste offsite.
- (2) Include with each offsite waste shipment a notice stating that the waste contains benzene which is required to be managed and treated in accordance with the provisions of this subpart.
- (g) Compliance with this subpart will be determined by review of facility records and results from tests and inspections using methods and procedures specified in §61.355 of this subpart.
- (h) Permission to use an alternative means of compliance to meet the requirements of §§61.342 through 61.352 of this subpart may be granted by the Administrator as provided in §61.353 of this subpart.

[55 FR 8346, Mar. 7, 1990, as amended at 58 FR 3095, Jan. 7, 1993; 65 FR 62159, 62160, Oct. 17, 2000]

#### § 61.343 Standards: Tanks.

(a) Except as provided in paragraph (b) of this section and in §61.351, the owner or operator must meet the standards in paragraph (a)(1) or (2) of this section for each tank in which the waste stream is placed in accordance

- with §61.342 (c)(1)(ii). The standards in this section apply to the treatment and storage of the waste stream in a tank, including dewatering.
- (1) The owner or operator shall install, operate, and maintain a fixed-roof and closed-vent system that routes all organic vapors vented from the tank to a control device.
- (i) The fixed-roof shall meet the following requirements:
- (A) The cover and all openings (e.g., access hatches, sampling ports, and gauge wells) shall be designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by the methods specified in §61.355(h) of this subpart.
- (B) Each opening shall be maintained in a closed, sealed position (e.g., covered by a lid that is gasketed and latched) at all times that waste is in the tank except when it is necessary to use the opening for waste sampling or removal, or for equipment inspection, maintenance, or repair.
- (C) If the cover and closed-vent system operate such that the tank is maintained at a pressure less than atmospheric pressure, then paragraph (a)(1)(i)(B) of this section does not apply to any opening that meets all of thefollowing conditions:
- (1) The purpose of the opening is to provide dilution air to reduce the explosion hazard;
- (2) The opening is designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by the methods specified in §61.355(h); and
- (3) The pressure is monitored continuously to ensure that the pressure in the tank remains below atmospheric pressure.
- (ii) The closed-vent system and control device shall be designed and operated in accordance with the requirements of §61.349 of this subpart.
- (2) The owner or operator must install, operate, and maintain an enclosure and closed-vent system that routes all organic vapors vented from the tank, located inside the enclosure, to a control device in accordance with

the requirements specified in paragraph (e) of this section.

- (b) For a tank that meets all the conditions specified in paragraph (b)(1) of this section, the owner or operator may elect to comply with paragraph (b)(2) of this section as an alternative to the requirements specified in paragraph (a)(1) of this section.
- (1) The waste managed in the tank complying with paragraph (b)(2) of this section shall meet all of the following conditions:
- (i) Each waste stream managed in the tank must have a flow-weighted annual average water content less than or equal to 10 percent water, on a volume basis as total water.
- (ii) The waste managed in the tank either:
- (A) Has a maximum organic vapor pressure less than 5.2 kilopascals (kPa) (0.75 pounds per square inch (psi));
- (B) Has a maximum organic vapor pressure less than 27.6 kPa (4.0 psi) and is managed in a tank having design capacity less than 151 m<sup>3</sup> (40,000 gal); or
- (C) Has a maximum organic vapor pressure less than 76.6 kPa (11.1 psi) and is managed in a tank having a design capacity less than 75 m<sup>3</sup> (20,000 gal).
- (2) The owner or operator shall install, operate, and maintain a fixed roof as specified in paragraph (a)(1)(i).
- (3) For each tank complying with paragraph (b) of this section, one or more devices which vent directly to the atmosphere may be used on the tank provided each device remains in a closed, sealed position during normal operations except when the device needs to open to prevent physical damage or permanent deformation of the tank or cover resulting from filling or emptying the tank, diurnal temperature changes, atmospheric pressure changes or malfunction of the unit in accordance with good engineering and safety practices for handling flammable, explosive, or other hazardous materials.
- (c) Each fixed-roof, seal, access door, and all other openings shall be checked by visual inspection initially and quarterly thereafter to ensure that no cracks or gaps occur and that access doors and other openings are closed and gasketed properly.

- (d) Except as provided in §61.350 of this subpart, when a broken seal or gasket or other problem is identified, or when detectable emissions are measured, first efforts at repair shall be made as soon as practicable, but not later than 45 calendar days after identification.
- (e) Each owner or operator who controls air pollutant emissions by using an enclosure vented through a closed-vent system to a control device must meet the requirements specified in paragraphs (e)(1) through (4) of this section.
- (1) The tank must be located inside a total enclosure. The enclosure must be designed and operated in accordance with the criteria for a permanent total enclosure as specified in "Procedure T-Criteria for and Verification of a Permanent or Temporary Total Enclosure" in 40 CFR 52.741, appendix B. The enclosure may have permanent or temporary openings to allow worker access; passage of material into or out of the enclosure by conveyor, vehicles, or other mechanical means; entry of permanent mechanical or electrical equipment: or direct airflow into the enclosure. The owner or operator must perform the verification procedure for the enclosure as specified in section 5.0 of Procedure T initially when the enclosure is first installed and, thereafter, annually. A facility that has conducted an initial compliance demonstration and that performs annual compliance demonstrations in accordance with the requirements for Tank Level 2 control requirements 40 CFR 264.1084(i) or 40 CFR 265(i) is not required to make repeat demonstrations of initial and continuous compliance for the purposes of this subpart.
- (2) The enclosure must be vented through a closed-vent system to a control device that is designed and operated in accordance with the standards for control devices specified in §61.349.
- (3) Safety devices, as defined in this subpart, may be installed and operated as necessary on any enclosure, closed-vent system, or control device used to comply with the requirements of paragraphs (e)(1) and (2) of this section.

(4) The closed-vent system must be designed and operated in accordance with the requirements of §61.349.

[55 FR 8346, Mar. 7, 1990, as amended at 55 FR 18331, May 2, 1990; 58 FR 3096, Jan. 7, 1993; 67 FR 68532, Nov. 12, 2002; 68 FR 6082, Feb. 6, 2003; 68 FR 67935, Dec. 4, 2003]

# § 61.344 Standards: Surface impoundments.

- (a) The owner or operator shall meet the following standards for each surface impoundment in which waste is placed in accordance with §61.342(c)(1)(ii) of this subpart:
- (1) The owner or operator shall install, operate, and maintain on each surface impoundment a cover (e.g., air-supported structure or rigid cover) and closed-vent system that routes all organic vapors vented from the surface impoundment to a control device.
- (i) The cover shall meet the following requirements:
- (A) The cover and all openings (e.g., access hatches, sampling ports, and gauge wells) shall be designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, initially and thereafter at least once per year by the methods specified in §61,355(h) of this subpart.
- (B) Each opening shall be maintained in a closed, sealed position (e.g., covered by a lid that is gasketed and latched) at all times that waste is in the surface impoundment except when it is necessary to use the opening for waste sampling or removal, or for equipment inspection, maintenance, or repair.
- (C) If the cover and closed-vent system operate such that the enclosure of the surface impoundment is maintained at a pressure less than atmospheric pressure, then paragraph (a)(1)(i)(B) of this section does not apply to any opening that meets all of the following conditions:
- The purpose of the opening is to provide dilution air to reduce the explosion hazard;
- (2) The opening is designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by the methods

specified in  $\S61.355(h)$  of this subpart; and

- (3) The pressure is monitored continuously to ensure that the pressure in the enclosure of the surface impoundment remains below atmospheric pressure.
- (D) The cover shall be used at all times that waste is placed in the surface impoundment except during removal of treatment residuals in accordance with 40 CFR 268.4 or closure of the surface impoundment in accordance with 40 CFR 264.228. (Note: the treatment residuals generated by these activities may be subject to the requirements of this part.)
- (ii) The closed-vent system and control device shall be designed and operated in accordance with §61.349 of this subpart.
- (b) Each cover seal, access hatch, and all other openings shall be checked by visual inspection initially and quarterly thereafter to ensure that no cracks or gaps occur and that access hatches and other openings are closed and gasketed properly.
- (c) Except as provided in §61.350 of this subpart, when a broken seal or gasket or other problem is identified, or when detectable emissions are measured, first efforts at repair shall be made as soon as practicable, but not later than 15 calendar days after identification.

[55 FR 8346, Mar. 7, 1990, as amended at 58 FR 3097, Jan. 7, 1993]

#### §61.345 Standards: Containers.

- (a) The owner or operator shall meet the following standards for each container in which waste is placed in accordance with §61.342(c)(1)(ii) of this subpart:
- (1) The owner or operator shall install, operate, and maintain a cover on each container used to handle, transfer, or store waste in accordance with the following requirements:
- (i) The cover and all openings (e.g., bungs, hatches, and sampling ports) shall be designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, initially and thereafter at least once per year by the methods specified in §61.355(h) of this subpart.

- (ii) Except as provided in paragraph (a)(4) of this section, each opening shall be maintained in a closed, sealed position (e.g., covered by a lid that is gasketed and latched) at all times that waste is in the container except when it is necessary to use the opening for waste loading, removal, inspection, or sampling.
- (2) When a waste is transferred into a container by pumping, the owner or operator shall perform the transfer using a submerged fill pipe. The submerged fill pipe outlet shall extend to within two fill pipe diameters of the bottom of the container while the container is being loaded. During loading of the waste, the cover shall remain in place and all openings shall be maintained in a closed, sealed position except for those openings required for the submerged fill pipe, those openings required for venting of the container to prevent physical damage or permanent deformation of the container or cover, and any openings complying with paragraph (a)(4) of this section.
- (3) Treatment of a waste in a container, including aeration, thermal or other treatment, must be performed by the owner or operator in a manner such that while the waste is being treated the container meets the standards specified in paragraphs (a)(3)(i) through (iii) of this section, except for covers and closed-vent systems that meet the requirements in paragraph (a)(4) of this section.
- (i) The owner or operator must either:
- (A) Vent the container inside a total enclosure which is exhausted through a closed-vent system to a control device in accordance with the requirements of paragraphs (a)(3)(ii)(A) and (B) of this section: or
- (B) Vent the covered or closed container directly through a closed-vent system to a control device in accordance with the requirements of paragraphs (a)(3)(ii)(B) and (C) of this section.
- (ii) The owner or operator must meet the following requirements, as applicable to the type of air emission control equipment selected by the owner or operator:
- (A) The total enclosure must be designed and operated in accordance with

- the criteria for a permanent total enclosure as specified in section 5 of the "Procedure T—Criteria for Verification of a Permanent or Temporary Total Enclosure" in 40 CFR 52.741, appendix B. The enclosure may have permanent or temporary openings to allow worker access; passage of containers through the enclosure by conveyor or other mechanical means; entry of permanent mechanical or electrical equipment; or direct airflow into the enclosure. The owner or operator must perform the verification procedure for the enclosure as specified in section 5.0 of "Procedure T-Criteria for and Verification of a Permanent or Temporary Total Enclosure" initially when the enclosure is first installed and, thereafter, annually. A facility that has conducted an initial compliance demonstration and that performs annual compliance demonstrations in accordance with the Container Level 3 control requirements in 40 CFR 40 264.1086(e)(2)(i) orCFR 265.1086(e)(2)(i) is not required to make repeat demonstrations of initial and continuous compliance for the purposes of this subpart.
- (B) The closed-vent system and control device must be designed and operated in accordance with the requirements of \$61.349.
- (C) For a container cover, the cover and all openings (e.g., doors, hatches) must be designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, initially and thereafter at least once per year by the methods specified in §61.355(h).
- (iii) Safety devices, as defined in this subpart, may be installed and operated as necessary on any container, enclosure, closed-vent system, or control device used to comply with the requirements of paragraph (a)(3)(i) of this section
- (4) If the cover and closed-vent system operate such that the container is maintained at a pressure less than atmospheric pressure, the owner or operator may operate the system with an opening that is not sealed and kept closed at all times if the following conditions are met:

- (i) The purpose of the opening is to provide dilution air to reduce the explosion hazard:
- (ii) The opening is designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by methods specified in §61.355(h); and
- (iii) The pressure is monitored continuously to ensure that the pressure in the container remains below atmospheric pressure.
- (b) Each cover and all openings shall be visually inspected initially and quarterly thereafter to ensure that they are closed and gasketed properly.
- (c) Except as provided in §61.350 of this subpart, when a broken seal or gasket or other problem is identified, first efforts at repair shall be made as soon as practicable, but not later than 15 calendar days after identification.

[55 FR 8346, Mar. 7, 1990, as amended at 58 FR 3097, Jan. 7, 1993; 67 FR 68532, Nov. 12, 2002; 68 FR 67936, Dec. 4, 2003]

# §61.346 Standards: Individual drain systems.

- (a) Except as provided in paragraph (b) of this section, the owner or operator shall meet the following standards for each individual drain system in which waste is placed in accordance with \$61.342(c)(1)(ii) of this subpart:
- (1) The owner or operator shall install, operate, and maintain on each drain system opening a cover and closed-vent system that routes all organic vapors vented from the drain system to a control device.
- (i) The cover shall meet the following requirements:
- (A) The cover and all openings (e.g., access hatches, sampling ports) shall be designed to operate with no detactable emissions as indicated by an instrument reading of less than 500 ppmv above background, initially and thereafter at least once per year by the methods specified in §61.355(h) of this subpart.
- (B) Each opening shall be maintained in a closed, sealed position (e.g., covered by a lid that is gasketed and latched) at all times that waste is in the drain system except when it is necessary to use the opening for waste

sampling or removal, or for equipment inspection, maintenance, or repair.

- (C) If the cover and closed-vent system operate such that the individual drain system is maintained at a pressure less than atmospheric pressure, then paragraph (a)(1)(i)(B) of this section does not apply to any opening that meets all of the following conditions:
- (1) The purpose of the opening is to provide dilution air to reduce the explosion hazard;
- (2) The opening is designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by the methods specified in §61.355(h); and
- (3) The pressure is monitored continuously to ensure that the pressure in the individual drain system remains below atmospheric pressure.
- (ii) The closed-vent system and control device shall be designed and operated in accordance with §61.349 of this subpart.
- (2) Each cover seal, access hatch, and all other openings shall be checked by visual inspection initially and quarterly thereafter to ensure that no cracks or gaps occur and that access hatches and other openings are closed and gasketed properly.
- (3) Except as provided in §61.350 of this subpart, when a broken seal or gasket or other problem is identified, or when detectable emissions are measured, first efforts at repair shall be made as soon as practicable, but not later than 15 calendar days after identification.
- (b) As an alternative to complying with paragraph (a) of this section, an owner or operator may elect to comply with the following requirements:
- (1) Each drain shall be equipped with water seal controls or a tightly sealed cap or plug.
- (2) Each junction box shall be equipped with a cover and may have a vent pipe. The vent pipe shall be at least 90 cm (3 ft) in length and shall not exceed 10.2 cm (4 in) in diameter.
- (i) Junction box covers shall have a tight seal around the edge and shall be kept in place at all times, except during inspection and maintenance.

- (ii) One of the following methods shall be used to control emissions from the junction box vent pipe to the atmosphere:
- (A) Equip the junction box with a system to prevent the flow of organic vapors from the junction box vent pipe to the atmosphere during normal operation. An example of such a system includes use of water seal controls on the junction box. A flow indicator shall be installed, operated, and maintained on each junction box vent pipe to ensure that organic vapors are not vented from the junction box to the atmosphere during normal operation.
- (B) Connect the junction box vent pipe to a closed-vent system and control device in accordance with §61.349 of this subpart.
- (3) Each sewer line shall not be open to the atmosphere and shall be covered or enclosed in a manner so as to have no visual gaps or cracks in joints, seals, or other emission interfaces.
- (4) Equipment installed in accordance with paragraphs (b)(1), (b)(2), or (b)(3) of this section shall be inspected as follows:
- (i) Each drain using water seal controls shall be checked by visual or physical inspection initially and thereafter quarterly for indications of low water levels or other conditions that would reduce the effectiveness of water seal controls.
- (ii) Each drain using a tightly sealed cap or plug shall be visually inspected initially and thereafter quarterly to ensure caps or plugs are in place and properly installed.
- (iii) Each junction box shall be visually inspected initially and thereafter quarterly to ensure that the cover is in place and to ensure that the cover has a tight seal around the edge.
- (iv) The unburied portion of each sewer line shall be visually inspected initially and thereafter quarterly for indication of cracks, gaps, or other problems that could result in benzene emissions.
- (5) Except as provided in §61.350 of this subpart, when a broken seal, gap, crack or other problem is identified, first efforts at repair shall be made as

soon as practicable, but not later than 15 calendar days after identification.

[55 FR 8346, Mar. 7, 1990, as amended at 55 FR 37231, Sept. 10, 1990; 58 FR 3097, Jan. 7, 1993]

# § 61.347 Standards: Oil-water separators.

- (a) Except as provided in §61.352 of this subpart, the owner or operator shall meet the following standards for each oil-water separator in which waste is placed in accordance with §61.342(c)(1)(ii) of this subpart:
- (1) The owner or operator shall install, operate, and maintain a fixed-roof and closed-vent system that routes all organic vapors vented from the oil-water separator to a control device.
- (i) The fixed-roof shall meet the following requirements:
- (A) The cover and all openings (e.g., access hatches, sampling ports, and gauge wells) shall be designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by the methods specified in §61.355(h) of this subpart.
- (B) Each opening shall be maintained in a closed, sealed position (e.g., covered by a lid that is gasketed and latched) at all times that waste is in the oil-water separator except when it is necessary to use the opening for waste sampling or removal, or for equipment inspection, maintenance, or repair.
- (C) If the cover and closed-vent system operate such that the oil-water separator is maintained at a pressure less than atmospheric pressure, then paragraph (a)(1)(i)(B) of this section does not apply to any opening that meets all of the following conditions:
- (1) The purpose of the opening is to provide dilution air to reduce the explosion hazard;
- (2) The opening is designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by the methods specified in §61.355(h); and
- (3) The pressure is monitored continuously to ensure that the pressure

in the oil-water separator remains below atmospheric pressure.

- (ii) The closed-vent system and control device shall be designed and operated in accordance with the requirements of §61.349 of this subpart.
- (b) Each cover seal, access hatch, and all other openings shall be checked by visual inspection initially and quarterly thereafter to ensure that no cracks or gaps occur between the cover and oil-water separator wall and that access hatches and other openings are closed and gasketed properly.
- (c) Except as provided in §61.350 of this subpart, when a broken seal or gasket or other problem is identified, or when detectable emissions are measured, first efforts at repair shall be made as soon as practicable, but not later than 15 calendar days after identification.

[55 FR 8346, Mar. 7, 1990, as amended at 58 FR 3098, Jan. 7, 1993]

# § 61.348 Standards: Treatment processes.

- (a) Except as provided in paragraph (a)(5) of this section, the owner or operator shall treat the waste stream in accordance with the following requirements:
- (1) The owner or operator shall design, install, operate, and maintain a treatment process that either:
- (i) Removes benzene from the waste stream to a level less than 10 parts per million by weight (ppmw) on a flowweighted annual average basis,
- (ii) Removes benzene from the waste stream by 99 percent or more on a mass basis, or
- (iii) Destroys benzene in the waste stream by incinerating the waste in a combustion unit that achieves a destruction efficiency of 99 percent or greater for benzene.
- (2) Each treatment process complying with paragraphs (a)(1)(i) or (a)(1)(ii) of this section shall be designed and operated in accordance with the appropriate waste management unit standards specified in §§61.343 through 61.347 of this subpart. For example, if a treatment process is a tank, then the owner or operator shall comply with §61.343 of this subpart.
- (3) For the purpose of complying with the requirements specified in para-

graph (a)(1)(i) of this section, the intentional or unintentional reduction in the benzene concentration of a waste stream by dilution of the waste stream with other wastes or materials is not allowed.

- (4) An owner or operator may aggregate or mix together individual waste streams to create a combined waste stream for the purpose of facilitating treatment of waste to comply with the requirements of paragraph (a)(1) of this section except as provided in paragraph (a)(5) of this section.
- (5) If an owner or operator aggregates or mixes any combination of process wastewater, product tank drawdown, or landfill leachate subject to §61.342(c)(1) of this subpart together with other waste streams to create a combined waste stream for the purpose of facilitating management or treatment of waste in a wastewater treatment system, then the wastewater treatment system shall be operated in accordance with paragraph (b) of this section. These provisions apply to above-ground wastewater treatment systems as well as those that are at or below ground level.
- (b) Except for facilities complying with §61.342(e), the owner or operator that aggregates or mixes individual waste streams as defined in paragraph (a)(5) of this section for management and treatment in a wastewater treatment system shall comply with the following requirements:
- (1) The owner or operator shall design and operate each waste management unit that comprises the wastewater treatment system in accordance with the appropriate standards specified in §§ 61.343 through 61.347 of this subpart.
- (2) The provisions of paragraph (b)(1) of this section do not apply to any waste management unit that the owner or operator demonstrates to meet the following conditions initially and, thereafter, at least once per year:
- (i) The benzene content of each waste stream entering the waste management unit is less than 10 ppmw on a flow-weighted annual average basis as determined by the procedures specified in §61.355(c) of this subpart; and
- (ii) The total annual benzene quantity contained in all waste streams

- managed or treated in exempt waste management units comprising the facility wastewater treatment systems is less than 1 Mg/yr (1.1 ton/yr). For this determination, total annual benzene quantity shall be calculated as follows:
- (A) The total annual benzene quantity shall be calculated as the sum of the individual benzene quantities determined at each location where a waste stream first enters an exempt waste management unit. The benzene quantity discharged from an exempt waste management unit shall not be included in this calculation.
- (B) The annual benzene quantity in a waste stream managed or treated in an enhanced biodegradation unit shall not be included in the calculation of the total annual benzene quantity, if the enhanced biodegradation unit is the first exempt unit in which the waste is managed or treated. A unit shall be considered enhanced biodegradation if it is a suspended-growth process that generates biomass, uses recycled biomass, and periodically removes biomass from the process. An enhanced biodegradation unit typically operates at a food-to-microorganism ratio in the range of 0.05 to 1.0 kg of biological oxygen demand per kg of biomass per day, a mixed liquor suspended solids ratio in the range of 1 to 8 grams per liter (0.008 to 0.7 pounds per liter), and a residence time in the range of 3 to 36 hours.
- (c) The owner and operator shall demonstrate that each treatment process or wastewater treatment system unit, except as provided in paragraph (d) of this section, achieves the appropriate conditions specified in paragraphs (a) or (b) of this section in accordance with the following requirements:
- (1) Engineering calculations in accordance with requirements specified in §61.356(e) of this subpart; or
- (2) Performance tests conducted using the test methods and procedures that meet the requirements specified in §61.355 of this subpart.
- (d) A treatment process or waste stream is in compliance with the requirements of this subpart and exempt from the requirements of paragraph (c) of this section provided that the owner or operator documents that the treatment process or waste stream is in

- compliance with other regulatory requirements as follows:
- (1) The treatment process is a hazardous waste incinerator for which the owner or operator has been issued a final permit under 40 CFR part 270 and complies with the requirements of 40 CFR part 264, subpart O:
- (2) The treatment process is an industrial furnace or boiler burning hazardous waste for energy recovery for which the owner or operator has been issued a final permit under 40 CFR part 270 and complies with the requirements of 40 CFR part 266, subpart D;
- (3) The waste stream is treated by a means or to a level that meets benzene-specific treatment standards in accordance with the Land Disposal Restrictions under 40 CFR part 268, and the treatment process is designed and operated with a closed-vent system and control device meeting the requirements of §61.349 of this subpart;
- (4) The waste stream is treated by a means or to a level that meets benzene-specific effluent limitations or performance standards in accordance with the Effluent Guidelines and Standards under 40 CFR parts 401–464, and the treatment process is designed and operated with a closed-vent system and control device meeting the requirements of §61.349 of this subpart; or
- (5) The waste stream is discharged to an underground injection well for which the owner or operator has been issued a final permit under 40 CFR part 270 and complies with the requirements of 40 CFR part 122.
- (e) Except as specified in paragraph (e)(3) of this section, if the treatment process or wastewater treatment system unit has any openings (e.g., access doors, hatches, etc.), all such openings shall be sealed (e.g., gasketed, latched, etc.) and kept closed at all times when waste is being treated, except during inspection and maintenance.
- (1) Each seal, access door, and all other openings shall be checked by visual inspections initially and quarterly thereafter to ensure that no cracks or gaps occur and that openings are closed and gasketed properly.
- (2) Except as provided in §61.350 of this subpart, when a broken seal or gasket or other problem is identified, first efforts at repair shall be made as

soon as practicable, but not later than 15 calendar days after identification.

- (3) If the cover and closed-vent system operate such that the treatment process and wastewater treatment system unit are maintained at a pressure less than atmospheric pressure, the owner or operator may operate the system with an opening that is not sealed and kept closed at all times if the following conditions are met:
- (i) The purpose of the opening is to provide dilution air to reduce the explosion hazard:
- (ii) The opening is designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by the methods specified in §61.355(h); and
- (iii) The pressure is monitored continuously to ensure that the pressure in the treatment process and wastewater treatment system unit remain below atmospheric pressure.
- (f) Except for treatment processes complying with paragraph (d) of this section, the Administrator may request at any time an owner or operator demonstrate that a treatment process or wastewater treatment system unit meets the applicable requirements specified in paragraphs (a) or (b) of this section by conducting a performance test using the test methods and procedures as required in §61.355 of this subnart.
- (g) The owner or operator of a treatment process or wastewater treatment system unit that is used to comply with the provisions of this section shall monitor the unit in accordance with the applicable requirements in §61.354 of this subpart.

[55 FR 8346, Mar. 7, 1990, as amended at 55 FR 37231, Sept. 10, 1990; 58 FR 3098, Jan. 7, 1993; 65 FR 62160, Oct. 17, 2000]

### § 61.349 Standards: Closedvent systems and control devices.

(a) For each closed-vent system and control device used to comply with standards in accordance with §§61.343 through 61.348 of this subpart, the owner or operator shall properly design, install, operate, and maintain the closed-vent system and control device

in accordance with the following requirements:

- (1) The closed-vent system shall:
- (i) Be designed to operate with no detectable emissions as indicated by an instrument reading of less than 500 ppmv above background, as determined initially and thereafter at least once per year by the methods specified in §61.355(h) of this subpart.
- (ii) Vent systems that contain any bypass line that could divert the vent stream away from a control device used to comply with the provisions of this subpart shall install, maintain, and operate according to the manufacturer's specifications a flow indicator that provides a record of vent stream flow away from the control device at least once every 15 minutes, except as provided in paragraph (a)(1)(ii)(B) of this section.
- (A) The flow indicator shall be installed at the entrance to any bypass line that could divert the vent stream away from the control device to the atmosphere.
- (B) Where the bypass line valve is secured in the closed position with a carseal or a lock-and-key type configuration, a flow indicator is not required.
- (iii) All gauging and sampling devices shall be gas-tight except when gauging or sampling is taking place.
- (iv) For each closed-vent system complying with paragraph (a) of this section, one or more devices which vent directly to the atmosphere may be used on the closed-vent system provided each device remains in a closed, sealed position during normal operations except when the device needs to open to prevent physical damage or permanent deformation of the closed-vent system resulting from malfunction of the unit in accordance with good engineering and safety practices for handling flammable, explosive, or other hazardous materials.
- (2) The control device shall be designed and operated in accordance with the following conditions:
- (i) An enclosed combustion device (e.g., a vapor incinerator, boiler, or process heater) shall meet one of the following conditions:
- (A) Reduce the organic emissions vented to it by 95 weight percent or greater;

- (B) Achieve a total organic compound concentration of 20 ppmv (as the sum of the concentrations for individual compounds using Method 18) on a dry basis corrected to 3 percent oxygen; or
- (C) Provide a minimum residence time of 0.5 seconds at a minimum temperature of 760 °C (1,400 °F). If a boiler or process heater issued as the control device, then the vent stream shall be introduced into the flame zone of the boiler or process heater.
- (ii) A vapor recovery system (e.g., a carbon adsorption system or a condenser) shall recover or control the organic emissions vented to it with an efficiency of 95 weight percent or greater, or shall recover or control the benzene emissions vented to it with an efficiency of 98 weight percent or greater
- (iii) A flare shall comply with the requirements of 40 CFR 60.18.
- (iv) A control device other than those described in paragraphs (a)(2) (i) through (iii) of this section may be used provided that the following conditions are met:
- (A) The device shall recover or control the organic emissions vented to it with an efficiency of 95 weight percent or greater, or shall recover or control the benzene emissions vented to it with an efficiency of 98 weight percent or greater.
- (B) The owner or operator shall develop test data and design information that documents the control device will achieve an emission control efficiency of either 95 percent or greater for organic compounds or 98 percent or greater for benzene.
- (C) The owner or operator shall identify:
- (I) The critical operating parameters that affect the emission control performance of the device:
- (2) The range of values of these operating parameters that ensure the emission control efficiency specified in paragraph (a)(2)(iv)(A) of this section is maintained during operation of the device; and
- (3) How these operating parameters will be monitored to ensure the proper operation and maintenance of the device
- (D) The owner or operator shall submit the information and data specified

- in paragraphs (a)(2)(iv) (B) and (C) of this section to the Administrator prior to operation of the alternative control device.
- (E) The Administrator will determine, based on the information submitted under paragraph (a)(2)(iv)(D) of this section, if the control device subject to paragraph (a)(2)(iv) of this section meets the requirements of §61.349. The control device subject to paragraph (a)(2)(iv) of this section may be operated prior to receiving approval from the Administrator. However, if the Administrator determines that the control device does not meet the requirements of §61.349, the facility may be subject to enforcement action beginning from the time the control device began operation.
- (b) Each closed-vent system and control device used to comply with this subpart shall be operated at all times when waste is placed in the waste management unit vented to the control device except when maintenance or repair of the waste management unit cannot be completed without a shutdown of the control device.
- (c) An owner and operator shall demonstrate that each control device, except for a flare, achieves the appropriate conditions specified in paragraph (a)(2) of this section by using one of the following methods:
- (1) Engineering calculations in accordance with requirements specified in §61.356(f) of this subpart; or
- (2) Performance tests conducted using the test methods and procedures that meet the requirements specified in §61.355 of this subpart.
- (d) An owner or operator shall demonstrate compliance of each flare in accordance with paragraph (a)(2)(iii) of this section.
- (e) The Administrator may request at any time an owner or operator demonstrate that a control device meets the applicable conditions specified in paragraph (a)(2) of this section by conducting a performance test using the test methods and procedures as required in §61.355, and for control devices subject to paragraph (a)(2)(iv) of this section, the Administrator may specify alternative test methods and procedures, as appropriate.

# **Environmental Protection Agency**

- (f) Each closed-vent system and control device shall be visually inspected initially and quarterly thereafter. The visual inspection shall include inspection of ductwork and piping and connections to covers and control devices for evidence of visable defects such as holes in ductwork or piping and loose connections.
- (g) Except as provided in §61.350 of this subpart, if visible defects are observed during an inspection, or if other problems are identified, or if detectable emissions are measured, a first effort to repair the closed-vent system and control device shall be made as soon as practicable but no later than 5 calendar days after detection. Repair shall be completed no later than 15 calendar days after the emissions are detected or the visible defect is observed.
- (h) The owner or operator of a control device that is used to comply with the provisions of this section shall monitor the control device in accordance with §61.354(c) of this subpart.

[55 FR 8346, Mar. 7, 1990; 55 FR 12444, Apr. 3, 1990, as amended at 55 FR 37231, Sept. 10, 1990; 58 FR 3098, Jan. 7, 1993; 65 FR 62160, Oct. 17, 2000]

### § 61.350 Standards: Delay of repair.

- (a) Delay of repair of facilities or units that are subject to the provisions of this subpart will be allowed if the repair is technically impossible without a complete or partial facility or unit shutdown.
- (b) Repair of such equipment shall occur before the end of the next facility or unit shutdown.

# § 61.351 Alternative standards for tanks.

- (a) As an alternative to the standards for tanks specified in §61.343 of this subpart, an owner or operator may elect to comply with one of the following:
- (1) A fixed roof and internal floating roof meeting the requirements in 40 CFR 60.112b(a)(1);
- (2) An external floating roof meeting the requirements of 40 CFR 60.112b (a)(2); or
- (3) An alternative means of emission limitation as described in 40 CFR 60.114b.

(b) If an owner or operator elects to comply with the provisions of this section, then the owner or operator is exempt from the provisions of §61.343 of this subpart applicable to the same facilities.

[55 FR 8346, Mar. 7, 1990, as amended at 55 FR 37231, Sept. 10, 1990]

# § 61.352 Alternative standards for oilwater separators.

- (a) As an alternative to the standards for oil-water separators specified in §61.347 of this subpart, an owner or operator may elect to comply with one of the following:
- (1) A floating roof meeting the requirements in 40 CFR 60.693-2(a); or
- (2) An alternative means of emission limitation as described in 40 CFR 60.694.
- (b) For portions of the oil-water separator where it is infeasible to construct and operate a floating roof, such as over the weir mechanism, a fixed roof vented to a vapor control device that meets the requirements in §§61.347 and 61.349 of this subpart shall be installed and operated.
- (c) Except as provided in paragraph (b) of this section, if an owner or operator elects to comply with the provisions of this section, then the owner or operator is exempt from the provisions in §61.347 of this subpart applicable to the same facilities.

# § 61.353 Alternative means of emission limitation.

- (a) If, in the Administrator's judgment, an alternative means of emission limitation will achieve a reduction in benzene emissions at least equivalent to the reduction in benzene emissions from the source achieved by the applicable design, equipment, work practice, or operational requirements in §§ 61.342 through 61.349, the Administrator will publish in the FEDERAL REGISTER a notice permitting the use of the alternative means for purposes of compliance with that requirement. The notice may condition the permission on requirements related to the operation and maintenance of the alternative means.
- (b) Any notice under paragraph (a) of this section shall be published only

after public notice and an opportunity for a hearing.

(c) Any person seeking permission under this section shall collect, verify, and submit to the Administrator information showing that the alternative means achieves equivalent emission reductions.

[55 FR 8346, Mar. 7, 1990, as amended at 58 FR 3099, Jan. 7, 1993]

#### § 61.354 Monitoring of operations.

- (a) Except for a treatment process or waste stream complying with §61.348(d), the owner or operator shall monitor each treatment process or wastewater treatment system unit to ensure the unit is properly operated and maintained by one of the following monitoring procedures:
- (1) Measure the benzene concentration of the waste stream exiting the treatment process complying with §61.348(a)(1)(i) at least once per month by collecting and analyzing one or more samples using the procedures specified in §61.355(c)(3).
- (2) Install, calibrate, operate, and maintain according to manufacturer's specifications equipment to continuously monitor and record a process parameter (or parameters) for the treatment process or wastewater treatment system unit that indicates proper system operation. The owner or operator shall inspect at least once each operating day the data recorded by the monitoring equipment (e.g., temperature monitor or flow indicator) to ensure that the unit is operating properly.
- (b) If an owner or operator complies with the requirements of §61.348(b), then the owner or operator shall monitor each wastewater treatment system to ensure the unit is properly operated and maintained by the appropriate monitoring procedure as follows:
- (1) For the first exempt waste management unit in each waste treatment train, other than an enhanced biodegradation unit, measure the flow rate, using the procedures of §61.355(b), and the benzene concentration of each waste stream entering the unit at least once per month by collecting and analyzing one or more samples using the procedures specified in §61.355(c)(3).

- (2) For each enhanced biodegradation unit that is the first exempt waste management unit in a treatment train, measure the benzene concentration of each waste stream entering the unit at least once per month by collecting and analyzing one or more samples using the procedures specified in §61.355(c)(3).
- (c) An owner or operator subject to the requirements in §61.349 of this subpart shall install, calibrate, maintain, and operate according to the manufacturer's specifications a device to continuously monitor the control device operation as specified in the following paragraphs, unless alternative monitoring procedures or requirements are approved for that facility by the Administrator. The owner or operator shall inspect at least once each operating day the data recorded by the monitoring equipment (e.g., temperature monitor or flow indicator) to ensure that the control device is operating properly.
- (1) For a thermal vapor incinerator, a temperature monitoring device equipped with a continuous recorder. The device shall have an accuracy of  $\pm 1$  percent of the temperature being monitored in °C or  $\pm 0.5$  °C, whichever is greater. The temperature sensor shall be installed at a representative location in the combustion chamber.
- (2) For a catalytic vapor incinerator, a temperature monitoring device equipped with a continuous recorder. The device shall be capable of monitoring temperature at two locations, and have an accuracy of  $\pm 1$  percent of the temperature being monitored in °C or  $\pm 0.5$  °C, whichever is greater. One temperature sensor shall be installed in the vent stream at the nearest feasible point to the catalyst bed inlet and a second temperature sensor shall be installed in the vent stream at the nearest feasible point to the catalyst bed outlet.
- (3) For a flare, a monitoring device in accordance with 40 CFR 60.18(f)(2) equipped with a continuous recorder.
- (4) For a boiler or process heater having a design heat input capacity less than 44 MW (150  $\times$  106 BTU/hr), a temperature monitoring device equipped with a continuous recorder. The device shall have an accuracy of  $\pm 1$  percent of the temperature being monitored in  $^{\circ}\mathrm{C}$

- or ±0.5 °C, whichever is greater. The temperature sensor shall be installed at a representative location in the combustion chamber.
- (5) For a boiler or process heater having a design heat input capacity greater than or equal to 44 MW (150 × 10<sup>6</sup> BTU/hr), a monitoring device equipped with a continuous recorder to measure a parameter(s) that indicates good combustion operating practices are being used.
  - (6) For a condenser, either:
- (i) A monitoring device equipped with a continuous recorder to measure either the concentration level of the organic compounds or the concentration level of benzene in the exhaust vent stream from the condenser; or
- (ii) A temperature monitoring device equipped with a continuous recorder. The device shall be capable of monitoring temperature at two locations, and have an accuracy of  $\pm 1$  percent of the temperature being monitored in °C or  $\pm 0.5$  °C, whichever is greater. One temperature sensor shall be installed at a location in the exhaust stream from the condenser, and a second temperature sensor shall be installed at a location in the coolant fluid exiting the condenser.
- (7) For a carbon adsorption system that regenerates the carbon bed directly in the control device such as a fixed-bed carbon adsorber, either:
- (i) A monitoring device equipped with a continuous recorder to measure either the concentration level of the organic compounds or the benzene concentration level in the exhaust vent stream from the carbon bed; or
- (ii) A monitoring device equipped with a continuous recorder to measure a parameter that indicates the carbon bed is regenerated on a regular, predetermined time cycle.
- (8) For a vapor recovery system other than a condenser or carbon adsorption system, a monitoring device equipped with a continuous recorder to measure either the concentration level of the organic compounds or the benzene concentration level in the exhaust vent stream from the control device.
- (9) For a control device subject to the requirements of §61.349(a)(2)(iv), devices to monitor the parameters as specified in §61.349(a)(2)(iv)(C).

- (d) For a carbon adsorption system that does not regenerate the carbon bed directly on site in the control device (e.g., a carbon canister), either the concentration level of the organic compounds or the concentration level of benzene in the exhaust vent stream from the carbon adsorption system shall be monitored on a regular schedule, and the existing carbon shall be replaced with fresh carbon immediately when carbon breakthrough is indicated. The device shall be monitored on a daily basis or at intervals no greater than 20 percent of the design carbon replacement interval, whichever is greater. As an alternative to conducting this monitoring, an owner or operator may replace the carbon in the carbon adsorption system with fresh carbon at a regular predetermined time interval that is less than the carbon replacement interval that is determined by the maximum design flow rate and either the organic concentration or the benzene concentration in the gas stream vented to the carbon adsorption system.
- (e) An alternative operation or process parameter may be monitored if it can be demonstrated that another parameter will ensure that the control device is operated in conformance with these standards and the control device's design specifications.
- (f) Owners or operators using a closed-vent system that contains any bypass line that could divert a vent stream from a control device used to comply with the provisions of this subpart shall do the following:
- (1) Visually inspect the bypass line valve at least once every month, checking the position of the valve and the condition of the car-seal or closure mechanism required under \$61.349(a)(1)(ii) to ensure that the valve is maintained in the closed position and the vent stream is not diverted through the bypass line.
- (2) Visually inspect the readings from each flow monitoring device required by §61.349(a)(1)(ii) at least once each operating day to check that vapors are being routed to the control device as required.
- (g) Each owner or operator who uses a system for emission control that is

maintained at a pressure less than atmospheric pressure with openings to provide dilution air shall install, calibrate, maintain, and operate according to the manufacturer's specifications a device equipped with a continuous recorder to monitor the pressure in the unit to ensure that it is less than atmospheric pressure.

[55 FR 8346, Mar. 7, 1990, as amended at 58 FR 3099, Jan. 7, 1993; 65 FR 62160, Oct. 17, 2000]

# § 61.355 Test methods, procedures, and compliance provisions.

- (a) An owner or operator shall determine the total annual benzene quantity from facility waste by the following procedure:
- (1) For each waste stream subject to this subpart having a flow-weighted annual average water content greater than 10 percent water, or a volume basis as total water, or is mixed with water or other wastes at any time and the resulting mixture has an annual average water content greater than 10 percent as specified in §61.342(a), the owner or operator shall:
- (i) Determine the annual waste quantity for each waste stream using the procedures specified in paragraph (b) of this section.
- (ii) Determine the flow-weighted annual average benzene concentration for each waste stream using the procedures specified in paragraph (c) of this section.
- (iii) Calculate the annual benzene quantity for each waste stream by multiplying the annual waste quantity of the waste stream times the flow-weighted annual average benzene concentration.
- (2) Total annual benzene quantity from facility waste is calculated by adding together the annual benzene quantity for each waste stream generated during the year and the annual benzene quantity for each process unit turnaround waste annualized according to paragraph (b)(4) of this section.
- (3) If the total annual benzene quantity from facility waste is equal to or greater than 10 Mg/yr (11 ton/yr), then the owner or operator shall comply with the requirements of §61.342 (c), (d), or (e).
- (4) If the total annual benzene quantity from facility waste is less than 10

Mg/yr (11 ton/yr) but is equal to or greater than 1 Mg/yr (1.1 ton/yr), then the owner or operator shall:

- (i) Comply with the recordkeeping requirements of §61.356 and reporting requirements of §61.357 of this subpart; and
- (ii) Repeat the determination of total annual benzene quantity from facility waste at least once per year and whenever there is a change in the process generating the waste that could cause the total annual benzene quantity from facility waste to increase to 10 Mg/yr (11 ton/vr) or more.
- (5) If the total annual benzene quantity from facility waste is less than 1 Mg/yr (1.1 ton/yr), then the owner or operator shall:
- (i) Comply with the recordkeeping requirements of §61.356 and reporting requirements of §61.357 of this subpart; and
- (ii) Repeat the determination of total annual benzene quantity from facility waste whenever there is a change in the process generating the waste that could cause the total annual benzene quantity from facility waste to increase to 1 Mg/yr (1.1 ton/yr) or more.
- (6) The benzene quantity in a waste stream that is generated less than one time per year, except as provided for process unit turnaround waste in paragraph (b)(4) of this section, shall be included in the determination of total annual benzene quantity from facility waste for the year in which the waste is generated unless the waste stream is otherwise excluded from the determination of total annual benzene quantity from facility waste in accordance with paragraphs (a) through (c) of this section. The benzene quantity in this waste stream shall not be annualized or averaged over the time interval between the activities that resulted in generation of the waste, for purposes of determining the total annual benzene quantity from facility waste.
- (b) For purposes of the calculation required by paragraph (a) of this section, an owner or operator shall determine the annual waste quantity at the point of waste generation, unless otherwise provided in paragraphs (b) (1), (2), (3), and (4) of this section, by one of the methods given in paragraphs (b) (5) through (7) of this section.

- (1) The determination of annual waste quantity for sour water streams that are processed in sour water strippers shall be made at the point that the water exits the sour water stripper.
- (2) The determination of annual waste quantity for wastes at coke byproduct plants subject to and complying with the control requirements of §61.132, 61.133, 61.134, or 61.139 of subpart L of this part shall be made at the location that the waste stream exits the process unit component or waste management unit controlled by that subpart or at the exit of the ammonia still, provided that the following conditions are met:
- (i) The transfer of wastes between units complying with the control requirements of subpart L of this part, process units, and the ammonia still is made through hard piping or other enclosed system.
- (ii) The ammonia still meets the definition of a sour water stripper in §61.341.
- (3) The determination of annual waste quantity for wastes that are received at hazardous waste treatment, storage, or disposal facilities from offsite shall be made at the point where the waste enters the hazardous waste treatment, storage, or disposal facility.
- (4) The determination of annual waste quantity for each process unit turnaround waste generated only at 2 year or greater intervals, may be made by dividing the total quantity of waste generated during the most recent process unit turnaround by the time period (in the nearest tenth of a year) between the turnaround resulting in generation of the waste and the most recent preceding process turnaround for the unit. The resulting annual waste quantity shall be included in the calculation of the annual benzene quantity as provided in paragraph (a)(1)(iii) of this section for the year in which the turnaround occurs and for each subsequent year until the unit undergoes the next process turnaround. For estimates of total annual benzene quantity as specified in the 90-day report, required under §61.357(a)(1), the owner or operator shall estimate the waste quantity generated during the most recent turnaround, and the time period between turnarounds in accordance with good

- engineering practices. If the owner or operator chooses not to annualize process unit turnaround waste, as specified in this paragraph, then the process unit turnaround waste quantity shall be included in the calculation of the annual benzene quantity for the year in which the turnaround occurs.
- (5) Select the highest annual quantity of waste managed from historical records representing the most recent 5 years of operation or, if the facility has been in service for less than 5 years but at least 1 year, from historical records representing the total operating life of the facility;
- (6) Use the maximum design capacity of the waste management unit; or
- (7) Use measurements that are representative of maximum waste generation rates.
- (c) For the purposes of the calculation required by §§61.355(a) of this subpart, an owner or operator shall determine the flow-weighted annual average benzene concentration in a manner that meets the requirements given in paragraph (c)(1) of this section using either of the methods given in paragraphs (c)(2) and (c)(3) of this section.
- (1) The determination of flow-weighted annual average benzene concentration shall meet all of the following criteria:
- (i) The determination shall be made at the point of waste generation except for the specific cases given in paragraphs (c)(1)(i)(A) through (D) of this section.
- (A) The determination for sour water streams that are processed in sour water strippers shall be made at the point that the water exits the sour water stripper.
- (B) The determination for wastes at coke by-product plants subject to and complying with the control requirements of §61.132, 61.133, 61.134, or 61.139 of subpart L of this part shall be made at the location that the waste stream exits the process unit component or waste management unit controlled by that subpart or at the exit of the ammonia still, provided that the following conditions are met:
- (1) The transfer of wastes between units complying with the control requirements of subpart L of this part, process units, and the ammonia still is

made through hard piping or other enclosed system.

- (2) The ammonia still meets the definition of a sour water stripper in §61.341.
- (C) The determination for wastes that are received from offsite shall be made at the point where the waste enters the hazardous waste treatment, storage, or disposal facility.
- (D) The determination of flow-weighted annual average benzene concentration for process unit turnaround waste shall be made using either of the methods given in paragraph (c)(2) or (c)(3) of this section. The resulting flow-weighted annual average benzene concentration shall be included in the calculation of annual benzene quantity as provided in paragraph (a)(1)(iii) of this section for the year in which the turnaround occurs and for each subsequent year until the unit undergoes the next process unit turnaround.
- (ii) Volatilization of the benzene by exposure to air shall not be used in the determination to reduce the benzene concentration.
- (iii) Mixing or diluting the waste stream with other wastes or other materials shall not be used in the determination—to reduce the benzene concentration.
- (iv) The determination shall be made prior to any treatment of the waste that removes benzene, except as specified in paragraphs (c)(1)(i)(A) through (D) of this section.
- (v) For wastes with multiple phases, the determination shall provide the weighted-average benzene concentration based on the benzene concentration in each phase of the waste and the relative proportion of the phases.
- (2) Knowledge of the waste. The owner or operator shall provide sufficient information to document the flow-weighted annual average benzene concentration of each waste stream. Examples of information that could constitute knowledge include material balances, records of chemicals purchases, or previous test results provided the results are still relevant to the current waste stream conditions. If test data are used, then the owner or operator shall provide documentation describing the testing protocol and the means by which sampling variability

and analytical variability were accounted for in the determination of the flow-weighted annual average benzene concentration for the waste stream. When an owner or operator and the Administrator do not agree on determinations of the flow-weighted annual average benzene concentration based on knowledge of the waste, the procedures under paragraph (c)(3) of this section shall be used to resolve the disagreement.

- (3) Measurements of the benzene concentration in the waste stream in accordance with the following procedures:
- (i) Collect a minimum of three representative samples from each waste stream. Where feasible, samples shall be taken from an enclosed pipe prior to the waste being exposed to the atmosphere.
- (ii) For waste in enclosed pipes, the following procedures shall be used:
- (A) Samples shall be collected prior to the waste being exposed to the atmosphere in order to minimize the loss of benzene prior to sampling.
- (B) A static mixer shall be installed in the process line or in a by-pass line unless the owner or operator demonstrates that installation of a static mixer in the line is not necessary to accurately determine the benzene concentration of the waste stream.
- (C) The sampling tap shall be located within two pipe diameters of the static mixer outlet.
- (D) Prior to the initiation of sampling, sample lines and cooling coil shall be purged with at least four volumes of waste.
- (E) After purging, the sample flow shall be directed to a sample container and the tip of the sampling tube shall be kept below the surface of the waste during sampling to minimize contact with the atmosphere.
- (F) Samples shall be collected at a flow rate such that the cooling coil is able to maintain a waste temperature less than 10 °C (50 °F).
- (G) After filling, the sample container shall be capped immediately (within 5 seconds) to leave a minimum headspace in the container.
- (H) The sample containers shall immediately be cooled and maintained at

### **Environmental Protection Agency**

- a temperature below 10 °C (50 °F) for transfer to the laboratory.
- (iii) When sampling from an enclosed pipe is not feasible, a minimum of three representative samples shall be collected in a manner to minimize exposure of the sample to the atmosphere and loss of benzene prior to sampling.
- (iv) Each waste sample shall be analyzed using one of the following test methods for determining the benzene concentration in a waste stream:
- (A) Method 8020, Aromatic Volatile Organics, in "Test Methods for Evaluating Solid Waste, Physical/Chemical Methods," EPA Publication No. SW-846 (incorporation by reference as specified in §61.18 of this part);
- (B) Method 8021, Volatile Organic Compounds in Water by Purge and Trap Capillary Column Gas Chromatography with Photoionization and Electrolytic Conductivity Detectors in Series in "Test Methods for Evaluating Solid Waste, Physical/Chemical Methods," EPA Publication No. SW-846 (incorporation by reference as specified in §61.18 of this part);
- (C) Method 8240, Gas Chromatography/Mass Spectrometry for Volatile Organics in "Test Methods for Evaluating Solid Waste, Physical/Chemical Methods," EPA Publication No. SW-846 (incorporation by reference as specified in §61.18 of this part);
- (D) Method 8260, Gas Chromatography/Mass Spectrometry for Volatile Organics: Capillary Column Technique in "Test Methods for Evaluating Solid Waste, Physical/Chemical Methods," EPA Publication No. SW-846 (incorporation by reference as specified in §61.18 of this part);
- (E) Method 602, Purgeable Aromatics, as described in 40 CFR part 136, appendix A, Test Procedures for Analysis of Organic Pollutants, for wastewaters for which this is an approved EPA methods; or
- (F) Method 624, Purgeables, as described in 40 CFR part 136, appendix A, Test Procedures for Analysis of Organic Pollutants, for wastewaters for which this is an approved EPA method.
- (v) The flow-weighted annual average benzene concentration shall be calculated by averaging the results of the sample analyses as follows:

$$\overline{C} = \frac{1}{Q_t} \times \sum_{i=1}^{n} (Q_i)(C_i)$$

Where:

- C=Flow-weighted annual average benzene concentration for waste stream, ppmw.
- Q<sub>t</sub>=Total annual waste quantity for waste stream, kg/yr (lb/yr).
- n=Number of waste samples (at least 3).
- $Q_i$ =Annual waste quantity for waste stream represented by  $C_i$ , kg/yr (lb/yr).
- C<sub>i</sub>=Measured concentration of benzene in waste sample i, ppmw.
- (d) An owner or operator using performance tests to demonstrate compliance of a treatment process with §61.348 (a)(1)(i) shall measure the flowweighted annual average benzene concentration of the waste stream exiting the treatment process by collecting and analyzing a minimum of three representative samples of the waste stream using the procedures in paragraph (c)(3) of this section. The test shall be conducted under conditions that exist when the treatment process is operating at the highest inlet waste stream flow rate and benzene content expected to occur. Operations during periods of startup, shutdown, and malfunction shall not constitute representative conditions for the purpose of a test. The owner or operator shall record all process information as is necessary to document the operating conditions during the test.
- (e) An owner or operator using performance tests to demonstrate compliance of a treatment process with §61.348(a)(1)(ii) of this subpart shall determine the percent reduction of benzene in the waste stream on a mass basis by the following procedure:
- (1) The test shall be conducted under conditions that exist when the treatment process is operating at the highest inlet waste stream flow rate and benzene content expected to occur. Operations during periods of startup, shutdown, and malfunction shall not constitute representative conditions for the purpose of a test. The owner or operator shall record all process information as is necessary to document the operating conditions during the test.
- (2) All testing equipment shall be prepared and installed as specified in the appropriate test methods.

(3) The mass flow rate of benzene entering the treatment process (Eb) shall be determined by computing the product of the flow rate of the waste stream entering the treatment process, as determined by the inlet flow meter, and the benzene concentration of the waste stream, as determined using the sampling and analytical procedures specified in paragraph (c)(2) or (c)(3) of this section. Three grab samples of the waste shall be taken at equally spaced time intervals over a 1-hour period. Each 1-hour period constitutes a run, and the performance test shall consist of a minimum of 3 runs conducted over a 3-hour period. The mass flow rate of benzene entering the treatment process is calculated as follows:

$$E_b = \frac{K}{n \times 10^6} \left[ \sum_{i=1}^n V_i C_i \right]$$

Where:

 $E_{\rm b}$  = Mass flow rate of benzene entering the treatment process, kg/hr (lb/hr).

K = Density of the waste stream, kg/m<sup>3</sup> (lb/ft<sup>3</sup>).

V<sub>i</sub> = Average volume flow rate of waste entering the treatment process during each run i, m<sup>3</sup>/hr (ft<sup>3</sup>/hr).

 $C_i$  = Average concentration of benzene in the waste stream entering the treatment process during each run i, ppmw.

n = Number of runs.

106 = Conversion factor for ppmw.

(4) The mass flow rate of benzene exiting the treatment process (Ea) shall be determined by computing the product of the flow rate of the waste stream exiting the treatment process, as determined by the outlet flow meter or the inlet flow meter, and the benzene concentration of the waste stream, as determined using the sampling and analytical procedures specified in paragraph (c)(2) or (c)(3) of this section. Three grab samples of the waste shall be taken at equally spaced time intervals over a 1-hour period. Each 1-hour period constitutes a run, and the performance test shall consist of a minimum of 3 runs conducted over the same 3-hour period at which the mass flow rate of benzene entering the treatment process is determined. The mass flow rate of benzene exiting the treatment process is calculated as follows:

$$E_a = \frac{K}{n \times 10^6} \left[ \sum_{i=1}^n V_i C_i \right]$$

Where:

E<sub>a</sub> = Mass flow rate of benzene exiting the treatment process, kg/hr (lb/hr).

K = Density of the waste stream, kg/m³ (lb/ ft³).

V<sub>i</sub> = Average volume flow rate of waste exiting the treatment process during each run i, m<sup>3</sup>/hr (ft<sup>3</sup>/hr).

C<sub>i</sub> = Average concentration of benzene in the waste stream exiting the treatment process during each run i, ppmw.

n = Number of runs.

 $10^6$  = Conversion factor for ppmw.

- (f) An owner or operator using performance tests to demonstrate compliance of a treatment process with §61.348(a)(1)(iii) of this subpart shall determine the benzene destruction efficiency for the combustion unit by the following procedure:
- (1) The test shall be conducted under conditions that exist when the combustion unit is operating at the highest inlet waste stream flow rate and benzene content expected to occur. Operations during periods of startup, shutdown, and malfunction shall not constitute representative conditions for the purpose of a test. The owner or operator shall record all process information necessary to document the operating conditions during the test.
- (2) All testing equipment shall be prepared and installed as specified in the appropriate test methods.
- (3) The mass flow rate of benzene entering the combustion unit shall be determined by computing the product of the flow rate of the waste stream entering the combustion unit, as determined by the inlet flow meter, and the benzene concentration of the waste stream, as determined using the sampling procedures in paragraph (c)(2) or (c)(3) of this section. Three grab samples of the waste shall be taken at equally spaced time intervals over a 1hour period. Each 1-hour period constitutes a run, and the performance test shall consist of a minimum of 3 runs conducted over a 3-hour period. The mass flow rate of benzene into the combustion unit is calculated as follows:

$$E_b = \frac{K}{n \times 10^6} \left[ \sum_{i=1}^n V_i C_i \right]$$

Where:

E<sub>b</sub> = Mass flow rate of benzene entering the combustion unit, kg/hr (lb/hr).

K = Density of the waste stream, kg/m3 (lb/

Vi = Average volume flow rate of waste entering the combustion unit during each run i, m³/hr (ft³/hr).

 $C_i$  = Average concentration of benzene in the waste stream entering the combustion unit during each run i, ppmw.

n = Number of runs.

106 = Conversion factor for ppmw.

(4) The mass flow rate of benzene exiting the combustion unit exhaust stack shall be determined as follows:

- (i) The time period for the test shall not be less than 3 hours during which at least 3 stack gas samples are collected and be the same time period at which the mass flow rate of benzene entering the treatment process is determined. Each sample shall be collected over a 1-hour period (e.g., in a tedlar bag) to represent a time-integrated composite sample and each 1-hour period shall correspond to the periods when the waste feed is sampled.
- (ii) A run shall consist of a 1-hour period during the test. For each run:
- (A) The reading from each measurement shall be recorded:
- (B) The volume exhausted shall be determined using Method 2, 2A, 2C, or 2D from appendix A of 40 CFR part 60, as appropriate.
- (C) The average benzene concentration in the exhaust downstream of the combustion unit shall be determined using Method 18 from appendix A of 40 CFR part 60.
- (iii) The mass of benzene emitted during each run shall be calculated as follows:

$$M_i = D_b VC(10^{-6})$$

Where:

 $M_i$  = Mass of benzene emitted during run i, kg (lb).

V = Volume of air-vapor mixture exhausted at standard conditions, m3 (ft3).

C = Concentration of benzene measured in

the exhaust, ppmv.  $D_b$  = Density of benzene, 3.24 kg/m³ (0.202 lb/

106 = Conversion factor for ppmv.

(iv) The benzene mass emission rate in the exhaust shall be calculated as follows:

$$\mathbf{E}_{\mathbf{a}} = \left(\sum_{i=1}^{n} \mathbf{M}_{i}\right) / \mathbf{T}$$

Where:

 $E_a$  = Mass flow rate of benzene emitted from the combustion unit, kg/hr (lb/hr).

M<sub>i</sub> = Mass of benzene emitted from the combustion unit during run i, kg (lb).

T = Total time of all runs, hr.

n = Number of runs.

(5) The benzene destruction efficiency for the combustion unit shall be calculated as follows:

$$R = \frac{E_b - E_a}{E_b} \times 100$$

Where:

R = Benzene destruction efficiency for the combustion unit, percent.

 $E_b$  = Mass flow rate of benzene entering the combustion unit, kg/hr (lb/hr).

Ea = Mass flow rate of benzene emitted from the combustion unit, kg/hr (lb/hr).

(g) An owner or operator using performance tests to demonstrate compliance of a wastewater treatment system unit with §61.348(b) shall measure the flow-weighted annual average benzene concentration of the wastewater stream where the waste stream enters an exempt waste management unit by collecting and analyzing a minimum of three representative samples of the waste stream using the procedures in paragraph (c)(3) of this section. The test shall be conducted under conditions that exist when the wastewater treatment system is operating at the highest inlet wastewater stream flow rate and benzene content expected to occur. Operations during periods of startup, shutdown, and malfunction shall not constitute representative conditions for the purpose of a test. The owner or operator shall record all process information as is necessary to document the operating conditions during the test.

(h) An owner or operator shall test equipment for compliance with no detectable emissions as required in §§ 61.343 through 61.347, and § 61.349 of this subpart in accordance with the following requirements:

- (1) Monitoring shall comply with Method 21 from appendix A of 40 CFR part 60.
- (2) The detection instrument shall meet the performance criteria of Method 21.
- (3) The instrument shall be calibrated before use on each day of its use by the procedures specified in Method 21.
  - (4) Calibration gases shall be:
- (i) Zero air (less than 10 ppm of hydrocarbon in air); and
- (ii) A mixture of methane or n-hexane and air at a concentration of approximately, but less than, 10,000 ppm methane or n-hexane.
- (5) The background level shall be determined as set forth in Method 21.
- (6) The instrument probe shall be traversed around all potential leak interfaces as close as possible to the interface as described in Method 21.
- (7) The arithmetic difference between the maximum concentration indicated by the instrument and the background level is compared to 500 ppm for determining compliance.
- (i) An owner or operator using a performance test to demonstrate compliance of a control device with either the organic reduction efficiency requirement or the benzene reduction efficiency requirement specified under \$61.349(a)(2) shall use the following procedures:
- (1) The test shall be conducted under conditions that exist when the waste management unit vented to the control device is operating at the highest load or capacity level expected to occur. Operations during periods of startup, shutdown, and malfunction shall not constitute representative conditions for the purpose of a test. The owner or operator shall record all process information necessary to document the operating conditions during the test.
- (2) Sampling sites shall be selected using Method 1 or 1A from appendix A of 40 CFR part 60, as appropriate.
- (3) The mass flow rate of either the organics or benzene entering and exiting the control device shall be determined as follows:
- (i) The time period for the test shall not be less than 3 hours during which

at least 3 stack gas samples are collected. Samples of the vent stream entering and exiting the control device shall be collected during the same time period. Each sample shall be collected over a 1-hour period (e.g., in a tedlar bag) to represent a time-integrated composite sample.

- (ii) A run shall consist of a 1-hour period during the test. For each run:
- (A) The reading from each measurement shall be recorded;
- (B) The volume exhausted shall be determined using Method 2, 2A, 2C, or 2D from appendix A of 40 CFR part 60, as appropriate;
- (C) The organic concentration or the benzene concentration, as appropriate, in the vent stream entering and exiting the control shall be determined using Method 18 from appendix A of 40 CFR part 60.
- (iii) The mass of organics or benzene entering and exiting the control device during each run shall be calculated as follows:

$$M_{aj} = \frac{K_1 V_{aj}}{10^6} \left( \sum_{i=1}^{n} C_{ai} M W_i \right)$$

$$M_{bj} = \frac{K_1 V_{bj}}{10^6} \left( \sum_{i=1}^{n} C_{bi} M W_i \right)$$

- $M_{aj} = Mass$  of organics or benzene in the vent stream entering the control device during run j, kg (lb).
- $M_{bj}$  = Mass of organics or benzene in the vent stream exiting the control device during run j, kg (lb).
- $V_{aj} = V$ olume of vent stream entering the control device during run j, at standard conditions, m<sup>3</sup> (ft<sup>3</sup>).
- $V_{bj}$  = Volume of vent stream exiting the control device during run j, at standard conditions.  $m^3$  (ft<sup>3</sup>).
- C<sub>ai</sub> = Organic concentration of compound i or the benzene concentration measured in the vent stream entering the control device as determined by Method 18, ppm by volume on a dry basis.
- $C_{\mathrm{bi}} = \mathrm{Organic}$  concentration of compound i or the benzene concentration measured in the vent stream exiting the control device as determined by Method 18, ppm by volume on a dry basis.
- MW<sub>i</sub> = Molecular weight of organic compound i in the vent stream, or the molecular weight of benzene, kg/kg-mol (lb/lbmole)

#### **Environmental Protection Agency**

- n = Number of organic compounds in the vent stream; if benzene reduction efficiency is being demonstrated, then n=1.
- K<sub>1</sub> = Conversion factor for molar volume at standard conditions (293 K and 760 mm Hg (527 R and 14.7 psia))
- = 0.0416 kg-mol/m<sup>3</sup> (0.00118 lb-mol/ft<sup>3</sup>)  $10^{-6}$ =Conversion factor for ppmv.
- (iv) The mass flow rate of organics or benzene entering and exiting the control device shall be calculated as follows:

$$E_a - \left(\sum_{j=1}^n \mathbf{M}_{aj}\right) / T$$

$$E_b - \left(\sum_{i=1}^n M_{bj}\right) / T$$

Where:

- E<sub>a</sub> = Mass flow rate of organics or benzene entering the control device, kg/hr (lb/hr).
- $E_b$  = Mass flow rate of organics or benzene exiting the control device, kg/hr (lb/hr).
- $M_{aj}=Mass$  of organics or benzene in the vent stream entering the control device during run j, kg (lb).
- M  $_{bj}$  = Mass of organics or benzene in the vent stream exiting the control device during run j, kg (lb).
- T = Total time of all runs, hr.
- n = Number of runs.
- (4) The organic reduction efficiency or the benzene reduction efficiency for the control device shall be calculated as follows:

$$R = \frac{E_a - E_b}{E_a} \times 100$$

Where:

- R = Total organic reduction of efficiency or benzene reduction efficiency for the control device, percent.
- E<sub>b</sub> = Mass flow rate of organics or benzene entering the control device, kg/hr (lb/hr).
- E<sub>a</sub> = Mass flow rate of organic or benzene emitted from the control device, kg/hr (lb/hr).
- (j) An owner or operator shall determine the benzene quantity for the purposes of the calculation required by §61.342 (c)(3)(ii)(B) according to the provisions of paragraph (a) of this section, except that the procedures in paragraph (a) of this section shall also

apply to wastes with a water content of 10 percent or less.

- (k) An owner or operator shall determine the benzene quantity for the purposes of the calculation required by §61.342(e)(2) by the following procedure:
- (1) For each waste stream that is not controlled for air emissions in accordance with §61.343. 61.344, 61.345, 61.346, 61.347, or 61.348(a), as applicable to the waste management unit that manages the waste, the benzene quantity shall be determined as specified in paragraph (a) of this section, except that paragraph (b)(4) of this section shall not apply, i.e., the waste quantity for process unit turnaround waste is not annualized but shall be included in the determination of benzene quantity for the year in which the waste is generated for the purposes of the calculation required by  $\S61.342(e)(2)$ .
- (2) For each waste stream that is controlled for air emissions in accordance with §61.343. 61.344, 61.345, 61.346, 61.347, or 61.348(a), as applicable to the waste management unit that manages the waste, the determination of annual waste quantity and flow-weighted annual average benzene concentration shall be made at the first applicable location as described in paragraphs (k)(2)(i), (k)(2)(ii), and (k)(2)(iii) of this section and prior to any reduction of benzene concentration through volatilization of the benzene, using the methods given in (k)(2)(iv) and (k)(2)(v) of this section.
- (i) Where the waste stream enters the first waste management unit not complying with §§61.343, 61.344, 61.345, 61.346, 61.347, and 61.348(a) that are applicable to the waste management unit,
- (ii) For each waste stream that is managed or treated only in compliance with §§ 61.343 through 61.348(a) up to the point of final direct discharge from the facility, the determination of benzene quantity shall be prior to any reduction of benzene concentration through volatilization of the benzene, or
- (iii) For wastes managed in units controlled for air emissions in accordance with §§61.343, 61.344, 61.345, 61.346, 61.347, and 61.348(a), and then transferred offsite, facilities shall use the first applicable offsite location as described in paragraphs (k)(2)(i) and

(k)(2)(ii) of this section if they have documentation from the offsite facility of the benzene quantity at this location. Facilities without this documentation for offsite wastes shall use the benzene quantity determined at the point where the transferred waste leaves the facility.

- (iv) Annual waste quantity shall be determined using the procedures in paragraphs (b)(5), (6), or (7) of this section, and
- (v) The flow-weighted annual average benzene concentration shall be determined using the procedures in paragraphs (c)(2) or (3) of this section.
- (3) The benzene quantity in a waste stream that is generated less than one time per year, including process unit turnaround waste, shall be included in the determination of benzene quantity as determined in paragraph (k)(6) of this section for the year in which the waste is generated. The benzene quantity in this waste stream shall not be annualized or averaged over the time interval between the activities that resulted in generation of the waste for purposes of determining benzene quantity as determined in paragraph (k)(6) of this section.
- (4) The benzene in waste entering an enhanced biodegradation unit, as defined in §61.348(b)(2)(ii)(B), shall not be included in the determination of benzene quantity, determined in paragraph (k)(6) of this section, if the following conditions are met:
- (i) The benzene concentration for each waste stream entering the enhanced biodegradation unit is less than 10 ppmw on a flow-weighted annual average basis, and
- (ii) All prior waste management units managing the waste comply with §§ 61.343, 61.344, 61.345, 61.346, 61.347 and 61.348(a).
- (5) The benzene quantity for each waste stream in paragraph (k)(2) of this section shall be determined by multiplying the annual waste quantity of each waste stream times its flow-weighted annual average benzene concentration.
- (6) The total benzene quantity for the purposes of the calculation required by §61.342(e)(2) shall be determined by adding together the benzene quantities determined in paragraphs (k)(1) and

- (k)(5) of this section for each applicable waste stream.
- (7) If the benzene quantity determined in paragraph (6) of this section exceeds 6.0 Mg/yr (6.6 ton/yr) only because of multiple counting of the benzene quantity for a waste stream, the owner or operator may use the following procedures for the purposes of the calculation required by §61.342(e)(2):
- (i) Determine which waste management units are involved in the multiple counting of benzene;
- (ii) Determine the quantity of benzene that is emitted, recovered, or removed from the affected units identified in paragraph (k)(7)(i) of this section, or destroyed in the units if applicable, using either direct measurements or the best available estimation techniques developed or approved by the Administrator.
- (iii) Adjust the benzene quantity to eliminate the multiple counting of benzene based on the results from paragraph (k)(7)(ii) of this section and determine the total benzene quantity for the purposes of the calculation required by §61.342(e)(2).
- (iv) Submit in the annual report required under §61.357(a) a description of the methods used and the resulting calculations for the alternative procedure under paragraph (k)(7) of this section, the benzene quantity determination from paragraph (k)(6) of this section, and the adjusted benzene quantity determination from paragraph (k)(7)(iii) of this section.

[55 FR 8346, Mar. 7, 1990; 55 FR 12444, Apr. 3, 1990, as amended at 55 FR 37231, Sept. 10, 1990; 58 FR 3099, Jan. 7, 1993; 65 FR 62160, Oct. 17, 2000]

### §61.356 Recordkeeping requirements.

- (a) Each owner or operator of a facility subject to the provisions of this subpart shall comply with the record-keeping requirements of this section. Each record shall be maintained in a readily accessible location at the facility site for a period not less than two years from the date the information is recorded unless otherwise specified.
- (b) Each owner or operator shall maintain records that identify each waste stream at the facility subject to this subpart, and indicate whether or

not the waste stream is controlled for benzene emissions in accordance with this subpart. In addition the owner or operator shall maintain the following records:

- (1) For each waste stream not controlled for benzene emissions in accordance with this subpart, the records shall include all test results, measurements, calculations, and other documentation used to determine the following information for the waste stream: waste stream identification, water content, whether or not the waste stream is a process wastewater stream, annual waste quantity, range of benzene concentrations, annual average flow-weighted benzene concentration, and annual benzene quantity.
- (2) For each waste stream exempt from §61.342(c)(1) in accordance with §61.342(c)(3), the records shall include:
- (i) All measurements, calculations, and other documentation used to determine that the continuous flow of process wastewater is less than 0.02 liters (0.005 gallons) per minute or the annual waste quantity of process wastewater is less than 10 Mg/yr (11 ton/yr) in accordance with §61.342(c)(3)(i), or
- (ii) All measurements, calculations, and other documentation used to determine that the sum of the total annual benzene quantity in all exempt waste streams does not exceed 2.0 Mg/yr (2.2 ton/yr) in accordance with §61.342(c)(3)(ii).
- (3) For each facility where process wastewater streams are controlled for benzene emissions in accordance with §61.342(d) of this subpart, the records shall include for each treated process wastewater stream all measurements, calculations, and other documentation used to determine the annual benzene quantity in the process wastewater stream exiting the treatment process.
- (4) For each facility where waste streams are controlled for benzene emissions in accordance with §61.342(e), the records shall include for each waste stream all measurements, including the locations of the measurements, calculations, and other documentation used to determine that the total benzene quantity does not exceed 6.0 Mg/yr (6.6 ton/yr).

- (5) For each facility where the annual waste quantity for process unit turnaround waste is determined in accordance with §61.355(b)(5), the records shall include all test results, measurements, calculations, and other documentation used to determine the following information: identification of each process unit at the facility that undergoes turnarounds, the date of the most recent turnaround for each process unit, identification of each process unit turnaround waste, the water content of each process unit turnaround waste, the annual waste quantity deaccordance termined in §61.355(b)(5), the range of benzene concentrations in the waste, the annual average flow-weighted benzene concentration of the waste, and the annual benzene quantity calculated in accordance with §61.355(a)(1)(iii) of this section.
- (6) For each facility where wastewater streams are controlled for benzene emissions in accordance with §61.348(b)(2), the records shall include all measurements, calculations, and other documentation used to determine the annual benzene content of the waste streams and the total annual benzene quantity contained in all waste streams managed or treated in exempt waste management units.
- (c) An owner or operator transferring waste off-site to another facility for treatment in accordance with §61.342(f) shall maintain documentation for each offsite waste shipment that includes the following information: Date waste is shipped offsite, quantity of waste shipped offsite, name and address of the facility receiving the waste, and a copy of the notice sent with the waste shipment.
- (d) An owner or operator using control equipment in accordance with §§61.343 through 61.347 shall maintain engineering design documentation for all control equipment that is installed on the waste management unit. The documentation shall be retained for the life of the control equipment. If a control device is used, then the owner or operator shall maintain the control device records required by paragraph (f) of this section.

- (e) An owner or operator using a treatment process or wastewater treatment system unit in accordance with §61.348 of this subpart shall maintain the following records. The documentation shall be retained for the life of the unit.
- (1) A statement signed and dated by the owner or operator certifying that the unit is designed to operate at the documented performance level when the waste stream entering the unit is at the highest waste stream flow rate and benzene content expected to occur.
- (2) If engineering calculations are used to determine treatment process or wastewater treatment system unit performance, then the owner or operator shall maintain the complete design analysis for the unit. The design analysis shall include for example the following information: Design specifications, drawings, schematics, piping and instrumentation diagrams, and other documentation necessary to demonstrate the unit performance.
- (3) If performance tests are used to determine treatment process or wastewater treatment system unit performance, then the owner or operator shall maintain all test information necessary to demonstrate the unit performance.
- (i) A description of the unit including the following information: type of treatment process; manufacturer name and model number; and for each waste stream entering and exiting the unit, the waste stream type (e.g., process wastewater, sludge, slurry, etc.), and the design flow rate and benzene content.
- (ii) Documentation describing the test protocol and the means by which sampling variability and analytical variability were accounted for in the determination of the unit performance. The description of the test protocol shall include the following information: sampling locations, sampling method, sampling frequency, and analytical procedures used for sample analysis.
- (iii) Records of unit operating conditions during each test run including all key process parameters.
  - (iv) All test results.
- (4) If a control device is used, then the owner or operator shall maintain

- the control device records required by paragraph (f) of this section.
- (f) An owner or operator using a closed-vent system and control device in accordance with §61.349 of this subpart shall maintain the following records. The documentation shall be retained for the life of the control device.
- (1) A statement signed and dated by the owner or operator certifying that the closed-vent system and control device is designed to operate at the documented performance level when the waste management unit vented to the control device is or would be operating at the highest load or capacity expected to occur.
- (2) If engineering calculations are used to determine control device performance in accordance with §61.349(c), then a design analysis for the control device that includes for example:
- (i) Specifications, drawings, schematics, and piping and instrumentation diagrams prepared by the owner or operator, or the control device manufacturer or vendor that describe the control device design based on acceptable engineering texts. The design analysis shall address the following vent stream characteristics and control device operating parameters:
- (A) For a thermal vapor incinerator, the design analysis shall consider the vent stream composition, constituent concentrations, and flow rate. The design analysis shall also establish the design minimum and average temperature in the combustion zone and the combustion zone residence time.
- (B) For a catalytic vapor incinerator, the design analysis shall consider the vent stream composition, constituent concentrations, and flow rate. The design analysis shall also establish the design minimum and average temperatures across the catalyst bed inlet and outlet.
- (C) For a boiler or process heater, the design analysis shall consider the vent stream composition, constituent concentrations, and flow rate. The design analysis shall also establish the design minimum and average flame zone temperatures, combustion zone residence time, and description of method and location where the vent stream is introduced into the flame zone.

#### **Environmental Protection Agency**

- (D) For a flare, the design analysis shall consider the vent stream composition, constituent concentrations, and flow rate. The design analysis shall also consider the requirements specified in 40 CFR 60.18.
- (E) For a condenser, the design analysis shall consider the vent stream composition, constituent concentration, flow rate, relative humidity, and temperature. The design analysis shall also establish the design outlet organic compound concentration level or the design outlet benzene concentration level, design average temperature of the condenser exhaust vent stream, and the design average temperatures of the coolant fluid at the condenser inlet and outlet.
- (F) For a carbon adsorption system that regenerates the carbon bed directly on-site in the control device such as a fixed-bed adsorber, the design analysis shall consider the vent stream composition, constituent concentration, flow rate, relative humidity, and temperature. The design analysis shall also establish the design exhaust vent stream organic compound concentration level or the design exhaust vent stream benzene concentration level, number and capacity of carbon beds, type and working capacity of activated carbon used for carbon beds, design total steam flow over the period of each complete carbon bed regeneration cycle, duration of the carbon bed steaming and cooling/drying cycles, design carbon bed temperature after regeneration, design carbon bed regeneration time, and design service life of carbon.
- (G) For a carbon adsorption system that does not regenerate the carbon bed directly on-site in the control device, such as a carbon canister, the design analysis shall consider the vent stream composition, constituent concentration, flow rate, relative humidity, and temperature. The design analysis shall also establish the design exhaust vent stream organic compound concentration level or the design exhaust vent stream benzene concentration level, capacity of carbon bed, type and working capacity of activated carbon used for carbon bed, and design carbon replacement interval based on the total carbon working capacity of

the control device and source operating schedule.

- (H) For a control device subject to the requirements of §61.349(a)(2)(iv), the design analysis shall consider the vent stream composition, constituent concentration, and flow rate. The design analysis shall also include all of the information submitted under §61.349 (a)(2)(iv).
  - (ii) [Reserved]
- (3) If performance tests are used to determine control device performance in accordance with §61.349(c) of this subpart:
- (i) A description of how it is determined that the test is conducted when the waste management unit or treatment process is operating at the highest load or capacity level. This description shall include the estimated or design flow rate and organic content of each vent stream and definition of the acceptable operating ranges of key process and control parameters during the test program.
- (ii) A description of the control device including the type of control device, control device manufacturer's name and model number, control device dimensions, capacity, and construction materials.
- (iii) A detailed description of sampling and monitoring procedures, including sampling and monitoring locations in the system, the equipment to be used, sampling and monitoring frequency, and planned analytical procedures for sample analysis.
  - (iv) All test results.
- (g) An owner or operator shall maintain a record for each visual inspection required by §§61.343 through 61.347 of this subpart that identifies a problem (such as a broken seal, gap or other problem) which could result in benzene emissions. The record shall include the date of the inspection, waste management unit and control equipment location where the problem is identified, a description of the problem, a description of the corrective action taken, and the date the corrective action was completed.
- (h) An owner or operator shall maintain a record for each test of no detectable emissions required by §§61.343

### § 61.356

through 61.347 and §61.349 of this subpart. The record shall include the following information: date the test is performed, background level measured during test, and maximum concentration indicated by the instrument reading measured for each potential leak interface. If detectable emissions are measured at a leak interface, then the record shall also include the waste management unit, control equipment, and leak interface location where detectable emissions were measured, a description of the problem, a description of the corrective action taken, and the date the corrective action was completed.

- (i) For each treatment process and wastewater treatment system unit operated to comply with §61.348, the owner or operator shall maintain documentation that includes the following information regarding the unit operation:
- (1) Dates of startup and shutdown of the unit.
- (2) If measurements of waste stream benzene concentration are performed in accordance with §61.354(a)(1) of this subpart, the owner or operator shall maintain records that include date each test is performed and all test results.
- (3) If a process parameter is continuously monitored in accordance with §61.354(a)(2) of this subpart, the owner or operator shall maintain records that include a description of the operating parameter (or parameters) to be monitored to ensure that the unit will be operated in conformance with these standards and the unit's design specifications, and an explanation of the criteria used for selection of that parameter (or parameters). This documentation shall be kept for the life of the unit.
- (4) If measurements of waste stream benzene concentration are performed in accordance with §61.354(b), the owner or operator shall maintain records that include the date each test is performed and all test results.
- (5) Periods when the unit is not operated as designed.
- (j) For each control device, the owner or operator shall maintain documentation that includes the following infor-

mation regarding the control device operation:

- (1) Dates of startup and shutdown of the closed-vent system and control device.
- (2) A description of the operating parameter (or parameters) to be monitored to ensure that the control device will be operated in conformance with these standards and the control device's design specifications and an explanation of the criteria used for selection of that parameter (or parameters). This documentation shall be kept for the life of the control device.
- (3) Periods when the closed-vent system and control device are not operated as designed including all periods and the duration when:
- (i) Any valve car-seal or closure mechanism required under §61.349(a)(1)(ii) is broken or the by-pass line valve position has changed.
- (ii) The flow monitoring devices required under §61.349(a)(1)(ii) indicate that vapors are not routed to the control device as required.
- (4) If a thermal vapor incinerator is used, then the owner or operator shall maintain continuous records of the temperature of the gas stream in the combustion zone of the incinerator and records of all 3-hour periods of operation during which the average temperature of the gas stream in the combustion zone is more than 28 °C (50 °F) below the design combustion zone temperature.
- (5) If a catalytic vapor incinerator is used, then the owner or operator shall maintain continuous records of the temperature of the gas stream both upstream and downstream of the catalyst bed of the incinerator, records of all 3-hour periods of operation during which the average temperature measured before the catalyst bed is more than 28 °C (50 °F) below the design gas stream temperature, and records of all 3-hour periods of operation during which the average temperature difference across the catalyst bed is less than 80 percent of the design temperature difference.
- (6) If a boiler or process heater is used, then the owner or operator shall maintain records of each occurrence when there is a change in the location at which the vent stream is introduced into the flame zone as required by

 $\S61.349(a)(2)(i)(C)$ . For a boiler or process heater having a design heat input capacity less than 44 MW (150  $\times$  106 BTU/hr), the owner or operator shall maintain continuous records of the temperature of the gas stream in the combustion zone of the boiler or process heater and records of all 3-hour periods of operation during which the average temperature of the gas stream in the combustion zone is more than 28 °C (50 °F) below the design combustion zone temperature. For a boiler or process heater having a design heat input capacity greater than or equal to 44 MW (150  $\times$  106 BTU/hr), the owner or operator shall maintain continuous records of the parameter(s) monitored in accordance with the requirements of §61.354(c)(5).

- (7) If a flare is used, then the owner or operator shall maintain continuous records of the flare pilot flame monitoring and records of all periods during which the pilot flame is absent.
- (8) If a condenser is used, then the owner or operator shall maintain records from the monitoring device of the parameters selected to be monitored in accordance with §61.354(c)(6). If concentration of organics or concentration of benzene in the control device outlet gas stream is monitored, then the owner or operator shall record all 3-hour periods of operation during which the concentration of organics or the concentration of benzene in the exhaust stream is more than 20 percent greater than the design value. If the temperature of the condenser exhaust stream and coolant fluid is monitored. then the owner or operator shall record all 3-hour periods of operation during which the temperature of the condenser exhaust vent stream is more than 6 °C (11 °F) above the design average exhaust vent stream temperature, or the temperature of the coolant fluid exiting the condenser is more than 6 °C (11 °F) above the design average coolant fluid temperature at the condenser outlet.
- (9) If a carbon adsorber is used, then the owner or operator shall maintain records from the monitoring device of the concentration of organics or the concentration of benzene in the control device outlet gas stream. If the concentration of organics or the concentration of organics or the

centration of benzene in the control device outlet gas stream is monitored, then the owner or operator shall record all 3-hour periods of operation during which the concentration of organics or the concentration of benzene in the exhaust stream is more than 20 percent greater than the design value. If the carbon bed regeneration interval is monitored, then the owner or operator shall record each occurrence when the vent stream continues to flow through the control device beyond the predetermined carbon bed regeneration time.

- (10) If a carbon adsorber that is not regenerated directly on site in the control device is used, then the owner or operator shall maintain records of dates and times when the control device is monitored, when breakthrough is measured, and shall record the date and time then the existing carbon in the control device is replaced with fresh carbon.
- (11) If an alternative operational or process parameter is monitored for a control device, as allowed in §61.354(e) of this subpart, then the owner or operator shall maintain records of the continuously monitored parameter, including periods when the device is not operated as designed.
- (12) If a control device subject to the requirements of §61.349(a)(2)(iv) is used, then the owner or operator shall maintain records of the parameters that are monitored and each occurrence when the parameters monitored are outside the range of values specified in §61.349(a)(2)(iv)(C), or other records as specified by the Administrator.
- (k) An owner or operator who elects to install and operate the control equipment in §61.351 of this subpart shall comply with the recordkeeping requirements in 40 CFR 60.115b.
- (1) An owner or operator who elects to install and operate the control equipment in §61.352 of this subpart shall maintain records of the following:
- (1) The date, location, and corrective action for each visual inspection required by 40 CFR 60.693-2(a)(5), during which a broken seal, gap, or other problem is identified that could result in benzene emissions.
- (2) Results of the seal gap measurements required by 40 CFR 60.693-2(a).

#### §61.357

- (m) If a system is used for emission control that is maintained at a pressure less than atmospheric pressure with openings to provide dilution air, then the owner or operator shall maintain records of the monitoring device and records of all periods during which the pressure in the unit is operated at a pressure that is equal to or greater than atmospheric pressure.
- (n) Each owner or operator using a total enclosure to comply with control requirements for tanks in §61.343 or the control requirements for containers in §61.345 must keep the records required in paragraphs (n)(1) and (2) of this section. Owners or operators may use records as required in 40 CFR 264.1089(b)(2)(iv) CFR or 265.1090(b)(2)(iv) for a tank or as required in 40 CFR 264.1089(d)(1) or 40 CFR 265.1090(d)(1) for a container to meet the recordkeeping requirement in paragraph (n)(1) of this section. The owner or operator must make the records of each verification of a total enclosure available for inspection upon request.
- (1) Records of the most recent set of calculations and measurements performed to verify that the enclosure meets the criteria of a permanent total enclosure as specified in "Procedure T—Criteria for and Verification of a Permanent or Temporary Total Enclosure" in 40 CFR 52.741, appendix B;
- (2) Records required for a closed-vent system and control device according to the requirements in paragraphs (d) (f), and (j) of this section.

[55 FR 8346, Mar. 7, 1990; 55 FR 12444, Apr. 3, 1990; 55 FR 18331, May 2, 1990, as amended at 58 FR 3103, Jan. 7, 1993; 65 FR 62161, Oct. 17, 2000; 67 FR 68533, Nov. 12, 2002]

### § 61.357 Reporting requirements.

(a) Each owner or operator of a chemical plant, petroleum refinery, coke byproduct recovery plant, and any facility managing wastes from these industries shall submit to the Administrator within 90 days after January 7, 1993, or by the initial startup for a new source with an initial startup after the effective date, a report that summarizes the regulatory status of each waste stream subject to §61.342 and is determined by the procedures specified in §61.355(c) to contain benzene. Each owner or oper-

ator subject to this subpart who has no benzene onsite in wastes, products, by-products, or intermediates shall submit an initial report that is a statement to this effect. For all other owners or operators subject to this subpart, the report shall include the following information:

- (1) Total annual benzene quantity from facility waste determined in accordance with §61.355(a) of this subpart.
- (2) A table identifying each waste stream and whether or not the waste stream will be controlled for benzene emissions in accordance with the requirements of this subpart.
- (3) For each waste stream identified as not being controlled for benzene emissions in accordance with the requirements of this subpart the following information shall be added to the table:
- (i) Whether or not the water content of the waste stream is greater than 10 percent;
- (ii) Whether or not the waste stream is a process wastewater stream, product tank drawdown, or landfill leachate;
- (iii) Annual waste quantity for the waste stream;
- (iv) Range of benzene concentrations for the waste stream;
- (v) Annual average flow-weighted benzene concentration for the waste stream; and
- (vi) Annual benzene quantity for the waste stream.
- (4) The information required in paragraphs (a) (1), (2), and (3) of this section should represent the waste stream characteristics based on current configuration and operating conditions. An owner or operator only needs to list in the report those waste streams that contact materials containing benzene. The report does not need to include a description of the controls to be installed to comply with the standard or other information required in §61.10(a).
- (b) If the total annual benzene quantity from facility waste is less than 1 Mg/yr (1.1 ton/yr), then the owner or operator shall submit to the Administrator a report that updates the information listed in paragraphs (a)(1) through (a)(3) of this section whenever

there is a change in the process generating the waste stream that could cause the total annual benzene quantity from facility waste to increase to 1 Mg/yr (1.1 ton/yr) or more.

- (c) If the total annual benzene quantity from facility waste is less than 10 Mg/vr (11 ton/vr) but is equal to or greater than 1 Mg/yr (1.1 ton/yr), then the owner or operator shall submit to the Administrator a report that updates the information listed in paragraphs (a)(1) through (a)(3) of this section. The report shall be submitted annually and whenever there is a change in the process generating the waste stream that could cause the total annual benzene quantity from facility waste to increase to 10 Mg/yr (11 ton/yr) or more. If the information in the annual report required by paragraphs (a)(1) through (a)(3) of this section is not changed in the following year, the owner or operator may submit a statement to that effect.
- (d) If the total annual benzene quantity from facility waste is equal to or greater than 10 Mg/yr (11 ton/yr), then the owner or operator shall submit to the Administrator the following reports:
- (1) Within 90 days after January 7, 1993, unless a waiver of compliance under §61.11 of this part is granted, or by the date of initial startup for a new source with an initial startup after the effective date, a certification that the equipment necessary to comply with these standards has been installed and that the required initial inspections or tests have been carried out in accordance with this subpart. If a waiver of compliance is granted under §61.11, the certification of equipment necessary to comply with these standards shall be submitted by the date the waiver of compliance expires.
- (2) Beginning on the date that the equipment necessary to comply with these standards has been certified in accordance with paragraph (d)(1) of this section, the owner or operator shall submit annually to the Administrator a report that updates the information listed in paragraphs (a)(1) through (a)(3) of this section. If the information in the annual report required by paragraphs (a)(1) through (a)(3) of this section is not changed in

the following year, the owner or operator may submit a statement to that effect.

- (3) If an owner or operator elects to comply with the requirements of §61.342(c)(3)(ii), then the report required by paragraph (d)(2) of this section shall include a table identifying each waste stream chosen for exemption and the total annual benzene quantity in these exempted streams.
- (4) If an owner or operator elects to comply with the alternative requirements of §61.342(d) of this subpart, then he shall include in the report required by paragraph (d)(2) of this section a table presenting the following information for each process wastewater stream:
- (i) Whether or not the process wastewater stream is being controlled for benzene emissions in accordance with the requirements of this subpart;
- (ii) For each process wastewater stream identified as not being controlled for benzene emissions in accordance with the requirements of this subpart, the table shall report the following information for the process wastewater stream as determined at the point of waste generation: annual waste quantity, range of benzene concentrations, annual average flow-weighted benzene concentration, and annual benzene quantity;
- (iii) For each process wastewater stream identified as being controlled for benzene emissions in accordance with the requirements of this subpart, the table shall report the following information for the process wastewater stream as determined at the exit to the treatment process: Annual waste quantity, range of benzene concentrations, annual average flow-weighted benzene concentration, and annual benzene quantity.
- (5) If an owner or operator elects to comply with the alternative requirements of §61.342(e), then the report required by paragraph (d)(2) of this section shall include a table presenting the following information for each waste stream:
- (i) For each waste stream identified as not being controlled for benzene emissions in accordance with the requirements of this subpart; the table shall report the following information

for the waste stream as determined at the point of waste generation: annual waste quantity, range of benzene concentrations, annual average flowweighted benzene concentration, and annual benzene quantity;

- (ii) For each waste stream identified as being controlled for benzene emissions in accordance with the requirements of this subpart; the table shall report the following information for the waste stream as determined at the applicable location described in §61.355(k)(2): Annual waste quantity, range of benzene concentrations, annual average flow-weighted benzene concentration, and annual benzene quantity.
- (6) Beginning 3 months after the date that the equipment necessary to comply with these standards has been certified in accordance with paragraph (d)(1) of this section, the owner or operator shall submit quarterly to the Administrator a certification that all of the required inspections have been carried out in accordance with the requirements of this subpart.
- (7) Beginning 3 months after the date that the equipment necessary to comply with these standards has been certified in accordance with paragraph (d)(1) of this section, the owner or operator shall submit a report quarterly to the Administrator that includes:
- (i) If a treatment process or waste-water treatment system unit is monitored in accordance with §61.354(a)(1) of this subpart, then each period of operation during which the concentration of benzene in the monitored waste stream exiting the unit is equal to or greater than 10 ppmw.
- (ii) If a treatment process or waste-water treatment system unit is monitored in accordance with §61.354(a)(2) of this subpart, then each 3-hour period of operation during which the average value of the monitored parameter is outside the range of acceptable values or during which the unit is not operating as designed.
- (iii) If a treatment process or wastewater treatment system unit is monitored in accordance with §61.354(b), then each period of operation during which the flow-weighted annual average concentration of benzene in the monitored waste stream entering the

unit is equal to or greater than 10 ppmw and/or the total annual benzene quantity is equal to or greater than 1.0 mg/yr.

- (iv) For a control device monitored in accordance with §61.354(c) of this subpart, each period of operation monitored during which any of the following conditions occur, as applicable to the control device:
- (A) Each 3-hour period of operation during which the average temperature of the gas stream in the combustion zone of a thermal vapor incinerator, as measured by the temperature monitoring device, is more than 28 °C (50 °F) below the design combustion zone temperature.
- (B) Each 3-hour period of operation during which the average temperature of the gas stream immediately before the catalyst bed of a catalytic vapor incinerator, as measured by the temperature monitoring device, is more than 28 °C (50 °F) below the design gas stream temperature, and any 3-hour period during which the average temperature difference across the catalyst bed (i.e., the difference between the temperatures of the gas stream immediately before and after the catalyst bed), as measured by the temperature monitoring device, is less than 80 percent of the design temperature difference.
- (C) Each 3-hour period of operation during which the average temperature of the gas stream in the combustion zone of a boiler or process heater having a design heat input capacity less than 44 MW (150 × 106 BTU/hr), as mesured by the temperature monitoring device, is more than 28 °C (50 °F) below the design combustion zone temperature.
- (D) Each 3-hour period of operation during which the average concentration of organics or the average concentration of benzene in the exhaust gases from a carbon adsorber, condenser, or other vapor recovery system is more than 20 percent greater than the design concentration level of organics or benzene in the exhaust gas.
- (E) Each 3-hour period of operation during which the temperature of the condenser exhaust vent stream is more than 6 °C (11 °F) above the design average exhaust vent stream temperature,

or the temperature of the coolant fluid exiting the condenser is more than 6 °C (11 °F) above the design average coolant fluid temperature at the condenser outlet.

- (F) Each period in which the pilot flame of a flare is absent.
- (G) Each occurrence when there is a change in the location at which the vent stream is introduced into the flame zone of a boiler or process heater as required by §61.349(a)(2)(i)(C) of this subpart.
- (H) Each occurrence when the carbon in a carbon adsorber system that is regenerated directly on site in the control device is not regenerated at the predetermined carbon bed regeneration time.
- (I) Each occurrence when the carbon in a carbon adsorber system that is not regenerated directly on site in the control device is not replaced at the predetermined interval specified in §61.354(c) of this subpart.
- (J) Each 3-hour period of operation during which the parameters monitored are outside the range of values specified in §61.349(a)(2)(iv)(C), or any other periods specified by the Administrator for a control device subject to the requirements of §61.349(a)(2)(iv).
- (v) For a cover and closed-vent system monitored in accordance with §61.354(g), the owner or operator shall submit a report quarterly to the Administrator that identifies any period in which the pressure in the waste management unit is equal to or greater than atmospheric pressure.
- (8) Beginning one year after the date that the equipment necessary to comply with these standards has been certified in accordance with paragraph (d)(1) of this section, the owner or operator shall submit annually to the Ad-

ministrator a report that summarizes all inspections required by §§61.342 through 61.354 during which detectable emissions are measured or a problem (such as a broken seal, gap or other problem) that could result in benzone emissions is identified, including information about the repairs or corrective action taken.

- (e) An owner or operator electing to comply with the provisions of §§61.351 or 61.352 of this subpart shall notify the Administrator of the alternative standard selected in the report required under §61.07 or §61.10 of this part.
- (f) An owner or operator who elects to install and operate the control equipment in §61.351 of this subpart shall comply with the reporting requirements in 40 CFR 60.115b.
- (g) An owner or operator who elects to install and operate the control equipment in §61.352 of this subpart shall submit initial and quarterly reports that identify all seal gap measurements, as required in 40 CFR 60.693—2(a), that are outside the prescribed limits.

[55 FR 8346, Mar. 7 1990; 55 FR 12444, Apr. 3, 1990, as amended at 55 FR 37231, Sept. 10, 1990; 58 FR 3105, Jan. 7, 1993; 65 FR 62161, Oct. 17, 2000]

### § 61.358 Delegation of authority.

- (a) In delegating implementation and enforcement authority to a State under section 112(d) of the Clean Air Act, the authorities contained in paragraph (b) of this section shall be retained by the Administrator and not transferred to a State.
- (b) Alternative means of emission limitation under §61.353 of this subpart will not be delegated to States.

### § 61.359 [Reserved]

Appendix J
40 CFR Part 63, Subpart ZZZZ
National Emission Standard for Stationary Reciprocating Internal Combustion Engines

### § 63.6580

63.8802 What methods must I use to demonstrate compliance with the emission limitation for loop slitter adhesive use?

63.8806 How do I demonstrate initial compliance with the emission limitations?

CONTINUOUS COMPLIANCE REQUIREMENTS

63.8810 How do I monitor and collect data to demonstrate continuous compliance?

63.8812 How do I demonstrate continuous compliance with the emission limitations?

NOTIFICATIONS, REPORTS, AND RECORDS

63.8816 What notifications must I submit and when?

63.8818 What reports must I submit and when?

63.8820 What records must I keep?

63.8822 In what form and how long must I keep my records?

OTHER REQUIREMENTS AND INFORMATION

63.8826 What parts of the General Provisions apply to me?

63.8828 Who implements and enforces this subpart?

63.8830 What definitions apply to this subpart?

TABLE 1 TO SUBPART MMMMMM OF PART 63— EMISSION LIMITS

Table 2 to Subpart MMMMM of Part 63— Operating Limits for New or Reconstructed Flame Lamination Affected Sources

Table 3 to Subpart MMMMM of Part 63— Performance Test Requirements for New or Reconstructed Flame Lamination Affected Sources

Table 4 to Subpart MMMMM of Part 63— Initial Compliance With Emission Limits

TABLE 5 TO SUBPART MMMMM OF PART 63— CONTINUOUS COMPLIANCE WITH EMISSION LIMITS AND OPERATING LIMITS

TABLE 6 TO SUBPART MMMMM OF PART 63— REQUIREMENTS FOR REPORTS

TABLE 7 TO SUBPART MMMMM OF PART 63— APPLICABILITY OF GENERAL PROVISIONS TO SUBPART MMMMM

AUTHORITY: 42 U.S.C. 7401 et seq.

SOURCE: 57 FR 61992, Dec. 29, 1992, unless otherwise noted.

### Subpart ZZZZ—National Emissions Standards for Hazardous Air Pollutants for Stationary Reciprocating Internal Combustion Engines

SOURCE: 69 FR 33506, June 15, 2004, unless otherwise noted.

### 40 CFR Ch. I (7-1-13 Edition)

WHAT THIS SUBPART COVERS

# \$63.6580 What is the purpose of subpart ZZZZ?

Subpart ZZZZ establishes national emission limitations and operating limitations for hazardous air pollutants (HAP) emitted from stationary reciprocating internal combustion engines (RICE) located at major and area sources of HAP emissions. This subpart also establishes requirements to demonstrate initial and continuous compliance with the emission limitations and operating limitations.

[73 FR 3603, Jan. 18, 2008]

#### § 63.6585 Am I subject to this subpart?

You are subject to this subpart if you own or operate a stationary RICE at a major or area source of HAP emissions, except if the stationary RICE is being tested at a stationary RICE test cell/stand.

(a) A stationary RICE is any internal combustion engine which uses reciprocating motion to convert heat energy into mechanical work and which is not mobile. Stationary RICE differ from mobile RICE in that a stationary RICE is not a non-road engine as defined at 40 CFR 1068.30, and is not used to propel a motor vehicle or a vehicle used solely for competition.

(b) A major source of HAP emissions is a plant site that emits or has the potential to emit any single HAP at a rate of 10 tons (9.07 megagrams) or more per year or any combination of HAP at a rate of 25 tons (22.68 megagrams) or more per year, except that for oil and gas production facilities, a major source of HAP emissions is determined for each surface site.

(c) An area source of HAP emissions is a source that is not a major source.

(d) If you are an owner or operator of an area source subject to this subpart, your status as an entity subject to a standard or other requirements under this subpart does not subject you to the obligation to obtain a permit under 40 CFR part 70 or 71, provided you are not required to obtain a permit under 40 CFR 70.3(a) or 40 CFR 71.3(a) for a reason other than your status as an area source under this subpart. Notwithstanding the previous sentence,

you must continue to comply with the provisions of this subpart as applicable.

- (e) If you are an owner or operator of a stationary RICE used for national security purposes, you may be eligible to request an exemption from the requirements of this subpart as described in 40 CFR part 1068, subpart C.
- (f) The emergency stationary RICE listed in paragraphs (f)(1) through (3) of this section are not subject to this subpart. The stationary RICE must meet the definition of an emergency stationary RICE in §63.6675, which includes operating according to the provisions specified in §63.6640(f).
- (1) Existing residential emergency stationary RICE located at an area source of HAP emissions that do not operate or are not contractually obligated to be available for more than 15 hours per calendar year for the purposes specified in §63.6640(f)(2)(ii) and (iii) and that do not operate for the purpose specified in §63.6640(f)(4)(ii).
- (2) Existing commercial emergency stationary RICE located at an area source of HAP emissions that do not operate or are not contractually obligated to be available for more than 15 hours per calendar year for the purposes specified in §63.6640(f)(2)(ii) and (iii) and that do not operate for the purpose specified in §63.6640(f)(4)(ii).
- (3) Existing institutional emergency stationary RICE located at an area source of HAP emissions that do not operate or are not contractually obligated to be available for more than 15 hours per calendar year for the purposes specified in §63.6640(f)(2)(ii) and (iii) and that do not operate for the purpose specified in §63.6640(f)(4)(ii).

[69 FR 33506, June 15, 2004, as amended at 73 FR 3603, Jan. 18, 2008; 78 FR 6700, Jan. 30, 2013]

### § 63.6590 What parts of my plant does this subpart cover?

This subpart applies to each affected source.

- (a) Affected source. An affected source is any existing, new, or reconstructed stationary RICE located at a major or area source of HAP emissions, excluding stationary RICE being tested at a stationary RICE test cell/stand.
  - (1) Existing stationary RICE.

- (i) For stationary RICE with a site rating of more than 500 brake horse-power (HP) located at a major source of HAP emissions, a stationary RICE is existing if you commenced construction or reconstruction of the stationary RICE before December 19, 2002.
- (ii) For stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions, a stationary RICE is existing if you commenced construction or reconstruction of the stationary RICE before June 12, 2006.
- (iii) For stationary RICE located at an area source of HAP emissions, a stationary RICE is existing if you commenced construction or reconstruction of the stationary RICE before June 12, 2006.
- (iv) A change in ownership of an existing stationary RICE does not make that stationary RICE a new or reconstructed stationary RICE.
- (2) New stationary RICE. (i) A stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions is new if you commenced construction of the stationary RICE on or after December 19, 2002.
- (ii) A stationary RICE with a site rating of equal to or less than 500 brake HP located at a major source of HAP emissions is new if you commenced construction of the stationary RICE on or after June 12. 2006.
- (iii) A stationary RICE located at an area source of HAP emissions is new if you commenced construction of the stationary RICE on or after June 12, 2006.
- (3) Reconstructed stationary RICE. (i) A stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions is reconstructed if you meet the definition of reconstruction in §63.2 and reconstruction is commenced on or after December 19, 2002.
- (ii) A stationary RICE with a site rating of equal to or less than 500 brake HP located at a major source of HAP emissions is reconstructed if you meet the definition of reconstruction in §63.2 and reconstruction is commenced on or after June 12, 2006.

#### § 63.6590

- (iii) A stationary RICE located at an area source of HAP emissions is reconstructed if you meet the definition of reconstruction in §63.2 and reconstruction is commenced on or after June 12, 2006.
- (b) Stationary RICE subject to limited requirements. (1) An affected source which meets either of the criteria in paragraphs (b)(1)(i) through (ii) of this section does not have to meet the requirements of this subpart and of subpart A of this part except for the initial notification requirements of §63.6645(f).
- (i) The stationary RICE is a new or reconstructed emergency stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions that does not operate or is not contractually obligated to be available for more than 15 hours per calendar year for the purposes specified in §63.6640(f)(2)(ii) and (iii).
- (ii) The stationary RICE is a new or reconstructed limited use stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions.
- (2) A new or reconstructed stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions which combusts landfill or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis must meet the initial notification requirements of §63.6645(f) and the requirements of §863.6625(c), 63.6650(g), and 63.6655(c). These stationary RICE do not have to meet the emission limitations and operating limitations of this subpart.
- (3) The following stationary RICE do not have to meet the requirements of this subpart and of subpart A of this part, including initial notification requirements:
- (i) Existing spark ignition 2 stroke lean burn (2SLB) stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions;
- (ii) Existing spark ignition 4 stroke lean burn (4SLB) stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions:
- (iii) Existing emergency stationary RICE with a site rating of more than

- 500 brake HP located at a major source of HAP emissions that does not operate or is not contractually obligated to be available for more than 15 hours per calendar year for the purposes specified in §63.6640(f)(2)(ii) and (iii).
- (iv) Existing limited use stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions:
- (v) Existing stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions that combusts landfill gas or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis:
- (c) Stationary RICE subject to Regulations under 40 CFR Part 60. An affected source that meets any of the criteria in paragraphs (c)(1) through (7) of this section must meet the requirements of this part by meeting the requirements of 40 CFR part 60 subpart IIII, for compression ignition engines or 40 CFR part 60 subpart JJJJ, for spark ignition engines. No further requirements apply for such engines under this part.
- (1) A new or reconstructed stationary RICE located at an area source;
- (2) A new or reconstructed 2SLB stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions;
- (3) A new or reconstructed 4SLB stationary RICE with a site rating of less than 250 brake HP located at a major source of HAP emissions:
- (4) A new or reconstructed spark ignition 4 stroke rich burn (4SRB) stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions;
- (5) A new or reconstructed stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions which combusts landfill or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis;
- (6) A new or reconstructed emergency or limited use stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions:
- (7) A new or reconstructed compression ignition (CI) stationary RICE with a site rating of less than or equal to 500

brake HP located at a major source of HAP emissions.

[69 FR 33506, June 15, 2004, as amended at 73 FR 3604, Jan. 18, 2008; 75 FR 9674, Mar. 3, 2010; 75 FR 37733, June 30, 2010; 75 FR 51588, Aug. 20, 2010; 78 FR 6700, Jan. 30, 2013]

# § 63.6595 When do I have to comply with this subpart?

- (a) Affected sources. (1) If you have an existing stationary RICE, excluding existing non-emergency CI stationary RICE, with a site rating of more than 500 brake HP located at a major source of HAP emissions, you must comply with the applicable emission limitations, operating limitations and other requirements no later than June 15, 2007. If you have an existing non-emergency CI stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, an existing stationary CI RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions, or an existing stationary CI RICE located at an area source of HAP emissions, you must comply with the applicable emission limitations, operating limitations, and other requirements no later than May 3, 2013. If you have an existing stationary SI RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions, or an existing stationary SI RICE located at an area source of HAP emissions, you must comply with the applicable emission limitations, operating limitations, and other requirements no later than October 19, 2013.
- (2) If you start up your new or reconstructed stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions before August 16, 2004, you must comply with the applicable emission limitations and operating limitations in this subpart no later than August 16, 2004.
- (3) If you start up your new or reconstructed stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions after August 16, 2004, you must comply with the applicable emission limitations and operating limitations in this subpart upon startup of your affected source.

- (4) If you start up your new or reconstructed stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions before January 18, 2008, you must comply with the applicable emission limitations and operating limitations in this subpart no later than January 18, 2008.
- (5) If you start up your new or reconstructed stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions after January 18, 2008, you must comply with the applicable emission limitations and operating limitations in this subpart upon startup of your affected source.
- (6) If you start up your new or reconstructed stationary RICE located at an area source of HAP emissions before January 18, 2008, you must comply with the applicable emission limitations and operating limitations in this subpart no later than January 18, 2008.
- (7) If you start up your new or reconstructed stationary RICE located at an area source of HAP emissions after January 18, 2008, you must comply with the applicable emission limitations and operating limitations in this subpart upon startup of your affected source.
- (b) Area sources that become major sources. If you have an area source that increases its emissions or its potential to emit such that it becomes a major source of HAP, the compliance dates in paragraphs (b)(1) and (2) of this section apply to you.
- (1) Any stationary RICE for which construction or reconstruction is commenced after the date when your area source becomes a major source of HAP must be in compliance with this subpart upon startup of your affected source.
- (2) Any stationary RICE for which construction or reconstruction is commenced before your area source becomes a major source of HAP must be in compliance with the provisions of this subpart that are applicable to RICE located at major sources within 3 years after your area source becomes a major source of HAP.
- (c) If you own or operate an affected source, you must meet the applicable

notification requirements in §63.6645 and in 40 CFR part 63, subpart A.

[69 FR 33506, June 15, 2004, as amended at 73 FR 3604, Jan. 18, 2008; 75 FR 9675, Mar. 3, 2010; 75 FR 51589, Aug. 20, 2010; 78 FR 6701, Jan. 30, 2013]

EMISSION AND OPERATING LIMITATIONS

§ 63.6600 What emission limitations and operating limitations must I meet if I own or operate a stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions?

Compliance with the numerical emission limitations established in this subpart is based on the results of testing the average of three 1-hour runs using the testing requirements and procedures in §63.6620 and Table 4 to this subpart.

- (a) If you own or operate an existing, new, or reconstructed spark ignition 4SRB stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, you must comply with the emission limitations in Table 1a to this subpart and the operating limitations in Table 1b to this subpart which apply to you.
- (b) If you own or operate a new or reconstructed 2SLB stationary RICE with a site rating of more than 500 brake HP located at major source of HAP emissions, a new or reconstructed 4SLB stationary RICE with a site rating of more than 500 brake HP located at major source of HAP emissions, or a new or reconstructed CI stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, you must comply with the emission limitations in Table 2a to this subpart and the operating limitations in Table 2b to this subpart which apply to you.
- (c) If you own or operate any of the following stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, you do not need to comply with the emission limitations in Tables 1a, 2a, 2c, and 2d to this subpart or operating limitations in Tables 1b and 2b to this subpart: an existing 2SLB stationary RICE; an existing 4SLB stationary RICE; a stationary RICE that combusts landfill gas or digester gas equivalent to 10 percent or more of the

gross heat input on an annual basis; an emergency stationary RICE; or a limited use stationary RICE.

(d) If you own or operate an existing non-emergency stationary CI RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, you must comply with the emission limitations in Table 2c to this subpart and the operating limitations in Table 2b to this subpart which apply to you.

[73 FR 3605, Jan. 18, 2008, as amended at 75 FR 9675, Mar. 3, 2010]

§ 63.6601 What emission limitations must I meet if I own or operate a new or reconstructed 4SLB stationary RICE with a site rating of greater than or equal to 250 brake HP and less than or equal to 500 brake HP located at a major source of HAP emissions?

Compliance with the numerical emission limitations established in this subpart is based on the results of testing the average of three 1-hour runs using the testing requirements and procedures in §63.6620 and Table 4 to this subpart. If you own or operate a new or reconstructed 4SLB stationary RICE with a site rating of greater than or equal to 250 and less than or equal to 500 brake HP located at major source of HAP emissions manufactured on or after January 1, 2008, you must comply with the emission limitations in Table 2a to this subpart and the operating limitations in Table 2b to this subpart which apply to you.

[73 FR 3605, Jan. 18, 2008, as amended at 75 FR 9675, Mar. 3, 2010; 75 FR 51589, Aug. 20, 2010]

§ 63.6602 What emission limitations and other requirements must I meet if I own or operate an existing stationary RICE with a site rating of equal to or less than 500 brake HP located at a major source of HAP emissions?

If you own or operate an existing stationary RICE with a site rating of equal to or less than 500 brake HP located at a major source of HAP emissions, you must comply with the emission limitations and other requirements in Table 2c to this subpart which

apply to you. Compliance with the numerical emission limitations established in this subpart is based on the results of testing the average of three 1-hour runs using the testing requirements and procedures in §63.6620 and Table 4 to this subpart.

[78 FR 6701, Jan. 30, 2013]

§ 63.6603 What emission limitations, operating limitations, and other requirements must I meet if I own or operate an existing stationary RICE located at an area source of HAP emissions?

Compliance with the numerical emission limitations established in this subpart is based on the results of testing the average of three 1-hour runs using the testing requirements and procedures in §63.6620 and Table 4 to this subpart.

- (a) If you own or operate an existing stationary RICE located at an area source of HAP emissions, you must comply with the requirements in Table 2d to this subpart and the operating limitations in Table 2b to this subpart that apply to you.
- (b) If you own or operate an existing stationary non-emergency CI RICE with a site rating of more than 300 HP located at an area source of HAP that meets either paragraph (b)(1) or (2) of this section, you do not have to meet the numerical CO emission limitations specified in Table 2d of this subpart. Existing stationary non-emergency CI RICE with a site rating of more than 300 HP located at an area source of HAP that meet either paragraph (b)(1) or (2) of this section must meet the management practices that are shown for stationary non-emergency CI RICE with a site rating of less than or equal to 300 HP in Table 2d of this subpart.
- (1) The area source is located in an area of Alaska that is not accessible by the Federal Aid Highway System (FAHS).
- (2) The stationary RICE is located at an area source that meets paragraphs (b)(2)(i), (ii), and (iii) of this section.
- (i) The only connection to the FAHS is through the Alaska Marine Highway System (AMHS), or the stationary RICE operation is within an isolated grid in Alaska that is not connected to

the statewide electrical grid referred to as the Alaska Railbelt Grid.

- (ii) At least 10 percent of the power generated by the stationary RICE on an annual basis is used for residential purposes.
- (iii) The generating capacity of the area source is less than 12 megawatts, or the stationary RICE is used exclusively for backup power for renewable energy.
- (c) If you own or operate an existing stationary non-emergency CI RICE with a site rating of more than 300 HP located on an offshore vessel that is an area source of HAP and is a nonroad vehicle that is an Outer Continental Shelf (OCS) source as defined in 40 CFR 55.2, you do not have to meet the numerical CO emission limitations specified in Table 2d of this subpart. You must meet all of the following management practices:
- (1) Change oil every 1,000 hours of operation or annually, whichever comes first. Sources have the option to utilize an oil analysis program as described in §63.6625(i) in order to extend the specified oil change requirement.
- (2) Inspect and clean air filters every 750 hours of operation or annually, whichever comes first, and replace as necessary.
- (3) Inspect fuel filters and belts, if installed, every 750 hours of operation or annually, whichever comes first, and replace as necessary.
- (4) Inspect all flexible hoses every 1,000 hours of operation or annually, whichever comes first, and replace as necessary.
- (d) If you own or operate an existing non-emergency CI RICE with a site rating of more than 300 HP located at an area source of HAP emissions that is certified to the Tier 1 or Tier 2 emission standards in Table 1 of 40 CFR 89.112 and that is subject to an enforceable state or local standard that requires the engine to be replaced no later than June 1, 2018, you may until January 1, 2015, or 12 years after the installation date of the engine (whichever is later), but not later than June 1, 2018, choose to comply with the management practices that are shown for stationary non-emergency CI RICE with a site rating of less than or equal

to 300 HP in Table 2d of this subpart instead of the applicable emission limitations in Table 2d, operating limitations in Table 2b, and crankcase ventilation system requirements in §63.6625(g). You must comply with the emission limitations in Table 2d and operating limitations in Table 2b that apply for nonemergency CI RICE with a site rating of more than 300 HP located at an area source of HAP emissions by January 1, 2015, or 12 years after the installation date of the engine (whichever is later). but not later than June 1, 2018. You must also comply with the crankcase ventilation system requirements in §63.6625(g) by January 1, 2015, or 12 years after the installation date of the engine (whichever is later), but not later than June 1, 2018.

(e) If you own or operate an existing non-emergency CI RICE with a site rating of more than 300 HP located at an area source of HAP emissions that is certified to the Tier 3 (Tier 2 for engines above 560 kilowatt (kW)) emission standards in Table 1 of 40 CFR 89.112, you may comply with the requirements under this part by meeting the requirements for Tier 3 engines (Tier 2 for engines above 560 kW) in 40 CFR part 60 subpart IIII instead of the emission limitations and other requirements that would otherwise apply under this part for existing non-emergency CI RICE with a site rating of more than 300 HP located at an area source of HAP emissions.

(f) An existing non-emergency SI 4SLB and 4SRB stationary RICE with a site rating of more than 500 HP located at area sources of HAP must meet the definition of remote stationary RICE in §63.6675 on the initial compliance date for the engine, October 19, 2013, in order to be considered a remote stationary RICE under this subpart. Owners and operators of existing non-emergency SI 4SLB and 4SRB stationary RICE with a site rating of more than 500 HP located at area sources of HAP that meet the definition of remote stationary RICE in §63.6675 of this subpart as of October 19, 2013 must evaluate the status of their stationary RICE every 12 months. Owners and operators must keep records of the initial and annual evaluation of the status of the engine. If the evaluation indicates that the

stationary RICE no longer meets the definition of remote stationary RICE in §63.6675 of this subpart, the owner or operator must comply with all of the requirements for existing non-emergency SI 4SLB and 4SRB stationary RICE with a site rating of more than 500 HP located at area sources of HAP that are not remote stationary RICE within 1 year of the evaluation.

[75 FR 9675, Mar. 3, 2010, as amended at 75 FR 51589, Aug. 20, 2010; 76 FR 12866, Mar. 9, 2011; 78 FR 6701, Jan. 30, 2013]

# § 63.6604 What fuel requirements must I meet if I own or operate a stationary CI RICE?

(a) If you own or operate an existing non-emergency, non-black start CI stationary RICE with a site rating of more than 300 brake HP with a displacement of less than 30 liters per cylinder that uses diesel fuel, you must use diesel fuel that meets the requirements in 40 CFR 80.510(b) for nonroad diesel fuel.

(b) Beginning January 1, 2015, if you own or operate an existing emergency CI stationary RICE with a site rating of more than 100 brake HP and a displacement of less than 30 liters per cvlinder that uses diesel fuel and operates or is contractually obligated to be available for more than 15 hours per calendar year for the purposes specified in §63.6640(f)(2)(ii) and (iii) or that operates for the purpose specified in §63.6640(f)(4)(ii), you must use diesel fuel that meets the requirements in 40 CFR 80.510(b) for nonroad diesel fuel, except that any existing diesel fuel purchased (or otherwise obtained) prior to January 1, 2015, may be used until depleted.

(c) Beginning January 1, 2015, if you own or operate a new emergency CI stationary RICE with a site rating of more than 500 brake HP and a displacement of less than 30 liters per cylinder located at a major source of HAP that uses diesel fuel and operates or is contractually obligated to be available for more than 15 hours per calendar year purposes for the specified §63.6640(f)(2)(ii) and (iii), you must use diesel fuel that meets the requirements in 40 CFR 80 510(b) for nonroad diesel fuel, except that any existing diesel fuel purchased (or otherwise obtained) prior to January 1, 2015, may be used until depleted.

(d) Existing CI stationary RICE located in Guam, American Samoa, the Commonwealth of the Northern Mariana Islands, at area sources in areas of Alaska that meet either \$63.6603(b)(1) or \$63.6603(b)(2), or are on offshore vessels that meet \$63.6603(c) are exempt from the requirements of this section.

[78 FR 6702, Jan. 30, 2013]

GENERAL COMPLIANCE REQUIREMENTS

# § 63.6605 What are my general requirements for complying with this subpart?

- (a) You must be in compliance with the emission limitations, operating limitations, and other requirements in this subpart that apply to you at all times.
- (b) At all times you must operate and maintain any affected source, including associated air pollution control equipment and monitoring equipment, in a manner consistent with safety and good air pollution control practices for minimizing emissions. The general duty to minimize emissions does not require you to make any further efforts to reduce emissions if levels required by this standard have been achieved. Determination of whether such operation and maintenance procedures are being used will be based on information available to the Administrator which may include, but is not limited to, monitoring results, review of operation and maintenance procedures, review of operation and maintenance records, and inspection of the source.

[75 FR 9675, Mar. 3, 2010, as amended at 78 FR 6702, Jan. 30, 2013]

TESTING AND INITIAL COMPLIANCE REQUIREMENTS

§ 63.6610 By what date must I conduct the initial performance tests or other initial compliance demonstrations if I own or operate a stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions?

If you own or operate a stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions you are subject to the requirements of this section.

- (a) You must conduct the initial performance test or other initial compliance demonstrations in Table 4 to this subpart that apply to you within 180 days after the compliance date that is specified for your stationary RICE in §63.6595 and according to the provisions in §63.7(a)(2).
- (b) If you commenced construction or reconstruction between December 19, 2002 and June 15, 2004 and own or operate stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, you must demonstrate initial compliance with either the proposed emission limitations or the promulgated emission limitations no later than February 10, 2005 or no later than 180 days after startup of the source, whichever is later, according to \$63.7(a)(2)(ix).
- (c) If you commenced construction or reconstruction between December 19, 2002 and June 15, 2004 and own or operate stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, and you chose to comply with the proposed emission limitations when demonstrating initial compliance, you must conduct a second performance test to demonstrate compliance with the promulgated emission limitations by December 13, 2007 or after startup of the source, whichever is later, according to §63.7(a)(2)(ix).
- (d) An owner or operator is not required to conduct an initial performance test on units for which a performance test has been previously conducted, but the test must meet all of the conditions described in paragraphs (d)(1) through (5) of this section.
- (1) The test must have been conducted using the same methods specified in this subpart, and these methods must have been followed correctly.
- (2) The test must not be older than 2 years.
- (3) The test must be reviewed and accepted by the Administrator.
- (4) Either no process or equipment changes must have been made since the test was performed, or the owner or operator must be able to demonstrate that the results of the performance test, with or without adjustments, reliably demonstrate compliance despite process or equipment changes.

§63.6611

(5) The test must be conducted at any load condition within plus or minus 10 percent of 100 percent load.

[69 FR 33506, June 15, 2004, as amended at 73 FR 3605, Jan. 18, 2008]

§ 63.6611 By what date must I conduct the initial performance tests or other initial compliance demonstrations if I own or operate a new or reconstructed 4SLB SI stationary RICE with a site rating of greater than or equal to 250 and less than or equal to 500 brake HP located at a major source of HAP emissions?

If you own or operate a new or reconstructed 4SLB stationary RICE with a site rating of greater than or equal to 250 and less than or equal to 500 brake HP located at a major source of HAP emissions, you must conduct an initial performance test within 240 days after the compliance date that is specified for your stationary RICE in §63.6595 and according to the provisions specified in Table 4 to this subpart, as appropriate.

[73 FR 3605, Jan. 18, 2008, as amended at 75 FR 51589, Aug. 20, 2010]

§ 63.6612 By what date must I conduct the initial performance tests or other initial compliance demonstrations if I own or operate an existing stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions or an existing stationary RICE located at an area source of HAP emissions?

If you own or operate an existing stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions or an existing stationary RICE located at an area source of HAP emissions you are subject to the requirements of this section.

- (a) You must conduct any initial performance test or other initial compliance demonstration according to Tables 4 and 5 to this subpart that apply to you within 180 days after the compliance date that is specified for your stationary RICE in §63.6595 and according to the provisions in §63.7(a)(2).
- (b) An owner or operator is not required to conduct an initial performance test on a unit for which a performance test has been previously con-

ducted, but the test must meet all of the conditions described in paragraphs (b)(1) through (4) of this section.

- (1) The test must have been conducted using the same methods specified in this subpart, and these methods must have been followed correctly.
- (2) The test must not be older than 2 years.
- (3) The test must be reviewed and accepted by the Administrator.
- (4) Either no process or equipment changes must have been made since the test was performed, or the owner or operator must be able to demonstrate that the results of the performance test, with or without adjustments, reliably demonstrate compliance despite process or equipment changes.

[75 FR 9676, Mar. 3, 2010, as amended at 75 FR 51589, Aug. 20, 2010]

# § 63.6615 When must I conduct subsequent performance tests?

If you must comply with the emission limitations and operating limitations, you must conduct subsequent performance tests as specified in Table 3 of this subpart.

## § 63.6620 What performance tests and other procedures must I use?

- (a) You must conduct each performance test in Tables 3 and 4 of this subpart that applies to you.
- (b) Each performance test must be conducted according to the requirements that this subpart specifies in Table 4 to this subpart. If you own or operate a non-operational stationary RICE that is subject to performance testing, you do not need to start up the engine solely to conduct the performance test. Owners and operators of a non-operational engine can conduct the performance test when the engine is started up again. The test must be conducted at any load condition within plus or minus 10 percent of 100 percent load for the stationary RICE listed in paragraphs (b)(1) through (4) of this section.
- (1) Non-emergency 4SRB stationary RICE with a site rating of greater than 500 brake HP located at a major source of HAP emissions.
- (2) New non-emergency 4SLB stationary RICE with a site rating of greater than or equal to 250 brake HP

located at a major source of HAP emissions.

(3) New non-emergency 2SLB stationary RICE with a site rating of greater than 500 brake HP located at a major source of HAP emissions.

(4) New non-emergency CI stationary RICE with a site rating of greater than 500 brake HP located at a major source of HAP emissions.

(c) [Reserved]

(d) You must conduct three separate test runs for each performance test required in this section, as specified in §63.7(e)(3). Each test run must last at least 1 hour, unless otherwise specified in this subpart.

§ 63.6620

(e)(1) You must use Equation 1 of this section to determine compliance with the percent reduction requirement:

$$\frac{C_{i} - C_{O}}{C_{i}} \times 100 = R \quad (Eq. 1)$$

Where:

C<sub>i</sub> = concentration of carbon monoxide (CO), total hydrocarbons (THC), or formaldehyde at the control device inlet,

C<sub>o</sub> = concentration of CO, THC, or formaldehyde at the control device outlet, and

R = percent reduction of CO, THC, or formaldehyde emissions.

(2) You must normalize the CO, THC, or formaldehyde concentrations at the inlet and outlet of the control device to a dry basis and to 15 percent oxygen, or an equivalent percent carbon dioxide

 $(\mathrm{CO}_2)$ . If pollutant concentrations are to be corrected to 15 percent oxygen and  $\mathrm{CO}_2$  concentration is measured in lieu of oxygen concentration measurement, a  $\mathrm{CO}_2$  correction factor is needed. Calculate the  $\mathrm{CO}_2$  correction factor as described in paragraphs (e)(2)(i) through (iii) of this section.

(i) Calculate the fuel-specific Fovalue for the fuel burned during the test using values obtained from Method 19, Section 5.2, and the following equation:

$$F_O = \frac{0.209 \ F_d}{F_C}$$
 (Eq. 2)

Where:

 $F_o$  = Fuel factor based on the ratio of oxygen volume to the ultimate  $CO_2$  volume produced by the fuel at zero percent excess air.

0.209 = Fraction of air that is oxygen, percent/100. 
$$\begin{split} \mathbf{F}_{\rm d} &= \mathbf{Ratio} \ \text{of the volume of dry effluent gas} \\ &\quad \text{to the gross calorific value of the fuel} \\ &\quad \text{from Method 19, dsm}^3 \! \mathbf{J} \ (dscf/10^6 \ Btu). \end{split}$$

 $F_{\rm c}$  = Ratio of the volume of CO<sub>2</sub> produced to the gross calorific value of the fuel from Method 19, dsm<sup>3</sup>/J (dscf/10<sup>6</sup> Btu)

(ii) Calculate the  $CO_2$  correction factor for correcting measurement data to 15 percent  $O_2$ , as follows:

$$X_{CO2} = \frac{5.9}{F_O}$$
 (Eq. 3)

Where:

 $X_{CO2} = CO_2$  correction factor, percent. 5.9 = 20.9 percent  $O_2$ —15 percent  $O_2$ , the defined  $O_2$  correction value, percent. (iii) Calculate the CO, THC, and formaldehyde gas concentrations adjusted to 15 percent  $O_2$  using  $CO_2$  as follows:

$$C_{adj} = C_d \frac{X_{CO2}}{%CO_2}$$
 (Eq. 4)

Where

 $C_{adj}$  = Calculated concentration of CO, THC, or formaldehyde adjusted to 15 percent  $O_2$ 

 $C_d$  = Measured concentration of CO, THC, or formaldehyde, uncorrected.

X<sub>CO2</sub> = CO<sub>2</sub> correction factor, percent. %CO<sub>2</sub> = Measured CO<sub>2</sub> concentration meas-

%CO<sub>2</sub> = Measured CO<sub>2</sub> concentration measured, dry basis, percent.

- (f) If you comply with the emission limitation to reduce CO and you are not using an oxidation catalyst, if you comply with the emission limitation to reduce formaldehyde and you are not using NSCR, or if you comply with the emission limitation to limit the concentration of formaldehyde in the stationary RICE exhaust and you are not using an oxidation catalyst or NSCR. you must petition the Administrator for operating limitations to be established during the initial performance test and continuously monitored thereafter; or for approval of no operating limitations. You must not conduct the initial performance test until after the petition has been approved by the Administrator.
- (g) If you petition the Administrator for approval of operating limitations, your petition must include the information described in paragraphs (g)(1) through (5) of this section.
- (1) Identification of the specific parameters you propose to use as operating limitations:
- (2) A discussion of the relationship between these parameters and HAP emissions, identifying how HAP emissions change with changes in these parameters, and how limitations on these parameters will serve to limit HAP emissions;
- (3) A discussion of how you will establish the upper and/or lower values for these parameters which will establish the limits on these parameters in the operating limitations;
- (4) A discussion identifying the methods you will use to measure and the instruments you will use to monitor these parameters, as well as the relative accuracy and precision of these methods and instruments; and

- (5) A discussion identifying the frequency and methods for recalibrating the instruments you will use for monitoring these parameters.
- (h) If you petition the Administrator for approval of no operating limitations, your petition must include the information described in paragraphs (h)(1) through (7) of this section.
- (1) Identification of the parameters associated with operation of the stationary RICE and any emission control device which could change intentionally (e.g., operator adjustment, automatic controller adjustment, etc.) or unintentionally (e.g., wear and tear, error, etc.) on a routine basis or over time:
- (2) A discussion of the relationship, if any, between changes in the parameters and changes in HAP emissions:
- (3) For the parameters which could change in such a way as to increase HAP emissions, a discussion of whether establishing limitations on the parameters would serve to limit HAP emissions:
- (4) For the parameters which could change in such a way as to increase HAP emissions, a discussion of how you could establish upper and/or lower values for the parameters which would establish limits on the parameters in operating limitations;
- (5) For the parameters, a discussion identifying the methods you could use to measure them and the instruments you could use to monitor them, as well as the relative accuracy and precision of the methods and instruments:
- (6) For the parameters, a discussion identifying the frequency and methods for recalibrating the instruments you could use to monitor them; and
- (7) A discussion of why, from your point of view, it is infeasible or unreasonable to adopt the parameters as operating limitations.
- (i) The engine percent load during a performance test must be determined by documenting the calculations, assumptions, and measurement devices

used to measure or estimate the percent load in a specific application. A written report of the average percent load determination must be included in the notification of compliance status. The following information must be included in the written report: the engine model number, the engine manufacturer, the year of purchase, the manufacturer's site-rated brake horsepower, the ambient temperature, pressure, and humidity during the performance test. and all assumptions that were made to estimate or calculate percent load during the performance test must be clearly explained. If measurement devices such as flow meters, kilowatt meters, beta analyzers, stain gauges, etc. are used, the model number of the measurement device, and an estimate of its accurate in percentage of true value must be provided.

[69 FR 33506, June 15, 2004, as amended at 75 FR 9676, Mar. 3, 2010; 78 FR 6702, Jan. 30, 2013]

# § 63.6625 What are my monitoring, installation, collection, operation, and maintenance requirements?

- (a) If you elect to install a CEMS as specified in Table 5 of this subpart, you must install, operate, and maintain a CEMS to monitor CO and either O<sub>2</sub> or CO<sub>2</sub> according to the requirements in paragraphs (a)(1) through (4) of this section. If you are meeting a requirement to reduce CO emissions, the CEMS must be installed at both the inlet and outlet of the control device. If you are meeting a requirement to limit the concentration of CO, the CEMS must be installed at the outlet of the control device.
- (1) Each CEMS must be installed, operated, and maintained according to the applicable performance specifications of 40 CFR part 60, appendix B.
- (2) You must conduct an initial performance evaluation and an annual relative accuracy test audit (RATA) of each CEMS according to the requirements in §63.8 and according to the applicable performance specifications of 40 CFR part 60, appendix B as well as daily and periodic data quality checks in accordance with 40 CFR part 60, appendix F, procedure 1.
- (3) As specified in §63.8(c)(4)(ii), each CEMS must complete a minimum of one cycle of operation (sampling, ana-

- lyzing, and data recording) for each successive 15-minute period. You must have at least two data points, with each representing a different 15-minute period, to have a valid hour of data.
- (4) The CEMS data must be reduced as specified in  $\S63.8(g)(2)$  and recorded in parts per million or parts per billion (as appropriate for the applicable limitation) at 15 percent oxygen or the equivalent  $CO_2$  concentration.
- (b) If you are required to install a continuous parameter monitoring system (CPMS) as specified in Table 5 of this subpart, you must install, operate, and maintain each CPMS according to the requirements in paragraphs (b)(1) through (6) of this section. For an affected source that is complying with the emission limitations and operating limitations on March 9, 2011, the requirements in paragraph (b) of this section are applicable September 6, 2011.
- (1) You must prepare a site-specific monitoring plan that addresses the monitoring system design, data collection, and the quality assurance and quality control elements outlined in paragraphs (b)(1)(i) through (v) of this section and in §63.8(d). As specified in §63.8(f)(4), you may request approval of monitoring system quality assurance and quality control procedures alternative to those specified in paragraphs (b)(1) through (5) of this section in your site-specific monitoring plan.
- (i) The performance criteria and design specifications for the monitoring system equipment, including the sample interface, detector signal analyzer, and data acquisition and calculations;
- (ii) Sampling interface (e.g., thermocouple) location such that the monitoring system will provide representative measurements;
- (iii) Equipment performance evaluations, system accuracy audits, or other audit procedures:
- (iv) Ongoing operation and maintenance procedures in accordance with provisions in 63.8(c)(1)(ii) and (c)(3); and
- (v) Ongoing reporting and recordkeeping procedures in accordance with provisions in §63.10(c), (e)(1), and (e)(2)(i).

### § 63.6625

- (2) You must install, operate, and maintain each CPMS in continuous operation according to the procedures in your site-specific monitoring plan.
- (3) The CPMS must collect data at least once every 15 minutes (see also §63.6635).
- (4) For a CPMS for measuring temperature range, the temperature sensor must have a minimum tolerance of 2.8 degrees Celsius (5 degrees Fahrenheit) or 1 percent of the measurement range, whichever is larger.
- (5) You must conduct the CPMS equipment performance evaluation, system accuracy audits, or other audit procedures specified in your site-specific monitoring plan at least annually.
- (6) You must conduct a performance evaluation of each CPMS in accordance with your site-specific monitoring plan.
- (c) If you are operating a new or reconstructed stationary RICE which fires landfill gas or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis, you must monitor and record your fuel usage daily with separate fuel meters to measure the volumetric flow rate of each fuel. In addition, you must operate your stationary RICE in a manner which reasonably minimizes HAP emissions.
- (d) If you are operating a new or reconstructed emergency 4SLB stationary RICE with a site rating of greater than or equal to 250 and less than or equal to 500 brake HP located at a major source of HAP emissions, you must install a non-resettable hour meter prior to the startup of the engine.
- (e) If you own or operate any of the following stationary RICE, you must operate and maintain the stationary RICE and after-treatment control device (if any) according to the manufacturer's emission-related written instructions or develop your own maintenance plan which must provide to the extent practicable for the maintenance and operation of the engine in a manner consistent with good air pollution control practice for minimizing emissions:
- (1) An existing stationary RICE with a site rating of less than 100 HP located at a major source of HAP emissions;

- (2) An existing emergency or black start stationary RICE with a site rating of less than or equal to 500 HP located at a major source of HAP emissions:
- (3) An existing emergency or black start stationary RICE located at an area source of HAP emissions;
- (4) An existing non-emergency, nonblack start stationary CI RICE with a site rating less than or equal to 300 HP located at an area source of HAP emissions:
- (5) An existing non-emergency, nonblack start 2SLB stationary RICE located at an area source of HAP emissions:
- (6) An existing non-emergency, nonblack start stationary RICE located at an area source of HAP emissions which combusts landfill or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis.
- (7) An existing non-emergency, nonblack start 4SLB stationary RICE with a site rating less than or equal to 500 HP located at an area source of HAP emissions:
- (8) An existing non-emergency, nonblack start 4SRB stationary RICE with a site rating less than or equal to 500 HP located at an area source of HAP emissions:
- (9) An existing, non-emergency, nonblack start 4SLB stationary RICE with a site rating greater than 500 HP located at an area source of HAP emissions that is operated 24 hours or less per calendar year; and
- (10) An existing, non-emergency, non-black start 4SRB stationary RICE with a site rating greater than 500 HP located at an area source of HAP emissions that is operated 24 hours or less per calendar year.
- (f) If you own or operate an existing emergency stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions or an existing emergency stationary RICE located at an area source of HAP emissions, you must install a non-resettable hour meter if one is not already installed.
- (g) If you own or operate an existing non-emergency, non-black start CI engine greater than or equal to 300 HP that is not equipped with a closed crankcase ventilation system, you

must comply with either paragraph (g)(1) or paragraph (2) of this section. Owners and operators must follow the manufacturer's specified maintenance requirements for operating and maintaining the open or closed crankcase ventilation systems and replacing the crankcase filters, or can request the Administrator to approve different maintenance requirements that are as protective as manufacturer requirements. Existing CI engines located at area sources in areas of Alaska that meet either §63.6603(b)(1) §63.6603(b)(2) do not have to meet the requirements of this paragraph (g). Existing CI engines located on offshore vessels that meet §63.6603(c) do not have to meet the requirements of this paragraph (g).

- (1) Install a closed crankcase ventilation system that prevents crankcase emissions from being emitted to the atmosphere, or
- (2) Install an open crankcase filtration emission control system that reduces emissions from the crankcase by filtering the exhaust stream to remove oil mist, particulates and metals.
- (h) If you operate a new, reconstructed, or existing stationary engine, you must minimize the engine's time spent at idle during startup and minimize the engine's startup time to a period needed for appropriate and safe loading of the engine, not to exceed 30 minutes, after which time the emission standards applicable to all times other than startup in Tables 1a, 2a, 2c, and 2d to this subpart apply.
- (i) If you own or operate a stationary CI engine that is subject to the work, operation or management practices in items 1 or 2 of Table 2c to this subpart or in items 1 or 4 of Table 2d to this subpart, you have the option of utilizing an oil analysis program in order to extend the specified oil change requirement in Tables 2c and 2d to this subpart. The oil analysis must be performed at the same frequency specified for changing the oil in Table 2c or 2d to this subpart. The analysis program must at a minimum analyze the following three parameters: Total Base Number, viscosity, and percent water content. The condemning limits for these parameters are as follows: Total Base Number is less than 30 percent of

the Total Base Number of the oil when new; viscosity of the oil has changed by more than 20 percent from the viscosity of the oil when new; or percent water content (by volume) is greater than 0.5. If all of these condemning limits are not exceeded, the engine owner or operator is not required to change the oil. If any of the limits are exceeded, the engine owner or operator must change the oil within 2 business days of receiving the results of the analysis; if the engine is not in operation when the results of the analysis are received, the engine owner or operator must change the oil within 2 business days or before commencing operation, whichever is later. The owner or operator must keep records of the parameters that are analyzed as part of the program, the results of the analysis, and the oil changes for the engine. The analysis program must be part of the maintenance plan for the engine.

(i) If you own or operate a stationary SI engine that is subject to the work, operation or management practices in items 6, 7, or 8 of Table 2c to this subpart or in items 5, 6, 7, 9, or 11 of Table 2d to this subpart, you have the option of utilizing an oil analysis program in order to extend the specified oil change requirement in Tables 2c and 2d to this subpart. The oil analysis must be performed at the same frequency specified for changing the oil in Table 2c or 2d to this subpart. The analysis program must at a minimum analyze the following three parameters: Total Acid Number, viscosity, and percent water content. The condemning limits for these parameters are as follows: Total Acid Number increases by more than 3.0 milligrams of potassium hydroxide (KOH) per gram from Total Acid Number of the oil when new; viscosity of the oil has changed by more than 20 percent from the viscosity of the oil when new; or percent water content (by volume) is greater than 0.5. If all of these condemning limits are not exceeded, the engine owner or operator is not required to change the oil. If any of the limits are exceeded, the engine owner or operator must change the oil within 2 business days of receiving the results of the analysis; if the engine is not in operation when the results of the analysis are received, the engine

owner or operator must change the oil within 2 business days or before commencing operation, whichever is later. The owner or operator must keep records of the parameters that are analyzed as part of the program, the results of the analysis, and the oil changes for the engine. The analysis program must be part of the maintenance plan for the engine.

[69 FR 33506, June 15, 2004, as amended at 73 FR 3606, Jan. 18, 2008; 75 FR 9676, Mar. 3, 2010; 75 FR 51589, Aug. 20, 2010; 76 FR 12866, Mar. 9, 2011; 78 FR 6703, Jan. 30, 20131

# § 63.6630 How do I demonstrate initial compliance with the emission limitations, operating limitations, and other requirements?

- (a) You must demonstrate initial compliance with each emission limitation, operating limitation, and other requirement that applies to you according to Table 5 of this subpart.
- (b) During the initial performance test, you must establish each operating limitation in Tables 1b and 2b of this subpart that applies to you.
- (c) You must submit the Notification of Compliance Status containing the results of the initial compliance demonstration according to the requirements in §63.6645.
- (d) Non-emergency 4SRB stationary RICE complying with the requirement to reduce formaldehyde emissions by 76 percent or more can demonstrate initial compliance with the formaldehyde emission limit by testing for THC instead of formaldehyde. The testing must be conducted according to the requirements in Table 4 of this subpart. The average reduction of emissions of THC determined from the performance test must be equal to or greater than 30 percent.
- (e) The initial compliance demonstration required for existing nonemergency 4SLB and 4SRB stationary RICE with a site rating of more than 500 HP located at an area source of HAP that are not remote stationary RICE and that are operated more than 24 hours per calendar year must be conducted according to the following reouirements:
- (1) The compliance demonstration must consist of at least three test runs.

- (2) Each test run must be of at least 15 minute duration, except that each test conducted using the method in appendix A to this subpart must consist of at least one measurement cycle and include at least 2 minutes of test data phase measurement.
- (3) If you are demonstrating compliance with the CO concentration or CO percent reduction requirement, you must measure CO emissions using one of the CO measurement methods specified in Table 4 of this subpart, or using appendix A to this subpart.
- (4) If you are demonstrating compliance with the THC percent reduction requirement, you must measure THC emissions using Method 25A, reported as propane, of 40 CFR part 60, appendix A
- (5) You must measure  $O_2$  using one of the  $O_2$  measurement methods specified in Table 4 of this subpart. Measurements to determine  $O_2$  concentration must be made at the same time as the measurements for CO or THC concentration.
- (6) If you are demonstrating compliance with the CO or THC percent reduction requirement, you must measure CO or THC emissions and  $O_2$  emissions simultaneously at the inlet and outlet of the control device.

[69 FR 33506, June 15, 2004, as amended at 78 FR 6704, Jan. 30, 2013]

CONTINUOUS COMPLIANCE REQUIREMENTS

# § 63.6635 How do I monitor and collect data to demonstrate continuous compliance?

- (a) If you must comply with emission and operating limitations, you must monitor and collect data according to this section.
- (b) Except for monitor malfunctions, associated repairs, required performance evaluations, and required quality assurance or control activities, you must monitor continuously at all times that the stationary RICE is operating. A monitoring malfunction is any sudden, infrequent, not reasonably preventable failure of the monitoring to provide valid data. Monitoring failures that are caused in part by poor maintenance or careless operation are not malfunctions.

(c) You may not use data recorded during monitoring malfunctions, associated repairs, and required quality assurance or control activities in data averages and calculations used to report emission or operating levels. You must, however, use all the valid data collected during all other periods.

[69 FR 33506, June 15, 2004, as amended at 76 FR 12867, Mar. 9, 2011]

# § 63.6640 How do I demonstrate continuous compliance with the emission limitations, operating limitations, and other requirements?

- (a) You must demonstrate continuous compliance with each emission limitation, operating limitation, and other requirements in Tables 1a and 1b, Tables 2a and 2b, Table 2c, and Table 2d to this subpart that apply to you according to methods specified in Table 6 to this subpart.
- (b) You must report each instance in which you did not meet each emission limitation or operating limitation in Tables 1a and 1b, Tables 2a and 2b, Table 2c, and Table 2d to this subpart that apply to you. These instances are deviations from the emission and operating limitations in this subpart. These deviations must be reported according to the requirements in §63.6650. If you change your catalyst, you must reestablish the values of the operating parameters measured during the initial performance test. When you reestablish the values of your operating parameters, you must also conduct a performance test to demonstrate that you are meeting the required emission limitation applicable to your stationary RICE.
- (c) The annual compliance demonstration required for existing non-emergency 4SLB and 4SRB stationary RICE with a site rating of more than 500 HP located at an area source of HAP that are not remote stationary RICE and that are operated more than 24 hours per calendar year must be conducted according to the following requirements:
- (1) The compliance demonstration must consist of at least one test run.
- (2) Each test run must be of at least 15 minute duration, except that each test conducted using the method in appendix A to this subpart must consist

of at least one measurement cycle and include at least 2 minutes of test data phase measurement.

- (3) If you are demonstrating compliance with the CO concentration or CO percent reduction requirement, you must measure CO emissions using one of the CO measurement methods specified in Table 4 of this subpart, or using appendix A to this subpart.
- (4) If you are demonstrating compliance with the THC percent reduction requirement, you must measure THC emissions using Method 25A, reported as propane, of 40 CFR part 60, appendix A.
- (5) You must measure  $O_2$  using one of the  $O_2$  measurement methods specified in Table 4 of this subpart. Measurements to determine  $O_2$  concentration must be made at the same time as the measurements for CO or THC concentration.
- (6) If you are demonstrating compliance with the CO or THC percent reduction requirement, you must measure CO or THC emissions and  $O_2$  emissions simultaneously at the inlet and outlet of the control device.
- (7) If the results of the annual compliance demonstration show that the emissions exceed the levels specified in Table 6 of this subpart, the stationary RICE must be shut down as soon as safely possible, and appropriate corrective action must be taken (e.g., repairs, catalyst cleaning, catalyst replacement). The stationary RICE must be retested within 7 days of being restarted and the emissions must meet the levels specified in Table 6 of this subpart. If the retest shows that the emissions continue to exceed the specified levels, the stationary RICE must again be shut down as soon as safely possible, and the stationary RICE may not operate, except for purposes of startup and testing, until the owner/operator demonstrates through testing that the emissions do not exceed the levels specified in Table 6 of this subpart.
- (d) For new, reconstructed, and rebuilt stationary RICE, deviations from the emission or operating limitations that occur during the first 200 hours of operation from engine startup (engine

### § 63.6640

burn-in period) are not violations. Rebuilt stationary RICE means a stationary RICE that has been rebuilt as that term is defined in 40 CFR 94.11(a).

(e) You must also report each instance in which you did not meet the requirements in Table 8 to this subpart that apply to you. If you own or operate a new or reconstructed stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions (except new or reconstructed 4SLB engines greater than or equal to 250 and less than or equal to 500 brake HP), a new or reconstructed stationary RICE located at an area source of HAP emissions, or any of the following RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, you do not need to comply with the requirements in Table 8 to this subpart: An existing 2SLB stationary RICE, an existing 4SLB stationary RICE, an existing emergency stationary RICE, an existing limited use stationary RICE, or an existing stationary RICE which fires landfill gas or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis. If you own or operate any of the following RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, you do not need to comply with the requirements in Table 8 to this subpart, except for the initial notification requirements: a new or reconstructed stationary RICE that combusts landfill gas or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis, a new or reconstructed emergency stationary RICE, or a new or reconstructed limited use stationary

(f) If you own or operate an emergency stationary RICE, you must operate the emergency stationary RICE according to the requirements in paragraphs (f)(1) through (4) of this section. In order for the engine to be considered an emergency stationary RICE under this subpart, any operation other than emergency operation, maintenance and testing, emergency demand response, and operation in non-emergency situations for 50 hours per year, as described in paragraphs (f)(1) through (4) of this section, is prohibited. If you do not op-

erate the engine according to the requirements in paragraphs (f)(1) through (4) of this section, the engine will not be considered an emergency engine under this subpart and must meet all requirements for non-emergency engines.

- (1) There is no time limit on the use of emergency stationary RICE in emergency situations.
- (2) You may operate your emergency stationary RICE for any combination of the purposes specified in paragraphs (f)(2)(1) through (iii) of this section for a maximum of 100 hours per calendar year. Any operation for non-emergency situations as allowed by paragraphs (f)(3) and (4) of this section counts as part of the 100 hours per calendar year allowed by this paragraph (f)(2).
- (i) Emergency stationary RICE may be operated for maintenance checks and readiness testing, provided that the tests are recommended by federal, state or local government, the manufacturer, the vendor, the regional transmission organization or equivalent balancing authority and transmission operator, or the insurance company associated with the engine. The owner or operator may petition the Administrator for approval of additional hours to be used for maintenance checks and readiness testing, but a petition is not required if the owner or operator maintains records indicating that federal, state, or local standards require maintenance and testing of emergency RICE beyond 100 hours per calendar year.
- (ii) Emergency stationary RICE may be operated for emergency demand response for periods in which the Reliability Coordinator under the North American Electric Reliability Corporation (NERC) Reliability Standard EOP-002-3, Capacity and Energy Emergencies (incorporated by reference, see §63.14), or other authorized entity as determined by the Reliability Coordinator, has declared an Energy Emergency Alert Level 2 as defined in the NERC Reliability Standard EOP-002-3.
- (iii) Emergency stationary RICE may be operated for periods where there is a deviation of voltage or frequency of 5 percent or greater below standard voltage or frequency.

- (3) Emergency stationary RICE located at major sources of HAP may be operated for up to 50 hours per calendar year in non-emergency situations. The 50 hours of operation in non-emergency situations are counted as part of the 100 hours per calendar year for maintenance and testing and emergency demand response provided in paragraph (f)(2) of this section. The 50 hours per year for non-emergency situations cannot be used for peak shaving or nonemergency demand response, or to generate income for a facility to supply power to an electric grid or otherwise supply power as part of a financial arrangement with another entity.
- (4) Emergency stationary RICE located at area sources of HAP may be operated for up to 50 hours per calendar year in non-emergency situations. The 50 hours of operation in non-emergency situations are counted as part of the 100 hours per calendar year for maintenance and testing and emergency demand response provided in paragraph (f)(2) of this section. Except as provided in paragraphs (f)(4)(i) and (ii) of this section, the 50 hours per year for nonemergency situations cannot be used for peak shaving or non-emergency demand response, or to generate income for a facility to an electric grid or otherwise supply power as part of a financial arrangement with another entity.
- (i) Prior to May 3, 2014, the 50 hours per year for non-emergency situations can be used for peak shaving or non-emergency demand response to generate income for a facility, or to otherwise supply power as part of a financial arrangement with another entity if the engine is operated as part of a peak shaving (load management program) with the local distribution system operator and the power is provided only to the facility itself or to support the local distribution system.
- (ii) The 50 hours per year for nonemergency situations can be used to supply power as part of a financial arrangement with another entity if all of the following conditions are met:
- (A) The engine is dispatched by the local balancing authority or local transmission and distribution system operator.
- (B) The dispatch is intended to mitigate local transmission and/or distribu-

- tion limitations so as to avert potential voltage collapse or line overloads that could lead to the interruption of power supply in a local area or region.
- (C) The dispatch follows reliability, emergency operation or similar protocols that follow specific NERC, regional, state, public utility commission or local standards or guidelines.
- (D) The power is provided only to the facility itself or to support the local transmission and distribution system.
- (E) The owner or operator identifies and records the entity that dispatches the engine and the specific NERC, regional, state, public utility commission or local standards or guidelines that are being followed for dispatching the engine. The local balancing authority or local transmission and distribution system operator may keep these records on behalf of the engine owner or operator.

[69 FR 33506, June 15, 2004, as amended at 71 FR 20467, Apr. 20, 2006; 73 FR 3606, Jan. 18, 2008; 75 FR 9676, Mar. 3, 2010; 75 FR 51591, Aug. 20, 2010; 78 FR 6704, Jan. 30, 2013]

NOTIFICATIONS, REPORTS, AND RECORDS

### § 63.6645 What notifications must I submit and when?

- (a) You must submit all of the notifications in §§63.7(b) and (c), 63.8(e), (f)(4) and (f)(6), 63.9(b) through (e), and (g) and (h) that apply to you by the dates specified if you own or operate any of the following:
- (1) An existing stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions.
- (2) An existing stationary RICE located at an area source of HAP emissions.
- (3) A stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions.
- (4) A new or reconstructed 4SLB stationary RICE with a site rating of greater than or equal to 250 HP located at a major source of HAP emissions.
- (5) This requirement does not apply if you own or operate an existing stationary RICE less than 100 HP, an existing stationary emergency RICE, or an existing stationary RICE that is not subject to any numerical emission standards.

### 33

- § 63.6650
- (b) As specified in §63.9(b)(2), if you start up your stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions before the effective date of this subpart, you must submit an Initial Notification not later than December 13, 2004.
- (c) If you start up your new or reconstructed stationary RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions on or after August 16, 2004, you must submit an Initial Notification not later than 120 days after you become subject to this subpart.
- (d) As specified in §63.9(b)(2), if you start up your stationary RICE with a site rating of equal to or less than 500 brake HP located at a major source of HAP emissions before the effective date of this subpart and you are required to submit an initial notification, you must submit an Initial Notification not later than July 16, 2008.
- (e) If you start up your new or reconstructed stationary RICE with a site rating of equal to or less than 500 brake HP located at a major source of HAP emissions on or after March 18, 2008 and you are required to submit an initial notification, you must submit an Initial Notification not later than 120 days after you become subject to this subpart.
- (f) If you are required to submit an Initial Notification but are otherwise not affected by the requirements of this subpart, in accordance with §63.6590(b), your notification should include the information in §63.9(b)(2)(i) through (v), and a statement that your stationary RICE has no additional requirements and explain the basis of the exclusion (for example, that it operates exclusively as an emergency stationary RICE if it has a site rating of more than 500 brake HP located at a major source of HAP emissions).
- (g) If you are required to conduct a performance test, you must submit a Notification of Intent to conduct a performance test at least 60 days before the performance test is scheduled to begin as required in §63.7(b)(1).
- (h) If you are required to conduct a performance test or other initial compliance demonstration as specified in Tables 4 and 5 to this subpart, you

must submit a Notification of Compliance Status according to §63.9(h)(2)(ii).

- (1) For each initial compliance demonstration required in Table 5 to this subpart that does not include a performance test, you must submit the Notification of Compliance Status before the close of business on the 30th day following the completion of the initial compliance demonstration.
- (2) For each initial compliance demonstration required in Table 5 to this subpart that includes a performance test conducted according to the requirements in Table 3 to this subpart, you must submit the Notification of Compliance Status, including the performance test results, before the close of business on the 60th day following the completion of the performance test according to §63.10(d)(2).
- (i) If you own or operate an existing non-emergency CI RICE with a site rating of more than 300 HP located at an area source of HAP emissions that is certified to the Tier 1 or Tier 2 emission standards in Table 1 of 40 CFR 89.112 and subject to an enforceable state or local standard requiring engine replacement and you intend to meet management practices rather than emission limits, as specified in §63.6603(d), you must submit a notification by March 3, 2013, stating that you intend to use the provision in §63.6603(d) and identifying the state or local regulation that the engine is subject to.

[73 FR 3606, Jan. 18, 2008, as amended at 75 FR 9677, Mar. 3, 2010; 75 FR 51591, Aug. 20, 2010; 78 FR 6705, Jan. 30, 2013]

### § 63.6650 What reports must I submit and when?

- (a) You must submit each report in Table 7 of this subpart that applies to you.
- (b) Unless the Administrator has approved a different schedule for submission of reports under §63.10(a), you must submit each report by the date in Table 7 of this subpart and according to the requirements in paragraphs (b)(1) through (b)(9) of this section.
- (1) For semiannual Compliance reports, the first Compliance report must cover the period beginning on the compliance date that is specified for your affected source in §63.6595 and ending

- on June 30 or December 31, whichever date is the first date following the end of the first calendar half after the compliance date that is specified for your source in §63.6595.
- (2) For semiannual Compliance reports, the first Compliance report must be postmarked or delivered no later than July 31 or January 31, whichever date follows the end of the first calendar half after the compliance date that is specified for your affected source in §63.6595.
- (3) For semiannual Compliance reports, each subsequent Compliance report must cover the semiannual reporting period from January 1 through June 30 or the semiannual reporting period from July 1 through December 31.
- (4) For semiannual Compliance reports, each subsequent Compliance report must be postmarked or delivered no later than July 31 or January 31, whichever date is the first date following the end of the semiannual reporting period.
- (5) For each stationary RICE that is subject to permitting regulations pursuant to 40 CFR part 70 or 71, and if the permitting authority has established dates for submitting semiannual repursuant to 70.6(a)(3)(iii)(A) or 40 CFR 71.6 (a)(3)(iii)(A), you may submit the first and subsequent Compliance reports according to the dates the permitting authority has established instead of according to the dates in paragraphs (b)(1) through (b)(4) of this section.
- (6) For annual Compliance reports, the first Compliance report must cover the period beginning on the compliance date that is specified for your affected source in §63.6595 and ending on December 31.
- (7) For annual Compliance reports, the first Compliance report must be postmarked or delivered no later than January 31 following the end of the first calendar year after the compliance date that is specified for your affected source in §63.6595.
- (8) For annual Compliance reports, each subsequent Compliance report must cover the annual reporting period from January 1 through December 31.
- (9) For annual Compliance reports, each subsequent Compliance report

- must be postmarked or delivered no later than January 31.
- (c) The Compliance report must contain the information in paragraphs (c)(1) through (6) of this section.
  - (1) Company name and address.
- (2) Statement by a responsible official, with that official's name, title, and signature, certifying the accuracy of the content of the report.
- (3) Date of report and beginning and ending dates of the reporting period.
- (4) If you had a malfunction during the reporting period, the compliance report must include the number, duration, and a brief description for each type of malfunction which occurred during the reporting period and which caused or may have caused any applicable emission limitation to be exceeded. The report must also include a description of actions taken by an owner or operator during a malfunction of an affected source to minimize emissions in accordance with §63.6605(b), including actions taken to correct a malfunction.
- (5) If there are no deviations from any emission or operating limitations that apply to you, a statement that there were no deviations from the emission or operating limitations during the reporting period.
- (6) If there were no periods during which the continuous monitoring system (CMS), including CEMS and CPMS, was out-of-control, as specified in §63.8(c)(7), a statement that there were no periods during which the CMS was out-of-control during the reporting period.
- (d) For each deviation from an emission or operating limitation that occurs for a stationary RICE where you are not using a CMS to comply with the emission or operating limitations in this subpart, the Compliance report must contain the information in paragraphs (c)(1) through (4) of this section and the information in paragraphs (d)(1) and (2) of this section.
- (1) The total operating time of the stationary RICE at which the deviation occurred during the reporting period.
- (2) Information on the number, duration, and cause of deviations (including unknown cause, if applicable), as applicable, and the corrective action taken.

### § 63.6650

- (e) For each deviation from an emission or operating limitation occurring for a stationary RICE where you are using a CMS to comply with the emission and operating limitations in this subpart, you must include information in paragraphs (c)(1) through (4) and (e)(1) through (12) of this section.
- (1) The date and time that each malfunction started and stopped.
- (2) The date, time, and duration that each CMS was inoperative, except for zero (low-level) and high-level checks.
- (3) The date, time, and duration that each CMS was out-of-control, including the information in \$63.8(c)(8).
- (4) The date and time that each deviation started and stopped, and whether each deviation occurred during a period of malfunction or during another period.
- (5) A summary of the total duration of the deviation during the reporting period, and the total duration as a percent of the total source operating time during that reporting period.
- (6) A breakdown of the total duration of the deviations during the reporting period into those that are due to control equipment problems, process problems, other known causes, and other unknown causes.
- (7) A summary of the total duration of CMS downtime during the reporting period, and the total duration of CMS downtime as a percent of the total operating time of the stationary RICE at which the CMS downtime occurred during that reporting period.
- (8) An identification of each parameter and pollutant (CO or formaldehyde) that was monitored at the stationary RICE.
- (9) A brief description of the stationary RICE.
- (10) A brief description of the CMS.
- (11) The date of the latest CMS certification or audit.
- (12) A description of any changes in CMS, processes, or controls since the last reporting period.
- (f) Each affected source that has obtained a title V operating permit pursuant to 40 CFR part 70 or 71 must report all deviations as defined in this subpart in the semiannual monitoring report required by 40 CFR 70.6 (a)(3)(iii)(A) or 40 CFR 71.6(a)(3)(iii)(A). If an affected source submits a Compli-

- ance report pursuant to Table 7 of this subpart along with, or as part of, the semiannual monitoring report required by 40 CFR 70.6(a)(3)(iii)(A) or 40 CFR 71.6(a)(3)(iii)(A), and the Compliance report includes all required information concerning deviations from any emission or operating limitation in this subpart, submission of the Compliance report shall be deemed to satisfy any obligation to report the same deviations in the semiannual monitoring report. However, submission of a Compliance report shall not otherwise affect any obligation the affected source may have to report deviations from permit requirements to the permit authority.
- (g) If you are operating as a new or reconstructed stationary RICE which fires landfill gas or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis, you must submit an annual report according to Table 7 of this subpart by the date specified unless the Administrator has approved a different schedule, according to the information described in paragraphs (b)(1) through (b)(5) of this section. You must report the data specified in (g)(1) through (g)(3) of this section.
- (1) Fuel flow rate of each fuel and the heating values that were used in your calculations. You must also demonstrate that the percentage of heat input provided by landfill gas or digester gas is equivalent to 10 percent or more of the total fuel consumption on an annual basis.
- (2) The operating limits provided in your federally enforceable permit, and any deviations from these limits.
- (3) Any problems or errors suspected with the meters.
- (h) If you own or operate an emergency stationary RICE with a site rating of more than 100 brake HP that operates or is contractually obligated to be available for more than 15 hours per calendar year for the purposes specified in §63.6640(f)(2)(ii) and (iii) or that operates for the purpose specified in §63.6640(f)(4)(ii), you must submit an annual report according to the requirements in paragraphs (h)(1) through (3) of this section.
- (1) The report must contain the following information:

- (i) Company name and address where the engine is located.
- (ii) Date of the report and beginning and ending dates of the reporting period.
- (iii) Engine site rating and model year.
- (iv) Latitude and longitude of the engine in decimal degrees reported to the fifth decimal place.
- (v) Hours operated for the purposes specified in §63.6640(f)(2)(ii) and (iii), including the date, start time, and end time for engine operation for the purposes specified in §63.6640(f)(2)(ii) and (iii).
- (vi) Number of hours the engine is contractually obligated to be available for the purposes specified in §63.6640(f)(2)(ii) and (iii).
- (vii) Hours spent for operation for the purpose specified in §63.6640(f)(4)(ii), including the date, start time, and end time for engine operation for the purposes specified in §63.6640(f)(4)(ii). The report must also identify the entity that dispatched the engine and the situation that necessitated the dispatch of the engine.
- (viii) If there were no deviations from the fuel requirements in §63.6604 that apply to the engine (if any), a statement that there were no deviations from the fuel requirements during the reporting period.
- (ix) If there were deviations from the fuel requirements in §63.6604 that apply to the engine (if any), information on the number, duration, and cause of deviations, and the corrective action taken.
- (2) The first annual report must cover the calendar year 2015 and must be submitted no later than March 31, 2016. Subsequent annual reports for each calendar year must be submitted no later than March 31 of the following calendar year.
- (3) The annual report must be submitted electronically using the subpart specific reporting form in the Compliance and Emissions Data Reporting Interface (CEDRI) that is accessed through EPA's Central Data Exchange (CDX) (www.epa.gov/cdx). However, if the reporting form specific to this subpart is not available in CEDRI at the time that the report is due, the written report must be submitted to the Ad-

ministrator at the appropriate address listed in §63.13.

[69 FR 33506, June 15, 2004, as amended at 75 FR 9677, Mar. 3, 2010; 78 FR 6705, Jan. 30, 2013]

### §63.6655 What records must I keep?

- (a) If you must comply with the emission and operating limitations, you must keep the records described in paragraphs (a)(1) through (a)(5), (b)(1) through (b)(3) and (c) of this section.
- (1) A copy of each notification and report that you submitted to comply with this subpart, including all documentation supporting any Initial Notification or Notification of Compliance Status that you submitted, according to the requirement in §63.10(b)(2)(xiv).
- (2) Records of the occurrence and duration of each malfunction of operation (i.e., process equipment) or the air pollution control and monitoring equipment.
- (3) Records of performance tests and performance evaluations as required in §63.10(b)(2)(viii).
- (4) Records of all required maintenance performed on the air pollution control and monitoring equipment.
- (5) Records of actions taken during periods of malfunction to minimize emissions in accordance with §63.6605(b), including corrective actions to restore malfunctioning process and air pollution control and monitoring equipment to its normal or usual manner of operation.
- (b) For each CEMS or CPMS, you must keep the records listed in paragraphs (b)(1) through (3) of this section.
- (1) Records described §63.10(b)(2)(vi) through (xi).
- (2) Previous (i.e., superseded) versions of the performance evaluation plan as required in \$63.8(d)(3).
- (3) Requests for alternatives to the relative accuracy test for CEMS or CPMS as required in §63.8(f)(6)(i), if applicable.
- (c) If you are operating a new or reconstructed stationary RICE which fires landfill gas or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis, you must keep the records of your daily fuel usage monitors.
- (d) You must keep the records required in Table 6 of this subpart to show continuous compliance with each

emission or operating limitation that applies to you.

- (e) You must keep records of the maintenance conducted on the stationary RICE in order to demonstrate that you operated and maintained the stationary RICE and after-treatment control device (if any) according to your own maintenance plan if you own or operate any of the following stationary RICE;
- (1) An existing stationary RICE with a site rating of less than 100 brake HP located at a major source of HAP emissions.
- (2) An existing stationary emergency RICE.
- (3) An existing stationary RICE located at an area source of HAP emissions subject to management practices as shown in Table 2d to this subpart.
- (f) If you own or operate any of the stationary RICE in paragraphs (f)(1) through (2) of this section, you must keep records of the hours of operation of the engine that is recorded through the non-resettable hour meter. The owner or operator must document how many hours are spent for emergency operation, including what classified the operation as emergency and how many hours are spent for non-emergency operation. If the engine is used the purposes specified § 63.6640(f)(2)(ii) (iii) oror §63.6640(f)(4)(ii), the owner or operator must keep records of the notification of the emergency situation, and the date, start time, and end time of engine operation for these purposes.
- (1) An existing emergency stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions that does not meet the standards applicable to non-emergency engines.
- (2) An existing emergency stationary RICE located at an area source of HAP emissions that does not meet the standards applicable to non-emergency engines.

[69 FR 33506, June 15, 2004, as amended at 75 FR 9678, Mar. 3, 2010; 75 FR 51592, Aug. 20, 2010; 78 FR 6706, Jan. 30, 2013]

# §63.6660 In what form and how long must I keep my records?

- (a) Your records must be in a form suitable and readily available for expeditious review according to §63.10(b)(1).
- (b) As specified in §63.10(b)(1), you must keep each record for 5 years following the date of each occurrence, measurement, maintenance, corrective action, report, or record.
- (c) You must keep each record readily accessible in hard copy or electronic form for at least 5 years after the date of each occurrence, measurement, maintenance, corrective action, report, or record, according to §63.10(b)(1).

[69 FR 33506, June 15, 2004, as amended at 75 FR 9678, Mar. 3, 2010]

OTHER REQUIREMENTS AND INFORMATION

### § 63.6665 What parts of the General Provisions apply to me?

Table 8 to this subpart shows which parts of the General Provisions in §§ 63.1 through 63.15 apply to you. If you own or operate a new or reconstructed stationary RICE with a site rating of less than or equal to 500 brake HP located at a major source of HAP emissions (except new or reconstructed 4SLB engines greater than or equal to 250 and less than or equal to 500 brake HP), a new or reconstructed stationary RICE located at an area source of HAP emissions, or any of the following RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, you do not need to comply with any of the requirements of the General Provisions specified in Table 8: An existing 2SLB stationary RICE, an existing 4SLB stationary RICE, an existing stationary RICE that combusts landfill or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis, an existing emergency stationary RICE, or an existing limited use stationary RICE. If you own or operate any of the following RICE with a site rating of more than 500 brake HP located at a major source of HAP emissions, you do not need to comply with the requirements in the General Provisions specified in Table 8 except for the initial notification requirements: A new stationary

RICE that combusts landfill gas or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis, a new emergency stationary RICE, or a new limited use stationary RICE.

[75 FR 9678, Mar. 3, 2010]

# § 63.6670 Who implements and enforces this subpart?

(a) This subpart is implemented and enforced by the U.S. EPA, or a delegated authority such as your State, local, or tribal agency. If the U.S. EPA Administrator has delegated authority to your State, local, or tribal agency, then that agency (as well as the U.S. EPA) has the authority to implement and enforce this subpart. You should contact your U.S. EPA Regional Office to find out whether this subpart is delegated to your State, local, or tribal agency.

(b) In delegating implementation and enforcement authority of this subpart to a State, local, or tribal agency under 40 CFR part 63, subpart E, the authorities contained in paragraph (c) of this section are retained by the Administrator of the U.S. EPA and are not transferred to the State, local, or tribal agency.

(c) The authorities that will not be delegated to State, local, or tribal agencies are:

(1) Approval of alternatives to the non-opacity emission limitations and operating limitations in §63.6600 under §63.6(g).

(2) Approval of major alternatives to test methods under §63.7(e)(2)(ii) and (f) and as defined in §63.90.

(3) Approval of major alternatives to monitoring under §63.8(f) and as defined in §63.90.

(4) Approval of major alternatives to recordkeeping and reporting under §63.10(f) and as defined in §63.90.

(5) Approval of a performance test which was conducted prior to the effective date of the rule, as specified in §63.6610(b).

# \$ 63.6675 What definitions apply to this subpart?

Terms used in this subpart are defined in the Clean Air Act (CAA); in 40 CFR 63.2, the General Provisions of this part; and in this section as follows:

Alaska Railbelt Grid means the service areas of the six regulated public utilities that extend from Fairbanks to Anchorage and the Kenai Peninsula. These utilities are Golden Valley Electric Association; Chugach Electric Association; Matanuska Electric Association; Homer Electric Association; Anchorage Municipal Light & Power; and the City of Seward Electric System.

Area source means any stationary source of HAP that is not a major source as defined in part 63.

Associated equipment as used in this subpart and as referred to in section 112(n)(4) of the CAA, means equipment associated with an oil or natural gas exploration or production well, and includes all equipment from the well bore to the point of custody transfer, except glycol dehydration units, storage vessels with potential for flash emissions, combustion turbines, and stationary RICE.

Backup power for renewable energy means an engine that provides backup power to a facility that generates electricity from renewable energy resources, as that term is defined in Alaska Statute 42.45.045(1)(5) (incorporated by reference, see §63.14).

Black start engine means an engine whose only purpose is to start up a combustion turbine.

*CAA* means the Clean Air Act (42 U.S.C. 7401 *et seq.*, as amended by Public Law 101-549, 104 Stat. 2399).

Commercial emergency stationary RICE means an emergency stationary RICE used in commercial establishments such as office buildings, hotels, stores, telecommunications facilities, restaurants, financial institutions such as banks, doctor's offices, and sports and performing arts facilities.

Compression ignition means relating to a type of stationary internal combustion engine that is not a spark ignition engine.

Custody transfer means the transfer of hydrocarbon liquids or natural gas: After processing and/or treatment in the producing operations, or from storage vessels or automatic transfer facilities or other such equipment, including product loading racks, to pipelines or any other forms of transportation. For the purposes of this subpart, the point at which such liquids or

natural gas enters a natural gas processing plant is a point of custody transfer

Deviation means any instance in which an affected source subject to this subpart, or an owner or operator of such a source:

- (1) Fails to meet any requirement or obligation established by this subpart, including but not limited to any emission limitation or operating limitation:
- (2) Fails to meet any term or condition that is adopted to implement an applicable requirement in this subpart and that is included in the operating permit for any affected source required to obtain such a permit; or
- (3) Fails to meet any emission limitation or operating limitation in this subpart during malfunction, regardless or whether or not such failure is permitted by this subpart.
- (4) Fails to satisfy the general duty to minimize emissions established by \$63.6(e)(1)(i).

Diesel engine means any stationary RICE in which a high boiling point liquid fuel injected into the combustion chamber ignites when the air charge has been compressed to a temperature sufficiently high for auto-ignition. This process is also known as compression ignition.

Diesel fuel means any liquid obtained from the distillation of petroleum with a boiling point of approximately 150 to 360 degrees Celsius. One commonly used form is fuel oil number 2. Diesel fuel also includes any non-distillate fuel with comparable physical and chemical properties (e.g. biodiesel) that is suitable for use in compression ignition engines.

Digester gas means any gaseous byproduct of wastewater treatment typically formed through the anaerobic decomposition of organic waste materials and composed principally of methane and CO<sub>2</sub>.

Dual-fuel engine means any stationary RICE in which a liquid fuel (typically diesel fuel) is used for compression ignition and gaseous fuel (typically natural gas) is used as the primary fuel.

Emergency stationary RICE means any stationary reciprocating internal combustion engine that meets all of the

criteria in paragraphs (1) through (3) of this definition. All emergency stationary RICE must comply with the requirements specified in §63.6640(f) in order to be considered emergency stationary RICE. If the engine does not comply with the requirements specified in §63.6640(f), then it is not considered to be an emergency stationary RICE under this subpart.

- (1) The stationary RICE is operated to provide electrical power or mechanical work during an emergency situation. Examples include stationary RICE used to produce power for critical networks or equipment (including power supplied to portions of a facility) when electric power from the local utility (or the normal power source, if the facility runs on its own power production) is interrupted, or stationary RICE used to pump water in the case of fire or flood, etc.
- (2) The stationary RICE is operated under limited circumstances for situations not included in paragraph (1) of this definition, as specified in  $\S63.6640(f)$ .
- (3) The stationary RICE operates as part of a financial arrangement with another entity in situations not included in paragraph (1) of this definition only as allowed in §63.6640(f)(2)(ii) or (iii) and §63.6640(f)(4)(i) or (ii).

Engine startup means the time from initial start until applied load and engine and associated equipment reaches steady state or normal operation. For stationary engine with catalytic controls, engine startup means the time from initial start until applied load and engine and associated equipment, including the catalyst, reaches steady state or normal operation.

Four-stroke engine means any type of engine which completes the power cycle in two crankshaft revolutions, with intake and compression strokes in the first revolution and power and exhaust strokes in the second revolution.

Gaseous fuel means a material used for combustion which is in the gaseous state at standard atmospheric temperature and pressure conditions.

Gasoline means any fuel sold in any State for use in motor vehicles and motor vehicle engines, or nonroad or stationary engines, and commonly or

commercially known or sold as gasoline

Glycol dehydration unit means a device in which a liquid glycol (including, but not limited to, ethylene glycol, diethylene glycol, or triethylene glycol) absorbent directly contacts a natural gas stream and absorbs water in a contact tower or absorption column (absorber). The glycol contacts and absorbs water vapor and other gas stream constituents from the natural gas and becomes "rich" glycol. This glycol is then regenerated in the glycol dehydration unit reboiler. The "lean" glycol is then recycled.

Hazardous air pollutants (HAP) means any air pollutants listed in or pursuant to section 112(b) of the CAA.

Institutional emergency stationary RICE means an emergency stationary RICE used in institutional establishments such as medical centers, nursing homes, research centers, institutions of higher education, correctional facilities, elementary and secondary schools, libraries, religious establishments, police stations, and fire stations.

ISO standard day conditions means 288 degrees Kelvin (15 degrees Celsius), 60 percent relative humidity and 101.3 kilopascals pressure.

Landfill gas means a gaseous by-product of the land application of municipal refuse typically formed through the anaerobic decomposition of waste materials and composed principally of methane and CO<sub>2</sub>.

Lean burn engine means any twostroke or four-stroke spark ignited engine that does not meet the definition of a rich burn engine.

Limited use stationary RICE means any stationary RICE that operates less than 100 hours per year.

Liquefied petroleum gas means any liquefied hydrocarbon gas obtained as a by-product in petroleum refining of natural gas production.

Liquid fuel means any fuel in liquid form at standard temperature and pressure, including but not limited to diesel, residual/crude oil, kerosene/naphtha (jet fuel), and gasoline.

Major Source, as used in this subpart, shall have the same meaning as in §63.2, except that:

- (1) Emissions from any oil or gas exploration or production well (with its associated equipment (as defined in this section)) and emissions from any pipeline compressor station or pump station shall not be aggregated with emissions from other similar units, to determine whether such emission points or stations are major sources, even when emission points are in a contiguous area or under common control;
- (2) For oil and gas production facilities, emissions from processes, operations, or equipment that are not part of the same oil and gas production facility, as defined in §63.1271 of subpart HHH of this part, shall not be aggregated;
- (3) For production field facilities, only HAP emissions from glycol dehydration units, storage vessel with the potential for flash emissions, combustion turbines and reciprocating internal combustion engines shall be aggregated for a major source determination; and
- (4) Emissions from processes, operations, and equipment that are not part of the same natural gas transmission and storage facility, as defined in §63.1271 of subpart HHH of this part, shall not be aggregated.

Malfunction means any sudden, infrequent, and not reasonably preventable failure of air pollution control equipment, process equipment, or a process to operate in a normal or usual manner which causes, or has the potential to cause, the emission limitations in an applicable standard to be exceeded. Failures that are caused in part by poor maintenance or careless operation are not malfunctions.

Natural gas means a naturally occurring mixture of hydrocarbon and non-hydrocarbon gases found in geologic formations beneath the Earth's surface, of which the principal constituent is methane. Natural gas may be field or pipeline quality.

Non-selective catalytic reduction (NSCR) means an add-on catalytic nitrogen oxides ( $NO_X$ ) control device for rich burn engines that, in a two-step reaction, promotes the conversion of excess oxygen,  $NO_X$ , CO, and volatile organic compounds (VOC) into  $CO_2$ , nitrogen, and water.

Oil and gas production facility as used in this subpart means any grouping of equipment where hydrocarbon liquids are processed, upgraded (i.e., remove impurities or other constituents to meet contract specifications), or stored prior to the point of custody transfer; or where natural gas is processed, upgraded, or stored prior to entering the natural gas transmission and storage source category. For purposes of a major source determination, facility (including a building, structure, or installation) means oil and natural gas production and processing equipment that is located within the boundaries of an individual surface site as defined in this section. Equipment that is part of a facility will typically be located within close proximity to other equipment located at the same facility. Pieces of production equipment or groupings of equipment located on different oil and gas leases, mineral fee tracts, lease tracts, subsurface or surface unit areas, surface fee tracts, surface lease tracts, or separate surface sites, whether or not connected by a road, waterway, power line or pipeline, shall not be considered part of the same facility. Examples of facilities in the oil and natural gas production source category include, but are not limited to, well sites, satellite tank batteries, central tank batteries, a compressor station that transports natural gas to a natural gas processing plant, and natural gas processing

Oxidation catalyst means an add-on catalytic control device that controls CO and VOC by oxidation.

Peaking unit or engine means any standby engine intended for use during periods of high demand that are not emergencies.

Percent load means the fractional power of an engine compared to its maximum manufacturer's design capacity at engine site conditions. Percent load may range between 0 percent to above 100 percent.

Potential to emit means the maximum capacity of a stationary source to emit a pollutant under its physical and operational design. Any physical or operational limitation on the capacity of the stationary source to emit a pollutant, including air pollution control

equipment and restrictions on hours of operation or on the type or amount of material combusted, stored, or processed, shall be treated as part of its design if the limitation or the effect it would have on emissions is federally enforceable. For oil and natural gas production facilities subject to subpart HH of this part, the potential to emit provisions in §63.760(a) may be used. For natural gas transmission and storage facilities subject to subpart HHH of this part, the maximum annual facility gas throughput for storage facilities may be determined according to §63.1270(a)(1) and the maximum annual throughput for transmission facilities may be determined according to §63.1270(a)(2).

Production field facility means those oil and gas production facilities located prior to the point of custody transfer.

Production well means any hole drilled in the earth from which crude oil, condensate, or field natural gas is extracted.

Propane means a colorless gas derived from petroleum and natural gas, with the molecular structure  $C_3H_8$ .

Remote stationary RICE means stationary RICE meeting any of the following criteria:

- (1) Stationary RICE located in an offshore area that is beyond the line of ordinary low water along that portion of the coast of the United States that is in direct contact with the open seas and beyond the line marking the seaward limit of inland waters.
- (2) Stationary RICE located on a pipeline segment that meets both of the criteria in paragraphs (2)(i) and (ii) of this definition.
- (i) A pipeline segment with 10 or fewer buildings intended for human occupancy and no buildings with four or more stories within 220 yards (200 meters) on either side of the centerline of any continuous 1-mile (1.6 kilometers) length of pipeline. Each separate dwelling unit in a multiple dwelling unit building is counted as a separate building intended for human occupancy.

(ii) The pipeline segment does not lie within 100 yards (91 meters) of either a building or a small, well-defined outside area (such as a playground, recreation area, outdoor theater, or other

place of public assembly) that is occupied by 20 or more persons on at least 5 days a week for 10 weeks in any 12-month period. The days and weeks need not be consecutive. The building or area is considered occupied for a full day if it is occupied for any portion of the day.

(iii) For purposes of this paragraph (2), the term pipeline segment means all parts of those physical facilities through which gas moves in transportation, including but not limited to pipe, valves, and other appurtenance attached to pipe, compressor units, metering stations, regulator stations, delivery stations, holders, and fabricated assemblies. Stationary RICE located within 50 vards (46 meters) of the pipeline segment providing power for equipment on a pipeline segment are part of the pipeline segment. Transportation of gas means the gathering, transmission, or distribution of gas by pipeline, or the storage of gas. A building is intended for human occupancy if its primary use is for a purpose involving the presence of humans.

(3) Stationary RICE that are not located on gas pipelines and that have 5 or fewer buildings intended for human occupancy and no buildings with four or more stories within a 0.25 mile radius around the engine. A building is intended for human occupancy if its primary use is for a purpose involving the presence of humans.

Residential emergency stationary RICE means an emergency stationary RICE used in residential establishments such as homes or apartment buildings.

Responsible official means responsible official as defined in 40 CFR 70.2.

Rich burn engine means any fourstroke spark ignited engine where the manufacturer's recommended operating air/fuel ratio divided by the stoichiometric air/fuel ratio at full load conditions is less than or equal to 1.1. Engines originally manufactured as rich burn engines, but modified prior to December 19, 2002 with passive emission control technology for NOx (such as pre-combustion chambers) will be considered lean burn engines. Also, existing engines where there are no manufacturer's recommendations regarding air/fuel ratio will be considered a rich burn engine if the excess oxygen content of the exhaust at full load conditions is less than or equal to 2 percent. Site-rated HP means the maximum manufacturer's design capacity at engine site conditions.

Spark ignition means relating to either: A gasoline-fueled engine; or any other type of engine with a spark plug (or other sparking device) and with operating characteristics significantly similar to the theoretical Otto combustion cycle. Spark ignition engines usually use a throttle to regulate intake air flow to control power during normal operation. Dual-fuel engines in which a liquid fuel (typically diesel fuel) is used for CI and gaseous fuel (typically natural gas) is used as the primary fuel at an annual average ratio of less than 2 parts diesel fuel to 100 parts total fuel on an energy equivalent basis are spark ignition engines.

Stationary reciprocating internal combustion engine (RICE) means any reciprocating internal combustion engine which uses reciprocating motion to convert heat energy into mechanical work and which is not mobile. Stationary RICE differ from mobile RICE in that a stationary RICE is not a nonroad engine as defined at 40 CFR 1068.30, and is not used to propel a motor vehicle or a vehicle used solely for competition.

Stationary RICE test cell/stand means an engine test cell/stand, as defined in subpart PPPPP of this part, that tests stationary RICE.

Stoichiometric means the theoretical air-to-fuel ratio required for complete combustion.

Storage vessel with the potential for flash emissions means any storage vessel that contains a hydrocarbon liquid with a stock tank gas-to-oil ratio equal to or greater than 0.31 cubic meters per liter and an American Petroleum Institute gravity equal to or greater than 40 degrees and an actual annual average hydrocarbon liquid throughput equal to or greater than 79,500 liters per day. Flash emissions occur when dissolved hydrocarbons in the fluid evolve from solution when the fluid pressure is reduced.

Subpart means 40 CFR part 63, subpart ZZZZ.

Surface site means any combination of one or more graded pad sites, gravel

### Pt. 63, Subpt. ZZZZ, Table 1a

### 40 CFR Ch. I (7-1-13 Edition)

pad sites, foundations, platforms, or the immediate physical location upon which equipment is physically affixed.

Two-stroke engine means a type of engine which completes the power cycle in single crankshaft revolution by combining the intake and compression operations into one stroke and the power and exhaust operations into a second

stroke. This system requires auxiliary scavenging and inherently runs lean of stoichiometric.

[69 FR 33506, June 15, 2004, as amended at 71 FR 20467, Apr. 20, 2006; 73 FR 3607, Jan. 18, 2008; 75 FR 9679, Mar. 3, 2010; 75 FR 51592, Aug. 20, 2010; 76 FR 12867, Mar. 9, 2011; 78 FR 6706, Jan. 30, 2013]

Table 1a to Subpart ZZZZ of Part 63—Emission Limitations for Existing, New, and Reconstructed Spark Ignition, 4SRB Stationary RICE >500 HP Located at a Major Source of HAP Emissions

As stated in  $\S 63.6600$  and 63.6640, you must comply with the following emission limitations at 100 percent load plus or minus 10 percent for existing, new and reconstructed 4SRB stationary RICE >500 HP located at a major source of HAP emissions:

For each	You must meet the following emission limitation, except during periods of startup	During periods of startup you must
1. 4SRB stationary RICE	Reduce formaldehyde emissions by 76 percent or more. If you commenced construction or reconstruction between December 19, 2002 and June 15, 2004, you may reduce formaldehyde emissions by 75 percent or more until June 15, 2007 or.     Limit the concentration of formaldehyde in the stationary RICE exhaust to 350 ppbvd or less at 15 percent O <sub>2</sub> .	ing of the engine, not to exceed 30 minutes, after which time the non-startup emission limi- tations apply. <sup>1</sup>

<sup>&</sup>lt;sup>1</sup> Sources can petition the Administrator pursuant to the requirements of 40 CFR 63.6(g) for alternative work practices.

 $[75~{\rm FR}~9679,\,{\rm Mar}.~3,\,2010,\,{\rm as}~{\rm amended}~{\rm at}~75~{\rm FR}~51592,\,{\rm Aug}.~20,\,2010]$ 

TABLE 1b TO SUBPART ZZZZ OF PART 63—OPERATING LIMITATIONS FOR EXISTING, NEW, AND RECONSTRUCTED SI 4SRB STATIONARY RICE >500 HP LOCATED AT A MAJOR SOURCE OF HAP EMISSIONS

As stated in  $\S 63.6600$ , 63.6603, 63.6603 and 63.6640, you must comply with the following operating limitations for existing, new and reconstructed 4SRB stationary RICE >500 HP located at a major source of HAP emissions:

For each	You must meet the following operating limitation, except during periods of startup
1. existing, new and reconstructed 4SRB stationary RICE >500 HP located at a major source of HAP emissions complying with the requirement to reduce formatlehyde emissions by 76 percent or more (or by 75 percent or more, if applicable) and using NSCR; or existing, new and reconstructed 4SRB stationary RICE >500 HP located at a major source of HAP emissions complying with the requirement to limit the concentration of formaldehyde in the stationary RICE exhaust to 350 ppbvd or less at 15 percent O <sub>2</sub> and using NSCR; 2. existing, new and reconstructed 4SRB stationary RICE >500 HP located at a major source of HAP emissions complying with the requirement to reduce formaldehyde emissions by 76 percent or more (or by 75 percent or more, if applicable) and not using NSCR; or existing, new and reconstructed 4SRB stationary RICE >500 HP located at a major source of HAP emissions complying with the requirement to limit the concentration of formaldehyde in the stationary RICE exhaust to 350 ppbvd or less at 15 percent O <sub>2</sub> and not using NSCR.	catalyst does not change by more than 2 inches of water at 100 percent load plus or minus 10 percent from the pressure drop across the catalyst measured during the initial performance test; and b. maintain the temperature of your stationary RICE exhaust so that the catalyst inlet temperature is greater than or equal to 750 °F and less than or equal to 1250 °F.1  Comply with any operating limitations approved by the Administrator.

Sources can petition the Administrator pursuant to the requirements of 40 CFR 63.8(f) for a different temperature range.

TABLE 2a TO SUBPART ZZZZ OF PART 63—EMISSION LIMITATIONS FOR NEW AND RECONSTRUCTED 2SLB AND COMPRESSION IGNITION STATIONARY RICE >500 HP AND NEW AND RECONSTRUCTED 4SLB STATIONARY RICE ≥250 HP LOCATED AT A MAJOR SOURCE OF HAP EMISSIONS

As stated in §§63.6600 and 63.6640, you must comply with the following emission limitations for new and reconstructed lean burn and new and reconstructed compression ignition stationary RICE at 100 percent load plus or minus 10 percent:

For each	You must meet the following emission limitation, except during periods of start-up	During periods of startup you must
1. 2SLB stationary RICE	Reduce CO emissions by 58 percent or more; or b. Limit concentration of formaldehyde in the stationary RICE exhaust to 12 ppmvd or less at 15 percent O <sub>2</sub> . If you commenced construction or reconstruction between December 19, 2002 and June 15, 2004, you may limit concentration of formaldehyde to 17 ppmvd or less at 15 percent O <sub>2</sub> until June 15, 2007.	Minimize the engine's time spent at idle and minimize the engine's startup time at startup to a period needed for appropriate and safe loading of the engine, not to exceed 30 minutes, after which time the non-startup emission limitations apply.1
2. 4SLB stationary RICE	Reduce CO emissions by 93 percent or more; or     b. Limit concentration of formaldehyde in the stationary RICE exhaust to 14 ppmvd or less at 15 percent O <sub>2</sub> .	
3. CI stationary RICE	a. Reduce CO emissions by 70 percent or more; or b. Limit concentration of formaldehyde in the stationary RICE exhaust to 580 ppbvd or less at 15 percent O <sub>2</sub> .	

<sup>&</sup>lt;sup>1</sup> Sources can petition the Administrator pursuant to the requirements of 40 CFR 63.6(g) for alternative work practices.

[75 FR 9680, Mar. 3, 2010]

TABLE 2b TO SUBPART ZZZZ OF PART 63—OPERATING LIMITATIONS FOR NEW AND RECONSTRUCTED 2SLB AND CI STATIONARY RICE >500 HP LOCATED AT A MAJOR SOURCE OF HAP EMISSIONS, NEW AND RECONSTRUCTED 4SLB STATIONARY RICE ≥250 HP LOCATED AT A MAJOR SOURCE OF HAP EMISSIONS, EXISTING CI STATIONARY RICE >500 HP

As stated in §§63.6600, 63.6601, 63.6603, 63.6630, and 63.6640, you must comply with the following operating limitations for new and reconstructed 2SLB and CI stationary RICE >500 HP located at a major source of HAP emissions; new and reconstructed 4SLB stationary RICE >2250 HP located at a major source of HAP emissions; and existing CI stationary RICE >500 HP:

For each . . .

You must meet the following operating limitation, except during periods of startup . . .

- New and reconstructed 2SLB and CI stationary RICE >500 HP located at a major source of HAP emissions and new and reconstructed 4SLB stationary RICE ≥250 HP located at a major source of HAP emissions complying with the requirement to reduce CO emissions and using an oxidation catalyst; and
- New and reconstructed 2SLB and CI stationary RICE >500 HP located at a major source of HAP emissions and new and reconstructed 4SLB stationary RICE ≥250 HP located at a major source of HAP emissions complying with the requirement to limit the concentration of formaldehyde in the stationary RICE exhaust and using an oxidation catalyst.
- tionary RICE exhaust and using an oxidation catalyst.

  2. Existing CI stationary RICE >500 HP complying with the requirement to limit or reduce the concentration of CO in the stationary RICE exhaust and using an oxidation catalyst.
- a. maintain your catalyst so that the pressure drop across the catalyst does not change by more than 2 inches of water at 100 percent load plus or minus 10 percent from the pressure drop across the catalyst that was measured during the initial performance test; and
- b. maintain the temperature of your stationary RICE exhaust so that the catalyst inlet temperature is greater than or equal to 450 °F and less than or equal to 1350 °F.¹
- a. maintain your catalyst so that the pressure drop across the catalyst does not change by more than 2 inches of water from the pressure drop across the catalyst that was measured during the initial performance test; and
   b. maintain the temperature of your stationary RICE exhaust
- b. maintain the temperature of your stationary RICE exhaust so that the catalyst inlet temperature is greater than or equal to 450 °F and less than or equal to 1350 °F.1

#### Pt. 63, Subpt. ZZZZ, Table 2c

#### 40 CFR Ch. I (7-1-13 Edition)

For each	You must meet the following operating limitation, except during periods of startup.
3. New and reconstructed 2SLB and CI stationary RICE >500 HP located at a major source of HAP emissions and new and reconstructed 4SLB stationary RICE ≥250 HP located at a major source of HAP emissions complying with the requirement to reduce CO emissions and not using an oxidation catalyst; and New and reconstructed 2SLB and CI stationary RICE >500 HP located at a major source of HAP emissions and new and reconstructed 4SLB stationary RICE ≥250 HP located at a major source of HAP emissions complying with the requirement to limit the concentration of formaldehyde in the stationary RICE exhaust and not using an oxidation catalyst; and existing CI stationary RICE >500 HP complying with the requirement to limit or reduce the concentration of CO in the stationary RICE exhaust and not using an oxidation catalyst.	

<sup>&</sup>lt;sup>1</sup> Sources can petition the Administrator pursuant to the requirements of 40 CFR 63.8(f) for a different temperature range.

[78 FR 6707, Jan. 30, 2013]

TABLE 2C TO SUBPART ZZZZ OF PART 63—REQUIREMENTS FOR EXISTING COMPRESSION IGNITION STATIONARY RICE LOCATED AT A MAJOR SOURCE OF HAP EMISSIONS AND EXISTING SPARK IGNITION STATIONARY RICE ≤500 HP LOCATED AT A MAJOR SOURCE OF HAP EMISSIONS

As stated in §§ 63.6600, 63.6602, and 63.6640, you must comply with the following requirements for existing compression ignition stationary RICE located at a major source of HAP emissions and existing spark ignition stationary RICE  $\leq$ 500 HP located at a major source of HAP emissions:

For each	You must meet the following requirement, except during periods of startup	During periods of startup you must
Emergency stationary CI RICE and black start stationary CI RICE <sup>1</sup> .	Change oil and filter every 500 hours of operation or annually, whichever comes first. <sup>2</sup> Inspect air cleaner every 1,000 hours of operation or annually, whichever comes first, and replace as necessary.     Inspect all hoses and belts every 500 hours of operation or annually, whichever comes first, and replace as necessary. <sup>3</sup>	Minimize the engine's time spent at idle and minimize the engine's startup time at startup to a period needed for appropriate and safe loading of the engine, not to exceed 30 minutes, after which time the non-startup emission limitations apply. <sup>3</sup>
Non-Emergency, non-black start sta- tionary CI RICE <100 HP.	a. Change oil and filter every 1,000 hours of operation or annually, whichever comes first. <sup>2</sup> b. Inspect air cleaner every 1,000 hours of operation or annually, whichever comes first, and replace as necessary; c. Inspect all hoses and belts every 500 hours of operation or annually, whichever comes first, and replace as necessary. <sup>3</sup>	
<ol> <li>Non-Emergency, non-black start CI stationary RICE 100≤HP≤300 HP.</li> </ol>	Limit concentration of CO in the sta- tionary RICE exhaust to 230 ppmvd or less at 15 percent O <sub>2</sub> .	
<ol> <li>Non-Emergency, non-black start CI sta- tionary RICE 300<hp≤500.< li=""> </hp≤500.<></li></ol>	a. Limit concentration of CO in the sta- tionary RICE exhaust to 49 ppmvd or less at 15 percent O <sub>2</sub> ; or b. Reduce CO emissions by 70 percent or more.	
<ol> <li>Non-Emergency, non-black start sta- tionary CI RICE &gt;500 HP.</li> </ol>	a. Limit concentration of CO in the stationary RICE exhaust to 23 ppmvd or less at 15 percent O <sub>2</sub> ; or b. Reduce CO emissions by 70 percent or more.	

#### Pt. 63, Subpt. ZZZZ, Table 2c

For each	You must meet the following requirement, except during periods of startup	During periods of startup you must
Emergency stationary SI RICE and black start stationary SI RICE. <sup>1</sup>	Change oil and filter every 500 hours of operation or annually, whichever comes first; <sup>2</sup> Inspect spark plugs every 1,000 hours of operation or annually, whichever comes first, and replace as necessary; c. Inspect all hoses and belts every 500 hours of operation or annually, whichever comes first, and replace as necessary, <sup>3</sup>	
<ol> <li>Non-Emergency, non-black start sta- tionary SI RICE &lt;100 HP that are not 2SLB stationary RICE.</li> </ol>	a. Change oil and filter every 1,440 hours of operation or annually, whichever comes first;     b. Inspect spark plugs every 1,440 hours of operation or annually, whichever comes first, and replace as necessary;     c. Inspect all hoses and belts every 1,440 hours of operation or annually, whichever comes first, and replace as necessary. <sup>3</sup>	
Non-Emergency, non-black start 2SLB stationary SI RICE <100 HP.	Change oil and filter every 4,320 hours of operation or annually, whichever comes first; <sup>2</sup> b. Inspect spark plugs every 4,320 hours of operation or annually, whichever comes first, and replace as necessary, c. Inspect all hoses and belts every 4,320 hours of operation or annually, whichever comes first, and replace as necessary. <sup>3</sup>	
<ol> <li>Non-emergency, non-black start 2SLB stationary RICE 100≤HP≤500.</li> </ol>		
10. Non-emergency, non-black start 4SLB stationary RICE 100≤HP≤500.	Limit concentration of CO in the stationary RICE exhaust to 47 ppmvd or less at 15 percent O <sub>2</sub> .	
<ol> <li>Non-emergency, non-black start 4SRB stationary RICE 100≤HP≤500.</li> </ol>	Limit concentration of formaldehyde in the stationary RICE exhaust to 10.3 ppmvd or less at 15 percent O <sub>2</sub> .	
12. Non-emergency, non-black start stationary RICE 100≤HP≤500 which combusts landfill or digester gas equivalent to 10 percent or more of the gross heat input on an annual basis.		

<sup>&</sup>quot;If an emergency engine is operating during an emergency and it is not possible to shut down the engine in order to perform the work practice requirements on the schedule required in Table 2c of this subpart, or if performing the work practice on the required schedule would otherwise pose an unacceptable risk under federal, state, or local law has abated. The work practice should be performed as soon as practicable after the emergency has ended or the unacceptable risk under federal, state, or local law has abated. Sources must report any failure to perform the work practice on the schedule required and the federal, state or local law has abated. Sources must report any failure to perform the work practice on the schedule required and the federal, state or local law under which the risk was deemed unacceptable.

2 Sources have the option to utilize an oil analysis program as described in §63.6625(i) or (j) in order to extend the specified oil change requirement in Table 2c of this subpart.

3 Sources can petition the Administrator pursuant to the requirements of 40 CFR 63.6(g) for alternative work practices.

[78 FR 6708, Jan. 30, 2013, as amended at 78 FR 14457, Mar. 6, 2013]

### 40 CFR Ch. I (7-1-13 Edition)

Table 2d to Subpart ZZZZ of Part 63—Requirements for Existing Stationary RICE Located at Area Sources of HAP Emissions

As stated in  $\S 63.6603$  and 63.6640, you must comply with the following requirements for existing stationary RICE located at area sources of HAP emissions:

For each	You must meet the following requirement, except during periods of startup	During periods of startup you must
Non-Emergency, non-black start CI stationary RICE ≤300 HP.	Change oil and filter every 1,000 hours of operation or annually, which-ever comes first;     Inspect air cleaner every 1,000 hours of operation or annually, whichever comes first, and replace as necessary.     Inspect all hoses and belts every 500 hours of operation or annually, whichever comes first, and replace as necessary.	Minimize the engine's time spent at idle and minimize the engine's startup time at startup to a period needed for appropriate and safe loading of the engine, not to exceed 30 minutes, after which time the non-startup emission limitations apply.
2. Non-Emergency, non-black start CI stationary RICE 300 <hp≤500.< td=""><td>a. Limit concentration of CO in the stationary RICE exhaust to 49 ppmvd at 15 percent O<sub>2</sub>; or b. Reduce CO emissions by 70 percent</td><td></td></hp≤500.<>	a. Limit concentration of CO in the stationary RICE exhaust to 49 ppmvd at 15 percent O <sub>2</sub> ; or b. Reduce CO emissions by 70 percent	
3. Non-Emergency, non-black start CI stationary RICE >500 HP.	or more.  a. Limit concentration of CO in the stationary RICE exhaust to 23 ppmvd at 15 percent O <sub>2</sub> ; or  b. Reduce CO emissions by 70 percent or more.	
Emergency stationary CI RICE and black start stationary CI RICE. <sup>2</sup>	a. Change oil and filter every 500 hours of operation or annually, whichever comes first;     b. Inspect air cleaner every 1,000 hours of operation or annually, whichever comes first, and replace as necessary; and     c. Inspect all hoses and belts every 500 hours of operation or annually, whichever comes first, and replace as necessary.	
5. Emergency stationary SI RICE; black start stationary SI RICE; non-emergency, non-black start 4SLB stationary RICE >500 HP that operate 24 hours or less per calendar year; non-emergency, non-black start 4SRB stationary RICE >500 HP that operate 24 hours or less per calendar year. <sup>2</sup>	essary.  a. Change oil and filter every 500 hours of operation or annually, whichever comes first; ':  b. Inspect spark plugs every 1,000 hours of operation or annually, whichever comes first, and replace as necessary; and  c. Inspect all hoses and belts every 500 hours of operation or annually, whichever comes first, and replace as necessary.	
Non-emergency, non-black start 2SLB stationary RICE.	a. Change oil and filter every 4,320 hours of operation or annually, whichever comes first;  b. Inspect spark plugs every 4,320 hours of operation or annually, whichever comes first, and replace as necessary; and  c. Inspect all hoses and belts every 4,320 hours of operation or annually, whichever comes first, and replace as	
Non-emergency, non-black start 4SLB stationary RICE s500 HP.	necessary.  a. Change oil and filter every 1,440 hours of operation or annually, whichever comes first;  b. Inspect spark plugs every 1,440 hours of operation or annually, whichever comes first, and replace as necessary; and  c. Inspect all hoses and belts every 1,440 hours of operation or annually, whichever comes first, and replace as necessary.	

For each	You must meet the following requirement, except during periods of startup	During periods of startup you must
Non-emergency, non-black start 4SLB remote stationary RICE >500 HP.	a. Change oil and filter every 2,160 hours of operation or annually, whichever comes first;     b. Inspect spark plugs every 2,160 hours of operation or annually, whichever comes first, and replace as necessary; and     c. Inspect all hoses and belts every 2,160 hours of operation or annually, whichever comes first, and replace as necessary.	
<ol> <li>Non-emergency, non-black start 4SLB stationary RICE &gt;500 HP that are not remote stationary RICE and that oper- ate more than 24 hours per calendar year.</li> </ol>	Install an oxidation catalyst to reduce HAP emissions from the stationary RICE.	
10. Non-emergency, non-black start 4SRB stationary RICE ≤500 HP.	a. Change oil and filter every 1,440 hours of operation or annually, which-ever comes first;     b. Inspect spark plugs every 1,440 hours of operation or annually, whichever comes first, and replace as necessary; and	
11. Non-emergency, non-black start 4SRB remote stationary RICE >500 HP.	c. Inspect all hoses and belts every 1,440 hours of operation or annually, whichever comes first, and replace as necessary.  a. Change oil and filter every 2,160 hours of operation or annually, whichever comes first;  b. Inspect spark plugs every 2,160 hours of operation or annually, whichever comes first, and replace as necessary; and	
12. Non-emergency, non-black start 4SRB stationary RICE >500 HP that are not remote stationary RICE and that operate more than 24 hours per calendar year.	Inspect all hoses and belts every 2,160 hours of operation or annually, whichever comes first, and replace as necessary.     Install NSCR to reduce HAP emissions from the stationary RICE.	
<ol> <li>Non-emergency, non-black start sta- tionary RICE which combusts landfill or digester gas equivalent to 10 percent or more of the gross heat input on an an- nual basis.</li> </ol>	hours of operation or annually, which- ever comes first; <sup>1</sup> b. Inspect spark plugs every 1,440 hours of operation or annually, whichever comes first, and replace as necessary; and	
	Inspect all hoses and belts every     1,440 hours of operation or annually, whichever comes first, and replace as necessary.	

[78 FR 6709, Jan. 30, 2013]

Sources have the option to utilize an oil analysis program as described in §63.6625(i) or (j) in order to extend the specified oil change requirement in Table 2d of this subpart.

2lf an emergency engine is operating during an emergency and it is not possible to shut down the engine in order to perform the management practice requirements on the schedule required in Table 2d of this subpart, or if performing the management practice on the required schedule would otherwise pose an unacceptable risk under federal, state, or local law, the management practice can be delayed until the emergency is over or the unacceptable risk under federal, state, or local law has abated. The management practice should be performed as soon as practicable after the emergency has ended or the unacceptable risk under federal, state, or local law has abated. Sources must report any failure to perform the management practice on the schedule required and the federal, state or local law under which the risk was deemed unacceptable.

### 40 CFR Ch. I (7-1-13 Edition)

### Pt. 63, Subpt. ZZZZ, Table 3

TABLE 3 TO SUBPART ZZZZ OF PART 63—SUBSEQUENT PERFORMANCE TESTS

As stated in  $\S 63.6615$  and 63.6620, you must comply with the following subsequent performance test requirements:

For each	Complying with the requirement to	You must
New or reconstructed 2SLB stationary RICE >500 HP located at major sources; new or reconstructed 4SLB stationary RICE ≥250 HP located at major sources; and new or reconstructed CI stationary RICE >500 HP located at major sources.	Reduce CO emissions and not using a CEMS.	Conduct subsequent performance tests semiannually. 1
2. 4SRB stationary RICE ≥5,000 HP located at major sources.	Reduce formaldehyde emissions	Conduct subsequent performance tests semiannually.1
<ol> <li>Stationary RICE &gt;500 HP located at major sources and new or recon- structed 4SLB stationary RICE 250<hp< li=""> <li>Stopher</li> </hp<></li></ol>	Limit the concentration of formaldehyde in the stationary RICE exhaust.	Conduct subsequent performance tests semiannually.1
<ol> <li>Existing non-emergency, non-black start CI stationary RICE &gt;500 HP that are not limited use stationary RICE.</li> </ol>	Limit or reduce CO emissions and not using a CEMS.	Conduct subsequent performance tests every 8,760 hours or 3 years, which- ever comes first.
<ol> <li>Existing non-emergency, non-black start CI stationary RICE &gt;500 HP that are limited use stationary RICE.</li> </ol>		Conduct subsequent performance tests every 8,760 hours or 5 years, which- ever comes first.

<sup>&</sup>lt;sup>1</sup> After you have demonstrated compliance for two consecutive tests, you may reduce the frequency of subsequent performance tests to annually. If the results of any subsequent annual performance test indicate the stationary RICE is not in compliance with the CO or formaldehyde emission limitation, or you deviate from any of your operating limitations, you must resume semi-annual performance tests.

[78 FR 6711, Jan. 30, 2013]

TABLE 4 TO SUBPART ZZZZ OF PART 63—REQUIREMENTS FOR PERFORMANCE TESTS

As stated in §§63.6610, 63.6611, 63.6612, 63.6620, and 63.6640, you must comply with the following requirements for performance tests for stationary RICE:

For each	Complying with the requirement to	You must	Using	According to the following requirements
2SLB, 4SLB, and CI sta- tionary RICE.	a. reduce CO emissions.	i. Measure the O <sub>2</sub> at the inlet and outlet of the control device; and	(1) Method 3 or 3A or 3B of 40 CFR part 60, appendix A, or ASTM Method D6522-00 (Reapproved 2005).**c	(a) Measurements to determine O <sub>2</sub> must be made at the same time as the measurements for CO concentration.
		ii. Measure the CO at the inlet and the outlet of the control device.	(1) ASTM D6522-00 (Reapproved 2005) and or Method 10 of 40 CFR part 60, appendix A.	(a) The CO concentration must be at 15 percent O <sub>2</sub> , dry basis.
2. 4SRB sta- tionary RICE.	a. reduce form- aldehyde emissions.	Select the sampling port location and the number of traverse points; and	(1) Method 1 or 1A of 40 CFR part 60, appendix A § 63.7(d)(1)(i).	(a) sampling sites must be located at the inlet and outlet of the control device.
		ii. Measure O <sub>2</sub> at the inlet and outlet of the control device; and	(1) Method 3 or 3A or 3B of 40 CFR part 60, appendix A, or ASTM Method D6522-00 (Reapproved 2005).a	(a) measurements to determine O <sub>2</sub> concentration must be made at the same time as the measurements for formaldehyde or THC concentration.
		iii. Measure moisture content at the inlet and outlet of the control device; and	(1) Method 4 of 40 CFR part 60, appendix A, or Test Method 320 of 40 CFR part 63, appendix A, or ASTM D 6348-03.*	(a) measurements to determine moisture content must be made at the same time and location as the measurements for formaldehyde or THC concentration.

### Pt. 63, Subpt. ZZZZ, Table 4

### **Environmental Protection Agency**

For each	Complying with the requirement to	You must	Using	According to the following requirements .
		iv. If demonstrating compli- ance with the formalde- hyde percent reduction re- quirement, measure form- aldehyde at the inlet and the outlet of the control device.	(1) Method 320 or 323 of 40 CFR part 63, appendix A; or ASTM D6348-03.º provided in ASTM D6348-03 Annex A5 (Analyte Spiking Technique), the percent R must be greater than or equal to 70 and less than or equal to 130.	(a) formaldehyde concentration must be at 15 percent O <sub>2</sub> , dry basis. Results of this test consist of the average of the three 1-hour or longer runs.
		v. If demonstrating compli- ance with the THC percent reduction requirement, measure THC at the inlet and the outlet of the con- trol device.	(1) Method 25A, reported as propane, of 40 CFR part 60, appendix A.	(a) THC concentration must be at 15 percent O₂, dry basis. Results of this test consist of the average of the three 1-hour or longer runs.
3. Stationary RICE.	a. limit the con- centration of formalde- hyde or CO in the sta- tionary RICE exhaust.	i. Select the sampling port location and the number of traverse points; and	(1) Method 1 or 1A of 40 CFR part 60, appendix A § 63.7(d)(1)(i).	(a) If using a control device, the sampling site must be located at the outlet of the control device.
		ii. Determine the O <sub>2</sub> con- centration of the stationary RICE exhaust at the sam- pling port location; and	(1) Method 3 or 3A or 3B of 40 CFR part 60, appendix A, or ASTM Method D6522–00 (Reapproved 2005).*	(a) measurements to determine O <sub>2</sub> concentration must be made at the same time and location as the measurements for formaldehyde or CO concentration.
		iii. Measure moisture content of the stationary RICE ex- haust at the sampling port location; and	(1) Method 4 of 40 CFR part 60, appendix A, or Test Method 320 of 40 CFR part 63, appendix A, or ASTM D 6348-03.ª	(a) measurements to determine moisture content must be made at the same time and location as the measurements for formaldehyde or CO concentration.
		iv. Measure formaldehyde at the exhaust of the sta- tionary RICE; or	(1) Method 320 or 323 of 40 CFR part 63, appendix A; or ASTM D6348-03, arovided in ASTM D6348-03 Annex A5 (Analyte Spiking Technique), the percent R must be greater than or equal to 70 and less than	(a) Formaldehyde concentra- tion must be at 15 percent O <sub>2</sub> , dry basis. Results of this test consist of the av- erage of the three 1-hour or longer runs.
		v. measure CO at the exhaust of the stationary RICE.	or equal to 730.  (1) Method 10 of 40 CFR part 60, appendix A, ASTM Method D6522-00 (2005)," Method 320 of 40 CFR part 63, appendix A, or ASTM D6348-03.*	(a) CO concentration must be at 15 percent O <sub>2</sub> , dry basis. Results of this test consist of the average of the three 1-hour or longer runs.

a Incorporated by reference, see 40 CFR 63.14. You may also obtain copies from University Microfilms International, 300 North Zeeb Road, Ann Arbor, MI 48108.
b You may also use Method 320 of 40 CFR part 63, appendix A, or ASTM D6348–03.
c ASTM-D6522–00 (2005) may be used to test both CI and SI stationary RICE.

[78 FR 6711, Jan. 30, 2013]

# Pt. 63, Subpt. ZZZZ, Table 5

Table 5 to Subpart ZZZZ of Part 63—Initial Compliance With Emission Limitations, Operating Limitations, and Other Requirements

As stated in  $\S\S63.6612$ , 63.6625 and 63.6630, you must initially comply with the emission and operating limitations as required by the following:

For each	Complying with the requirement to	You have demonstrated initial compliance if
1. New or reconstructed non-emergency 2SLB stationary RICE >500 HP located at a major source of HAP, new or reconstructed non-emergency 4SLB stationary RICE ≥250 HP located at a major source of HAP, non-emergency stationary CI RICE >500 HP located at a major source of HAP, and existing non-emergency stationary CI RICE >500 HP located at an area source of HAP.  2. Non-emergency stationary CI RICE >500 HP located at a major source of HAP, and existing non-emergency stationary CI RICE >500 HP located at a major source of HAP, and existing non-emergency stationary CI RICE >500 HP located at an area source of HAP.	Reduce CO emissions and using oxidation catalyst, and using a CPMS.      Limit the concentration of CO, using oxidation catalyst, and using a CPMS.	i. The average reduction of emissions of CO determined from the initial performance test achieves the required CO percent reduction; and ii. You have installed a CPMS to continuously monitor catalyst inlet temperature according to the returnments in § 63.6625(b); and iii. You have recorded the catalyst pressure drop and catalyst inlet temperature during the initial performance test. i. The average CO concentration determined from the initial performance test is less than or equal to the CO emission limitation; and
302 302 30 1 M .		ii. You have installed a CPMS to continuously monitor catalyst inlet temperature according to the requirements in § 63.6625(b); and iii. You have recorded the catalyst pressure drop and catalyst inlet temperature during the initial performance test.
New or reconstructed non-emergency 2SLB stationary RICE >500 HP located at a major source of HAP, new or reconstructed non-emergency 4SLB stationary RICE >250 HP located at a major source of HAP, non-emergency stationary CI RICE >500 HP located at a major source of HAP, and existing non-emergency stationary CI RICE >500 HP located at an area source of HAP.	Reduce CO emissions and not using oxidation catalyst.	i. The average reduction of emissions of CO determined from the initial performance test achieves the required CO percent reduction; and ii. You have installed a CPMS to continuously monitor operating parameters approved by the Administrator (if any) according to the requirements in §63.662(b); and iii. You have recorded the approved operating parameters (if any) during the initial performance test.
<ol> <li>Non-emergency stationary CI RICE &gt;500 HP located at a major source of HAP, and existing non-emergency sta- tionary CI RICE &gt;500 HP located at an area source of HAP.</li> </ol>	a. Limit the concentration of CO, and not using oxidation catalyst.	i. The average CO concentration determined from the initial performance test is less than or equal to the CO emission limitation; and ii. You have installed a CPMS to continuously monitor operating parameters approved by the Administrator (if any) according to the requirements in §63.6625(b); and iii. You have recorded the approved operating parameters (if any) during the initial performance test.
5. New or reconstructed non-emergency 2SLB stationary RICE >500 HP located at a major source of HAP, new or reconstructed non-emergency 4SLB stationary RICE ≥250 HP located at a major source of HAP, non-emergency stationary CI RICE >500 HP located at a major source of HAP, and existing non-emergency stationary CI RICE >500 HP located at an area source of	a. Reduce CO emissions, and using a CEMS.	i. You have installed a CEMS to continuously monitor CO and either O <sub>2</sub> or CO <sub>2</sub> at both the finlet and outlet of the oxidation catalyst according to the requirements in § 63.6625(a); and ii. You have conducted a performance evaluation of your CEMS using PS 3 and 4A of 40 CFR part 60, appendix B; and
нар.		iii. The average reduction of CO cal- culated using § 63.6620 equals or ex- ceeds the required percent reduction. The initial test comprises the first 4- hour period after successful validation of the CEMS. Compliance is based on the average percent reduction achieved during the 4-hour period.

# Pt. 63, Subpt. ZZZZ, Table 5

For each	Complying with the requirement to	You have demonstrated initial compliance if
<ol> <li>Non-emergency stationary CI RICE &gt;500 HP located at a major source of HAP, and existing non-emergency sta- tionary CI RICE &gt;500 HP located at an area source of HAP.</li> </ol>	a. Limit the concentration of CO, and using a CEMS.	i. You have installed a CEMS to continuously monitor CO and either O2 or CO2 at the outlet of the oxidation catalyst according to the requirements in §63.6625(a); and ii. You have conducted a performance evaluation of your CEMS using PS 3 and 4A of 40 CFR part 60, appendix B; and iii. The average concentration of CO calculated using §63.6620 is less than or equal to the CO emission limitation. The initial test comprises the first 4-hour period after successful validation of the CEMS. Compliance is based on the average concentration measured during the 4-hour period.
<ol> <li>Non-emergency 4SRB stationary RICE &gt;500 HP located at a major source of HAP.</li> </ol>	Reduce formaldehyde emissions and using NSCR.	The average reduction of emissions of formaldehyde determined from the initial performance test is equal to or greater than the required formaldehyde percent reduction, or the average reduction of emissions of THC determined from the initial performance test is equal to or greater than 30 percent; and     ii. You have installed a CPMS to continuously monitor catalyst inlet temperature according to the requirements in § 63.6625(b); and     iii. You have recorded the catalyst pressure drop and catalyst inlet temperature during the initial performance test.
Non-emergency 4SRB stationary RICE >500 HP located at a major source of HAP.	Reduce formaldehyde emissions and not using NSCR.	i. The average reduction of emissions of formaldehyde determined from the initial performance test is equal to or greater than the required formaldehyde percent reduction or the average reduction of emissions of THC determined from the initial performance test is equal to or greater than 30 percent; and ii. You have installed a CPMS to continuously monitor operating parameters approved by the Administrator (if any) according to the requirements in § 63.6625(b); and iii. You have recorded the approved operating parameters (if any) during the
<ol> <li>New or reconstructed non-emergency stationary RICE &gt;500 HP located at a major source of HAP, new or recon- structed non-emergency 4SLB sta- tionary RICE 250SHP≤500 located at a major source of HAP, and existing non- emergency 4SRB stationary RICE &gt;500 HP located at a major source of HAP.</li> </ol>	a. Limit the concentration of formaldehyde in the stationary RICE exhaust and using oxidation catalyst or NSCR.	initial performance test.  i. The average formaldehyde concentration, corrected to 15 percent O <sub>2</sub> , dry basis, from the three test runs is less than or equal to the formaldehyde emission limitation; and  ii. You have installed a CPMS to continuously monitor catalyst inlet temperature according to the requirements in § 63.6625(b); and  iii. You have recorded the catalyst pressure drop and catalyst inlet temperature during the initial performance test.
10. New or reconstructed non-emergency stationary RICE >500 HP located at a major source of HAP, new or reconstructed non-emergency 4SLB stationary RICE 250≤HP≤500 located at a major source of HAP, and existing non-emergency 4SRB stationary RICE >500 HP located at a major source of HAP.	a. Limit the concentration of formaldehyde in the stationary RICE exhaust and not using oxidation catalyst or NSCR.	i. The average formaldehyde concentration, corrected to 15 percent O <sub>2</sub> , dry basis, from the three test runs is less than or equal to the formaldehyde emission limitation; and ii. You have installed a CPMS to continuously monitor operating parameters approved by the Administrator (if any) according to the requirements in § 63.6625(b); and

### 40 CFR Ch. I (7-1-13 Edition)

For each	Complying with the requirement to	You have demonstrated initial compli- ance if
11. Existing non-emergency stationary RICE 100≾HP≤500 located at a major source of HAP, and existing non-emergency stationary CI RICE 300≺HP≤500 located at an area source of HAP.	a. Reduce CO emissions	iii. You have recorded the approved op- erating parameters (if any) during the initial performance test. i. The average reduction of emissions of CO or formaldehyde, as applicable de- termined from the initial performance test is equal to or greater than the re- quired CO or formaldehyde, as appli-
12. Existing non-emergency stationary RICE 100≤HP≤500 located at a major source of HAP, and existing non-emergency stationary CI RICE 300 <hp≤500 an="" area="" at="" hap.<="" located="" of="" source="" td=""><td>a. Limit the concentration of formaldehyde or CO in the stationary RICE exhaust.</td><td>cable, percent reduction.  i. The average formaldehyde or CO concentration, as applicable, corrected to 15 percent O<sub>2</sub>, dry basis, from the three test runs is less than or equal to the formaldehyde or CO emission limitation, as applicable.</td></hp≤500>	a. Limit the concentration of formaldehyde or CO in the stationary RICE exhaust.	cable, percent reduction.  i. The average formaldehyde or CO concentration, as applicable, corrected to 15 percent O <sub>2</sub> , dry basis, from the three test runs is less than or equal to the formaldehyde or CO emission limitation, as applicable.
13. Existing non-emergency 4SLB stationary RICE >500 HP located at an area source of HAP that are not remote stationary RICE and that are operated more than 24 hours per calendar year.	a. Install an oxidation catalyst	i. You have conducted an initial compli- ance demonstration as specified in § 63.6630(e) to show that the average reduction of emissions of CO is 93 percent or more, or the average CO concentration is less than or equal to 47 ppmvd at 15 percent O <sub>2</sub> ;
		II. You have installed a CPMS to continuously monitor catalyst inlet temperature according to the requirements in §63.6625(b), or you have installed equipment to automatically shut down the engine if the catalyst inlet temperature exceeds 1350 °F.
<ol> <li>Existing non-emergency 4SRB sta- tionary RICE &gt;500 HP located at an area source of HAP that are not remote stationary RICE and that are operated more than 24 hours per calendar year.</li> </ol>	a. Install NSCR	<ol> <li>You have conducted an initial compli- ance demonstration as specified in §63.6630(e) to show that the average reduction of emissions of CO is 75 percent or more, the average CO con- centration is less than or equal to 270 ppmvd at 15 percent O<sub>2</sub>, or the aver- age reduction of emissions of THC is 30 percent or more;</li> </ol>
		ii. You have installed a CPMS to con- tinuously monitor catalyst inlet tem- perature according to the require- ments in §63.6625(b), or you have in- stalled equipment to automatically shut down the engine if the catalyst inlet temperature exceeds 1250 °F.

[78 FR 6712, Jan. 30, 2013]

# Table 6 to Subpart ZZZZ of Part 63—Continuous Compliance With Emission Limitations, and Other Requirements

As stated in  $\S63.6640$ , you must continuously comply with the emissions and operating limitations and work or management practices as required by the following:

For each	Complying with the requirement to	You must demonstrate continuous compliance by			
1. New or reconstructed non-emergency 2SLB stationary RICE >500 HP located at a major source of HAP, new or reconstructed non-emergency 4SLB stationary RICE ≥250 HP located at a major source of HAP, and new or reconstructed non-emergency CI stationary RICE >500 HP located at a major source of HAP.	Reduce CO emissions and using an oxidation catalyst, and using a CPMS.	Conducting semiannual performance tests for CO to demonstrate that the required CO percent reduction is achieved *; and     Collecting the catalyst inlet temperature data according to §63.6625(b); and     iii. Reducing these data to 4-hour rolling averages; and     iv. Maintaining the 4-hour rolling averages within the operating limitations for the catalyst inlet temperature; and			

# Pt. 63, Subpt. ZZZZ, Table 6

For each	Complying with the requirement to	You must demonstrate continuous com- pliance by
		Measuring the pressure drop across the catalyst once per month and demonstrating that the pressure drop across the catalyst is within the operating limitation established during the performance test.
<ol> <li>New or reconstructed non-emergency 2SLB stationary RICE &gt;500 HP located at a major source of HAP, new or re- constructed non-emergency 4SLB sta- tionary RICE ≥250 HP located at a major source of HAP, and new or re- constructed non-emergency CI sta- tionary RICE &gt;500 HP located at a major source of HAP.</li> </ol>	Reduce CO emissions and not using an oxidation catalyst, and using a CPMS.	Conducting semiannual performance tests for CO to demonstrate that the required CO percent reduction is achieved and     ii. Collecting the approved operating parameter (if any) data according to §63.6625(b); and     iiii. Reducing these data to 4-hour rolling averages; and     iv. Maintaining the 4-hour rolling averages within the operating limitations for the operating parameters estab.
3. New or reconstructed non-emergency 2SLB stationary RICE >500 HP located at a major source of HAP, new or reconstructed non-emergency 4SLB stationary RICE ≥250 HP located at a major source of HAP, new or reconstructed non-emergency stationary CI RICE >500 HP located at a major source of HAP, and existing non-emergency stationary CI RICE >500 HP.	a. Reduce CO emissions or limit the concentration of CO in the stationary RICE exhaust, and using a CEMS.	lished during the performance test.  i. Collecting the monitoring data according to §63.6625(a), reducing the measurements to 1-hour averages calculating the percent reduction to concentration of CO emissions according to §63.6620; and  ii. Demonstrating that the catalys achieves the required percent reduction of CO emissions over the 4-hou averaging period, or that the emission remain at or below the CO concentration limit; and
Non-emergency 4SRB stationary RICE >500 HP located at a major source of HAP.	Reduce formaldehyde emissions and using NSCR.	<ul> <li>iii. Conducting an annual RATA of you CEMS using PS 3 and 4A of 40 CFF part 60, appendix B, as well as daily and periodic data quality checks in ac cordance with 40 CFR part 60, appen dix F, procedure 1.</li> <li>i. Collecting the catalyst inlet tempera ture data according to §63.6625(b) and</li> <li>iii. Reducing these data to 4-hour rolling averages; and</li> <li>iiii. Maintaining the 4-hour rolling averages within the operating limitations for the catalyst inlet temperature; and</li> <li>iv. Measuring the pressure drop across</li> </ul>
<ol> <li>Non-emergency 4SRB stationary RICE &gt;500 HP located at a major source of HAP.</li> </ol>	Reduce formaldehyde emissions and not using NSCR.	the catalyst once per month and dem- onstrating that the pressure drog across the catalyst is within the oper ating limitation established during the performance test.  i. Collecting the approved operating pa rameter (if any) data according to § 63.6625(b); and  ii. Reducing these data to 4-hour rolling averages; and  iii. Maintaining the 4-hour rolling aver
6. Non-emergency 4SRB stationary RICE with a brake HP ≥5,000 located at a major source of HAP.	a. Reduce formaldehyde emissions	ages within the operating limitations for the operating parameters established during the performance test.  Conducting semiannual performance tests for formaldehyde to demonstrate that the required formaldehyde percent reduction is achieved, or to demonstrate that the average reduction on emissions of THC determined from the performance test is equal to or greater

# 40 CFR Ch. 1 (7-1-13 Edition)

For each	Complying with the requirement to	You must demonstrate continuous compliance by
<ol> <li>New or reconstructed non-emergency stationary RICE &gt;500 HP located at a major source of HAP and new or recon- structed non-emergency 4SLB sta- tionary RICE 250≤HP≤500 located at a major source of HAP.</li> </ol>	Limit the concentration of formaldehyde in the stationary RICE exhaust and using oxidation catalyst or NSCR.	i. Conducting semiannual performance tests for formaldehyde to demonstrate that your emissions remain at or below the formaldehyde concentration limit*; and ii. Collecting the catalyst inlet temperature data according to §63.6625(b); and iii. Reducing these data to 4-hour rolling averages; and iv. Maintaining the 4-hour rolling averages within the operating limitations for the catalyst inlet temperature; and v. Measuring the pressure drop across the catalyst once per month and demonstrating that the pressure drop across the catalyst is within the operating limitation established during the performance test.
New or reconstructed non-emergency stationary RICE >500 HP located at a major source of HAP and new or reconstructed non-emergency 4SLB stationary RICE 250≤HP≤500 located at a major source of HAP.	Limit the concentration of formaldehyde in the stationary RICE exhaust and not using oxidation catalyst or NSCR.	i. Conducting semiannual performance tests for formaldehyde to demonstrate that your emissions remain at or below the formaldehyde concentration limita; and ii. Collecting the approved operating parameter (if any) data according to § 63.6625(b); and iii. Reducing these data to 4-hour rolling averages; and iv. Maintaining the 4-hour rolling averages within the operating limitations for the operating parameters established during the performance test.
9. Existing emergency and black start stationary RICE ≤500 HP located at a major source of HAP, existing non-emergency stationary RICE <100 HP located at a major source of HAP, existing emergency and black start stationary RICE located at an area source of HAP, existing non-emergency stationary CI RICE ≤300 HP located at an area source of HAP, existing non-emergency 2SLB stationary RICE located at an area source of HAP, existing non-emergency stationary SI RICE located at an area source of HAP, existing non-emergency stationary SI RICE located at an area source of HAP existing non-emergency 4SLB and 4SRB stationary RICE ≤500 HP located at an area source of HAP, existing non-emergency 4SLB and 4SRB stationary RICE >500 HP located at an area source of HAP existing non-emergency 4SLB and 4SRB stationary RICE >500 HP located at an area source of HAP existing non-emergency 4SLB and 4SRB stationary RICE >500 HP located at an area source of HAP existing non-emergency 4SLB and 4SRB stationary RICE >500 HP located at an area source of HAP existing non-emergency 4SLB and 4SRB stationary RICE >500 HP located at an area source of HAP existing non-emergency 4SLB and 4SRB stationary RICE >500 HP located at an area source of existing non-emergency 4SLB and 4SRB stationary RICE >500 HP located at an area source of existing non-emergency 4SLB and 4SRB stationary RICE >500 HP located at an area source of existing non-emergency 4SLB and 4SRB stationary RICE >500 HP located at an area source of existing non-emergency 4SLB and 4SRB stationary RICE >500 HP located at an area source of existing non-emergency 4SLB and 4SRB stationary RICE >500 HP located at an area source of existing non-emergency 4SLB and 4SRB stationary RICE >500 HP located at an area source of existing non-emergency 4SLB and 4SRB stationary RICE >500 HP located at an area source of existing non-emergency 4SLB and 4SRB stationary RICE >500 HP located at an area source of existing non-emergency 4SLB and 4SRB stationary RICE >500 HP located at an area source of ex		Operating and maintaining the stationary RICE according to the manufacturer's emission-related operation and maintenance instructions; or     Develop and follow your own maintenance plan which must provide to the extent practicable for the maintenance and operation of the engine in a maner consistent with good air pollution control practice for minimizing emissions.
HAP that are remote stationary RICE.  10. Existing stationary CI RICE >500 HP that are not limited use stationary RICE.	Reduce CO emissions, or limit the concentration of CO in the stationary RICE exhaust, and using oxidation catalyst.	8,760 hours or 3 years, whichever

### Pt. 63, Subpt. ZZZZ, Table 6

For each	Complying with the requirement to	You must demonstrate continuous com- pliance by
11. Existing stationary CI RICE >500 HP that are not limited use stationary RICE.	Reduce CO emissions, or limit the concentration of CO in the stationary RICE exhaust, and not using oxidation catalyst.	iii. Reducing these data to 4-hour rolling averages; and iv. Maintaining the 4-hour rolling averages within the operating limitations for the catalyst inlet temperature; and v. Measuring the pressure drop across the catalyst once per month and demonstrating that the pressure drop across the catalyst is within the operating limitation established during the performance test.  i. Conducting performance tests every 8,760 hours or 3 years, whichever comes first, for CO or formaldehyde, as appropriate, to demonstrate that the required CO or formaldehyde, as appropriate, percent reduction is achieved or that your emissions re-
		achieved or that your emissions re- main at or below the CO or formalde- hyde concentration limit; and ii. Collecting the approved operating pa- rameter (if any) data according to §63.6625(b); and iii. Reducing these data to 4-hour rolling
		averages; and iv. Maintaining the 4-hour rolling aver- ages within the operating limitations for the operating parameters estab- lished during the performance test.
<ol> <li>Existing limited use CI stationary RICE &gt;500 HP.</li> </ol>	Reduce CO emissions or limit the concentration of CO in the stationary RICE exhaust, and using an oxidation catalyst.	<ol> <li>Conducting performance tests every 8,760 hours or 5 years, whichever comes first, for CO or formaldehyde, as appropriate, to demonstrate that the required CO or formaldehyde, as appropriate, percent reduction is achieved or that your emissions re- main at or below the CO or formalde- hyde concentration limit; and ii. Collecting the catalyst inlet tempera- ture data according to §63.6625(b);</li> </ol>
		and iii. Reducing these data to 4-hour rolling averages; and iv. Maintaining the 4-hour rolling averages within the operating limitations for the catalyst inlet temperature; and v. Measuring the pressure drop across the catalyst once per month and dem- onstrating that the pressure drop
13. Existing limited use CI stationary RICE >500 HP.	Reduce CO emissions or limit the concentration of CO in the stationary RICE exhaust, and not using an oxidation catalyst.	across the catalyst is within the operating limitation established during the performance test.  i. Conducting performance tests every 8,760 hours or 5 years, whichever comes first, for CO or formaldehyde, as appropriate, to demonstrate that the required CO or formaldehyde, as appropriate, percent reduction is achieved or that your emissions remain at or below the CO or formaldehyde concentration limit; and
		ii. Collecting the approved operating parameter (if any) data according to § 63.665(b); and iii. Reducing these data to 4-hour rolling averages; and iv. Maintaining the 4-hour rolling averages within the operating limitations for the operating parameters established during the performance test.

### Pt. 63, Subpt. ZZZZ, Table 6

### 40 CFR Ch. I (7-1-13 Edition)

For each	Complying with the requirement to	You must demonstrate continuous compliance by			
14. Existing non-emergency 4SLB stationary RICE >500 HP located at an area source of HAP that are not remote stationary RICE and that are operated more than 24 hours per calendar year.	a. Install an oxidation catalyst	i. Conducting annual compliance demonstrations as specified in §63.6640(c) to show that the average reduction of emissions of CO is 93 percent or more, or the average CO concentration is less than or equal to 47 ppmvd at 15 percent O <sub>2</sub> ; and either ii. Collecting the catalyst inlet temperature data according to §63.6625(b), reducing these data to 4-hour rolling averages; and maintaining the 4-hour rolling averages within the limitation of greater than 450 °F and less than or equal to 1350 °F for the catalyst inlet temperature; or iii. Immediately shutting down the engine if the catalyst inlet temperature exceeds 1350 °F.			
15. Existing non-emergency 4SRB stationary RICE >500 HP located at an area source of HAP that are not remote stationary RICE and that are operated more than 24 hours per calendar year.	a. Install NSCR	i. Conducting annual compliance demonstrations as specified in § 63.6640(c) to show that the average reduction of emissions of CO is 75 percent or more, the average CO concentration is less than or equal to 270 ppmvd at 15 percent O <sub>2</sub> or the average reduction of emissions of THC is 30 percent or more; and either ii. Collecting the catalyst inlet temperature data according to §63.6625(b), reducing these data to 4-hour rolling averages; and maintaining the 4-hour rolling averages within the limitation of greater than or equal to 750 °F and less than or equal to 1250 °F for the catalyst inlet temperature; or iii. Immediately shutting down the engine if the catalyst inlet temperature exceeds 1250 °F.			

aAfter you have demonstrated compliance for two consecutive tests, you may reduce the frequency of subsequent performance tests to annually. If the results of any subsequent annual performance test indicate the stationary RICE is not in compliance with the CO or formaldehyde emission limitation, or you deviate from any of your operating limitations, you must resume semi-annual performance tests.

[78 FR 6715, Jan. 30, 2013]

# Pt. 63, Subpt. ZZZZ, Table 7

 $\label{thm:constraint} TABLE~7~TO~SUBPART~ZZZZ~OF~PART~63\\ --REQUIREMENTS~FOR~REPORTS~$  As stated in §63.6650, you must comply with the following requirements for reports:

For each	You must submit a	The report must contain	You must submit the report
1. Existing non-emergency, non-black start stationary RICE 100≤HP≤500 located at a major source of HAP; existing non-emergency, non-black start stationary CI RICE >500 HP located at a major source of HAP; existing non-emergency 4SRB stationary RICE >500 HP located at a major source of HAP; existionary CI RICE >300 HP located at an area source of HAP; new or reconstructed non-emergency stationary CI RICE >300 HP located at a major source of HAP; and new or reconstructed non-emergency 4SLB stationary RICE >500 HP located at a major source of HAP; and new or reconstructed non-emergency 4SLB stationary RICE 250≤HP≤500 located at a major source of HAP.	Compliance report	a. If there are no deviations from any emission limitations or operating limitations that apply to you, a statement that there were no deviations from the emission limitations or operating limitations during the reporting period. If there were no periods during which the CMS, including CEMS and CPMS, was outof-control, as specified in §53.8(c)(7), a statement that there were not periods during which the CMS was out-of-control during the reporting period; or	i. Semiannually according to the requirements in § 63.6650(b)(1)–(5) for engines that are not limited use stationary RICE subject to numerical emission limitations; and ii. Annually according to the requirements in § 63.6650(b)(6)–(9) for engines that are limited use stationary RICE subject to numerical emission limitations.
		b. If you had a deviation from any emission limitation or operating limitation during the reporting period, the information in § 63.6650(d). If there were periods during which the CMS, including CEMS and CPMS, was out-of-control, as specified in § 63.8(c)(7), the information in § 63.6650(e); or	i. Semiannually according to the requirements in § 63.6650(b).
New or reconstructed non- emergency stationary RICE that combusts landfill gas or digester gas equivalent to 10 percent or more of the gross	Report	c. If you had a malfunction during the reporting period, the information in § 63.6650(c)(4).  a. The fuel flow rate of each fuel and the heating values that were used in your calculations, and you must demonstrate that the per-	i. Semiannually according to the requirements in § 63.6650(b).     i. Annually, according to the requirements in § 63.6650.
heat input on an annual basis.		centage of heat input pro- vided by landfill gas or di- gester gas, is equivalent to 10 percent or more of the gross heat input on an an- nual basis; and	l Cu tura d'
		b. The operating limits provided in your federally enforceable permit, and any deviations from these limits; and     c. Any problems or errors	i. See item 2.a.i.
<ol> <li>Existing non-emergency, non-black start 4SLB and 4SRB stationary RICE &gt;500 HP located at an area source of HAP that are not remote stationary RICE and that operate more than 24 hours per calendar year.</li> </ol>	Compliance report	suspected with the meters. a. The results of the annual compliance demonstration, if conducted during the reporting period.	i. Semiannually according to the requirements in § 63.6650(b)(1)–(5).

### Pt. 63, Subpt. ZZZZ, Table 8

### 40 CFR Ch. I (7-1-13 Edition)

For each	You must submit a	The report must contain	You must submit the report		
4. Emergency stationary RICE that operate or are contractually obligated to be available for more than 15 hours per year for the purposes specified in § 63.6640(f)(2)(ii) and (iii) or that operate for the purposes specified in § 63.6640(f)(4)(ii).	nergency stationary RICE to operate or are contractly obligated to be available for more than 15 hours right year for the purposes scilided in § 63.6840(f)(2)(ii) did (iii) or that operate for purposes specified in		i. annually according to the requirements in § 63.6650(h)(2)–(3).		

[78 FR 6719, Jan. 30, 2013]

Table 8 to Subpart ZZZZ of Part 63—Applicability of General Provisions to Subpart ZZZZ.

As stated in §63.6665, you must comply with the following applicable general provisions.

General provisions citation	Subject of citation	Applies to sub- part	Explanation
§ 63.1	General applicability of the General Provisions.	Yes.	
§ 63.2	Definitions	Yes	Additional terms defined in § 63.6675.
§ 63.3	Units and abbreviations	Yes.	
§ 63.4	Prohibited activities and circumvention.	Yes.	10
§ 63.5	Construction and reconstruction	Yes.	
§ 63.6(a)	Applicability	Yes.	
§ 63.6(b)(1)–(4)	Compliance dates for new and reconstructed sources.	Yes.	Ti.
§ 63.6(b)(5) § 63.6(b)(6)	Notification    [Reserved]	Yes.	
§ 63.6(b)(7)	Compliance dates for new and reconstructed area sources that become major sources.	Yes.	
§ 63.6(c)(1)–(2)	Compliance dates for existing sources.	Yes.	
§ 63.6(c)(3)-(4)	[Reserved]		
§ 63.6(c)(5)	Compliance dates for existing area sources that become major sources.	Yes.	
§ 63.6(d)	[Reserved]	i	
§ 63.6(e)	Operation and maintenance	No.	
§ 63.6(f)(1)	Applicability of standards	No.	
§ 63.6(f)(2)	Methods for determining compliance	Yes.	
§ 63.6(f)(3)	Finding of compliance	Yes.	
§ 63.6(g)(1)–(3)	Use of alternate standard	Yes.	l
§ 63.6(h)	Opacity and visible emission stand- ards.	No	Subpart ZZZZ does not contain opac ity or visible emission standards.
§ 63.6(i)	Compliance extension procedures and criteria.	Yes.	
§ 63.6(j)	Presidential compliance exemption	Yes.	0. ha - 4 7777
§ 63.7(a)(1)–(2)	Performance test dates	Yes	Subpart ZZZZ contains performance test dates at §§ 63.6610, 63.6611 and 63.6612.
§ 63.7(a)(3)	CAA section 114 authority	Yes.	
§ 63.7(b)(1)	Notification of performance test	Yes	Except that § 63.7(b)(1) only applie as specified in § 63.6645.
§ 63.7(b)(2)	Notification of rescheduling	Yes	Except that §63.7(b)(2) only applie as specified in §63.6645.
§ 63.7(c)	Quality assurance/test plan	Yes	Except that §63.7(c) only applies a specified in §63.6645.
§ 63.7(d)	Testing facilities	Yes.	
§ 63.7(e)(1)	Conditions for conducting performance tests.	No	Subpart ZZZZ specifies conditions for conducting performance tests a § 63.6620.
§ 63.7(e)(2)	Conduct of performance tests and reduction of data.	Yes	Subpart ZZZZ specifies test method at § 63.6620.
§ 63.7(e)(3)	Test run duration	Yes.	
§ 63.7(e)(4)	Administrator may require other test-	Yes.	{
	ing under section 114 of the CAA.	1	
8.63.7(f)	Alternative test method provisions	l Yes.	I .

# Pt. 63, Subpt. ZZZZ, Table 8

# **Environmental Protection Agency**

prformance test data analysis, recordkeeping, and reporting. aiver of tests	Yes. Yes. Yes. No. Yes. Yes. Yes. Yes. No. Yes.	Subpart ZZZZ contains specific requirements for monitoring at § 63.6625.
oplicability of monitoring requirements.  arformance specifications	Yes.  Yes.  No. Yes. Yes. Yes. Yes. Yes.	quirements for monitoring at
ments.  erformance specifications	Yes. No. Yes. Yes. Yes. No. Yes.	quirements for monitoring at
leserved] onitoring for control devices uitiple effluents and multiple monitoring systems. onitoring system operation and maintenance. outine and predictable SSM SM not in Startup Shutdown Malfunction Plan. ompliance with operation and maintenance requirements.	No. Yes. Yes. Yes. No. Yes.	
leserved] onitoring for control devices uitiple effluents and multiple monitoring systems. onitoring system operation and maintenance. outine and predictable SSM SM not in Startup Shutdown Malfunction Plan. ompliance with operation and maintenance requirements.	Yes. Yes. Yes. No. Yes.	
onitoring	Yes. Yes. Yes. No. Yes.	
ultiple effluents and multiple moni- toring systems operation and maintenance. outine and predictable SSM SM not in Startup Shutdown Mal- function Plan. ompliance with operation and main- tenance requirements.	Yes. Yes. No. Yes.	
toring systems. onitoring system operation and maintenance. outine and predictable SSM SM not in Startup Shutdown Mal- function Plan. ompliance with operation and main- tenance requirements.	Yes. No. Yes.	
maintenance.  putine and predictable SSM  SM not in Startup Shutdown Malfunction Plan.  ompliance with operation and maintenance requirements.	No. Yes.	
SM not in Startup Shutdown Mal- function Plan. ompliance with operation and main- tenance requirements.	Yes.	
function Plan.  ompliance with operation and maintenance requirements.		
tenance requirements.	l No.	
onitoring system installation		
	Yes.	<b>_</b>
ontinuous monitoring system (CMS)	Yes	Except that subpart ZZZZ does not
requirements.	N	require Continuous Opacity Monitoring System (COMS).
OMS minimum procedures	No	Subpart ZZZZ does not require COMS.
MS requirements  MS quality control	Yes	Except that subpart ZZZZ does not require COMS.
MS performance evaluation	Yes	Except for §63.8(e)(5)(ii), which ap-
, , , , , , , , , , , , , , , , , , ,	Except that § 63.8(e)	plies to COMS.
	only applies as specified in §63.6645.	
ternative monitoring method		Except that §63.8(f)(4) only applies as specified in §63.6645.
•		Except that §63.8(f)(6) only applies as specified in §63.6645.
ata reduction	165	Except that provisions for COMS are not applicable. Averaging periods for demonstrating compliance are specified at \$\$63.6635 and 63.6640.
pplicability and State delegation of notification requirements.	Yes.	
itial notifications	Yes	Except that § 63.9(b)(3) is reserved.
	Except that § 63.9(b) only applies as specified	
equest for compliance extension	in § 63.6645. Yes	Except that §63.9(c) only applies as
otification of special compliance re-	Yes	specified in § 63.6645. Except that § 63.9(d) only applies as
quirements for new sources. otification of performance test	Yes	specified in § 63.6645.  Except that § 63.9(e) only applies as
otification of visible emission (VE)/	No	specified in § 63.6645. Subpart ZZZZ does not contain opac-
opacity test. otification of performance evaluation	Yes	ity or VE standards.  Except that § 63.9(g) only applies as
otification of use of COMS data	No	specified in § 63.6645. Subpart ZZZZ does not contain opac-
	Yes	ity or VE standards. If alternative is in use.
otification that criterion for alternative to RATA is exceeded.		a anemanye is at use.
t a priit	plicability and State delegation of notification requirements. It is in the compliance extension	plicability and State delegation of notification requirements. ial notifications

### Pt. 63, Subpt. ZZZZ, App. A

### 40 CFR Ch. I (7-1-13 Edition)

General provisions citation	Subject of citation	Applies to sub- part	Explanation			
§ 63.9(h)(1)–(6)	Notification of compliance status	Yes	Except that notifications for sources using a CEMS are due 30 days after completion of performance evaluations. § 63.9(h)(4) is reserved.			
			Except that §63.9(h) only applies as specified in §63.6645.			
§ 63.9(i)	Adjustment of submittal deadlines	Yes.				
§ 63.9(j)	Change in previous information	Yes.				
§ 63.10(a)	Administrative provisions for record- keeping/reporting.	Yes.				
§ 63.10(b)(1)	Record retention	Yes	Except that the most recent 2 years of data do not have to be retained on site.			
§ 63.10(b)(2)(i)–(v)	Records related to SSM	No.				
§ 63.10(b)(2)(vi)–(xi)	Records	Yes.				
§ 63.10(b)(2)(xii)	Record when under waiver	Yes.				
§ 63.10(b)(2)(xiii)	Records when using alternative to RATA.	Yes	For CO standard if using RATA alternative.			
§ 63.10(b)(2)(xiv)	Records of supporting documentation	Yes.				
§ 63.10(b)(3)	Records of applicability determination	Yes.				
§ 63.10(c)	Additional records for sources using CEMS.	Yes	Except that §63.10(c)(2)-(4) and (9) are reserved.			
§ 63.10(d)(1)	General reporting requirements	Yes.	G. 5 1000, 1000			
§ 63.10(d)(2)	Report of performance test results	Yes.				
§ 63.10(d)(3)	Reporting opacity or VE observations	No	Subpart ZZZZ does not contain opacity or VE standards.			
§ 63.10(d)(4)	Progress reports	Yes.	.,			
§ 63.10(d)(5)	Startup, shutdown, and malfunction reports.	No.				
§ 63.10(e)(1) and (2)(i)	Additional CMS Reports	Yes.				
§ 63.10(e)(2)(ii)	COMS-related report	No	Subpart ZZZZ does not require COMS.			
§ 63.10(e)(3)	Excess emission and parameter exceedances reports.	Yes	Except that §63.10(e)(3)(i) (C) is reserved.			
§ 63.10(e)(4)	Reporting COMS data	No	Subpart ZZZZ does not require COMS.			
§ 63.10(f)	Waiver for recordkeeping/reporting	Yes.	* = ····= *			
§ 63.11	Flares	No.				
§ 63.12	State authority and delegations	Yes.				
§ 63.13	Addresses	Yes.				
§ 63.14	Incorporation by reference	Yes.				
§ 63.15	Availability of information	Yes.				
3 00.10	Transcinty of intoffiction	103.	<u> </u>			

[75 FR 9688, Mar. 3, 2010, as amended at 78 FR 6720, Jan. 30, 2013]

APPENDIX A—PROTOCOL FOR USING AN ELECTROCHEMICAL ANALYZER TO DETERMINE OXYGEN AND CARBON MONOXIDE CONCENTRATIONS FROM CERTAIN ENGINES

1.0 Scope and Application. What is this Protocol?

This protocol is a procedure for using portable electrochemical (EC) cells for meas-

uring carbon monoxide (CO) and oxygen  $(O_2)$  concentrations in controlled and uncontrolled emissions from existing stationary 4-stroke lean burn and 4-stroke rich burn reciprocating internal combustion engines as specified in the applicable rule.

1.1 Analytes. What does this protocol determine?

This protocol measures the engine exhaust gas concentrations of carbon monoxide (CO) and oxygen  $(O_2)$ .

Analyte	CAS No.	Sensitivity
Carbon monoxide (CO)	630-08-0	Minimum detectable limit should be 2 percent of the nominal range or 1 ppm, whichever is less restrictive.
Oxygen (O <sub>2</sub> )	7782 <del>-44-</del> 7	

# 1.2 Applicability. When is this protocol acceptable?

This protocol is applicable to 40 CFR part 63, subpart ZZZZ. Because of inherent cross sensitivities of EC cells, you must not apply this protocol to other emissions sources without specific instruction to that effect.

# 1.3 Data Quality Objectives. How good must my collected data be?

Refer to Section 13 to verify and document acceptable analyzer performance.

# 1.4 Range. What is the targeted analytical range for this protocol?

The measurement system and EC cell design(s) conforming to this protocol will determine the analytical range for each gas component. The nominal ranges are defined by choosing up-scale calibration gas concentrations near the maximum anticipated flue gas concentrations for CO and  $O_2$ , or no more than twice the permitted CO level.

# 1.5 Sensitivity. What minimum detectable limit will this protocol yield for a particular gas component?

The minimum detectable limit depends on the nominal range and resolution of the specific EC cell used, and the signal to noise ratio of the measurement system. The minimum detectable limit should be 2 percent of the nominal range or 1 ppm, whichever is less restrictive.

#### 2.0 SUMMARY OF PROTOCOL

In this protocol, a gas sample is extracted from an engine exhaust system and then conveyed to a portable EC analyzer for measurement of CO and O<sub>2</sub> gas concentrations. This method provides measurement system performance specifications and sampling protocols to ensure reliable data. You may use additions to, or modifications of vendor supplied measurement systems (e.g., heated or unheated sample lines, thermocouples, flow meters, selective gas scrubbers, etc.) to meet the design specifications of this protocol. Do not make changes to the measurement system from the as-verified configuration (Section 3.12).

#### 3.0 DEFINITIONS

- 3.1 Measurement System. The total equipment required for the measurement of CO and  $O_2$  concentrations. The measurement system consists of the following major subsystems:
- 3.1.1 Data Recorder. A strip chart recorder, computer or digital recorder for logging measurement data from the analyzer output. You may record measurement data from the digital data display manually or electronically.

- 3.1.2 Electrochemical (EC) Cell. A device, similar to a fuel cell, used to sense the presence of a specific analyte and generate an electrical current output proportional to the analyte concentration.
- 3.1.3 Interference Gas Scrubber. A device used to remove or neutralize chemical compounds that may interfere with the selective operation of an EC cell.
- 3.1.4 Moisture Removal System. Any device used to reduce the concentration of moisture in the sample stream so as to protect the EC cells from the damaging effects of condensation and to minimize errors in measurements caused by the scrubbing of soluble gases.
- 3.1.5 Sample Interface. The portion of the system used for one or more of the following: sample acquisition; sample transport; sample conditioning or protection of the EC cell from any degrading effects of the engine exhaust effluent; removal of particulate matter and condensed moisture.
- 3.2 Nominal Range. The range of analyte concentrations over which each EC cell is operated (normally 25 percent to 150 percent of up-scale calibration gas value). Several nominal ranges can be used for any given cell so long as the calibration and repeatability checks for that range remain within specifications.
- 3.3 Calibration Gas. A vendor certified concentration of a specific analyte in an appropriate balance gas.
- 3.4 Zero Calibration Error. The analyte concentration output exhibited by the EC cell in response to zero-level calibration gas.
- 3.5 Up-Scale Calibration Error. The mean of the difference between the analyte concentration exhibited by the EC cell and the certified concentration of the up-scale calibration gas.
- 3.6 Interference Check. A procedure for quantifying analytical interference from components in the engine exhaust gas other than the targeted analytes.
- 3.7 Repeatability Check. A protocol for demonstrating that an EC cell operated over a given nominal analyte concentration range provides a stable and consistent response and is not significantly affected by repeated exposure to that gas.
- 3.8 Sample Flow Rate. The flow rate of the gas sample as it passes through the EC cell. In some situations, EC cells can experience drift with changes in flow rate. The flow rate must be monitored and documented during all phases of a sampling run.
- 3.9 Sampling Run. A timed three-phase event whereby an EC cell's response rises and plateaus in a sample conditioning phase, remains relatively constant during a measurement data phase, then declines during a refresh phase. The sample conditioning phase exposes the EC cell to the gas sample for a length of time sufficient to reach a constant response. The measurement data phase is the

#### 40 CFR Ch. I (7-1-13 Edition)

time interval during which gas sample measurements can be made that meet the acceptance criteria of this protocol. The refresh phase then purges the EC cells with CO-free air. The refresh phase replenishes requisite O2 and moisture in the electrolyte reserve and provides a mechanism to de-gas or desorb any interference gas scrubbers or filters so as to enable a stable CO EC cell response. There are four primary types of sampling runs: pre-sampling calibrations; stack gas sampling; post-sampling calibration checks; and measurement system repeatability checks. Stack gas sampling runs can be chained together for extended evaluations, providing all other procedural specifications are met.

3.10 Sampling Day. A time not to exceed twelve hours from the time of the pre-sampling calibration to the post-sampling calibration check. During this time, stack gas sampling runs can be repeated without repeated recalibrations, providing all other sampling specifications have been met.

3.11 Pre-Sampling Calibration/Post-Sampling Calibration Check. The protocols executed at the beginning and end of each sampling day to bracket measurement readings with controlled performance checks.

3.12 Performance-Established Configuration. The EC cell and sampling system configuration that existed at the time that it initially met the performance requirements of this protocol.

#### 4.0 INTERFERENCES.

When present in sufficient concentrations, NO and NO<sub>2</sub> are two gas species that have been reported to interfere with CO concentration measurements. In the likelihood of this occurrence, it is the protocol user's responsibility to employ and properly maintain an appropriate CO EC cell filter or scrubber for removal of these gases, as described in Section 6.2.12.

5.0 SAFETY. [RESERVED]

6.0 EQUIPMENT AND SUPPLIES.

6.1 What equipment do I need for the measurement system?

The system must maintain the gas sample at conditions that will prevent moisture condensation in the sample transport lines, both before and as the sample gas contacts the EC cells. The essential components of the measurement system are described below.

#### 6.2 Measurement System Components.

6.2.1 Sample Probe. A single extractionpoint probe constructed of glass, stainless steel or other non-reactive material, and of length sufficient to reach any designated sampling point. The sample probe must be designed to prevent plugging due to condensation or particulate matter. 6.2.2 Sample Line. Non-reactive tubing to transport the effluent from the sample probe to the EC cell.

6.2.3 Calibration Assembly (optional). A three-way valve assembly or equivalent to introduce calibration gases at ambient pressure at the exit end of the sample probe during calibration checks. The assembly must be designed such that only stack gas or calibration gas flows in the sample line and all gases flow through any gas path filters.

6.2.4 Particulate Filter (optional). Filters before the inlet of the EC cell to prevent accumulation of particulate material in the measurement system and extend the useful life of the components. All filters must be fabricated of materials that are non-reactive to the gas mixtures being sampled.

6.2.5 Sample Pump. A leak-free pump to provide undiluted sample gas to the system at a flow rate sufficient to minimize the response time of the measurement system. If located upstream of the EC cells, the pump must be constructed of a material that is non-reactive to the gas mixtures being sampled.

6.2.8 Sample Flow Rate Monitoring. An adjustable rotameter or equivalent device used to adjust and maintain the sample flow rate through the analyzer as prescribed.

6.2.9 Sample Gas Manifold (optional). A manifold to divert a portion of the sample gas stream to the analyzer and the remainder to a by-pass discharge vent. The sample gas manifold may also include provisions for introducing calibration gases directly to the analyzer. The manifold must be constructed of a material that is non-reactive to the gas mixtures being sampled.

6.2.10~EC~cell. A device containing one or more EC cells to determine the CO and  $O_2$  concentrations in the sample gas stream. The EC cell(s) must meet the applicable performance specifications of Section 13 of this protocol.

6.2.11 Data Recorder. A strip chart recorder, computer or digital recorder to make a record of analyzer output data. The data recorder resolution (i.e., readability) must be no greater than 1 ppm for CO; 0.1 percent for  $O_2$ ; and one degree (either °C or °F) for temperature. Alternatively, you may use a digital or analog meter having the same resolution to observe and manually record the analyzer responses.

6.2.12 Interference Gas Filter or Scrubber. A device to remove interfering compounds upstream of the CO EC cell. Specific interference gas filters or scrubbers used in the performance-established configuration of the analyzer must continue to be used. Such a filter or scrubber must have a means to determine when the removal agent is exhausted. Periodically replace or replenish it in accordance with the manufacturer's recommendations.

7.0 REAGENTS AND STANDARDS. WHAT CALIBRATION GASES ARE NEEDED?

7.1 Calibration Gases. CO calibration gases for the EC cell must be CO in nitrogen or CO in a mixture of nitrogen and  $O_2$ . Use CO calibration gases with labeled concentration values certified by the manufacturer to be within  $\pm 5$  percent of the label value. Dry ambient air (20.9 percent  $O_2$ ) is acceptable for calibration of the  $O_2$  cell. If needed, any lower percentage  $O_2$  calibration gas must be a mixture of  $O_2$  in nitrogen.

7.1.1 Up-Scale CO Calibration Gas Concentration. Choose one or more up-scale gas concentrations such that the average of the stack gas measurements for each stack gas sampling run are between 25 and 150 percent of those concentrations. Alternatively, choose an up-scale gas that does not exceed twice the concentration of the applicable outlet standard. If a measured gas value exceeds 150 percent of the up-scale CO calibration gas value at any time during the stack gas sampling run, the run must be discarded and repeated.

7.1.2 Up-Scale O<sub>2</sub> Calibration Gas Concentration.

Select an  ${\rm O}_2$  gas concentration such that the difference between the gas concentration and the average stack gas measurement or reading for each sample run is less than 15 percent  ${\rm O}_2$ . When the average exhaust gas  ${\rm O}_2$  readings are above 6 percent, you may use dry ambient air (20.9 percent  ${\rm O}_2$ ) for the upscale  ${\rm O}_2$  calibration gas.

7.1.3 Zero Gas. Use an inert gas that contains less than 0.25 percent of the up-scale CO calibration gas concentration. You may use dry air that is free from ambient CO and other combustion gas products (e.g., CO<sub>2</sub>).

#### 8.0 SAMPLE COLLECTION AND ANALYSIS

8.1 Selection of Sampling Sites.

8.1.1 Control Device Inlet. Select a sampling site sufficiently downstream of the engine so that the combustion gases should be well mixed. Use a single sampling extraction point near the center of the duct (e.g., within the 10 percent centroidal area), unless instructed otherwise.

8.1.2 Exhaust Gas Outlet. Select a sampling site located at least two stack diameters downstream of any disturbance (e.g., turbo-charger exhaust, crossover junction or recirculation take-off) and at least one-half stack diameter upstream of the gas discharge to the atmosphere. Use a single sampling extraction point near the center of the duct (e.g., within the 10 percent centroidal area), unless instructed otherwise.

8.2 Stack Gas Collection and Analysis. Prior to the first stack gas sampling run, conduct that the pre-sampling calibration in accordance with Section 10.1. Use Figure 1 to record all data. Zero the analyzer with zero gas. Confirm and record that the scrubber

media color is correct and not exhausted. Then position the probe at the sampling point and begin the sampling run at the same flow rate used during the up-scale calibration. Record the start time. Record all EC cell output responses and the flow rate during the "sample conditioning phase" once per minute until constant readings are obtained. Then begin the "measurement data phase" and record readings every 15 seconds for at least two minutes (or eight readings). or as otherwise required to achieve two continuous minutes of data that meet the specification given in Section 13.1. Finally, perform the "refresh phase" by introducing dry air, free from CO and other combustion gases, until several minute-to-minute readings of consistent value have been obtained. For each run use the "measurement data phase" readings to calculate the average stack gas CO and O2 concentrations.

8.3 EC Cell Rate. Maintain the EC cell sample flow rate so that it does not vary by more than ± 10 percent throughout the pre-sampling calibration, stack gas sampling and post-sampling calibration check. Alternatively, the EC cell sample flow rate can be maintained within a tolerance range that does not affect the gas concentration readings by more than ± 3 percent, as instructed by the EC cell manufacturer.

#### 9.0 QUALITY CONTROL (RESERVED)

#### 10.0 Calibration and Standardization

10.1 Pre-Sampling Calibration. Conduct the following protocol once for each nominal range to be used on each EC cell before performing a stack gas sampling run on each field sampling day. Repeat the calibration if you replace an EC cell before completing all of the sampling runs. There is no prescribed order for calibration of the EC cells; however, each cell must complete the measurement data phase during calibration. Assemble the measurement system by following the manufacturer's recommended protocols including for preparing and preconditioning the EC cell. Assure the measurement system has no leaks and verify the gas scrubbing agent is not depleted. Use Figure 1 to record all data.

 $10.1.1\ Zero\ Calibration.$  For both the  $O_2$  and CO cells, introduce zero gas to the measurement system (e.g., at the calibration assembly) and record the concentration reading every minute until readings are constant for at least two consecutive minutes. Include the time and sample flow rate. Repeat the steps in this section at least once to verify the zero calibration for each component gas.

10.1.2 Zero Calibration Tolerance. For each zero gas introduction, the zero level output must be less than or equal to  $\pm 3$  percent of the up-scale gas value or  $\pm 1$  ppm, whichever is less restrictive, for the CO channel and

#### 40 CFR Ch. I (7-1-13 Edition)

less than or equal to  $\pm$  0.3 percent  $O_2$  for the  $O_2$  channel.

10.1.3 Up-Scale Calibration. Individually introduce each calibration gas to the measurement system (e.g., at the calibration assembly) and record the start time. Record all EC cell output responses and the flow rate during this "sample conditioning phase" once per minute until readings are constant for at least two minutes. Then begin the "measurement data phase" and record readings every 15 seconds for a total of two minutes, or as otherwise required. Finally, perform the "refresh phase" by introducing dry air, free from CO and other combustion gases. until readings are constant for at least two consecutive minutes. Then repeat the steps in this section at least once to verify the calibration for each component gas. Introduce all gases to flow through the entire sample handling system (i.e., at the exit end of the sampling probe or the calibration assembly).

10.1.4 Up-Scale Calibration Error. The mean of the difference of the "measurement data phase" readings from the reported standard gas value must be less than or equal to  $\pm 5$  percent or  $\pm 1$  ppm for CO or  $\pm 0.5$  percent O<sub>2</sub>, whichever is less restrictive, respectively. The maximum allowable deviation from the mean measured value of any single "measurement data phase" reading must be less than or equal to  $\pm 2$  percent or  $\pm 1$  ppm for CO or  $\pm 0.5$  percent O<sub>2</sub>, whichever is less restrictive, respectively.

10.2 Post-Sampling Calibration Check. Conduct a stack gas post-sampling calibration check after the stack gas sampling run or set of runs and within 12 hours of the initial calibration. Conduct up-scale and zero calibration checks using the protocol in Section 10.1. Make no changes to the sampling system or EC cell calibration until all post-sampling calibration checks have been recorded. If either the zero or up-scale calibration error exceeds the respective specification in Sections 10.1.2 and 10.1.4 then all measurement data collected since the previous successful calibrations are invalid and re-calibration and re-sampling are required. If the sampling system is disassembled or the EC cell calibration is adjusted, repeat the calibration check before conducting the next analvzer sampling run.

#### 11.0 ANALYTICAL PROCEDURE

The analytical procedure is fully discussed in Section 8.

#### 12.0 CALCULATIONS AND DATA ANALYSIS

Determine the CO and  $O_2$  concentrations for each stack gas sampling run by calculating the mean gas concentrations of the data recorded during the "measurement data phase".

13.0 PROTOCOL PERFORMANCE

Use the following protocols to verify consistent analyzer performance during each field sampling day.

13.1 Measurement Data Phase Performance Check. Calculate the mean of the readings from the "measurement data phase". The maximum allowable deviation from the mean for each of the individual readings is  $\pm$  2 percent, or  $\pm$  1 ppm, whichever is less restrictive. Record the mean value and maximum deviation for each gas monitored. Data must conform to Section 10.1.4. The EC cell flow rate must conform to the specification in Section 8.3.

Example: A measurement data phase is invalid if the maximum deviation of any single reading comprising that mean is greater than  $\pm$  2 percent or  $\pm$  1 ppm (the default criteria). For example, if the mean = 30 ppm, single readings of below 29 ppm and above 31 ppm are disallowed).

13.2 Interference Check. Before the initial use of the EC cell and interference gas scrubber in the field, and semi-annually thereafter, challenge the interference gas scrubber with NO and  $NO_2$  gas standards that are generally recognized as representative of diesel-fueled engine NO and  $NO_2$  emission values. Record the responses displayed by the CO EC cell and other pertinent data on Figure 1 or a similar form.

13.2.1 Interference Response. The combined

13.2.1 Interference Response. The combined NO and NO<sub>2</sub> interference response should be less than or equal to  $\pm$  5 percent of the upscale CO calibration gas concentration.

13.3 Repeatability Check. Conduct the following check once for each nominal range that is to be used on the CO EC cell within 5 days prior to each field sampling program. If a field sampling program lasts longer than 5 days, repeat this check every 5 days. Immediately repeat the check if the EC cell is replaced or if the EC cell is exposed to gas concentrations greater than 150 percent of the highest up-scale gas concentration.

13.3.1 Repeatability Check Procedure. Perform a complete EC cell sampling run (all three phases) by introducing the CO calibration gas to the measurement system and record the response. Follow Section 10.1.3. Use Figure 1 to record all data. Repeat the run three times for a total of four complete runs. During the four repeatability check runs, do not adjust the system except where necessary to achieve the correct calibration gas flow rate at the analyzer.

13.3.2 Repeatability Check Calculations. Determine the highest and lowest average "measurement data phase" CO concentrations from the four repeatability check runs and record the results on Figure 1 or a similar form. The absolute value of the difference between the maximum and minimum average values recorded must not vary more than  $\pm$  3 percent or  $\pm$  1 ppm of the up-scale gas value, whichever is less restrictive.

14.0 POLLUTION PREVENTION (RESERVED)

15.0 WASTE MANAGEMENT (RESERVED)

16.0 ALTERNATIVE PROCEDURES (RESERVED)

#### 17.0 REFERENCES

- (1) "Development of an Electrochemical Cell Emission Analyzer Test Protocol", Topical Report, Phil Juneau, Emission Monitoring, Inc., July 1997.
- toring, Inc., July 1997.

  (2) 'Determination of Nitrogen Oxides,
  Carbon Monoxide, and Oxygen Emissions

### Pt. 63, Subpt. ZZZZ, App. A

from Natural Gas-Fired Engines, Boilers, and Process Heaters Using Portable Analyzers", EMC Conditional Test Protocol 30 (CTM-30), Gas Research Institute Protocol GRI-96/0008, Revision 7, October 13, 1997.

- (3) "ICAC Test Protocol for Periodic Monitoring", EMC Conditional Test Protocol 34 (CTM-034), The Institute of Clean Air Companies, September 8, 1999.
- (4) "Code of Federal Regulations", Protection of Environment, 40 CFR, Part 60, Appendix A, Methods 1-4; 10.

Pt. 63, Subpt. ZZZZ, App. A

ື Fa	acility			_ En	gine I.D.				Date		
Run Type:(X)		Pre-Sam		ration Stack Gas Sample			Post-Sa	(ample Cal. Che	(_) eck Repeatability Check		
Run #	1 O <sub>2</sub>	1 CO	2 O <sub>2</sub>	2 CO	3 O <sub>2</sub>	3 CO	4 O <sub>2</sub>	4 CO	Time	Scrub. OK	Flow- Rate
Sample Cond											
Phase	••••••										
,,	************							***************************************	***************************************		
,,								***********	***************************************		
,			***************************************					***************************************	***************************************		
								************			
Measurement											
Data Phase			•••••			************					
[								***************************************			
[						***********					
[					·····						
[_ ·····										***************************************	
[											
<u> </u>											
( <u> </u>											
[											
[	[							•			
ſ											
,				***************************************						***************************************	
Mean					]						
Refresh		••••••								***************************************	
Phase				ĺ	[	[					
, , , , , , , , , , , , , , , , , , , ,											
"											
"										***************************************	
"								l		***************************************	
									********************	*******************	

TABLE 1: APPENDIX A-SAMPLING RUN DATA.

[78 FR 6721, Jan. 30, 2013]

Appendix K
40 CFR Part 63, Subpart CCCCCC
National Emission Standards for Hazardous Air Pollutants for Source Category: Gasoline
Dispensing Facilities

#### § 63.11110

#### 40 CFR Ch. I (7-1-13 Edition)

Citation	Subject	Brief description	Applies to subpart BBBBBB
§ 63.10(e)(3)(iv)-(v)	Excess Emissions Reports.	Requirement to revert to quarterly submission if there is an excess emissions and parameter monitor exceedances (now defined as deviations); provision to request semiannual reporting after compliance for 1 year; submit report by 30th day following end of quarter or calendar half; if there has not been an exceedance or excess emissions (now defined as deviations), report contents in a statement that there have been no deviations; must submit report containing all of the information in \$\\$ 63.8(c)(7)-(8) and 63.10c)(5)-(13).	Yes, § 63.11095 speci- fies excess emission events for this sub- part.
§ 63.10(e)(3)(vi)-(viii)	Excess Emissions Re- port and Summary Report.	Requirements for reporting excess emissions for CMS; requires all of the information in §§ 63.8(c)(7)–(8) and 63.10(c)(5)–(13).	Yes.
§ 63.10(e)(4)	Reporting COMS Data	Must submit COMS data with performance test data.	Yes.
§ 63.10(f)	Waiver for Record- keeping/Reporting.	Procedures for Administrator to waive	Yes.
§ 63.11(b)	Flares	Requirements for flares	Yes, the section ref- erences § 63.11(b).
§ 63.12	Delegation	State authority to enforce standards	Yes.
§ 63.13	Addresses	Addresses where reports, notifications, and requests are sent.	Yes.
§ 63.14	Incorporations by Ref- erence.	Test methods incorporated by reference	Yes.
§ 63.15	Availability of Information.	Public and confidential information	Yes.

[73 FR 1933, Jan. 10, 2008, as amended at 76 FR 4180, Jan. 24, 2011]

# Subpart CCCCCC—National Emission Standards for Hazardous Air Pollutants for Source Category: Gasoline Dispensing Facilities

SOURCE: 73 FR 1945, Jan. 10, 2008, unless otherwise noted.

WHAT THIS SUBPART COVERS

# §63.11110 What is the purpose of this subpart?

This subpart establishes national emission limitations and management practices for hazardous air pollutants (HAP) emitted from the loading of gasoline storage tanks at gasoline dispensing facilities (GDF). This subpart also establishes requirements to demonstrate compliance with the emission limitations and management practices.

# § 63.11111 Am I subject to the requirements in this subpart?

(a) The affected source to which this subpart applies is each GDF that is located at an area source. The affected source includes each gasoline cargo tank during the delivery of product to

- a GDF and also includes each storage tank.
- (b) If your GDF has a monthly throughput of less than 10,000 gallons of gasoline, you must comply with the requirements in §63.11116.
- (c) If your GDF has a monthly throughput of 10,000 gallons of gasoline or more, you must comply with the requirements in §63.11117.
- quirements in §63.11117.

  (d) If your GDF has a monthly throughput of 100,000 gallons of gasoline or more, you must comply with the requirements in §63.11118.
- (e) An affected source shall, upon request by the Administrator, demonstrate that their monthly throughput is less than the 10,000-gallon or the 100,000-gallon threshold level, as applicable. For new or reconstructed afsources, as specified in §63.11112(b) and (c), recordkeeping to document monthly throughput must begin upon startup of the affected source. For existing sources, as specified in §63.11112(d), recordkeeping to document monthly throughput must begin on January 10, 2008. For existing sources that are subject to this subpart only because they load gasoline into fuel tanks other than those in motor

vehicles, as defined in §63.11132, recordkeeping to document monthly throughput must begin on January 24, 2011. Records required under this paragraph shall be kept for a period of 5 years.

- (f) If you are an owner or operator of affected sources, as defined in paragraph (a) of this section, you are not required to obtain a permit under 40 CFR part 70 or 40 CFR part 71 as a result of being subject to this subpart. However, you must still apply for and obtain a permit under 40 CFR part 70 or 40 CFR part 71 if you meet one or more of the applicability criteria found in 40 CFR 70.3(a) and (b) or 40 CFR 71.3(a) and (b).
- (g) The loading of aviation gasoline into storage tanks at airports, and the subsequent transfer of aviation gasoline within the airport, is not subject to this subpart.
- (h) Monthly throughput is the total volume of gasoline loaded into, or dispensed from, all the gasoline storage tanks located at a single affected GDF. If an area source has two or more GDF at separate locations within the area source, each GDF is treated as a separate affected source.
- (i) If your affected source's throughput ever exceeds an applicable throughput threshold, the affected source will remain subject to the requirements for sources above the threshold, even if the affected source throughput later falls below the applicable throughput threshold.
- (j) The dispensing of gasoline from a fixed gasoline storage tank at a GDF into a portable gasoline tank for the on-site delivery and subsequent dispensing of the gasoline into the fuel tank of a motor vehicle or other gasoline-fueled engine or equipment used within the area source is only subject to §63.11116 of this subpart.
- (k) For any affected source subject to the provisions of this subpart and another Federal rule, you may elect to comply only with the more stringent provisions of the applicable subparts. You must consider all provisions of the rules, including monitoring, record-keeping, and reporting. You must identify the affected source and provisions with which you will comply in your Notification of Compliance Status required under §63.11124. You also must

demonstrate in your Notification of Compliance Status that each provision with which you will comply is at least as stringent as the otherwise applicable requirements in this subpart. You are responsible for making accurate determinations concerning the more stringent provisions, and noncompliance with this rule is not excused if it is later determined that your determination was in error, and, as a result, you are violating this subpart. Compliance with this rule is your responsibility and the Notification of Compliance Status does not alter or affect that responsibility.

[73 FR 1945, Jan. 10, 2008, as amended at 76 FR 4181, Jan. 24, 2011]

# §63.11112 What parts of my affected source does this subpart cover?

- (a) The emission sources to which this subpart applies are gasoline storage tanks and associated equipment components in vapor or liquid gasoline service at new, reconstructed, or existing GDF that meet the criteria specified in §63.1111. Pressure/Vacuum vents on gasoline storage tanks and the equipment necessary to unload product from cargo tanks into the storage tanks at GDF are covered emission sources. The equipment used for the refueling of motor vehicles is not covered by this subpart.
- (b) An affected source is a new affected source if you commenced construction on the affected source after November 9, 2006, and you meet the applicability criteria in §63.11111 at the time you commenced operation.
- (c) An affected source is reconstructed if you meet the criteria for reconstruction as defined in §63.2.
- (d) An affected source is an existing affected source if it is not new or reconstructed.

# § 63.11113 When do I have to comply with this subpart?

- (a) If you have a new or reconstructed affected source, you must comply with this subpart according to paragraphs (a)(1) and (2) of this section, except as specified in paragraph (d) of this section.
- (1) If you start up your affected source before January 10, 2008, you

must comply with the standards in this subpart no later than January 10, 2008.

- (2) If you start up your affected source after January 10, 2008, you must comply with the standards in this subpart upon startup of your affected source.
- (b) If you have an existing affected source, you must comply with the standards in this subpart no later than January 10, 2011.
- (c) If you have an existing affected source that becomes subject to the control requirements in this subpart because of an increase in the monthly throughput, as specified in §63.11111(c) or §63.11111(d), you must comply with the standards in this subpart no later than 3 years after the affected source becomes subject to the control requirements in this subpart.
- (d) If you have a new or reconstructed affected source and you are complying with Table 1 to this subpart, you must comply according to paragraphs (d)(1) and (2) of this section.
- (1) If you start up your affected source from November 9, 2006 to September 23, 2008, you must comply no later than September 23, 2008.
- (2) If you start up your affected source after September 23, 2008, you must comply upon startup of your affected source.
- (e) The initial compliance demonstration test required under  $\S 63.11120(a)(1)$  and (2) must be conducted as specified in paragraphs (e)(1) and (2) of this section.
- (1) If you have a new or reconstructed affected source, you must conduct the initial compliance test upon installation of the complete vapor balance system.
- (2) If you have an existing affected source, you must conduct the initial compliance test as specified in paragraphs (e)(2)(i) or (e)(2)(ii) of this section.
- (i) For vapor balance systems installed on or before December 15, 2009, you must test no later than 180 days after the applicable compliance date specified in paragraphs (b) or (c) of this section.
- (ii) For vapor balance systems installed after December 15, 2009, you must test upon installation of the complete vapor balance system.

- (f) If your GDF is subject to the control requirements in this subpart only because it loads gasoline into fuel tanks other than those in motor vehicles, as defined in §63.11132, you must comply with the standards in this subpart as specified in paragraphs (f)(1) or (f)(2) of this section.
- (1) If your GDF is an existing facility, you must comply by January 24, 2014
- (2) If your GDF is a new or reconstructed facility, you must comply by the dates specified in paragraphs (f)(2)(i) and (ii) of this section.
- (i) If you start up your GDF after December 15, 2009, but before January 24, 2011, you must comply no later than January 24, 2011.
- (ii) If you start up your GDF after January 24, 2011, you must comply upon startup of your GDF.

[73 FR 1945, Jan. 10, 2008, as amended at 73 FR 35944, June 25, 2008; 76 FR 4181, Jan. 24, 2011]

EMISSION LIMITATIONS AND MANAGEMENT PRACTICES

# § 63.11115 What are my general duties to minimize emissions?

Each owner or operator of an affected source under this subpart must comply with the requirements of paragraphs (a) and (b) of this section.

- (a) You must, at all times, operate and maintain any affected source, including associated air pollution control equipment and monitoring equipment, in a manner consistent with safety and good air pollution control practices for minimizing emissions. Determination of whether such operation and maintenance procedures are being used will be based on information available to the Administrator which may include, but is not limited to, monitoring results, review of operation and maintenance procedures, review of operation and maintenance records, and inspection of the source.
- (b) You must keep applicable records and submit reports as specified in  $\S63.11125(d)$  and  $\S63.11126(b)$ .

[76 FR 4182, Jan. 24, 2011]

# § 63.11116 Requirements for facilities with monthly throughput of less than 10,000 gallons of gasoline.

- (a) You must not allow gasoline to be handled in a manner that would result in vapor releases to the atmosphere for extended periods of time. Measures to be taken include, but are not limited to, the following:
  - (1) Minimize gasoline spills;
- (2) Clean up spills as expeditiously as practicable;
- (3) Cover all open gasoline containers and all gasoline storage tank fill-pipes with a gasketed seal when not in use:
- (4) Minimize gasoline sent to open waste collection systems that collect and transport gasoline to reclamation and recycling devices, such as oil/water separators.
- (b) You are not required to submit notifications or reports as specified in §63.11125, §63.11126, or subpart A of this part, but you must have records available within 24 hours of a request by the Administrator to document your gasoline throughput.
- (c) You must comply with the requirements of this subpart by the applicable dates specified in §63.11113.
- (d) Portable gasoline containers that meet the requirements of 40 CFR part 59, subpart F, are considered acceptable for compliance with paragraph (a)(3) of this section.

[73 FR 1945, Jan. 10, 2008, as amended at 76 FR 4182, Jan. 24, 2011]

# § 63.11117 Requirements for facilities with monthly throughput of 10,000 gallons of gasoline or more.

- (a) You must comply with the requirements in section §63.11116(a).
- (b) Except as specified in paragraph (c) of this section, you must only load gasoline into storage tanks at your facility by utilizing submerged filling, as defined in §63.11132, and as specified in paragraphs (b)(1), (b)(2), or (b)(3) of this section. The applicable distances in paragraphs (b)(1) and (2) shall be measured from the point in the opening of the submerged fill pipe that is the greatest distance from the bottom of the storage tank.
- (1) Submerged fill pipes installed on or before November 9, 2006, must be no more than 12 inches from the bottom of the tank.

- (2) Submerged fill pipes installed after November 9, 2006, must be no more than 6 inches from the bottom of the tank.
- (3) Submerged fill pipes not meeting the specifications of paragraphs (b)(1) or (b)(2) of this section are allowed if the owner or operator can demonstrate that the liquid level in the tank is always above the entire opening of the fill pipe. Documentation providing such demonstration must be made available for inspection by the Administrator's delegated representative during the course of a site visit.
- (c) Gasoline storage tanks with a capacity of less than 250 gallons are not required to comply with the submerged fill requirements in paragraph (b) of this section, but must comply only with all of the requirements in \$63.11116.
- (d) You must have records available within 24 hours of a request by the Administrator to document your gasoline throughput.
- (e) You must submit the applicable notifications as required under §63.11124(a).
- (f) You must comply with the requirements of this subpart by the applicable dates contained in §63.11113.

[73 FR 1945, Jan. 10, 2008, as amended at 73 FR 12276, Mar. 7, 2008; 76 FR 4182, Jan. 24, 2011]

# § 63.11118 Requirements for facilities with monthly throughput of 100,000 gallons of gasoline or more.

- (a) You must comply with the requirements in §§63.11116(a) and 63.11117(b).
- (b) Except as provided in paragraph (c) of this section, you must meet the requirements in either paragraph (b)(1) or paragraph (b)(2) of this section.
- (1) Each management practice in Table 1 to this subpart that applies to your GDF.
- (2) If, prior to January 10, 2008, you satisfy the requirements in both paragraphs (b)(2)(i) and (ii) of this section, you will be deemed in compliance with this subsection.
- (i) You operate a vapor balance system at your GDF that meets the requirements of either paragraph (b)(2)(i)(A) or paragraph (b)(2)(i)(B) of this section.

#### §63.11120

- (A) Achieves emissions reduction of at least 90 percent.
- (B) Operates using management practices at least as stringent as those in Table 1 to this subpart.
- (ii) Your gasoline dispensing facility is in compliance with an enforceable State, local, or tribal rule or permit that contains requirements of either paragraph (b)(2)(i)(A) or paragraph (b)(2)(i)(B) of this section.
- (c) The emission sources listed in paragraphs (c)(1) through (3) of this section are not required to comply with the control requirements in paragraph (b) of this section, but must comply with the requirements in §63.11117.
- (1) Gasoline storage tanks with a capacity of less than 250 gallons that are constructed after January 10, 2008.
- (2) Gasoline storage tanks with a capacity of less than 2,000 gallons that were constructed before January 10, 2008
- (3) Gasoline storage tanks equipped with floating roofs, or the equivalent.
- (d) Cargo tanks unloading at GDF must comply with the management practices in Table 2 to this subpart.
- (e) You must comply with the applicable testing requirements contained in §63.11120.
- (f) You must submit the applicable notifications as required under \$63.11124.
- (g) You must keep records and submit reports as specified in §§ 63.11125 and 63.11126.
- (h) You must comply with the requirements of this subpart by the applicable dates contained in §63.11113.

[73 FR 1945, Jan. 10, 2008, as amended at 73 FR 12276, Mar. 7, 2008]

# TESTING AND MONITORING REQUIREMENTS

# § 63.11120 What testing and monitoring requirements must I meet?

- (a) Each owner or operator, at the time of installation, as specified in §63.11113(e), of a vapor balance system required under §63.11118(b)(1), and every 3 years thereafter, must comply with the requirements in paragraphs (a)(1) and (2) of this section.
- (1) You must demonstrate compliance with the leak rate and cracking pressure requirements, specified in

- item 1(g) of Table 1 to this subpart, for pressure-vacuum vent valves installed on your gasoline storage tanks using the test methods identified in paragraph (a)(1)(i) or paragraph (a)(1)(ii) of this section.
- (i) California Air Resources Board Vapor Recovery Test Procedure TP-201.1E,—Leak Rate and Cracking Pressure of Pressure/Vacuum Vent Valves, adopted October 8, 2003 (incorporated by reference, see §63.14).
- (ii) Use alternative test methods and procedures in accordance with the alternative test method requirements in §63.7(f).
- (2) You must demonstrate compliance with the static pressure performance requirement specified in item 1(h) of Table 1 to this subpart for your vapor balance system by conducting a static pressure test on your gasoline storage tanks using the test methods identified in paragraphs (a)(2)(i), (a)(2)(ii), or (a)(2)(iii) of this section.
- (i) California Air Resources Board Vapor Recovery Test Procedure TP-201.3,—Determination of 2-Inch WC Static Pressure Performance of Vapor Recovery Systems of Dispensing Facilities, adopted April 12, 1996, and amended March 17, 1999 (incorporated by reference, see §63.14).
- (ii) Use alternative test methods and procedures in accordance with the alternative test method requirements in §63.7(f).
- (iii) Bay Area Air Quality Management District Source Test Procedure ST-30—Static Pressure Integrity Test—Underground Storage Tanks, adopted November 30, 1983, and amended December 21, 1994 (incorporated by reference, see §63.14).
- (b) Each owner or operator choosing, under the provisions of §63.6(g), to use a vapor balance system other than that described in Table 1 to this subpart must demonstrate to the Administrator or delegated authority under paragraph §63.11131(a) of this subpart, the equivalency of their vapor balance system to that described in Table 1 to this subpart using the procedures specified in paragraphs (b)(1) through (3) of this section.
- (1) You must demonstrate initial compliance by conducting an initial performance test on the vapor balance

system to demonstrate that the vapor balance system achieves 95 percent reduction using the California Air Resources Board Vapor Recovery Test Procedure TP-201.1,—Volumetric Efficiency for Phase I Vapor Recovery Systems, adopted April 12, 1996, and amended February 1, 2001, and October 8, 2003, (incorporated by reference, see §63.14).

- (2) You must, during the initial performance test required under paragraph (b)(1) of this section, determine and document alternative acceptable values for the leak rate and cracking pressure requirements specified in item 1(g) of Table 1 to this subpart and for the static pressure performance requirement in item 1(h) of Table 1 to this subpart.
- (3) You must comply with the testing requirements specified in paragraph (a) of this section.
- (c) Conduct of performance tests. Performance tests conducted for this subpart shall be conducted under such conditions as the Administrator specifies to the owner or operator based on representative performance (i.e., performance based on normal operating conditions) of the affected source. Upon request, the owner or operator shall make available to the Administrator such records as may be necessary to determine the conditions of performance tests.
- (d) Owners and operators of gasoline cargo tanks subject to the provisions of Table 2 to this subpart must conduct annual certification testing according to the vapor tightness testing requirements found in §63.11092(f).

[73 FR 1945, Jan. 10, 2008, as amended at 76 FR 4182, Jan. 24, 2011]

NOTIFICATIONS, RECORDS, AND REPORTS

# § 63.11124 What notifications must I submit and when?

- (a) Each owner or operator subject to the control requirements in §63.11117 must comply with paragraphs (a)(1) through (3) of this section.
- (1) You must submit an Initial Notification that you are subject to this subpart by May 9, 2008, or at the time you become subject to the control requirements in §63.11117, unless you meet the requirements in paragraph (a)(3) of this

section. If your affected source is subject to the control requirements in §63.11i17 only because it loads gasoline into fuel tanks other than those in motor vehicles, as defined in §63.11132, you must submit the Initial Notification by May 24, 2011. The Initial Notification must contain the information specified in paragraphs (a)(1)(1) through (iii) of this section. The notification must be submitted to the applicable EPA Regional Office and delegated State authority as specified in §63.13.

- (i) The name and address of the owner and the operator.
- (ii) The address (i.e., physical location) of the GDF.
- (iii) A statement that the notification is being submitted in response to this subpart and identifying the requirements in paragraphs (a) through (c) of §63.11117 that apply to you.
- (2) You must submit a Notification of Compliance Status to the applicable EPA Regional Office and the delegated State authority, as specified in §63.13, within 60 days of the applicable compliance date specified in §63.11113, unless you meet the requirements in paragraph (a)(3) of this section. The Notification of Compliance Status must be signed by a responsible official who must certify its accuracy, must indicate whether the source has complied with the requirements of this subpart, and must indicate whether the facilities' monthly throughput is calculated based on the volume of gasoline loaded into all storage tanks or on the volume of gasoline dispensed from all storage tanks. If your facility is in compliance with the requirements of this subpart at the time the Initial Notification required under paragraph (a)(1) of this section is due, the Notification of Compliance Status may be submitted in lieu of the Initial Notification provided it contains the information required under paragraph (a)(1) of this section.
- (3) If, prior to January 10, 2008, you are operating in compliance with an enforceable State, local, or tribal rule or permit that requires submerged fill as specified in §63.11117(b), you are not required to submit an Initial Notification or a Notification of Compliance Status under paragraph (a)(1) or paragraph (a)(2) of this section.

#### § 63.11125

- (b) Each owner or operator subject to the control requirements in §63.11118 must comply with paragraphs (b)(1) through (5) of this section.
- (1) You must submit an Initial Notification that you are subject to this subpart by May 9, 2008, or at the time you become subject to the control requirements in §63.11118. If your affected source is subject to the control requirements in §63.11118 only because it loads gasoline into fuel tanks other than those in motor vehicles, as defined in §63.11132, you must submit the Initial Notification by May 24, 2011. The Initial Notification must contain the information specified in paragraphs (b)(1)(i) through (iii) of this section. The notification must be submitted to the applicable EPA Regional Office and delegated State authority as specified in §63.13.
- (i) The name and address of the owner and the operator.
- (ii) The address (i.e., physical location) of the GDF.
- (iii) A statement that the notification is being submitted in response to this subpart and identifying the requirements in paragraphs (a) through (c) of §63.11118 that apply to you.
- (2) You must submit a Notification of Compliance Status to the applicable EPA Regional Office and the delegated State authority, as specified in §63.13, in accordance with the schedule specified in §63.9(h). The Notification of Compliance Status must be signed by a responsible official who must certify its accuracy, must indicate whether the source has complied with the requirements of this subpart, and must indicate whether the facility's throughput is determined based on the volume of gasoline loaded into all storage tanks or on the volume of gasoline dispensed from all storage tanks. If your facility is in compliance with the requirements of this subpart at the time the Initial Notification required under paragraph (b)(1) of this section is due, the Notification of Compliance Status may be submitted in lieu of the Initial Notification provided it contains the information required under paragraph (b)(1) of this section.
- (3) If, prior to January 10, 2008, you satisfy the requirements in both paragraphs (b)(3)(i) and (ii) of this section,

- you are not required to submit an Initial Notification or a Notification of Compliance Status under paragraph (b)(1) or paragraph (b)(2) of this subsection.
- (i) You operate a vapor balance system at your gasoline dispensing facility that meets the requirements of either paragraphs (b)(3)(i)(A) or (b)(3)(i)(B) of this section.
- (A) Achieves emissions reduction of at least 90 percent.
- (B) Operates using management practices at least as stringent as those in Table 1 to this subpart.
- (ii) Your gasoline dispensing facility is in compliance with an enforceable State, local, or tribal rule or permit that contains requirements of either paragraphs (b)(3)(i)(A) or (b)(3)(i)(B) of this section.
- (4) You must submit a Notification of Performance Test, as specified in §63.9(e), prior to initiating testing required by §63.11120(a) and (b).
- (5) You must submit additional notifications specified in §63.9, as applicable

[73 FR 1945, Jan. 10, 2008, as amended at 73 FR 12276, Mar. 7, 2008; 76 FR 4182, Jan. 24, 2011]

# § 63.11125 What are my recordkeeping requirements?

- (a) Each owner or operator subject to the management practices in §63.11118 must keep records of all tests performed under §63.11120(a) and (b).
- (b) Records required under paragraph (a) of this section shall be kept for a period of 5 years and shall be made available for inspection by the Administrator's delegated representatives during the course of a site visit.
- (c) Each owner or operator of a gasoline cargo tank subject to the management practices in Table 2 to this subpart must keep records documenting vapor tightness testing for a period of 5 years. Documentation must include each of the items specified in §63.11094(b)(2)(i) through (viii). Records of vapor tightness testing must be retained as specified in either paragraph (c)(1) or paragraph (c)(2) of this section.
- (1) The owner or operator must keep all vapor tightness testing records with the cargo tank.

- (2) As an alternative to keeping all records with the cargo tank, the owner or operator may comply with the requirements of paragraphs (c)(2)(i) and (ii) of this section.
- (i) The owner or operator may keep records of only the most recent vapor tightness test with the cargo tank, and keep records for the previous 4 years at their office or another central location.
- (ii) Vapor tightness testing records that are kept at a location other than with the cargo tank must be instantly available (e.g., via e-mail or facsimile) to the Administrator's delegated representative during the course of a site visit or within a mutually agreeable time frame. Such records must be an exact duplicate image of the original paper copy record with certifying signatures.
- (d) Each owner or operator of an affected source under this subpart shall keep records as specified in paragraphs (d)(1) and (2) of this section.
- (1) Records of the occurrence and duration of each malfunction of operation (i.e., process equipment) or the air pollution control and monitoring equipment.
- (2) Records of actions taken during periods of malfunction to minimize emissions in accordance with §63.11115(a), including corrective actions to restore malfunctioning process and air pollution control and monitoring equipment to its normal or usual manner of operation.

[73 FR 1945, Jan. 10, 2008, as amended at 76 FR 4183, Jan. 24, 2011]

# § 63.11126 What are my reporting requirements?

- (a) Each owner or operator subject to the management practices in §63.11118 shall report to the Administrator the results of all volumetric efficiency tests required under §63.11120(b). Reports submitted under this paragraph must be submitted within 180 days of the completion of the performance testing.
- (b) Each owner or operator of an affected source under this subpart shall report, by March 15 of each year, the number, duration, and a brief description of each type of malfunction which occurred during the previous calendar year and which caused or may have

caused any applicable emission limitation to be exceeded. The report must also include a description of actions taken by an owner or operator during a malfunction of an affected source to minimize emissions in accordance with §63.11115(a), including actions taken to correct a malfunction. No report is necessary for a calendar year in which no malfunctions occurred.

[76 FR 4183, Jan. 24, 2011]

OTHER REQUIREMENTS AND INFORMATION

# §63.11130 What parts of the General Provisions apply to me?

Table 3 to this subpart shows which parts of the General Provisions apply to you.

# § 63.11131 Who implements and enforces this subpart?

- (a) This subpart can be implemented and enforced by the U.S. EPA or a delegated authority such as the applicable State, local, or tribal agency. If the U.S. EPA Administrator has delegated authority to a State, local, or tribal agency, then that agency, in addition to the U.S. EPA, has the authority to implement and enforce this subpart. Contact the applicable U.S. EPA Regional Office to find out if implementation and enforcement of this subpart is delegated to a State, local, or tribal agency.
- (b) In delegating implementation and enforcement authority of this subpart to a State, local, or tribal agency under subpart E of this part, the authorities contained in paragraph (c) of this section are retained by the Administrator of U.S. EPA and cannot be transferred to the State, local, or tribal agency.
- (c) The authorities that cannot be delegated to State, local, or tribal agencies are as specified in paragraphs (c)(1) through (3) of this section.
- (1) Approval of alternatives to the requirements in §§63.11116 through 63.11118 and 63.11120.
- (2) Approval of major alternatives to test methods under §63.7(e)(2)(ii) and (f), as defined in §63.90, and as required in this subpart.
- (3) Approval of major alternatives to recordkeeping and reporting under

#### §63.11132

§63.10(f), as defined in §63.90, and as required in this subpart.

# § 63.11132 What definitions apply to this subpart?

As used in this subpart, all terms not defined herein shall have the meaning given them in the Clean Air Act (CAA), or in subparts A and BBBBB of this part. For purposes of this subpart, definitions in this section supersede definitions in other parts or subparts.

Dual-point vapor balance system means a type of vapor balance system in which the storage tank is equipped with an entry port for a gasoline fill pipe and a separate exit port for a vapor connection.

Gasoline means any petroleum distillate or petroleum distillate/alcohol blend having a Reid vapor pressure of 27.6 kilopascals or greater, which is used as a fuel for internal combustion engines.

Gasoline cargo tank means a delivery tank truck or railcar which is loading or unloading gasoline, or which has loaded or unloaded gasoline on the immediately previous load.

Gasoline dispensing facility (GDF) means any stationary facility which dispenses gasoline into the fuel tank of a motor vehicle, motor vehicle engine, nonroad vehicle, or nonroad engine, including a nonroad vehicle or nonroad engine used solely for competition. These facilities include, but are not limited to, facilities that dispense gasoline into on- and off-road, street, or highway motor vehicles, lawn equipment, boats, test engines, landscaping equipment, generators, pumps, and other gasoline-fueled engines and equipment.

Monthly throughput means the total volume of gasoline that is loaded into, or dispensed from, all gasoline storage tanks at each GDF during a month. Monthly throughput is calculated by summing the volume of gasoline loaded into, or dispensed from, all gasoline storage tanks at each GDF during the current day, plus the total volume of gasoline loaded into, or dispensed from, all gasoline storage tanks at each GDF

during the previous 364 days, and then dividing that sum by 12.

Motor vehicle means any self-propelled vehicle designed for transporting persons or property on a street or highway.

Nonroad engine means an internal combustion engine (including the fuel system) that is not used in a motor vehicle or a vehicle used solely for competition, or that is not subject to standards promulgated under section 7411 of this title or section 7521 of this title.

Nonroad vehicle means a vehicle that is powered by a nonroad engine, and that is not a motor vehicle or a vehicle used solely for competition.

Submerged filling means, for the purposes of this subpart, the filling of a gasoline storage tank through a submerged fill pipe whose discharge is no more than the applicable distance specified in §63.11117(b) from the bottom of the tank. Bottom filling of gasoline storage tanks is included in this definition.

Vapor balance system means a combination of pipes and hoses that create a closed system between the vapor spaces of an unloading gasoline cargo tank and a receiving storage tank such that vapors displaced from the storage tank are transferred to the gasoline cargo tank being unloaded.

Vapor-tight means equipment that allows no loss of vapors. Compliance with vapor-tight requirements can be determined by checking to ensure that the concentration at a potential leak source is not equal to or greater than 100 percent of the Lower Explosive Limit when measured with a combustible gas detector, calibrated with propane, at a distance of 1 inch from the source.

Vapor-tight gasoline cargo tank means a gasoline cargo tank which has demonstrated within the 12 preceding months that it meets the annual certification test requirements in §63.11092(f) of this part.

[73 FR 1945, Jan. 10, 2008, as amended at 76 FR 4183, Jan. 24, 2011]

Table 1 to Subpart CCCCCC of Part 63—Applicability Criteria and Management Practices for Gasoline Dispensing Facilities With Monthly Throughput of 100,000 Gallons of Gasoline or More<sup>1</sup>

If you own or operate	Then you must	
A new, reconstructed, or existing GDF subject to § 63.11118.	Install and operate a vapor balance system on your gasoline storage tanks that meets the design criteria in paragraphs (a) through (h).  (a) All vapor connections and lines on the storage tank shall be equipped with closures that seal upon disconnect.  (b) The vapor line from the gasoline storage tank to the gasoline cargo tank shall be vapor-tight, as defined in §63.11132.  (c) The vapor balance system shall be designed such that the pressure in the tank truck does not exceed 18 inches water pressure or 5.9 inches water vacuum during product transfer.  (d) The vapor recovery and product adaptors, and the method of connection with the delivery elbow, shall be designed so as to prevent the overtightening or loosening of fittings during normal delivery operations.  (e) If a gauge well separate from the fill tube is used, it shall be provided with a submerged drop tube that extends the same distance from the bottom of the storage tank as specified in §63.11117(b).  (f) Liquid fill connections for all systems shall be equipped with vapor-tight caps.  (g) Pressure/vacuum (PV) vent valves shall be installed on the storage tank vent pipes. The pressure specifications for PV vent valves shall be: a positive pressure setting of 6.0 to 10.0 inches of water. The total leak rate of all PV vent valves at an affected facility, including connections, shall not exceed 0.17 cubic foot per hour at a pressure of 2.0 inches of water.  (h) The vapor balance system shall be capable of meeting the static pressure supperformance requirement of the following equation:	
	Pf = 2e <sup>-500.887/v</sup>	
	Where:	
A new or reconstructed GDF, or any storage tank(s) constructed after November 9, 2006, at	Pf = Minimum allowable final pressure, inches of water. v = Total ullage affected by the test, gallons. e = Dimensionless constant equal to approximately 2.718. 2 = The initial pressure, inches water. Equip your gasoline storage tanks with a dual-point vapor balance system as defined in §63.11132, and comply with the requirements of item 1 in	

¹ The management practices specified in this Table are not applicable if you are complying with the requirements in §63.11118(b)(2), except that if you are complying with the requirements in §63.11118(b)(2)(i)(B), you must operate using management practices at least as stringent as those listed in this Table.

[73 FR 1945, Jan. 10, 2008, as amended at 73 FR 35944, June 25, 2008; 76 FR 4184, Jan. 24, 2011]

TABLE 2 TO SUBPART CCCCCC OF PART 63—APPLICABILITY CRITERIA AND MANAGEMENT PRACTICES FOR GASOLINE CARGO TANKS UNLOADING AT GASOLINE DISPENSING FACILITIES WITH MONTHLY THROUGHPUT OF 100,000 GALLONS OF GASOLINE OR MORE

If you own or operate	Then you must	
A gasoline cargo tank	Not unload gasoline into a storage tank at a GDF subject to the control requirements in this subpart unless the following conditions are met:  (i) All hoses in the vapor balance system are properly connected,  (ii) The adapters or couplers that attach to the vapor line on the storage tank have closures that seal upon disconnect,  (iii) All vapor return hoses, couplers, and adapters used in the gasoline delivery are vapor tight,  (iv) All tank truck vapor return equipment is compatible in size and forms a vapor-tight connection with the vapor balance equipment on the GDF storage tank, and  (v) All hatches on the tank truck are closed and securely fastened.  (vi) The filling of storage tanks at GDF shall be limited to unloading from vapor-tight gasoline cargo tanks. Documentation that the cargo tank has met the specifications of EPA Method 27 shall be carried with the cargo tank, as specified in §63.11125(c).	

TABLE 3 TO SUBPART CCCCCC OF PART 63—APPLICABILITY OF GENERAL PROVISIONS

Citation	Subject	Brief description	Applies to subpart CCCCCC
§ 63.1	Applicability	Initial applicability determination; applicability after standard established; permit requirements; extensions, notifications.	Yes, specific require- ments given in § 63.11111.
§ 63.1(c)(2)	Title V Permit	Requirements for obtaining a title V permit from the applicable permitting authority.	Yes, §63.11111(f) of subpart CCCCC ex empts identified area sources from the obligation to obtain title operating permits.
§ 63.2	Definitions	Definitions for part 63 standards	Yes, additional defini- tions in §63.11132.
§ 63.4	Units and Abbreviations Prohibited Activities and Circumvention.	Units and abbreviations for part 63 standards Prohibited activities; Circumvention, severability	Yes. Yes.
§ 63.5	Construction/Reconstruction.	Applicability; applications; approvals	Yes, except that these notifications are not required for facilities subject to § 63.11116
§ 63.6(a)	Compliance with Stand- ards/Operation & Maintenance—Appli- cability.	General Provisions apply unless compliance ex- tension; General Provisions apply to area sources that become major.	Yes.
§ 63.6(b)(1)–(4)	Compliance Dates for New and Recon- structed Sources.	Standards apply at effective date; 3 years after effective date; upon startup; 10 years after construction or reconstruction commences for CAA section 112(f).	Yes.
§ 63.6(b)(5)	Notification	Must notify if commenced construction or reconstruction after proposal.	Yes.
§ 63.6(b)(7)	[Reserved]. Compilance Dates for New and Recon- structed Area Sources That Be- come Major.	Area sources that become major must comply with major source standards immediately upon becoming major, regardless of whether required to comply when they were an area source.	No.
§ 63.6(c)(1)–(2)	Compliance Dates for Existing Sources.	Comply according to date in this subpart, which must be no later than 3 years after effective date; for CAA section 112(f) standards, comply within 90 days of effective date unless compliance extension.	No, § 63.11113 speci- fies the compliance dates.
§ 63.6(c)(3)–(4) § 63.6(c)(5)	[Reserved]. Compliance Dates for Existing Area Sources That Be- come Major. [Reserved].	Area sources That become major must comply with major source standards by date indicated in this subpart or by equivalent time period (e.g., 3 years).	No.
§ 63.6(e) (1)(i)	General duty to mini- mize emissions.	Operate to minimize emissions at all times; in- formation Administrator will use to determine if operation and maintenance requirements were met.	No. See § 63.11115 fo general duty require- ment.
63.6(e)(1)(ii)	malfunctions ASAP.	Owner or operator must correct malfunctions as soon as possible.	No.
§ 63.6(e)(2) § 63.6(e)(3)	[Reserved]. Startup, Shutdown, and Malfunction (SSM) Plan.	Requirement for SSM plan; content of SSM plan; actions during SSM.	No.
§ 63.6(f)(1)	Compliance Except During SSM.	You must comply with emission standards at all times except during SSM.	No.
§ 63.6(f)(2)–(3)	Methods for Deter- mining Compliance.	Compliance based on performance test, operation and maintenance plans, records, inspection.	Yes.
§ 63.6(g)(1)–(3) § 63.6(h)(1)	Alternative Standard Compliance with Opacity/Visible Emission (VE) Standards.	Procedures for getting an alternative standard You must comply with opacity/VE standards at all times except during SSM.	Yes. No.
§ 63.6(h)(2)(i)	Determining Compli- ance with Opacity/VE Standards.	If standard does not State test method, use EPA Method 9 for opacity in appendix A of part 60 of this chapter and EPA Method 22 for VE in appendix A of part 60 of this chapter	]

#### **Environmental Protection Agency**

#### Pt. 63, Subpt. CCCCCC, Table 3

Citation	Subject	Brief description	Applies to subpart CCCCCC		
§ 63.6(h)(2)(iii)	Using Previous Tests To Demonstrate Compliance With Opacity/VE Standards.	Criteria for when previous opacity/VE testing can be used to show compliance with this subpart.	No.		
63.6(h)(3) 63.6(h)(4)	[Reserved]. Notification of Opacity/	Must notify Administrator of anticipated date of observation.	No.		
63.6(h)(5)(i), (iii)-(v)	VE Observation Date. Conducting Opacity/VE Observations.	Dates and schedule for conducting opacity/VE observations.	No.		
63.6(h)(5)(ii)	Opacity Test Duration and Averaging Times.	Must have at least 3 hours of observation with 30 6-minute averages.	No.		
63.6(h)(6)	Records of Conditions During Opacity/VE Observations.	Must keep records available and allow Administrator to inspect.	No.		
63.6(h)(7)(i)	Report Continuous Opacity Monitoring System (COMS) Monitoring Data From Performance Test.	Must submit COMS data with other performance test data.	No.		
§ 63.6(h)(7)(ii)	Using COMS Instead of EPA Method 9.	Can submit COMS data instead of EPA Method 9 results even if rule requires EPA Method 9 in appendix A of part 60 of this chapter, but must notify Administrator before performance test.	No.		
§ 63.6(h)(7)(iii)	Averaging Time for COMS During Performance Test.	To determine compliance, must reduce COMS data to 6-minute averages.	No.		
§ 63.6(h)(7)(iv)	COMS Requirements	Owner/operator must demonstrate that COMS performance evaluations are conducted according to §63.8(e); COMS are properly maintained and operated according to §63.8(c) and data quality as §63.8(d).	No.		
683.6(h)(7)(v)	Determining Compli- ance with Opacity/VE Standards.	COMS is probable but not conclusive evidence of compliance with opacity standard, even if EPA Method 9 observation shows otherwise. Requirements for COMS to be probable evidence-proper maintenance, meeting Performance Specification 1 in appendix B of part 60 of this chapter, and data have not been altered.	No.		
§ 63.6(h)(8)	Determining Compli- ance with Opacity/VE Standards.	Administrator will use all COMS, EPA Method 9 (In appendix A of part 60 of this chapter), and EPA Method 22 (in appendix A of part 60 of this chapter) results, as well as information about operation and maintenance to determine compliance.	No.		
63.6(h)(9)	Adjusted Opacity Stand- ard.	Procedures for Administrator to adjust an opacity standard.	No.		
63.6(i)(1)–(14)	Compliance Extension	Procedures and criteria for Administrator to grant compliance extension.	Yes.		
63.6(j)	Presidential Compliance Exemption. Performance Test	President may exempt any source from require- ment to comply with this subpart.	Yes.		
63.7(a)(3)	Dates. CAA Section 114 Au-	Dates for conducting initial performance testing; must conduct 180 days after compliance date. Administrator may require a performance test	Yes.		
63.7(b)(1)	thority. Notification of Perform-	under CAA section 114 at any time.  Must notify Administrator 60 days before the	Yes.		
63.7(b)(2)	ance Test. Notification of Re- scheduling.	test.  If have to reschedule performance test, must notify Administrator of rescheduled date as	Yes.		
§ 63.7(c)	Quality Assurance (QA)/ Test Plan.	soon as practicable and without delay.  Requirement to submit site-specific test plan 60 days before the test or on date Administrator agrees with; test plan approval procedures; performance audit requirements; internal and external QA procedures for testing.	Yes.		
§ 63.7(d)	Testing Facilities Conditions for Conducting Performance Tests.	Requirements for testing facilities	Yes. No, § 63.11120(c) specifies conditions for conducting per- formance tests.		

#### 40 CFR Ch. I (7-1-13 Edition)

Citation	Subject	Brief description	Applies to subpart CCCCCC		
§ 63.7(e)(2)	Conditions for Conducting Performance Tests.	Must conduct according to this subpart and EPA test methods unless Administrator approves alternative.	Yes.		
§ 63.7(e)(3)	Test Run Duration	Must have three test runs of at least 1 hour each; compliance is based on arithmetic mean of three runs; conditions when data from an additional test run can be used.	Yes.		
§ 63.7(f)	Alternative Test Method	Procedures by which Administrator can grant approval to use an intermediate or major change, or alternative to a test method.	Yes.		
§ 63.7(g)	Performance Test Data Analysis.	Must include raw data in performance test re- port; must submit performance test data 60 days after end of test with the Notification of Compliance Status; keep data for 5 years.	Yes.		
§ 63.7(h)	l i	Procedures for Administrator to waive perform- ance test.	Yes.		
63.8(a)(1)	Applicability of Monitoring Requirements.	Subject to all monitoring requirements in standard.	Yes.		
§ 63.8(a)(2)	Performance Specifications. [Reserved].	Performance Specifications in appendix B of 40 CFR part 60 apply.	Yes.		
§ 63.8(a)(3) § 63.8(a)(4)		Monitoring requirements for flares in §63.11 apply.	Yes.		
§ 63.8(b)(1)	, and the second	Must conduct monitoring according to standard unless Administrator approves alternative.	Yes.		
§ 63.8(b)(2)–(3)	Multiple Effluents and Multiple Monitoring Systems.	Specific requirements for installing monitoring systems; must install on each affected source or after combined with another affected source before it is released to the atmosphere provided the monitoring is sufficient to demonstrate compliance with the standard; if more than one monitoring system on an emission point, must report all monitoring system results, unless one monitoring system is a backup.	No.		
63.8(c)(1)	Monitoring System Op- eration and Mainte- nance.	Maintain monitoring system in a manner con- sistent with good air pollution control practices.	No.		
§ 63.8(c)(1)(i)–(iii)	Operation and Mainte- nance of Continuous Monitoring Systems (CMS).	Must maintain and operate each CMS as speci- fied in §63.6(e)(1); must keep parts for rou- tine repairs readily available; must develop a written SSM plan for CMS, as specified in §63.6(e)(3).	No.		
§ 63.8(c)(2)–(8)	CMS Requirements	Must install to get representative emission or parameter measurements; must verify oper- ational status before or at performance test.	No.		
§ 63.8(d)	CMS Quality Control	Requirements for CMS quality control, including calibration, etc.; must keep quality control plan on record for 5 years; keep old versions for 5 years after revisions.	No.		
§ 63.8(e)	uation.	Notification, performance evaluation test plan, reports.	No.		
63.8(f)(1)–(5)	Method.	Procedures for Administrator to approve alternative monitoring.  Procedures for Administrator to approve alter-	No.		
§ 63.8(f)(6)	Alternative to Relative Accuracy Test.	native relative accuracy tests for continuous emissions monitoring system (CEMS).	110.		
§ 63.8(g)	Data Reduction	COMS 6-minute averages calculated over at least 36 evenly spaced data points; CEMS 1 hour averages computed over at least 4 equally spaced data points; data that cannot be used in average.	No.		
§ 63.9(a)	ments.	Applicability and State delegation	Yes.		
§ 63.9(b)(1)–(2), (4)–(5)	Initial Notifications	Submit notification within 120 days after effective date, notification of Intent to construct/reconstruct, notification of commencement of construction/reconstruction, notification of startup; contents of each.			
§ 63.9(c)	Request for Compliance Extension.	Can request if cannot comply by date or if in- stalled best available control technology or lowest achievable emission rate.	Yes.		

#### **Environmental Protection Agency**

#### Pt. 63, Subpt. CCCCC, Table 3

Citation	Subject	Brief description	Applies to subpart CCCCCC
§ 63.9(d)	Notification of Special Compliance Require- ments for New Sources.	For sources that commence construction be- tween proposal and promulgation and want to comply 3 years after effective date.	Yes.
§ 63.9(e)	Notification of Perform- ance Test.	Notify Administrator 60 days prior	Yes.
§ 63.9(f)	Notification of VE/Opac- ity Test.	Notify Administrator 30 days prior	No.
§ 63.9(g)	Additional Notifications when Using CMS.	Notification of performance evaluation; notifica- tion about use of COMS data; notification that exceeded criterion for relative accuracy alter- native.	Yes, however, there are no opacity standards.
§ 63.9(h)(1)–(6)	Notification of Compli- ance Status.	Contents due 60 days after end of performance test or other compliance demonstration, except for opacity/VE, which are due 30 days after; when to submit to Federal vs. State authority.	Yes, however, there are no opacity standards.
§ 63.9(i)	Adjustment of Submittal Deadlines.	Procedures for Administrator to approve change when notifications must be submitted.	Yes.
§ 63.9(j)	Change in Previous Information.	Must submit within 15 days after the change	Yes.
§ 63.10(a)	Recordkeeping/Report- ing.	Applies to all, unless compliance extension; when to submit to Federal vs. State authority; procedures for owners of more than one source.	Yes.
§ 63.10(b)(1)	Recordkeeping/Report- ing.	General requirements; keep all records readily available; keep for 5 years.	Yes.
§ 63.10(b)(2)(i)	Records related to SSM	Recordkeeping of occurrence and duration of startups and shutdowns.	No.
§ 63.10(b)(2)(ii)	Records related to SSM	Recordkeeping of malfunctions	No. See § 63.11125(d) for recordkeeping of (1) occurrence and duration and (2) ac- tions taken during malfunction.
§ 63.10(b)(2)(iii)		Recordkeeping of maintenance on air pollution control and monitoring equipment.	Yes.
§ 63.10(b)(2)(iv)	Records Related to SSM.	Actions taken to minimize emissions during SSM.	No.
§ 63.10(b)(2)(v)	Records Related to SSM.	Actions taken to minimize emissions during SSM.	No.
§ 63.10(b)(2)(vi)–(xi)	CMS Records	Malfunctions, inoperative, out-of-control periods	No.
§ 63.10(b)(2)(xii) § 63.10(b)(2)(xiii)	Records	Records when under waiver	Yes.
§ 63.10(b)(2)(xiii)	Records	Records when using alternative to relative accuracy test.  All documentation supporting initial Notification	Yes.
		and Notification of Compliance Status.	163.
§ 63.10(b)(3)	Records	Applicability determinations	Yes.
§ 63.10(d)(1)§	Records	Additional records for CMS	No. Yes.
§ 63.10(d)(2)	quirements.  Report of Performance  Test Results.	When to submit to Federal or State authority	Yes.
§ 63.10(d)(3)	Reporting Opacity or	What to report and when	No.
§ 63.10(d)(4)	VE Observations. Progress Reports	Must submit progress reports on schedule if under compliance extension.	Yes.
§ 63.10(d)(5)	SSM Reports	Contents and submission	No. See § 63.11126(b) for malfunction report- ing requirements.
§ 63.10(e)(1)–(2)	Additional CMS Reports	Must report results for each CEMS on a unit; written copy of CMS performance evaluation; two-three copies of COMS performance evaluation.	No.
§ 63.10(e)(3)(i)-(iii)	Reports	Schedule for reporting excess emissions	No

Citation	Subject	Brief description	Applies to subpart CCCCC
§ 63.10(e)(3)(iv)–(v)	Excess Emissions Reports.	Requirement to revert to quarterly submission if there is an excess emissions and parameter monitor exceedances (now defined as deviations); provision to request semiannual report by 30th day following end of quarter or calendar half; if there has not been an exceedance or excess emissions (now defined as deviations), report contents in a statement that there have been no deviations; must submit report containing all of the information in \$\frac{85}{858.86(7)\to 48} and \frac{63}{10c(16)=113}.	No.
§ 63.10(e)(3)(iv)–(v)	Excess Emissions Reports.	Requirement to revert to quarterly submission if there is an excess emissions and parameter monitor exceedances (now defined as deviations); provision to request semiannual reporting after compliance for 1 year; submit report by 30th day following end of quarter or calendar half; if there has not been an exceedance or excess emissions (now defined as deviations), report contents in a statement that there have been no deviations; must submit report containing all of the information in §§ 63.8(c)(7)–(8) and 63.10(c)(5)–(13).	No, §63.11130(K) specifies excess emission events for this subpart.
§ 63.10(e)(3)(vi)–(viii)	Excess Emissions Report and Summary Report.	Requirements for reporting excess emissions for CMS; requires all of the information in §§ 63.10(c)(5)–(13) and 63.8(c)(7)–(8).	No.
§ 63.10(e)(4)	Reporting COMS Data	Must submit COMS data with performance test data.	No.
§ 63.10(f)	Waiver for Record- keeping/Reporting.	Procedures for Administrator to waive	Yes.
§ 63.11(b)	Flares	Requirements for flares	No.
§ 63.12	Delegation	State authority to enforce standards	Yes.
§ 63.13	Addresses	Addresses where reports, notifications, and requests are sent.	Yes.
§ 63.14	Incorporations by Ref- erence.	Test methods incorporated by reference	Yes.
§ 63.15	Availability of Informa- tion.	Public and confidential information	Yes.

[73 FR 1945, Jan. 10, 2008, as amended at 76 FR 4184, Jan. 24, 2011]

# Subpart DDDDDD—National Emission Standards for Hazardous Air Pollutants for Polyvinyl Chloride and Copolymers Production Area Sources

Source: 72 FR 2943, Jan. 23, 2007, unless otherwise noted.

APPLICABILITY AND COMPLIANCE DATES

# §63.11140 Am I subject to this subpart?

(a) On or before April 17, 2012, you are subject to this subpart if you own or operate a plant specified in §61.61(c) of this chapter that produces polyvinyl chloride (PVC) or copolymers and is an area source of hazardous air pollutant (HAP) emissions. After April 17, 2012, you are subject to the requirements in

this subpart if you own or operate one or more polyvinyl chloride and copolymers process units (PVCPU), as defined in §63.12005, that are located at, or are part of, an area source of HAP.

(b) On or before April 17, 2012, this subpart applies to each new or existing affected source. The affected source is the collection of all equipment and activities in vinyl chloride service necessary to produce PVC and copolymers. An affected source does not include portions of your PVC and copolymers production operations that meet the criteria in §61.60(b) or (c) of this chapter. After April 17, 2012, this subpart applies to each polyvinyl chloride and copolymers production affected source. The polyvinyl chloride and copolymers production affected source is the facility-wide collection of PVCPU, storage vessels, heat exchange systems, surge Appendix K
Examples of Loading Rack Calculations

#### Martin Operating Partnership, L.P. Emission Calculations for SN-14, SN-15, and SN-16

SN-14

Distillate Loading Rack

VOC Emissions

0.5 lb/hr

HAP Emission

0.2 ton/yr

Pollutant	Emission Factor	Emission Rate	Emission Rate	Comments
	(wt fraction in vapor)	(lb/hr)	(ton/yr)	
Benzene	1.17E-03	5.85E-04	2.34E-04	Specific Condition 23 of Permit No. 1227-AOP-21
Cumene	1.80E-04	9.00E-05	3.60E-05	Specific Condition 23 of Permit No. 1227-AOP-21
Cyclohexane	1.22E-02	6.10E-03	2.44E-03	Specific Condition 23 of Permit No. 1227-AOP-21
Ethylbenzene	5.89E-04	2.95E-04	1.18E-04	Specific Condition 23 of Permit No. 1227-AOP-21
n-Hexane	4.64E-03	2.32E-03	9.28E-04	Specific Condition 23 of Permit No. 1227-AOP-21
Naphthalene	1.54E-04	7.70E-05	3.08E-05	Specific Condition 23 of Permit No. 1227-AOP-21
Toluene	2.24E-03	1.12E-03	4.48E-04	Specific Condition 23 of Permit No. 1227-AOP-21
m,p-Xylene	2.55E-03	1.28E-03	5.10E-04	Specific Condition 23 of Permit No. 1227-AOP-21
o-Xylene	8.10E-04	4.05E-04	1.62E-04	Specific Condition 23 of Permit No. 1227-AOP-21

#### SN-15

#### Asphalt and Black Oil Truck Loading #1

In a given hour, MOP can only load either asphalt or black oil, but not both. Maximum hourly emission will be the higher of the two VOC emissions calculated. Yearly emissions will be the sum of the two VOC emissions calculated.

Asphalt Loading Emissions VOC Emissions

0.09 lb/hr

0.08 ton/yr

#### **HAP Emissions**

Pollutant	Emission	Emission	Emission	Comments
	Factor	Rate	Rate	
	(wt %)	(lb/hr)	(ton/vr)	
Benzene	0.0345%	3.11E-05	2.76E-05	Specific Condition 22 of Permit No. 1227-AOP-21
Biphenyl	0.0100%	9.00E-06	8.00E-06	Emission Factor from the Radian Corporation in 1993.
Cumene	3.68%	3.31E-03	2.94E-03	Emission Factor from the Radian Corporation in 1993.
Ethylbenzene	75.9%	6.83E-02	6.07E-02	Emission Factor from the Radian Corporation in 1993.
Toluene	100.0%	9.00E-02	8.00E-02	Emission Factor from the Radian Corporation in 1993.
Xvlene	100.0%	9.00E-02	8.00E-02	Emission Factor from the Radian Corporation in 1993

Black Oil Loading Emissions

**VOC Emissions** 

0.10 lb/hr

0.04 ton/yr

#### HAP Emissions

Pollutant	Emission	Emission	Emission	Comments
	Factor (wt %)	Rate (lb/hr)	Rate (ton/vr)	
Benzene	0.274%	2.74E-04	1.10E-04	Emission factor based on 80% lube oil and 20% asphalt
Biphenyl	0.002%	2.00E-06	8.00E-07	Emission factor based on 80% lube oil and 20% asphalt
Cumene	0.784%	7.84E-04	3.14E-04	Emission factor based on 80% lube oil and 20% asphalt
Cyclohexane	4.77%	4.77E-03	1.91E-03	Emission factor based on 80% lube oil and 20% asphalt
Ethylbenzene	15.3%	1.53E-02	6.12E-03	Emission factor based on 80% lube oil and 20% asphalt
n-Hexane	2.19%	2.19E-03	8.76E-04	Emission factor based on 80% lube oil and 20% asphalt
Naphthalene	0.046%	4.60E-05	1.84E-05	Emission factor based on 80% lube oil and 20% asphalt
Toluene	20.5%	2.05E-02	8.20E-03	Emission factor based on 80% lube oil and 20% asphalt
m,p-Xylene	20.5%	2.05E-02	8.20E-03	Emission factor based on 80% lube oil and 20% asphalt
o-Xylene	0.143%	1.43E-04	5.72E-05	Emission factor based on 80% lube oil and 20% asphalt
Xylene		2.06E-02	8.26E-03	Emission factor based on 80% lube oil and 20% asphalt

#### Total Emissions for SN-15

Pollutant	Emission Rate (lb/hr)	Emission Rate (ton/vr)		
Benzene	2.74E-04	1.37E-04		
Biphenyl	9.00E-06	8.80E-06		
Cumene	3.31E-03	3.26E-03		
Cyclohexane	4.77E-03	1.91E-03		
Ethylbenzene	6.83E-02	6.68E-02		
n-Hexane	2.19E-03	8.76E-04		
Naphthalene	4.60E-05	1.84E-05		
Toluene	9.00E-02	8.82E-02		
Xylene	9.00E-02	8.83E-02		

SN-16

Asphalt and Black Oil Truck Loading #2

Emissions are the same as SN-15

# Martin Operating Partnership, L.P. Emission Calculations for SN-17, SN-18, and SN-21

SN-17 SN-18 SN-21 Lube Oil Truck Loading Rack Lube Oil Truck Loading Rack Lube Oil Rail Car Loading Rack

VOC Emissions

SN-17 SN-18 SN-21 0.1 lb/hr 0.1 lb/hr 0.2 lb/hr 0.3 ton/yr

Total

HAP Emission

Pollutant	Emission Factor (wt fraction in	SN-17 Emission Rate	SN-18 Emission Rate	SN-21 Emission Rate	Total Emission Rate	Comments
· · · · · · · · · · · · · · · · · · ·	vapor)	(lb/hr)	(lb/hr)	(lb/hr)	(ton/vr)	RUNER 08
Benzene	3.34E-03	3.34E-04	3.34E-04	6.68E-04	1.00E-03	Specific Condition 23 of Permit No. 1227-AOP-21
Cumene	6.06E-04	6.06E-05	6.06E-05	1.21E-04	1.82E-04	Specific Condition 23 of Permit No. 1227-AOP-21
Cyclohexane	5.96E-02	5.96E-03	5.96E-03	1.19E-02	1.79E-02	Specific Condition 23 of Permit No. 1227-AOP-21
Ethylbenzene	1.54E-03	1.54E-04	1.54E-04	3.08E-04	4.62E-04	Specific Condition 23 of Permit No. 1227-AOP-21
n-Hexane	2.73E-02	2.73E-03	2.73E-03	5.46E-03	8.19E-03	Specific Condition 23 of Permit No. 1227-AOP-21
Naphthalene	5.72E-04	5.72E-05	5.72E-05	1.14E-04	1.72E-04	Specific Condition 23 of Permit No. 1227-AOP-21
Toluene	6.49E-03	6.49E-04	6.49E-04	1.30E-03	1.95E-03	Specific Condition 23 of Permit No. 1227-AOP-21
m,p-Xylene	6.31E-03	6.31E-04	6.31E-04	1.26E-03	1.89E-03	Specific Condition 23 of Permit No. 1227-AOP-21
o-Xylene	1.79E-03	1.79E-04	1.79E-04	3.58E-04	5.37E-04	Specific Condition 23 of Permit No. 1227-AOP-21

Appendix M Inventory of Tanks

Martin Operating Partnership, L.P. Summary Table for Tanks (SN-27a through SN-27I)

Source No.	Product	Reid Vapor Pressure	Tank Number	Date Built	Material	Tank Type	Height	Diameter	Gallons	Sum of Gallons	No. of Turnovers	Annual Usage (gal/yr)	Totals (gal/yr)	Tank VOC Working Loss (lb/yr)	Tank VOC Breathing Loss (lb/yr)	Tank VOC Total Emissions (lb/yr)
			256	1981	Steel	Fixed	20'	12'	16,800		4	70,588		57.18	150.35	207,53
SN-27a	Refinery Additive	N/A	259	1970	Steel	Fixed	30'	6'4"	6,300		4	26,471		23.87	55.12	78.99
011 274	Tronnery Additive	INA	260	1970	Steel	Fixed	30'	6'4"	6,300		4	26,471		23.87	55.12	78.99
			261	1970	Steel	Fixed	30'	6'4"	6,300	35,700	4	26,471	150,000	23.87	55.12	78.99
			P053	1962	Steel	Fixed	12'	10	7,056		10	71,429		69.6	81.65	151.25
			P054	1962	Steel	Fixed	12'	10	7,056		10	71,429		69.6	81.65	151.25
	Packaging Plant		P055	1962	Steel	Fixed	12'	10	7,056		10	71,429		69.6	81.65	151.25
SN-27b	Additive	N/A	P056	1962	Steel	Fixed	12'	10	7,056		10	71,429		69.6	81.65	151.25
			P057	1962	Steel	Fixed	12'	10	7,056		10	71,429		69.6	81.65	
			P058	1962	Steel	Fixed	12'	10	7,056		10	71,429		69.6	81.65	151.25
			P059	1962	Steel	Fixed	12'	10	7,056	49,392	10	71,429	500,000	69.6	81.65	
			100	2011	Steel	Fixed	50'	110'	3,360,000		5	18,191,399		3,13	10.83	
			222	1970	Steel	Fixed	19'7"	37'9"	147,000		5	795,874		0.14	0.5	
SN-27c	Asphalt	too thick (<0.01)	225 228	1970 1981	Steel	Fixed Fixed	18'5" 20'	19'8" 12'	42,000 16,800		<u>5</u>	227,392 90,957		0.04	0.11	0.15 0.06
014-270	Vabileir	100 trick (<0.01)	279	1970	Steel	Fixed	21'	20'	42,000		5	227,392		0.01	0.05	
	<u> </u>		280	1970	Steel	Fixed	21'	20'	42,000		5	227,392		0.04	0.13	
			281	1970	Steel	Fixed	21'	20'	42,000	3,691,800	5	227,392	19,987,800	0.04	0.13	
	<del></del>		231	2008	Steel	Fixed	18'	20	42,000	3,091,000	31	1,320,000	19,907,000	1.47	0.13	
			232	2008	Steel	Fixed	18'	20'	42,000		31	1,320,000		1.47	0.05	
	l i		233	2008	Steel	Fixed	18'	20'	42,000		31	1,320,000		1,47	0.05	
SN-27d	Black Oil	N/A	275	1970	Steel	Fixed	21'	19'	42,000		31	1,320,000		1.54	0.04	
	1		276	1970	Steel	Fixed	21'	19'	42,000		31	1,320,000		1.54	0.04	
	1		277	1970	Steel	Fixed	21'	19'	42,000		31	1,320,000		1.54	0.04	
			278	1970	Steel	Fixed	21'	19'	42,000	294,000	31	1,320,000	9,240,000	1.54	0.04	
SN-27e	Crude Oil	0.41	109	1992	Steel	Fixed	32'	82'	1,260,000	1,260,000	94	118,041,000	118,041,000	11737.4	895.14	12632.54
CN 974	Untreated Distillate	-0.4	120	2013	Steel	Fixed	52'	60'	1,050,000		18	18,728,091		503.79	320.3	824.09
SN-27f	Unit eated Distillate	<0.1	321	1985	Steel	Floating	32'	47'4"	420,000	1,470,000	18	7,491,237	26,219,328	38.19	8.29	46.48
SN-27g	Gasoline	N/A	Gasoline Tank	1999	Steel	Fixed	21'3"	8'	8,022		3	24,066	24,066	189.41	76.56	1
			210	1970	Steel	Floating	24'	30'	126,000		24	3,071,082		19.01	0.52	
			216	1970	Steel	Floating	24'5"	23'4"	84,000		24	2,047,388		16.3	0.45	
			218	1970	Steel	Fixed	24'5"	23'4"	84,000		24	2,047,388		4.18	2.03	
	1		246	2010	Steel	Fixed	40'	52'	630,000		24	15,355,409		23.06	1.23	
			314	1980	Steel	Fixed	18'	20'	42,000		24	1,023,694		2.3		
		İ	315	1980	Steel	Fixed	18'	20'	42,000	, .,	24	1,023,694		2.3		
	i l		318	1990	Steel	Fixed	18'	20'	42,000		24	1,023,694		2.3		3.4
			319	1990	Steel	Fixed	18'	20'	42,000		24	1,023,694		2.3		
			320	1990	Steel	Fixed	18'	201	42,000	ļ <b>.</b>	24	1,023,694		2,3		
SN-27h	Un-Treated Lube Oil	<0.1	322	1993	Steel	Floating	25'2"	20'7"	61,488		24	1,498,688		13.52	0.41	13.93
	Refinery		323	1994	Steel	Floating	42'2"	42'	404,250		24	9,853,054		43.57	0.68	
			325	1995	Steel	Floating	32'	34'	199,500		24	4,862,546		26.56	0.57	
	]		326	1995	Steel	Floating	40'	43'	397,152		24	9,680,050		41.81		
			328	2009	Steel	Fixed	18'	20	42,177		24	1,028,008		2.3		
			330	1998(2002)	Steel	Fixed	40'	42'	420,000	ļ	24	10,236,939		22.58		
	j 1		332	2008	Steel	Fixed	28'	28'	126,000	<b> </b>	24	3,071,082		7.03		
			333	2008	Steel	Fixed	24'	25'	87,024		24	2,121,094		4.8		
			334	2008	Steel	Fixed	30,	35'	215,670		24	5,256,668		11.76	2.28	
			335	2008	Steel	Fixed	24'	25	87,150	0 000 001	24	2,124,165	00 000 700	11.76		
	I		336	2008	Steel	Fixed	30'	35'	215,670	3,390,081	24	5,256,668	82,628,700	11./6	5.00	17.4.

#### Martin Operating Partnership, L.P. Summary Table for Tanks (SN-27a through SN-27I)

Causes N	5-4-4	5.7117	1	T = . =					or lanks (SN-							
Source No.	Product	Reid Vapor Pressure	Tank Number	Date Built	Material	Tank Type	Height	Diameter	Gallons	Sum of Gallons	No. of Turnovers	Annual Usage (gal/yr)	Totals (gal/yr)	Tank VOC Working Loss (lb/yr)	Tank VOC Breathing Loss (lb/yr)	Tank VOC Total Emissions (lb/yr)
			113	1980	Steel	Fixed	32'	68'	840,000 -		6	5,397,247	<del></del>	8.07	2.75	10.82
]			197	1980	Steel	Fixed	24'	30,	126,000		6	809,587		1,18	0.24	
			206	1980	Steel	Floating	24'	30'	126,000		6	809,587		4.76	0.52	
			223	1970	Steel	Fixed	14'7"	19'8"	31,500		6	202,397		0.31	0.07	
			224	1970	Steel	Fixed	14'7"	19'8"	31,500		6	202,397		0.31	0.07	
			229	2008	Steel	Fixed	18'	20'	42,000		6	269,862		0.4	0.07	
			230	2008	Steel	Fixed	18'	20'	42,000		6	269,862		0.4	0.07	
	1		247	2013	Steel	Fixed	52'	48'	672,000		6	4,317,797		5.37	1.28	
	ĺ		263	1970	Steel	Fixed	21'	20'	46,200		6	296,849		0.46	0.07	
			266	1980	Steel	Fixed	24'9"	30'	126,000		6	809,587		1.78	3.54	
1			269	1970	Steel	Fixed	24°	29'	126,000		6	809,587		1.18	0.24	
			270	1970	Steel	Fixed	24'	29'	126,000		6	809,587		1.18	0.24	1.42
	ļ		287	1980	Steel	Fixed	18'7"	20'	42,000		6	269,862		0.4	0.07	
	[		288	1980	Steel	Fixed	18'7*	20'	42,000		6	269,862		0.4	0.07	
	ĺ		289	1980	Steel	Fixed	18'7"	20'	42,000		6	269,862		0.4	0.07	
ļ			1 290	1980	Steel	Fixed	18'7"	20'	42,000		6	269,862		0.4	0.07	0.47
1			291	1980	Steel	Fixed	25'	26'	105,000		6	674,656		0.92	0.15	
-			292	1980	Steel	Fixed	28'	26'	115,500		. 6	742,121		1.52	2.83	
i			1 293	1980	Steel	Fixed	11'	13'	10,500		6	67,466		0.1	0.02	
1			1 294	1980	Steel	Fixed	11'	13'	10,500		6	67,466		0.1	0.02	0.12
1			_ ∟ 295	1980	Steel	Fixed	11'	13'	10,500		6	67,466		0.1	0.02	
			299	1980	Steel	Fixed	24'	30'	126,000		6	809,587		1.18	0.24	
			300	1990	Steel	Fixed	24'	30'	126,000		6	809,587		1.18	0.24	
			301	1990	Steel	Fixed	24'	18'	45,150		6	290,102		0.42	0.05	
	<b>.</b>		302	1990	Steel	Fixed	24'	18'	45,150		6	290,102		0.42	0.05	
SN-27i	Treated Lube Oil Refinery	<0.1	303	1990	Steet	Fixed	24'	18'	45,150		6	290,102		0.42	0.05	
1	Retinery		304	1990	Steel	Fixed	24'	18'	45,150		6	290,102		0.42	0.05	
	İ		305	1990	Steel	Fixed	24'	18'	45,150		6	290,102		0.42	0.05	
1			306	1990	Steet	Fixed	12'6"	12'	10,500		6	67,466		0.1	0.02	
-	1		307	1990	Steel	Fixed	12'6"	12'	10,500		6	67,466		0.1	0.02	
i			308	1970	Steel	Fixed	18'6"	32'6"	115,500		6	742,121		1.07	0.3	
			309	1993	Steel	Fixed	18'6"	20'	42,000		66	269,862		0.4		
1			<u> </u>	1993	Steel	Fixed	18'6"	20'	42,000		6	269,862		0.4	0.07	
				1998	Steel	Fixed Fixed	40'	43' 42'	420,000 -		6	2,698,623 2,698,623		4.03	0.7	
1			331	1998(2004) 2008	Steel	Fixed	30'	35'	215,000		6	1,381,438		4.03		
			338	2008	Steel	Fixed	30'	35'	215,000		6	1,381,438				2.30
1			\ 339	2008	Steel	Fixed	30'	35	215,000		6	1,381,438			0.36	
į	1		340	2008	Steel	Fixed	30'	50'	438,000 -		6	2,814,279		4.09		
1			341	2010	Steel	Fixed	31'	49'	420,000 -		6	2,698,623		4.06		
1			342	2010	Steel	Fixed	31'	60'	630,000 -		6	4,047,935		6.09		
1			343	2013	Steel	Fixed	60,	52'	1,050,000 -		6	6,746,558		9.9		
Ī			344	2013	Steel	Fixed	60,	52'	1,050,000		6	6,746,558		9.9		
	1		345	2013	Steel	Fixed	60'	52'	1,050,000		6	6,746,558		9.9		
1	ì		346	2013	Steel	Fixed	48'	52'	672,000 -		6	4,317,797		7.08		
- 1			347	2013	Steel	Fixed	60	52'	1,050,000 -		6	6,746,558		7.08		3 8.3
1			/ 348	2013	Steel	Fixed	48'	52'	672,000		6	4,317,797		7.08		
1			349	2013	Steel	Fixed	48'	52'	672,000		6	4,317,797		7.08		
ţ			350	2013	Steel	Fixed	28'	52'	210,000		6	1,349,312		4.13		3 5.3
1	į		351	2013	Steel	Fixed	28'	52'	210,000		6	1,349,312		4.13		
ì			/ 352	2013	Steel	Fixed	28'	52'	210,000		6	1,349,312		4.13		3 5.3
I			353	2013	Steel	Fixed	28'	52'	210,000	13,482,450	6	1,349,312	86,628,700	4.13		

## Martin Operating Partnership, L.P. Summary Table for Tanks (SN-27a through SN-27i)

Source No.	Product	Reid Vapor	Tank Number	Date Built	Material	Tank Type	Height	Diameter	Gallons	Sum of	No. of	Annual Usage	Totals	Tank VOC	Tank VOC	Tank VOC
		Pressure					_			Gallons	Turnovers	(gal/yr)	(gal/yr)	Working Loss (lb/yr)		Total Emissions (lb/yr)
			: P001	1989(2003)	Steel	Fixed	20'4"	12'	16,296		15	250,706		0.58	0.41	0.99
			P002	1989(2003)	Steel	Fixed	20'4"	12'	16,296		15	250,706		0.58		
	i		P003	1989(2003) 1988(2003)	Steel Steel	Fixed Fixed	20'6" 9'1"	12' 8'	16,422 3,444		15	252,644		0.58		
- 1	1		P005	1988(2003)	Steel	Fixed	9'1"	8'	3,444		15 15	52,984 52,984		0.12 0.12		
			P006	1988(2003)	Steel	Fixed	9'1"	8'	3,444		15	52,984		0.12		
			P007	1988(2003)	Steel	Fixed	20'4"	12'	16,296		15	250,706		0.58		
			P008	1988(2003)	Steel	Fixed	20'6"	12'	16,422		15	252,644		0.58		
			P009 P010	1988(2003) 2004	Steel	Fixed Fixed	24'4" 9'6"	12' 8'	20,500 3,654		15 15	315,382 56,215		0.7		
			P011	2004	Steel	Fixed	9'6"	8,	3,654		15	56,215		0.12		
			P012	1991	Steel	Fixed	31'	13'	31,500		15	484,612		1.05		1.8
			P013	1980(2004)	Steel	Fixed	20'	12'	16,800		15	258,460		0.58		
			P014 P015	1980(2004) 1990(2004)	Steel	Fixed	20'	12' 12'	16,800		15	258,460		0.58		
			P015	1990(2004)	Steel Steel	Fixed Fixed	20'	12'	16,800 16,800	2.4	15	258,460 258,460		0.58		
İ			P017	1990	Steel	Fixed	20'	12	16,800		15	258,460		0.58		
			P018	1990	Steel	Fixed	20'	12'	16,800		15	258,460		0.58		0.9
			P019	1991	Steel	Fixed	31'	13'	31,500		15	484,612		1.05		
			P020 P021	1993	Steel	Fixed	30'8"	13'	29,785		15	458,227		1.01		
			P021	1993 1990	Steel Steel	Fixed Fixed	30'	13'	19,864 29,785		15 15	305,598 458,227		0.66		
1	[		P023	1990	Steel	Fixed	30'	13'	29,785		15	458,227		1.01		
ŀ			P024	1990	Steel	Fixed	30'	13'	29,785		15	458,227		1.01	0.71	1.7
			P025	1989	Steel	Fixed	30'	13'	29,785		15	458,227		1.01		
	i		P026 P027	1982 1982	Steel	Fixed Fixed	50'	13'3" 12'	51,562 42,300		15 15	793,256 650,765		1.76		
:			P027	1983	Steel	Fixed	50'	13'	49,642		15	763,718		1.69		
			P029	1982	Steel	Fixed	40'	12'5"	36,219		15	557,211	**	1.24		
			P030	2006	Steel	Fixed	30,	35'	215,900		15	3,321,515		7.35	5.66	
SN-27j	Lube Oil Packaging	N/A	P031	1980	Steel	Fixed	50'	13'	49,642		15	763,718		1.69		
02.,	Plant	,	P032 P033	1981 1990	Stee! Stee!	Fixed Fixed	50' 32'	12 <sup>'</sup> 3"	44,072 11,466		15 15	678,026 176,399		0,41		
	1		P033	1990	Steel	Fixed	35.	8'	11,466		15	176,399		0.41		
	1		P035	1990	Steel	Fixed	32'	8'	11,466		15	176,399		0.41		0.6
	1		P036	1982	Steel	Fixed	20'	10'	11,760		15	180,922		0.4		
· ·	}		P037	1982	Steel	Fixed	20'	10'	11,760		15 15	180,922 180,922		0.4		
	1		P038	1982 1982	Steel Steel	Fixed Fixed	20'	10'	11,760 11,760		15	180,922		0.4		
ł	1		P040	1982	Steel	Fixed	20'	10'	11,760		15	180,922		0.4		
			P041	1982	Steel	Fixed	35'	10'	20,580		15	316,613		0.7		1.
	-		P042	1982	Steel	Fixed	35'	10'	20,580		15	316,613		0.7		
j			P043	1982	Steel	Fixed	35'	10'	20,580		15 15	316,613 316,613		0.7		1.
			P044 P045	1982 1982	Steel Steel	Fixed Fixed	35' 35'	10'	20,580 20,580		15	316,613		0.7		
			P046	1982	Steel	Fixed	40'	10'	23,520	·····	15	361,844		0.8		5 1.3
			P047	1982	Steel	Fixed	40'	10'	23,520		15	361,844		0.8		5 1.3
			P048	1982	Steel	Fixed	40	10'	23,520		15	361,844		0.8		
			P049 P050	1982 1982	Steel Steel	Fixed Fixed	40' 40'	10'	23,520 23,520		15 15	361,844 361,844		0.0		
1			P051	1982	Steel	Fixed	40'	10	23,520		15	361,844		0.0		
			P052	1982	Steel	Fixed	40'	10'	23,520		15	361,844		0.8	0.5	5 1.3
			P060	2012	Steet	Fixed	47'	13'6"	49,250		15	757,687		1.7		
			P061	2012	Steel	Fixed	47'	13'6"	49,250		15	757,687		1.7		
			P062	2012 2012	Steel Steel	Fixed Fixed	47°	13'6"	49,250 49,250		15 15	757,687 757,687	Barrata #* a	1.7		
			P080	2012	Steel	Fixed	52'	48'	672,000		15	10,338,387		23.9		
			/ P081	2013	Steel	Fixed	52'	48'	672,000 -		15	10,338,387		23.90		
-	1		└ P082	2013	Steel	Fixed	52'	48'	672,000		15	10,338,387		23.9		
			r- P296	1980	Steel	Fixed	24'7"	30	128,164		15	1,971,740		4.3		5 7.8
			P297	1980	Steel	Fixed	24'7"	30'	128,164		15	1,971,740		4.3		
	Ì		L P298	1980	Steel	Fixed	24'7"	30'	128,164	3,900,028	15	1,971,740	60,000,000	4.3	2 3.	5 7.8

#### Martin Operating Partnership, L.P. Summary Table for Tanks (SN-27a through SN-27l)

Source No.	Product	Reid Vapor Pressure	Tank Number	Date Built	Material	Tank Type	Height	Diameter	Gallons	Sum of Gallons	No. of Turnovers	Annual Usage (gal/yr)	Totals (gal/yr)	Tank VOC Working Loss (lb/yr)	Tank VOC Breathing Loss (lb/yr)	Tank VOC Total Emissions (lb/yr)
		·	226	1998	Steel	Fixed	16'	6'	3,465		12	43,043		2.45	2.14	4.59
			284	1970	Steel	Fixed	10'	20'	21,000		12	260,867		16.98	18.04	35.02
			501	2007	Steel	Fixed	20'	12'	16,800		12	208,694		12.23	10.99	23.22
Ì			502	2007	Steel	Fixed	20'	12'	16,800		12	208,694		12.23	10.99	23.22
ì			503	1980	Steel	Fixed	20'	12"	16,800		12	208,694		12.23	10.99	23.22
			504 505	2011 1970	Steel Steel	Fixed Fixed	21' 16'	15' 22'	28,224 42,000		12	350,605 521,734		20.06 32.88	19.06 32.49	39.12 65.37
			512	1970	Steel	Fixed	16'	22'	42,000		12	521,734		32.88	32.49	65.37
SN-27k	Reclaimed Oil	N/A	514	2011	Steel	Fixed	24'	15'6"	31,500		12	391,301		24.48	22.08	46.56
			515	2011	Steel	Fixed	24'	15'6"	31,500		12	391,301		24.48	22.08	46.56
1			516	2011	Steel	Fixed	24'	15'6"	31,500		12	391,301		24.48	22.08	46.56
			517	2011	Steel	Fixed	24'	15'6"	31,500		12	391,301		24.48	22.08	46.56
			518	2007	Steel	Fixed	20'	10'	11,750	324,839	12	145,961	4,035,229	8.49	7.54	16.03
			500 (gun)	1970	Steel	Fixed	25'	12'	21,000		91	1,912,431		57.53	0.84	58.37
			509 (gun)	1997	Steel	Fixed	24'	5'5"	2,310		91	210,367		11.6	0.08	11.68
			513 (gun)	1970	Steel	Fixed	25'	12'	21,000	44,310	91	1,912,431	4,035,229	57.53	0.84	58.37
SN-27!	Treated Distillate	<0.1	121	2013	Steel	Fixed	52'	76'	1,680,000 -		8	13,109,664	الم وستمالينية	251.25	368.48	619.73
			122	2013	Steel	Fixed	52'	76'	1,680,000	3,360,000	8	13,109,664	26,219,328	251.25	368.48	619.73
}	Caustic	N/A	312	1980	Steel	Floating	18'	20'	42,000		NA					
	Caustic	N/A	400	1970	Steel	Fixed	18'	25'	82,992		NA		· · · · · · · · · · · · · · · · · · ·	<del>.</del>		<del></del>
[ <b>}</b>	Caustic Chemical	N/A N/A	402 922	1994 1970	Steel Steel	Fixed Fixed	16'6* 6'	18' 6'	14,994		NA NA			· · · · · · · · · · · · · · · · · · ·		
i }	Chemical	N/A N/A	922	2005		Fixed	6'	5'5"	1,260		NA NA	<del>   </del>		· · · · ·		
	Chemical	N/A	932	2005	Steel Steel	Fixed	14'	7'1"	1,000 4,000		NA NA					
· •	Chemical	N/A	933	2005	Plastic	Fixed	9'1"	5'	1,300		NA NA	<del></del>			<u> </u>	
<u> </u>	Chemical	N/A	934	2005	Plastic	Fixed	- 6'	5'5"	1,000		NA NA					
· •	Chemical	N/A	935	2005	Plastic	Fixed	6'	5'5"	1,000		NA NA				· · · · · · · · · · · · · · · · · ·	
0	Chemical	N/A	936	2005	Plastic	Fixed	6'	5'5"	1,000		NA NA					
Other Tanks as part of	Naphtha	0.54	130	1942	Steel	Fixed	54' length	12'	42,301		47	1,967,350				-
SN-27 that	Naphtha	0.54	131	1942	Steel	Fixed	54' length	12'	42,301		47	1,967,350				
do not have	Naphtha	0.54	132	1942	Steel	Fixed	54' length	12'	42,301		47	1,967,350	5,902,050		••	
emissions	NASH (Spent Caustic)	N/A	401	1970	Steel	Fixed	12'	34'	21,000		NA	-	<u>-</u>			
	NASH	N/A	313	1980	Steel	Fixed	18'	20'	42,000		NA					
ļ	NASH	N/A	316	1990	Steel	Fixed	18'	20	42,000		NA.		4 500 500			
	Heavy Condensate	3.83	133	1963	Steel	Fixed	95'6" length	12'	77,400		20	1,533,000	1,533,000			
	Water Water	N/A N/A	317 329	1990 1980	Steel Steel	Fixed Floating	18' 18'	20'	42,000 42,000		NA NA		<u></u>		<del></del> -	
-	Process Water	N/A N/A	506	1970	Steel	Fixed	24'	30'	126,000		NA NA					
<u> </u>	Process Water	N/A	507	1970	Steel	Fixed	16'	75'	500.010		NA NA	<del></del>		<del></del>	<del></del>	
ŀ	Process Water	N/A	508	2013	Steel	Fixed	42'	85'	1,782,690		NA NA		-			
	Process Water	N/A	510	2013	Steel	Fixed	42'	85'	1,782,690		NA NA			<del></del>		
-	.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		117 (bullet tank)	2011	Steel	horizontal	54' length	12'	42,301	42,301	473	20,000,000	20,000,000	1022.46	181.08	1203.54
			104	1992	Steel	Fixed	31'	111'	2,310,000 -	2,310,000	38	87,381,000		16577.07	2166.59	18743.66
SN-28	Crude Oil	0.79	110	1991	Steel	Fixed	30'	114'	2,310,000	2,310,000	7	15,330,000		3193.75	2341.19	5534.94
	i		111	1980	Steel	Fixed	32'	111'	2,310,000 -	2,310,000	7	15,330,000	118,041,000	3193.75	2341.19	5534.94
211.00	2		114	1995	Steel	Fixed	32'	72'	970,200	970,200	15	14,755,125		2972.36	611.19	
SN-29	Crude Oil	0.79	115	1997	Steel	Fixed	40'	110'	2,843,400 ·	2,843,400	5	14,755,125	29,510,250	2673.94	2110.32	4784.26
											Tota	Il Emissions (lb/yr):	SN-27a	Working Loss 128.79	Breathing Loss 315.71	Total Loss 444.50
				I	l		1	I					SN-27b	487.20	571.55	1,058.75
			I	1			L				1		SN-27c	3.44	11.88	15.32
				1				ļ				ļ	SN-27d	10.57	0.31	10.88
													SN-27e	11,737.40	895.14	12,632.54
											ļ		SN-27f	541.98	328.59	870,57 265,97
						1					ļ	ļ	SN-27g	189.41	76.56 43.09	307.63
						<b>∤</b> .		1	<b>.</b>			<b>.</b>	SN-27h	264.54	34.46	173.57
							1					· · · · · · · · · · · · · · · · · · ·	SN-27i	139.11	100,70	236.87
				ļ		+							SN-27j	136.17	234.81	609.82
i				ļ -									SN-27k SN-27l	375.01 502.50	736.96	1,239.46
								+					SN-271 SN-28	23,987.03	7.030.05	31,017.08
1				1	1			1			1	1	SN-29	5,646.30	2,721.51	8.367.81
1	;		·	<u> </u>	<u> </u>		<u> </u>		<u></u>		<u> </u>	<u></u>	- O14-52	3,040.30	4,141.01	3,507.51

## **CERTIFICATE OF SERVICE**

I, Pam Owen, hereby certify that a copy of this permit has been mailed by first class mail to
Martin Operating Partnership L.P., 484 East 6th Street, Smackover, AR, 71762, on this day of, 2014.
Pam Owen, AAII, Air Division